Diss. ETH No. 6413

# **Body-Mounted Antennas**

The Effect of the Human Body on the RF Transmission of Small Body-Mounted Biotelemetry- and Portable Radio Antennas in the Frequency Range 10-1000 MHz and Safety Considerations

A Dissertation submitted to the

SWISS FEDERAL INSTITUTE OF TECHNOLOGY ZURICH for the degree of Doctor of Technical Sciences

presented by

PETER A. NEUKOMM Dipl. El. Ing. ETH bom July 28, 1943 Citizen of Hallau, Schaffhausen

Accepted on the Recommendation of Prof. H. Baggenstos, referee Prof. Dr. E. Baumann, co-referee

> Juris Druck + Verlag Zurich 1979

# ISBN 3 260 04682 8

٠

## Preface

This work is an attempt to present a survey on the problems dealing with body-mounted antennas. The effect of the human body on the radiation patterns has been investigated by theoretical models and experiments. A polarization transformation effect has been discovered which leads to a new class of antennas for the resonance frequency range of man. The safety aspects have been investigated by studying the available literature on biological effects of radio- and microwaves.

The author wishes to thank all who have contributed to this work. Valuable scientific support came especially from :

- Prof. H.Baggenstos in electromagnetic theory (ETH, Inst. Electronics)
- Prof. R.F.Harrington in computer programs and theory (Syracuse Univ.)
- Dr. R.M.Bevensee in near-field analysis and computer programs (Lawrence Livermore Laboratory)
- Prof. 0.P.Gandhi in resonance absorption (University of Utah)
- Prof. A.W.Guy in phantom techniques (University of Seattle)
- Dr. Z.R.Glaser and Dr. D.L.Conover in safety aspects (NIOSH)

The experiments could be only performed by the technical assistance of:

- R.Graner and M.Knaute in antegna measurements (Division of Military aerodromes)
- W.Kerle in RF-instrumentation (PTT Switzerland)

I extend my thanks to the directors of the Laboratory of Biomechanics, ETH Zürich:

 Dr. B.M.Nigg and the late, much admired, Prof. J.Wartenweiler (ETH) for the general support of this work as a side branch of biomechanics.

Finally I appreciated the valuable comments and suggestions by

- Dr. J.Denoth (ETH, Laboratory of Biomechanics)
- Mrs. P.Fritz (Zürich)
- Prof. E.Baumann (ETH, Institute of Applied Physics)
- Prof. D.Kaufmann (University of Florida, visiting professor ETH)

and the drawing of the figures by Michael Fritz

#### P.A. Neukomm

to Vreni

ſοN	Ť	ENTS	
CON	۱	EW12	

PREI	FACE	
1.	INTRODUCTION	9
2.	PROJECT	13
3.	SYMBOLS AND DEFINITIONS	15
4.	SAFETY ASPECTS OF RADIO- AND MICROWAVES	23
	4.1. Introduction and Historical Background	23
	4.2. Electric and Magnetic Properties of Biological Materials	27
	4.3. Absorption of Electromagnetic Energy in Biological Material	29
	4.4. Observed Biological Effects of RF and MW	34
	4.5. High Frequency Fields from Electrically Small Antennas near a Body	38
	4.6. Current Trends in International Safety Standards Development	44
	4.7. Recommendations for Safety Limits for Body-Mounted Antennas	46
5.	ANALYSIS OF THE ANTENNA-BODY SYSTEM	49
	5.1. Description of the Antenna-Body Problem	49
	5.1.1. Definition of the Basic Goals in Antenna-Body Modelling	49
	5.1.2. Parameter description of the General Antenna- Body System	49
	5.2. Parameter Evaluation and Modelling of the Antenna-	50
	5.2.1. Transformation of the Problem with the	55
	Reciprocity Theorem	53
	5.2.2. Antenna Length and Field Homogeneity	55 50
	5.2.4. Influence of the Human Body's Material on	55
	the Scattered Field	61
	5.3. General Considerations on Antenna Measurements	63
	5.3.1. Antenna Measurements in Proximity to the Ground	63 64
	5.3.3. Field Homogeneity along the Body Axis at	04
	Vertical Polarization	68
	5.4. Antenna-Body Models for Computation and Experiment	71
	5.4.1. Body Models 5.4.2. Antonna Body Models for Computation	71 73
	5.4.3. Antenna-Body Models for Experiment	73
	•	

page

6.	FUND	AMENTAL THEORY FOR THE COMPUTATION OF SCATTERING FROM UCTING BODIES	75
	6.1.	Purpose of the Theory	75
	6.2.	Penetration Depth of the EM Field in Conducting Bodies	75
	6.3.	Charge- and Current Densities at the Surface of a Conducting Body 6.3.1. Boundary-Value Problem 6.3.2. The Effect of the Surface Current Density J	77 77 80
		6.3.4. The Field Outside of the Conducting Medium 6.3.5. Determination of the Current Density $\vec{J}$ and the Charge Density $\sigma$	81 82
	6.4.	Scattering from Bodies of Revolution with the Method	
		of Moments	84
		of Revolution	84
		6.4.2. Impedance Matrices	88
		6.4.2.1. Evaluation of the Impedances	88
		Computations of the Impedances	91
		6.4.3. Measurement Matrices	91
		6.4.4. General Plane-Wave Scattering	94
		6.4.5.1. Method of Solution	94
		6.4.5.2. Limitations of the Near-Field Computations	98
	6.5.	Scattering from Long Circular Cylinders: An Analytical	
		Approach	99
		6.5.1. Purpose of the Analytical Approach 6.5.2. Method of Solution	99
		6.5.3. Limitations of the Analytical Near-Field Computation	101
7	דע∩_ו	NIMENSIONAL COMPUTATION OF SCATTERING FROM AN INFINITE	
· •	CIRCI	JLAR CYLINDER	103
	7.1.	Computational Model and Goals	103
	7.2.	Computer Program PANA: Near-Field Pattern Computation of the	
		Infinite Cylinder IZYL	103
		7.2.1. Computational Formulas and Parameters	103
		7.2.2. Program Description PANA 7.2.3. Program Limitations and Accuracy	104
	73	Computed Results of the Two-Dimensional Model IZYL	108
	/	7.3.1. Azimuthal and Directive Radiation Patterns of Antenna-IZYL Model	108
		7.3.2. Minimum Gain depending on Frequency and Cylinder Radius	112
8.	MEASU	JRING METHOD	113
	8.1.	Purpose of the Experiment	113
	8.2.	Description of the Antenna-Body Test Set-Up	113

	8.3.	Antennas and Feeding d.3.1. Body-Mounted Antenna Aj 8.3.2. RF-Chokes 8.3.3. Remote Antenna A <sub>2</sub>	117 117 118 119
	8.4.	Measuring Equipment 8.4.1. Revolving Stage for Antenna-Body Rotation 8.4.2. Trackway for Antenna-Body Translation 8.4.3. Field Measuring Equipment	119 119 121 122
	8.5.	Antenna Set-Up Testing and Experimental Procedure	123
9.	COMP	RISON OF EXPERIMENTAL DATA WITH THEORETICAL DATA	127
	9.1.	Investigated Parameters 9.1.1. Effect of Frequency and Body Material 9.1.2. Effect of Antenna-Body Distance and Body Material 9.1.3. Effect of the Azimuthal Angle 9.1.4. Effect of the Body Material 9.1.5. Verification of the Reciprocity Theorem	127 127 129 133 134 135
	9.2.	Discussion of the Limitations of Experiment and Computation	135
10.	THREE OF RE	-DIMENSIONAL COMPUTATION OF SCATTERING FROM FINITE BODIES	137
	10.1.	Computational Models and Goals	137
	10.2.	Computer Programs for Near-Field Computations 10.2.1. General Overview 10.2.2. Parameter Description 10.2.3. Program Description HARRA 10.2.4. Program Description PANB 10.2.5. Program Description PANC	137 137 138 141 142 148
	10.3.	<pre>Investigation of Program Limitations and Computational Accuracy 10.3.1. Program Limitations 10.3.2. Computational Time Limitations 10.3.3. Storage Capacity Limitations 10.3.4. Investigation of Computational Accuracy 10.3.4.1. Minimum Mode Number KK 10.3.4.2. Difference between the two Azimuthal Field Components 10.3.4.3. Difference between Results at Different Test Segment Lengths 10.3.5. Field Homogeneity around a Near Field Point 10.3.5.1. Significance of the Field Homogeneity and Computational Data 10.3.5.2. Computational Data for Antenna Design</pre>	152 153 154 154 154 154 156 157 158 158 158
	10.4.	Results from Three-Dimensional Computations on Antenna- Body Models 10.4.1. Overview of Investigated Parameters and Explanations 10.4.2. Effect of the Frequency on Vertical and Radial Field 10.4.3. Effect of the Antenna-Body Distance 10.4.4. Effect of the Azimuthal Angle 10.4.5. Effect of the Irradiation Angle 10.4.6. Effect of the Relative Antenna Height 10.4.7. Effect of the Frequency on Azimuthal Radiation Pattern	163 164 163 167 168 169 170

		<ul> <li>10.4.8. Effect of the Frequency on Directive Radiation Pattern</li> <li>10.4.9. Effect of Different Body Shapes on the Fields in the Shadow Zone</li> <li>10.4.10. Effect of Different Body Shapes on Azimuthal Patterns</li> </ul>	174 178 182
п.	EXTEN	DED MEASURING METHOD FOR FIELD COMPONENTS SEPARATIONS	187
	11.1. 11.2. 11.3. 11.4.	Purpose of the Extended Experiments Antenna Manipulator Electrically Small Dipole Antennas with Built-In Oscillator Test Program and some Experimental Results Obtained with AO 1 to AO 4	187 188 189
12.	COMPAN COMPUT	RISON OF IMPROVED EXPERIMENTAL DATA WITH THREE-DIMENSIONAL TATIONAL DATA	195
	12.1.	Investigated Parameters	195
		<ul><li>12.1.1. Effect of the Frequency on the Field Components at FZYL and MET</li><li>12.1.2. Effect of Antenna Height and Proximity to the Ground</li><li>12.1.3. Effect of the Antenna-Body Distance</li></ul>	195 197 200
		<ul> <li>12.1.4. Effect of the Frequency on the Field Components at a Human Body</li> <li>12.1.5. Azimuthal Radiation Patterns of MET, SUB, FZYL</li> </ul>	202
	10.0	and MARMUUI & 2	203
	12.2.	Discussion of the Limitations of Experiment and computation	208
13.	CONCLU	JSIONS AND PERSPECTIVES	211
	13.1.	Important Investigated Parameters of the Antenna-Body System 13.1.1. Overview of the Investigated Antenna-Body System 13.1.2. The Effect of the Frequency on the Field Distribution 13.1.3. The Effect of the Antenna-Body Distance 13.1.4. The Dominant Rule of the Radial E-Field Component	211 211 212 213 213 214
	13.2.	Interesting Additional Features of the Antenna-Body System 13.2.1. The Broad-Band Characteristics of the Human Body 13.2.2. Body-Mounted Antenna Arrays	217 217 219
	13.3.	Proposals for Efficient, Body-Mounted Antennas 13.3.1. Vertical Polarized Antennas 13.3.2. Radial Polarized Antennas	220 220 220
	13.4.	The Optimal Frequency Range for Body-Mounted Antennas 13.4.1. Conclusions from the Obtained Data 13.4.2. Future Frequencies for Biotelemetry	223 223 224
14.	SUMMAR	RIES	
	14.1. 14.2.	Summary Zusammenfassung	225 227
15.	REFERE	INCES	229
16.	APPEND	XIX	235

### 1. INTRODUCTION

Biotelemetry is concerned with the obtaining and transmission of measuring data from a free-moving subject. Among other transmission methods such as infrared and ultrasonic it is utilized for all modulated radiowaves in the frequency range 10 to 1000 MHz.

In radio telemetry one distinguishes between tracking (radio bearing and identification of subjects) and real measuring data transmission. In contrast to voice communication systems a non-interrupted data flow is required from a moving subject, because the redundancy of the signals is small and the test situation happens only once in many cases.

Multichannel telemetry equipment for the continuous recording of physiological, chemical, biomechanical and other data are primarily applied on human subjects in patient monitoring, exercise physiology and sport research. The encumbrance for the subject due to weight and volume of the equipment and the feedback of the apparatus on the measuring data should be as little as possible. With today's technology it is possible to produce miniature transducers and transmitters. Missing are, however, small, body-mounted antennas with good omnidirectional properties.

A small, trunk-mounted, or even a non-visible, efficient antenna on the subject would open new fields of application not only for biotelemetry. Many applications of mobile voice communication for security personnel, police agents, etc., require a camouflaged antenna (GOUBAU and SCHWERING [32], KING [48]). Experience demonstrates however, that the transmission loss of small, body-mounted antennas amounts up to about 20-30 dB. This means that less than one percent of the RF energy can be utilized for transmission in unfavorable conditions. An enhancement of the power of the transmitter output for the improvement of transmission performance cannot be recommended for two reasons: In modern equipment the battery determines the final weight and volume of the transmitter, and RF-power in excess of 100 mW may exceed the safety limits for uncontrolled RF exposition (DDR-Standard [18], NEUKOMM [64]).

In the last few years various experimentalists attempted to quantify the influence of the human body on the radiation pattern of body-mounted antennas. Azimuthal radiation pattern measurements with horizontal polarized antennas were performed in the frequency range 6 to 280 MHz (BUCHANAN, MOORE and RICHTER [12]). Investigations with vertical polarized  $\lambda/2$  dipoles demonstrated a dominant, but not explicable influence of the antenna-body distance at 450 and 900 MHz (KING and WONG [49]). The fitting of relatively large antennas to the body was investigated in the frequency range 33 to 170 MHz by means of the VSWR, and a main resonance of the human body was postulated to occur at 60-80 MHz (KRUPKA [53]). As a result from these works one may conclude, that the human body acts as a director, reflector or absorber at frequencies above 30 MHz. In spite of considerable effort no systematic relationship was found between body geometry, antenna-body-distance, frequency and radiation pattern.

A first attempt to compute the radiation pattern of an antenna-body system came out in 1977 (NYQUIST, CHEN and GURU [66]). The model consisted of a short dipole antenna with assumed sinusoidal current distribution, parallel to a rectangular cylinder subdivided in dielectrical volume elements. By means of tensor Green's functions various results at 50 MHz were computed, such as power depositions in the body, impedance change of the antenna and also the azimuthal radiation pattern for some antenna locations.

Up to now a general theory about the radiation characteristics of an antenna-body system is missing. The reasons for that gap are mainly: - An antenna is a complicated radiation source. The fields around an antenna can be roughly categorized in near zone (r< $\lambda/2\pi$ ), far zone (r> $\lambda$ ) and transition zone. Within the near zone a strong reactive near field exists which is partially converted within the transition zone into an effective, real, radiating field. The final far field in the far zone is clearly described by the antenna parameter, as long as the near zone is not disturbed. But exactly this happens in the practical application of body-mounted antennas. Any antenna, especially an electrically small antenna, will be detuned by the body proximity. In spite of VSWR measurements one knows little about the radiated power and its radiation characteristics. If the antenna is combined with a fix transmitter (e.g., walkie-talkie) it is difficult to define a radiation reference level. If the antenna is remotely fed, surface waves on the feeding coaxial cables may radiate more than the antenna itself. Reflections from the ground effect a further, but estimable influence. In general, it is quite difficult to construct an antenna test set-up for antenna-body distances  $(d_{at})$  below 0.2 m and signal levels below - 15 dB (0 dB = antenna in free space) with a measuring error of less than 3 dB. Therefore, systematic effects could not be detected by experiments in the past.

- Most experiments have been performed at some fixed frequencies and with some fixed antenna-body distances where the future use of the transmitter was planned. A systematic relationship can be recognized only if these parameters are changed in little steps over a large range.
- The human body exhibits a complicated, variable shape. The dielectric inherent properties of the individual body organs vary around 1:10. In addition they are strongly dependent on the frequency, with a distinct change at about 100 MHz (JOHNSON and GUY [45]).
- The frequency range of interest covers the resonance region of the human body. The largest circumference of a body (human body : measured from head to feet) is roughly equal to the wavelength of the first resonant frequency. In fact this is demonstrated by absorption computation and thermographic investigations, where the maximum absorption occurs at about 65 MHz, if a human body model is irradiated by a plane wave (GANDHI, HAGMANN and D'ANDREA [24], CHEN and GURU [14].

From the literature neither analytical resolvable models nor approximative methods are available which explain the radiation characteristics of actual body-mounted antennas in the entire frequency range. The method of NYQUIST, CHEN and GURU [66] could lead to a systematic explanation, if the model could be improved by parameter variation. However, an extension of that method exceeds the limited storage capacity and the computation time limits of our ETH computer. Thus, other models and computation methods have to be found in order to understand the systematic correlations in a antenna-body system and in order to develop new, efficient antenna configurations.

# Leer - Vide - Empty

## 2. PROJECT

The fundamentals have to be prepared for the development of efficient, electrically small, body-mounted antennas with omnidirectional radiation patterns in the horizontal plane.

An antenna-body model has to be created which allows the computation of the systematic relation among frequency, body geometry, relative position of the antenna to the body and transmission loss. The model should be applicable for the entire frequency range from 10 to 1000 MHz.

A measuring method has to be developed which allows radiation pattern recording of body-mounted antennas in the entire frequency range from 10 to 1000 MHz. The measuring error should be less than 3 dB.

An investigation about the possible risks of body-mounted antennas has to be performed. The safety standard of some countries would prohibit the use of transmitters with sufficient power in combination with electrically small antennas. The international findings on biological effects of radio- and microwaves are controversial and the safety standards vary greatly from country to country. Biological effects may have an influence on the accuracy of biotelemetrical data and could lead to health hazards. The investigation should conclude in recommendations for reliable and safe use of transmitting devices with body-mounted antennas.

# Leer - Vide - Empty

## 3. SYMBOLS AND DEFINITIONS

SYMBOL	NAME AND DEFINITION	UNIT
a	radius	m
a <sub>s</sub>	radius of the radiansphere (4.5.) , $\lambda/21$	m
ā	spherical radius vector (5.2.1.)	}
ARP	azimuthal radiation pattern (7.2.1.)	
A	body-mounted antenna (5.1.2.) , usually electrically small (h< $\lambda/4$ )	
A <sub>2</sub>	remote antenna (5.1.2., 8.3.3.) , large broadband antenna	
Ā	magnetic vector potential (6.3.2.)	Vs/m
Ascat	scattered magnetic vector potential (6.3.2.)	Vs/m
В	bandwidth , $2\Delta f_{450}$ or frequency range between - 3 dB, (4.5.,16.1.)	MHz·
BK	wave propagation constant k in computer programs (10.2.2.)	1/m
BMR	basal metabolic rate (4.1.), metabolic power dissipation	W/kg
B	magnetic induction $\mu_{\mu\mu_0} H$	Vs/m²
с	velocity of light in vacuum, 2.9979.10° m/s	m/s
curl.	curl vector function, $\overrightarrow{\operatorname{curl}} \overrightarrow{E} = \overrightarrow{\nabla} \times \overrightarrow{E}$	
С	capacitance, 1 Farad = 1 Coulomb/Volt	As/V
C	central nervous system (4.4.)	
CW	continuous wave	
C <sub>max</sub>	maximum circumference of the body in wavelengths	1
ď	transmission distance $(d > \lambda)$ (5.1.2., 8.2.)	m
div	divergence vector function, div $\vec{E} = \vec{\nabla} \cdot \vec{E}$	
dat	antenna-body distance , distance from $A_1$ to body surface (5.1.2)	m
dg	thickness of the reflecting layer (5.3.2.)	m
dB	decibel, relative measure for power or field strength $(5.1.2.)$ power: 10 log $(P/P_0)$ , field: 20 log $(E/E_0)$	1
D	refracted wave (5.3.2.)	{
DRP	directive radiation pattern (7.2.1.)	1
D <sub>c</sub>	diameter of the infinite cylinder IZYL (5.4.1., 6.5.2.)	m
<sup>D</sup> h	diameter of the helical antenna (4.5., 16.1.1.)	m
D <sub>B</sub>	mean diameter of the trunk of a TS or test body (10.4.9.)	m
DTEST	1/4 of the test segment length in program PANB (6.4.5.1.,10.2.2.)	m

SYMBOL	NAME AND DEFINITION	UNIT
e	constant, e = 2.718	
EEG	electroencephalogram	
EM	electromagnetic (-wave)	
e <sub>o</sub>	reference electric field strength for 0 dB, received field strength at optimal antenna polarization in quasi-free-space condition (5.2.l)	V/m
E	horizontal component ( $\phi$ -component) of an E-field near a body (5.2.1.)	V/m ′
E	perpendicular (normal) component of an E-field to a surface (6.3.)	V/m
E	radial component of an E-field with respect to body axis (5.2.1.)	V/m
Et	tangential component of an E-field with respect to a surface (6.3.)	V/m
Ev	vertical component of an E-field near a body (   body axis) (5.2.1.)	V/m
Ezinc	incident electric field strength in z-direction	V/m
Ezind	induced electric field strength in z-direction	V/m
Ezscat	scattered electric field strength in z-direction	V/m
Eztot	total electric field strength in z-direction	V/m
Ē	electric field intensity, complex vector	V/m
É inc	general incident electric field	V/m
	general reflected electric field	V/m
<u>É</u> (a)	electric field at the relative position $\dot{a}$ to the body (5.2.1.)	V/m
Ēθ	$\theta$ - ('vertical') polarized incident E-field (6.4.3.)	V/m
Ē	$\phi$ - ('horizontal') polarized incident E-field (6.4.3.)	V/m
f	frequency in MHz, $f = \omega/21$	1/s
fres	resonant frequency (usually in MHz)	MHz
fliml	lower frequency limit due to Fresnel condition (5.3.1.)	MHz
f <sub>lim2</sub>	lower frequency limit due to Rayleigh criterion (5.3.2.)	MHz
f <sub>lim 3</sub>	upper frequency limit due to plane wave condition (5.3.3.)	MHZ
f <sub>lim 4</sub>	lower frequency limit due to far-field condition (5.3.3.)	MHz
f <sub>lim5</sub>	maximum computational frequency in program HARRA (10.3.1.)	MHz
F	frequency in MHz in all computer programs	MHz
FFHD	flat folded helical dipole (16.1.2.)	
FHD	flat helical dipole (16.1.2.)	
FSL	free-space level = $E_0 = 0  dB  (5.1.2.)$	dB

SYMBOL	NAME AND DEFINITION	UNIT
FZYL	finite cylinder, computational body model (5.4.1.)	+
9.	Green's function (6.4.2.1.)	
grad	gradient vector operator, $\overrightarrow{\text{grad}} \phi = \overrightarrow{\nabla} \phi$	1
G2	gain of the remote antenna $A_2$ (5.3.2.)	1
GA	ground plane antenna (16.1.2.)	
Gn	evaluated Green's function (6.4.2.1.)	
Gain <sub>B</sub>	transmission gain, gain of antenna $A_1$ when body-mounted with respect to the quasi-free-space performance (5.1.2.)	dB
h	hour	h
h	physical length of an electrically small (monopole) antenna (4.5.)	m
h	height of $A_1$ above ground (5.1.2.)	m
h <sub>2</sub>	height of $A_2$ above ground (5.1.2.)	m
<sup>h</sup> eff	effective height of an antenna (for U <sub>ind</sub> computation)	
h <sub>B</sub>	relative height of the center of $A_1$ with respect to the feet of the test body, z-coordinate of $A_1$ in computation (5.1.2.,10.2.5.)	m
HDR	heat development rate, usually relative value (4.3.)	1
н <sub>n</sub>	perpendicular (normal) component of a H-field to a surface (6.3.)	A/m
H <sub>t</sub>	tangential component of an H-field with respect to a surface (5.3.)	A/m
H <sub>x</sub> inc	incident magnetic field strength in x-direction	A/m
$H_x^{scat}$	scattered magnetic field strength in x-direction	A/m
$H_x^{tot}$	total magnetic field strength in x-direction	A/m
Ħ	magnetic field intensity, complex vector	A/m
Hinc	general incident magnetic field	A/m
Href	general reflected magnetic field	A/m
<u>H</u> n(kr)	Hankel function, second kind, order n (6.5.2.)	
I	current	A
j	square root of -1, imaginary number	1
j	electric current density in a volume element (6.3.4.)	A/m²
J <sub>t</sub>	tangential electric current density (6.3.1.)	A/m
J <sub>n</sub> (kr)	Bessel function, order n (6.5.2.)	
J J	electric current density on a surface (6.3.1.)	A/m
k <sub>m</sub>	amount of the wave propagation factor in a medium	1/m

SYMBOL	NAME AND DEFINITION	UNIT
ĸ	wave propagation factor $2\pi/\lambda$ in free space	1/m
кк	number of regarded modes n in computer program PANB	1
L	inductance, 1 Henry = 1 Vs/A	Vs/A
LPD	logarithmic periodic antenna (remote antenna A <sub>2</sub> ) (8.3.3.)	
LB	length of the TS or test body (10.4.9.)	m
Loss <sub>B</sub>	transmission loss caused by the body, (= - $Gain_{B}$ ) (5.1.2.)	dB
L(J)	integro-differential operator (6.4.1.)	
m	refraction index, m = k <sub>m</sub> /k (5.3.2.)	1
м	number of division of the interval O to $\P$ in computer programs	1
MANMOD	conducting human body model, sagittal or lateral view (5.4.1.)	
MET	metallic cylinder (for experiments) (5.4.1.)	ł
MHz	megahertz, 10 <sup>6</sup> Hz	1/s
MW	microwave, frequencies above 300 MHz	
n	mode number (in expansion functions) (6.4., 6.5., 10.3.1.)	1
N	number of tangential units along the contour curve of a body of revolution (6.4.)	1
NN	number of modes to be computed in program HARRA	1
NP	number of body contour points, $(NP-1)/2 = N$ , program HARRA and PANB	1
NPHI	number of division of the interval O to $\P$ in program HARRA and PANB	1
NNPHI	number of computed azimuthal field points in program PANB and PANC	1
NTEST	number of test segments in program PANB (10.2.4.)	1
Nh	number of turns of a helical monopole antenna (16.1.)	1
,, 1 p1	polarization of antenna $A_1$ , orientation of the main polarization axis in the space, vertical, radial or horizontal (5.1.2.,10.2.5.)	-
P2	polarization of antenna A <sub>2</sub> , vertical or horizontal (5.1.2.)	
P	real power in watt	VA
рна	phantom cylinder (for experiments) (5.4.1.)	
Pabs	absorbed power in a lossy medium, watt (4.3.)	VA
Pin	input power at a transmitting antenna, watt (5.3.2.)	VA
Ploss	dissipated power in a transmitting antenna, watt (4.5.)	VA
Prad	radiated power from a transmitting antenna, watt (4.5.)	VA
Preac	reactive power near a transmitting antenna (4.5.)	VA

SYMBOL	NAME AND DEFINITION	UNIT
Preal	real power (rad+loss) from and in a transmitting antenna (4.5.)	VA
Pscat	total scattered power from an object (5.2.3.)	VA
Þ	power density, Poynting vector $\vec{P} = \vec{E} \times \vec{H}$ , usually in mW/cm <sup>2</sup> (4.5.)	VA/m²
P <sub>real</sub>	real power density, computed from P <sub>rad</sub> only, mW/cm <sup>2</sup> (4.5.)	VA/m²
Preac	reactive power density, $ \vec{P}_{real}  \cdot Q$ , mVA/cm <sup>2</sup> (4.5.)	VA/m²
Q	Q-factor, ratio f <sub>res</sub> /B or stored energy/radiated+lossed energy in an RLC network (4.5.)	1
r	radius or distance	m
ř	radius vector from origin of coordinate system (6.3.2.)	m
r'	radius vector of a source point from origin of coordinate system	m
R	amount of the distance between source and observation point, $ \vec{R}  =  \vec{r} - \vec{r'} $ (6.3.2., 6.4.)	m
R	resistance, Ohm	V/A
RACS	relative absorption cross section (4.3.)	1
RCS	radar cross section, usually related to shadow area (5.2.3.)	m²
RF	radio frequencies, frequencies below 300 MHz	
RH	radius of a contour point in programs HARRA and PANB (10.2.2.)	m
RHD	round helical dipole (16.1.)	
RTEST	radius of the test segment center point (6.4.5.1., 10.2.2.)	m
Rloss	loss resistance of an antenna (4.5., 16.1.1.)	V/A
Rrad	radiation resistance of a transmitting antenna (4.5., 16.1.1.)	V/A
<u></u> Ε	reflection coefficient of the E-field of a TE-wave (5.2.4.)	1
<u>R</u> H	reflection coefficient of the H-field of a TM-wave (5.2.4.)	1
[R]	measurement row matrix (6.4.3.)	{ {
\$	distance from ground to feet of TS (5.1.2.)	m
S	surface of a body	m²
SAR	specific absorption rate, in $W/m^2$ or $W/kg$ (4.3.)	VA/g
SUB	standard human test subject (5.4.1.)	
t	time	s
t	index, means tangential (to a surface) component (6.3.)	
t <sub>1'2</sub>	tangential unit vectors on a surface of a body (6.3.5.)	1
TE	transversal electric mode, horizontal polar. (5.2.4., 5.3.2.)	

SYMBOL	NAME AND DEFINITION	UNIT
TEM	transversal electro-magnetic mode, general case (5.2.4.)	
м	transversal magnetic mode, vertical polarization (5.2.4., 5.3.2.)	
TS	test subject	
U	voltage	۷
U <sub>ind</sub>	induced voltage at the antenna terminals (5.2.2.)	v
v	volume of a body	m <sup>3</sup>
VSWR	voltage standing-wave ratio, ratio of $U_{max}^{\prime}/U_{min}$ (16.1.)	τ
V <sub>eff</sub>	effective volume of an antenna, computational value (4.5.)	m <sup>3</sup>
٧s	volume of the radiansphere = $\lambda^3/6 \ 1^2$ (4.5.)	m <sup>3</sup>
[V]	excitation matrix (6.4.1.)	
w,	testing function (6.4.1.)	
tY1	admittance matrix (6.4.1., 10.2.2.)	
Y <sub>n</sub> (kr)	Neumann function (Bessel function of second kind) (6.5.2.)	
z	Impedance	V/A
ZH	height of a contour point in programs HARRA and PANB (10.2.2.)	m
ZTEST	height of the test segment center point (6.4.5.1., 10.2.2.)	m
Zo	characteristic impedance of vacuum, $Z_0 = 377$ Ohm	V/A
Zm	characteristic impedance of a medium (5.3.2.)	V/A
[Z]	impedance matrix (6.4.1., 10.2.2.)	
α	incident angle with respect to a body surface (5.2.4.)	٩
β	phase factor (5.2.2.)	1
a <sub>n</sub>	Fourier coefficient (6.5.2.)	
Y	reflection angle (glancing angle) of a wave to the ground $(5.1.2.)$	•
Υ <sub>B</sub>	Brewster angle, total refraction of a TM-wave (5.1.2, 5.3.2.)	0
<u>r</u>	refraction coefficient of the air-medium interface at perpendicular (normal) wave incidence (4.2., 5.2.4.)	
δ	penetration depth of a wave into a medium (4.2., 5.2.4., 6.2.)	m
δE	maximum field strength variation along antenna axis (5.2.2.)	dB
δΦ	maximum phase variation of the field along antenna axis $(5.2.2.)$ ( $\delta E$ and $\delta \Phi$ are measurements for the field homogeneity, 10.3.5.)	0

.

SYMBOL	NAME AND DEFINITION	UNIT
ΔU	ratio actual induced voltage $U_{ind}$ to approximated induced voltage $\overline{U_{ind}}$ (from center field) in dB (5.2.2., mean error 10.2.5.)	dB
Δ <sub>s1</sub>	path difference for the first Fresnel Ellipsoid (5.3.1.)	m
∆ <sub>s2</sub>	path difference of a rough surface (Rayleigh criterion) (5.3.2.)	m
Δ <sub>s3</sub>	path difference of a plane wave along the body axis $(5.3.3.)$	m
Δ <sub>s3'</sub>	path difference of a spherical wave along body axis (5.3.3.)	m
Δ <sub>ψ2</sub>	phase difference due to the reflection of the wave at a rough surface, Rayleigh criterion : $\Delta_{\psi 2}$ < ¶/2 (5.3.2.)	0
ε	total permittivity	As/Vm
ε <sub>m</sub>	permittivity of a medium (total permittivity)	As/Vm
ε,	dielectric constant, $\epsilon_{ m o}$ = 8.854 · 10 <sup>-12</sup> F/m or As/Vm	As/Vm
٤ <sub>r</sub>	relative dielectric constant of a medium	1
θ <sub>e1</sub>	elevation angle of an incident wave above ground (5.1.2.)	0
θ <sub>d</sub>	refraction angle (5.3.2.)	0
θi	incident angle of a wave (to the body axis) (5.1.2.)	0
θŢ	angle of the test segment to the body axis (6.4.5.1.)	0
λ	wavelength	m ·
۲ <u>m</u>	wavelength of a wave in a medium (4.2.)	m
<b>ک</b>	wavelength of a wave in vacuum	m
μ	total permeability	¥s/Am
μm	permeability of a medium (total)	Vs/Am
μ	permeability of vacuum, $\mu_0 = 41 \cdot 10^{-7}$ H/m or Vs/Am	Vs/Am
μ <sub>r</sub>	relative permeability of a medium	1
p	radius of a point on the surface of the body of revolution (6.4.)	ព
۶m	distributed charges in a medium (charge density) (6.2.)	As/m³
σ, σ <sub>n</sub>	conductivity of a medium in S/m or mho/m (4.2.)	A/Vm
σ <sub>su</sub>	surface charge density on a body (6.3.)	As/m²
φ	azimuthal rotation angle (in the horizontal plane) (5.1.2.)	0
٥	electric potential (6.3.3.)	v
ψ	phase of the reflection coefficient (5.3.2.)	0
Ψ(z)	current distribution function along the antenna axis (5.2.2.)	1
ω	angular frequency, ω = 2¶f in rad/s	1/s

# Leer - Vide - Empty

## 4. SAFETY ASPECTS OF RADIO- AND MICROWAVES

### 4.1. INTRODUCTION AND HISTORICAL BACKGROUND

Body-mounted antennas are used in biotelemetry and walkie-talkies. Biotelemetry transmitters are applied on human test subjects and animals in order to record physiological and other data with minimum encumbrance for the test subject. Great efforts are made to reduce the influence of the measuring equipment on the data to be measured. In this chapter we are therefore not only interested in possible health hazards of radio transmitters but also in effects which may falsify the recorded data.

The subject is within the near-zone of a radiating source if body-mounted antennas are used. Due to the smallness of the antenna and due to the small antenna-body distance  $(d_{at})$ , the power density  $(\vec{P})$  at the subject's surface may exceed the maximum permissible values of some safety standards. As an example the consequent application of the German Democratic Safety Standard [18] prohibit such transmitting devices if the radiated power exceeds a few mW.

The non-ionizing electromagnetic (EM) spectrum encompasses frequencies from 1 Hz to 10<sup>19</sup> Hz. In general, frequencies from about 0.03 MHz to 300 MHz are called radio frequencies (RF) and frequencies from 300 MHz to 300,000 MHz are designated as microwaves (MW). In analogy to the well established safety standards for ionizing radiation the purpose of the less known safety standards for non-ionizing radiation is to protect a large population from uncontrolled exposure. A safety standard is always a compromise between absolute safety and practical realization. The permissible limits should exclude health hazards based on the present state of science. Under certain, well-described conditions the safety limits may be exceeded willingly if the risks resulting from other factors can be considerably reduced. For example, the application of a powerful ECG telemetry transmitter for the monitoring of heart disease patients is justified if the physician in charge considers a permanent heart monitoring as urgent.

A meaningful application of the safety standard requires the knowledge of the risks and often also the history of the standard's development. Until about 1945 "low-power" non-ionizing EM radiation was generally considered completely harmless. It was known that dielectrical materials can be heated internally with high RF power, an effect which is applied in diathermy for the clinical warming-up of certain body regions (MOOR [60], SCHWAN [73]). During World War II the U.S. Department of Defense medical services

became interested and concerned about possible hazards associated with the development, operation and maintenance of the increasing numbers of radar sets and other RF emitting electronic equipment. The main reason for that interest was reports about human microwave cataractogenesis (see MILROY and MICHAELSON [59]), HIRSCH and PARKER [44]) in radar repairmen who have been exposed to power densities in excess of 100 mW/cm<sup>2</sup>. After some investigations by the U.S. Navy and the U.S. Air Force, responsibility for research on biomedical aspects was delegated in 1957 to "Tri-Service-Program" directed by the USAF. This program, well described by MICHAELSON [58], included investigations on effects of exposure on whole-body, selective organs and tissues, single cells and enzyme systems, using various power levels, for pulsed and continuous waves in the frequency spectrum from 200 to 24,500 MHz. Basic work for the understanding of thermal effects was performed by SCHWAN and PIERSOL [74] on the field of power matching, absorption, penetration depth, etc.. Non-thermal effects such as field force effects on molecules (pearl chain formation, MUTH [61]), orientation of macromolecules (HELLER [43]), activation of membranes and neurons (e.g., LIVESHITS [55]), macromolecular resonance denaturation (e.g., BACH, LUZZIO and BROWNELL [4]) and many other effects have been investigated in that period. The listed non-thermal effects occurred only at high field intensities, so that the thermal effects were considered as dominant. A safe limit of 10 mW/cm<sup>2</sup> power density was defined which is still valid now (ANSI 1974 [2]) in the USA and in most of the western countries.

The power density number of  $10 \text{ mW/cm}^2$  has the following origin: the metabolic processes of the human body amount to about 1 W/kg when averaged over the total body mass for the sleeping state (BELDING and HATCH [8], GUY [35]). This Basal Metabolic Rate (BMR) for the whole body may be exceeded by that of individual organs; for example, the heart muscle has a metabolic rate of 33 W/kg, the brain 11 W/kg, the liver 6.7 W/kg, the skeletal muscle 0.7 W/kg (GUY [35]). As a fundamental, limiting criterion an artifical power deposition of 1 W/kg averaged over the whole body was defined. Such an external heating increases theoretically the head core temperature by about 0.15 °C and the body muscle temperature by about 1°C (EMERY et al. [22]). This heating is considered harmless since it is comparable with the heat produced by physical exercise. It was argued that since the BMR of the human body is in the order of 75 watts for a 70 kg man with a body surface of about 1.9 m<sup>2</sup>, this represents an equivalent areal heat production rate of about 4 mW/cm<sup>2</sup>. Since half of the surface area would be available for single sided exposure to a MW field, by limiting the maximum continuous MW exposure on 10 mW/cm<sup>2</sup>, one would expect no more than a doubling of the BMR in the body. (See limitations of this simplistic model in the next sections and TELL [77].) The philosophy of permissible heat loading concerns mainly the protection against destruction. Furthermore, clearly defined test conditions were choosen, e.g., in animal experiments an isolated test subject was exposed to a plane wave and great effort has been made to keep the field homogeneous. From the power density 10 mW/cm<sup>2</sup> the equivalent free space field intensities were derived for near-field conditions (E-field 200 V/m, H-field 0.5 A/m).

The Russian and generally the Eastern safety regulations are based on other considerations. The regulations concerning the power density are up to 10,000 times more stringent and the exposure duration is limited. As an example the GDR safety standard [18] demands that the power density in the MW region should not exceed 10  $\mu$ W/cm<sup>2</sup> at 8 h/day, and for the same exposure duration the E-field should not exceed 2 V/m in the frequency range 50 to 300 MHz. The H-field is not limited up to now, but since GUY [36] could prove that the H-field induced E-fields inside a human body are larger than the E-field induced E-fields in the frequency range 1 to 20 MHz, an appendix regarding permissible H-fields is to be expected. These very limiting safety standards are based mainly on effects studied in encephalography, biochemistry, cardiovascular pathophysiology etc. and are often connected with investigations in occupational medicine. Soviet investigators have stressed that the central nervous system is highly sensitive to all modes of radiation exposure. Their conceptional approach is based to a large extent on Pavlovian methods as can be seen from the many publications about changes in conditioned reflexes (GORDON, ROSCIN and BYCKOV [30]), see also summary of MICHAELSON [57]). Without discussing the details of the Eastern investigations it is evident that alterations of functions of complicated biological systems occur at much lower field intensities (nonthermal or microthermal effects) than material alterations or destruction. In the late 1960's the Eastern literature was reaching the American microwave community and initiated a broad research on lowlevel effects. One reason for this new interest was the introduction of microwave ovens in the USA, where the microwave leakage is in the range

from 0.7 to 20 mW/cm<sup>2</sup> (CONSUMER REPORTS [17]) in contrast to the reported effects at a power density level of 10  $\mu$ W/cm<sup>2</sup> to 5 mW/cm<sup>2</sup> (GORDON [30,31]). In 1971 GLASER [29] from the U.S. Electromagnetic Radiation Project Office began with a bibliography on reported effects and clinical manifestations attributed to MW and RF radiation. During this time the ninth supplement came out so that about 4,600 citations are available.

Some of the Eastern findings could not be reproduced in the West. Criticized were the insufficiently described test methods (see e.g. PROCEEDINGS WARSAW [69]). If more than one animal is kept in a cage for animal experiments, the field homogeneity may be disturbed by a factor of 100. Reflections from the cage walls and ground effects lead to enhanced absorption. If a ratisplaced in a reflector corner, an averaged incident power density of 10 mW/cm<sup>2</sup> may lead to an absorption of 200 W/kg (GANDHI [26,27]). On the other hand it must be pointed out, that under working conditions, reflections exist and that no long-term investigations have been performed with controlled conditions. The stringent Eastern safety standards which have been prepared above all for the protection of workers in factories are reasonable from this point of view.

In the following sections the physical background of RF and MW absorption and recent investigations on biological effects will be presented in order to estimate safety recommendations for body-mounted antennas.

## 4.2. ELECTRIC AND MAGNETIC PROPERTIES OF BIOLOGICAL MATERIALS

The investigations of the electric and magnetic properties were mainly performed during the last 30 years. Only the most important results can be discussed here; additional data and computation formulas can be found e.g., by TOLER and SEALS [79], SCHWAN and LI [75], JOHNSON and GUY [45]. Biological material is non-magnetic and can be characterized by their dielectric properties: conductivity  $\sigma$  and relative dielectric constant  $\varepsilon_r$ . The dielectric properties depend on the material and the frequency. In TABLE 1 these dielectric properties and the properties of the electromagnetic waves in the media are shown for two typical biological material groups:

PROPERTIES OF ELECTROMAGNETIC WAVES IN TWO GROUPS OF BIOLOGICAL MEDIA									
	GROUP A Muscle,Skin, and Tissue with High Water Content				GROUP B Fat,Bone, and Tissue with Low Water Content				
Frequency f [MHz]	40.7	100	300	915	40.7	100	300	915	
Wavelength in Air λ₀ [cm]	738	300	100	32.8	738	300	100	32.8	
Wavelength in Media λm [cm]	51.3	27.0	11.9	4.5	187	106	41.0	13.7	
Relative Diel. Constant ɛr [l]	97.3	71.7	54.0	51.0	14.6	7.5	5.7	5.6	
Conductivity σ [mmho/m] or σ [mS/m]	690	890	1370	1600	13-53	19-75	32-107	56-147	
Depth of Penetration δ [cm]	11.2	6.66	3.89	3.04	118	60.4	32.1	17.7	
Refl.Coeff. <u>r</u>   <u>r</u>   [1] arg( <u>r</u> ) [°]	0.91 +176	0.88 +175	0.83 +175	0.77 +177	0.62 +173	0.51 +168	0.44 +169	0.42 +173	

TABLE 1 Properties of electromagnetic waves in two groups of biological media. (Source: JOHNSON and GUY [45]).

The dielectric behavior of the two groups of biological materials listed in TABLE 1 has been evaluated most thoroughly by SCHWAN and his associates [73,74,75]. The biological tissues are composed of cells encapsulated by thin membranes containing an intracellular fluid. The increase of the conductivity  $\sigma$  and the decrease of the relative dielectric constant  $\varepsilon_r$  with increasing frequency can be explained for group A by the interfacial polarization across the cell membrane. The cell membranes, with a capacity of about  $1F/cm^2$ , act as insulating layers at low frequencies so that currents flow only in the extracellular medium. At higher frequencies the reactance decreases, resulting in increasing currents in the intracellular medium. The most noticeable change can be observed at about 100 MHz. A current density of about 1 mA/cm<sup>2</sup> produces a heat equal to that due to the BMR (SCHWAN [72]).

The depth of penetration  $\delta$  of RF and MW power into the material is defined as the distance required to reduce the power by  $e^2$ . The indicated values are valid only for a plane slab, an extended investigation of KRITIKOS and SCHWAN [52] on the distribution of heating potential inside lossy spheres has revealed hot spots in depths which may be larger than  $\delta$ . The hot spots appear inside only for spheres with radii from 0.1 to 8 cm and frequencies from 300 to 12,000 MHz and are of importance mainly in animal experiments (see e.g.,GUY [35]).

The reflection coefficient  $\underline{\Gamma}$  of the air-media interface will be important for the later computation of the scattering properties of a biological body. A  $\underline{\Gamma}$  of 0.88/+175<sup>0</sup> (group A, TABLE 1) means that the reflected wave will be only about 1 dB less than whose reflected from a perfect conductor and will show the same phase (dB: 20 log E/E<sub>0</sub>, E<sub>0</sub> reference E-field, E measured E-field, see chapter 5.1.2). For the computation of the absorption of electromagnetic energy in a multi-layer medium, also the reflection coefficients from one layer to the other are of importance as can be seen in the next section.

#### 4.3. ABSORPTION OF ELECTROMAGNETIC ENERGY IN BIOLOGICAL MATERIAL

The determination of the absorbed power in an arbitrarily shaped inhomogeneous biological medium needs a great computational effort. However, the multi-layer plane slab model is well investigated and may serve as a first approach for the absorption phenomena.

A simple plane slab model consists of two infinite layers of a certain thickness which are irradiated perpendicularly by a plane wave. (FIGURE 2: irradiation from the left, first layer = 3 cm fat, second layer = 10 cm muscle)



FIGURE 2 Relative Heat Development Rate (HDR) in a two-layer model. (Source: TELL [76], SCHWAN and PIERSOL [74])

The power which is absorbed in a volume element V and which is converted into heat is given by the formula (1):

$$P_{abs} = \frac{1}{2} \int_{V} \sigma |\vec{E}(\vec{r})|^2 dV$$
 (1)

The specific absorption rate (SAR) can be obtained by relating  $P_{abs}$  to the volume (W/cm<sup>3</sup>) or to the specific gravity (W/kg). The electric field strength  $\vec{E}(\vec{r})$  can be computed by the reflection coefficients and the attenuation factors in the two layers. Since the impedances of the materi-

als are complex, a feasible method of solution is to apply the Smith Chart as demonstrated by TELL [76].

The SAR leads directly to the Relative Heat Development Rate (HDR) by dividing the obtained SAR's by the maximum obtained SAR (FIGURE 2). The heat actually produced could be computed by applying the laws of thermodynamics, but since thermal data (heat conduction, external cooling, etc.) are difficult to obtain with the required accuracy, thermographic methods are better suited (GUY, WEBB and SORENSEN [36]).

In FIGURE 2 it is interesting to see that the HDR in the fat layer depends directly on the intrinsic wavelength  $\lambda_{\rm m}$ . At 915 MHz the fat layer is about  $\lambda_{\rm m}/4$  (TABLE 1) and acts therefore as a  $\lambda/4$  impedance transformer. The result is a high HDR at the irradiated surface and a sharp rise just inside the muscle layer. At 300 MHz the heating in the fat is much less than in the muscle, and at2,450 MHz the surface heating of the fat layer is about 68 percent of the maximum heating which occurs deeper within the fat.

This simple plane slab model is a good model for local application of a guided plane wave (diathermy applicators etc.), but does not adequately describe RF and MW absorption in complicated biological structures with irregular geometry, especially if the dimension of the body is comparable to the wavelength.

The absorption of EM energy in a three-dimensional body depends greatly on the body geometry and the wavelength. An adequate measure to describe this phenomenon is the Relative Absorption Cross-Section (RACS):

In FIGURE 3 a dielectric sphere with a radius a is shown which is irradiated by a plane wave (SCHWAN [72]). The RACS is defined as the ratio of absorbed power to the incident power. The incident power can be computed from the incident power density (in free space) multiplied by the shadow area  $\pi a^2$ ; the absorbed power can be computed by several methods or determined by thermal measurements. An RACS smaller than 1 means that a part of the incident power is reflected or transmitted through the sphere. An RACS greater than 1 means that the effective shadow area is greater than the physical area  $\pi a^2$  or, in other words, that EM power is extracted also from outer regions around the sphere.

Where are 3 different regions (FIGURE 3) : for small radius a (or for long wavelengths  $\lambda)$  the RACS is small, but increases rapidly with size.

- 30 -



FIGURE 3 Relative Absorption Cross-Section (RACS) of a sphere of tissuelike dielectric properties as a function of the relative frequency  $f_{rel}$  (sphere circumference 2% /wavelength  $\lambda$ ) (Source: SCHWAN [721]).

For large radius a (or short wavelengths  $\lambda$ ) the RACS is about 0.5 since the sphere reflects a part of the incident power. High RACS's occur, if the circumference of the sphere is almost equal to the wavelength. For  $2\pi/\lambda$  between 0.4 and 1.5 the RACS may exceed 1. In that resonant case not only is the absorption very high, but also the field homogeneity around the sphere is disturbed; an effect which will be important in the later field computation outside the irradiated body.

GANDHI et al.[25] continued RACS investigations with dielectric ellipsoids and various field polarizations. An ellipsoid with the axis a/b of 6.34 shows an RACS of about 4.2, if the incident E-field is polarized parallel to the main axis a and if the length L (2a) of the ellipsoid is about  $\lambda/2$ .

Specific Absorption Rates (SAR, see above) of body elements were determined by GANDHI, HAGMANN and D'ANDREA [24] and are shown in FIGURE 4: A salinefilled man model was irradiated in free space with a power density of 1  $mW/cm^2$ . The maximum averaged SAR for the whole body amounts to 0.2 W/kg and occur at 68 MHz for a model length of 1.75 m. The SAR of the leg and of the neck may reach 0.4 W/kg as can be seen from FIGURE 4.



FIGURE 4 Specific Absorption Rate (SAR) for a 1.75 m man model at an incident power density of  $1 \text{ mW/cm}^2$  in free space, vertical polarization. (Source: GANDHI, HAGMANN and D'ANDREA [24])



### FIGURE 5

The z-component (vertical) of induced electric field. Incident EM-Wave: vertical polarization, incident  $E_z^{inc}$  1 V/m, 80 MHz.

Compared with FIGURE 4 at an incident power density of 1 mW/cm<sup>2</sup> and 70 MHz the maximum SAR's occur at the knee (0.4-0.5 W/kg) and in the neck (0.2 W/kg). (Source: CHEN and GURU [15]) In FIGURE 5 the computed induced E-field components in the z-direction are shown in a body model irradiated by a vertical polarized EM-wave (CHEN and GURU [15]. The incident E-field is 1 V/m and the frequency 80 MHz. The highest induced E-field occurs in the knee region and is about 0.44 times the incident field. The computed SAR's agree with the measurement of GANDHI et al. [24,25]. High values are obtained mainly in the leg, the thigh and the neck. At horizontal polarization the largest SAR's occur at about 200 MHz and are located in the chest amounting to 0.4 W/kg at 1 mW<sup>2</sup> incident power density level (CHEN and GURU [15]). The dielectric properties  $\sigma$  and  $\varepsilon$  in experiment and computation are similar to those of group A in TABLE 1. If the computer capacity is available, they could be varied for each cube for future refinement with more cubes.

Up to here the human body was considered to be in free space. The effects of the presence of reflecting surfaces and ground effects were studied by GANDHI et al. [24,26,27]. The SAR's for the whole body and some intact anatomical parts of a man for an incident power density of  $lmW/cm^2$  is shown in FIGUR 6, where man is in good electrical contact with a high conducting ground plane:



FIGURE 6 Specific Absorption Rate (SAR) for a 1.75 m man model in good electrical contact with a high conducting ground plane. The incident power density is 1 mW/cm<sup>2</sup>. (Source: GANDHI, HAGMANN and D'ANDREA [24])

The resonant frequency is now near 35 MHz, and the SAR's are about twice as high than in the ungrounded condition. The SAR of the whole body amounts to about 0.3 W/kg, the SAR of the leg to about 1 W/kg at an incident power density of 1 mW/cm<sup>2</sup>. The indicated SAR's are those integrated over basic anatomical structures and do not reveal the worst case. A further increase of the SAR in the whole body can be observed when the man is placed in front of a flat reflector (1W/kg), in a 90° corner reflector (6W/kg) and in a corner reflector with ground contact (12W/kg) (all values related to a power density of 1 mW/cm<sup>2</sup>).

In the introduction it has been mentioned that the fundamental, limiting criterion was a power deposition of 1 W/kg which is equal to the BMR. At frequencies below 20 MHz and above 300 MHz the present U.S. safety standard of 10 mW/cm<sup>2</sup> fulfills this criterion. However, a more stringent safety standard seems to be reasonable for the resonance frequency range 20 to 300 MHz.

#### 4.4. OBSERVED BIOLOGICAL EFFECTS OF RF AND MW

Although some thousand recent investigations on biological effects are available (see bibliography of GLASER et al. [29]), the effects at lowpower densities are not yet understood in a larger context. Most of the experiments were carried out with small animals at frequencies above 300 MHz, therefore the results of such investigations cannot be transferred directly onto large animals or humans. Some few examples should give an overview on the variety of the documented effects.

The teratogenic effects of MW in insects were studied by LIU, ROSENBAUM and PICKARD [56] by irradiating the pupae of the darkling beetle *Tenebrio Molitor* during its metamorphosis. A statistically significant increase in malformations in the adult insect was observed at power levels as low as 170  $\mu$ W/cm<sup>2</sup>. The pupation time increased monotonically with the power density at a constant (2 h) irradiation duration. The damages increased linearly with the logarithm of the dosage, and the effects started at approximately 40  $\mu$ W/cm<sup>2</sup> power density and 0.1 mWh/cm<sup>2</sup> energy density. Exposure of various durations (max. 16 h) and powers (max. 16 mW/cm<sup>2</sup>) strongly suggested that it is the total dosage which determines the level of teratological damage. Since irradiation at 16 mW/cm<sup>2</sup> is known to produce a

- 34 -

measured rise in pupal temperature of less than  $2^{\circ}$  C, and since heating by conventional thermal techniques appears not to be teratogenic, the effects seem to be not (macro-) thermal in origin.

A widely observed and accepted biological effect of low-average power EM energy is the auditory sensation evoked in an exposure to MW. Among other researchers GUY et al. [37] describe the effect as an audible clicking or buzzing sensation that originates from within and near the back of the head and that corresponds in frequency to the recurrence rate of the MW pulses. The loudness of the sensation correlates with the average incident power density. The threshold energy density per pulse is about 40  $\mu$ J/cm<sup>2</sup> (corresponds to about 0.01  $\mu$ Wh/cm<sup>2</sup>) and is five order of magnitudes smaller than the permissible U.S. safety standard value of 1 mWh/cm<sup>2</sup> for peak power averaged over any 6-minute period (ANSI [2]). However, it should be mentioned that the average power density for the threshold of  $120 \mu W/cm^2$  (about two order of magnitudes lower than the permissible U.S. safety standard value) requires a pulse width of 1 to 32 us, with peak power from 1.25 to 40 W/cm<sup>2</sup>. The presented data are valid for 2,450 MHz for humans. Experiments at 918 MHz with cats have shown that depending on the pulse width (3 to 32  $\mu$ s) average energy densities of 17 to 28  $\mu$ J/cm<sup>2</sup> per pulse, average densities of 17 to 28  $\mu$ W/cm<sup>2</sup> and peak power density of 0.8 to 5.8  $W/cm^2$  are required to produce the auditory effects. Although an energy density of 40  $\mu$ J/cm<sup>2</sup> is capable of increasing the tissue temperature by only 5.10<sup>-6</sup> <sup>O</sup>C, the auditory effect could be explained by microthermal expansion of the liquid in the cochlea, producing a pressure wave similar to the normal input of acoustic signals.

Microwave-induced chronotropic effects in the isolated rat heart are described in a recent report by OLSEN, LORDS and DURNEY [68]. Continuous (CW) MW irradiation at 960 MHz causes bradycardía (lowered heart rate) in isolated, perfused rat heart maintained at 20  $^{\circ}$ C. The observed bradycardía occured at a power deposition of 1.3 to 2.2 W/kg that should have caused mild tachycardía (increased heart rate) based on the theromogenic properties of the irradiation. The observed bradycardía,moreover, exhibits neurologic features, because atropinized hearts showed strong tachycardía during irradiation, and hearts treated with propranolol showed significantly stronger bradycardia during irradiation than seen without drugs. It is assumed tha MW interacts with the autonomic nervous system by changing the neurotransmitter release mechanism. Because the temperature rise was limited to 0.1 <sup>o</sup>C, macrothermal mechanisms are not possible, but the possibility exists, that microheating, i.e., strong thermal gradients over small regions could be responsible for this chronotropic effect of MW. Similar effects, but at lower SAR in living rats, are reported by East European researchers [28,29,69].

A considerable body of literature has grown in the East European countries on transient functional changes following low dose RF and MW irradiation. A sample of clinical and experimental data is presented in TABLE 7.

A SAMPLING OF THE GENERAL BIOLOGICAL EFFECTS OF MICROWAVES AT POWER DENSITIES OF 10 mW/cm <sup>2</sup> OR LESS (EAST EUROPEAN SOURCES)							
Clinical Effects	Experimental Effects						
I. General subjective complaints (sensations,fatigue,loss of ap- petite,asthenia, etc.)	<ol> <li>Decreased physical endurance and re- tarded weight gain (rats).</li> </ol>						
II. Functional CNS and perceptual changes.	II. General inactivation of CNS electri- cal activity; domination of hypo- thalamic function; altered afferent function (rabbits, cats).						
	Inhibition of conditioned reflexes; increased motor activity; weakening of excitation/inhibition reactions (rats,mice,birds).						
	Morphological changes in nervous sys- tems (rats,guinea pigs,rabbits)						
	Altered reactivity in response to drugs (rats,rabbits).						
III. Cardiovascular and associated autonomic changes.	III. Altered blood pressure and heart rate (rats,rabbits).						
IV. Altered blood chemistry.	IV. Altered blood neuroendocrine chemi- stry (rats,rabbits).						
V. Altered metabolism.	V. Altered amino acid and ascorbic acid metabolism (rats)						
VI. Depressed endocrine function.	VI. Altered reproductive cycle; decreas- ed viability of offspring (rats).						
VII. Increased susceptibility to infectious diseases.	VII. Altered immune reactions (rabbits).						

TABLE 7 A sampling of the general biological effects of MW power densities of 10 mW/cm<sup>2</sup> or less as reported by Soviet, Czechoslovakian and Polish researchers. (Source: GLASER and DODGE [28,19])
In the Warsaw Proceedings "Biological Effects and Health Hazards of MW Radiation" [69] and in the recent book by BARANSKI and CZERSKI [6] a review of the East European research is presented. GORDON, ROSCIN and BYCKOV [30] describe functional disturbances in the Central Nervous System (CNS), physiological alterations and behavioral changes which occur at power levels down to a few  $\mu W/cm^2$ . Various low-level effects may be considered as selective absorption of radiation at the interfaces of heterogeneous biological systems, e.g., hypothalamic-hypophyseal-suprarenal system. Electrophysical investigations of isolated nerves and muscle fibers in frogs at 5  $\mu$ W/cm<sup>2</sup> have revealed slowed conduction of impulses, an increased synaptic delay, a lengthening of latent and refractionary periods and changes in action potentials. DUMANSKIJ and SANDALA [20] investigated alterations in the EEG, in conditioned reflex activity (longer latent period, weakend reaction to positive stimuli) and in several metabolic processes in rats and rabbits after irradiation with less than 10  $\mu\text{W/cm}^2$  at 50 MHz and 12 h/ day exposition. KALADA, FUKOLOVA and GONCAROVA [46] and others [69] demonstrated effects in occupational exposure. The effects are manifested by weakness, fatigue, headache, etc. and dysfunctions in the autonomic nervous system, which are apparently reversible.

As pointed out by many Western researchers some of the Eastern findings could not be reproduced in the West at the same low-power density level (see e.g.,CHOU and GUY [16] and ROMERO-SIERRA, HALTER and TANNER [70]). However, there is an increasing number of investigations in the West which lead now to similar results (EEG-changes, altered conditioned reflexes, behavioral changes, pathological changes in nerve tissue and brain, increased sensitivity to drugs, etc.), and it has been well established that certain birds, fish and invertebrates can exhibit sensitivity at very weak fields of all kinds (see e.g.,discussion by DODGE and GLASER [19]). Very little is known about RF- and MW receptors, the effect of irradiation on children and non-healthy persons, and the significance of long-term irradiation.

- 37 -

# 4.5. HIGH FREQUENCY FIELDS FROM ELECTRICALLY SMALL ANTENNAS NEAR A BODY

The purpose of this section is to estimate the quantities of the E- and Hfields on the surface of a subject in close contact with a transmitting antenna and to compare these quantities with the safety standards.

Let us consider a small  $(2h < \lambda)$ , see FIGURE 8) dipole antenna  $A_1$  radiating an RF power  $P_{rad}$ . In a large distance r  $(r \gg \lambda)$  from the antenna one may assume that the propagating wave is plane, so that the E- and H-vectors are rectangular to each other and show the same phase. The amount of the vector power density  $\vec{P}$  and the amounts of the vectors  $\vec{E}$  and  $\vec{H}$  can be computed from  $P_{rad}$ , r and the characteristic impedance of vacuum  $Z_0$ :

$$|\vec{P}| = |P_{rad}| / 4 \cdot r^2 \cdot \Pi$$
 (2)

$$\vec{P} = \vec{E} \times \vec{H}$$
 (Definition Poynting) (3)

$$Z_0 = (\mu_0/\varepsilon_0)^{\frac{1}{2}} = |\vec{E}| / |\vec{H}|$$
(4)

$$\vec{E}| = (|\vec{P}| \cdot Z_0)^{V_2}$$
 (5)

$$\vec{H}$$
 =  $(|\vec{P}| / Z_0)^{V_2}$ 

In the vicinity of an actual antenna  $(r<\lambda)$  the  $\vec{E}$ - and  $\vec{H}$ -vectors are neither rectangular to each other nor in phase.  $\vec{P}$  becomes a rotating vector of variable amount, and the time averaged power density  $|\vec{P}|$  is

$$|\vec{P}| = \frac{V_2}{Re(\vec{E} \times \vec{H}^*)}$$
 (6)

The total power density  $\vec{P}$  can be considered to be a superposition of a real power density  $\vec{P}_{real}$  and a reactive power density  $\vec{P}_{reac}$ . The energy associated with the reactive power  $P_{reac}$  pulses back and forth and represents stored energy (similar to the energy stored in an inductor or capacitor). The energy flow associated with the real power  $P_{real}$  is always positive in direction of propagation and represents a real energy flow. For the following estimation we define  $\vec{P}_{real}$  as the power density which is produced from a 'hypothetical point source' with the radiating power  $P_{rad}$ :

$$|P_{real}| = |P_{rad}| / 4 \cdot r^2 \cdot \P$$
 (8)

An antenna can be considered as a resonator for the nominal frequency  $f_{res}$  which loses energy by radiation. In the vicinity of the antenna exists a large reactive power which is converted into radiating (real) power, and at about  $r = \lambda/21$  the radiating power dominates over the rapidly decreasing reactive power.

An electrically small antenna is defined as an aerial, one whose size is a small fraction of the wavelength. It is a capacitor or inductor, and is tuned to resonance by a reactor of opposite kind (WHEELER [83]). From this definition it is evident that an electrically small antenna will show a considerable amount of reactive power.

The 'Helical Normal-Mode Antenna' is today one of the most applied type of electrically small antennas for walkie-talkies and biotelemetry transmitters. The dipole version (see FIGURE 8) of the helical antenna consists of a helical conductor in the shape of a long cylinder with the diameter  $D_h(D_h \ll \lambda)$  and with the axial length 2h ( $2h < \lambda/2$ ). The computation of the helical antenna and its features will be discussed in chapter 16.1. At the moment we have to know only, that the main radiation direction is radial to the axis and that the main polarization axis is parallel to the antenna axis (similar to a full-size dipole antenna).

With the theory of WHEELER [83,84,85] the ratio of real to reactive power can be computed in a situation as shown in FIGURE 8.



# FIGURE 8

Model for the estimation of the nearfield reactive power

- A<sub>1</sub> : helical dipole antenna
- 2h : physical antenna length  $(<\lambda)$
- $D_h$  : diameter of the helical coil (<< $\lambda$ )
- $a_s$ : radius of the radiansphere (see definition in the text,  $a_s = \lambda / 2$ ¶)
- $\lambda$  : wavelength
  - : point on the surface of the body, located in a distance h from the antenna

The 'radiansphere' is defined by WHEELER [84] as the boundary between the near field and the far field of a small antenna. Its circumference is  $\lambda$ , and the radius  $a_s$  is one radianlength ( $\lambda/2\pi$ ), at which distance the three terms of the field (from R,L and C of the antenna impedance) are equal in

P

magnitude. The volume V of the radiansphere is:

$$V_{\rm S} = \frac{4\pi}{3} \left( \frac{\lambda}{2\pi} \right)^3 = \frac{\lambda^3}{6\pi^2}$$
(9)

An electrically small antenna is somewhat smaller than the radiansphere, but it has a sphere of influence occupying the radiansphere. From the computation of radiation power factor an effective volume V<sub>eff</sub> has been defined (WHEELER [85]) which is very roughly a value between the physical volume of the antenna and the volume of a sphere containing the antenna:

$$\frac{\text{1Dh}^2}{4} \cdot 2h \ll V_{\text{eff}} < \frac{41}{3} \cdot h^3$$
 (10)

The effective volume of a slender helical antenna is about  $2/3 \cdot h^3 \cdot \P$ . The ratio between radiating power  $P_{rad}$  to reactive power  $P_{reac}$  is given by the ratio  $V_{eff}$  to  $V_s$  as discovered by WHEELER [83,85].

$$\frac{P_{reac}}{P_{rad}} = 4.5 \frac{V_s}{V_{eff}}$$
(11)

An antenna can also be considered as a resonant R-L-C network. The socalled Q-factor of such a network is defined by the ratio of the resonant frequency  $f_{res}$  to the bandwidth B and results from the ratio of stored power (in L and C) to real power (in R). The real power is the sum of the radiated power (in the radiation resistance  $R_{rad}$ ) and the dissipated power  $P_{loss}$  (in the loss resistances  $R_{loss}$ ):

$$Q = \frac{f_{res}}{B} = \frac{stored \ power}{real \ power} = \frac{P_{reac}}{P_{rad} + P_{loss}}$$
(12)

By combining equation (11) and (12) we obtain for the lossless antenna:

$$\frac{f_{res}}{B} = 4.5 \frac{V_s}{V_{eff}} = \frac{P_{reac}}{P_{rad}}$$
(13)

Equation (13) leads to the following interesting conclusions:

- By decreasing the size of a distinct antenna type the bandwidth decreases considerably, if the resonant frequency is kept constant. This law (WHEELER [83]) is often not noticed in practical antennas, because the radiation resistance decreases and the loss resistance increases.
- 2. By decreasing the size of a distinct antenna, the reactive power in-

creases considerably, if the radiated power and the resonant frequency are kept constant. The Q of an actual antenna is about 5 to 30, so that the electrically small antenna represents a strong, concentrated reactive power source.

With the assumption that the total real and reactive power is contained in a sphere of radius h (FIGURE 8) and is distributed homogeneously, the averaged quantities of  $\vec{P}_{real}$ ,  $\vec{P}_{reac}$ ,  $\vec{H}$  and  $\vec{E}$  at the subject's surface point P can be estimated as follows:

a) The radiated power  $P_{rad}$  can be computed from the electrical fieldstrength  $E_V$  in the far-field at the distance r: (BECKER [7])

$$P_{rad} \stackrel{\sim}{=} \frac{E_{v}^{2} \cdot r^{2}}{45 \Omega} ; E_{v} \text{ in mV/m, r in m, } P_{rad} \text{ in } \mu W \qquad (14)$$

b) The real power density  $\vec{P}_{real}$  originating from an assumed point source with the real radiating power  $P_{rad}$  is with equation (8) :

$$|P_{real}| = P_{rad} / 4 \pi^2$$
(15)

c) The Q-factor can be obtained by measuring the resonant frequency  $f_{res}$  and the - 3 dB bandwidth B. For the lossless antenna we obtain the reactive power density  $\vec{P}_{reac}$  with equation (12) :

$$|\vec{P}_{reac}| = |\vec{P}_{rea1}| \cdot Q \tag{16}$$

For an antenna with high losses it is recommended to determine the losses with the efficiency measuring method or to compute the theoretical Q (see Appendix 16.1.)

 d) If we assume that the total reactive power is stored magnetically, the H-field component is about

$$|\vec{H}| \cong Q \cdot (|\vec{P}_{rea1}|/Z_0)^{V_2}$$
(17)

and if we assume that the total reactive power is stored electrically, the E-field component is about

$$|\vec{E}| \cong Q \cdot (|\vec{P}_{real}| \cdot Z_0)^{V_2}$$
(18)

The obtained results agree with actual measurements of helical antennas (TELL and O'BRIEN [78]) within a factor of 2. For generally small antennas an error factor of about 5 is to be expected, which is acceptable, because the threshold for biological effects is very variable. Two typical examples should illustrate the significance of radiation of body-mounted antennas with respect to safety:

<u>Mobile communication systems.</u> Security personnel, police, trafficcontrol agents, the crew on railroad yards and many other groups are equipped more and more with body-mounted transmitters. The position of the antenna during transmission is close to the hip, chest or head, the standard power is 1 to 5 watts; the standard frequencies are about 170, 450 and recently also 900 MHz. For a 450 MHz walkietalkie the general specifications are as follows: monopole antenna with a length of h=4 cm, antenna-body distance  $d_{at}$ =4 cm (situation as depicted in FIGURE 8), input power 5 W, antenna efficiency 50 percent and bandwidth 10 percent. From these data we compute a radiated power  $P_{rad} = 2.5$  W and a Q-factor of 10. The radiation intensities on the surface of the body computed with equation (15) to (18) are: real power density  $\vec{P}_{real} = 10$  mW/cm<sup>2</sup>, reactive power density  $\vec{P}_{reac} = 100$  mVA /cm<sup>2</sup> and maximum possible  $\vec{E}$  or  $\vec{H} = 2,000$  V/m or 5 A/m.

These intensities are comparable to the measurements of TELL and O'BRIEN [78] at a 3.8 W/450 MHz walkie-talkie equipped with a 15 cm helical antenna. At a distance of 5 cm a maximum power density of 24  $mW/cm^2$  was measured with an E-field probe (EDM-3 from NBS, see e.g., [9]).

If we assume a daily transmission duration of 20 minutes, the radiation of standard professional walkie-talkies exceeds the U.S. safety standard by a factor 1 to 10 and the East-European safety standard by a factor of 100 to 1,000. Macro-thermal effects are not to be expected, because the small irradiated area is well-cooled, but microthermal or non-thermal effects probably occur. The main risk is not only the high intensities, but the uncontrolled, frequent, world-wide application of walkie-talkies. The actual Polish regulation (see ref. 461 in BARANSKI and CZERSKI [6]) requires that any candidate for work necessitating exposure to MW must undergo a medical examination and obtain a medical certificate for fitness, and periodic examination of MW workers are compulsory. It would be wise to collect medical data on personnel equipped with mobile communication systems in order to decide if similar examinations are necessary for such personnel. <u>Miniature biotelemetry transmitters</u>. A common antenna for biotelemetry transmitters is a small coil around the housing. We assume the following data: RF-input power 1 mW, antenna diameters 2h = 2 cm, radiation efficiency 10 percent and a Q-factor of 30. With that data the real power density  $\vec{P}_{real}$  is about 0.01 mW/cm<sup>2</sup> and the reactive power density  $\vec{P}_{reac}$  is about 0.30 mVA/cm<sup>2</sup>.

The radiation intensities of miniature biotelemetry transmitters are between the safety standard limits of the U.S. and East Europe. The radiation duration is generally a few weeks, and often electrodes to sensitive body regions (EEG) are implanted . Often pulse position modulation (sharp peak power) is applied in animal experiments. Health hazards are notlikely to occur , but micro-thermal gradients may cause biological effects which may lead to wrong physiological measuring data. Therefore, it would be wise to check the probable influence of the biotelemetry transmitter radiation with respect to artifacts.

The accurate computation of the E- and H-fields near an antenna and near or in the body is complicated and vary from one antenna type to the other. Near-field results of a slender monopole antenna have been presented by CHANG, HALBGEWACHS and HARRISON [13]. NYQUIST, CHEN and GURU [66] investigated the coupling of a 50 MHz  $\lambda/4$  dipole antenna with a man-model consisting of dielectrical cubes. At an antenna-body distance of 10 cm they computed a total power deposition of 0.28W at an input power of 3.14W. They concluded that an input power of 20W results in potentially hazardous intensities comparable to a plane-wave irradiation with 10mW/cm<sup>2</sup>. It should be mentioned that a  $\lambda/4$  antenna is 1.5m long so that the critical power level may be expected below 2W for electrically small antennas.

Accurate near-field measurements in and outside the boby still pose a problem. EGGERT, GOLTZ and KUPFER [21] developed the near-field strength meter NFM-1 which allows E-field measurements 10 cm away from a source with less than 15 percent error in the frequency range 10 to 350 MHz. BELSHER [9] developed the near-field electric energy density meter EDM-2 with a E-probe consisting of three orthogonal miniature dipoles. The probe is imbedded in a 2 cm Østick and allows E-field measurements with less than 10 percent error in the frequency range 10 to 500 MHz. GREENE [33,34] described an H-field probe and a near-field exposure synthesizer for the frequency range 10 to 40 MHz. The best method to determine the field <u>inside</u> a body is the thermographic recording of the absorption in a model (e.g., GUY,WEBB and SORENSEN [36]). An overview on the present safety standards and on the actual existing exposures is given in FIGURE 9:



FIGURE 9 International safety standards and actual exposures. Some standards are related gradually to the duration, all are valid for partial and whole body exposure. The GDR limit is valid for pregnant and nursing women. The urban environmental exposure regards approximately 20,000 people in Washington and Chicago. Sources: MICHAELSON [57], DODGE and GLASER [19], ANSI [2], TGL [18], HANKIN et al. [39] and NEUKOMM [64].

The U.S. Safety Standard recommendations (ANSI [2]) apply to all radiation within the frequency range from 10 MHz to 100 GHz except for deliberate exposure of patients by or under the direction of practitioners of the healing arts. The recommendations pertain to both whole body and partial body irradiation. For normal environmental conditions the CW (continuous wave) radiation guide is 10 mW/cm<sup>2</sup>, and the equivalent free-space electric and magnetic field strengths are approximately 200 V/m RMS and 0.5 A/m. For modulated fields, the power densities and the field strengths are averaged over any 0.1 hour period, and they should also not exceed an energy density of 1 mWh/cm<sup>2</sup>. The US Army (MICHAELSON [57] recommends further, that short exposures should not exceed 100 mW/cm<sup>2</sup>, and that the exposure duration (in minutes) is limited by the expression 6000/((x mW/cm<sup>2</sup>)<sup>2</sup>). Sweden decreased step by step the maximum permissible power density from 10 mW/cm<sup>2</sup> (1970) over 5 mW/cm<sup>2</sup> (1973) to now 1 mW/cm<sup>2</sup>, and Canada intends to follow (DODGE and GLASER [19]).

The East European safety standards in FIGURE 9 apply for all occupational radiation in the frequency range 300 MHz to 300 GHz. Remarkable are the stepped curve (constant dosis) and the low values for permanent exposure. At lower frequencies somewhat higher values are permissible. A typical example for Eastern safety standards is presented in TABLE 10 with the German Democratic Republic's (GDR) safety standard:

FREQUENCY	MAXIMUM PERMISSIBLE FIELD INTENSITIES IN THE GERMAN DEMOCRATIC REPUBLIC (1978)						
RANGE	OCCUPATIONAL EXPOSURE TO IRRADIATION GENERAL PREGNANT AND NURSING WOMEN	COMMUNAL HYGIENE (RECOMMENDED) OPEN TERRITORY DWELLING HOUSES					
60 kHz - 3 MHz 3 MHz - 30 MHz 30 MHz - 300 MHz 300 MHz - 300 GHz	50 V/m 10 V/m per 8 h 20 V/m 4 V/m per 8 h 5 V/m 2 V/m per 8 h 10 µW/cm <sup>2</sup> per 8 h 1 µW/cm <sup>2</sup> per 8 h 100µW/cm <sup>2</sup> per 2 h 1 mW/cm <sup>2</sup> per 0.3 h	10 V/m         10 V/m           4 V/m         0.4 V/m           2 V/m         0.2 V/m           5 μW/cm² (pulsed,         2 μW/cm² (pulsed,           rot. antenna)         rot. antenna)           1 μW/cm² (CW)         0.5 μW/cm² (CW)					

TABLE 10 Maximum permissible field intensities for RF and MW irradiation in occupational exposure and communal hygiene in the GDR. (Sources: DDR-Standard and appendix [18])

The actual exposure to RF and MW is shown in FIGURE 8 for three different categories. In urban areas with distributed Radio- and TV stations many thousand people are living day and night in EM fields. HANKIN et al. [39] investigated the power densities of UHF-TV stations and found that about 20,000 people in Washington, 20,000 people in Chicago and 3,000 people in Philadelphia are exposed to more than 4  $\mu$ W/cm<sup>2</sup>. Up to now little data are available about hazardous effects, from the work of VREELAND, SHEPHERD and HUTCHINSON [82] it is known, however, that TV-stations may affect the correct operation of pacemakers. Mobile communication systems and especially the UHF walkie-talkies are of greater significance as discussed in section 4.5.. One may assume, that about 10 million people are exposed to more than 1 mW/cm<sup>2</sup> by such sources, and it is worth to mention that in the East European countries the power of professional walkie-talkies is legally limited to about 100 mW. Biotelemetry transmitters are relatively safe, but deserve attention to possible artifacts.

# 4.7. RECOMMENDATIONS FOR SAFETY LIMITS FOR BODY-MOUNTED ANTENNAS

With the present poor knowledge about long-time effects of RF and MW on humans it is very difficult to state generalized recommendations. If very low permissible values are recommended, many sensible applications for biotelemetry and mobile communications have to be excluded. If very high values are recommended, we have to bear the responsibility for health hazards. Summarizing the facts collected in this chapter , we may come to the following conclusions:

Averaged permissible values related to the transmitting frequency: Below 20 MHz the power absorption is about proportional to  $f^2$ , is determined mainly by the H-field, and at 10 mW/cm<sup>2</sup> the SAR's are well below 1 W/kg. A maximum power deposition of 10 mW/cm<sup>2</sup> and maximum nearfield strengths of 200 V/m and 0.5 A/m are conservative limits.

Above 300 MHz the penetration depth is small, but local hot spots are possible under certain conditions. A maximum power density of  $lmW/cm^2$  and maximum near-field strengths of 63 V/m and 0.16 A/m must not be exceeded.

In the resonance region of 20 to 300 MHz excessive local absorption is only possible, if the human body is irradiated by a remote source (whole body exposure), and if the power density is more than 1 mW/cm<sup>2</sup>. For partial body exposure, like irradiation from body-mounted antennas, a distribution of the available radiation power may be expected. For low-power transmitters (e.g., < 25 W) the maximum power density should not exceed 1 mW/cm<sup>2</sup> and the near-field strengths should not exceed 63 V/m and 0.16 A/m. For a high-powered transmitter the coupling conditions and the power distribution have to be investigated.

<u>Peak power and dosis:</u> The reported phenomena seem to be effects from the dosis and effects from the peak values. The above indicated maximum ratings are conservative for CW and for maximum 2 hour exposure per day. For shorter durations and pulse modulated sources the above indicated averaged power densities may be multiplied by a factor of 10 and the above indicated averaged field intensities may be multiplied by a factor of 3 in order to obtain the permissible peak values. For long-time exposure, however, the above indicated values should by divided by the factors 10 and 3', respectively. <u>Risk factors</u>: Some of the risk factors are: conducting objects inside and outside the body (e.g., pace makers, electrodes, microphone cables, headphones, transducers), decreased state of health, extreme environmental conditions (heat), stress, immobility, pregnancy, etc. The present state of bio-research leads to the conclusion that some reversible biological effects may occur, but that real health hazards can be excluded at such low maximum safety limits. With respect to biotelemetry one has to take into account possible artifacts which may lead to wrong results. In animal experiments, especially with small animals, one should consider the wavelength/size ratio, the different biological functions (e.g., thermoregulation, metamorphosis) and the environmental (e.g., cage reflections) conditions.

# Leer - Vide - Empty

# 5. ANALYSIS OF THE ANTENNA-BODY SYSTEM

#### 5.1. DESCRIPTION OF THE GENERAL PROBLEM

# 5.1.1. DEFINITION OF THE BASIC GOALS IN ANTENNA BODY MODELLING

Comparatively speaking, there are two kinds of antenna engineers. First, the experienced practitioner who develops in a short time an exotic, welloperating antenna, but who is not able to deliver computational data, because there are too many variable, undefined parameters. Second, the theoretically-trained engineer, who computes for assumed idealized conditions an excellent theoretical antenna which operates badly under the given difficult environmental conditions.

Similar to above the same dilemma is manifested in our modelling problem:

The antenna-body model should contain on one hand all significant parameters which describe a realistic situation, on the other hand the selected model should be computable with a reasonable effort.

The basic goals may be defined as follows:

The computation of the antenna-body model should explain the systematic relation among frequency, body geometry, relative position of the antenna to the body and transmission loss to a remote antenna.

The results of the computation should be verified by a sufficiently accurate measuring method.

The obtained results from both theory and experiment should deliver fundamental data for the development of efficient, electrically small, body-mounted antennas. With the test subject standing on the earth, the antenna-body system should radiate omnidirectionally in the horizontal plane.

## 5.1.2. PARAMETER DESCRIPTION OF THE GENERAL ANTENNA-BODY SYSTEM

The general test situation is shown in Figure 11. Given is a test subject (TS), a body-mounted antenna  $(A_1)$  and a remote antenna  $(A_2)$ . The TS is electrically isolated from ground by the small space (s) between ground and feet. The relative position of  $A_1$  to the TS is defined by the azimuthal rotation angle ( $\phi$ ), the relative antenna height ( $h_B$ ) and by the antenna-body distance ( $d_{at}$ ), which is the distance between the center of  $A_1$  and the surface of the TS. The absolute position of the antennas is de-

fined as follows: The transmitting distance (d) is much greater than the wavelength ( $\lambda$ ) and d<sub>at</sub>; therefore we regard d as constant. The antenna height (h<sub>1</sub>) is the height of A<sub>1</sub> above ground, the antenna height (h<sub>2</sub>) is the height of A<sub>2</sub> above ground. The elevation angle ( $\theta_{e1}$ ) is the vertical angle of the beam  $\overline{A_1A_2}$  to the horizontal ground; the incident angle ( $\theta_i$ ) is the vertical angle between the beam  $\overline{A_1A_2}$  and the vertical body axis. The reflection angle ( $\gamma_B$ ) is the Brewster angle (later discussed in 5.3.2.).



FIGURE 11 General test situation. Parameters described in the text.

The point of interest is the transmission from an EM signal from A<sub>1</sub> to  $A_2$  when a body (TS) is near to the antenna A<sub>1</sub>. Because we want to investigate the systematic influence of the TS and not the properties of a specific antenna type, we define the test situation closer :

The two antennas  $A_1$  and  $A_2$  have to fulfill the following requirements:

- The physical size of A<sub>1</sub> should be smaller than any relevant dimension of the test set-up.
- A1 should have only one dominant E-polarization axis (p1)
- $A_1$  should radiate omnidirectionally in free space (e.g., radial radiation independent on the rotation angle of the axis of  $A_1$ )
- A1 should not change its input impedance due to body proximity.
- A<sub>2</sub> should have a strict linear E-polarization (p<sub>2</sub>)

As an additional regulation the input power  $(P_{in})$  at  $A_1$ , the absolute positions  $h_1, h_2, d$  and the polarization  $p_2$  are kept constant for each experiment performed at a given frequency (f).

The reference field strength  $E_0$  is measured at  $A_2$ , when no subject is present.  $A_1$  is oriented for maximum radiation in direction of  $A_2$ ; in the first case (see FIGURE 11)  $p_1$  and  $p_2$  are vertical.

The actual field strength E is measured at  $A_2$  when the TS is positioned, varying  $d_{at}$ ,  $h_B$ ,  $\phi$  and  $p_1$ .

The "transmission loss" ( $Loss_B$ ) and the "transmission gain" (Gain<sub>B</sub>) is defined as follows:

$$Loss_{B} = -20 \log \left(\frac{|E|}{|E_{O}|}\right) \text{ in decibels [dB]}$$
(19)  

$$Gain_{B} = 20 \log \left(\frac{|E|}{|E_{O}|}\right) \text{ in decibels [dB]}$$
(20)

Loss<sub>B</sub> is a measure for the negative (or positive) influence of the TS on the radiation characteristics of the antenna  $A_1$ . Loss<sub>B</sub> is a function of many parameters which are discussed in TABLE 12. The main radiation pattern of the antenna-body system are described by Loss<sub>B</sub> versus  $\phi$  and versus  $\theta_{e1}$ .

Because  $E_0$  represents the maximum possible field strength for a given test antenna in the best practical conditions (near ground, but far away from a body), we call Loss<sub>B</sub> of the free antenna ( $E_0$ ) "free-space level" (FSL):

Free-space level (FSL) = 0 dB = reverence level  $E_0$  (21) The polarization  $p_2$  of  $A_2$  may also be horizontal (parallel to the ground). In that case the FSL (0 dB) is obtained by orienting  $p_1$  of  $A_1$  horizontal, so that  $p_1$  and  $p_2$  are parallel.

In the case of reverse transmission, that is, transmission from  $A_2$  to  $A_1$ , the FSL (O dB) is calibrated to the electrical field strength  $E_0$  of the incident, plane wave in the region, where the TS is intended to be placed.

Finally, it is pointed out once more that  $E_0$  or the FSL is not identical with the field strength produced by an ideal isotropic radiator. At the moment we are not interested in the directional gain and in the efficiency of A<sub>1</sub>; such questions will be treated later in Appendix 16.1.

A list of Loss<sub>B</sub> determining parameters are presented in TABLE 12:

A SAMPLE OF SIGNIFICANT PARAMETERS DETERMINING GENERALLY THE TRANSMISSION LOSS					
PARAMETER	SYMBOL	RANGE	DESCRIPTION OF THE PARAMETER AND ESTIMATION OF ITS INFLUENCE		
Frequency Wavelength	f λ	10-1000 MHz 0.3 - 30 m	At frequencies from 50-300 MHz the dimensions of the human body are in the same order of magnitude as the wavelength. Resonance phenomena of unknown influence are to be expected.		
Polari- zation of A <sub>1</sub> and A <sub>2</sub>	<sup>թ</sup> լ <sup>թ</sup> շ	vert./hor./ radial vert./hor.	The main resonance of the human body occurs at vertical po- larization, a weaker resonance at horizontal polarization. Field irregularities outside the body may likely occur es- specially in the case of resonance for $p_1$ and $p_2$ vertical.		
Body- mounted antenna	Aı	h < 0.15 m	The requirements concerning length, polarization, omnidirect- ionality and impedance have been defined in 5.1.2. In the ex- periment $A_1$ has to be operated off-resonance in order to keep the impedance stable. An antenna for practical use, however, has to be tuned exactly on resonance for the provided mount- ing. Details are described in section 8. and 16.1.		
Remote antenna	A2	-	$A_2$ has no influence, if $A_2$ is a calibrated precision antenna with strict linear polarization $p_2$ .		
Height of A <sub>l</sub> above ground	hj	0.8 - 1.5 m	An h <sub>1</sub> of this range corresponds to $0.03 \lambda$ up to $5 \lambda$ . The radiation pattern in the vertical plane will show varying lobes for an elevation angle $\theta_{e1}$ of $5-20^{\circ}$ . The electrical center of the antenna-body system is not known, so that errors of about 4 dB are likely due to ground reflections. See $\gamma_B$ below.		
Height of A <sub>2</sub> above ground	h <sub>2</sub>	fix 6.2 m	At frequencies above 20 MHz the $h_2$ has no influence, as long as $h_1,d$ and thus $\theta_{e1}$ are kept constant.		
Transmis- sion distance	d	fix 31 m	As long as d is greater than $\lambda$ and if $\theta_{e1}$ is kept constant, the distance d has no influence. A small variation of the actual A1-A2 distance with dat of $\pm$ 1m causes a 0.3 dB change.		
Relative Al height	hB	0.6 - 1.3 m	$h_B$ has an influence on $h_1$ and causes a secondary influence. The primary influence of $h_B$ is a subject of this study.		
Azimuthal angle	\$	0 - 360 <sup>0</sup>	Experiments showed a large loss in the shadow zone at 180°. The influence of $\phi$ is a subject of this study.		
Aj-body- distance	dat	0.05 - 4 m	Experiments showed a large loss at small $d_{at}$ . A large range variation of $d_{at}$ is a subject of this study.		
Space TS to ground	s	0.2 - 1 m	Grounding effects will probably affect Lossg. The TS has to be isolated from the ground to minimize that influence.		
Geometry of the TS	LB DB	1.5 - 1.5 m 0.2 - 0.5 m	The length $L_{B}$ and the diameter $D_{B}$ of the TS might be related to the $\lambda/2$ resonances. A standard TS or phantom is required.		
Material of the TS	εr σ	6 - 100 0.01-1.6S/m	See section 4.2. The EM density is so high that the material might be of little significance at large $d_{at}$ .		
Elevation angle	θe1	5 <b>-</b> 20 <sup>0</sup>	The elevation angle $\theta_{\text{el}}$ should be kept constant. See discussion at h1 above.		
Brewster angle	Υ <sub>Β</sub>	12 - 170	The ground reflected vertically polarized wave is more than 10 dB smaller than the directly transmitted wave, if the Brewster angle $\gamma_B$ is adjusted properly. For details see section 5.3.2.		

TABLE 12 A sample of significant parameters which generally determine the transmission loss  $(Loss_B)$  in an antenna-body system.

# 5.2.1. TRANSFORMATION OF THE PROBLEM WITH THE RECIPROCITY THEOREM

The EM fields in the vicinity of an electrically small transmitting antenna  $A_1$  are of a complicated nature. If  $A_1$  is in the vicinity of the body, the near-fields are disturbed. The integral effects of the near-fields at the remote antenna  $A_2$  determine the transmission loss Loss<sub>B</sub>. The computation of this problem is quite difficult and needs much additional input data concerning the specific antenna and the specific body.

Because we are only interested in the transmission loss from a point  $A_1$  to a point  $A_2$ , we may ask also for the transmission loss from a point  $A_2$  to a point  $A_1$ . If both losses are of the same magnitude, we had only to look at the more simple second case.

The reciprocity theorem (HEILMANN [42]) runs as follows:

Assumed are two arbitrary antennas  $A_1$  and  $A_2$  at arbitrary relative orientation and distance. If the same voltage is applied either to  $A_1$  or  $A_2$ , the same current will flow in the other antenna  $A_2$  or  $A_1$ .



FIGURE 13 Test set-up for the verification of the reciprocity theorem (Source: HEILMANN [42]).

The reciprocity theorem is valid for any linear medium, where  $\mu$ , $\epsilon$  and k are scalar, but arbitrary quantities. These electromagnetic parameters may depend on their locations, but are not functions of the field vectors.

In our model we can assume that the body material is linear (i.e., not dependent on the magnitude of the EM-field, no thermal alterations of the material) and isotropic (i.e., the properties do not depend on the direction of the EM field vectors). If both antennas are terminated with the same impedance, we may conclude that  $Loss_B$  is the same for A1-A2 and A2-A1 transmission.

As it will be shown in section 9.1.5. the reciprocity theorem is valid for our application. The verification measurements revealed a difference of less than 2 dB when the direction of transmission was reversed (transmitter and receiver changed over, antennas unchanged). This held true for all test bodies (see 5.4.1.), for all measurements at 100 to 1000 MHz, for all  $d_{at}$ 's above 0.05 m and for an applied input power  $P_{in}$  of 1 mW.



FIGURE 14 Transformed test situation. Antenna  $A_2$  generates a plane wave in the region of the body, the wave is scattered at the body and the disturbed wave is picked-up by antenna  $A_1$ .

The transformed problem, which we have to <u>compute</u> in this study is: a body is irradiated by a plane wave with an FSL field strength  $E_0$  at 0. We have to quantify the  $\vec{E}(\vec{a})$  vector components  $\underline{E}_V$  (vertical),  $\underline{E}_h$  (horizontal) and  $\underline{E}_r$  (radial) around the body as a function of  $\vec{a}$  (FIGURE 14).

- 54 -

In the test situation FIGURE 14 the monitored E-vector  $\vec{E}(\vec{a})$  is given by

$$\underline{\vec{E}}(\vec{a}) = \underline{\vec{E}}_{inc}(\vec{a}) + \underline{\vec{E}}_{scat}(\vec{a})$$
(23)

where  $\underline{\vec{E}}_{inc}(\overline{\vec{a}})$  is the incident E-vector from  $A_2$ , and  $\overline{\vec{E}}_{scat}(\overline{\vec{a}})$  is the scattered E-vector from the body. The total E-vector  $\underline{\vec{E}}(\overline{\vec{a}})$  is varying in direction, amount and phase for variable positions  $\overline{\vec{a}}$ . Because the antenna  $A_1$  will have a certain length 2h (FIGURE 15), the E-field irradiating  $A_1$ is not constant along the antenna axis.

Because the voltage induced at the terminals of the antenna  $A_1$  will become a measure for the transmission loss, we have to investigate the relation between  $\vec{E}(\vec{a})$ , antenna length 2h and the induced voltage Uind.



# FIGURE 15

Induced voltage in a linear dipole antenna immersed in a inhomogeneous electrical field.  $\vec{E}(\vec{a})$  : variable applied E-field

 $\underline{E}_{z}(z)$ : z-component of  $\underline{\underline{E}}(\overline{a})$ for  $\overline{a}$  on the z-axis z : axis of the antenna 2h : length of the antenna  $U_{ind}$  : induced voltage at the terminal of the antenna  $\overline{a}, \overline{a}_{0}$  : position vector (see also FIGURE 14)

Let us assume a linear dipole antenna  $A_1$  with the axis z and the length 2h (FIGURE 15). Let us further assume that we know the current distribution function  $\Psi(z)$  along the z-axis for an incident plane wave (see e.g., HEILMANN [42]). The induced voltage  $U_{ind}$  is then given by:

$$U_{\text{ind}} = \int_{-h}^{+h} \underline{E}_{z}(z) \Psi(z) dz \qquad (24)$$

If the length 2h of  $A_1$  is adequately small, the variations of  $\underline{E}_Z(z)$  along the z-axis are so small, that we are allowed to replace  $\underline{E}_Z(z)$  by the plane wave equivalent field strength  $\underline{E}_Z(o)$  of  $\underline{E}(a_0) \cdot e_Z^2$ .

For a given antenna  $A_1$  a certain relation between the axial antenna length 2h and the variability of the axial field  $\underline{E}_z(z)$  has to exist, if the above approximation should lead to induced voltages  $U_{ind}$  representing the field-strength. We assume the following:

- $A_1$  should measure the relative field strength  $E_z(z)$  compared to an FSL reference field  $E_0$  with an accuracy of 2 dB.
- $A_1$  is small compared with the wavelength (2h < 0.2 $\lambda$ ).
- The total  $\underline{\vec{E}}(\vec{a})$ -vector is polarized in z-direction, which is the main polarization axis  $p_1$  of  $A_1$ .
- We assume for the worst case a constant current distribution function  $\Psi(z)$  = constant. An  $\underline{E}_Z$  at the ends of the antenna has the same weight on  $U_{ind}$  as an  $\underline{E}_Z$  at the antenna center.
- We assume that  $\underline{E}_{Z}(+h)=E_{0}=0$  dB and that  $|\underline{E}_{Z}(z)|$  increases monotonously from -h<z<+h.  $\underline{E}_{Z}(z)$  may be approximated as:

$$|\underline{E}_{z}(z)| = E(x) = a_{0} + a_{1}x + a_{2}x^{2}, \quad x = \frac{z+n}{2h}, \quad 0 < x < 1$$
(25)

- We assume that the phase angle  $\arg(\underline{E}_z(z))$  increases monotonously from -h < z < +h and depends linearly on z :

 $\arg(E_{\tau}(z)) = \text{const.} + z \cdot \beta \cdot k$ ;  $2h \cdot \beta \cdot k < 1/2$ ,  $\beta = \text{const.}$  (26)





$$\begin{split} |\underline{E}_{Z}(-h)| &= E(x=0) = \text{minimum at one antenna end} = -6.00 \text{ dB} \\ |\underline{E}_{Z}(0)| &= E(x=V_{2}) = \text{nominal center field} = -4.08 \text{ dB} \\ |\underline{E}_{Z}(+h)| &= E(x=1) = \text{maximum at other antenna end} = 0.00 \text{ dB} \\ |\underline{E}_{Z}(0)| &= \overline{E}(x=V_{2}) = \text{logarithmic mean value} = -3.00 \text{ dB} \end{split}$$
(27)

The induced voltage U<sub>ind</sub> at the antenna terminals is with (25)

$$U_{ind} = \int_{-h}^{+h} \frac{E_z(z) dz}{c} = 2h \int_0^{+1} E(x) dx = 2h \left(a_0 + \frac{a_1}{2} + \frac{a_2}{3}\right)$$
(28)

The approximated induced voltage  $\overline{U}_{ind}$  with a constant (center) field is

$$\overline{U}_{ind} = \int_{-h}^{+n} \frac{E_z(0)}{-h} dz = 2h \int_{0}^{+1} E(\frac{1}{2}) dx = 2h \left(a_0 + \frac{a_1}{2} + \frac{a_2}{4}\right)$$
(29)

The logarithmic mean value  $\overline{|\underline{E}_z(o)|}_{dB} = 20 \log(|\underline{E}_z(o)|/|\underline{E}_z(+h)|)$  is

$$\overline{|\underline{E}_{z}(0)|}_{dB} = \overline{E}(\frac{1}{2})_{d\overline{B}} \frac{1}{2} \left[ 20 \log 1 + 20 \log(\frac{a_{0}}{a_{0} + a_{1} + a_{2}}) \right]$$
(30)  
$$= 20 \log\left[ \sqrt{1 + \frac{a_{1}}{a_{0}} + \frac{a_{2}}{a_{0}}} / \left( 1 + \frac{a_{1}}{a_{0}} + \frac{a_{2}}{a_{0}} \right) \right]$$

The logarithmic center field strenght  $|\underline{E}_z(0)|_{dB}$  is 20 log  $|\underline{E}_z(0)|/|\underline{E}_z(+h)|$ :

$$|\underline{E}_{z}(0)|_{dB} = E(\frac{1}{2})_{dB} = 20 \log \left[ \left( 1 + \frac{a_{1}}{2a_{0}} + \frac{a_{2}}{4a_{0}} \right) / \left( 1 + \frac{a_{1}}{a_{0}} + \frac{a_{2}}{a_{0}} \right) \right]$$
(31)

From (28) and (29) we obtain the logarithmic difference  $\Delta U$  :

$$\Delta U = 20 \log U_{\text{ind}} - 20 \log \overline{U}_{\text{ind}} = 20 \log \left| \frac{1 + \frac{a_0}{2a_0} + \frac{a_2}{3a_0}}{1 + \frac{a_1}{2a_0} + \frac{a_2}{4a_0}} \right|$$
(32)

From (30) and (31) we obtain the logarithmic difference  $\Delta E$  :

$$\Delta E = 20 \log E(\frac{1}{2}) - 20 \log \overline{E}(\frac{1}{2}) = 20 \log \left| \frac{1 + \frac{a_1}{2a_0} + \frac{a_2}{4a_0}}{\sqrt{1 + \frac{a_1}{a_0} + \frac{a_2}{a_0}}} \right|$$
(33)

If we replace  $\frac{a_1}{a_0}$  by  $\alpha_1$  and  $\frac{a_2}{a_0}$  by  $\alpha_2$  , we obtain

$$\Delta U = 20 \log \left| \frac{1 + \frac{\alpha_1}{2} + \frac{\alpha_2}{3}}{1 + \frac{\alpha_1}{2} + \frac{\alpha_2}{4}} \right|$$
(34)

$$\Delta E = 20 \log \left| \frac{1 + \frac{\alpha_1}{2} + \frac{\alpha_2}{4}}{\sqrt{1 + \alpha_1 + \alpha_2}} \right|$$
(35)

As we can see from equation (35),  $\Delta E$  becomes zero for specific pairs of  $\alpha_1$  and  $\alpha_2$ , also when  $\Delta U$  is not zero. Therefore, the number  $\Delta E$  cannot be used as an indicator for  $\Delta U$ .

 $\Delta U$ , however, varies only little for  $0 < \alpha_1, \alpha_2 < 1$ . As can be seen from (34),  $\Delta U$  increases with  $\alpha_2$  and decreases with increasing  $\alpha_1$ . A good measure for the maximum  $|\Delta U|$  can be obtained with  $|\delta E|$ , assuming that the E(x) function is of the type E(x) =  $a_0 + a_2x^2$ . In FIGURE 16 such a function has been assumed with  $a_0 = 0.5$ ,  $a_2 = 0.5$  and thus  $\alpha_1 = 0$ ,  $\alpha_2 = 0$ . The obtained results  $\delta E$  and  $\Delta U$  are:

$$|\delta E| = 6 dB$$
;  $|\Delta U| = 0.56 dB$  (36)

If we regard only the  $\Delta U/\alpha_2$  ratio of the same quadratic equation  $E(x) = a_0 + a_2 x^2$ , we obtain for a permissible  $|\Delta U|$  of 1 dB an  $\alpha_2$  of 2.3. Thus, the field strength variation  $\delta E$  along the antenna should not exceed :

$$\left|\delta E\right| = \left|\left|\underline{E}_{Z}(-h)\right|_{dB} - \left|\underline{E}_{Z}(+h)\right|_{dB}\right| < 20 \log\left(\frac{1}{1+2.3}\right) = 10 dB \qquad (37)$$

The permissible phase variation  $\delta\Phi$  along the antenna is limited by:

$$|U_{ind}| = \iint_{-h}^{+h} \hat{E} \sin(\beta \cdot z \cdot k) dz| = |\frac{\hat{E}}{k} (\cos(2h\beta \cdot k))|$$
(38)

If we require a resulting real-part variation of less than 10 dB (which causes a  $|\Delta U| < 1~\text{dB}$ , see (37)), we obtain a phase variation limit  $\delta\Phi$  of

$$|\delta\Phi| = |\arg(\underline{E}_{z}(-h)) - \arg(\underline{E}_{z}(+h))| < \arccos(10^{-10}_{-20}) = 71^{\circ}$$
 (39)

For the practical applications the field homogeneity requirements along a dipole test antenna A<sub>1</sub> are completely specified by (38) and (39). Within these limits the field strength  $E_z(o)$  at the <u>center</u> of the antenna is <u>representative</u> for the whole field around the antenna, at an accuracy of better than 2 dB. Under these conditions the transmission loss Loss<sub>B</sub> for  $p_1 = z$ -polarization can be computed as:

$$Loss_B = -20 \log \frac{|E_z(0)|}{|E_0|} \qquad E_0 = FSL, z = polar. axis, (40)$$
$$\frac{|E_0|}{|E_0|} \qquad E_z(0) = center field strength$$

If equations (36) and (38) are not fulfilled, equation (24) has to be evaluated for both theoretical Loss $_{\rm B}$  and verification experiments.

In section 4.3. and FIGURE 3 it has been shown that the absorption of EM energy inside of a dielectric sphere depends on the ratio of the spherecircumference 2% to the wavelength  $\lambda$  (see RACS).

With the Radar Cross Section (RCS) it should be shown that also the scattered fields outside a body depend on the same ratio. If a body is irradiated by an EM wave, a part of the EM energy will be absorbed as discussed in 4.3. and the remaining EM energy will be scattered in all directions. That part of EM-energy which will be back-scattered toward the EMsource can be quantified by the RCS. The RCS is defined as:

$$RCS = \frac{P_{scat}}{|\vec{P}|} = \lim_{R \to \infty} 4\pi R^2 \frac{|\vec{E}_{scat}|^2}{|\vec{E}_{inc}|^2}$$
(41)

 $P_{scat}$  is the total scattered power,  $\vec{P}$  is the incident power density of the EM wave at the body, R is the distance between the remote source and the body,  $\vec{E}_{scat}$  is the back scattered E-field and  $\vec{E}_{inc}$  is the incident Efield. The RCS of a sphere is shown in FIGURE 17 versus the ratio  $2\pi a/\lambda$ .



FIGURE 17 Radar Cross Section (RCS) of a sphere with the radius a, related to the shadow area  $\pi a^2$ , versus the relative frequency  $f_{rel}$  (sphere circumference  $2\pi a/a$  wavelength  $\lambda$ ). (Source: BECKER [7]).

The shape of the human body can be approximated by an ellipsoid. From the literature it is known that the RCS of an ellipsoid is similar to the RCS of a sphere, if the largest circumference is equal and if the incident wave is polarized parallel to the main axis of the ellipsoid. In FIGURE 18 a simplified man-model is shown and the approximated main resonant frequency for vertical polarization is indicated:



FIGURE 18 Simplified Man-Model Vertical axis : 2c = 1.8 m Sagittal axis : 2a = 0.2 m Lateral axis : 2b = 0.3 m Incident vertical polarized waves: Sagittal incidence :  $\vec{E}_{ca}, \vec{k}_{ca}$ Lateral incidence :  $\vec{E}_{cb}, \vec{k}_{cb}$ Circumferences for vertical polarized incident waves:  $C_{ca} \cong \pi(1.5(c+a) - \sqrt{c \cdot a}) = 3.77 \text{ m}$  $C_{cb} \cong \pi(1.5(c+b) - \sqrt{c \cdot b}) = 3.79 \text{ m}$ 

Resonant frequency for maximum RCS f<sub>res</sub> <sup>°</sup> ≥ 80 MHz (vertical polar.)

Comparing FIGURE 3 with FIGURE 17 we notice a certain relationship. If the absorption in a body and if the scattered field <u>far away</u> from the body is highly dependent on the circumference/wavelength ratio, we may assume that the scattered fields <u>near</u> the body are also affected by the same ratio:

- At frequencies below 40 MHz the RACS and the RCS are small. That means that the integral effect of the body on the incident EM-wave is small. However, local field disturbances in the vicinity of the body has to be expected, because the body is not transparent to EM-waves.
- At high frequencies above about 200 MHz the RACS and RCS are high but almost constant. The resonance effects may be neglected, and the three dimensional problem may be reduced to a two-dimensional problem, e.g., the scattering from an infinite cylinder.
- In the resonance region, that is about 40 to 200 MHz, both RACS and RCS are high and depend greatly on the frequency. Numerical field computations on a three dimensional model are urgently needed.

The dielectric properties of biological materials have been discussed in section 4.2.. They depend on material and frequency, but, in general, the human body represents a dense, lossy medium. With respect to the field distribution around the irradiated body we have to look closer to the transmission of EM-energy through the body and the reflection of an EM-wave at the body's surface:

<u>The transmission</u> of EM-energy through the body can be estimated as follows: We assume a vertical, circular cylinder, which is irradiated by a plane, vertical polarized wave with the incident E-field  $\vec{E}_{inc}$  (FIGURE 19):



FIGURE 19 Transmission of EMenergy through a circular cylinder. 2a : diameter of the cylinder 1,3: medium air 2 : medium body, lossy material E1 : incident wave,  $|\vec{E}_{inc}|=0$  dB R12: reflected E-field in 1 D21: refracted E-field in 2 E2 : attenuated E-field in 2 R23: reflected E-field in 2 D32: transmitted E-field in 3

The EM wave enters into medium 2 only for small angles  $\alpha$  and is refracted close to the center of the cylinder. The entered wave  $D_{21}$  is attenuated during the travelling through the body and amounts finally to  $E_2$ . The second refraction produced a small transmitted and scattered wave,  $D_{32}$ . Maximum transmission occurs at  $\alpha = 0^{\circ}$  and at low frequencies. For this case the complex computation of the EM transmission through a plane slab model of 2a thickness was performed by a computer, using the method of TELL [76]:

<u>Input_data</u> :	Output data:						
Thickness plane slab	:	2a	= 0.25 m	Reflection	R <sub>12</sub> =	-0.8	dB
Frequency	:	f	= 75 MHz	Attenuated field	E <sub>2</sub> =	-38	dB
Material : $\varepsilon_r = 50$ ,		σ	= 1.25 S/m	Transmitted field	$D_{32} =$	-52	dB

Without EM-transmission the Gaing(5.1.2.) in the shadow zone amounts up to -30 dB at extremely small  $d_{at}$ . Because the transmitted wave amounts to -52 dB, is scattered and is even smaller at higher frequencies, the transmitted wave through the human body has no effect on the transmission Loss<sub>B</sub>.

<u>The reflection</u> of the incident EM-wave at the surface of the human body determines the scattered field around the body. The EM-wave reflected at the air-body interface is described by the three reflection coefficients  $\Gamma$  (see TABLE 1) and <u>R<sub>E</sub>, R<sub>H</sub></u> (see also section 5.3.2. and FIGURE 24):

- $\underline{\Gamma}$ : reflection coefficient for the E-vector of an TEM-wave with rectangular incidence on the surface. The complex number  $\underline{\Gamma}$  is derived only from the intrinsic impedances of the interface media (for computation see e.g.,TELL [76]). An average value for our frequency range is about 0.7-0.9 /  $\times$  175-177<sup>0</sup> (TABLE 1).
- $\underline{R}_E$ : reflection coefficient for the E-vector of an TE-wave. Here the E-vector is parallel to the surface.  $R_E$  is the ratio of  $\underline{E}_{ref}$  to  $\underline{E}_{inc}$  and  $\underline{H}_{ref}$  to  $\underline{H}_{inc}$ . For a vertical body and a vertical polarized incident TE-wave (FIGURE 19)  $\underline{R}_E$  is about  $1/x180^{\circ}$  (larger than  $\underline{\Gamma}$ ) for all angles  $\alpha > 10^{\circ}$ . (For reference see FIGURE 24, horizontal reflection at earth,  $\alpha = 90^{\circ} - \gamma$ ).
- $\underline{R}_{H}$ : reflection coefficient for the H-vector of an TM-wave. Here the H-vector is parallel to the surface.  $R_{H}$  is the ratio of  $\underline{H}_{ref}$  to  $\underline{H}_{inc}$  and -  $\underline{E}_{ref}$  to  $\underline{E}_{inc}$ . For a vertical body and a horizontal polarized incident TM-wave,  $\underline{R}_{H}$  is near 1 /  $\times 0^{\circ}$  for all angles  $\alpha$  smaller than 70°. Total refraction and signum change occur only at  $\alpha > 80^{\circ}$  and are of little significance on the integrally scattered fields. The average reflection coefficient for the Evector is greater than  $\underline{\Gamma}$  and amounts to about 1/  $\times 180^{\circ}$ . (For reference see FIGURE 24, vertical reflection at earth,  $\alpha = 90^{\circ} - \gamma$ ).

The reflection coefficient  $\underline{\Gamma}$  represents at least for vertical polarization the smallest occuring reflection coefficient. The reflection coefficient for any incident E-vector at the interface air to a perfect conductor is -1. Compared with  $\underline{\Gamma}$  the amplitude of the body-reflected E-vector differs only within - 3 dB. This means that we are allowed to assume the human body as a <u>perfectly conducting</u> body, if we are only interested in the fields outside of the human body.

However, the penetration depth  $\delta$  of the EM-wave in the human body is not zero as in a perfect conductor. The surface charges and the surface currents (see section 6.) which are responsible for the scattered field are distributed in the outer layers but also with decreasing amplitude in deeper regions (TABLE 1). Therefore, the perfectly conducting man-model is only accurate for d<sub>at</sub> > 50 mm and verification measurements are needed.

# 5.3.1. ANTENNA MEASUREMENTS IN PROXIMITY TO THE GROUND

For the field measurements the TS has to be rotated together with the bodymounted antenna A<sub>1</sub>, and the antenna-body distance  $d_{at}$  must be varied up to 4 m. Theoretically, antenna measurements with quasi-free-space conditions would be possible with anechoic chambers or elevated platforms. However, at measuring frequencies from 10 to 1000 MHz both methods are not suitable. Below 200 MHz an anechoic chamber has to be huge, and the common absorber pyramid plates has to be matched to the frequency (length of an absorber pyramid approximately  $\lambda/4!$ ). An elevated platform is prohibitive due to material and stability problems. Thus, the measurements have to be performed in proximity to the ground, and we have to analyse the influence of the ground on the relative transmitted signal from A<sub>1</sub> to A<sub>2</sub>:



FIGURE 20 Effects in proximity to the ground in antenna measurements. TS : Test subject Fr.El.: Fresnel Ellipsoid (b'+b"-a) =  $\lambda/2$ A<sub>1</sub> : Body-mounted antenna C<sub>S</sub> : stray capacitors from TS to ground A<sub>2</sub> : Remote antenna  $\gamma$  : reflection angle (glancing angle)

At a small antenna height hj (FIGURE 20) disturbing effects occur by:

- Capacitive coupling of the TS to the ground

- Entrance of material into the first Fresnel Ellipsoid and depending on h1,h2 and d :

- Ground reflections with significant interferences if  $(g'+g"-a) = n\frac{\lambda}{2}$ and if the reflection coefficient of the ground is high. The capacitive coupling of the TS to the ground depends on the space s and the material of both the TS and the ground. With great effort the stray capacitors  $C_S$  might be computed, but its effect on the scattered field around the TS would require a further study. Because such a study does not help. much in the understanding of the antenna-body problem, it is not necessary to further scrutinize an investigation. In order to reduce the capacitive coupling, a constant space s of 0.2 m is now defined for the measurements. Verification measurements with varying s from 0.2 to 1 m will show that the capacitive coupling can be neglected with this restriction.

The first Fresnel Ellipsoid is defined as the geometrical locus for all points which satisfy the condition  $(b'+b'-a) = \lambda/2$ . (FIGURE 20). If no obstacles interfere with the first Fresnel Ellipsoid, one speaks of optical line-of-sight propagation (BECKER [7]). The lower frequency limit  $f_{lim l}$  for this free propagation can be approximately determined for the data:

Assuming the ellipsoid touches the ground with the reflected beam g'+g" we obtain the path difference  $\Delta s_1$  and the frequency limit  $f_{lim 1}$ :

$$\Delta_{s1} = g' + g'' - a = \sqrt{(h_2 + h_1)^2 + d^2} - \sqrt{(h_2 - h_1)^2 + d^2} = 0.455 m \quad (43)$$

$$f_{lim_1} = c/2\Delta_{s_1} = 330 \text{ MHz}$$
 (44)

It has to be clearly stated that accurate absolute antenna measurements are not possible for frequencies below 350 MHz with such a test set-up. However, relative measurements with an accuracy of about 2 dB (experimental experience) are possible, if one looks carefully on the reflection angle  $\gamma$ .

# 5.3.2. REFLECTIONS FROM THE GROUND AND WAVE POLARIZATION

The following considerations are based on the condition that there is an optical line-of-sight propagation as discussed above.

The antenna measurements are performed on a very large lawn. Because the grass is not an ideal reflector, we have to find out the frequency limit  $f_{lim2}$  at which a certain grass thickness  $d_g$  changes from an EM smooth to an EM rough surface (Rayleigh criterion, BECKER [7]):



FIGURE 21 Phase difference of two beams, reflected on different heights.

 $\lambda$  : reflection angle

d<sub>g</sub> : thickness of the reflecting layer (grass) U : upper beam

: lower beam

∆s<sub>2</sub>: path difference

In FIGURE 21 an EM-wave is shown which is reflected by the reflection angle  $\gamma$  from a grass surface. The upper beam U is reflected from the top of the grass layer, the lower beam L from the actual earth. We assume a reflection coefficient <u>R</u> of +1. The path difference  $\Delta s_2$  depend on  $\gamma$  and d<sub>o</sub>:

$$\Delta_{s2} = 2 \cdot d_q \cdot \sin \gamma \tag{45}$$

If the resulting phase difference  $\Delta \psi_2$  is near 0, the surface can be regarded as smooth. If  $\Delta \psi_2$  is larger than ¶, the surface is rough and represents a random scatterer. With the Rayleigh criterion  $\Delta \psi_2 < \P/2$  we obtain finally the lower frequency limit  $f_{\lim 2}$  for a smooth surface:

$$f_{\lim 2} < \frac{c}{8 \cdot d_g \cdot \sin \gamma}$$
(46)

If we assume a thickness  $d_g \stackrel{<}{=} 0.05~m$  and a reflection angle  $\gamma \stackrel{<}{=} 17^0$  we obtain:

$$f_{1im2} < 2500 \text{ MHz}$$
 (47)

Thus we study the ground reflection at the smooth earth (FIGURE 22):



FIGURE 22 Reflection of a wave.  
I,
$$\theta_i$$
: incident wave and angle  
D, $\theta_d$ : refracted wave and angle  
R, $\theta_r$ : reflected wave and angle  
1 : medium air ( $\varepsilon_1$ , $\mu_1$ , $\sigma_1$ , $m_1$ )  
2 : medium earth ( $\varepsilon_2$ , $\mu_2$ , $\sigma_2$ , $m_2$ )  
m<sub>1</sub>,m<sub>2</sub>: refraction indices  
 $\gamma$  : reflection angle 900- $\theta_i$ 

The computation of reflection and refraction of waves at the interface of two media is completely described by e.g., BAGGENSTOS [5] and BECKER [7]. The derivation of the formulas is quite long, but well known so that only the significant final formulas should be indicated here.

The correlation of the refraction index m, the wave factor k and the characteristic impedance  $\rm Z_m$  in a medium is given by BECKER [7] :

$$Z_{\rm m} = \left(\frac{j\omega \ \mu_0 \ \mu_r}{j\omega \ \varepsilon_0 \ \varepsilon_r - \sigma}\right)^{V_2} = \frac{\omega \ \mu_0 \ \mu_r}{k} = \frac{\mu_r}{m} Z_0 \tag{48}$$

The correlation of the incident angle  $\theta_i$  with the refraction angle  $\theta_d$  and with the reflection angle  $\theta_r$  ( $\gamma = 90^{0}-\theta_r$ ) is:

$$\theta_i = \theta_r$$
,  $k_1 \sin \theta_i = k_2 \sin \theta_d$ ,  $m_1 \sin \theta_i = m_2 \sin \theta_d$  (49)

The two different polarizations have to be treated separately (TABLE 23):

THE DIFFERENT REFLECTION OF A VERTICALLY AND A HORIZONTALLY POLARIZED WAVE				
VERTICAL POLARIZATION (TM-WAVE)	HORIZONTAL POLARIZATION (TE-WAVE)			
The E-vector is in the plane of incidence. The H-vector is parallel to the interface. Directly computable : reflection coefficient R <sub>H</sub> of the H-vector : $R_{H} = \frac{\mu_{1}m_{2}^{2}\cos\theta_{i}-\mu_{2}m_{1}\left(m_{2}^{2}-m_{1}^{2}\sin^{2}\theta_{i}\right)^{V_{2}}}{\mu_{1}m_{2}^{2}\cos\theta_{i}+\mu_{2}m_{1}\left(m_{2}^{2}-m_{1}^{2}\sin^{2}\theta_{i}\right)^{V_{2}}}$	The H-vector is in the plane of incidence. The E-vector is parallel to the interface. Directly computable : reflection coefficient R <sub>E</sub> of the E-vector : $R_E = \frac{\mu_2 m_1 \cos\theta_i - \mu_1 m_2 \left(1 - \left(\frac{m_1}{m_2} \sin\theta_i\right)^2\right)^{V_2}}{\mu_2 m_1 \cos\theta_i + \mu_1 m_2 \left(1 - \left(\frac{m_1}{m_2} \sin\theta_i\right)^2\right)^{V_2}}$			
For $\mu_1 = \mu_2 = \mu_0$ and $\sigma_1 = \sigma_2 = 0$ : $R_H = \frac{\tan(\theta_i - \theta_d)}{\tan(\theta_i + \theta_d)}$	For $\mu_1 = \mu_2 = \mu_0$ and $\sigma_1 = \sigma_2 = 0$ : $R_E = -\frac{\sin(\theta_i - \theta_d)}{\sin(\theta_i + \theta_d)}$			
If $\theta_i^{+}\theta_d = \pi/2$ , the $R_H$ becomes zero and there is no reflected wave. The glancing reflection angle $\gamma$ ( $\gamma = 90^{\circ} - \theta_i$ ) for which this extinction occurs is here defined as BREWSTER ANGLE $\gamma_B$	R <sub>E</sub> varies only a little and does not become zero. There is always a ground reflection.			

TABLE 23 Computation of the reflected wave for vertically and horizontally polarized incident waves. (Source: BECKER [7]). The amounts and phases of the reflection coefficients  $R_H$  and  $R_E$  for a TMand a TE-wave reflected by a realistic earth surface ( $\epsilon_2$ =10 $\epsilon_0$ , $\sigma_2$ =1mS/m) is shown in FIGURE 24:



FIGURE 24 Amplitudes  $R_H$  and  $R_E$  (left) and phases  $\psi$  of  $R_H$  and  $R_E$  (right) versus the glancing reflection angle  $\gamma$  for an average earth surface. (Source: BECKER [7]).

If the conductivity is very low, the refraction index  $m_2$  and the Brewster angle  $\gamma_B'$  become for  $\varepsilon_r = 10, \mu_r = 0, \sigma_2 = 0$ :  $(\gamma_B' = 90^0 - \gamma_B)$ 

$$m_2 = \sqrt{\epsilon_r}$$
;  $\gamma_B' = 90^\circ - \arctan m_2 = 17^\circ$  (50)

which is close to the value depicted in FIGURE 24.

At <u>vertical polarization</u> the influence of the reflected wave is small, if the glancing reflection angle  $\gamma$  is near  $\gamma_B$ '. With the data in FIGURE 24 applied to the antenna configuration in FIGURE 20 we can compute according to BECKER [7] the influence of the reflected wave on the transmission from A<sub>2</sub> to A<sub>1</sub>. The field strength E<sub>oeff</sub>' at A<sub>1</sub> produced by the direct beam (a) from A<sub>2</sub> is (input power P<sub>in</sub>, antenna gain G<sub>2</sub>):

$$|E_{oeff'}| = \frac{(30 P_{in} G_2)^{1/2}}{d}$$
(51)

The actual  $E_{neff}$  at  $A_1$  produced by the beams (a) and (g'+g'') is:

$$|\mathsf{E}_{\mathsf{oeff}}| \approx |\mathsf{E}_{\mathsf{oeff}}'| \left\{ 1 + 2|\mathsf{R}_{\mathsf{H}}| \cos(\psi + \frac{4\pi h_1 h_2}{\lambda \, \mathsf{d}}) + |\mathsf{R}_{\mathsf{H}}|^2 \right\}^{V_2}$$
(52)

If we insert the data (42), we obtain a  $\gamma$  of 13.4° and thus an  $|R_H|$  of less than 0.2 (FIGURE 24). At worst case  $\psi$  conditions and for all frequencies

between 30 to 1000 MHz the field strength E<sub>oeff</sub> varies within :

$$0.8 |E_{oeff'}| < |E_{oeff}| < 1.2 |E_{oeff'}|$$
 (53)

so that  $|E_{oeff}|$  differs from -2.0 to +1.6 dB. With respect to the transmission loss determination the effect of the ground reflected wave will be smaller, because the reference  $E_0$  will be determined for each measuring frequency, and because the direct and reflected E-vectors are not parallel.

At <u>horizontal polarization</u> the influence of the reflected wave might be important, if we look on FIGURE 24 and equation (52). With  $|R_E| = 1$  we obtain from (52) the interference equation:

$$|E_{oeff}| \approx |E_{oeff}'| \left| 2sin\left(\frac{2\pi h_1 h_2}{\lambda d}\right) \right|$$
 (54)

With the data (42) we obtain at 323 MHz a maximum (+6 dB) and at 646 MHz an extinction (< -10 dB). Because  $h_1$  of  $A_1$  is not identical with the height of the antenna-body center, transmission loss measurements are very inaccurate at arbitrary frequencies. Reasonable measurements at horizontal polarizations are only possible at certain selected frequencies and only with large  $h_1$  and  $h_2$ . Because the horizontal polarization is of little significance for omnidirectionally radiating antenna-body systems, measurement at horizontal polarization will not be performed.

# 5.3.3. FIELD HOMOGENEITY ALONG THE BODY AXIS AT VERTICAL POLARIZATION

For the computation a homogeneous, plane wave has been assumed. For the measurements this is not absolutely true as can be seen from FIGURE 25:



FIGURE 25 Field homogeneity along the body axis of the TS.

For this consideration we assume that  $A_1$  is on the vertical axis of the TS and that the following parameters are given:

p1,p2	:	polarization = vertica				
hj	:	nominal antenna height	=	1.16	m	
h2	:	fixed antenna height	=	6.2	m	
d	:	transmission distance	=	31	m (5	55)
s	:	space TS to ground	=	0.2	m	
Լը	:	length of the TS	=	1.8	m	

For the computer computations (section 6.4., FIGURE 33) one assumes a plane wave (dashed lines in FIGURE 25) irradiating the TS with the nominal  $\theta_i$ :

$$\theta_i$$
 = nominal incident irradiation angle = 80.8° (56)

The program computes the phase difference among any field point and the origin 0 for the incident <u>plane</u> wave. In this case the path difference  $\Delta_{s3}$  of the plane wave along the axis of the TS amounts to :

$$\Delta_{c3} = L_B \cos \theta_i = 0.2878 \, \mathrm{m} \tag{57}$$

In the actual measurements (solid lines in FIGURE 25) the TS is irradiaated by a <u>spherical</u> wave with an averaged incident angle  $\theta_i$ . The path difference  $\Delta_{s3}$ ' of the spherical wave along the axis of the TS is:

$$\Delta_{s3'} = g - f = (d^2 + (h_2 - s)^2)^{V_2} - (d^2 + (h_2 - L_B - s)^2)^{V_2} = 0.2921 \text{ m}$$
(58)

The difference of the plane to the spherical wave is expressed by the difference of the path differences  $\Delta_{S3} - \Delta_{S3}'$ . If we allow a phase difference of max. 1/2, we obtain a maximum permissible frequency limit flim 3 for which the spherical wave can be still regarded as a plane wave:

$$f_{\lim 3} \stackrel{\leq}{=} \frac{c}{4|\Delta_{s3} - \Delta_{s3'}|} = 17,000 \text{ MHz}$$
 (59)

In addition we have to fulfill the condition  $d > \lambda$ , i.e.,  $A_1$  has to be outside of the near-field of  $A_2$ . For small antennas we obtain the lower limit

$$f_{\lim 4} \stackrel{\geq}{=} \frac{c}{d} = 9.7 \text{ MHz}$$
 (60)

Neglecting the ground reflections, we may assume a plane wave for both experiment and computation, if the operation frequency f is between:

$$9.7 \text{ MHz} < f < 17,000 \text{ MHz}$$
 (61)

. . . .

The effect of the ground on the field homogeneity  $(E_0(h_1))$  along the vertical axis is described in ARRL [3] for some selected cases. Generally, the field strength  $E_0(h_1)$  oscillates around the free space value  $E_0(\infty)$ , with minima and maxima spaced about  $\lambda/2$ . Because the field homogeneity is of fundamental interest for the later experiments, the field homogeneity along the vertical axis of the TS (without TS) has been measured by varying h1 from  $\lambda/8$  up to  $\lambda$  with the following method :

Electrically small dipole antennas  $A_1$  (2h = 0.1 m) with autonomic RF-oscillators (see section 11.3.) were moved along the vertical axis with a special antenna manipulator (see section 11.2.). The field strength  $E_0(h_1)$  was measured with an LPD antenna  $A_2$  (see section 8.3.3.) and was calibrated to the field strength for  $h_1 = 1.2 m$ . The obtained data are presented in TABLE 26:

VERTICALLY POLARIZED FIELD AMPLITUDE VARIATION AT VARIABLE ANTENNA HEIGHTS ${\sf H}_1$						
FREQUENCY [MHz]	RELATIVE F h <sub>l</sub> = 0.8 m	REE-SPACE F h <sub>l</sub> = 1.0 m	FIELD STRENG h <sub>l</sub> = 1.2 m	5TH E <sub>O</sub> (h <sub>1</sub> ) h <sub>1</sub> = 1.4 m	IN DECIBELS h <sub>l</sub> = 1.6 m	AT HEIGHT h <sub>l</sub> = 1.8 m
65	+ 0.5	+ 0.0	+ 0.0	- 0.0	- 0.5	- 0.5
74	+ 1.0	+ 0.5	+ 0.0	- 0.5	- 1.0	- 1.0
101	+ 1.5	+ 0.5	+ 0.0	+ 0.0	+ 0.0	+ 0.5
164	- 1.5	- 1.0	+ 0.0	+ 0.5	+ 1.5	+ 2.0

TABLE 26 Homogeneity of the field along the vertical axis of the TS at vertical polarization. Values of  $E_0(h_1)$  related to  $E_0$  at  $h_1 = 1.2$  m. (Measuring data obtained by the standard test set-up,  $h_2 = 6.2$  m, d = 31 m).

The maximum field strength variation amounts to  $\pm 2 \text{ dB}$  in a full  $\lambda/2-h_1$  range (TABLE 26: f = 164 MHz). If we apply the equations (52,53),there is a good agreement; the changing factor is  $R_H$ , but because  $s < h_1 < s+L_B$ , the reflection angle is  $11.6^0 < \gamma < 14.8^0$  and thus  $|R_H| < 0.2$ . The variations of  $|E_0(h_1)|$  along the vertical axis are the same as given by equation 53 and amount to -2.0 to + 1.6 dB for all frequencies between 30 to 1000 MHz.

As a conclusion from this section 5.3. we may state:

- The test set-up allows antenna measurements with an accuracy of  $\pm 2 \text{ dB}$  in the proximity to ground at vertical polarization p<sub>2</sub> of A<sub>2</sub>.
- The TS must not be coupled with the ground.

#### 5.4.1. BODY MODELS

An adequate body model should respond to the investigated parameter like the original with a required accuracy but should be so simple that the phenomena can be computed with adequate effort.

The computation of the antenna-body model requires a stepwise increase of the complexity of the body model. In this study we start with a infinite model for two-dimensional computation and we end with a conducting model of human shape for three-dimensional computation.

The experiment with the antenna-body model requires a stepwise decrease of the complexity of the body model. The aim is twofold: at one side the body model should quantify the difference between original and model; on the other hand the model should allow the verification of the computation.

The development of the different body models is depicted in FIGURE 27:

BODY MODELS FOR EXPERIMENT

BODY MODELS FOR COMPUTATION



FIGURE 27 Body modelling for experiment and computation.

The different body models are specified as follows:

#### HUMAN TEST SUBJECT (SUB)

The same TS has been used for all experiments (dress without metallic parts) Body length : 1.68 m Averaged trunk diameter : 0.25 m Lateral diameter : 0.3 m Sagittal diameter : 0.2 m

# PHANTOM CYLINDER (PHA)

A cylindrical vessel has been constructed using a PVC tube filled with a kind of Ringer solution:

Cylinder length: 1.8 mDielectric properties of the RingerCylinder diameter: 0.25 msolution at 750 MHz:Wall thickness: 6.0 mmRelative permittivity  $\varepsilon_r$  : 50Relative  $\varepsilon_r$  of the wall:2.7ConductivityConductivity $\sigma$  : 1.25 S/m

The Ringer solution has been composed according to a prescription by GUY [38] and consisted of:

Glycol Ethandiol	носн <sub>2</sub> -сн <sub>2</sub> он	48.2 liters
Destilled water	H <sub>2</sub> 0	35.4 liters
Natrium Chloride	NaCL	1.86 kg

## METALLIC CYLINDER (MET)

A cylindrical vessel without caps has been constructed with copper plates: Cylinder length : 1.8 m Cylinder diameter : 0.25 m Wall thickness : 1.0 mm

# INFINITE METALLIC CYLINDER (IZYL)

For the two-dimensional (off-resonance) computation a rotational symmetric cylinder of infinite length and infinite conductivity has been assumed: Cylinder length : infinite Cylinder diameter : 0.25 m (nominal value, used as a parameter)

#### FINITE METALLIC CYLINDER (FZYL)

For the general three-dimensional computations a rotational symmetric cylinder with hemispherical caps and infinite conductivity has been assumed: Cylinder length : 1.8 m Cylinder diameter : 0.25 m Caps diameter : 0.25 m (other caps used as a parameter)

# METALLIC MAN MODEL (MANMOD 1 & MANMOD 2)

The sagittal and lateral projection of the human test subject(SUB) was used for modelling rotational symmetric, perfectly conducting body models. The shapes can be seen in FIGURE 28 and are described in section 16.2.4.


FIGURE 28 Body models used in experiments and computations

#### 5.4.2. ANTENNA-BODY MODELS FOR COMPUTATION

The off-resonance computation for frequencies above 200 MHz can be performed with a two-dimensional antenna-body model, using the body model IZYL. We assume a plane wave with  $\theta_i = 900$  which is scattered by a vertical cylinder of infinite length. The total field at an arbitrary point outside of the cylinder is the superposition of the incident field and the scattered field. This problem can be analytically solved by Bessel functions.

The general computation has to be performed with a three-dimensional antenna-body model, using the body models FZYL and MANMOD. We assume a plane wave with  $\theta_i = 80.8^{\circ}$  which is scattered by a rotational symmetric body. The total field at an arbitrary point outside of the cylinder is again the superposition of the incident field and the scattered field. This problem can only be solved by numerical methods, e.g., by the method of moments.

#### 5.4.3. ANTENNA-BODY MODELS FOR EXPERIMENT

The antenna-body model consists of the body models SUB, PHA and MET in a test set-up as shown in FIGURE 11. The parameters d (nominal 31 m),  $\theta_i$  (nominal 80.8°) and  $\gamma_B'(12-17°)$  should be kept as constant as possible during varying  $d_{at}$  and  $\phi$ . The antennas have to fulfill the requirements indicated in section 5.1.2..

# Leer - Vide - Empty

# 6. FUNDAMENTAL THEORY FOR THE COMPUTATION OF SCATTERING FROM CONDUCTING BODIES

#### 6.1. PURPOSE OF THE THEORY

The purpose of this theoretical section is to present all needed steps from the Maxwell equations up to the numerical solution concept. The theoretical background is described in BAGGENSTOS [5], VAN BLADEL [81], ANDRE-ASEN [1], KING and WU [50], HARRINGTON and MAUTZ [40] and BEVENSEE [10].

#### 6.2. PENETRATION DEPTH OF THE EM FIELD IN CONDUCTING BODIES

Most of the computations of fields neara conducting body are based on the assumption that the body is a perfect conductor and that there is no field inside the body. This assumption should be proven for our application.

If the conducting body and the observer are not in relative motion to each other and if the conducting body is an isotropic and linear medium, the Maxwell equations within the conducting body can be written as:

$$\overrightarrow{\operatorname{curl}} \overrightarrow{H} = \sigma_{\mathrm{m}} \overrightarrow{E} + \varepsilon_{\mathrm{m}} \frac{\partial \overrightarrow{E}}{\partial t}$$
 (101)

$$\operatorname{curl} \mathbf{E} = -\mu_{\mathrm{m}} \frac{\partial \mathbf{H}}{\partial \mathbf{t}} \tag{102}$$

$$div \vec{H} = 0 \tag{103}$$

div 
$$\dot{E} = \frac{\rho_m}{\varepsilon_m}$$
 (104)

In these equations  $\sigma_m$  is the conductivity of the medium,  $\varepsilon_m$  is the dielectric constant  $\varepsilon_r$  of the medium multiplied by the permittivity  $\varepsilon_0$  of vacuum,  $\mu_m$  is the relative permeability  $\mu_r$  of the medium multiplied by the permeability  $\mu_0$  of vacuum and  $\rho_m$  is the electric charge density in the medium. If static fields can be excluded, we obtain with (104) inserted in (101) :

div 
$$\overrightarrow{\operatorname{curl}} \overrightarrow{H} = 0 = \frac{\sigma_{m}}{\varepsilon_{m}}\rho_{m} + \frac{\partial\rho_{m}}{\partial t}$$
 and with  
 $\rho_{m} = \rho_{m_{0}} e^{-t/\tau}$  we obtain  
 $\tau = \frac{\varepsilon_{m}}{\sigma_{m}}$  (105)

Consider now a time interval T in which a charge of the fields E and H should be observed. If this time interval T (usually a fraction of the

$$\frac{I}{\tau} >> 1 \tag{106}$$

when the electric charge density  $\rho_{\rm m}$  in the medium can be assumed to be zero, since it disappears rapidly.

With the restriction (106) the Maxwell equations inside the medium are:

$$\overrightarrow{\operatorname{curi}} \overrightarrow{H} = \sigma_{\mathrm{m}} \overrightarrow{E} + \varepsilon_{\mathrm{m}} \frac{\partial \overrightarrow{E}}{\partial t}$$
(101)

$$\operatorname{curl} \vec{\mathsf{E}} = -\mu_{\mathrm{m}} \frac{\partial n}{\partial t} \tag{102}$$

$$\operatorname{div} \vec{H} = 0 \tag{103}$$

$$div \vec{E} = 0$$
 (107)

Using the curl function on (101) and (102) and the identity

$$-$$
 = curl curl - grad div

we obtain in a Cartesian coordinate system (x,y,z) the formula

$$\frac{\partial^2 F}{\partial x^2} + \frac{\partial^2 F}{\partial y^2} + \frac{\partial^2 F}{\partial z^2} + \mu_m \sigma_m \frac{\partial F}{\partial t} + \mu_m \varepsilon_m \frac{\partial^2 F}{\partial t^2} = 0$$
(108)

where F stands for  $E_x, E_y, E_z, H_x, H_y, H_z$ . The solution of this differential equation can be found in the case of a homogeneous body material with

$$F = F_0 e^{\left(-\vec{k}_m \cdot \vec{r} + j\omega t\right)} \quad \text{where} \qquad (109)$$

$$k_{m}^{2} = k_{mx}^{2} + k_{my}^{2} + k_{mz}^{2} = j_{\omega}\mu_{m}\sigma_{m} - \varepsilon_{m}\mu_{m}\omega^{2}$$
(110)

$$|\vec{k}_{m}| = \pm \sqrt{\omega \mu_{m} \sigma_{m}} \cdot \sqrt{j - \varepsilon_{m} \omega / \sigma_{m}}$$
(111)

With the restriction (106) and with the period time T of the applied wave we obtain the maximum frequency  $f_{max}$  at which the charge density within a well conducting medium (copper) may be still ignored:

$$T = \frac{2\eta}{\omega} ; \frac{\omega \varepsilon_{m}}{2 \pi \sigma_{m}} << 1$$

$$\varepsilon_{m} \sim \varepsilon_{0} = 8.86 \cdot 10^{-12} \text{ A} \cdot \text{s} \cdot \text{V}^{-1} \cdot \text{m}^{-1}$$

$$\sigma_{m} \sim 10^{7} \text{ A} \cdot \text{V}^{-1} \cdot \text{m}^{-1}$$

$$f_{max} \sim 10^{18} \text{ Hz} >> 1000 \text{ MHz} \qquad (112)$$

The wave factor  $\vec{k_m}$  from (111) becomes at frequencies well below  $f_{max}$ :

$$\vec{k}_{m} \sim \pm \sqrt{\pi} f \mu_{m} \sigma_{m}$$
 (1+j) (113)

Considering a wave travelling in positive direction r within an isotropic medium, we may write:

$$-\vec{k}_{m}\cdot\vec{r} = -k_{m}\cdot r$$

The attenuated wave is described by (113) inserted in equation (109):

$$F = F_{o} e^{-\sqrt{\pi}f\mu_{m}\sigma_{m}} \cdot r \cdot e^{j(\omega t - \sqrt{\pi}f\mu_{m}\sigma_{m}} \cdot r)}$$
(114)

 $F_0$  may be regarded as a component of  $\vec{E}$  or  $\vec{H}$  below the surface of the conducting body. The penetration depth  $\delta$ , defined as the distance which the propagating component will travel before the amplitude is decreased by a factor of  $e^{-1}$ , can be quantified as

$$\delta = \frac{1}{\sqrt{\pi f \mu_m \sigma_m}}$$
(115)

The lowest frequency in our investigation is 10 MHz. With a permeability  $\mu_m = \mu_0$  and with  $\delta$  we are now able to compute the maximum thickness of the field carrying surface layer of a well conducting medium. In a depth of 5 times  $\delta$  the fields are almost zero, and this layer thickness  $5\delta$  is :

$$5\delta = 5 \cdot \delta_{10 \text{ MHz}} \sim 0.2 \text{ mm} \text{ (copper)}$$
 (116)

This layer is much smaller than all dimensions of the body (see FIGURE 28) so that the field inside the conducting body models (copper) can be considered to be zero. This means that the computations on a perfectly conducting body will be also representative for an actual metallic body model as used for the experiments.

## 6.3. CHARGE- AND CURRENT-DENSITIES AT THE SURFACE OF A CONDUCTING BODY

#### 6.3.1. BOUNDARY-VALUE PROBLEM

We apply the Maxwell equations (101) and (102) on the fields near the surface of the medium and assume a thin layer 56 in which the fields may exist. The deeper regions are field-free as calculated with (114). In the computational model FIGURE 29 we assume that:

a) The E- and H- components within the test area  $\Delta A$  are finite b) The E- and H- components within  $\Delta I$  are constant in planes || surface.



FIGURE 29 Interface between vacuum and a well-conducting medium.

 $5\delta$  = thickness field-carrying layer  $\Delta I$  = length of the test area  $\Delta A$   $\Delta A$  = test area  $5\delta \cdot \Delta I$ s = integration path  $E_{to}$  = E-comp. tangential outside  $E_{ti}$  = E-comp. tangential inside  $H_{to}$  = H-comp. tangential outside  $H_{ti}$  = H-comp. tangential in layer  $E_{t5\delta}$  = E-comp. tangential in layer  $E_{n5\delta}$  = H-comp. tangential in layer  $H_{t5\delta}$  = H-comp. tangential in layer  $H_{n5\delta}$  = H-comp. normal in layer

The integration along s contains the parts :

$$F_{to} \cdot \Delta 1 - F_{t1} \cdot \Delta 1 + \int_{0}^{5\delta} F_{n5\delta}(s) ds + \int_{5\delta}^{0} F_{n5\delta}(s) ds$$

so that only  $F_{to} \cdot \Delta I$  is left. Thus, the integrations of (101) and (102) are:

$$E_{to} = -\frac{\mu_{m}}{\Delta I} \int_{\Delta A} \frac{\partial \vec{H}}{\partial t} \cdot d\vec{A} ; \vec{H} = H_{t5\delta}(s)$$
(117)

$$H_{to} = \frac{1}{\Delta 1} \left( \int_{\Delta A} \sigma_m \vec{E} \cdot d\vec{A} + \varepsilon_m \int_{\Delta A} \frac{\partial \vec{E}}{\partial t} \cdot d\vec{A} \right) ; \vec{E} = E_{t5\delta}(s)^{\prime}$$
(118)

For the integration of the Maxwell equations (103) and (104) we consider a subvolume  $\Delta V$  containing the field carrying layer (FIGURE 30) :



FIGURE 30 Interface between vacuum and medium (see also FIGURE 29 above).  $\Delta w = width of the subvolume \Delta V$  $\Delta V = subvolume \Delta A \cdot \Delta w$  $E_{no} = E-comp. normal outside$  $E_{ni} = E-comp. normal inside$  $H_{no} = H-comp. normal outside$  $H_{ni} = H-comp. normal inside$  In analogy to the assumptions a) and b) we assume the field conditions: c) The E- and H-components within the test subvolume  $\Delta V$  are finite d) The E- and H-components within  $\Delta 1 \cdot \Delta w$  are constant in planes || surface. The integrations of (103) and (104) over the layer thickness 5 $\delta$  lead to :

$$H_{no} = 0$$
(119)  
$$E_{no} = \frac{1}{\Delta 1 \cdot \Delta w} \int_{V} \frac{\rho_{m}}{\epsilon_{m}} \cdot dV$$
(120)

We assume now that there is an outer wave travelling parallel to the surface. The Poynting vector  $\vec{P}$  is therefore parallel to the surface, and with FIGURE 30 we obtain the only tangential component  $P_t$ :

$$P_t = E_{to} \cdot H_{no} - E_{no} \cdot H_{to} \neq 0$$
  
Since  $H_{no}$  is zero (119) it remains :

$$P_{t} = -E_{no} \cdot H_{to} \quad \text{and} \quad E_{no} \neq 0, \quad H_{to} \neq 0 \quad (121)$$

 $E_{no} \neq 0$  means that the right side of (120) is not zero. Since we have proven with (105) that there are no charges  $\rho_m$  inside the medium at low frequencies, then charges have to exist on the surface of the medium:

$$E_{no} = \frac{\sigma_{su}}{\epsilon_m}$$
;  $\sigma_{su}$  = surface charge density [C·m<sup>-2</sup>] (122)

 $H_{to} \neq 0$  means that the right side of (118) is not zero. Thus currents flowing in the outer layer have to exist which can be expressed by:

 $H_{to} = J_t$ ;  $\vec{J} = current density vector [A·m<sup>-1</sup>]$  (123) The direction of  $J_t$  is tangential to the surface and perpendicular to  $H_{to}$ 

Now we consider a wave perpendicular (normal) to the surface. If we neglect the losses in the medium, there is no wave entering the medium and thus the wave will be totally reflected. This means that the Poynting vector has no normal component  $P_n$  (see FIGURE 29) :

 $P_n = E_{to} \cdot H_{to} = 0$  (upper and lower case) (124) Either  $E_{to}$  or  $H_{to}$  could be zero in order to fulfill (124). If  $H_{to}$  would be zero, no wave could exist outside the medium since  $H_{no}$  is already zero as proven by (119). Because the wave cannot vanish outside the medium, the other field component  $E_{to}$  has to be zero, and we obtain:

$$E_{to} = 0$$
 (125)

The conclusion of (119,122,123,125) is: at the surface of a good conductor the  $\vec{E}$ -field has only a component normal to the surface, produced by a surface charge density  $\sigma_{su}$ , and the  $\vec{H}$ -field has only a component tangential to the surface, produced by the surface current density  $\vec{J}$  normal to  $\vec{H}$ .

This surface charge density  $\sigma_{su}$  and the current density  $\vec{J}$  are the origins of the scattered fields from a conducting body which is irradiated by a wave.

## 6.3.2. THE EFFECT OF THE SURFACE CURRENT DENSITY J

In FIGURE 31 a conducting body is depicted with the area element dS , containing the surface current density  $\vec{J}$  and the surface charge density  $\sigma_{su}$ :



FIGURE 31 Conducting body with  $\vec{J}$  and  $\sigma_{su}$  produced by an incident wave, and scattered parameters.

- 0 = Origin of coordinate system
- Q = Source point
- P = Observation point
- r' = Position of the source point
- r = Position of the observ.point  $\frac{\Phi^{scat}}{A} = Scattered electr.potential$   $\frac{A}{A} = Scattered mag.vector pot.$   $\frac{J}{J} = Surface current density$   $\sigma_{su} = Surface charge density$  dS = Area element on the body

The source is the current density  $\vec{J}$  in the area element dS positioned at  $Q(\vec{r}')$ . The effect is the scattered magnetic vector potential  $\vec{A}^{scat}$  at the remote observation point  $P(\vec{r})$ . Because the distance between the source and the observation point is comparable with the wavelength  $\lambda$ , a phase difference occurs which has to be treated by the time retardation t':

$$R = |\vec{R}| = |\vec{r} - \vec{r'}|$$
  
t' = t -  $\frac{R}{c}$ ; t = actual time (126)  
t' = time retardation

The scattered magnetic vector potential Ascat is defined as

$$\vec{A}^{\text{scat}}(\vec{r},t) = \frac{\mu_0}{4\pi} \int_{S} \vec{J}(\vec{r'},\vec{t'}) \frac{1}{R} dS \qquad (127)$$

$$\frac{\mathbf{j}}{\mathbf{j}}(\mathbf{r}',\mathbf{t}') = \mathbf{j}(\mathbf{r}') e^{\mathbf{j}\omega\mathbf{t}'}$$
(128)

and using the retardation (126) we insert (128) in (127) :

$$\vec{A}^{\text{scat}}(\vec{r},t) = \frac{\mu_0}{4\pi} \int \vec{J}(\vec{r}') e^{j\omega(t-R/c)} \frac{1}{R} dS$$

Introducing the wave factor k for free space (111)

$$\mathbf{k} = 2 \pi / \lambda = \omega / c \tag{129}$$

the exponential function becomes

$$e^{j\omega(t-R/c)} = e^{jk(t-c-R)}$$

and if we delete the time dependency we obtain finally :

$$\frac{\vec{A}^{\text{scat}}(\vec{r})}{S} = \frac{\mu_0}{4\pi} \int_{S} \vec{J}(\vec{r}') e^{-jkR} \frac{1}{R} dS \qquad (130)$$

#### 6.3.3. THE EFFECT OF THE SURFACE CHARGE DENSITY JSU

The treatment of the effect of the charge density  $\sigma_{SU}$  is analogous to 6.3.2.. The source is the charge density  $\sigma_{SU}$  in the area element dS positioned at  $Q(\vec{r})$ . The effect is the scattered electric (scalar) potential  $\phi^{SCat}$  at  $P(\vec{r})$ . The scattered electric potential  $\phi^{SCat}$  is defined as

$$\Phi^{\text{scat}}(\vec{r},t) = \frac{1}{41\varepsilon_0} \int_{S} \sigma_{\text{su}}(\vec{r'},t') \frac{1}{R} dS$$
(131)

and analogous to 6.3.2., we obtain with deleted time dependency :

$$\underline{\phi}^{\text{scat}}(\vec{r}) = \frac{1}{4 \pi \varepsilon_0} \int_{S} \sigma_{\text{su}}(\vec{r}') e^{-jkR} \frac{1}{R} dS$$
(132)

### 6.3.4. THE FIELD OUTSIDE OF THE CONDUCTING MEDIUM

With the magnetic vector potential  $\vec{A}^{\text{scat}}$  (130) and the electric potential  $\Phi^{\text{scat}}$  (132) the field strengths of the scattered  $\vec{E}^{\text{scat}}$  and  $\vec{H}^{\text{scat}}$  are:

$$\vec{H}^{scat} = \frac{1}{\mu_0} \frac{1}{curl} \vec{A}^{scat}$$
(133)

$$\vec{E}$$
scat =  $-\frac{\partial \vec{A} \operatorname{scat}}{\partial t} - \operatorname{grad}_{\Phi} \operatorname{scat}$  (134)

The total fields outside of the conducting medium are the superpositions of the incident and the scattered fields:

$$\vec{E}$$
 tot =  $\vec{E}$  inc +  $\vec{E}$  scat (135)

$$\vec{H}$$
 tot =  $\vec{H}$  inc +  $\vec{H}$  scat (130)

The index 'tot' means total outer field, 'inc' stands for the incident field and 'scat' stands for the scattered field.

The combinations of (134)(135) and (133)(136) lead to :

$$\vec{E} \operatorname{tot} = \frac{\partial \vec{A} \operatorname{scat}}{\partial t} - \operatorname{grad}_{\Phi} \operatorname{scat}_{+} \vec{E} \operatorname{inc}$$
(137)

$$\dot{H}^{tot} = \frac{1}{\mu_0} \underbrace{\overrightarrow{url}}_{scat}^{scat} + \dot{H}^{inc}$$
(138)

## 6.3.5. DETERMINATION OF THE CURRENT DENSITY J AND THE CHARGE DENSITY JSU

From equations (119) and (123) we know that at the margin of the conductor only the tangential H-components are existing. Since the coordinate system is notyet choosen, we define  $\vec{H}_t$  as a H-vector parallel to the test area dS.

$$\vec{H}_{t}^{tot} = \vec{H}_{t}^{inc} + \vec{H}_{t}^{scat}$$

$$\mu_{0}\vec{H}_{t}^{tot} = \mu_{0}\vec{H}_{t}^{inc} + \mu_{0}\vec{H}_{t}^{scat}$$
(139)

With the equation (123) and the relation:

$$\mu_{0}\vec{H}_{t}^{scat} = \text{Induction } \vec{B}^{scat} = \vec{curl}_{t}\vec{A}^{scat} = \vec{J}_{t}\mu_{0}$$

$$\vec{curl}_{t} = \text{tangential components of } \vec{curl}$$
(140)

we obtain finally

$$\vec{J} = \frac{\vec{curlt} \vec{A} \text{ scat}}{\mu_0} + \vec{H}_t^{\text{inc}}$$
(141)

Equation (141) is a transcendental differential equation for the two tangential components of  $\vec{J}$ .

Similar as above we obtain the normal component of  $\vec{E}$  at the margin of the conductor with equations (122) and (137) :

$$\vec{\text{grad}}_n, A_n, E_n = \text{normal component of } \vec{\text{grad}}, \vec{\text{A}}, \text{ and } \vec{\text{E}}$$
 (142)

$$E_{n}^{tot} = -\frac{\partial A_{n}^{scat}}{\partial t} - \overline{\operatorname{grad}}_{n} \Phi^{scat} + E_{n}^{inc}$$
(143)

By inserting (122) we obtain a transcendental differential equation :

$$\frac{\sigma_{su}}{\varepsilon_0} = -\frac{\partial \vec{A}_n \operatorname{scat}}{\partial t} - \overrightarrow{\operatorname{grad}}_n \Phi^{\operatorname{scat}} + \vec{E}_n^{\operatorname{scat}}$$
(144)

Equation (134) can be rearranged considering (109) :

$$\vec{E}^{\text{scat}}(\vec{r}) = -j\omega \vec{A}^{\text{scat}}(\vec{r}) - \overline{\text{grad}} \, \underline{\phi}^{\text{scat}}(\vec{r})$$
(145)

and with (135) and (125) are

$$\underline{\vec{E}}_{t}^{inc} = \underline{\vec{E}}_{t}^{tot} - \underline{\vec{E}}_{s}^{scat} \qquad \text{with } \underline{\vec{E}}_{t}^{tot} = 0 \qquad (146)$$

we obtain with  $\vec{A}_t$ ,  $\vec{\text{grad}}_t$  = tangential components of  $\vec{A}$  and  $\vec{\text{grad}}$ 

$$\vec{\underline{E}}_{t}^{scat}(\vec{r}) = j_{\omega}\vec{\underline{A}}_{t}^{scat}(\vec{r}) + \vec{\mathrm{grad}}_{t} \Phi^{scat}(\vec{r})$$
(147)

We may apply now (147) on a local coordinate system  $\vec{t} = \hat{t}_1, \hat{t}_2$  on the surface of the conducting body where the sources are located. Thus, we build the scalar product  $\vec{t}$  with  $\vec{E}^{inc}$  as follows :

$$\vec{t} \cdot \underline{\vec{E}}^{\text{inc}}(\vec{r}) = \vec{t} \cdot \left( j_{\omega} \underline{\vec{A}}^{\text{scat}}(\vec{r}) + g_{\text{rad}} \underline{\Phi}^{\text{scat}}(\vec{r}) \right)$$
(148)

By inserting (127) and (132) the equation (148) can be shaped in a form of an integral equation which does no more contain  $\vec{A}$  and  $\Phi$  :

$$\vec{t} \cdot \underline{\vec{E}}^{\text{inc}}(\vec{r}) = \frac{j\omega\mu_0}{4\pi} \vec{t} \cdot \left( \int_{S} \underline{\vec{j}}(\vec{r}') \frac{e^{-jkR}}{R} dS + \frac{1}{j\omega\mu_0\varepsilon_0} \overrightarrow{\text{grad}}_t \int_{S} \underline{\sigma}_{su}(r') \frac{e^{-jkR}}{R} dS \right)$$
(149)

The current density vector  $\vec{J}$  and the surface charge density  $\sigma_{su}$  are <u>not</u> independent of each other but are linked by the continuity equation:

$$\frac{\partial}{\partial t}\rho(\vec{r},t) = -\operatorname{div}\vec{j}(\vec{r},t)$$
(150)

In our case the continuity equation becomes

$$\frac{\partial}{\partial t}\sigma_{su}(\vec{r},t) = -\operatorname{div}_{t}\vec{J}(\vec{r},t)$$
(151)

By inserting (109) and (122) in equation (151) we obtain

$$\underline{\sigma}_{su}(\vec{r}') = -\frac{1}{j\omega} \operatorname{div}_{t} \underline{j}(\vec{r}')$$
(152)

The surface current density  $\underline{\sigma}_{su}$  is now expressed by a function of  $\underline{J}$  for each position r' of the area element dS. With the free-space wave factor

$$k^2 = \omega^2 \varepsilon_0 \mu_0 \tag{153}$$

we combine (149) with (152). We obtain two equations for the tangential components of the incident E-field as functions of  $\vec{j}$  only :

$$\vec{t} \cdot \vec{\underline{E}}^{inc}(\vec{r}) =$$

$$\frac{j\omega\mu_{0}}{4\pi} \stackrel{\ddagger}{t} \cdot \left( \int_{S} \stackrel{\ddagger}{\underline{J}(\vec{r}')} \frac{e^{-jkR}}{R} dS + \frac{1}{k^{2}} \stackrel{\ddagger}{grad}_{t} \int_{S} \frac{e^{-jkR}}{R} div_{t} \stackrel{\ddagger}{\underline{J}(\vec{r}')} dS \right)$$
(154)

The general scattering problem is a very complicated mathematical boundary - value problem which so far has resisted exact analytical treatment except in such special cases as the sphere and the infinite cylinder. For all other cases only numerical methods can be applied.

#### 6.4. SCATTERING FROM BODIES OF REVOLUTION WITH THE METHOD OF MOMENTS

#### 6.4.1. GENERALIZED NETWORK PARAMETERS FOR BODIES OF REVOLUTION

The integral equation (154) can be solved numerically by the method of moments (HARRINGTON [41]). In the years 1968 and 1969 a theory and some computer programs have been developed for the scattering from bodies of revolution by HARRINGTON AND MAUTZ [40]. The following is a summary of this report as it applies to the present problem. The first step is the determination of the generalized network parameters:

The equation (154) can be rewritten in the simpler form:

$$\vec{E}_{t}^{inc} = L(\vec{J})$$
(155)

where  $L(\vec{J})$  is the integro-differential operator and which corresponds to the right side of equation (154).  $L(\vec{J})$  is also similar to (148) :

L(Ĵ)	= [jω Ă	+	$\overrightarrow{\operatorname{grad}} \Phi]_{t}$	(156)
Ă	= Åscat		(see (130))	
Φ	$= \Phi$ scat		(see (132))	

A solution of (155) gives the current  $\vec{J}$  on the surface S. Usually we are interested in some functional of  $\vec{J}$ , which can be computed once  $\vec{J}$  is known.

In order to effect a solution by the method of moments, let the inner product be defined as : (see definition of <f,g> in HARRINGTON [41])

$$\langle \vec{W}, \vec{J} \rangle = \int_{S} \vec{W} \cdot \vec{J} \, dS$$
 (157)

Both  $\vec{W}$  and  $\vec{J}$  are tangential vectors on S. A set of expansion functions  $\{\vec{J}_i\}$  is next defined, and the current on S is approximated by

$$\vec{j} = \sum_{j} I_{j} \vec{j}_{j}$$
(158)

I, are constants to be determined. Equation (158) is substituted into (155) which, because of the linearity of L, reduces to

$$\vec{E}_{t_{inc}} = \sum_{j} I_{j} L(\vec{J}_{j})$$
(159)

A set of testing functions  $\{\vec{W}_i\}$  is defined, and the inner productof (159) with each  $\vec{W}_i$  is taken. The result is

$$\sum_{j} I_{j} \langle \vec{W}_{i}, L \vec{J}_{j} \rangle = \langle \vec{W}_{i}, \vec{E}_{inc} \rangle \qquad i = 1, 2, 3, ...$$
(160)

The index 't' has been dropped from  $\vec{E}_{inc}$  because the inner product involves only tangential components. We now define the generalized network matrices

$$[Z] = [\langle W_i, L(\hat{J}_j) \rangle]$$
(161)

$$[V] = [\langle \dot{W}_{i}, \dot{E}_{inc} \rangle]$$
(162)

$$[I] = [I_i]$$
(163)

and rewrite the set (160) as

$$\begin{bmatrix} \mathsf{Z} \end{bmatrix} \begin{bmatrix} \mathsf{I} \end{bmatrix} = \begin{bmatrix} \mathsf{V} \end{bmatrix} \tag{164}$$

[Z] is the generalized impedance matrix, and  $[Y] = [Z]^{-1}$  is the generalized admittance matrix. The inverse of (164)

$$\begin{bmatrix} I \end{bmatrix} = \begin{bmatrix} Y \end{bmatrix} \begin{bmatrix} V \end{bmatrix}$$
(165)

gives the coefficients  $I_j$  of the current expansion (158) and hence is an approximate solution of the problem.

The impedance elements of (161) are explicitly using (156) and (157) :

$$Z_{ij} = \int_{S} W_{i} \cdot (j\omega \dot{A}_{j} + grad \phi_{j}) dS \qquad (166)$$

The subscript j denotes that  $\vec{A}_j$  and  $\Phi_j$  are potentials due to  $\vec{j}$  and  $\sigma_{su}$ . In order to match the equations in 6.3. to those used in [40] we replace

$$\vec{\text{grad}} \Phi = \vec{\nabla} \Phi$$

$$\vec{\text{div}} \vec{\text{J}} = \vec{\nabla} \cdot \vec{\text{J}}$$
(167)

Regarding  $\bar{W}_{1}$  as a current density from  $\sigma_{SU}$ , we rewrite (152) with (167)

$$\sigma_{su_i} = \frac{-1}{j\omega} \vec{\nabla} \cdot \vec{W}_i$$
 (169)

Now (166) can be written as

$$Z_{ij} = j\omega \begin{cases} (\vec{W}_i \cdot \vec{A}_j + \sigma_{su_i} \Phi_j) \end{cases}$$
(170)

Equation (170) is more convenient for computation than (166) or (154).

So far the discussion has been for an arbitrary conducting body. Now we restrict considerations to the surface S generated by revolving a plane curve about the z-axis. The surface and the coordinate systems are shown in FIGURE 32:



FIGURE 32 Body of revolution and coordinate systems.

S ≈ surface of the body

NP = number of points describing
 the generating curve

RH = radius parameter of gen.curve

ZH = height parameter of gen.curve

 $t_N = tangent unit elements, N = (\frac{NP-1}{2})$ 

- t = length variable along the curve generating S
- $\rho$  = radius of a point on S
- $\phi$  = angle of a point on S
- z = height of a point on S

 $\vec{u}_+$  = local tangential coord. syst.

 $\dot{\vec{u}}_{\phi}$  = local tangential coord. syst. (suffix ' = source point)

The body of revolution is described by the generating curve given by the curve parameters RH and ZH. The variable 't' is now a tangential length variable along the curve generating the surface S. We desire the expansion (158) to be general enough to approximate an arbitrary  $\vec{J}$  on S. Hence, independent sets of functions are defined as

$$\vec{j}_{mj}^{t} = \vec{u}_{t} f_{j}(t) e^{jm\phi}$$
(171)

$$\vec{J}_{mj}^{\phi} = \vec{u}_{\phi} f_{j}(t) e^{jm\phi}$$
(172)

where  $u_t$  and  $u_{\phi}$  are unit vectors t-directed and  $\phi$ -directed, respectively. The  $f_j(t)$  has been chosen in both sets to be the same, but it is not necessary to do so [40]. The current expansion (158) now becomes

$$\vec{J} = \sum_{m,j} (I_{mj}^{t} \vec{J}_{mj}^{t} + I_{mj}^{\phi} \vec{J}_{mj}^{\phi})$$
(173)

For testing functions, choose

$$\vec{W}_{ni}^{t} = \vec{u}_{t} f_{i}(t) e^{-jn\phi}$$
(174)

$$\vec{h}_{ni}^{\phi} = \vec{u}_{\phi} f_{i}(t) e^{-jn\phi}$$
(175)

which differ from (171,172) only in the sign of the exponent. The  $\vec{W}_n$  are orthogonal to  $\vec{J}_m$ ,  $m \neq n$ , over 0 to 21 on  $\phi$ , and also to  $L(\vec{J}_m)$  (the field from  $\vec{J}_m$ ). Hence, all impedance elements are zero except those for which m = n, and each mode n can be treated separately. This is the major simplification introduced by the rotational symmetry of the body. For the computation of non-rotational symmetric bodies, such as shown in FIGURE 18, the further procedure had to be already changed here.

The use of (171)(172)(174) and (175) to evaluate the elements of (170) results in the partitioned matrix equation

$$\begin{bmatrix} [Z_n tt] & [Z_n t\phi] \\ [Z_n^{\phi t}] & [Z_n^{\phi \phi}] \end{bmatrix} = \begin{bmatrix} [I_n^t] \\ [I_n^{\phi}] \end{bmatrix} = \begin{bmatrix} [V_n^t] \\ [V_n^{\phi}] \end{bmatrix}$$
(176)

Here the elements of the Z submatrices are

$$(Z_n^{tt})_{ij} = \langle \vec{W}_{ni}, L(\vec{J}_{nj}) \rangle$$
, etc. for t $\phi, \phi t$  and  $\phi \phi$  (178)

The elements of the I submatrices are the coefficients in (173), and the elements of the V submatrices are

Note that, for N terms in the Fourier series of  $\varphi,$  there are N sets of matrix equations (176).

The solution to (176) can be also written in partitioned form as

$$\begin{bmatrix} [\mathbf{I}_{n}^{\mathsf{t}}] \\ [\mathbf{I}_{n}^{\Phi}] \end{bmatrix} = \begin{bmatrix} [\mathbf{Y}_{n}^{\mathsf{tt}}] & [\mathbf{Y}_{n}^{\mathsf{t}\Phi}] \\ [\mathbf{Y}_{n}^{\Phi\mathsf{t}}] & [\mathbf{Y}_{n}^{\Phi\Phi}] \end{bmatrix} \begin{bmatrix} [\mathbf{V}_{n}^{\mathsf{t}}] \\ [\mathbf{V}_{n}^{\Phi}] \end{bmatrix}$$
(180)

The Y submatrices must in general be obtained after inversion of the entire Z matrix and are not the inverse of the corresponding Z submatrices. However, as shown in [40], the -n mode matrices are related to the +n mode matrices, so that only the  $n \ge 0$  mode matrices need to be inverted.

Finally, for an explicit solution one has to choose the t expansion functions  $f_i(t)$ . A triangle expansion function gives a piecewise-linear approximation which converges rapidly:

$$f_{i}(t) = \frac{1}{\rho} T(t - t_{i})$$
(181)

$$T(t) = 1 - |t|$$
 for  $|t| < 1$  and 0 for  $|t| > 1$  (182)

When using these functions, distance and frequency are scaled so that the  $t_i$ 's are one unit apart. The generating curve is determined by NP body points (FIGURE 32). There are (NP-1)/2 = N tangential units, and if one triangle function covers 2 units, there are N-1 peaks at 1,2,3...N-1.

#### 6.4.2. IMPEDANCE MATRICES

#### 6.4.2.1. EVALUATION OF THE IMPEDANCES

The generalized impedances for a body of arbitrary shape is the integral over all source points (dS') and the integral over all field points (dS) :

$$Z_{ij} = \int_{S} dS' \int_{S} dS \left[ j_{\omega\mu_0} \vec{W}_i \cdot \vec{J}_j + \frac{1}{j_{\omega\epsilon_0}} (\vec{\nabla} \cdot \vec{W}_i) (\vec{\nabla} \cdot \vec{J}_j) \right] \frac{e^{-jKR}}{4\pi R}$$
(183)

For bodies of revolution the integrals have the t-elements 0 to N and the  $\boldsymbol{\varphi}\text{-}$ 

elements 0 to 21. The radius R (see FIGURE 32) can be expressed by  $\vec{r} - \vec{r'}$ :

$$R = \left[ \rho^{2} + \rho^{\prime 2} - 2\rho\rho^{\prime} \cos(\phi - \phi^{\prime}) + (z - z^{\prime})^{2} \right]^{V_{2}}$$
(184)

The inner products in (183) are of the type

$$\vec{\nabla} \cdot \vec{J} = \frac{1}{\rho} \frac{\partial}{\partial t} (\rho J_t) + \frac{1}{\rho} \frac{\partial}{\partial \phi} (J_{\phi})$$
 (185)

Four types of impedances are defined by (178). To evaluate them, we use (171)(172) and (174)(175) to obtain the  $\vec{W} \cdot \vec{J}$  terms in (183) as

$$\vec{w}_{ni}^{p} \cdot \vec{j}_{q} = e^{jn(\phi-\phi')} f_{i}(t') f_{j}(t) \vec{u}_{p'} \cdot \vec{u}_{q}$$
(186)

where p and q represent permutations of t and  $\phi$ . The unit vector inner products in terms of the body coordinates defined by FIGURE 32 are

$$\vec{u}_{t}' \cdot \vec{u}_{t} = \sin v \sin v' \cos(\phi - \phi') + \cos v \cos v'$$
$$\vec{u}_{t}' \cdot \vec{u}_{\phi} = -\sin v' \sin(\phi - \phi')$$
$$\vec{u}_{\phi}' \cdot \vec{u}_{t} = \sin v \sin(\phi - \phi')$$
$$\vec{u}_{\phi}' \cdot \vec{u}_{\phi} = \cos(\phi - \phi')$$

Here v is the angle between the t direction and the z axis, being positive if  $\vec{u}_t$  points away from the z-axis. Changing ( $\phi$ - $\phi$ ') to a new variable, and expressing the sine and cosine terms of (187) as exponentials, one  $\phi$  integration of (183) can be performed. The remaining  $\phi$  integration defines the Green's function :

$$g_n = \int_0^{\pi} d\phi \, \frac{e^{-jkR_o}}{R_o} \, \cos n\phi \qquad (188)$$

where  $R_0$  is given by (184) with  $\phi'=0$ . With  $f_i$  given by (181), the resultant expression for the impedance elements (178) are (only  $(Z_n^{tt})_{i,i}$  shown):

$$(Z_n^{tt})_{ij} = \int_0^N dt' \int_0^N dt \left[ j\omega\mu_0 T(t'-i) T(t-j)(\sin v \sin v' \frac{g_{n+1} + g_{n-1}}{2} + \cos v \cos v' g_n) + \frac{1}{j\omega\varepsilon_0} T'(t'-i) T'(t-j) g_n \right]$$
(189)

Here T' is the derivative of the triangle function

$$\begin{array}{rl} 1, & -1 < t < 0 \\ T'(t) &= -1, & 0 < t < 1 \\ 0, & |t| > 1 \end{array} \tag{190}$$

The integrations of (189) involve many different integrands, and to reduce the number of integrations the following approximations are made. For the t-integration, the T function is approximated by four pulses of amplitude 1/4, 3/4, 3/4, 1/4, and the derivative of T (denoted here as T') is represented exactly by four pulses of amplitude 1, 1, -1, -1. The functions  $\rho$ , sin v, and cos v are assumed constant over each pulse, equal to their values at the midpoints of the pulses. For the t' integration, the T function is approximated by four impulse functions of strengths 1/8, 3/8, 3/8, 1/8, and the derivative of T is approximated by four impulse functions of strengths 1/2, 1/2, -1/2, -1/2 ( $T_1$ ,  $T_2$ ,  $T_3$ ,  $T_4$  and  $T_1'$ ,  $T_2'$ ,  $T_3'$ ,  $T_4'$ ).

The midpoints of the pulses and the pulse Green's functions are defined :

$$t_p = i + \frac{p-2.5}{2}; t_q = j + \frac{q-2.5}{2}$$
 (191)

$$G_{n} = 2 \int_{i+\frac{q-2}{2}}^{j+\frac{q-2}{2}} dt \int_{0}^{\pi} d\phi \, \frac{e^{-jkR_{p}}}{R_{p}} \cos n\phi \qquad (192)$$

$$R_{p} = \left[\rho^{2} + \rho_{p}^{2} - 2\rho\rho_{p}\cos\phi + (z - z_{p})^{2}\right]^{V_{2}}$$
(193)

In terms of these definitions and approximations, the matrix elements of (189) reduce to:

$$(Z_{n}^{tt})_{ij} = \sum_{p=1}^{4} \sum_{q=1}^{4} \left[ j\omega\mu_{0} T_{p}T_{q} (\sin v_{p} \sin v_{q} \frac{G_{n+1} + G_{n-1}}{2} + \cos v_{p} \cos v_{q} G_{n}) + \frac{1}{j\omega\varepsilon_{0}} T_{p} T_{q} G_{n} \right]$$
(194)  
$$(Z_{n}^{t\phi})_{ij} = \sum_{p=1}^{4} \sum_{q=1}^{4} \left[ -\omega\mu_{0} T_{p}T_{q} \sin v_{p} \frac{G_{n+1} - G_{n-1}}{2} + \frac{n}{\omega\varepsilon_{0}} T_{p} \frac{T_{q}}{\rho_{q}} G_{n} \right]$$
(194)  
$$(Z_{n}^{\phi t})_{ij} = \sum_{p=1}^{4} \sum_{q=1}^{4} \left[ +\omega\mu_{0} T_{p}T_{q} \sin v_{q} \frac{G_{n+1} - G_{n-1}}{2} - \frac{n}{\omega\varepsilon_{0}} \frac{T_{p}}{\rho_{p}} T_{q} G_{n} \right]$$
( $Z_{n}^{\phi \phi})_{ij} = \sum_{p=1}^{4} \sum_{q=1}^{4} \left[ j\omega\mu_{0} T_{p}T_{q} \frac{G_{n+1} - G_{n-1}}{2} + \frac{n^{2}}{j\omega\varepsilon_{0}} \frac{T_{p}}{\rho} \frac{T_{q}}{\rho} G_{n} \right]$ 

Here  $\rho_p$ ,  $v_p$ ,  $\rho_q$ ,  $v_q$  are the  $\rho$  and v evaluated at  $t_p$  and  $t_q$  respectively. Finally, the  $G_n$  (192) was prepared for the numerical computation by dividing the integration interval 0 to  $\pi$  into M equal intervals. In the actual computer program the  $G_n$ 's were further divided by k in order to make them insensitive to the absolute size of the body. In addition , the  $T_p$  and the  $T'_p$  were modified, as can be seen in the description of the computer program A by HARRINGTON and MAUTZ [40].

#### 6.4.2.2. LIMITATIONS OF THE NUMERICAL COMPUTATIONS OF THE IMPEDANCES

The computer program HARRA is based on the above theory. It will be discussed later in section 10.2.. The purpose of the program HARRA is to compute the four Z-matrices (194) and its inverse four Y-admittance matrices (164). The Y-matrices for each mode n from 0 to  $n_{nn}(192)$  are stored in a file for the later use by following programs computing the scattering.

Because the solution for the Y-matrices are obtained by matrix inversions, the matrix size has to be limited to a reasonable value in order to save computation time and storage capacity. The body shape will be approximated by 20 tangents (N=20), thus one obtaines 4 matrices of the size 19 x 19 for each mode. The integral intervals of the  $G_n$ 's of 0 to ¶ will be divided into 20 subintervals (M=20).

With these specifications acurate Y-matrices up to the mode n=6 can be obtained for a conducting model of man up to frequencies of 400 MHz. The proof for these statments will be presented in the program description in section 10.3.1.

#### 6.4.3. MEASUREMENT MATRICES

Any linear measurement of the field from the current  $\vec{J}$  on the body S can be expressed as a linear functional of  $\vec{J}$ , that is

measurement = 
$$\int_{S} \vec{E}^{r} \cdot \vec{J} dS$$
 (195)

where  $\vec{E}^r$  is a known function. For a moment solution, the current is given by a superposition J =  $\Sigma I_j \vec{J}_j$ , and (195) reduces to

$$measurement = [R] [I]$$
(196)

where [I] is the matrix (163) and [R] is a measurement row matrix:

$$[R] = [\langle \vec{J}_{j}, \vec{E}^{r} \rangle]$$
(197)

[R] is similar to the excitation matrix [V] (162), and with the matrix solution (165) substituted in (196), one has

$$measurement = [R] [Y] [V]$$
(198)

For bodies of revolution, the expansion for  $\vec{J}$  can be separated into t and  $\phi$  directed components, according to (173). It is then convenient to partition [R] into t and  $\phi$  component terms as

$$(R_n^t)_i = \langle \vec{j}_{ni}^t, \vec{E}^r \rangle$$

$$(R_n^\phi)_i = \langle \vec{j}_{ni}^\phi, \vec{E}^r \rangle$$

$$(199)$$

The analogous partition for excitation [V] is given by (179). Now one can rewrite (198) in the partitioned form as

measurement = 
$$\begin{bmatrix} \begin{bmatrix} R_{n}^{t} \end{bmatrix} & \begin{bmatrix} R_{n}^{\phi} \end{bmatrix} \begin{bmatrix} \begin{bmatrix} Y_{n}^{tt} \end{bmatrix} & \begin{bmatrix} Y_{n}^{t\phi} \end{bmatrix} \begin{bmatrix} \begin{bmatrix} v_{n}^{t} \end{bmatrix} \\ \begin{bmatrix} Y_{n}^{\phi t} \end{bmatrix} & \begin{bmatrix} Y_{n}^{\phi\phi} \end{bmatrix} \begin{bmatrix} \begin{bmatrix} v_{n}^{t} \end{bmatrix} \end{bmatrix}$$
(200)

where the Y submatrices are obtained after the Z matrix is inverted and are not the inverses of the corresponding Z submatrices.

An important special case is that of radiation field measurements. HAR-RINGTON [41] has shown that the radiation field from currents  $\vec{j}$  on S is given by (201) :

$$\vec{E} \cdot \vec{u} = \frac{-j\omega\mu_0}{4\pi} e^{-jkr} [R] [I]$$
(201)

where the elements of [R] are given by (197) with

$$\vec{E}^{r} = \vec{u} e^{-jk\vec{r}\cdot\vec{r}}$$
(202)

This is a unit plane wave with polarization vector  $\vec{u}$  and propagation vector  $\vec{k}$ . An arbitrary plane wave is a superposition of two orthogonal components :

$$\vec{E}_{\theta}$$
 = 'vertical' polarization (see 5.3.2.) (203)  
 $\vec{E}_{.}$  = 'horizontal' polarization (see 5.3.2.) (204)

Hence, one can treat the general case as two applications of (202), one for  $\vec{u} = \vec{u}_{\theta}$  and the other for  $\vec{u} = \vec{u}_{\phi}$ . To distinguish between the two cases let us denote the measurement matrices as follows: (205),(206)

θ-polarized case	<pre></pre>	
$(R_n^{t\theta})_i = \langle \tilde{J}_{ni}^t, \tilde{E}_{\theta}^r \rangle$	$(R_n^{t\phi})_i = \langle \hat{J}_{ni}^t, \hat{E}_{\phi}^r \rangle$	θ: (205)
$(R_n^{\phi\theta})_i = \langle \hat{J}_{ni}^{\phi}   \hat{E}_{\theta}^r \rangle$	$(R_n^{\phi\phi})_i = \langle \hat{J}_{ni}^{\phi}   \hat{E}_{\phi}^{r} \rangle$	φ: (206)

The excitation matrices can now be evaluated as follows. Let

$$\vec{E}_{\theta}^{r} = \vec{u}_{\theta}^{r} e^{jk(\rho \sin \theta_{r} \cos \phi + z \cos \theta_{r})}$$
(207)

where  $\theta_r$  and  $\phi_r$  = 0 are the angles to the field point of measurement. The inner products required in (205) are given by

$$\vec{u}_{t} \cdot \vec{u}_{\theta}^{r} = \cos \theta_{r} \sin v \cos \phi - \sin \theta_{r} \cos v$$

$$\vec{u}_{\phi} \cdot \vec{u}_{\theta}^{r} = -\cos \theta_{r} \sin \phi$$
(208)

Using the integral formula for Bessel functions

••

...

$$J_{n}(\rho) = \frac{j^{n}}{2\pi} \int_{0}^{2\pi} e^{-j\rho \cos \phi} e^{-jn\phi} d\phi \qquad (209)$$

one can evaluate the  $\boldsymbol{\phi}$  integrations in (205), obtaining

$$(R_{n}^{t\theta})_{i} = 2\pi j^{n+1} \int_{0}^{N} dt \rho f_{i}(t) e^{jkz \cos \theta} r \left[ \cos \theta_{r} \sin v \frac{J_{n+1} - J_{n-1}}{2} + j \sin \theta_{r} \cos v J_{n} \right]$$
(210)  
$$(R_{n}^{\phi\theta})_{i} = -2\pi j^{n+1} \int_{0}^{N} dt \rho f_{i}(t) e^{jkz \cos \theta} r \cos \theta_{r} \frac{J_{n+1} + J_{n-1}}{2j}$$

and one can similarly evaluate the  $\varphi$  integration for the  $\varphi$  case :

$$(R_{n}^{t\phi})_{i} = 2\pi j^{n+1} \int_{0}^{N} dt \rho f_{i}(t) e^{jkz \cos \theta} r \sin v \frac{J_{n+1} + J_{n-1}}{2j}$$

$$(R_{n}^{\phi\phi})_{i} = 2\pi j^{n+1} \int_{0}^{N} dt \rho f_{i}(t) e^{jkz \cos \theta} r \frac{J_{n+1} - J_{n-1}}{2}$$

$$(211)$$

where J<sub>n</sub> is

$$J_{n} = J_{n} (k\rho \sin \theta_{r})$$
 (212)

For computations, the  $\rho f_i(t)$  in (210) and (212) were the triangle functions (182). For plane-wave excitation of the body the excitation matrix [V] is

$$(V_n^{pq})_i = (R_{-n}^{pq})_i$$
 (213)

where pqrepresents t0,  $\phi0$ , t $\phi$ , or  $\phi\phi$ . Equation (213) means that the V<sub>i</sub> are given by (210) and (211) with n replaced by -n and 0<sub>r</sub> by 0<sub>t</sub>.

#### 6.4.4. GENERAL PLANE-WAVE SCATTERING

The radar scattering problem consists of a plane wave incident on a scattering body, plus measurement of the far-zone scattering. Because we are interested in the measurement of the near-zone scattering, only a few details should be discussed here. The solution of this radar scattering problem requires the determination of the t-directed and  $\phi$ -directed surface currents. In program B the currents are computed and printed for the axial incidence of a plane wave, and in program D the currents are partly computed for an oblique incidence (HARRINGTON and MAUTZ [40])

Using the formulas (127),(132) and the continuity equation (152), the scattered field in a point near the body (see FIGURE 31) could be computed if  $J^{t}$  and  $J^{\phi}$  are known at the peaks of each triangle function. A special program E (not enclosed in this book) has been prepared to compute these currents for arbitrary wave incidence. In principle, the current densities for each mode are obtained by program D, have to be summed over all needed modes and transferred in program B for linearization and printing.

#### 6.4.5. NEAR-FIELD COMPUTATION

#### 6.4.5.1. METHOD OF SOLUTION

The solution of our actual problem, that is the computation of the scattered field and the incident field <u>near</u> the conducting body became possible with the extension of the theory by BEVENSEE [10] (see FIGURE 33) :



FIGURE 33 Coordinates and nomenclature for near-field computations at oblique incidence of a plane wave. (Source: BEVENSEE [10])

The method of solution (BEVENSEE [10] is an extension of the method by HARRINGTON and MAUTZ [40] discussed above in sections 6.4.1. to 6.4.3.. The formulas (155) up to (163) can be summarized as follows:

The operation equation

$$\vec{E}_{t}^{\text{inc}} = L(\vec{J}), \quad L(\vec{J}) = (j\omega\vec{A} + \vec{\nabla}\Phi)_{t}$$
(214)

is solved by expanding

$$\vec{J} = \sum_{j} I_{j} \vec{J}_{j}$$
(215)

for the n<sup>th</sup> azimuthally varying mode by a set of functions :  $\vec{J}_j = \vec{u} f_j e^{jn\phi}$ ( $\vec{u}$  = unit vector in the t- or  $\phi$ -direction on the surface of the body) and by a set of functionals  $\vec{W}_j = \vec{u} f_j e^{-jn\phi}$  in succession :

$$\langle \vec{W}_{i}, \vec{E}_{t}^{inc} \rangle = \sum_{j} \langle \vec{W}_{i}, L(\vec{J}_{j}) \rangle I_{j}, i = 1, 2, ... N$$
 (216)

The < > denotes a spatial integral over the localized range of the W<sub>i</sub>-function. Defining  $Z_{ij}$  as  $\langle \vec{W}_{i}, L(\vec{J}_{j}) \rangle$  one obtains the network representation of (214) as :

$$V_{i} = \sum_{j} Z_{ij} I_{j}, \quad V_{i} = \langle \vec{W}_{i}, \vec{E}_{t}^{inc} \rangle$$
(217)

Analogously, <u>a test segment</u>, subscript T , is inserted to measure the scattered field  $\vec{E}^{scat} = L(\vec{J})$  as (BEVENSEE [10])

$$\langle \vec{W}_{T}, \vec{E}^{\text{scat}} \rangle = \langle \vec{W}_{T}, -L(\vec{J}) \rangle = -\sum_{j} \langle \vec{W}_{T}, L(\vec{J}_{j}) \rangle I_{j}$$
 (218)

Defining the measurement matrix [ZM] as  $(ZM)_{1j} = -\langle W_{T}, L(J_{j}) \rangle$ , where the index '1' means test segment 1 and 'j' the source element (166), one has

$$\langle \vec{W}_{T}, \vec{E}^{scat} \rangle = \sum_{j} (ZM)_{1j} I_{j}$$
 (219)

and the electric field in the T-direction along the test segment is approximately

$$E_{T}^{\text{scat}} = \frac{1}{\langle \overline{W}_{T} \rangle} \sum_{j} (ZM)_{jj} I_{jj}$$
(220)

As the area  $\langle \vec{W}_T \rangle$  of the test segment approaches zero,  $\vec{E}_T^{scat}$  approaches the correct scattered field value.

The computation of the fields at a point near the conducting body (see FIGURE 33) involves the following steps:

First, the [Z] matrices (194) and its inverse [Y] matrices have to be computed for all needed modes. This computation is only dependent on the dimensions of the body and the applied frequency but not on the irradiation angle  $\theta_i$  or the polarization.

Second, the [R] matrices (210) and (211) have to be computed for a specific incident wave in order to obtain the exitation matrices [V] and the coefficient [I] of the current expansion (164).

Third, at a given test point a test segment is defined by five points and four sections of equal length DTEST (FIGURE 33). This test segment is first positioned along the spherical radius vector  $\vec{a}_r$  (for IT = 1, as shown in FIGURE 33) to measure the  $\vec{a}_r$ - and  $\vec{a}_{\phi}$ -components of the  $\vec{E}$  field. It is then positioned along the spherical  $\vec{a}_{\theta}$ -vector (for IT = 2) to measure the  $\vec{a}_{\theta}$ - and  $\vec{a}_{\phi}$ -components of the  $\vec{E}$  field. The two measured  $E_{\phi}$  fields approach each other as the test segment length 4DTEST approaches zero; their discrepancy gives an estimate of the accuracy obtained with this segment.

The total E-field  $\vec{E}^{tot}$  along an  $\vec{a}$ -vector is the superposition of  $\vec{E}^{inc}$ and  $\vec{E}^{scat}$  (see (135)). The  $\vec{E}^{inc}$ - components are obtained by using the subroutine PLANE (which computes the [R] and [V] matrices, see (213)) applied a second time on the test segment. The  $\vec{E}^{scat}$ -components are computed with a new subroutine NEARZ which is well described by BEVENSEE [10] The matrix elements of [ZM] (220) are determined similar to the elements of [Z] (194) but with an altered Green's function  $G_{nT}$ . If we denote a surface element on the body as 2  $y_0 \cdot x_0$  (2  $y_0 = 2$  tangent units as in FIGURE 32,  $x_0$  = circumference unit  $\rho_q \ \pi/M$ , M = number of  $\ \pi$  intervals), the  $G_{nT}$  for the test segment may be written in the form

$$G_{nT}^{(1)} = \frac{1}{2\rho_{q}y_{o}} \int_{-y_{o}}^{+y_{o}} dy \int_{0}^{x_{o}} dx \frac{1 + jk[x^{2} + (y - y_{a})^{2} + \Delta_{a}^{2}]^{V_{2}}}{[x^{2} + (y - y_{a})^{2} + \Delta_{a}^{2}]^{V_{2}}}$$
(221)

where  $dx = \rho_q d\phi$ , x,y = center of a local coordinate system at the center of the J<sup>th</sup> source element, y<sub>a</sub> = tangential distance from the test section and  $\Delta_a$  = projection distance from the test section (see FIGURE 34). Equation (221) can be evaluated as follows:



FIGURE 34 Coordinates and nomenclature for a test segment section very near to a triangle segment on the body. The x and y of the local coordinate system are measured from the center of the source segment J.  $y_a = \vec{v}_1 \cdot \vec{a}_1$ ,  $|\vec{a}_1| = 1$ ,  $\Delta_a^2 = v_1^2 - y_a^2$  (Source : BEVENSEE [10]) With  $y_A = y_0 + y_a \ge 0$  and  $y_B = y_0 - y_a \ge 0$  one obtains:  $G_{nT}^{(1)} = \frac{1}{2 \rho_{p} \gamma_{p}} \int^{y_{B}} dy \int^{x_{0}} dx \frac{1 - jk \left[x^{2} + y^{2} + \Delta_{a}^{2}\right]^{V_{2}}}{\left[x^{2} + y^{2} + \Delta_{a}^{2}\right]^{V_{2}}}$ (222)

In NEARZ the near-charge contribution (222) of source segment J to the potential at test section I is computed only if all three of these conditions are true : 
$$K = 1$$
 ( $0 \le \phi \le \pi/M$  in the azimuthal integration over the source).  
 $\Delta_a < DTEST$  (the center of the test section is closer to the source segment than the test section length), and  $|y_A| < DH(J)/2$  (i.e., a perpendicular

t а

t dropped from the test section falls somewhere on the source segment). If one or more of these conditions is not valid the standard  $G_n$  (192) will be computed by the distant-source formulas . The evaluation of  $G_n$  for the numerical computation is described in HARRINGTON and MAUTZ [40] and that of  $G_{nT}^{(1)}$  in BEVENSEE [10].

Because of the approximations in separating the charge contributions to the potential on the test segment into a smoothed near-charge and discrete far-charge components, the test fields should be regarded as suspect

unless  $\Delta_a/\lambda > 1$  and also 4 DTEST/ $\lambda < 1/4$ , where 4 DTEST is the full length of the test segment in FIGURE 33.

The remainder of NEARZ is essentially the same as in program A and HARRA (6.4.2.2.) except that the measurement matrix [ZM] is computed for the test segment instead of the impedance matrix [Z] of the body.

The output of the source program HARRDF (BEVENSEE [10]) consists of the field components  $E^{inc}, E^{scat}, E^{tot}$  in the directions of  $\vec{a}_{\theta}, \vec{a}_{r}$  and (2x)  $\vec{a}_{\phi}$  for  $\phi = 0^{\circ}$ , a selected incident angle  $\theta_{i}$  and the selected mode n.

The extended program PANB computes these quantities for all needed modes n and for  $\phi = 0, 5, 10, ...180^{\circ}$ . The contribution of each mode is summed separately, and one obtains the complete  $E^{\text{tot}}$  in  $\vec{a}_{\theta}, \vec{a}_{r}$  and  $\vec{a}_{\phi}$  (mean value) directions. After coordination transformation the  $E^{\text{tot}}(\phi)$  are available in vertical, horizontal and radial directions. The amounts of  $E^{\text{tot}}(\phi)$  related to |E| of the incident plane wave deliver the chosen transmission Loss<sub>B</sub> for the three polarization axes  $p_1$  of the antenna  $A_1$  in FIGURE 11.

#### 6.4.5.2. LIMITATIONS OF THE NEAR-FIELD COMPUTATIONS

The basic limitations are already given by the computation of the [Z] matrices with program HARRA (see discussion in section 6.4.2.2.)

A further problem is to select the "best" segment length. If the segment is short, one would measure fluctuating components of the tangential fields, since the boundary condition of  $E_t = 0$  (125) is only satisfied in an integral sense with respect to the triangle functions. If the segment is long, one obtains an averaged value of the surrounding fields of the test point. Near field computations of a <u>point</u> field with a test segment tend to be inaccurate unless both these conditions are fulfilled:

a.) Minimum distance of the test segment center to the body surface >  $\lambda$ . b.) Test segment length <  $\lambda/4$ .

The accuracy of the obtained results will be discussed in section 10.3. In general, the field components can be computed with less than 2 dB error for a human-sized conducting model, if the frequency is below 500 MHz and if the distance of the test segment (antenna-body distance  $d_{at}$ ) from the surface is larger than 0.1 m. It is required to investigate the influence of each additional mode n and to perform two computations with different test segment lengths, in our case 0.08 and 0.2 m, to check fluctuations.

#### 6.5. SCATTERING FROM LONG CIRCULAR CYLINDERS: AN ANALYTICAL APPROACH

#### 6.5.1. PURPOSE OF THE ANALYTICAL APPROACH

The numerical computation of a finite body of revolution according to section 6.4. is primarily limited by computation time and storage capacity of the computer. Tremendous efforts are needed if the frequency is above 400 MHz and if the antenna-body distance is smaller than 0.1 m.

On the other hand, as shown in section 5.2.3., the human body is large compared with the wavelength at frequencies above 200 MHz if we consider the radar cross section as a measure for quasi-off-resonance behavior.

If we neglect the resonance phenomena and are looking only on the total field at vertical polarization at frequencies above 200 MHz, the threedimensional problem can be converted in a more simple two-dimensional problem. Thus, the adequate model is an infinite, circular cylinder, which is irradiated by a plane wave at an angle  $\theta_i = 90^\circ$ . The theory to solve this problem originates from KING and WU [50] and VAN BLADEL [81].

#### 6.5.2. METHOD OF SOLUTION

The scattered fields associated with a circular cylindrical scattered can be determined by separation of variables. The configuration of interest is shown in FIGURE 35, the body model is the IZYL defined in 5.4.1.:



FIGURE 35 Two-dimensional model of man for analytical computation (IZYL). Vertical infinite cylinder in a vertical polarized incident plane wave:  $D_c = cylinder$  diameter,  $A_1 = antenna$  positioned at r,  $\phi$ ,  $\underline{E}_7^{inc} = incident \vec{E}$  For the following computation it is convenient to operate with the angle  $\phi$ ' and the cylinder radius a :

$$\phi' = \phi + \P$$
;  $a = D_c/2$  (223)

The incident  $\vec{E}$ -wave has only a z-component and is of the type :

$$\underline{E}_{z}^{\text{inc}} = e^{-jkx} = e^{-jkr\cos\phi}$$
(224)

The incident field is expanded in a Fourier series in  $\phi$ ' whose r-dependent coefficients are found by insertion in the wave equation to satisfy Bessel's equation. Thus, with  $\underline{E}_{\tau}^{\text{inc}}$  finite at r = 0

$$\underline{E}_{z}^{\text{inc}} = \sum_{n=-\infty}^{\infty} j^{-n} J_{n}(kr) e^{jn\phi'}$$
(225)

where  $\underline{J}_n(kr)$  is a Bessel function of order n. The scattered field must have a Fourier series of the form

$$\underline{\underline{E}}_{z}^{scat} = \sum_{n=-\infty}^{\infty} \underline{\alpha}_{n} \quad \underline{j}^{-n} \quad \underline{\underline{H}}_{n}^{(2)}(kr) \quad e^{jn\phi'}$$
(226)

where  $\underline{H}_{n}^{(2)}(kr)$  is a Hankel function of the second kind and order n. The value of  $\underline{\alpha}_{n}$  follows directly from the boundary condition (125) :

$$\underline{E}_{z} = \underline{E}_{z}^{\text{inc}} + \underline{E}_{z}^{\text{scat}} = 0 \quad \text{at } r = a \qquad (227)$$

according to which the  $\underline{\alpha}_n$  are

$$\underline{\alpha}_{n} = -\frac{J_{n}(ka)}{\underline{H}_{n}^{(2)}(ka)}$$
 (228)

The above formulas by KING and WU [50] and VAN BLADEL [81] are now evaluated for near-field conditions with respect to the computer program PANA (see also program description in section 7.3.and listings in 16.2.1.) A series of complex parameters can be expressed as

$$\sum_{n=1}^{+\infty} \frac{c_n}{n} = \frac{c_n}{n} + \sum_{n=1}^{+\infty} \frac{(c_n + c_{-n})}{(c_n + c_{-n})}$$
(229)

In equation (225) the  $\underline{c}_n$  and  $\underline{c}_{-n}$  are

$$\underline{c}_{n} = j^{-n} J_{n}(kr) e^{jn\phi'}$$
  
 $\underline{c}_{-n} = j^{-n} J_{n}(kr) e^{-jn\phi'} = j^{n}(-1)^{n} J_{n}(kr) e^{-jn\phi'}$  (230)

With the relations(-j)<sup>n</sup> = (j)<sup>-n</sup> and  $e^{jn\phi} = \cos \phi' = e^{-jn\phi}, \underline{E}_z^{inc}$  becomes

$$\frac{E}{z}_{z}^{inc} = J_{o}(kr) + 2\sum_{n=1}^{+\infty} j^{-n} J_{n}(kr) \cos n\phi'$$
(231)

The series for  $\underline{E}_{z}^{scat}$  is evaluated as follows:

$$\frac{c}{n} = \underline{\alpha}_{n} j^{-n} \underline{H}_{n}^{(2)}(kr) e^{jn\phi'}$$

$$\underline{c}_{-n} = \underline{\alpha}_{-n} j^{n} \underline{H}_{-n}^{(2)}(kr) e^{-jn\phi'}$$

$$\underline{H}_{-n}^{2}(kr) = e^{-jn\pi} \underline{H}_{n}^{(2)}(kr) \qquad (232)$$

Because  $(-1)^n = (e^{j\pi})^n$ ,  $(-1)^n e^{jn\pi} = (e^{j2\pi})^n = +1$ , the  $\alpha_{-n}$  becomes

$$\underline{\alpha}_{-n} = -\frac{(-1)^n J_n(ka)}{e^{-jn \prod \underline{\mu}_n^{(2)}}(ka)} = (-1)^n e^{jn \prod \underline{\alpha}_n} = \underline{\alpha}_n \qquad (233)$$

and the  $\underline{c}_n$  becomes

$$\underline{c}_{-n} = \underline{\alpha}_{n} j^{n} e^{-jn \prod \underline{H}_{n}^{(2)}} (kr) e^{-jn \phi'}$$
(234)

With the relations  $e^{-jn\pi} = (-1)^n$ ,  $j^n e^{-jn\pi} = j^{-n}$  one obtains for  $E_z^{scat}$ 

$$\underline{\underline{E}}_{z}^{\text{scat}} = \underline{\underline{\alpha}}_{0} \underline{\underline{H}}_{0}^{2}(kr) + 2 \sum_{n=1}^{+\infty} \underline{\underline{\alpha}}_{n} \underline{\underline{H}}_{n}^{(2)}(kr) j^{-n} \cos n\phi' \qquad (235)$$

The combined fields from (231) and (235) represent the solution  $\underline{E}_{z}(r,\phi')$  at the location of the antenna  $A_{1}$ .

For the computer program the Hankel function may be replaced by the expression:

$$\underline{H}_{n}^{(2)}(kr) = J_{n}(kr) - j Y_{n}(kr)$$
(236)

where  $J_n$  is a Bessel function of the first kind and order n,and  $Y_n$  is a Bessel function of the second kind and order n or a Neumann function.

#### 6.5.3. LIMITATIONS OF THE ANALYTICAL NEAR-FIELD COMPUTATION

The convergence of the series (231) and (235) is quite slow  $(r \gg a)$  for big values of ka. Whereas six terms (modes n) give satisfactory results for ka = 3, over 1000 terms are needed for ka = 100 (KING and WU [50]). Our ka is about 3 at 1000 MHz, and accurate near-field data are possible with n=25.

## Leer - Vide - Empty

.

## 7. Two-Dimensional Computation of Scattering from an Infinite Circular Cylinder

#### 7.1. COMPUTATIONAL MODEL AND GOALS

The antenna-body model consists of a perfectly conducting, circular cylinder of infinite length (FIGURE 28: Model IZYL), a small antenna  $A_1$  polarized parallel to the cylinder axis with the coordinates r (radius) and  $\phi$  (azimuthal angle), and an incident plane wave with the E-field polarized parallel to the cylinder axis. The computational situation is shown in FIGURE 35, and the method of solution is described in section 6.5..

The selection of this model follows from the analysis in section 5.2.3. and 5.2.4. The vertical IZYL represents a standing human TS at frequencies above 200 MHz, irradiated by a plane wave with the incident angle  $\theta_i = 90^\circ$ .

The model allows the computation of all <u>vertical</u> polarized E-field components and should give a preliminary answer about the correlation among antenna-body distance  $d_{at}$ , azimuthal angle  $\phi$ , frequency f, cylinder diameter  $D_c$  and transmission  $Gain_B$ .

### 7.2. COMPUTER PROGRAM PANA: NEAR-FIELD PATTERN COMPUTATION OF THE IN-FINITE CYLINDER IZYL

### 7.2.1. COMPUTATIONAL FORMULAS AND PARAMETERS

The total field (vertical component always)  $\underline{E}(\mathbf{r}, \phi')$  at the antenna  $A_1$  is determined by the formulas (223) to (236) in section 6.5...With  $\phi' = \phi + \mathbf{1}$ :

$$\underline{E}(\mathbf{r},\phi') = \underline{E}^{\mathrm{inc}}(\mathbf{r},\phi') + \underline{E}^{\mathrm{scat}}(\mathbf{r},\phi') \quad (= 0 \text{ at } \mathbf{r} = a) \quad (227)$$

$$\underline{E}^{inc}(r,\phi') = J_{0}(kr) + 2\sum_{n=1}^{+\infty} j^{-n} J_{n}(kr) \cos n\phi'$$
(231)

$$\underline{E}^{\text{scat}}(\mathbf{r},\phi') = \underline{\alpha}_{0} \underline{H}_{0}^{(2)}(\mathbf{kr}) + 2\sum_{n=1}^{+\infty} \underline{\alpha}_{n} \underline{H}_{n}^{(2)}(\mathbf{kr}) \mathbf{j}^{-n} \cos n\phi' \qquad (235)$$

$$\frac{\alpha}{n} = -\frac{J_n(ka)}{\underline{H}_n^{(2)}(ka)} ; a = D_c/2$$
 (228)

$$\underline{H}_{n}^{(2)}(kr) = J_{n}(kr) - j Y_{n}(kr)$$
(236)

The transmission  $\operatorname{Gain}_{\mathsf{R}}$  (see definition (20) in section 5.1.2.) is then:

$$Gain_{B} = 20 \log \left[ 1 + \frac{\underline{E}^{Scat}(r,\phi')}{\underline{E}^{inc}(r,\phi')} \right]$$
(237)

In the program the variables and parameters are denoted as follows:

* A (REAL) a, radius of the cylinder [m] AK (REAL) ka DAT (REAL) dat, antenna-body distance [m] * DMAX1 (REAL) maximum dat for ARP [m] * DMAX2 (REAL) maximum dat for DRP [m] * DMIN1 (REAL) minimum dat for DRP [m] * DMIN2 (REAL) minimum dat for DRP [m] EZI (COMPL) Einc EZSC (COMPL) Escat [m] * [NTEG] f, integer number for the frequency [MHz] G (REAL) Gaing [dB] HNKR (COMPL) H <sub>n</sub> , Hankel function, second kind JN (REAL) J <sub>n</sub> , Bessel function, first kind K (REAL) k, wave propagation factor LAM (REAL) $\lambda$ , wavelength [m] * M1 (INTEG) maximum mode n for ARP * M2 (INTEG) number of dat for ARP * M2 (INTEG) number of dat for DRP [* MR1 (INTEG) number of dat for DRP [* MR1 (INTEG) number of dat for DRP PHI (INTEG) $\phi$ , integer azimuthal angle, 0, 5, 10, 180 [ <sup>0</sup> ] PHH (REAL) $\phi'$ , $\phi + \pi$ [ <sup>0</sup> ]	INPUT	PARAME	ETER NAME	MEANING	UNITS
YN (REAL) Kr YN (REAL) Y <sub>n</sub> , Bessel function, second kind	* * * * * * * *	A AK DAT DMAX1 DMAX2 DMIN1 DMIN2 EZI EZSC F G HNKR JN K LAM M1 M2 MR1 M2 MR1 M2 MR1 M2 MR1 RK YN	(REAL) (REAL) (REAL) (REAL) (REAL) (REAL) (COMPL) (COMPL) (INTEG) (REAL) (REAL) (INTEG) (INTEG) (INTEG) (INTEG) (INTEG) (REAL) (REAL) (REAL) (REAL) (REAL)	a, radius of the cylinder ka $d_{at}$ , antenna-body distance maximum $d_{at}$ for ARP maximum $d_{at}$ for DRP minimum $d_{at}$ for DRP Einc Escat f, integer number for the frequency Gaing H <sub>n</sub> , Hankel function, second kind J <sub>n</sub> , Bessel function, first kind k, wave propagation factor $\lambda$ , wavelength maximum mode n for ARP number of $d_{at}$ for DRP number of $d_{at}$ for DRP $\phi$ , integer azimuthal angle, 0, 5, 10, 180 $\phi', \phi + \Pi$ kr Y <sub>n</sub> , Bessel function, second kind	[m] [m] [m] [m] [MHz] [dB] [ <sup>0</sup> ]

TABLE 36 Variables and parameters used in program PANA (ARP: Azimuthal radiation pattern, DRP: Directive radiation pattern)

#### 7.2.2. PROGRAM DESCRIPTION PANA

The listing of the program PANA is enclosed in Appendix 16.2.1. It consists of one main program and one subroutine BESS. The main program consists of two parts preceded by an input section.

PANA 44 to 60 : Input and preliminary computations PANA 61 to 98 : Computation of the azimuthal radiation pattern PANA 100 to 143 : Computation of the directive radiation pattern

The input data set consists of the following punched cards (PANA 174-179):

NAMES	INPUT	PARAME	TER	FORMATS	INPUT	PARA	METER	SAMPLE	INPUT	PARA	METER
DMINI	DMAX1	MR1	MI	F6.2	F6.2	14	 I4	0.05	0.25	05	12
DMIN2	DMAX2	MR2	M2	F6.2	F6.2	14	14	0.05	1.00	20	25
А				F7.3				0.125	5		
F				13				150			
:				:				:			
F				13				250			

TABLE 37 Input parameter set (punched cards PANA 174-179) Card No. 1 : data for the azimuthal radiation pattern Card No. 2 : data for the directive radiation pattern Card No. 3 : data for the cylinder radius Card No. 4+i : data for the frequencies to be computed (i=0,1,2,..)

In the present program the azimuthal radiation pattern computation (part 1) consists of the computation of the  $\phi$ -dependence of Gain<sub>B</sub> for five (MR1) fixed d<sub>at</sub>'s of 0.05 to 0.25 m in the  $\phi$ -range 0 to 180°. The spacing of the  $\phi$ -steps is determined to 5° due to the statement

DO 2 I=1,37

The maximum number of  $d_{at}$ 's (MR1) is limited by the output procedure PANA 69-71; the values of the  $d_{at}$ 's may vary from 0.01 to 9.99 m (PANA 70).

PANA 81

In the present program the directive radiation pattern computation (part 2) consists of the computation of the  $d_{at}$ -dependence of Gaing at the irradiated side ( $\phi = 0^\circ$ ) and in the shadow zone ( $\phi = 180^\circ$ ) of the cylinder. The  $d_{at}$ 's vary from 0.05 to 1.00 m in 0.05 m steps, determined by the input cards. The maximum number of  $d_{at}$ 's (MR2) is limited only by the computational time; the values of the  $d_{at}$ 's may vary from 0.01 to 9.99 m (PANA 109).

The program computes the two radiation patterns for as many frequencies as frequency input cards are added in the data set. If the last frequency is executed, the program stops due to the statements:

IF(EOF (1)) 52,53 PANA 55 52 STOP PANA 145

The radius a (A) is limited by PANA 64 and 104 and may vary from 0.005 to 4.999 m. The frequency f (F) may vary from 1 to 999 MHz (PANA 62 and 101).

The computation of the first Gain<sub>B</sub> for the azimuthal radiation pattern starts with the defining of AK (PANA 60), selecting the first  $d_{at}$  (PANA 78 and 79) and selecting the first  $\phi = 0^{\circ}$  (PANA 81 and 82). The corresponding RK is given by PANA 88 and the PHH by PANA 83. The maximum mode number M1 is transferred to M, and the computation parameters M,AK,RK,PHH are transferred to the subroutine BESS for the computation of Gain<sub>B</sub> (G) :

The subroutine BESS will be discussed below. The result G (Gain<sub>B</sub>) is returned from the subroutine and is checked for correct size in printing (PANA 90) and plotting (PANA 92). With the statements

DO 2 1-1 31	PANA 85
	PANA 86
3 DI(L)=IN	PANA 93
KK=1F1X(G+30.5)	DANA Q
D1(KK)=1RO+J)	FARA 3

the number J (first J=1) of the J-th  $d_{at}$  is plotted as an amplitude marker in the Gain<sub>B</sub> versus  $\phi$  diagram in FIGURE 38.

The computation of the first  $Gain_B$  for the directive radiation pattern uses the same AK and starts with the defining of the first  $d_{at}$  (PANA 124) and the first PHH (PANA 118). The corresponding RK is computed in PANA 131 and 132. The following computation of  $Gain_B$  in the subroutine BESS and the plotting of Gain\_B as an amplitude marker in the Gain\_B versus  $d_{at}$  diagram is similar to above with the exception that a (\*) is printed. The complete diagram is shown in FIGURE 39.

The subroutine BESS uses the routine BESYN of the library BRUSLIB VIMCODE C306. Because other Bessel routines might be used in other computer centers, a few comments are helpful. In FORTRAN IV the transfer of the parameter 0 causes difficulties. Thus, the mode numbers n = 0,1,2,...M are changed into MP1 = 1,2,3,...M+1 (PANA 153). The statement

CALL BESYN(-AK,MP1,JN,YN) PANA 154

executes the computation of  $J_n(ka)$  and  $Y_n(ka)$  for the modes 0,1,2,..MP1-1. With PANA 156 one obtains the  $\alpha_n$  of (228) in array AN(I). Similarly

CALL BESYN(-RK,MP1,JN,YN) PANA 157

executes the computation of  $J_n(kr)$  and  $Y_n(kr)$ . Finally, the <u>E</u><sup>inc</sup> and <u>E</u><sup>scat</sup> are computed in PANA 158-167. The returned G in PANA 169 represents Gaing of equation (237) for the selected r and  $\phi$ '.

#### 7.2.3. PROGRAM LIMITATIONS AND ACCURACY

The convergence of the series in equation 231 and 235 is quite slow for large values of kr and similar in equation 228 for large values of ka. The mimimum required modes  $n_{min}$  to solve equation 237 accurately depends therefore primarily on the maximum frequency f and the maximum  $d_{at}$ . The subroutine BESS allows the computation of maximum 99 modes, and since the computational time increases with n,  $n_{min}$  should be evaluated as follows:

For large values of  $d_{at}$ 's (or distance r)  $Gain_B$  should approach 0 dB for  $\phi = 180^{\circ}$  (shadow zone). If the chosen  $n_{min}$  is too small,  $Gain_B$  increases first monotonously with increasing  $d_{at}$  as expected; it begins to oscillate for larger  $d_{at}$ 's, and finally the program stops with an error message. Introducing the additional statements:

	GOTO 5	before	PANA	61
5	CONTINUE	before	PANA	100
	IJ=IJ+2	before	PANA	118

the program executes only the directive radiation pattern computation for  $\phi = 180^{\circ}$ . At a given frequency f, a given radius a and a chosen minimum mode number  $n_{min}$  (M2) the largest correctly computed  $d_{at}$  can be seen in the Gaing versus  $d_{at}$  diagram (e.g., FIGURE 39,bottom, if M2 would be <12). In our application it was found that even for the maximum frequency 999 MHz a mode number of 25 (M2 = 25) is sufficient for  $d_{at}$  below 2 m, with a = 0.125 m and an accuracy of better than 0.1 dB.

The dimensions of the input parameters depend mainly on the selected output formats and have been discussed in 7.2.2.. The units are [m] and [MHz]:

DMINI	: 0.01	DMIN2	: 0.01	A : 0.005-4999	
DMAX1	: 9.99	DMAX2	: 9.99	F : 1 - 999	
MR1	: 1-5	MR2	: 1-∞	No. of F : 1 - ∞	(238)
M1	: 1-99	M2	: 1-99	Δφ steps : 1 - ∞ (PANA 81)	

Scaled model computations are possible with a factor cscale:

Fmode1	=	Factual • <sup>C</sup> scale		
A <sub>mode1</sub>	=	A <sub>actual</sub> ÷ <sup>c</sup> scale		(239)
R <sub>mode</sub> 1	=	<sup>R</sup> actual <sup>÷ C</sup> scale	$(R = d_{at} + a)$	

The execution of the program PANA on a CDC 6500 computer requires a storage of 20,000 to 60,000 octal, depending on the compiler. The standard program (with 3 frequencies) requires 30 s excecution time. 7.3.1. AZIMUTHAL AND DIRECTIVE RADIATION PATTERNS OF ANTENNA-IZYL MODEL

Samples for the frequency 567 MHz (FIGURES 38 and 39) and 150 MHz (FIGURES 40 and 41) are presented here; additional samples are in Appendix 16.2.1..

#### AZIMUTHAL RADIATION PATTERN FREQUENCY 567 MHZ TWO-DIMENSIONAL ANTENNA-BODY SYSTEM NUMBER OF MODI: 12 TESTBODY: INFINITE ROT.SYM.CYLINDER POLARIZATION: VERTICAL/VERTICAL DIAMETER: .25 M : ANTENNA-BODY DISTANCE IN METERS DAT(I) : GAIN, FIELD STRENGTH AT THE ANTENNA A1 IN DB, 0 DB = FREE SPACE : HORIZONTAL ROTATION ANGLE IN DEGREES GAI(I) PHT DAT(1) DAT(2) DAT(3) DAT(4) DAT(5) .200 . 100 .250 .050 . 150 + 0 DB +10 DB PHI GAI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB 5 -5.3 0 3.9 3.7 .5 •5 -5.3 -5.1 5555 3.7 .5 •5 5 3.9 • 14 .6 10 4 3.9 44 -4.9 ر 5 5 5 5 5 •3 .8 15 3.8 3.8 ٠ 4 3.8 1.1 -4.4 .2 3.8 3.7 20 . Ŀ .0 -.2 1.4 -3.9 25 3.8 4 3.5 1.8 -3.1 30 3.9 • 423 423 3.4 3.2 3.9 3.9 -.5 2.1 -2.3 35 40 5 -.8 • 43 2.8 -.4 45 3.0 2.7 3.9 -1.1 .4 43 -1.5 3.8 3.1 • 50 .5243 3.3 3.5 3.5 3.7 1.2 55 60 -1.9 2.4 • 543 3.6 3.4 -2.4 2.0 1.9 • ٠ 534 2.4 -2.9 • 65 1.6 .2 54 .2 5 3.1 3.5 2.9 -3.5 1.2 70 ٠ 3.4 .6 3.1 • Ż5 2.7 ٠ 2 35 2. 35 3.3 -4.8 .1 2.3 3.3 80 ٠ -.6 3.3 85 -5.5 1.8 • ٠ 2.3 5 -6.3 -7.1 1.2 -1.3 -2.0 2.6 3.1 90 2 <u>9</u>5 35 -2.8 -3.6 1.5 2.5 2.0 2 -7.9 -.1 100 3.45 2 -.8 .9 105 • 2 3 45 1.4 -4.5 110 -9.7 -1.7 34. 2345 34 3 4.5 -.7 .7 -5.5 -2.6 -10.7 • 115 2 -3.5 -.2 -6.5 -1.6 120 -11.8 ٠ 2 -2.6 -1.2 -7.6 125 -12.9 .2 345 -2.3 -5.9 -3.8 . -14.2 -8.9 130 2 -3.7 3 45 -10.3 -7.3 -5.2 -15.6 135 2 .3 4 5 2 .3 45 -5.2 -6.8 -8.8 140 -17.1 -11.8 2 -8.4 -6.9 -10.5 145 -18.7 -13.4 3 45 2 -8.6 150 -20.2 -15.0 -12.1 -10.1 34 5 2 -13.4 -11.5 -10.1 -21.4 -16.2 155 3 45 3 45. 2 2 -12.3 -11.0 160 -21.9 -16.8 -14.1 ٠ -12.4 -14.1 -11.2 165 -21.8 -16.8 22 3 45. -10.9 -13.9 170 175 -21.5 -16.5 -12.2 -11.9 2 345. -16.1 -13.4 -11.7 -10.5 . 180 -21.0

FIGURE 38 Azimuthal radiation pattern at 567 MHz (result of program PANA) Left: Gain<sub>B</sub> for 5  $d_{at}$ 's and  $\phi = 0 - 180^{\circ}$ , right: Gain<sub>B</sub> versus  $\phi$  for 5  $d_{at}$ 's.
In FIGURE 38 and 39 the diameter of the IZYL is 0.25 m and the frequency is 567 MHz. The azimuthal radiation pattern (FIGURE 38) shows a clear minimum of the gain (GAI(I)) at  $\phi = 160-165^{\circ}$ , decreasing with decreasing  $d_{at}$  (DAT(I)). The directive radiation pattern (FIGURE 39) at  $\phi = 0^{\circ}$  reveals a gain oscillation, with maxima at  $d_{at}$  of about  $\lambda/4 + n \cdot \lambda/2$ .

DIRECTIVE RA	DIATION PA	TTERN FREQU	Е N C Y 567 M H Z				
TWO-DIMENSIONAL ANTENNA	-BODY SYSTEM	ROT.ANGLE PHI : 0	ROT.ANGLE PHI : O ABOVE, 180 BELOW				
TESTBODY: INFINITE ROT. DIAMETER: .25	SYM.CYLINDER	NUMBER OF MODI: 25 POLARIZATION: VERTICAL/VERTICAL					
DAT(I) GAI(I) PHI M DB DEG	-20 DB -15 DB	-10 DB - 5 DB	+ 0 DB + 5 DB				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$							

FIGURE 39 Directive radiation pattern at 567 MHz (resultof program PANA). •Above: Gain<sub>B</sub> versus  $d_{at}$  at  $\phi = 0^{\circ}$ , below: Gain<sub>B</sub> versus  $d_{at}$  at  $\phi = 180^{\circ}$ . In FIGURE 40 and 41 the same IZYL is shown at a frequency of 150 MHz. This frequency is just below the application range of this two-dimensional computational model. The data obtained with this model should be compared with the data from the three-dimensional model in section 10.4..FIGURE 40 shows the azimuthal- and FIGURE 41 the directive radiation pattern:

AZIMUTHAL RADIATION PATTERN FREQUENCY 150 MHZ						
TWO-DIMENSIONAL ANTENNA-BODY SYSTEM						
TESTBODY: INFINITE ROT.SYM.CYLINDER       NUMBER OF MODI: 12         DIAMETER: .25 M       POLARIZATION: VERTICAL/VERTICAL         DAT(I) : ANTENNA-BODY DISTANCE IN METERS         GAI(I) : GAIN, FIELD STRENGTH AT THE ANTENNA A1 IN DB, 0 DB = FREE SPACE         PHI : HORIZONTAL ROTATION ANGLE IN DEGREES						
DAT(1) DAT(2) DAT(3) DAT(4) DAT(5 .050 .100 .150 .200 .250						
PHI GAI(1) GAI(2) GAI(3) GAI(4) GAI(5	) -20 DB -10 DB + 0 DB +10 DB					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$					

FIGURE 40 Azimuthal radiation pattern at 150 MHz (result of program PANA). Left: Gain<sub>B</sub> for  $5 d_{at}$ 's and  $\phi = 0 - 180^{\circ}$ , right: Gain<sub>B</sub> versus  $\phi$  for  $5 d_{at}$ 's.

41 shows at  $\varphi=0^\circ$  a maximum gain at  $d_{at}$  =  $\lambda/4$  (0.375m). The shadow zone  $\varphi=180^\circ$  is within 1.5dB identical with those at 567 MHz.

DIRECTIVE RA	DIATION PATTERN FREQUENCY 15	50 M H Z						
TWO-DIMENSIONAL ANTENNA-BODY SYSTEM ROT.ANGLE PHI : 0 ABOVE, 180 BELOW								
TESTBODY: INFINITE ROT. DIAMETER: .25	SYM.CYLINDER NUMBER OF MODI: 25 5 M POLARIZATION: VERTICAL/VERTI	ICAL						
DAT(I) GAI(I) PHI M DB DEG	-20 DB -15 DB -10 DB - 5 DB + 0 DB	+ 5 DB						
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		· · · · · · · · · · · · · · · · · · ·						
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		· · · · · · · · · · · · · · · · · · ·						

FIGURE 41 Directive radiation pattern at 150 MHz (result of program PANA). Above: Gaing versus  $d_{at}$  at  $\phi = 0^{\circ}$ , below: Gaing versus  $d_{at}$  at  $\phi = 180^{\circ}$ . The minimum Gain<sub>B</sub> occurs at frequencies above 300 MHz not at  $\phi = 180^{\circ}$  but at about 165° as shown in FIGURE 42 (dashed lines: model not accurate). Small diameter changes are of little significance at all d<sub>at</sub>'s (FIGURE 43).







FIGURE 43 Gain<sub>B</sub> versus d<sub>at</sub> for different cylinder diameters  $D_{c}$  ( $\phi$  = 180°).

## 8. MEASURING METHOD

### 8.1. PURPOSE OF THE EXPERIMENT

The experiments should answer the following questions:

<u>Assumption verification</u>: The computational models are based on assumptions which need to be verified. The main assumptions, as discussed in section 5.2., are reciprocity, quasi-perfect conductivity, and simplified body shape.

<u>Off-resonance model verification</u> : The two-dimensional computation of the radiation characteristics of the IZYL-antenna model (section 7.) delivers data for frequencies above 200 MHz. The experiment at frequencies above 200 MHz should deliver near-field data for large scale  $d_{at}$  and  $\phi$  variation. If an acceptable correlation between experiment and computation exists, an extension of the theory on three dimensional computation of conducting bodies is reasonable.

<u>Resonance phenomena</u> : The analysis of the antenna-body system (section 5.2.3.) predicts resonance effects at about 40 to 200 MHz. The experiment should verify this prediction.

<u>Practical body-mounted antennas</u>: A practical body-mounted antenna is usually a monopole antenna mounted on a transmitting device. If the housing of the transmitter is small (e.g., maximum dimension about 0.25 m), the counterpoise for the monopole antenna is not ideal (too small) for frequencies below 300 MHz. Thus, the experiment should also deliver data for standardized transmitting devices operating in proximity to the body in the entire regarded frequency range.

#### 8.2. DESCRIPTION OF THE ANTENNA-BODY TEST SET-UP

Real-size antenna-body experiments have to be performed outdoors and in proximity to the ground. The reasons for this decision and the problems dealing with ground reflections have been discussed in section 5.3.. The outdoor FR experiments are performed on a military airport (AMF Dübendorf). With respect to RF-emissions one has to act very cautiously. Not only the RF-power emitted at the measuring frequency has to be kept with-in permissible limits but also unwanted harmonic distortions have to be controlled. On the other hand there are at any frequency distortions from the outside. Our experiment requires a signal/noise (S/N) ratio of > 30 dB

since the effect to be studied are up to 25 dB below the maximum field strength level. Further, one is obliged to apply small radiation sources (electrically small antennas, limited counterpoise) with a generally limited efficiency. Thus, the accurate measuring frequency and the necessary RF-power cannot be determined in advance.

Quartz-stabilized RF-generators for <u>all</u> measuring frequencies are out of question. The manufacturing of miniaturized transmitters is too expensive (especially for frequencies above 200 MHz), the frequency is fixed and the power can only be controlled within small margins (see section 11.3.).

Free-oscillating RF-generators tend to be unstable with respect to power and frequency, need a careful tuning in order to avoid harmonic distortions and get detuned if the antenna environment is disturbed. Thus, such transmitting devices are not suited for our experiments.

In order to choose arbitrary antennas, frequencies and power levels, a dummy system (FIGURE 44) is used instead of an autonomic RF-generator. The monopole antenna  $A_1$  and its counterpoise represent an idealized field point source (transmitting case) or a small field probe (receiving case).  $A_1$  is connected by a long coaxial cable to a precision RF-generator or in the receiving case to a field strength meter.

Let us first consider A1 as a transmitting antenna. Remote feeding causes severe problems concerning radiation from the feeding coaxial cable, if  $A_1$ is a monopole with an electrically small ( <  $\lambda/2$  diameter) counterpoise. For the experiments  $A_1$  has to be approached up to 0.05 m to the test body. As a consequence, the counterpoise in FIGURE 44, No. 2, represents the maximum acceptable size with a diameter of 70 mm and a length of 100 mm. If no precautions would be taken, the outer sleeve of the cable would become a part of the counterpoise at frequencies below 500 MHz. Radiating currents on the outer sleeve of the cable would make accurate field measurements impossible. Methods for attenuating such sleeve currents are described in ARRL [3], KRUPKA [53] and ROTHAMMEL [71]. Usually a  $\lambda/4$  hollow cylinder is mounted around the coax, opened towards the antenna and contacted with the coax on the opposite cylinder side. Such a  $\lambda/4$  RF choke is efficient if the diameter is about 3-times the coax diameter. A smaller RF-choke can be obtained if the feeding coax cable is shaped in a coil (FIGURE 44, No. 4) of an electrical length of  $\lambda/4$ . The design of such a helical RF-choke is similar to that of a helical antenna and will be discussed in 8.3.2.. Because the helical RF-choke radiates itself a certain amount of RF power, it is covered by an absorbing tube (FIGURE 44, No.5.). The remaining surface waves on the feeding coax are attenuated by covering the whole coax with absorber material in the proximity of the antenna  $A_1$ . Finally load variations for the coax and the generator are prevented by inserting a 20 dB attenuator (FIGURE 44, No.3) below the antenna's foot point.

The complete antenna test set-up is shown in FIGURE 44. It follows the specifications evaluated in section 5.3. and allows measurements of vertical polarized E-fields in the frequency range 10-1000 MHz. The dummy system discussed above has a shape of a circular cylinder of 70 mm diameter and is supported by an antenna holder (No.6.). Depending on the frequency the test antenna A<sub>1</sub> (No.1.) is a helical monopole as shown or a whip (see 8.3.1.). The revolving stage (No.7.) rotates the test-body together with the antenna A<sub>1</sub>. A directive, broadband antenna (8.3.3.) A<sub>2</sub> (No.8.) completes the antenna test set-up.



FIGURE 44 Measuring antenna test set-up

- test (body-mounted) antenna A1
   electrical counterpoise for A1
   20 dB attenuator in series
- 4: matched RF-choke

- 5: EM absorber material
- 6: wooden antenna holder
- 7: wooden revolving stage
- 8: remote antenna A<sub>2</sub>

For the computational models the transmission distance d has been defined in section 5.1.2. as follows:

For the experiment it is better to keep the distance between A1 and A2 constant, thus we define the experimental transmission distance d:

Essentially there is no difference between (240) and (241), because  $d_{at}$  is small compared with d and because we are not interested in the absolute phase of the fields. However, the reference field strength  $E_0$  is more constant when the antennas and the cables are not moved.

The experiments require rotation of the antenna-body system ( $\phi$  = variable,  $d_{at}$  = parameter) and translation of the body in respect to the antenna A<sub>1</sub> ( $d_{at}$  = variable,  $\phi$  = parameter). These two experiments are shown schematically in FIGURE 45:

TEST 2

TEST 1



FIGURE 45 Rotation and translation of test bodies

- 1: wooden revolving stage
- 2: plastic trackway section
- 4: displacement transducer
  - (rubber band goniometer)
- 5: rubber thread

3: wagon

### 8.3. ANTENNAS AND FEEDING

### 8.3.1. BODY-MOUNTED ANTENNA A1

Theoretically the body-mounted antenna A<sub>1</sub> should fulfill the following requirements as defined in section 5.1.2. :

- The radiation intensity should be constant, but the efficiency is not of interest.
- The physical size of  $A_1$  should be smaller than any relevant dimension of the test set-up.
- Only one dominant E-polarization axis pj should exist.
- The antenna should radiate omnidirectionally.
- The impedance and thus the radiation should not change due to body proximity.

In practice there are physical and technical limitations:

- The efficiency should be so high, that the radiation of  $A_1$  is much higher (30 dB) than the leakage radiation of all involved equipment. Such an efficiency can only be obtained if  $A_1$  is operated near resonance.
- An antenna length h of 0.15 m is acceptable for the purpose of the experiment. This antenna length corresponds to  $\lambda/4$  at 500 MHz. Thus, efficient, resonant (no external tuning), electrically small antennas have to be used for all frequencies below 500 MHz.
- A strict linear polarization is difficult to obtain with resonant electrically small antennas. Any internal frequency tuning element leads to certain field irregularities.
- Omnidirectional radiation in a horizontal plane depends not only on the antenna but also on the feeding cable. A dipole antenna is not suited because the cable had to be mounted rectangular to the antenna axis. Thus, monopole antennas have to be used with a limited counterpoise.
- If an antenna has to be operated near resonance, the impedance depends on body proximity, because the bandwidth of any electrically small antenna is narrow (see section 4.5.).

The normal mode helical monopole (16.1.) offers a good compromise for antenna lengths  $h > \lambda/20$ . The polarization is elliptical, with a dominant vertical E-component which is larger than that of an equal-sized whip.

Ant. Type No.	Resonant Frequency f <sub>res</sub> [MHz]	Approx3dB Bandwidth min/max[MHz]	Antenna length h [mm]	Antenna diameter D <sub>h</sub> [mm]	Number of turns <sup>N</sup> h	Wire diam. d <sub>W</sub> [mm]
ATI hel. AT2 hel. AT3 hel.	$120 \pm 3$ $180 \pm 3$ $210 \pm 2$ $298 \pm 2$	110 - 125 170 - 185 190 - 215 260 - 310	185 155 145 145	11.5 12.0 12.0	42 26 24	1.5 2.0 2.0
AT5 hel. AT5 hel. AT6 whip AT7 whip AT8 whip	298 ± 2 363 ± 4 490 ± 5 630 ± 5 1060 ± 8	200 - 310 335 - 403 450 - 560 605 - 700 960 -1150	143 130 150 110 75	12.0	8 - -	2.0 1.5 2.0 1.0

TABLE 46 Monopole antennas  $A_1$  for field experiments. The resonant frequency  $f_{res}$  and the bandwidth have been measured with a network analyser, when  $A_1$  was mounted on the counterpoise, with RF-choke, but without attenuator, and in an anechotic chamber.

The specifications of the experimental antennas  $A_1$  are shown in TABLE 46. It should be mentioned that these data may vary from the theoretical data in section 16.1., because the antennas  $A_1$  are operated on the later used limited counterpoise.

### 8.3.2. RF-CHOKES

The purpose of the RF-chokes is to attenuate surface currents on the feeding coaxial cable. They are constructed from an RG-58 coaxial cable, wrapped in a helical shape. The specifications are shown in TABLE 47:

Operating Range f <sub>C</sub> [MHz]	Choke length l <sub>c</sub> [mm]	Choke diameter d <sub>c</sub> [mm]	Cable length l <sub>f</sub> [mm]	Number of turns <sup>n</sup> c	Cable diam. d <sub>f</sub> [mm]
50 <del>-</del> 150	100	20	1000	14	6
100 - 200	70	16	700	12	6
150 - 200	50	16	500	8.5	6
200 - 250	37	16	330	6	6
250 - 350	22	16	220	4	6
350 - 500	13	16	130	3	6
	$\begin{array}{c} \text{Operating} \\ \text{Range} \\ \text{f}_{\text{c}}  [\text{MHz}] \\ \hline 50 & - 150 \\ 100 & - 200 \\ 150 & - 200 \\ 200 & - 250 \\ 200 & - 250 \\ 250 & - 350 \\ 350 & - 500 \end{array}$	Operating Range         Choke length $f_c$ [MHz]           50         - 150           100         - 200           150         - 200           50         - 200           200         - 250           37           250         - 350           22           350         - 500	$\begin{array}{c c} \mbox{Operating} \\ \mbox{Range} \\ \mbox{f}_c \ [MHz] \\ \mbox{f}_c \ [MHz] \\ \mbox{Imm} $	$\begin{array}{c c} \mbox{Operating} \\ \mbox{Range} \\ \mbox{f}_c \ [MHz] \\ \mbox{f}_c \ [MHz] \\ \mbox{l}_c \ [mm] \\ \mbox{l}_c \ [mm] \\ \mbox{l}_d \ [mm] \\ \mbox{l}_d \ [mm] \\ \mbox{l}_d \ [mm] \\ \mbox{l}_d \ [mm] \\ \mbox{l}_f \ [mm] \ [mm] \\ \mbox{l}_f \ [mm] \ $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

TABLE 47 RF-chokes for field experiments.  $f_{C}$  theoretical values.

The effectiveness of the coax-line RF-chokes in TABLE 47 decreases at higher frequencies because of the distributed capacitance among the turns. Since they have to be mounted in absorbing tubes (attenuation of the choke radiation), the Q-factor is not very high, resulting in a favourable broad band range, but with further decreases effectiveness. There is a simple method to test RF-chokes: when the complete antenna test set-up according to FIGURE 44 is assembled for the selected test frequency, the received field strength should not change more than 2 dB if the coaxial feeding cable is touched by hand at any location along the cable.

## 8.3.3. REMOTE ANTENNA A2

Because the helical antennas  $A_1$  are elliptically polarized and we have to measure only the vertical polarized E-field component, the remote antenna  $A_2$  has to be strictly linear polarized. In addition  $A_2$  should be a directive antenna with broadband characteristics. It was found that a logarithmic periodic antenna (LPD) fulfills these requirements best. The chosen Cross LPD (R. GRANER, AMF) operates from 100 to 1000 MHz with a directive gain of about 6 dB (aperture angle  $\pm$  60°). By simple line-switching the vertical polarization and the horizontal polarization can be measured independently, which is needed to check the radiation properties of  $A_1$  before the actual experiment can be started.

### 8.4. MEASURING EQUIPMENT

### 8.4.1. REVOLVING STAGE FOR ANTENNA-BODY ROTATION

In cooperation with R. GRANER from the AMF a revolving stage and an electronic plotter control unit has been developed. It consists of

- revolving stage with basement, stage, engine and angle sensors (see FIGURE 48, shown with inverted separate stage).
- angle data transmitter, combined with remote control of the stage rotator and with an intercom system
- plotter control unit with angle display, synchronizer and x-y driver (see FIGURE 49).

This equipment allows the automatic plotting of azimuthal gain charts when combined with the RF-equipment and the power supply for the stage engine. The control unit synchronizes on the zero marker (FIGURE 48) and plots one full 0-360°  $E/\phi$  sequence with 1° resolution on an ordinary x-y recorder in a rectangular diagram as recommended by CEI Publication 138.



FIGURE 48 Revolving stage

- 1 : basement
- 2 : hollow axle (for antenna coaxial cable with rotational N-connector)
- 3 : supporting wheels
- 4 : engine with driving
  wheel (24 V DC)
- 5 : angle sensors with sensor protection bolts
- 6 : line driver and intercom station
- 7 : stage platform (upside down)
- 8 : angle markers (s: synchronization marker)
- 9 : angle marker protection



FIGURE 49 Plotter control unit
1 : command switch : next revolution = plot diagram
2 : x-y driver output

### 8.4.2. TRACKWAY FOR ANTENNA-BODY TRANSLATION

According to FIGURE 45 a trackway was constructed in order to move the heavy test subject (phantom weight 90 kg !) continuously toward the stationary antenna A<sub>1</sub>. A wooden basement of 5 meter length equipped with plastic rails was mounted on the  $\phi$ -coordinates 0-180° and later 90-270°. A small, ball-bearing equipped wagon carried the test subjects, with the footpoint spaced s = 0.2m appart from the ground. By help of deflection pullies, mounted each 10 meters away from A<sub>1</sub> on the trackway axis, and long plastic strings the wagon could be precisely moved manually.

The monitoring of  $d_{at}$ , which is the distance from the center of A<sub>1</sub> and the nearest surface of the test body, is not easy. First, this distance has to be actually measured and not e.g., the position of the wagon, because of practical accuracy considerations. Second, the space between A<sub>1</sub> and TS should not be disturbed by measuring devices, because any metallic or dielectric material causes field disturbances. Third, an accuracy of ± 10 mm is required at least in the low dat regions.

The rubber band goniometry, developed by NEUKOMM [65] for the biomechanical research, solves this difficult problem with little effort. A low torque conductive plastic potentiometer is mounted rectangularly to the trackway axis in the distance a = 2 m from the vertical A1 axis. (FIGURE 45, No. 4.). At the <u>vertical</u> axis of the potentiometer a beryllium bronze arm of 0.1mm thickness and 100 mm length is attached. The arm can be bent up and down without angular changes or significant forces on the axle. From this arm a thin rubber thread ( $\emptyset < 0.3 mm$ ) is stretched (about 1 N tensil force) to the test point on the TS and fixed with self-adhesive tape (see FIGURE 45, No. 5.). The linear motion dat is thus transformed into an angular motion  $\beta$  according to

$$\beta = arctg(dat/a)$$

(242)

There is a non-linear, but one-to-one correspondence between  $\beta$  and dat. In a linear  $\beta$  presentation the interesting range  $0 < d_{at} < 0.35 \text{ m}$  corresponds to  $0 < \beta < 10^{\circ}$  but contains also the large range  $0.35 < d_{at} < 2.38 \text{ m}$  corresponding to  $10 < \beta < 50^{\circ}$ . The accuracy in the low dat range is better than 10 mm, because the hysteresis of the rubber band goniometer is less than 0.3°, the resolution is quasi-infinite (better than 0.01°). The non-linearity of the potentiometer of 0.5 % F.S. would cause moderate errors in (242), thus the  $\beta$ -scale is calibrated directly in the actual dat-scale. The mechanical  $\beta$  signal is converted by the potentiometer into a proportional electrical tension, and with the built-in impedance converter the signal is transmitted over a long shielded cable to the x-imput of the plotter. If the field strength signal is on the y-imput of the plotter, one obtains a calibrated E/dat plot with 0 < dat < 2.3 m for  $\phi = 0^{\circ}$  and  $\phi = 180^{\circ}$ . Of course, the switching from  $\phi = 0^{\circ}$  to  $\phi = 180^{\circ}$  requires a tip down of the A<sub>1</sub> antenna tower and to connect the rubber thread on the reverse side of the test body when it has been rolled over with the wagon. The d<sub>at</sub>-scale has to be calibrated once by means of markers on the middle line of the trackway.

## 8.4.3. FIELD MEASURING EQUIPMENT

Besides the mentioned antennas and  $\phi/d_{at}$  recording devices the following materials have been applied in the experiments:

- Field-strength measuring unit: main unit RHODE + SCHWARZ VHF-UHF ESUM BN 15076/5/P with the plug-in units 25-230 MHz, 160-470 MHz, and 850-1300 MHz. The effective accuracy (as tested) is  $\pm$  0.5 dB in the +5 to -20 dB range and  $\pm$  2 dB in the -20 to -35 dB range. (0 dB = 80 % full scale of the recorder output, operating on the self calibrated "linear" range of the ESUM)
- RF-Generator : HEWLETT PACKARD 8640B, 25-1000 MHz. The stability of the amplitude is specified to  $\pm 0.1$  dB and could be checked by a free space recording before and after an experiment at a specific frequency (over-all test revealed a stability of  $\pm$  0.5 dB at frequencies above 200 MHz, depending mostly on the antenna A1)
- X-Y-Recorder: BRYANS 26000 A3. The accuracy is specified to ± 1mm which could be affirmed by a test,
- Absorber material: blocks of 0.3 x 0.3 x 1 m. Standard absorbers of the PTT antennna development division, efficient above about 100 MHz. Has been used to attenuate sleeve currents on the feeding coaxial cable (see FIGURES 44 and 53).
- Coaxial cables: double shielded coax of 9 mm diameter with N-connectors. Standard materials of the AMF antenna development division. At frequencies below 30 MHz the RF-radiation leakage cannot be neglected if inefficient antennas and cable lengths in excess of 40 m are used.

## 8.5. ANTENNA SET-UP TESTING AND EXPERIMENTAL PROCEDURE

Field measurements with non-resonant antennas in proximity to the ground require careful preliminary tests in order to exclude artifacts. With a human test subject TEST 1 (FIGURE 45,  $\phi = 0.90, 180, 270^{\circ}$ ,  $d_{at} = 0$  to 4 m) and TEST 2 (FIGURE 45,  $d_{at} = 0.035, 0.077, 0.135$  m) have been performed at the frequencies 25,50,75,100,150,200,300,400,600,700,800 and 900 MHz. In October 1976 a further experiment with the three test bodies SUB, PHA and MET (see specifications in section 5.4.1) was performed according TEST 1 (FIGURE 45,  $\phi = 0$  and 180°,  $d_{at} = 0.035$  to 2 m) at 11 frequencies from 74 to 897 MHz. TABLE 50 shows the preliminary test preparations, TABLE 51 the antenna parameter and TABLE 52 the experiment check list.

1. LABORATORY PREPARATIONS
1.1. Computation and construction of antennas $A_1$ 1.2. Network analysis in anechoic chamber (TABLE 46) 1.3. Construction and testing of RF-chokes (TABLE 47) 1.4. Construction of rubber band goniometer with lawn anchor 1.5. Goniometer test with 100 m cable and strong RF disturbances
2. MEASURING SET-UP PREPARATIONS
2.1. Warming-up of RF equipment and recorder, initial calibration 2.2. Trackway mounting with levelling rod under 100 kg load 2.3. Goniometer mounting with levelling rod. (In this test series is $a=1m$ , corresponding to $d_{at}$ : $2 m \triangleq \beta$ : $63.43^{\circ}$ ) 2.4. Calibration of the $d_{at}$ scale with reference markers on trackway 2.5. Verification of the $d_{at}$ accuracy when rubber thread is attached on phantom at $h_1 = 1.2 m$ .
3. TRANSMISSION TEST (for each measuring frequency)
<ul> <li>3.1. Search for a free RF-channel</li> <li>3.2. Evaluation of antenna A<sub>1</sub> and RF choke, sleeve current tests.</li> <li>3.3. Checking if no obstacles are around within a radius of 50 m.</li> <li>3.4. Polarization and reciprocity test: <ul> <li>Transmitter out on A<sub>1</sub>, polarization A<sub>2</sub> = vertical , reading:</li> <li>Receiver input on A<sub>1</sub>, polarization A<sub>2</sub> = vertical , reading:</li> <li>Receiver input on A<sub>1</sub>, polarization A<sub>2</sub> = horizontal , reading:</li> <li>Transmitter out on A<sub>1</sub>, polarization A<sub>2</sub> = horizontal , reading:</li> <li>Transmitter out on A<sub>1</sub>, polarization A<sub>2</sub> = horizontal , reading:</li> <li>Transmitter out on A<sub>1</sub>, polarization A<sub>2</sub> = horizontal , reading:</li> <li>Transmitter out on A<sub>1</sub>, polarization A<sub>2</sub> = horizontal , reading:</li> <li>Receiver input on A<sub>1</sub> cable, polar. A<sub>2</sub> = vertical , reading:</li> <li>Receiv. in on " " " " = horizontal , reading:</li> <li>Receiv. in on A<sub>1</sub> cable, polar. A<sub>2</sub> = horizontal , reading:</li> </ul> </li> </ul>

TABLE 50 Preliminary test preparations and checks

- 124 -

The accuracy of the data obtained in the following antenna-body experiments depends directly on the results of the transmission test 3.4. and 3.5. in TABLE 50. The field strength at vertical polarization should be at least 15 dB higher than the field strength at horizontal polarization, and should be at least 20 dB higher than the noise or the residual signal picked-up when  $A_1$  (transmitting and receiving case) is replaced by a 50  $\Omega$  terminator. The reciprocity test reveals RF-leakage in the control center, theoretically the transmission in both directions should give the same reading. These important antenna parameters are listed in TABLE 51 :

Freq	A٦	RF	A <sub>l</sub> = tr	ansmit	ting ar	ntenna	A <sub>1</sub> = receiving antenna			
		choke	A2: V6	erti.	A2 : h0	oriz.	A <sub>2</sub> :ve	rti.	A <sub>2</sub> :ho	riz.
[MHz]	No.	No.	ANT [dB]	TERM [dB]	ANT [dB]	TERM [dB]	ANT [dB]	TERM [dB]	ANT [dB]	TERM [dB]
74	AT1	2x CH1	25	3	13	0	17	25	18	20
101	AT1	3х СН1	25	10	0	0	26	15	0	0
125	ATI	- 2x CH1	25	0	6	0	25	0	6	0
158	AT2	СН1	25	0	4	0	26	0	3	2
205	AT3	СН2	25	2	9	0	25	7	9	0
250	AT4	снз	25	0	2	0	25	0	3	0
300	AT4	снз	25	3	. 5	0	24	7	5	0
400	AT5	CH4	25	0	0	0	25	0	0	0
562	AT6	СН5	25	3	11	0	24	0	7	0
700	AT7	CH6	25	0	0	0	25	0	5	0
897	ATS	СН6	25	0	2	0	25	0	0	0

TABLE 51 Antenna parameter of the test set-up used in the later antennabody experiments. ANT: specified antenna  $A_1$  connected to generator or receiver, TERM:  $A_1$  replaced by a 50  $\Omega$  terminator. Frequency 74 allows no accurate measurements (but qualitative measurements for  $A_1$  = transmitting antenna), 101 MHz is at the border line of the applicability.

If we compare TABLE 51 with TABLE 46, one can predict that the frequency

300 MHz may cause difficulties, because AT4 resonates at 298 MHz. A change in the proximity of the antenna (body) will effect an important change of the antenna impedance. The specifications at 74 MHz are not satisfactory not only due to AT1, but also due to the lower frequency limit of 100 MHz of the remote antenna  $A_2$ .

The check list for the actual antenna-body experiment (TEST 1 in FIGURE 45) is shown in the next TABLE 52 :

ANTENNA-BODY EXPERIMENT CHECK LIST (for each measuring frequency) 4.1. Final calibration of RF-equipment and recorder 4.2. Repetition of points 3.4. and 3.5. according TABLE 50 4.3. Zero calibration of the goniometer 4.4. Calibration of the FSL (free-space level) on + 15 dB for A1 = transmitting antenna and  $A_2$  = vertical polarization 4.5. Experiments with the test bodies MET,PHA and SUB in the  $d_{at}$ interval 0.035 up to 2 m: (A2 = always vertical polarization) 4.5.1.1. MET,  $\phi = 0^{\circ}$ , A<sub>1</sub> = transmitting antenna 2. A<sub>1</sub> = receiving antenna 4.5.2.1. PHA,  $\phi = 0^{\circ}$ , A] = transmitting antenna 2. A] = receiving antenna 4.5.3.1. PHA,  $\phi=180^{\circ}$ , A<sub>1</sub> = receiving antenna 2. A1 = transmitting antenna 4.5.4.1. MET,  $\phi$ =180°, A<sub>1</sub> = transmitting antenna 2. A] = receiving antenna 4.5.5.1. SUB,  $\phi=180^\circ$ , A<sub>1</sub> = receiving antenna 2. A<sub>1</sub> = transmitting antenna 4.5.6.1. SUB,  $\phi = 0^{\circ}$ ,  $A_1 =$  transmitting antenna 2. A<sub>1</sub> = receiving antenna 4.6. When all test bodies are dislocated, reading of the FSL for A<sub>1</sub> = transmitting antenna. 4.7. Reading of the goniometer at  $d_{at} = 0$ .

TABLE 52 Antenna-body experiment check list (shortened)

The procedure for TEST 2 (FIGURE 45) is similar to TABLE 50, 51, and 52. In FIGURE 53 a picture is shown of such a TEST 2. The measured results are discussed in the next section together with the theoretical predictions. The sketch in FIGURE 54 is not only a joke : the directive gain from the bodies of three men amounts to about 4 to 6 dB as measured with a MOTOROLA HT 220 walkie-talkie with a 4 inch helix at 174 MHz.



## FIGURE 53

Test set-up for azimuthal radiation experiments

- 1: test antenna A<sub>1</sub>
- 2: spacer (d<sub>at</sub> = parameter)
- 3: electrical counterpoise
   with attenuator
- 4: absorber tubes (RF-choke
   within the tubes)
- 5: wooden antenna holder
- 6: wooden revolving stage
- 7: absorber blocks on feeding coaxial cable



FIGURE 54 "The soldier directive antenna". An application of the Gaing obtained at  $d_{at} = \lambda/4$  at  $\phi = 0^{\circ}$  and  $d_{at} = \lambda/2$  at  $\phi = 90$  and  $270^{\circ}$ .

## 9.1. INVESTIGATED PARAMETERS

9.1.1. EFFECT OF FREQUENCY AND BODY MATERIAL



FIGURE 55 Gain<sub>B</sub> versus f in the shadow zone from the three test bodies MET, PHA, SUB (measured data) compared with the computed data from IZYL. Parameter:  $d_{at} = 0.1$  and 0.2m.  $E_{v}$ : E-field strength, vertical polarization.

FIGURE 55 is a comparison between the experimental data for the three test bodies MET, PHA, SUB and the computational data for the IZYL (see definition of the bodies in 5.4.1.). Shown are the data for the constant parameter  $d_{at} = 0.1$  m and 0.2 m in the shadow zone  $\phi = 180^{\circ}$ , obtained from the Gain<sub>B</sub>/d<sub>at</sub> experiment in FIGURES 56,57 and 58.

The experiment, performed according to the check lists in 8.5., includes the experimental frequencies (74), 101, 125, 158, 205, 250, (300), 400, 562, 700 and 897 MHz. The results at (74) and (300) MHz are not accurate enough for a quantitative consideration, the reasons are mentioned below TABLE 51 and in section 9.2..

The computational data for the infinite cylinder (IZYL) are valid only for frequencies above 200 MHz for finite bodies of more than 1.8 m length, as explained in section 7.1.. The most interesting result is the parallelism of the Gain<sub>B</sub> curves for  $d_{at} = 0.1$  m and 0.2 m. The difference is almost constant versus the frequency and amounts to 4 ± 0.5 dB.

The experimental data reveal a similar tendency. For all test bodies the gain difference between  $d_{at} = 0.1m$  and  $d_{at} = 0.2m$  amounts to  $4 \pm 3 dB$  and often to  $4 \pm 1 dB$ . Thus, one may assume for a first approach that the measuring accuracy is better than  $\pm 3 dB$ .

The experimental data oscillate around the computed data, at frequencies below 200 MHz with a considerable amplitude and above 200 MHz with a much smaller amplitude. The experimental data reveal clearly four regions:

- $\lambda/2$  resonance at 74 to 101 MHz (exp. data of insufficient accuracy)
- 3  $\lambda/4$  anti-resonance at about 101 to 125 MHz
- $\lambda$  second resonance at about 125 to 158 MHz
- >  $\lambda$  off-resonance at frequencies above 200 MHz.

The MET and PHA have the same length of 1.8 m, the SUB is only 1.68 m long. It seems that the body material does not influence the resonant frequencies much, and that the anti-resonance is especially weaker in lossy materials.

Above 200 MHz the difference between experiment and theory is less than  $\pm$  3 dB for all three test bodies. Taking into account that field measurements in the shadow zone are very difficult due to the large field gradients, the correlation is encouraging.





FIGURE 56 Gain<sub>B</sub> versus  $d_{at}$  at  $\phi = 0$  and  $180^{\circ}$  from the three test bodies MET, PHA. SUB (measured data) compared with the computed data from IZYL.



FIGURE 57 Gaing versus  $d_{at}$  at  $\phi = 0$  and 180<sup>o</sup> from the three test bodies MET, PHA, SUB (measured data) compared with the computational data from IZYL (continuation of FIGURE 56).



FIGURE 58 Gain<sub>B</sub> versus  $d_{at}$  at  $\phi = 0$  and 180<sup>o</sup> from the three test bodies MET, PHA, SUB (measured data) compared with the computational data from IZYL (continuation of FIGURES 56 and 57).

FIGURES 56,57 and 58 are comparisons between the experimental data for the three test bodies MET,PHA,SUB and the computational data for the IZYL.

The experiments include the experimental frequencies 101, 125, 158, 205, 250, 400, 562, 700 and 897 MHz and were performed according to the check lists in section 8.5..

The computational data for the infinite cylinder (IZYL) are valid only for frequencies above 200 MHz, as explained in section 7.1.. The Gaing decreases with decreasing d<sub>at</sub> in the shadow zone  $\phi = 180^{\circ}$  and oscillates around the FSL in the irradiated zone  $\phi = 0^{\circ}$ , with maxima at d<sub>at</sub>  $\circ n \lambda/4$ , n = 1,3,5,... and minima at d<sub>at</sub>  $\circ n \lambda/2$ , n = 0,1,2,...

The experimental data agree qualitatively with the theoretical data for all frequencies and antenna-body distances. The quantitative agreement depends on the frequency range:

- Resonance region below 200 MHz. At  $\phi = 0^{\circ}$  the differences are smaller than 2 dB, except for 158 MHz at d<sub>at</sub> below 0.1m. At  $\phi = 180^{\circ}$  the differences amount up to about 5 dB at d<sub>at</sub> above 0.2m. Generally, Gain<sub>B</sub> decreases from SUB to PHA to MET. The somewhat larger differences at very small d<sub>at</sub>'s seem to be caused by near-field effects.
- Off-resonance region between 200 and 400 MHz. There is an excellent agreement between experiment and theory. Generally, the accuracy is better than 3 dB for all dat's and  $\phi$ 's. There are no significant differences between the three test bodies.
- High frequencies above 400 MHz. It should be mentioned that the test antenna  $A_1$  has a length of  $\sim \lambda/4$  in this frequency region, so that  $A_1$  cannot be regarded as an actual point source. However, the difference amounts to less than 4 dB. The data for MET and PHA are very similar but differ from SUB, leading to the hypothesis that the shape of the body becomes more important than the body material.

The Gain<sub>B</sub> versus  $d_{at}$  diagrams demonstrate clearly a formerly unknown, systematical relation between these two quantities.

Analogous experiments at  $\phi = 90^{\circ}$  and 270° revealed similar results as shown at  $\phi = 0^{\circ}$ , with maxima at  $d_{at} \stackrel{\sim}{=} n \lambda/2$ , n = 1,3,... as predicted. An application of these maxima at 0,90 and 270° is shown in FIGURE 54. Similar effects occur also, if a person approaches a mobile receiver (FM, 80 - 120 MHz) when tuned to a weak radio station (Try it with your radio !).

### 9.1.3. EFFECT OF THE AZIMUTHAL ANGLE

If one speaks of the influence of the human body on the radiation pattern of body-mounted antennas, one means generally the azimuthal radiation pattern. Such experiments were performed by many authors, e.g., BUCHANAN, MOORE and RICHTER [12], KING and WU [50], etc.. Generally, the published results differed greatly, offering hypotheses about directive properties of the human body and impedance changes.

After the discovery of the dominant  $d_{at}$  effect, the azimuthal radiation pattern is well explainable in the off-resonance region. FIGURE 59 shows a typical azimuthal radiation pattern recording for  $d_{at}$ =0.035, 0.077 and 0.135 m compared with the computed results at  $d_{at}$ =0.050 and 0.150 m :



FIGURE 59 Azimuthal radiation pattern at 600 MHz. Gain<sub>B</sub> versus  $\phi$  for the human test subject SUB compared with the computational data from IZYL.

The experimental data SUB,  $d_{at} = 0.077$  and 0.135 m are in between the computational data from IZYL,  $d_{at} = 0.05$  and 0.15 m. Thus, the agreement between experiment and computation is better than 3 dB for  $d_{at}$  above 0.05 m. The antenna-body system is an efficient directive antenna with a frontto-back ratio of up to 20 dB and a gain up to 2-3 dB. The main- and sidelobes are completely controlled by  $d_{at}$  for a given  $\phi$  (FIGURES 59, 38 and the other computer results in Appendix 16.2.1.). The experimental difference in gain and  $\phi$ , and the asymmetric shape of the curve at  $d_{at} = 0.035$  m may be caused by the asymmetry of the SUB and by an inclination of A<sub>1</sub>. A summary of the results obtained by TEST 2 (see FIGURE 45) is shown in TABLE 60. Listed are the minimum  $Gain_B$  measured with the human test subject SUB at  $d_{at}$  = 0.035, 0.077 and 0.135 m and the minimum  $Gain_B$  computed from IZYL :

MINIMUM GAIN AT VERTICAL POLARIZATION IN SHADOW ZONE FOR SUB AND IZYL													
Frequency [MHz]	25	50	75	100	150 2	200	300	400	500	600	700	800	900
SUB, Gaing dat=0.035m [dB]	0	-8	+2	-4	-18 -	-20	-23	-20	-22	<b>-</b> 28	-23	-21	-26
SUB, GainB d <sub>at</sub> =0.077m [dB]	0	-3	+4	-7	-13 -	-17	-21	-19	-18	-21	-21	-18	-24
SUB, Gaing $d_{at} = 0.135 \text{ m}$ [dB]	0	-4		-5	-12 -	-11	-15	-18	-17	-17	-21	~16	-20
IZYL,GainB d <sub>at</sub> =0.100 m [dB]	-	-	-	-		-15	-15	-16	-17	-17	-18	-18	-19

TABLE 60 Minimum Gain<sub>B</sub> at vertical polarization in the shadow zone 135<sup>o</sup>  $< \phi < 225^{o}$ . Comparison between measured data (SUB) and computational data (IZYL) at different antenna-body distances d<sub>at</sub>.

The experimental minima occur somethere between 135 <  $\phi$  < 225<sup>0</sup>. They are symmetrical for d<sub>at</sub> > 0.05 m (see also FIGURE 59, the most asymmetrical recording of the whole test series). In contrast, the computed minima are always near ± 165<sup>0</sup> (see also section 10.4.) in the IZYL-model. Nevertheless, the IZYL data agree within 3 dB with the averaged experimental data at d<sub>at</sub> 0.077 and 0.135 m. Taking into account the small signals (up to -26 dB below FSL), and the large field gradients, the agreement is satisfactory for frequencies above 200 MHz (IZYL-model limit).

### 9.1.4. EFFECT OF THE BODY MATERIAL

In the analysis 5.2.4. it was shown that the reflection coefficient for TEM, TE and TM waves is close to -1 for the E-vector and that probably differences occur only due to larger penetration depth  $\delta$ . In fact, the experimental results prove these hypotheses. The differences among MET, PHA and SUB are generally below  $\pm$  3 dB (FIGURES 56,57,58 and also 55) at d<sub>at</sub> above 0.1 m. Below 200 MHz and at extreme small d<sub>at</sub>'s there are somewhat larger differences caused perhaps by  $\delta$ , but also by the different shapes of the bodies. From the practical point of view, these differences

are of little interest, as long as the antenna-body distance is above approximately 0.050 m. Smaller  $d_{at}$  are only sensible with much smaller antennas, but one should take into account that  $Gain_B$  becomes very small and that the antenna gets detuned due to the extreme body proximity.

### 9.1.5. VERIFICATION OF THE RECIPROCITY THEOREM

The complete test series TEST 1 in section 9.1.2. was performed for both transmission directions ( $A_1$  = transmitting antenna /  $A_2$  = receiving antenna and  $A_1$  = receiving antenna /  $A_2$  = transmitting antenna). The transmission loss (or Gain<sub>B</sub>) differed only within ± 2 dB (usually ± 0.5 dB). This holds true for all three test bodies, for all frequencies above 100 MHz and for all antenna-body distances above 0.05 m. The RF-power was always below 1 mW; it might be that at higher power levels with considerable heating effects significant differences could occur, but such power levels are beyond our application.

### 9.2. DISCUSSION OF THE LIMITATIONS OF EXPERIMENT AND COMPUTATION

The agreement between experiment and theory is  $\pm$  3 dB at frequencies above 200 MHz and antenna-body distances above 0.1 m. Taking into account the large signal range from -26 dB up to + 4 dB, the agreement is more than satisfactory. A difference of  $\pm$  3 dB around the - 20 dB level corresponds to a power variation of only  $\pm$  1%, related to 0 dB = FSL = 100 %.

The relative simple IZYL model explains the off-resonance effects at frequencies above 200 MHz, at  $d_{at}$  above 0.1 m and for all test bodies.

The assumptions in section 5.2. could be verified. The reciprocity theorem is valid for our application as shown in 9.1.5.. The human body can be regarded as a perfectly conducting body with respect to scattering, at least for  $d_{at}$  above 0.05 m as shown in 9.1.1. and 9.1.2..

Experimental data at frequencies between 100 and 900 MHz (except 300 MHz) could be measured accurately at distances from 4 m up to 0.1 m. The lower frequency limit is mainly determined by the performance of the test antenna  $A_1$  and also by the remote antenna  $A_2$ . The main problem is the bad efficiency of electrically small antennas when operated off - resonance. Measurements with antennas tuned on resonance tend to be inaccurate as can be seen in FIGURE 55 at 300 MHz: the extreme loss with MET is caused by detuning effects due to body proximity, less accentuated with PHA and

SUB. The experimental lower limit for  $d_{at}$  is determined by the antenna dimensions and by the extremely low signal level in the shadow zone.

A problem related to all frequencies below 300 MHz is the insufficient  $\cdot$  counterpoise of the monopole antennas A<sub>1</sub>. With the precautions in FIGURE 44 the resulting radiation from the feeding coaxial cable could be attenuated. A better solution with dipole antennas and built-in RF-generators will be shown in section 11.3.

The IZYL-antenna model has proven its usability for two-dimensional offresonance computations at vertical E-field polarization at  $\theta_i = 90^\circ$ . An extension of the computations for frequencies at or below resonance, for arbitrary polarizations and arbitrary wave incidence is only possible with a finite body-antenna model. Such three-dimensional computations and experiments of verifications will be performed in the next sections.

# 10. THREE-DIMENSIONAL COMPUTATION OF SCATTERING FROM FINITE BODIES OF REVOLUTION

### 10.1. COMPUTATIONAL MODELS AND GOALS

The antenna-body models consist of perfectly conducting, finite bodies of revolution (FIGURE 28: Body models FZYL, MANMOD1 and MANMOD2), a small antenna A<sub>1</sub> positioned at h<sub>B</sub>,d<sub>at</sub> and  $\phi$ , and an incident plane wave with an irradiation angle  $\theta_i$  to the vertical axis of the body. The computational situation is shown in the FIGURES 14 and 33 : the E-field vector of the incident wave may be  $\theta$ -polarized ('vertical', def. 203) or  $\phi$ -polarized ('horizontal', def. 204); the E-field components at A<sub>1</sub> are computed in  $\vec{a}_{\theta}$ ,  $\vec{a}_r$  and  $\vec{a}_{\phi}$  directions by help of test segments located at RTEST and ZTEST, rotated with the angle  $\phi$  around the z-axis. With these data one obtains the vertical, radial and horizontal field components  $E_{V}$ ,  $E_r$  and  $E_h$  at A<sub>1</sub>.

The method of solution is described in section 6.4. and is based on the extended works of HARRINGTON and MAUTZ [40] and BEVENSEE [10]. The purpose of the following (very expensive) computation is to collect numerical data in the important frequency range 10 to 500 MHz (extended resonance region of man) with regard to near-field components which influence Gaing. The parameters of interest are the body geometry (actual shape of the body), frequency f, antenna-body distance dat, azimuthal angle  $\phi$ , antenna polarization  $p_1$  and  $p_2$ , incident irradiation angle  $\theta_1$  and the relative antenna height  $h_B$ .

## 10.2. COMPUTER PROGRAMS FOR NEAR-FIELD COMPUTATIONS

#### 10.2.1. GENERAL OVERVIEW

Due to storage capacity- and computational time limits the computation is split-up into three independent programs connected by one file:

- Program HARRA : computation of the Y-matrices
- Program PANB : computation of near-field data for some Aj-positions
- Program PANC : computation of the field homogeneity along A1.

These programs are written in Fortran IV for a CDC 7600 computer and require a minimum storage capacity of 160,000 octal in the core memory. The data are stored in a COLLECT FILE, catalogued in HARRA, read and extended in PANB and read in PANC. There is only an auxiliary print output in HARRA and PANB, because the final data are plotted by a special routine. The following subroutines are used several times in program HARRA, PANB:

- Subroutine LINEQ : replaces a 19 by 19 matrix Z by its inverse	
- Subroutine PLANE : provides the measurement matrices for the bod	у
of revolution and for the test segments	
- Subroutine PROGA : prepares the matrix variables in order to ac-	
commodate the Y-matrices read from the file	
- Subroutine REORD : arranges a number of values in descending or-	
der (needed if no plotter is available)	
- Subroutine NEARZ : computes the coordinates for the test segment	s,
similar to the body coordinates in HARRA.	

The complete programs and output samples are enclosed in Appendix 16.2.. Not enclosed is the plot routine, because its application is limited to the ETH computer center. The procedure to compute the near-field data can be summarized as follows:

- definition of the body geometry and frequency in HARRA
- computation of the Y-matrices for some modes and storage in a collect file in HARRA
- definition of the test segments (location and length) in PANB
- computation of the near-fields for each mode, manual test of the convergence of the results in PANB
- summation of contribution of each mode and for each azimuthal angle, manual checking of the minimum number of needed modes in PANB
- coordinate transformation of the results in order to obtain the E-field components in vertical, radial and horizontal direction, separate storage of the data in the collect file in PANB
- auxiliary output of Gaing for some significant cases in PANB
- reading of the file data, final processing and plotting or
- reading of the file data, final processing of the field homogeneity and printing in PANC.

### 10.2.2. PARAMETER DESCRIPTION

The input parameters are described in TABLE 61; the body parameters for FZYL, MANMOD1 and MANMOD2 are enclosed in the listings of PANC in Appendix 16.2.4. CAUTION: For convenience the computational frequency is ten times smaller than the actual frequency, and all computational dimensions are ten times larger than actual. This means that the input data

have to be scaled to the computational data. The output tables describing the body parameters, the tables with the complex field data for each test segment for the convergence test and the tables showing the contribution of each mode are presented in the computational scale. However, the final graphical outputs and the final result tables about the field homogeneity are presented in the actual scale.

INPUT	PARAMETER	MEANING	UNITS
BK	(REAL)	Computational wave factor = $21/\lambda_{actual} * 0.1$ .	[1/m]
DIESI		computational DTEST = actual size * 10 .	[m]
F	(REAL)	Computational frequency = f <sub>actual</sub> *0.1 in HARRA.	ÍMH z 1
F	(REAL)	Actual frequency used in PANC.	[MHz]
КК	(INTEGER)	Number of computed modes. KK = 8 means that the	
		ding Y-matrices are available.	
NN	(INTEGER)	Mode number. NN is the same as n appearing in (194).	
MNPHI	(INTEGER)	from $0^{\circ}$ to 180°. The standard NNPHI is 37.	
NP	(INTEGER)	Number of points describing the generating curve,	
		see FIGURE 32. The arc length t <sub>tot</sub> along this curve is divided in (NP-1)/2 = N tangential unit	
		elements t from one peak of a triangle function	
		to the next peak. It is <u>not</u> necessary that the body points are equally spaced.Standard NP is 41.	
NPHI	(INTEGER)	Number of equal subdivisions of the body $\phi$ axis	[
NT	(INTEGER)	from 0 to ¶. The standard NPHI is 20. Number of irradiation angles. The standard NT is	
	(INTEGEN)	1, but the program is prepared for larger NT's.	
NTEST	(INTEGER)	Number of test points (A) positions d <sub>at</sub> 's and here's). The standard NTEST is 9 for PANB and 5 for	
		PANC.	
RH	(REAL)	Radius of the points describing the generation	
		actual body radius * 10.	[m]
RTEST	(REAL)	Radius from the body axis to the test point, see	{
		radius * 10.	[m]
RUN	(INTEGER)	Control variable. If RUN = 1, program PANB com-	
		wave, if RUN = 2, of the ground reflected wave.	
ZH	(REAL)	Height of the points describing the generation	]
		the actual height * 10.	[m]
ZTEST	(REAL)	Height $h_B$ of the test point, see FIGURE 33. The	[ [m]
[		computational ZIESI IS nBactual * 10.	[m]

TABLE 61 Input parameter names used in programs HARRA, PANB and PANC

OUTPU Parami	T AND FILE ETER NAMES	MEANING		UNITS					
DA DAV DAR DI	(REAL) (REAL) (REAL) (REAL)	length of the body in logarithmic differenc logarithmic differenc antenna-body distance 22 def at PANE 336	length of the body in z-axis, actual scale logarithmic difference $\delta E, p_1$ vertical or hor. logarithmic difference $\delta E, p_1$ radial antenna-body distance dat, from body point No.						
DU	(REAL)	22, def. at PANB 336, actual scale diameter of the body, at body point No. 20, def. at PANB 332, actual scale							
DPV	(REAL)	phase variation $\delta \Phi$ al vertical or horizonta	phase variation $\delta \phi$ along the antenna axis, pj vertical or horizontal						
DPR EAV EAR	(REAL) (REAL) (REAL)	phase variation $\delta \Phi$ al logarithmic difference actual to averaged U <sub>i</sub>	ong antenna axis,pj rad. = ΔU,j pj vertical or hor. nd J pj radial	[º] [dB] [dB]					
EPH ERAD ETH	(COMPLEX) "	E-field $\vec{a}_{\phi}$ -direction " $\vec{a}_{r}$ " " $\vec{a}_{\theta}$ "	amplitude of one mode, of one test point, for $\phi = 0^\circ$ , $\theta_i$ , Einc and Einc	[V/m]					
EPTOT ERTOT ETTOT	(COMPLEX) "	E-field $\vec{a}_{\phi}$ -direction " $\vec{a}_{r}$ " " $\vec{a}_{\theta}$ "	added amplitudes up to the last computed mode, else identical to above	[V/m]					
EHTOT ESTOT EVTOT	(COMPLEX) "	E-field horizontal "radial "vertical	field strength components including all modes for all $\phi$ 's, one test point	[V/m]					
GHTOT GSTOT GVTOT	(REAL) "	E-field horizontal "radial "vertical	amounts of the components in dB, including all modes for all $\phi$ 's, one test p.	[V/m]					
PHTOT PSTOT PVTOT	(REAL)	Phase horizontal of radial E-field vertical	phases of the field com- ponents GHTOT,GSTOT and GVTOT	[°]					
HO IT ITE L M NST NZ POI	(REAL) (INTEGER) (INTEGER) (INTEGER) (INTEGER) (INTEGER) (INTEGER)	relative antenna heig control variable for index test point index irradiation ang index mode number of stored fina number of stored Y-el control variable:1=50	relative antenna height h <sub>B</sub> , actual size control variable for test segment orientation index test point index irradiation angle index mode number of stored final data:NTEST*NNPHI(H,S,V) number of stored Y-elements:(NP-3) <sup>2</sup> (Y1,Y2,Y3,Y4)						
SYH	(COMPLEX)	stored final data, ho stored final data, ra	rizontal components dial components	[V/m] [V/m]					
SYV Tj	(COMPLEX) (REAL)	stored final data, ver arc length along gener triangle function (HA	rtical components rating curve to the jth RRA) computational scale	[V/m] [m]					
X1 Y1-4 Z	(REAL) (COMPLEX) (COMPLEX)	irradiation angle $\theta_i$ Y-matrices elements ( $(Y_n^{\varphi\varphi})ij$ , first run n=	or $180^{\circ} - \theta_i$ , (RUN 1,2) $Y_{1}^{t+})ij$ , $(Y_{1}^{\phi})ij$ , $(Y_{1}^{t+})ij$ , (Y_{1}^{\phi})ij, 0, next run n = 1,2,NN	[°]					
S164M0 S164D9 F100D5	7 (FILE) (FILE) (FILE)	S = code MANMOD 2 ,164 = S = code MANMOD 2 ,164 = F = code FZYL, 100 =	164 MHz,M07 = max.mode 7 164 MHz,D9 = 9 $d_{at}$ 's 100 MHz,D5 = 5 $d_{at}$ 's						

TABLE 62 Output and file parameter names in programs HARRA, PANB, PANC

### 10.2.3. PROGRAM DESCRIPTION HARRA

The computer program HARRA is based on the theory in section 6.4.1. and 6. 4.2.. In the presented form (Appendix 16.2.2.) it computes the Z-matrices for the body MAMMOD2 for the actual frequency 164 MHz, performs the matrix inversion and stores the Y-matrices Y1 to Y4 for the modes 0 to 7 in the collect file AMAT2 with the code name S164M07.

The source program is the program A developed and well described by HAR-RINGTON and MAUTZ [40]. It runs on an IBM computer. Because our computer center is equipped with a CDC-computer, the source program has been altered to the present program HARRA. The modifications regard mainly the structure of the program parts, the conversion of some characters and the file generation. Only the input and file procedure are discussed here; details can be studied in the above mentioned source program description. Punched card data about mode number, number of body points,  $\phi$ -subdivisions, k-factor and frequency (computational scale) is read early in the main program:

50	READ (1,51) NN, NP,	NPHI, BK,	F	HARRA	47
51	FORMAT (313, E14.7,	F8.2)		HARRA	48

The number F is only used for punched card data identifications and is not used for computations. Because we need all modes from 0,1,2,...7 in our example, there are 8 data cards, only differing in the first number for NN. If the last mode is executed, the program stops due to the statement:

	IF (EOF (1)) 52,49	HARRA	49
49	CONTINUE	HARRA	50
52	STOP	HARRA2	259

Punched card data about the body contour (MANMOD2) is read later:

	READ (1,53) (RH(I),I=1,NP)	HARRA	51
	READ (1,53) (ZH(I), I=1, NP)	HARRA	52
53	FORMAT (10F8.4)	HARRA	53

For each mode a complete set of data is required. With an earlier reading of the number NP only one reading of the body data would be required, but for a better overview on the program the presented procedure is more convenient.

For each mode the Y-matrices are computed and printed separately. The

output (see example in Appendix 16.2.2.) consists of the listing of the input data, the 19 arc lengths along the generation curve and 4 Y-matrices of 19 x 19 complex elements (see theory in section 6.4.2.2.). The Y-matrices are stored on the collect file according to:

PERMF,AMAT2. (first run)	HARRA 4
PUBLIC,COLLECT. (first run)	HARRA 7
COLLECT,N=AMAT2. (first run)	HARRA 8
CATALOG,AMAT2. (first run)	HARRA 9
or	
ATTACH,AMAT2,PW (following runs)	HARRA 13
PUBLIC,COLLECT. (following runs)	HARRA 16
COLLECT,M=AMAT2. (following runs)	HARRA 17
PROGRAM HARRA (TAPE 6 = DISK)	HARRA 21
WRITE (6) (Z(I),I=1,NZ)	HARRA242
ADD, S164M07, DISK, RO.	HARRA334
LIST	HARRA335
END	HARRA336

Due to the LIST statement all existing names of the files in AMAT2 are listed.

## 10.2.4. PROGRAM DESCRIPTION PANE

The computer program PANB is based on the theory in section 6.4.5.. In the presented form (Appendix 16.2.3.) it reads the Y-matrices from the collect file, computes the current densities on the body of revolution MANMOD2 for the specified incident wave, computes the incident, the scattered and the total E-field at a specified test segment near the body, determines the field components in vertical, radial and horizontal directions, delivers some tables for accuracy considerations, prints an auxiliary output and stores the final data in the same collect file AMAT2 with the code name S164D9 (or in a new collect field HOMOG with the code name F100D5, see description program PANC).

The source program is the program HARRDF, developed and well described by BEVENSEE [10].HARRDF is an extension of the program D by HARRINGTON and MAUTZ [40], so that both reports have to be consulted for a detailed understanding. HARRDF runs on an IBM computer and delivers the field components in  $\vec{a}_{\theta}, \vec{a}_{\tau}$  and  $\vec{a}_{\phi}$  directions at one test point, for  $\phi = 0^{\circ}$  and for each single mode. The source program has been altered to the present program PANB. The modifications regard mainly the structure of the program parts, the preceding computation of the Y-matrices (now in HARRA), the conversion of some characters (CDC-notations), the  $\theta_i$ -handling, the extension on the computation of the  $\phi$ -dependence, the summation of the contributions of each mode, the coordinate transformation of the results, the output procedure and the file generation. A major problem was the reduction of the storage requirement from the original 662,513 octal to 160, 000 octal. Only the input and some important procedures are discussed here; the principle of the method is described in section 6.4.5. and the computational details in the above mentioned reports.

Punched card data about the number of modes, number of body points, $\phi$ -subdivisions, number of incident angles, number of test points, number of azimuthal steps, k-factor and wave origin is read early in the main program in computational scale:

50	READ (1,51)	KK, NP,	NPHI, NT,	NTEST, BI	K, RUN	PANB	62
51	FORMAT (613	, E14.7,	2X, I2)			PANB	63

Usually only one corresponding data card is necessary, but more than one is possible simliar to the input section in program HARRA. Next the card data about the body contour (MANMOD 2) are read:

	READ (1,53) (RH(I),I=1,NP)	PANB	67
	READ (1,53) (ZH(I),I=1,NP)	PANB	68
53	FORMAT (10F8.4)	PANB	69

Next an integer array TEXT(9) is filled with the letters for the words "VERTIKAL", "HORIZONTAL", "RADIAL", "DIREKTE EINSTRAHLUNG" (means direct irradiation) and "REFLEKTIERTE EINSTRAHLUNG" (means irradiation by a wave reflected from the ground) to be used in the output tables. The input data appear first in the output listing due to the statements

JA=RUN*3+1	PANB	79
•	•	
•	•	
WRITE (3,46) (ZH(I),I=1,NP)	PANB	91

The irradiation angle  $\theta_i$  is set to 80.78° (direct wave, RUN = 1) and 103.36° (reflected wave, RUN = 2) due to the statements

DT = 0.394	[DT = PI/(NT-1)]	PANB	111
DO 1 J=1,NT	[DO 1 J=1,NT ]	PANB	112
THR(J) = DT*(J-1) + 1.410	$[THR(J) = DT^{*}(J-1)]$	PANB	113
IF(RUN.EQ.2) THR(J) = DT*J+1.410		PANB	114

For equally spaced  $\theta_i$ -steps the [statements] must be used and all arrays concerning the  $\theta_i$ -variable (L=1,NT) have to be changed (PANB 39 to 50).

After some initial computations the punched card data about the 9 test points are read :

READ (1,49) (RTEST(J), ZTEST(J), DTEST(J), J=1,NTEST 49 FORMAT (3F8.4) PANB 174

For a given mode subroutine PLANE is called with its  $5^{th}$  argument IT=1 to compute the incident plane wave components VVR(1,J) on the J<sup>th</sup> field triangle on the body:

In contrast to the source program the Y-matrices are read from the collect file, called by PROGA which prepares only the matrices accommodations.

In the DO 41 loop E3(L,J) and E3(L,J+NM) measure the current densities for the L<sup>th</sup> incident angle of the  $E_{\theta}^{inc}$  (p<sub>2</sub> = vertical, see FIGURE 33) while E4(L,J) and E4(L,J+NM) measure corresponding current densities of the  $E_{\phi}^{inc}$  (p<sub>2</sub> = horizontal, see FIGURE 33) in t and  $\phi$  directions (FIGURE 32):

$$E3(L,J) = E3(L,J) + Y(J1)*VVR(1,I1)-Y(J2)*VVR(1,I2)$$
 PANB 221  
 $E4(L,J) = E4(L,J) - Y(J1)*VVR(1,I3)+Y(J2)*VVR(1,I4)$  PANB 222

With  $J_t$  and  $J_{\varphi}$  the current densities per unit length in azimuth and along t, respectively, at the peak of the J<sup>th</sup> triangle, for a certain  $\theta^{inc}$ ,

$$E_{\theta}^{\text{inc}} \qquad J_{Jt}(\phi) \begin{cases} \frac{E3(L,J)}{RH(J2)} & NN = n = 0 \\ \frac{E3(L,J)}{RH(J2)} 2 \cos n\phi & n \ge 1 \end{cases}$$
(240)  
polari-  
zation  $J_{J\phi}(\phi) \begin{cases} 0 & n = 0 \\ \frac{E3(L,J+NM)}{RH(J2)} j 2 \sin n\phi & n \ge 1 \end{cases}$ (241)

These formulas are similar for  $E_{\phi}^{inc}$  polarization (see BEVENSEE [10]) and can be used to compute the current density  $\vec{j}$  in equation 128 at any point on the body surface, if one sums the contributions of all modes for each azimuthal angle separately.

In the following discussions only the  $E_{\theta}^{inc}$  polarization will be considered, but all the data from  $E_{\phi}^{inc}$  are computed, printed and stored. With E3(L,J)
stored for a given mode, for each incident angle 1, and for each triangle function J, the scattered field for that mode can be determined from them for all test segments in succession. Thus, the DO loop 7100 over the test segments, NEARZ determines the near-field matrix ZM for a given test segment, and PLANE yields its sampled incident field for all angle of incidence. Both test segment orientations for IT = 1 and IT = 2 are treated:

The test segment fields are approximated according to equation 220. The ESC(IT,1) and ESC(IT,2) at PANB 257 are proportional to the scattered field mode amplitudes in the  $\vec{a}_r$  and  $\vec{a}_{\phi}$ -directions for IT = 1, and in the  $\vec{a}_{\theta}$  and  $\vec{a}_{\phi}$ -directions for IT = 2, respectively. From statement 702 to 711 the incident, the scattered and the total E-fields in  $\vec{a}_r$  (ERAD),  $\vec{a}_{\theta}$  (ETH) and twice in  $\vec{a}_{\phi}$  (EPH) - directions are computed and printed for the regarded mode M, irradiation angle L, test segment ITE and both wave polarizations. The use of the two EPH's will be discussed in section 10.3.4.2.

With the advice of BEVENSEE [10] the program has been extended to compute the total field components for different azimuthal angles. The key formulas are: (ERAD:ETR, ETH:ETH, EPH: ETP1 and ETP2 inside the program)

E <sup>tot</sup> (r <sub>T</sub> ,z <sub>T</sub> ,φ)	KK-1 = ∑ ETR(M,ITE,L) M=0	$\cos M_{\phi}, E_{\theta}^{inc}$ j sin M $_{\phi}$ , $E_{\phi}^{inc}$	(242)
$E_{\theta}^{tot}(r_T, z_T, \phi)$	KK-1 = ∑ ETH(M,ITE,L) M=0	$\begin{array}{c} \cos \ M\varphi, E_{\theta}^{inc} \\ j \ sin \ M\varphi, \ E_{\phi}^{inc} \end{array}$	(243)
$E_{\phi}^{tot}(r_{T}, z_{T}, \phi)$	KK~1 = ∑ ETP½(M,ITE,L) M=0	j sin M $\phi$ , E <sup>inc</sup> cos M $\phi$ ,E <sup>inc</sup>	(244)

These computations are performed by the statements

909 DO 506 M=1,KK	PANB 361
NN = M-1	PANB 362
ERTOT = ERTOT + ETR(M,ITE,L)*COPHI(M,J)	PANB 363
ETTOT = ETTOT + ETT(M,ITE,L)*COPHI(M,J)	PANB 364
EPTOT = EPTOT + ETP(M,ITE,L)*SIPHI(M,J)	PANB 365

where ETP is the averaged value of the two EPH's and the trigonometric functions COPHI and SIPHI are the previously computed numbers:

$EPR = (ETP1R+ETP2R)/2 \ EPI = (ETP1I+ETP2I)/2$	PANB	295
ETP(M, ITE, L) = COMPLX(EPR, EPI)	PANB	296
COPHI(M,J) = COS(NN*PHI(J))	PANB	189
SIPHI(M,J) = U*SIN(NN*PHI(J))	PANB	190

In order to monitor the contribution of each mode (see later in section 10.3.4.1.) the continuously summed up field components are listed in the output for  $\phi = 0$ , 90 and 180°. The sample output shows the results for one  $\theta_i$ , the 9 test points (h<sub>B</sub> = constant = 1 m, d<sub>at</sub> = DIST = parameter), for  $E_{inc}^{inc}$  and for the actual frequency 164 MHz, according to:

40	CONTINUE	PANB	330
	F=3000.*BK/(2*PI)	PANB	331
913	IF(J.EQ.19) WRITE (3,545) NN,ERTOT,EPTOT,ETTOT CONTINUE	PANB PANB	376 378

From the last ERTOT, EPTOT, ETTOT the corresponding field components are computed in vertical, horizontal and radial directions : EVTOT, EHTOT and ESTOT. They are also stored as SYV(JDI), SYH(JDI) and SYS(JDI) :

JDI=0	PANB	350
JDI=JOI+1	PANB	379
EVTOT = ERTOT * COTN(ITE) - ETTOT * SITN(ITE)	PANB	380
SYV(JDI) = EVTOT	PANB	381
EHTOT = EPTOT	PANB	382
SYH(JDI) = EPTOT	PANB	383
ESTOT = ERTOT * SITN(ITE) + ETTOT * COTN(ITE)	PANB	384
SYS(JDI) = ESTOT	PANB	385

The numbers COTN(ITE) and SITN(ITE) are the sin and cos of the a-vector already computed in PANB 177 for the different test points. Next the field components are scaled in dB and the phase angles are computed:

	PP = CMPLX(1.0E-32, 1.0E-32) QQ = 1.0E-32	PANB	333 334
	IF(CABS(EVTOT).LT.QQ) EVTOT = PP	PANB	386
	GVTOT(ITE,L,J) = 20.*ALOG10(CABS(EVTOT))	PANB	389
	PVTOT(ITE,L,J) = ATAN2(REAL(EVTOT),AIMAG(EVTOT))*PR	PANB	392
504	CONTINUE	PANB	396

For the later use of the results in the plot programs or in PANC, the complex original data are stored in the collect file:

ATTACH,AMAT2,PW	PANB	4
CALL,S164M07,P=AMAT2,B=DISK. (for Y-matrices)	PANB	5
PROGRAM PANB (TAPE 6 = DISK, TAPE 7 = RESULT)	PANB	13
WRITE (7) (SYV(1),I=1,NST)	PANB	397
WRITE (7) (SYH(I),I=1,NST)	PANB	398
WRITE (7) (SYS(I),I=1,NST)	PANB	399
ADD, S164D9, RESULT, RO. (or S100D5 for later PANC)	PANB	768

The remaining of the program PANB is concerned with the graphical output of the amplitude and the phase of  $Gain_B$ . Due to the statements

only the results for  $\phi = 0$ , 90 and 180° are printed. By changing these cards into D0 918 I=1,37 and J = I all azimuthal results would be printed. Only the p<sub>1</sub> = vertical polarization is executed by the output procedure. By changing PANB 402, 403 and 431 and duplicating the program from PANB 404 to 463 (change labels) one also obtains the other p<sub>1</sub>-data. The  $E_{\phi}^{\text{inc}}$  output (p<sub>2</sub> = horizontal) is already incorporated in the program due to the D0 loop 912 in PANB 239,but not shown in the sample output.

Similarly, the results for the next program PANC are computed for all  $p_1$  and  $p_2$ . The only difference is, that the results are stored in the collect file HOMOG, and that only 5 test points are needed.

The computer program PANC is an extension to program PANB and is based on the investigation in section 5.2.2.. It reads the vertical polarized Edata of P1, P4, P5, the radial polarized E-data of P1, P2, P3 and the horizontal polarized E-data of P1 (see FIGURE 63) for all azimuthal angles 0  $\leq \varphi \leq$  180° (J = 1, NNPHI) from the collect file HOMOG. It computes the horizontal E-data of P6 and P7, so that the fields at the center and at the ends of a dipole antenna of 2h = 0.1m are available for  $p_1 = ver$ tical/radial at p2 = vertical and p1 = horizontal/radial at p2 = horizontal. Then it computes the amplitude variation  $\delta E$  and the phase variation  $\delta \Phi$  along the antenna polarization axes, the logarithmic difference ∆U between the induced voltage computed by Ecenter • 2h and the numeric integral  $\int_{-h}^{+h} E(\xi) \cdot d\xi$  where  $\xi = p_1$ , and prints all data versus  $\phi$  in tables which can be directly applied for field homogeneity considerations and antenna design. The sample program is specified for the body model FZYL at 100 MHz and  $d_{at} \approx 0.1 \text{ m}$ . The listing of the program and results for the frequencies 65,75,100,125,300 and 425 MHz are enclosed in Appendix 16.2.4..



### FIGURE 63

Body model FZYL and antenna A<sub>1</sub> with its center at P1. Antenna polarization p<sub>1</sub>: p<sub>1</sub> = vertical : P4, P1, P5 p<sub>1</sub> = radial : P2, P1, P3 p<sub>1</sub> = horizontal: P6, P1, P7 Antenna length  $2h = 2\Delta = 0.1 \text{ m}$ Antenna height  $h_B = 1 \text{ m}$ Ant.body dist.dat = 0.1 m Body diameter  $D_B = 0.25 \text{ m}$ Body length  $L_B = 1.8 \text{ m}$  There is only one punched card data read concerning the actual frequency:

The other parameters as used in PANB for the field computations are defined by statements:

DA is the body length, DI(1) the dat of the center of the antenna, DU the body diameter, HO the relative antenna height h<sub>B</sub>, X1 the irradiation angle  $\theta_i$ , NNPHI the number of azimuthal  $\phi$ -steps and NST the number NTESTx NNPHI of stored data for each p1 at p2 = vertical (POL = 1) and horizontal (POL = 2). In order to compute the missing field data at P6 and P7 the arc length CS between P1(J) and P1(J+1) is determined by

$$CS = (DU/2.+DI(1))*PI/(NNPHI-1)$$
 PANC 50

In FIGURE 61 the approximation method is shown for the computation of the field data at P6 and P7:



FIGURE 64 Approximation method for the E-field data at P6 and P7 ( $p_1$  = horizontal, amplitude |E| and phase  $\Phi$ ) from the data at P1(J-2), P1(J) and P1(J+2). Correction factor FAC : 2h/4CS where CS is the arc length between P1(J) and P1(J+1) |E| at P6 : |E| at P1(J) - FAC\*D1 |E| at P7 : |E| at P1(J) - FAC\*D2  $\delta\Phi$  : FAC\* $\delta\Phi$ '

The correction factor is computed by (here  $2h = d_{at} = 0.1 m$ )

FAC = DI(1)/(4.\*CS)

PANC 51

In the first run POL =1 (PANC 58) the  $p_2$  = vertical- data are read from the collect file by:

READ	(6)	(SYV(I),	I=1,NST)	PANC	59
READ	(6)	(SYH(I),	I=1,NST)	PANC	60
READ	(6)	(SYS(I),	I=1,NST)	PANC	61

After the statements describing the output the field data are rearranged.

In the D0 loop 700 the J-dependent variables ITE1 for P1, ITE2 for P2,.. ITE5 for P5 are computed by the statements PANC 97 to 101. For each azimuthal angle  $0 \le \phi \le 180^{\circ}$  (J=1,NNPHI) the amounts of the fields at P1,P5,P4 (vertical components) and at P1,P3,P2 (radial components) are computed according to

AMV1 = CABS(SYV(ITE1)) \$ AMR1 = CABS(SYS(ITE1))**PANC 114** \$ PANC 115 AMV5 ITE5 AMR3 ITE3 11 **PANC 116** AMV4 ITE4 \$ AMR2 ITE2

The phases of the fields at these points are computed due to

PV1 = ATAN2(REAL(SYV(ITE1)),AIMAG(SYV(ITE1))) \* PR PANC 117 : :

The phase differences  $\delta \phi$  along the p<sub>1</sub>-vertical/radial axes is computed by:

DPV1 = ABS(PV5-PV4) \$ DPV2 = 360.-DPV1 PANC 123

Then the dB-values of the fields are computed according to

The numerical integration of the E-field along the antenna axis is simply the sum of (AMV5+AMV1)•h/2 + (AMV4+AMV1)•h/2 and corresponds to the actual induced voltaged U<sub>ind</sub> (see section 5.2.2.). The mean induced voltage  $\overline{U}_{ind}$ is AMV1•2h. The logarithmic difference  $\Delta U$  in dB is 20 log ( $\overline{U}_{ind}/U_{ind}$ ) :

The amplitude variation  $\delta E$  is computed according to

$$DAV = ABS(AV5-AV4) + SDAR = ABS(AR3-AR2)$$
 PANC 138

After the checking of the size of AV1 and AR1 (PANC 139 and 142) the results  $\phi$ , E(dB) center,  $\phi$  center,  $\Delta U$ ,  $\delta E$  and  $\delta \Phi$  are printed for p<sub>1</sub> = vertical and radial at p<sub>2</sub> = horizontal due to the statement PANC 145.

In the second run POL= 2 (PANC 58) the  $p_2$  = horizontal data are read from the collect file by the same statements PANC 59,60 and 61. After the output statements the field data are rearranged as follows in DO loop 700:

IH1 = J-2 \$	IH2 = J+2	<pre>\$ NC = NNPHI-1</pre>		PANC 94
IF(J.EQ.1)	IH1 = J+2	\$ IF(J.EQ.2)	IHI = J	PANC 95
IF(J.EQ.NC)	IH2 = J	<pre>\$ IF(J.EQ.NNPHI)</pre>	IH2 = J-2	PANC 96

At POL = 2 the vertical components SYV(J) are not used. In order to apply the same procedure as for POL = 1 the horizontal components around the center point P1 ( $\phi = \pm 10^{\circ}$ , see FIGURE 64) P1(J-2), P1(J) and P1(J+2) are transferred into the vertical component arrays of P1, P5 and P4:

	IF(POL.EQ.1) GOTO 4	PANC	103
	SYV(ITE1) = SYH(ITE1)	PANC	104
	SYV(ITE5) = SYH(IH1)	PANC	105
	SYV(ITE4) = SYH(IH2)	PANC	106
ļ	CONTINUE	PANC	107

Now the amplitude difference D1 and D2 (FIGURE 64) is computed and the approximated amounts at P6 (now called P5) and P7 (now called P4) are determined, and the phase difference  $\delta \Phi$  is obtained by  $\delta \Phi^+ \cdot FAC$ :

2

	IF(POL.EQ.1) GOTO 5	PANC 127
	D1 = AMV1 - AMV5  \$ $D2 = AMV1 - AMV4$	PANC 128
	AMV5 = AMV1-D1*FAC $AMV4 = AMV1-D2*FAC$	PANC 129
	IF(AMV5.LT.00) AMV5 = 00  \$ $IF(AMV4.LT.00) AMV4 = 00$	PANC 130
	DPV = DPV*FAC	PANC 131
5	CONTINUE	PANC 132

The rest of the procedure is analogous to the computation of the field parameters at  $p_2$  = vertical.

The output consists of two tables. The first table contains the field data  $p_1$  = vertical/radial at  $p_2$  = vertical , and the second table contains the field data  $p_1$  = horizontal/radial at  $p_2$  = horizontal. The model body and the antenna positions are explained in the table head. The data are ranging from 0 <  $\phi$  < 180°. The output parameters are:

GAIN CENTER DB	<pre>= amount of the field at Pl in [dB]</pre>
PHASE CENTER DEG	= phase of the field at Pl in [ <sup>0</sup> ]
MEAN ERROR DB	= logarithmic difference $\Delta U$ in [dB]
MAXIMUM GAINVAR DB	= logarithmic difference $\delta E$ in [dB]
MAXIMUM PHASEVAR DE	G= phase variation δΦ in [ <sup>0</sup> ]

The results of PANC are discussed in section 10.3.5.1.

### 10.3. INVESTIGATION OF PROGRAM LIMITATIONS AND COMPUTATIONAL ACCURACY

## 10.3.1. PRJGRAM LIMITATIONS

The limitations of the program HARRA and PANB are defined by BEVENSEE [10] and HARRINGTON and MAUTZ [40] and are interpreted as follows:

- The programs are written for perfectly conducting bodies of revolution in free space.
- The wave-number k =  $21/\lambda$  should be such that the peaks of the triangle functions are not more than  $\lambda/21$  apart. For a given  $\lambda$  this condition determines the number NP (number of points describing the generation curve). The arc length along the generation curve  $t_{tot}$  (see FIGURE 32) is divided in HARRA by (NP-1)/2 = N tangential unit elements t of equal lengths. This length should not be larger than  $\lambda/21$  in order to prevent field oscillation between two peaks of the triangle functions. If we assume the standard NP of 41 and an arc length of the human body of  $t_{tot} = 2m$ , we obtain the maximum permissible frequency  $f_{lim5}$ :

$$f_{1im5} = \frac{(NP-1)/2 \cdot c}{t_{tot} \cdot 2\pi} = 500 \text{ MHz}$$
 (243)

- The number of subdivisions of the  $\phi$ -axis, NPHI, should be large enough so that  $\P(n_{\max})/NPHI < 1 \text{ rad}, n_{\max}$  being the number of the last azimuthal mode employed. In addition,  $(2\P/\lambda) (\P/NPHI) \cdot \rho_{\max} < 1 \text{ rad}$ ,  $\rho_{\max}$  being the maximum radial cylindrical coordinate of the body contour. These conditions are interpreted as follows:
- We assume a plane wave at  $\theta_i = 90^\circ$ ,  $\vec{k} || x-axis$ . The  $\phi$ -axis is subdivided in  $1 \cdot n_{max}$ /NPHI sectors with the arc length w and the angle  $\phi_W$ . The phase along w should not vary more than  $\lambda/21$  in order to prevent field oscillations within a sector. For small  $\phi_W$  the projection of w to the  $\vec{k}$ -axis amounts to w sin  $\phi$ . The maximum permissible  $n_{max}$  for a given f = 300 MHz,  $\rho_{max} = 0.125 m$  and NPHI = 20 for all sectors amounts to:

$$n_{\max} = \frac{\lambda \cdot NPHI}{2\pi \cdot \pi \cdot \rho_{\max}} \cdot \frac{1}{\sin \phi} = 8$$
(244)

Equation (244) is valid for all  $\phi$ . If we consider the interesting special case  $\phi = 0^{\circ}$  (or  $180^{\circ}$ ) and  $d_{at} < 0.2 \text{ m}$ , the sectors with  $\phi$  near  $0^{\circ}$  (or  $180^{\circ}$ ) are of greater influence than those with  $\phi$  near 1/2. Thus,  $n_{max}$  may be as large as 10 for 500 MHz without large accuracy loss.

- The convergence of the computed current densities to their correct values along the surface increases with NP and NPHI and should be rapid if both circumferences (2.t<sub>tot</sub> and  $\rho_{max}$ .21) remain <  $\chi$ . In our case with  $\theta_i \sim 90^\circ$  the body radius sets the limit. With a  $\rho_{max}$ =0.125 m the convergence may worsen at frequencies above 380 MHz.
- Near-field computations of a <u>point</u> field with a test segment tend to be inaccurate unless both these conditions are fulfilled:
  - a.) minimum distance of the test segment center to the body surface >  $\lambda$

b.) test segment length <  $\lambda/4$ .

When the test segment is very near to the body surface it usually does not measure the <u>point</u> field accurately. But if it has a length equal to one of the triangle functions it measures, at the position of that triangle function, the integral of electric field according to the network equation obtained with that function. The length of a triangle element is about 0.09 m (actual) and the length of a test segment is  $4 \cdot DTEST = 0.08$  and 0.20 m (actual) which is smaller than  $\lambda/4$  at frequencies below 375 MHz. The distance of the test segment to the body surface, however, is generally much smaller than  $\lambda$  (d<sub>at</sub> from 0.05 m to about 1 m). Thus, computations have to be performed with different test segment lengths, at test frequencies around the wanted frequency and the influence of each mode has to be monitored.

The program PANC is limited on  $d_{at} = 0.1 \text{ m}$  but is valid for all frequencies. For other  $d_{at}$ 's see the program description in 10.2.5..

### 10.3.2. COMPUTATIONAL TIME LIMITATIONS

With the standard NP of 41 and the standard NPHI of 20 one obtains for each mode 4 matrices of the size  $19 \times 19$  (complex elements). The computational time depends on the matrix inversion time and thus on the matrix size and on the number of modes. As an example program HARRA requires for the computation of 9 modes at the computational frequency 16.4 MHz 325 sec on a CDC 6500. The execution of program PANB at the same frequency for 7 test points with the modes 0 to 7 requires 320 sec. The execution of program PANC needs only about 5 sec. Thus, one should limit  $n_{max}$  on the absolutely needed number (increases with dat and frequency, see 10.3.4.1.), and one should limit the number of test segments, especially those with large dat.

### 10.3.3. STORAGE CAPACITY LIMITATIONS

The original program by BEVENSEE [10] requires a total storage of 662,513 octal which exceeds the permissible limit of 160,000 octal of the ETH computer by a factor of 4. With the assistance of BEVENSEE and the specialists at the computer center the storage requirement could be reduced. The main steps of reduction were:

- Separate computation of the Y-matrices in HARRA
- Reduction of the ZM-matrix from (2x10,000) to (2x76)
- Reduction of NT to 2, KK to 13 and NTEST to 9
- Reduction of the VVR from (2x14,400) to (2x760), Y from (10,000) to (1,444), and G from (30,603) to (4764)
- Extensive use of the COMMON BLANK and COMMON/A

The reduction of the array sizes and the COMMON operations is critical with respect to writing over the reserved array lengths. Several debugging procedures are needed which cannot be discussed here.

The final storage requirements of the programs in Appendix 16.2. are:

- Program HARRA: 61,200 octal (specified 70,000)
- Program PANB : 121,600 octal (specified 130,000)
- Program PANC : < 10,000 octal (specified 70,000)

### 10.3.4. INVESTIGATION OF THE COMPUTATIONAL ACCURACY

### 10.3.4.1. MINIMUM MODE NUMBER KK

The preliminary study in section 7.2.3. has shown that up to 25 modes contribute to the total field at  $f \le 1000$  MHz and  $d_{at} \le 2$  m. It is not possible to compute so many modes due to the limitations mentioned above. With the following method the absolutely needed minimum KK is evaluated:

Program PANB offers for  $p_1$  = vertical and horizontal a table denoted as "Einfluss der Anzahl der berücksichtigten Modi auf E<sup>tot</sup>" (see Appendix 16.2.3.) For each test segment the values for ERTOT,EPTOT and ETTOT are listed for the azimuthal angles  $\phi = 0$ , 90 and 180° as the summation from the modes 0 to M, where  $0 \le M \le (KK-1)$ . In each column one checks first if the (KK-1) result is >10<sup>-3</sup> and if the (KK-2) result differs not more than ±1%. Then one chooses that result in the column which is not more than ±5% different from the (KK-1) result and notes the corresponding minimum M.

TEST No.	SEGMENT d <sub>at</sub> [m]	ERT( 0° M	0T-c 90° M	omp.at 180º M	EPT 0° M	0T-c 90° M	omp.at 180º M	ETT( 0° M	0T-c 90° M	omp.at 180º M	Recom- mended KK(M+1)
1	0.08	3	1	3	0	3	(3)	1	2	3	<u>&gt;</u> 4
2	0.13	2	2	3	0	3	(3)	2	2	2	<u>&gt;</u> 4
3	0.18	3	4	3	0	3	(4)	2	2	3	<u>&gt;</u> 4
4	0.28	3	4	4	0	3	(4)	3	3	3	<u>&gt;</u> 5
5	0.38	3	2	4	0	3	(4)	4	3	4	<u>&gt;</u> 5
6	0.53	4	4	4	0	3	(4)	4	4	4	<u>&gt;</u> 5
7	0.68	4	4	5	0	5	(5)	4	4	6	<u>&gt;</u> 7
8	0.83	6	5	6	0	5	(6)	5	4	6	<u>&gt;</u> 7
9	1.03	7	6	6	0	7	(7)	6	6	7	<u>&gt;</u> 8

TABLE 65 shows the evaluation of the minimum mode number from the data in Appendix 16.2.3. for the actual frequency 164 MHz and for 5 % relative accuracy.

TABLE 65 Determination of the minimum mode number KK for 5% relative accuracy at 9  $d_{at}$ 's at the actual frequency 164 MHz and model MANMOD2.

In general it is sufficient to monitor the results at  $\phi = 180^{\circ}$ . In the example in TABLE 65 (164 MHz,  $p_1 = vertical$ ,  $\theta_1 = 80.8^{\circ}$ ) the horizontal component EPTOT is very small ( $10^{-7}$ ) so that those differences are of little meaning. Very roughly the minimum KK increases linarly with  $d_{at}$  and frequency. The computational data in section 10.4. at  $d_{at}$  from 0.1 to 0.4 m were computed with the NN and KK listed in TABLE 66, checked according to TABLE 65 at a 2% level below 300 MHz and 5% level above 300 MHz.

FREQ	UENCY RANGE	MAXIMUM NN in HARRA	SELECTED KK in PANB
1	to 30 MHz	4 ok	5 ok
50	to 100 MHz	5 ok	6 ok
101	to 200 MHz	7 ok	8 ok
250	to 300 MHz	9 ok	10 ok
350	to 500 MHz	10 ∿ok	11 ∿ok
600	to 800 MHz	12 (?)	13 (?)

TABLE 66 Mode number NN and KK versus frequency at  $d_{at}$  from 0.1 to 0.4 m. At frequencies below 500 MHz the relative accuracy is better than 5 %, the absolute accuracy (FSL = 100 %) is better than 0.2 %.

# 10.3.4.2. DIFFERENCE BETWEEN THE TWO AZIMUTHAL FIELD COMPONENTS

Program PANB computed the azimuthal field components EPH twice (for E<sup>inc</sup>, E<sup>scat</sup> and E<sup>tot</sup>, for  $\phi = 0^{\circ}$ ), first for IT = 1 (test segment oriented along the  $\vec{a}_r$ -vector) and then for IT = 2 (test segment oriented along the  $\vec{a}_{\phi}$ -vector) for both incident polarizations ETHETA INC (p<sub>2</sub> = vertical) and EPHI INC (p<sub>2</sub> = horizontal). The results are listed after the output of the body contour (see Appendix 16.2.3.) as follows:

 $\theta$  SIG  $\theta\theta$  MAG S $\theta\theta$  SIG  $\phi\phi$  MAG S $\phi\phi$  MODE NN = 0

RTEST = 1.7500 ZTEST = 10.0000 DTEST = 0.2000

EINC ETHETA INC ERAD(x,x), EPH (x,x), ETH(x,x), EPH (x,x) ESCAT ETHETA INC ERAD(x,x), EPH (x,x), ETH(x,x), EPH (x,x) ETOT ETHETA INC ERAD(x,x), EPH (x,x), ETH(x,x), EPH (x,x)

etc. for EPHI INC

RTEST = 2.2500 ZTEST = 10.0000 DTEST = 0.2000

etc. for all RTEST up to

RTEST = 11.2500 ZTEST = 10.0000 DTEST = 0.2000

and the same output is now repeated for all modes up to MODE NN = 7

The data in Appendix 16.2.3. were used as an indicator for the computational accuracy. The results, obtained with the following method, are listed in TABLE 67 :

Error in  $0/00 = ||EPH1| - |EPH2|| \cdot 1000 E_0$  (246)

The results in TABLE 67 are therefore related to the free-space level = 1000 O/oo = 0 dB. Generally, the error increases with decreasing dat, and the first modes determine the final accuracy. Below 500 MHz an increase of the error could not be noticed depending on the frequency. In normal conditions (see next section) the total error is well below 1 %.

TEST No.	SEGMENT d <sub>at</sub> [m]	Differ NN=0	ence be NN = 1	tween t NN = 2	he two NN = 3	EPH in NN = 4	<sup>0</sup> / <sub>00</sub> at NN = 5	164 MH NN = 6	lz. Mode: NN = 7
1	0.08	0	1.926	0.257	0.029	0.016	<0.03	~0	~0
2	0.13	0	0.273	0.103	0.045	0.021	∿0	∿0	<b>∿</b> 0
3	0.18	0	0.011	0.065	0.041	0.024	∿0	∿0	∿0
4	0.28	0	0.061	0.060	0.024	<0.03	$\sim 0$	∿0	∿0
5	0.38	0	0.053	0.047	0.012	∿0	∿0	∿0	∿0
6	0.53	0	0.034	0.036	<0.03	∿0	∿0	∿0·	∿0
7	0.68	0	0.029	0.017	∿0	∿0	$\sim 0$	∿0	~0
8	0.83	0	0.026	<0.03	~0	∿0	∿0	∿0	∿0
9	1.03	0	<0.03	∿0	~0	∿0	∿0	∿0	$\sim 0$

TABLE 67 Computational error in  $^{\rm O}/{\rm oo}$  versus mode number NN and versus  $d_{at}$  at the actual frequency 164 MHz and model MANMOD2.

Considering the results in TABLE 67 one could conclude that the computational accuracy is satisfactory for all applications, but this is not absolutely true. The exceptions are mentioned in 10.3.4.3..

### 10.3.4.3. DIFFERENCE BETWEEN RESULTS AT DIFFERENT TEST SEGMENT LENGTHS

Due to the approximation method by BEVENSEE [10] (see description in 6.4. 5.1. and program limitations in 10.3.1.) problems may occur at some few frequencies at small  $d_{at}$ . Without checking carefully the result according to the method in 10.3.4.2. the computational error had to be specified to about 10 %. With the following method inaccurate results can be detected easily:

All computations are performed twice, first with a test segment length of 0.08 m (DTEST = 0.2) and second with 0.2 m (DTEST = 0.5). The results of the test points are plotted versus the frequency (all results in section 10.4. contain the computational data from both test segments). At some arbitrary frequencies discrepancies occur between the two results (see FIGURE 80,82). If the difference exceeds 1 dB, the results are checked according to 10.3.4.2.. Up to now the inaccuracy was always manifested by a large EPH difference, in all of the 11 problematic cases of about 100 complete near-field computations. The computations are repeated at a new frequency, differing from the "disturbed" frequency by about 2 %. There is no strict rule to prevent "disturbed" frequencies, but  $d_{at}$ 's below 0.2m, frequencies above 300 MHz and complicated body shapes are risk factors.

## 10.3.5. FIELD HOMOGENEITY AROUND A NEAR FIELD POINT

# 10.3.5.1. SIGNIFICANCE OF THE FIELD HOMOGENEITY AND COMPUTATIONAL DATA

The body-mounted antenna  $A_1$  is not infinitesimally small. Thus, the induced voltage at the anntenna terminals (receiving case) depends generally not only on the field at the (computed) antenna center, but also on the field along the antenna axis. The study in section 5.2.2. concluded in the statements:

If the amount of the E-field can be described by a polynome of second degree and if the phase of the E-field changes monotonously along the antenna axis, the logarithmic difference  $\Delta U$  (i.e., the ratio induced voltage from center field / induced voltage from actual field, see 5.2.2.) is less than 1 dB, if the following conditions are fulfilled:

- $\delta E$  = variation of the amount of the E-field along < 10 dB (37) the antenna axis in direction of the regarded p<sub>1</sub>
- $\delta \Phi$  = phase variation of the E-field along the an- < 71 ° (38) tenna axis in direction of the regarded pl

In section 10.2.5. the near-field data along the p<sub>1</sub>-axis vertical, radial and horizontal were computed at the center and at the ends of a dipole antenna of the length 2h = 0.1 m, with test segment lengths of 0.08 m. The test segments at +h and -h are separated by a gap of 0.02 m, so that oscillations (if existing) can be detected without computational artifacts which may occur at smaller test segments (10.3.1.). The quantities  $\Delta U$ ,  $\delta E$ and  $\delta \Phi$  were computed for d<sub>at</sub> = 0.1 m, h<sub>B</sub> = 1.0 m,  $\theta_i$  = 80.8<sup>o</sup> and for the body model FZYL at several frequencies (see TABLE 68, 69 and the results in Appendix 16.2.4.).

The computational results confirm the validity of equation (37) and (38). At  $p_1$  = vertical and radial the  $\delta E$  and  $\delta \Phi$  are below the critical level and thus  $\Delta U$  remains smaller than 1 dB. With  $p_1$ ,  $p_2$  = horizontal the antenna A<sub>1</sub> is oriented perpendicular to the E<sup>tot</sup> wavefront at about  $\phi$  = 80 to 100°, resulting in large phase changes and thus large  $\Delta U$ . The significant data of the homogeneity investigation are summarized in TABLE 70 for the frequencies 65, 75, 100, 125, 150, 300 and 425 MHz.

AZ	IMU	THAL	RAD	IATI	ONPA	TTER	NFR	EQU	ENCY	150 M H Z		
HOM	HOMOGENEITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA											
TEST	IBODY:	ROT.SYM. AXIAL LE DIAMETER	CYLIND NGTH = =	ER 1.80 M .25 M	DINT IO M DO M	INCID POLAR THETA	ENT WA • = VE = 80	VE RTICAL .8 DEG				
	VERTI	CAL POLA	RIZED	ANTENNA		RADIAL	POLARI	ZED AN	TENNA			
PHI DEG	GAIN CENTE DB	PHASE R CENTER DEG	MEAN ERROR DB	MAXIMUM GAINVAR DB	MAXIMUM Phasevar Deg	GAIN CENTER DB	PHASE CENTER DEG	MEAN ERROR DB	MAXIMUM GAINVAR DB	MAXIMUM Phasevar Deg		
0 5 10 15 20 25 30 35 40 45 0 55 00 65 70 75 80 85 90 95 10 11 12 20 25 30 35 40 45 0 55 00 65 70 75 80 85 90 95 10 11 12 10 12 10 10 11 11 10 10 10 10 10 10 10 10 10	-3.6677901368148261628852964229 -3.77901368148261628852964229 -111233962122 -1112339622691222 -1112339622691222	151.9 152.0 152.4 153.2 154.2 155.4 157.0 158.8 160.8 163.1 155.7 174.8 160.5 174.8 -165.3 -169.8 -169.8 -143.9 -143.9 -137.4 -122.9 -137.4 -123.9 -143.9 -143.9 -123.9 -143.9 -123.9 -143.9 -123.9 -143.9 -123.9 -123.4 -123.9 -123.4 -123.9 -123.4 -123.9 -123.4 -123.9 -123.4 -123.9 -123.5 -698.1 -89.9 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -698.1 -75.5 -698.8	.00 .00 .00 .00 .00 .00 .00 .00 .00 .00	44444444443333322110123456788998	2.88 2.2.7 2.65 2.2.1 1.97 1.2.2 2.197 1.2.97 1.0 3.69 2.4 7.899 1.17 1.997 1.17 1.997 1.17 1.997 1.17 1.997 1.17 1.997 1.17 1.997 1.17 1.997 1.17 1.97 1.2.2 2.2.5 2.2.197 1.2.2 2.2.197 1.2.2 2.2.197 1.2.2 2.2.197 1.2.2 2.2.2 2.2.197 1.2.2 2.2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2 2.2.2.2 2.2.2.2 2.2.2.2 2.2.2.2 2.2.2.2.2 2.	1.444332211098765431012334455555555 	$\begin{array}{c} 40.8\\ 40.9\\ 41.3\\ 41.3\\ 42.7\\ 43.3\\ 43.8\\ 44.4\\$	2652525252524 225252524 225252524 22222222222222222222222222222222222	$\begin{array}{c} 4.1\\ 4.1\\ 4.1\\ 4.1\\ 4.1\\ 4.0\\ 4.0\\ 4.0\\ 4.0\\ 4.0\\ 4.0\\ 4.0\\ 4.0$	.2 .0 .247.92581468000985273832726924		
170 175 180	-16.1 -16.1 -16.1	-62.2 -60.3 -59.7	01 01 01	.8 .8 .8	3.1 3.2 3.3	5 5 5	36.9 36.8 36.7	23 23	3.9 3.9 3.9	2.7		

TABLE 68 Field homogeneity at 150 MHz at  $d_{at} = 0.1 \text{ m}$  and  $p_2 = \text{vertical}$ . Left:  $p_1 = \text{vertical}$ , right:  $p_1 = \text{radial}$ ; dB-values related to 0 dB = FSL. Gain center: Gain<sub>B</sub> [dB], Phase center: phase  $\Phi[^0]$ , Mean error:  $\Delta U$  [dB], Maximum gainvar:  $\delta E$  [dB], Maximum phasevar:  $\delta \Phi[^0]$ .

AZIMUTHAL RADIATION PATTERN FREQUENCY 150 MHZ										
HOMOGENEITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA										
TESTBODY: ROT. SYM. CYLINDERFIELD POINTINCIDENT WAVEAXIAL LENGTH = 1.80 MDAT = .10 MPOLAR. = HORIZONTALDIAMETER= .25 MHB =1.00 MTHETA = 80.8 DEG										
	HORIZONTAL F	OLARIZED A	NTENNA		RADIAL	POLARIZ	ZED ANT	TENNA		
PHI DEG	GAIN PHASE CENTER CENTE DB DEG	MEAN MA R ERROR GA DB DB	XIMUM MA INVAR PH 3 DE	AXIMUM IASEVAR CG	GAIN CENTER DB	PHASE CENTER DEG	MEAN ERROR DB	MAXIMUM GAINVAR DB	MAXIMUM Phasevar Deg	
0 55 100 15 205 300 355 600 555 600 555 600 555 600 555 600 555 800 905 1005 1105 11205 11205 11495 11495 11495 11495 11495 11495 11495 11495 11495 11495 11495 11495 11495 1150 1155 115 115 115 115 115 115 115	$\begin{array}{c} -4.7 & 16.6 \\ -4.7 & 17.1 \\ -4.8 & 17.9 \\ -5.0 & 19. \\ -5.3 & 20.9 \\ -5.7 & 23.5 \\ -6.6 & 29.9 \\ -7.2 & 33.6 \\ -8.8 & 43.2 \\ -9.8 & 49.0 \\ -11.0 & 55.4 \\ -12.3 & 63.6 \\ -13.9 & 73.4 \\ -13.9 & 73.4 \\ -19.3 & -162.4 \\ -13.9 & 73.4 \\ -19.3 & -162.4 \\ -10.8 & -140.4 \\ -19.3 & -162.4 \\ -10.8 & -107.4 \\ -2.8 & -67.4 \\ -2.8 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.8 & -67.4 \\ -2.8 & -67.4 \\ -2.7 & -67.4 \\ -2.7 & -67.4 \\ -2.8 & -67.4 \\ -2.7 & -67.4 \\ -2.8 & -67.4 \\ -2.8 & -67.4 \\ -2.8 & -67.4 \\ -2.7 & -67.4 \\ -2.8 & -67.4 \\ -$	$\begin{array}{c} & 10 \\ & 10 \\ & 10 \\ & 10 \\ & 09 \\ & 09 \\ & 08 \\ & 08 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 06 \\ & 07 \\ & 08 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 1.96 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & -28 \\ & 00 \\ & 0$	$\begin{array}{c} 0.0\\\\\\\\\\\\\\$	0.66 7.95 7.55 7.95 7.55 7.95 7.55	***** -19.4 -13.4 -9.9 -7.5 -4.8 -1.8 4.8 	****** 36.2 36.6 37.2 38.1 39.7 42.3 40.7 42.3 40.7 42.3 40.7 42.3 40.7 42.3 40.7 55.2 58.9 63.7 55.2 66.3 72.8 87.5 58.6 972.4 82.0 87.5 82.6 87.5 82.6 87.5 99.4 93.3 99.4 99.9 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.4 99.5 99.5	****** 18 18 18 18 18 18 18 18 18 17 18 *******	***** ** 2.7 2.6 2.6 2.6 2.5 2.5 2.2 2.2 2.2 2.2 2.2 2.2 2.2 2.1 2.1 2.1	***** 17.1 16.5 16.5 16.5 11.4 13.8 10.7 5.2 8.9 5.2 8.9 5.2 1.0 1.2 7 8.9 5.2 8.9 5.2 8.9 5.2 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.9 5.2 8.0 1.2 7 8.0 5.2 8.0 1.2 7 8.0 5.2 8.0 1.2 7 8.0 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 5.2 8.0 1.2 7 8.0 9 9 1.1 8.5 7 8.0 9 9 1.1 8.5 7 8.0 9 9 1.1 8.5 7 8.0 9 9 1.1 8.5 7 8.0 9 9 1.1 8.5 7 8.0 9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 8.0 9 1.1 8.5 7 8.9 9 1.1 8.5 7 8.9 9 8.0 9 8.5 1.3 9 8.5 8 9 8.5 8 9 8.5 8 9 8.5 8 8 9 8 8 8 8 8 8 8 8 9 8 8 8 8 8 8 9 8 8 8 8 8 8 8 8 9 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 9 8 8 8 8 8 8 8 8 8 9 8 8 8 8 8 8 8 8 8 8 8 8 8 9 8	

TABLE 69 Field homogeneity at 150 MHz at  $d_{at} = 0.1 \text{ m}$  and  $p_2 = \text{horizontal}$ . Left:  $p_1 = \text{horizontal}$ , right:  $p_1 = \text{radial}$ ; dB-values related to 0 dB = FSL. Gain center : Gain<sub>B</sub> [dB]. Phase center: phase  $\phi$  [°], Mean error:  $\Delta U$  [dB], Maximum gainvar:  $\delta E$  [dB], Maximum phasevar:  $\delta \phi$  [°].

FRE-	Maxi	mum ra	atings	; p2 =	VER	TICAL	Maximum ratings p <sub>2</sub> = HORIZONTAL						
QUEN-	p1 =	VER	TICAL	P1	= RAI	DIAL	p1 = 1	HORIZ	ONTAL	p] =	RAD	<b>FAL</b>	
CY [MHz]	∆U [dB]	δE [dB]	δΦ] [0]	∆U [dB]	δE [dB]	δΦ [0]	∆U [dB]	δE [dB]	δΦ [0]	∆U [dB]	δE [dB]	δΦ [0]	
65	0.1	1.0	10	0.3	4.5	4	8.7	29.9	200	0.2	2.1	7	
75	0.2	3.8	18	0.3	4.3	2	7.6	25.0	194	0.2	2.1	8	
100	0.0	1.3	3	0.3	4.5	2	5.4	18.1	180	0.2	2.3	11	
125	0.0	1.9	5	0.4	4.5	3	3.7	13.5	165	0.2	2.5	14	
150	0.0	0.8	3	0.3	4.1	3	2.4	11.3	149	0.2	2.7	17	
300	0.0	1.2	8	0.5	5.1	31	>10	>20	222	0.3	2.2	45	
425	0.1	1.7	19	1.0	5.9	65	3.1	13.3	100	0.3	2.3	50	

The maximum ratings of the computed field homogeneity parameters are listed in TABLE 70 :

TABLE 70 Maximum ratings from the field homogeneity computations. The data concern the model FZYL with the antenna position  $d_{at} = 0.1$  m and  $h_B = 1.0$  m, related to a 2h = 0.1 m dipole antenna A1.

The field homogeneity is satisfactory except for  $p_1 = p_2 = horizontal$  which is not suited for omnidirectional transmission. From the data in TABLE 70 one can conclude that all further computations have only to be performed for the <u>center test point</u> which is representative for the field around the test point, if d<sub>at</sub> is larger than 0.1 m and if the frequency is below 500 MHz.

# 10.3.5.2. COMPUTATIONAL DATA FOR ANTENNA DESIGN

With the knowledge of the amplitude- and especially the phase conditions along a certain antenna axis the design of radiation systems become possible which perform better than an usual antenna in the proximity to a body. If we look at TABLE 68 at p<sub>1</sub> = radial we notice a field amplification effect produced by the body. There is strong radial field around the body, differing only from +1.4 to -0.5 dB at  $0 < \phi < 180^{\circ}$ . The  $\delta E$  and the  $\delta \Phi$  are within reasonable limits (4 dB and  $3^{\circ}$ ), so that an excellent antenna with omnidirectional radiation characteristics could be designed. In the following study the near-field data are discussed and compared with experimental data; a special radial antenna will be shown in 13.3.3.

# Leer - Vide - Empty

# 10.4. RESULTS FROM THREE-DIMENSIONAL COMPUTATIONS ON ANTENNA-BODY MODELS 10.4.1. OVERVIEW OF INVESTIGATED PARAMETERS AND EXPLANATIONS

In the following sections the influence of certain parameters on the near-field will be demonstrated by computer plots. Unless otherwise specified, the regarded body model is the finite conducting cylinder (FZYL), the irradiation angle  $\phi$  amounts to 80.8°, p<sub>2</sub> is vertical ( $E_{\theta}^{inc}$ ) and the relative antenna height h<sub>B</sub> is 1.0 m. The field data are related to the free-space level FSL = 0 dB and specify the E-field in dB in direction of the investigated polarization axis p<sub>1</sub> = vertical, radial or horizontal.

The computations were performed according to section 10.2. and 10.3., and the results were checked according to 10.3.4..The accuracy of the following data is better than  $\pm 1$  %, corresponding to  $\pm 1$  dB at signal levels above - 21 dB, at frequencies from 10 to 500 MHz. All data have been computed with two different test segment lengths (0.08 m and 0.2 m, actual scale) and generally both results are plotted, as can be noticed by the thicker or double lines in the plots. The accuracy of the frequencies above 500 MHz is in the region of 5 % F.S. and was not further investigated, because the previous computational model presented in section 7. covers the frequency range from 200 to 1000 MHz, and because the interesting effects occur below 500 MHz. In addition, more accurate computations above 500 MHz are prohibitive due to storage and computational time limitations (see limitations in section 10.3.2. and 10.3.3.).

The following sections treat the specific effects:

- 10.4.2. Effect of the frequency on vertical and radial field
- 10.4.3. Effect of the antenna-body distance
- 10.4.4. Effect of the azimuthal angle
- 10.4.5. Effect of the irradiation angle
- 10.4.6. Effect of the relative antenna height
- 10.4.7. Effect of the frequency on the azimuthal radiation patterns
- 10.4.8. Effect of the frequency on the directive radiation patterns
- 10.4.9. Effect of differentbody shapes on the fields in the shadow zone
- 10.4.10. Effect of different body shapes on azimuthal radiation patterns

## 10.4.2. EFFECT OF THE FREQUENCY ON VERTICAL AND RADIAL FIELD

FIGURE 71 provides a first impression. The Gain<sub>B</sub> data are shown for 4  $d_{at}$ 's of 0.1, 0.2, 0.3 and 0.4 m in the shadow zone  $\phi = 180^{\circ}$  at  $p_1, p_2 =$  vertical.









Unless otherwise specified one always considers the E-field components  $E_v$ ,  $E_r$  and  $E_h$  in the shadow region  $\phi = 180^\circ$ . In FIGURE 71 one distinguishes 5 frequency regions similar to those noticed in the experiments in section 9.

- <  $\lambda/2$  below resonance at frequencies below 50 MHz
- $\lambda/2$  first resonance at about 65 MHz
- 3  $\lambda/4$  anti-resonance at about 80 to 110 MHz
- λ second resonance at about 140 MHz
- $> \lambda$  off-resonance at frequencies above 200 MHz

If we compare the FZYL results with the previous IZYL results in FIGURE 72 one observes an oscillation of the FZYL data around the IZYL data with an amplitude of maximum 2.5 dB at frequencies above 200 MHz. This means that both computational methods agree at higher frequencies. From the practical point of view this agreement is disappointing for two reasons: first, there is no theoretical chance to operate with vertical polarized antennas above 75 MHz due to the high transmission losses, second, the experimental data do not agree with the computational data below 150 MHz.

Fortunately, an astonishing radial field effect occurs in the resonance region which provides new hope for both application and experiment :



FIGURE 73 Radial field component  $E_r$  versus frequency f, parameter  $d_{at}$ .

As can be seen in FIGURE 73 a <u>radial field component</u>  $E_r$  is developed at frequencies from about 40 to 300 MHz, caused by a <u>vertical polarized</u> incident wave and the body. In the proximity to the body this  $E_r$  is very strong (-4 to 0 dB at  $d_{at}$  = 0.1 m) and decreases with increasing  $d_{at}$ . Below 30 MHz the  $E_r$  is small (< -12 dB at  $d_{at}$  > 0.3m) and above 500 MHz the  $E_r$  is extremely small (< -16 dB at  $d_{at}$  > 0.3m). The significance of the strong  $E_r$  is demonstrated in the next section.

### 10.4.3. EFFECT OF THE ANTENNA-BODY DISTANCE

Let us consider first the vertical  $(E_v)$  and radial  $(E_r)$  field component within the resonance region at 150 MHz and at  $\phi = 0^{\circ}$  (irradiated zone) and at  $\phi = 180^{\circ}$  (shadow zone). FIGURE 74 shows the dependence of the field components from the antenna-body distance  $d_{at}$ :



FIGURE 74 Vertical  $(E_v)$  and radial  $(E_r)$  field component versus  $d_{at}$ . Parameter:  $E_v$  and  $E_r$  at  $A_1$  at  $\phi = 180^\circ$  (left) and  $0^\circ$  (right) Constant : frequency = 150 MHz,  $p_2$  = vertical,  $\theta_i = 80.8^\circ$ ,  $h_B = 1.0$  m At  $\phi = 0^\circ$  the  $E_r$  becomes larger than  $E_v$  at  $d_{at} < 0.15$  m and at  $d_{at} = 0.05$  m  $E_r$  is about 12 dB larger than  $E_v$ . The situation is even more extreme at  $\phi = 180^\circ$ ; the  $E_r$  becomes larger than  $E_v$  at  $d_{at} < 0.42$  m and at  $d_{at} = 0.05$  m  $E_r$  is about 22 dB larger than  $E_v$ ! This effect explains the discrepancy of the experimental results at small  $d_{at}$ , because the transverse sensitivity of a probe antenna  $A_1$  is rarely below -15 dB. It is not possible to measure an  $E_v$  of only -20 dB, if there is also an  $E_r$  of about 0 dB.



FIGURE 75 Vertical ( $E_v$ ), radial ( $E_r$ ) and horizontal ( $E_h$ ) field components versus azimuthal angle  $\phi$ . Parameter:  $E_v$ ,  $E_r$ ,  $E_h$  and  $d_{at} = 0.1, 0.2, 0.3, 0.4 m$ . Constant: frequency = 150 MHz, p<sub>2</sub> = vertical,  $\theta_i = 80.8^\circ$ ,  $h_B = 1.0 m$ . The  $E_r$  component is almost constant throughout the full  $\phi = 0.180^\circ$  range.

# 10.4.4. FFFECT OF THE AZIMUTHAL ANGLE



FIGURE 76 Vertical ( $E_V$ ), radial ( $E_r$ ) and horizontal ( $E_h$ ) field components at constant  $d_{at}$  = 0.1 m versus  $\phi$  for different irradiation angles  $\theta_i$  = 90,80,70 and 60°. Constant: f = 150 MHz, p<sub>2</sub> = vertical, h<sub>B</sub> = 1.0 m. E<sub>r</sub> increases with decreasing  $\theta_i$ , E<sub>V</sub> remains about constant, E<sub>h</sub> disappears at  $\theta_i$  = 90°.

HB=1. OM HB=1.2M 1280 1440 DAT=0**.** 1M, EINC V 80.8D DAT=0.1M 8 80.80 8 8 EINC V MHZ PHI-AXIS A2 MHZ PHI-AXIS A2 8 யீ ш ш 150 20 a àz 00 CRIN DB R1 PØL VERTHRØND LZYL CAIN DB A1 POL VERINIA UNITE 1727 **Ð**. HB=1.1M HB=0. 9M DAT=0. 1M. 9 8 80, 80 ( 8 EINC V 8 MHZ PHI-AXIS A2 шī ш шī 3 20 œ. 01 ĊC. CAIN DS A1 POL VEAT/HOR/RAD JYZĘ CAIN DB AI POL VERT/HOR/TRAD ر. الاكرار

FIGURE 77 Vertical ( $E_v$ ), radial ( $E_r$ ) and horizontal ( $E_h$ ) field components at constant  $d_{at} = 0.1 \text{ m}$  versus  $\phi$  for different relative antenna heights  $h_B=0.9$ , 1.0, 1.1 and 1.2 m. Constant: f = 150 MHz, p<sub>2</sub> = vertical,  $\theta_i = 80.8^{\circ}$ . Even at  $h_B = 0.9$  (body center) all field components are only little influenced by h<sub>B</sub>.

10.4.6. EFFECT OF THE RELATIVE ANTENNA HEIGHT



10.4.7. EFFECT OF THE FREQUENCY ON THE AZIMUTHAL RADIATION PATTERN

FIGURE 77a Field components  $E_V$ ,  $E_r$  and  $E_h$  versus  $\phi$  with the parameter f 11 to 100 MHz. Constant:  $d_{at}$  = 0.1m, p<sub>2</sub> = vertical,  $\theta_i$  = 80.8<sup>o</sup>,  $h_B$  = 1.0m.

- 171 -



FIGURE 77b Field components  $E_v, E_r$  and  $E_h$  versus  $\phi$  with the parameter f 125 to 800 MHz. Constant:  $d_{at} = 0.1 \text{ m}, p_2 = \text{vertical}, \theta_1 = 80.8^\circ, h_B = 1.0 \text{ m}.$ 



FIGURE 77c Field components  $E_v, E_r$  and  $E_h$  versus  $\phi$  with the parameter f 60 to 100 MHz. Constant:  $d_{at}$ = 0.1m,  $p_2$ = vertical,  $\theta_1$ = 80.8°,  $h_B$ = 1.0m.

FIGURES 77a and 77b show the azimuthal radiation pattern with constant  $d_{at}$  of 0.1m in the full frequency range 11 to 800 MHz, FIGURE 77c in the resonance frequency range 60 to 100 MHz.

# Vertical E-field component Ev

- Below resonance: at frequencies below 50 MHz  $E_V$  depends not on  $\phi$ . Although the far-field scattering (RCS, see section 5.3.2.) is small,the attenuation of the near-field is considerable (-11 dB). However,omnidirectional transmission is possible with a reasonable transmission loss.
- Above first resonance: at frequencies above 60 MHz E<sub>V</sub> depends much on  $\phi$ . Generally E<sub>V</sub> decreases with increasing  $\phi$  and the maximum difference amounts to 12 dB at 125 MHz up to 23 dB at 800 MHz from  $0 < \phi < 180^{\circ}$ . Thus, omnidirectional transmission is difficult due to the large E<sub>V</sub> variation and due to the small E<sub>V</sub> (minimum: 19 dB/800 MHz). Compared with the IZYL data the FZYL data vary not more than 2.5 dB above 125 MHz.

# Radial E-field component Er

- Below resonance: at frequencies below 50 MHz  $E_r$  depends much on  $\phi$ . Generally  $E_r$  is very small at  $\phi = 0^{\circ}$  (-30 dB at 11 MHz, but increases with f up to -10 dB at 50 MHz) and is relatively high at  $\phi = 180^{\circ}$  (-7 dB at 11 MHz and -3 dB at 50 MHz).  $E_r$  increases with  $\phi$ , and the maximum difference within  $0^{\circ} < \phi < 180^{\circ}$  amounts to 23 dB at 11 MHz and 7 dB at 50 MHz.  $E_r$  becomes permanently larger than  $E_v$  above 50 MHz and is therefore suited for omnidirectional transmission above 50 MHz.
- Above first resonance: at frequencies from (60)-200 MHz  $E_r$  depends not much on  $\phi$  and is always larger than -4 dB. These properties of  $E_r$  are ideal for omnidirectional transmission. Above 250 MHz  $E_r$  depends on  $\phi$ , first decreases  $E_r$  at  $\phi = 0^{\circ}$ , becomes small at  $0 < \phi < 180^{\circ}$  and above 500 MHz (Appendix 16.2.5.) the minimum  $E_r$  becomes very small (-30 dB).

# Horizontal E-field component Eh

- At all frequencies the  $E_h$  depends much on  $\phi$  and is smaller then -14 dB. Resonance region 60 to 100 MHz

- Vertical E<sub>v</sub> component: with increasing frequency the minimum shifts from  $\phi=0^{0}$  to  $\phi=180^{0}$ . The minimum occurs at about 73 MHz (at about  $\phi=90^{0}$ ) and is smaller than -30 dB. The maximum E<sub>v</sub> amounts to -7 dB at 100 MHz,  $\phi=0^{0}$ .
- Radial E<sub>r</sub> component: E<sub>r</sub> depends little on  $\phi$  and is always >4.5 dB.



10.4.8. EFFECT OF THE FREQUENCY ON THE DIRECTIVE RADIATION PATTERN

FIGURE 78a Field components  $E_V$  and  $E_r$  versus  $d_{at}$  at  $\phi = 0$  and  $180^{\circ}$ , with the parameter f 11 to 100 MHz. Constant:  $p_2 = vertica1, \theta_1 = 80.8^{\circ}$ ,  $h_B = 1.0 \text{ m}$ .

- 175 -



FIGURE 78b Field components  $E_V$  and  $E_r$  versus  $d_{at}$  at  $\phi=0$  and  $180^{0}$ , with the parameter f 125 to 800 MHz. Constant:  $p_2$  = vertical,  $\theta_i$  = 80.8°,  $h_B$  = 1.0 m.



FIGURE 78c Field components  $E_V$  and  $E_r$  versus  $d_{at}$  at  $\phi = 0$  and  $180^{\circ}$ , with the parameter f 60 to 100 MHz. Constant:  $p_2 = vertical$ ,  $\theta_1 = 80.8^{\circ}$ ,  $h_B = 1.0 \text{ m}$ .

.

- 177 -

FIGURES 78a and 78b show the directive radiation pattern at  $\phi = 0^{\circ}$  and 180° at variable d<sub>at</sub> in the full frequency range from 11 to 800 MHz,FIGURE 78c in the resonance frequency range 60 to 100 MHz.

### Vertical E-field component Ev

- Below resonance : Below 30 MHz the  $E_v$ 's at  $\phi = 0$  and 180° are symmetrical, are largest with large  $d_{at}$  (-3 dB at  $d_{at} = 0.4$  m) and decrease with decreasing  $d_{at}$  (-11 dB at  $d_{at} = 0.1$  m). Above 50 MHz the two  $E_v$ 's become asymmetric but still decrease with decreasing  $d_{at}$ . Thus, the body shows no directive characteristics, and the body has little influence on  $E_v$ as long as  $d_{at}$  is larger than 0.4 m.
- Above first resonance: At frequencies above 125 MHz E<sub>V</sub> is always smaller at  $\phi = 180^{\circ}$  than at  $\phi = 0^{\circ}$ . At  $\phi = 180^{\circ}$  E<sub>V</sub> decreases constantly with decreasing d<sub>at</sub>, with an amplitude of about -11 dB at d<sub>at</sub> = 0.4 m and of about -18 dB at d<sub>at</sub> = 0.1 m. At  $\phi = 0^{\circ}$  E<sub>V</sub> oscillates around 0 dB according to the results in section 7.3.1.: maxima occur at d<sub>at</sub>  $\sim n \cdot \lambda/4$ , n = 1,3,5,..and minima at d<sub>at</sub>  $\sim n \cdot \lambda/2$ , n = 0,1,2,... The body acts like an efficient reflector if d<sub>at</sub> amounts to  $n \cdot \lambda/4$  and is a good absorber if d<sub>at</sub> < 0.2 m. As an example the forward/backward ratio is 17 dB at d<sub>at</sub> = 0.2 m/350MHz.

# Radial E-field component Er

- Below resonance: at 11 MHz  $E_r$  is very small at  $\phi = 0^\circ$  but increases with increasing frequency. Above 50 MHz both  $E_r$ 's increase with decreasing  $d_{at}$  and become larger than  $E_v$  at about  $d_{at} < 0.15 \text{ m}$ . Above 60 MHz both  $E_r$ 's are about symmetrical with about -2 dB at  $d_{at} = 0.1$  and -9 dB at  $d_{at} = 0.4 \text{ m}$ . At very small  $d_{at}$  the body acts like a director (11 MHz) and as a director/reflector (> 60 MHz).
- Above first resonance: from (60) to 200 MHz both  $E_r$ 's at  $\phi = 0$  and  $180^{\circ}$  are about symmetrical with high values at small  $d_{at}$ . Generally the  $E_r$ 's are larger than the  $E_v$ 's at  $d_{at} < 0.1 \text{ m}$  ( < 0.3 at  $\phi = 180^{\circ}$ ), and the body acts like a director/reflector for small  $d_{at}$ 's. Above 250 MHz  $E_r$  decreases first at  $\phi = 0^{\circ}$  and above 350 MHz also at  $\phi = 180^{\circ}$ . Above 450 MHz and  $d_{at} > 0.3 \text{ m}$  the  $E_r$ 's become very small, smaller than the  $E_v$ 's.

# Resonance region 60 to 200 MHz

- Vertical E<sub>V</sub> components: if  $d_{at} < \lambda/4$ , the E<sub>V</sub>'s decrease with decreasing  $d_{at}$  and are smallest at about 85 MHz at  $\phi = 180^{\circ}$  (-22 dB at  $d_{at} = 0.1$  m).
- Radial  $E_r$  components: Both  $E_r$ 's are larger (> 5 dB) at  $d_{at} < 0.15 m$ .

10.4.9. EFFECT OF DIFFERENT BODY SHAPES ON THE FIELDS IN THE SHADOW ZONE The former computations with the simple body model FZYL revealed the systematic relations between antenna location and field quantities. The most important results are summarized in FIGURE 80 for  $\phi = 180^{\circ}$ :

- $E_{\rm V}/f$  diagram: sharp anti-resonance at about 75-105 MHz, recovery of  $E_{\rm V}$ at frequencies above 125 MHz and oscillation around - 16 dB ( $d_{at}$ =0.1m)
- Er/f diagram: two peaks, one around 70 MHz (+0.5 dB) and a second of similar amplitude around 150 MHz (+0 dB) at  $d_{at} = 0.1 m$ .
- $E_y$ - $E_r$ /f diagram at dat=0.1m:  $E_r$  is up to 16 dB larger than  $E_y$  at frequencies below 550 MHz
- $E_v$ - $E_r$ /f diagram at  $d_{at}$  = 0.4 m :  $E_r$  is usually smaller than  $E_v$

The corresponding data has been computed for the man-models in FIGURE 79:



FIGURE 79 Computational body models - FZYL : finite rotational symmetric

- cylinder with round end caps  $L_B = 1.8 m$ ,  $D_B = 0.25 m$ .
- MANMOD1:rotational symmetric human body, front view = contour curve  $L_B = 1.68 m$ ,  $D_B = 0.296 m$  at  $d_{at} = 1.0 m$ .
- MANMOD 2 : rotational symmetric human body, side view = contour curve  $L_B = 1.68 \,\text{m}$ ,  $D_B = 0.196 \,\text{m}$  at  $d_{at} = 1.0 \,\text{m}$ .

(dimensions listed in Appendix 16.2.4.)

Comparing FIGURE 81 (MANMOD1) with FIGURE 80 (FZYL) we find:

- E<sub>v</sub>/f diagram: there is still an anti-resonance at about 80-125 MHz but less sharp and without recovery.  $E_V$  drops with increasing frequency.
- Er/f diagram: only the first peak is well developed, amounts to +3 dB at dat=0.1m and occurs at 75 MHz.
- $E_r-E_v$  diagram at  $d_{at} = 0.1 \text{ m} : E_r > E_v$  (max.16 dB) above 600 MHz Comparing FIGURE 82 (MANMOD 2) with FIGURE 80 (FZYL) we find:
- $E_v/f$  diagram: very similar to FZYL (but shifted in frequency)
- Er/f diagram: very similar to FZYL but larger amplitude variations. First peak around 80 MHz(+4 dB) and second peak around 160 MHz(+2.5 dB).



FIGURE 80 Summarized computational results from body model FZYL  $E_V$  versus f, parameter  $d_{at}$   $E_r$  versus f, parameter  $d_{at}$   $E_V$  and  $E_r$  at  $d_{at} = 0.1$  m  $E_V$  and  $E_r$  at  $d_{at} = 0.4$  m Constant:  $\phi = 180^\circ$ ,  $p_2 =$  vertical,  $\theta_i = 80.8^\circ$ ,  $h_B = 1.0$  m.



FIGURE 81 Summarized computational results from body model MANMOD1  $E_V$  versus f, parameter  $d_{at}$   $E_r$  versus f, parameter  $d_{at}$   $E_V$  and  $E_r$  at  $d_{at} = 0.1$  m  $E_V$  and  $E_r$  at  $d_{at} = 0.4$  m Constant:  $\phi = 180^{\circ}$ ,  $p_2 =$  vertical,  $\theta_i = 80.8^{\circ}$ ,  $h_B = 1.0$  m.


FIGURE 82 Summarized computational results from body model MANMOD 2  $E_V$  versus f, parameter  $d_{at}$   $E_r$  versus f, parameter  $d_{at}$   $E_V$  and  $E_r$  at  $d_{at} = 0.1$  m  $E_V$  and  $E_r$  at  $d_{at} = 0.4$  m Constant:  $\phi = 180^\circ$ ,  $p_2 =$  vertical,  $\theta_i = 80.8^\circ$ ,  $h_B = 1.0$  m.

### 10.4.10. EFFECT OF DIFFERENT BODY SHAPES ON AZIMUTHAL RADIATION PATTERNS

If we look at the azimuthal radiation patterns in 10.4.7., we notice the following frequency dependent changes of the field components:

- Ev changes drastically between 60 and 100 MHz
- Er changes rapidly at about 50 MHz and above 250 MHz
- Eh shows the same pattern up to 350 MHz

Changes of the field components occur only if the body dimensions are in a special relation to the wavelength. If the vertical circumference ( $\sim 2 L_B$ , see also FIGURE 18) is about  $\lambda$ , the E<sub>V</sub> and E<sub>r</sub> are affected. If the horizontal circumference DB+¶ is about  $\lambda$ , E<sub>h</sub> and E<sub>r</sub> are affected.

Considering these facts it is easy to understand the changes in the azimuthal radiation patterns due to body shape alterations. Small changes of  $L_B$  and  $D_B$  provokes similar effects like small changes of the frequency.

Significant azimuthal radiation patterns are shown for the three bodies FZYL, MANMOD1 and MANMOD2 in FIGURES 83 a,b,c at 65 MHz and in FIGURES 84 a,b,c at 150 MHz. Additional samples for 11, 50, 75, 85, 200 and 800 MHz are presented in FIGURES 100 to 105 in Appendix 16.2.5..



FIGURE 83a Azimuthal radiation patterns FZYL at 65 MHz.  $E_{\rm V},~E_{\rm r}$  and  $E_{\rm h}\,com-$  ponents at dat=0.1, 0.2, 0.3 and 0.4 m . See also FIGURE 75.



FIGURE 83b Azimuthal radiation patterns MANMOD1 at 65 MHz.



FIGURE 83c Azimuthal radiation patterns MANMOD 2 at 65 MHz.

Comparing FIGURES 83a, b, c. at 65 MHz we observe an L<sub>B</sub>-effect:

- $E_v$  changes drastically from FZYL to MANMOD1 ( $L_B$  = 1.8 and 1.68 m), but there are only small differences between MANMOD1 and 2 ( $L_B$  = 1.68 m). As can be seen in Appendix 16.2.5., FIGURE 102, the MANMOD's resonate at about 75 MHz in contrast to FZYL at 65 MHz due to the 6.7% shorter  $L_B$ .
- Er varies only within 2 dB.
- E<sub>h</sub> varies only within 2 dB.

Comparing FIGURES 84a,b,c at 150 MHz we observe an L<sub>B</sub> and D<sub>B</sub>-effect:

- E<sub>v</sub> varies within 5 dB at  $\phi = 180^{\circ}$ . The second ( $\lambda$ ) resonance is best developed at MANMOD 2 which has the largest L<sub>B</sub>/D<sub>B</sub>-ratio. The weakest  $\lambda$  resonance occurs at MANMOD 1 with the smallest L<sub>B</sub>/D<sub>B</sub>-ratio.
- $E_r$  varies within 3 dB, and the values of FZYL are between those of MAN-MOD1 and 2, corresponding to the  $D_B$  ratio of the three bodies.
- $E_h$  varies within 3 dB. The theoretical horizontal resonant frequencies are 301 MHz (MANMOD1), 380 MHz (FZYL) and 487 MHz (MANMOD2). The asymmetry of  $E_h$  of MANMOD1 is caused by a subresonance, because 150 MHz is close to 301 MHz, the  $E_h$  of FZYL is already better and the  $E_h$  of MANMOD2 is very symmetrical because 150 MHz is well below 487 MHz.



FIGURE 84a Azimuthal radiation patterns FZYL at 150 MHz.  $E_v, E_r$  and  $E_h$  components at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4 m. See also FIGURE 75.



FIGURE 84b Azimuthal radiation patterns MANMOD1 at 150 MHz.



FIGURE 84c Azimuthal radiation patterns MANMOD 2 at 150 MHz.

# Leer - Vide - Empty

# 11. EXTENDED MEASURING METHOD FOR FIELD COMPONENTS SEPARATION

#### 11.1. PURPOSE OF THE EXTENDED EXPERIMENTS

The experimental data in section 9. were obtained with the measuring method described in section 8. The computational data agreed with the experimental data within  $\pm$  3 dB for all three test bodies, but only at frequencies above 200 MHz and antenna-body distances above 0.1m.

A poor agreement between experimental data and computation was noticed at frequencies below 200 MHz if  $d_{at}$  was smaller than 0.2 m. The reasons for this discrepancy can now be explained by the computational data found in section 10.:

- In the extreme proximity of the body the radial field exceeds the vertical polarized field, especially at  $\lambda/2$  resonance ( $\sim$  65 MHz) and at  $\lambda$  resonance ( $\sim$  150 MHz)
- Electrically small monopole antennas with insufficient counterpoise exhibit a considerable transverse sensitivity (receiving case) or radiate a not wholly pure vertical polarized field (transmitting case). The antenna tests in TABLE 51 tell us that the horizontal (or radial) sensitivity is only about 0-12 dB below the vertical sensitivity at 75 MHz, 19 dB at 125 MHz and 16-18 dB at 205 MHz. The antenna data at 101 MHz (25 dB) and 158 MHz (21 dB) are satisfactory, resulting in a better agreement as can be seen in FIGURES 56 and 57.
- Remotely fed test antennas disturb the fields in the proximity of the body, and especially at low signal levels a part of the signal is picked up by the feeding cable, even when surface waves are attenuated according to FIGURE 44.

Another problem was the fixed relative  $(h_B)$  and absolute  $(h_1)$  antenna height and the missing data concerning the field homogeneity.

The purpose of the extended experiments are therefore defined as:

- Separate measurement or generation of radial and vertical field components in the frequency region 50 to 200 MHz without disturbing the fields around the test body.
- Measurement of the  $h_1$  and  $h_B$ -dependence of the field components with and without test bodies for field homogeneity studies.

In section 9.1.5. the reciprocity theorem has been verified, so that the test antenna could be a receiving or a transmitting antenna for our extended experiments. Both methods have their advantages for special applications:

<u>Probe receiving antenna A1</u>. A common probe antenna consists of a small dipole (whip, helical or conical for broadband) equipped with a rectifier attached to highly resistive, twisted cables. A remote precision DC amplifier measures the signal without range switching from about 1 mV to 1 V (see e.g. BELSHER [9]) almost linearly. The problem is the very low signal at A1 (e.g.  $E_V$  at 100 MHz and  $d_{at} < 0.1$  m) and the relative high field around the connecting cables. A better method would be to build the DC-amplifier and a fiber-optic transmitter close to A1, but then problems have to be solved concerning power consumption (LED's!) and amplifier stability. This method would be best if broadband characteristic is urgently required and if the signal to be measured is much larger than the other RF-signals in the air. Selective built-in receivers cannot be recommended due to limited amplitude range, tuning problems and stability.

<u>Probe transmitting antenna A1</u>. Free-oscillating miniature transmitters cannot be recommended due to amplitude and especially frequency stability problems, because the antenna load is not stable. Quartz-stabilized frequency synthesizers are not suited due to power consumption and space requirements. Thus separate, quartz-stabilized fixed-frequency transmitters offer the best solution concerning volume, power consumption, stability and costs. The main problem is the preset frequency, but on the other hand each probe transmitter can be matched properly to the suited antenna with best long-time stability. Such probe transmitters will be shown in section 11.3. and were used in the following experiments.

#### 11.2. ANTENNA MANIPULATOR

A special antenna manipulator (FIGURE 85 and 86) has been developed with the following features:

- Translation of a complete transmitter along the vertical axis from  $0.1 < h_B < 1.7$  m. Continuous remote translation with permanent  $h_B$  recording (rubber band goniometry, accuracy better than 5 cm)
- Rotation of A1 around the antenna center for p1 vertical to radial
- Accurate and stable positioning of dat at 0.1, 0.2, 0.3 and 0.4m.



FIGURE 85 Antenna manipulator

- 1 : test body MET
- 2 : d<sub>at</sub>-spacers, 15 mm Ø PVC tubes screwed perpendicular to the surface of the test body
- 4 : anchoring strings and deflection pully for strings (6)
- 5 : basement with deflection pully
  (screwed on revolving stage)
- 6 : thin plastic strings for remote manual up/down pulling of the wagon carrying the antenna
- 7 : wagon, PVC plate with four PVC wheels shaped like cotton reels
- 8 : small revolving disk with lock for antenna rotation around the antenna center (details:FIG.86)

The main parts of the antenna manipulator are shown in FIGURE 85. In addition a rubber band goniometer (similar to FIGURE 45) in the horizontal plane on the basement (5) measures the antenna height. For outdoor experiments the top (4) of the trackway is fixed to the revolving stage by stretched strings to prevent mechanical oscillations.

For the experiments with standard body-earth spacing s (s = 0.2 m) the test body is standing directly on the supporting revolving stage (FIGURE 45) for experiments with s = 0.7 m on wooden precision spacers.

#### 11.3. ELECTRICALLY SMALL DIPOLE ANTENNAS WITH BUILT-IN OSCILLATORS

A test transmitter consists of an electrically small dipole antenna and of an autonomous RF-generator (FIGURE 86). The test transmitters have been developed in order to obtain very stable, independent and miniature field sources and are denoted as AO1 (65 MHz), AO2 (74 MHz), AO3 (101 MHz) and AO4 (164 MHz). The helical dipole antenna of 2h < 0.1m and  $D_h = 11mm$  have been tuned to resonance (TABLE 88) and the RF-generators are mounted perpendicular to the center of the antennas in the neutral antenna plane.



FIGURE 86 Test antenna A1 with built-in oscillator on antenna manipulator

- 1 : dat-spacer (see also FIGURE 85) 6 : lock of the revolving disk
- 2 : plexiglass vertical trackway 7 : helical dipole antenna Aj
- 3 : quick-fixing device for dat
- 4 : antenna wagon with pulling strings 9 : 9V alkaline battery 540 mAh
- 5 : small revolving disk carrying A1 10 : quartz (3<sup>th</sup> harmonic mode) for p1 = vertical to radial 11 : RF-antenna coupler

- 8 : RF-generator with:

The construction of miniature, unshielded RF-generators of high amplitude stability is quite difficult at frequencies above 50 MHz. A solution was found in the modification of available high-standard RF-suboscillators of professional walkie-talkies (65,74 and 164 MHz) and in a special construction for 101 MHz. The main specifications of the final RF-oscillators are listed in TABLE 87, the output power amounts to 1-10 mW.

FREQ [MHz]	SUBOSCILLATOR manufacturer type	ADDITIONAL ELEMENTS (RF-coupling etc.)	INPUT CURR. [mA]	AMPLITUDE STABILITY (*) [dB]	DIMENSIONS without bat. [mm]
65	AUTOPHON (special)	ferrit 50 to 50 Ω balun,mod.TV balun	7	0.5	27 x 27 x 30
74	MOTOROLA KXN1067A	external 5 V reg., antenna center coil used as oscil. coil	8	0.1	19 x 9 x 28
101	WAFFEN FABRIK THUN (spec.)	inductive coupling 3 turns around osc.	6	1.0	50 x 40 x 10 (few elem.)
164	MOTOROLA KXN1041A 54.667 MHz	5 V reg.,RF-amplif. freq. multiplier antenna center coil used as 164MHz coil	40	0.5 (*): dur- ing 2 hours	19 x 9 x 28 + 25 x 17 x 15

TABLE 87 Specifications of the built-in quartz RF oscillators

The helical dipole antennas were computed according to Appendix 16.1. and tuned with the help of a network analyser. The application of lossy conductors (PVC insulation) increases the bandwidth (damping effect in a resonante RLC network) but decreases the (non important) efficiency. The data of the antennas are listed in TABLE 88.

FREQ [MHz]	TOTAL TURNS No.	CENTER TURNS No.	TOTAL LENGTH [mm]	CENTER LENGTH [mm]	CONDUCTOR MATERIAL DIAMETER,INSULAT. [mm]	-3 dB BANDWIDTH (analyser data) [MHz]
65	121	7	90	8	0.4 PVC center sec. 0.2 enamelled wire	64 to 66
74	97	7	94	9	0.5 enamelled wire	72 to 75
101	72	6	82	9	0.8 enamelled wire	99 to 102
164	45	5	95	9	0.8 PVC	159 to 167

,

TABLE 88 Specifications of the helical dipole antennas A1

During the network analyser measurements the center turns of the dipole antenna were coupled to the  $50 \,\Omega$  coaxial measuring cable with a miniature  $50 \text{ to } 50 \,\Omega$  ferrit balun (Appendix 16.1.1.). The obtained - 3 dB bandwidth may be different when coupled to the actual RF-oscillator and is generally larger at imperfect matching (reduced efficiency).

From the test transmitters A01 to A04 one cannot expect a totally omnidirectional radiation pattern with strict linear polarization. The transmitter's tests, however, revealed a very stable (0.5 dB) azimuthal radiation pattern at  $p_1$  = vertical (see TABLE 90), and the actual experiments revealed that the transverse polarization (polarization perpendicular to the antenna axis) is about 10 to 15 dB smaller than the main polarization.

# 11.4. TEST PROGRAM AND SOME EXPERIMENTAL RESULTS OBTAINED WITH A0 1 TO A0 4

1. PREPARATIONS						
1.1. Mounting of the h <sub>1</sub> -goniometer and calibrations 1.2. Warming-up of the RF-equipment and recorder, initial calibrations 1.3. Measuring of the FSL at p <sub>1</sub> = vertical and radial versus $\phi$ at h <sub>1</sub> = 1.2m and p <sub>2</sub> = vertical. (see results in TABLE 90) 1.4. Field homogeneity measurements at 0.7 < h <sub>1</sub> < 2.0m at $\phi$ = 0 <sup>0</sup> 1.5. Calibration of FSL to 0 dB at p <sub>1</sub> = vertical, h <sub>1</sub> = 1.2m, $\phi$ = 0 <sup>0</sup>						
2. TRANSMISSION EXPERIMENTS WITH VARIABLE ANTENNA HEIGHTS h						
2.1. MET at $\phi = 0^{\circ}$ , $d_{at} = 0.1m$ , $s = 0.2m$ , $p_1 = vertical/radial, 0.7 < h_1 < 1.5$ 2.2. " $d_{at} = 0.2m$ , " " " " " " " " " " " " " " " " " " "						
3. TRANSMISSION EXPERIMENTS WITH VARIABLE AZIMUTHAL ANGLE $\boldsymbol{\varphi}$						
3.1. MET at $d_{at} = 0.1m$ , $s = 0.2m$ , $h_1 = 1.2m$ , $p_1 = vertical/radial, 0 < \phi < 360^{\circ}$ 3.2. " $d_{at} = 0.2m$ , " " " " " " " " " " " " " " " " " " "						

TABLE 89 Summarized experiments with test transmitters A01, A02, A03, A04

The test set-up consists of a revolving stage, carrying the test body and the antenna manipulator, and the remote receiving antenna A<sub>2</sub> at d = 31 m and h<sub>2</sub>=6.2m as shown in the similar test set-up in FIGURE 44. The performance of the LPD-antenna A<sub>2</sub> is only specified for 100 - 1000 MHz. Below 100 MHz A<sub>2</sub> is suited for relative field measurements of vertical polarized field components (p<sub>2</sub>= vertical) but will also pick-up field components of other polarizations. The performance of the test transmitters A01 to A04 was measured by recording the azimuthal radiation pattern at p<sub>1</sub> = vertical/radial without TS. If both A<sub>1</sub> and A<sub>2</sub> would be strictly linear polarized, and if there would be no ground reflections, the following data had to be obtained: 1.) p<sub>1</sub> = vertical: E<sub>0</sub> stable at 0 dB from 0 < $\phi$  < 360°. 2.) p<sub>1</sub> = radial : A maximum of - 16 dB (E<sub>0</sub> cosθ<sub>1</sub>) should occur at  $\phi$  = 0, 180 and 360°, and the signal should drop to -∞dB at  $\phi$  = 90 and 270°. The actual experimental data are listed in TABLE 90 :

FREQ.	VERTICAL POL.	RADIAL POLARIZATION				
	Amplitude	Maximum Amplitude	۲ Ampli	linimum tude	n Signal Angle o	
[MHz]	[dB]	[dB]	[0]	[dB	[0]	
65	± 0.5	-1, -3	80,260	-9,	-10	165, 350
74	± 0.25	-10, -7	45,220	-15,	-32	115, 320
101	± 0.5	-6,-9,-6	0,185,360	-16,	-13	105,260
164	± 0.25	-6,-8,-6	0,190,360	-14,	-12	110,270

TABLE 90 Experimental data of the performance of the test transmitters A01 to A04 at h<sub>1</sub> = 1.2 m, p<sub>2</sub> = vertical,  $\theta_i = 80.8^{\circ}$  in proximity to ground.

The test transmitters A03 and A04 perform best because the azimuthal radiation patterns are symmetrical and follow the predicted pattern. The test transmitters A01 (65 MHz) and A02 (74 MHz) have a disturbed azimuthal radiation pattern at  $p_1$  = radial, caused primarily by ground reflections and by the elliptical polarization of the helical A<sub>1</sub> antennas. However, measurements of the dominant field components in the proximity of the TS should be possible with a reduced accuracy.

The next test is concerned with the field homogeneity at  $p_1$  = vertical and radial at variable antenna heights without TS. The results are shown in TABLE 91 for both polarizations at 0.8 <  $h_1$  < 1.8 m.

VERTICALLY POLARIZED FIELD AMPLITUDE VARIATION AT VARIABLE ANTENNA HEIGHTS H_1						
FREQUENCY [MHz]	RELATIVE F h <sub>l</sub> = 0.8 m	REE-SPACE F h <sub>l</sub> = 1.0 m	ield strend h, = 1.2 m	бтн Е <sub>0</sub> (h]) h <sub>]</sub> = 1.4 m	IN DECIBELS h <sub>l</sub> = 1.6 m	AT HEIGHT h <sub>l</sub> = 1.8 m
65	+ 0.5	+ 0.0	+ 0.0	- 0.0	- 0.5	- 0.5
74	+ 1.0	+ 0.5	+ 0.0	- 0.5	- 1.0	- 1.0
101	+ 1.5	+ 0.5	+ 0.0	+ 0.0	+ 0.0	+ 0.5
164	- 1.5	- 1.0	+ 0.0	+ 0.5	+ 1.5	+ 2.0

TOTAL FIELD AMPLITUDE AT P1 = RADIAL / P2 = VERTICAL AT VARIABLE ANTENNA HEIGHTS						
FREQUENCY [MHz]	RELATIVE F hj = 0.8 m	REE-SPACE F hj = 1.0 m	IELD STREN( hן = 1.2 m	STH RELATED իլ = 1.4 m	TO Ev AT ⊨ h] = 1.6 m	lι=1.2m,φ=0 <sup>0</sup> hι = 1.8 m
65	-10.5	-10.2	-10.0	- 9.5	- 9.0	- 8.2
74	-14.5	-13.5	-13.0	-12.5	-12.0	-11.0
101	- 7.0	- 6.6	- 6.0	- 5.0	- 4.5	- 4.5
164	- 6.0	- 5.5	- 6.0	- 7.0	- 8.0	- 9.0

TABLE 9] Field homogeneities of the field components at  $p_1$  = vertical (above) and  $p_1$  = radial (below), measured along the theoretical vertical axis of the TS (without TS) at  $\phi = 0^{\circ}$  and  $p_2$  = vertical. The reference field strength is the FSL (0 dB), measured with  $p_1$  = vertical at  $h_1$  = 1.2 m. The upper data in TABLE 91 show that the field homogeneity at  $p_1$  =  $p_2$  = vertical is within 1.5 dB at antenna heights  $h_1$  from 0.8 to 1.6 m.

The data below in TABLE 91 are a measure for the transversal Gain<sub>T</sub> of the antenna  $A_1$ , if  $A_2$  is assumed to be strictly linear polarized. The measured  $E_{tot}$  is the superposition of:

 $E_{tot} = E_0 \cos \theta_i + E_0 \sin \theta_i \cdot \cos \Delta \psi \cdot Gain_T$ (245)

 $\Delta \psi$  is the unknown argument of the transversal Gain<sub>T</sub>. With (245), E<sub>tot</sub> and  $\Delta \psi = 0^{\circ}$  one obtains a Gain<sub>T</sub>\*, which is the minimum of the actual Gain<sub>T</sub>( $\Delta \psi$ ):

Measured E<sub>tot</sub> at p<sub>1</sub> = radial,p<sub>2</sub> = vert. Minimum transversal Gain<sub>T</sub>

< - 6.8 dB	>	- 10.5 dB
< - 9.9 dB	>	- 14.0 dB
< -11.8 dB	>	- 20.0 dB

Thus, one may assume that the transverse polarization is about - 9 to -20 dB (mean values of TABLE 91, which are depending on the ground reflection)

# 12. COMPARISON OF IMPROVED EXPERIMENTAL DATA WITH THREE-DI-MENSIONAL COMPUTATIONAL DATA

#### 12.1. INVESTIGATED PARAMETERS

#### 12.1.1. EFFECT OF THE FREQUENCY ON THE FIELD COMPONENTS AT FZYL AND MET

Let us first compare the previous experimental data of section 9. with the new FZYL computational data of section 10.4.:



FIGURE 92 Gain<sub>B</sub> versus f at  $d_{at} = 0.1 \text{ m}$  and  $\phi = 180^{\circ}$ . Comparison between experimental MET data (9.1.1.) and computational FZYL data  $E_v$  and  $E_r$ .

FIGURE 92 shows the experimental MET data measured with the previous monopole antennas AT1 to AT8 (8.3.1.) at 74,101,125,158, 205, 250, 400, 562, 700 and 897 MHz. Theoretically these measuring data should be close to the computes  $E_V$  curve. In fact, a good agreement is achieved at higher frequencies 250, 400, 562, 700 and 897 MHz. At lower frequencies, however, the agreement is poor; the experimental data are generally much higher than the computed  $E_V$ 's. It seems to be evident that the monopole antennas with their insufficient counterpoise do not only respond to the weak vertical field component but also to the very strong radial field component. The experiments with the new test transmitters A01 to A04 prove the existence of the strong radial field components. The experiments are based on the verified reciprocity theorem (9.1.5.) and in order to quantify the effect of the proximity to the ground the experiments have been performed twice: first experiment with s = 0.2 m,  $h_1 = 1.2 \text{ m}$  ( $h_B = 1.0 \text{ m}$ ) and second experiment with s = 0.7 m,  $h_1 = 1.7 \text{ m}$  ( $h_B = 1.0 \text{ m}$ ). The results of the experiments are shown in FIGURE 93 :



FIGURE 93 Gain<sub>B</sub> versus f at  $d_{at} = 0.1 \text{ m}$  and  $\phi = 180^{\circ}$ . Comparison between experimental MET data (obtained with the test transmitters A01 to A04) and computational FZYL data at vertical and radial polarizations. Standard experiments : full symbols,  $h_1 = 1.2 \text{ m}$ ,  $h_B = 1.0 \text{ m}$ , s = 0.2 m. Experiments with reduced ground effects: empty symbols,  $h_1 = 1.7 \text{ m}$ ,  $h_B =$ 1.0 m, s = 0.7 m. (Test body separated by wooden spacers from the stage)

<u>Vertical field components  $E_V$ </u>: The experimental data agree within 3 dB with the computational data, except at 75 MHz where the difference amounts to 5 dB. If one would shift the computed FZYL-data by 5 % to the left the agreement would be 2 dB. Thus, it could be that the actual resonant frequency is about 5 % lower than computed. The effect of the proximity to the ground is not very important, the differences are smaller than 3 dB.

- 197 -

<u>Radial field components Er</u> : FIGURE 93 proves the existence of the theoretical predicted radial field components. The Er are more than 10 dB larger than Ev at  $d_{at} = 0.1$ . The experimental data agree within 2.5 dB with the computational data, except at 164 MHz where the difference amounts to 4 dB. Similar to the Ev-components, a better agreement could be achieved (but only for 65, 74 and 101 MHz) by shifting the FZYL-data by about 5 % to the left. The proximity to the ground results in a 2 dB difference at 65 and 74 MHz and a 4 dB difference at 101 and 164 MHz (s : 0.7/0.2m).

#### 12.1.2. EFFECT OF ANTENNA HEIGHT AND PROXIMITY TO THE GROUND

The computational data are valid for a body in free space, for a standard relative antenna height  $h_B = 0.1 \,\mathrm{m}$  and for a standard irradiation angle  $\theta_i$  of 80.8°. The experimental data are obtained from a test body in proximity to the ground due to the reasons explained in section 5.3.1. The antenna-body system is a resonant circuit as we can see from FIGURE 93. Any resonant system is very sensitive to external influences so that in our case the antenna-body system may change its function (e.g., resonant frequency) in proximity to the ground. In order to check if  $h_B$  is an exceptional relative antenna height (i.e., not a representantive height) and in order to quantify the effect of the proximity to the ground both  $h_B$  (relative antenna height) and  $h_1$  (absolute antenna height) were varied and compared with the computational data in FIGURES 94 a,b,c,d.

Effect of the proximity to the ground : The FSL vary only by  $\Delta < 2$  dB from  $1.2 < h_1 < 1.7 m$ . The computed data are calibrated to FSL (s = 0.2 m) and to 'FSL' (s = 0.7 m). A strong influence of the proximity of the ground is only noticed at  $E_V$  (65 MHz) and  $E_V$  (74 MHz) due to the different  $E_V(h_B)$  pattern. At a fixed  $h_B$  of 1.0 m the differences due to ground proximity are well below 3 dB for all  $E_V$  and  $E_r$ .

<u>Effect of the relative antenna height</u> : Generally, the  $E_r$  components increase with increasing  $h_B$  (except at 164 MHz, where  $E_r$  is almost constant) and vary up to  $\pm 5 \, dB$  in the range  $0.8 < h_B < 1.2 \, m$ . The  $E_v$  components are much lower than  $E_r$  and vary up to  $\pm 5 \, dB$  in the same range. An  $h_B$  of 1.0 m is not an extraordinary antenna height: only at 74 MHz the  $E_v$  of the isolated body (s = 0.7m) is close to a point of inflexion.







FIGURE 94b Effect of ground proximity and relative antenna height, 74 MHz



FIGURE 94 c Effectof ground proximity and relative antenna height, 101 MHz



FIGURE 94 d Effect of ground proximity and relative antenna height, 164 MHz

#### 12.1.3. EFFECT OF THE ANTENNA-BODY DISTANCE

FIGURE 92 and 93 revealed very low Ev's above 100 MHz and large Er's at frequencies between 50 to 300 MHz at  $d_{at} = 0.1 \text{ m}$  and  $\phi = 180^{\circ}$ . FIGURES 95 a, b and c show the corresponding data for  $d_{a+} = 0.2$ , 0.3, and 0.4m. Similar to FIGURE 92 the experimental  $E_v$  data above 250 MHz agree best with the computational data, the typical error is less than 2 dB. At lower frequencies the experimental  $E_r$  and  $E_V$  follow the computed patterns and the agreement between experiment and computations is generally better than 4 dB. The comparison of the experimental and computational pattern leads to the assumption that the MET body resonates about 5% lower than the FZYL. This effect could be explained by the larger circumference of MET (sharp cylinder ends) compared with FZYL (round end caps). FIGURES 95 a,b,c show clearly the increase of  $E_v$  with increasing  $d_{at}$  and the decrease of  $E_r$  with increasing dat. FIGURE 95 c demonstrates the equilibrium of Ey and Er at  $d_{at}$  = 0.4 m in the frequency region 75 to 170 MHz experimentally and theoretically. At smaller  $d_{at}$  the radial component  $E_r$  is dominant and can be observed from minimum 40 MHz to maximum 500 MHz.



FIGURE 95 a Gain<sub>B</sub> versus f at  $\phi = 180^{\circ}$  and  $d_{at} = 0.2 \text{ m}$ . Comparison between experimental MET data and computational FZYL data. Standard experiments with  $\theta_1 = 80.8^{\circ}$ ,  $h_1 = 1.2 \text{ m}$ , s = 0.2 m,  $h_B = 1.0 \text{ m}$  with A01-4 and AT3-8.







FIGURE 95 c Gaing versus f at  $\varphi$  = 180° and dat = 0.4 m. Comparison like 95 a.

The experimental data in section 9.1.2. revealed only small differences between the test bodies MET, PHA and SUB at larger  $d_{at}$ 's. Thus, the experimental SUB data and the two computational data MANMOD1 and 2 are presented in FIGURE 96 for the most critical antenna-body distance  $d_{at} = 0.1$ in the shadow zone  $\phi = 180^{\circ}$ :



FIGURE 96 Gain<sub>B</sub> versus f at dat = 0.1 and  $\phi$  = 180<sup>O</sup>. Comparison between experimental SUB data and computational MANMOD1&2 data for E<sub>r</sub> and E<sub>v</sub> at two different absolute antenna heights. Full symbols : h<sub>1</sub> = 1.2 m, h<sub>B</sub> = 1.0 m, s = 0.2 m; empty symbols : h<sub>1</sub> = 1.7 m, h<sub>B</sub> = 1.0 m, s = 0.7 m.

<u>Vertical field components  $E_V$ </u>: 9 of the ll experimental data agree with the the MANMOD l or 2 - data within 3 dB. The experimental 400 MHz  $E_V$  is 4.5 dB higher than computed and the experimental 101 MHz  $E_V$  is 6.5 dB larger.

<u>Radial field components  $E_r$ </u>: All investigated frequencies prove the existence of the very large radial field components. The maximum difference between experiment and theory amounts to 3 dB (s = 0.7) and 5 dB (s = 0.2 m). The radial field component at a human body is up to 10 dB larger than the vertical component. Much higher radial fields were recorded with decreased dat, but are not presented here due to insufficient accuracy.

#### 12.1.5. AZIMUTHAL RADIATION PATTERNS OF MET, SUB, FZYL AND MANMOD 1 & 2

The azimuthal radiation patterns at frequencies above 200 MHZ (above resonance) have been treated in section 9.1.2. (TABLE 60) and 9.1.3. (FIG-URE 59) since they could be explained with the simple two-dimensional computations on the IZYL model.

The following FIGURES 97 a,b,c,d show the experimental and the computational azimuthal radiation patterns at 65, 74, 101 and 164 MHz at  $d_{at} =$  0.1m of MET and FZYL, and FIGURES 98 a,b,c,d that of SUB and MANMOD1&2. The experimental data have been recorded with the test transmitters A01 to A04 (11.3.) in proximity to the ground (s = 0.2m, h<sub>1</sub> = 1.2m) and represent realistic azimuthal radiation patterns for practical applications. Because complete 0-180-360° revolutions have always been recorded, two 0-180° recordings may appear for the vertical component E<sub>V</sub> and the radial component E<sub>r</sub> due to the asymmetry of the test set-up (antenna manipulator, radial antenna not perfectly adjusted in the horizontal plane). For the following discussion of the comparison between experimental data (MET and SUB) and computational data (FZYL and MANMOD1&2) the mean value of the experimental data at the distinct angle  $\phi$  will be regarded.

<u>65 MHz</u>: The MET E<sub>V</sub> agree with FZYL E<sub>V</sub> at  $\phi > 100^{\circ}$  within 3 dB, but at  $0^{\circ}$  the MET E<sub>V</sub> are up to 14.5 dB higher than computed. Comparing MET with the FZYL data at 50 and 60 MHz one may assume that the discrepancy is caused by a lower resonant frequency of MET. The MET E<sub>r</sub> agree, however, with FZYL E<sub>r</sub> for all  $\phi$ 's within 3 dB.

The SUB E<sub>v</sub> agree with MANMOD1&2 at  $\phi > 90^{\circ}$  within 4 dB, but at  $0^{\circ}$  the SUB E<sub>v</sub> are up to 10 dB higher than computed. The SUB E<sub>r</sub> agree, however, with MANMOD1 E<sub>r</sub> for all  $\phi$ 's within 3 dB.

The minimum radial component is always stronger than the maximum vertical component. Using the radial component for omnidirectional transmission an improvement of 7 to 13 dB (MET) and of 0 to 4 dB (SUB) can be achieved in realistic conditions.

<u>74 MHz</u>: The MET  $E_v$  agree with FZYL  $E_v$  at  $\phi > 140^\circ$  within 3 dB, but do not show the extreme loss at 110°, caused perhaps by a cross talk of the horizontal component of the elliptical polarized A<sub>1</sub> antenna. The SUB  $E_r$  agree with the FZYL  $E_r$  for all  $\phi$ 's within 4.5 dB.







FIGURE 98a Azimuthal radiation pattern SUB and MANMOD1&2, same dat and f.







FIGURE 98b Azimuthal radiation pattern SUB and MANMOD1&2, same  $d_{\mbox{at}}$  and f.











FIGURE 97 d Azimuthal radiation pattern MET and FZYL at  $d_{at} = 0.1 \text{ m}$ , 164 MHz.



FIGURE 98d Azimuthal radiation pattern SUB and MANMOD1&2, same  ${\rm d}_{{\rm at}}$  and f.

<u>74 MHz</u> : (continued) The SUB  $E_v$  agree with MANMOD 2  $E_v$  at  $\phi > 90^{\circ}$  within 3 dB, but at 0-90° the SUB  $E_v$  are up to 13 dB higher than computed. The SUB  $E_r$  agree with MANMOD 1  $E_r$  for all  $\phi$ 's within 6 dB.

Generally the minimum radial component is always stronger than the maximum vertical component (except SUB:  $\phi = 0^{\circ}$ ). Using the radial component, the omnidirectional transmission can be improved by 12 to 14 dB (MET) and 5.5 to 8 dB (SUB) in realistic conditions.

<u>101 MHz</u> : The MET E<sub>V</sub> agree with FZYL E<sub>V</sub> at all  $\phi$ 's within 5 dB. A much better agreement could be achieved by comparing MET E<sub>V</sub> with FZYL E<sub>V</sub> at 85 to 95 MHz (see FIGURE 77 c). One may assume that MET resonates 10 % lower than computed (MET: sharp cylinder ends, FZYL: round end caps). The MET E<sub>V</sub> agree with FZYL E<sub>r</sub> at all  $\phi$ 's within 5 dB, compared with FZYL E<sub>r</sub> at 95 MHz within 3 dB.

The SUB  $E_v$  agree with MANMOD1  $E_v$  at  $\phi < 90^\circ$  within 3 dB, but at 90 to 180<sup>°</sup> the SUB  $E_v$  are up to 8 dB higher than computed. The SUB  $E_r$  agree with MANMOD1  $E_r$  at all  $\phi$ 's within 5 dB.

The minimum radial component is always stronger than the maximum vertical component. Using the radial component, the omnidirectional transmission can be improved by 15 to 17 dB (MET) and 5 to 8 dB (SUB) in realistic conditions.

<u>164 MHz</u> : The MET  $E_v$  agree with FZYL  $E_v$ (150 MHz) at all  $\phi$ 's within 5 dB, and with FZYL  $E_v$  (162 MHz) within 3 dB. The MET  $E_r$  agree with FZYL  $E_r$  (162 MHz) at all  $\phi$ 's within 5 dB.

The SUB E<sub>v</sub> agree with MANMOD 2 E<sub>v</sub> at all  $\phi$ 's within 4 dB. The SUB E<sub>r</sub> agree with MANMOD 1 E<sub>r</sub> at all  $\phi$ 's within 3.5 dB. (Only MANMOD 1 & 2 data at 150 MHz are available)

Generally the minimum radial component is always stronger than the maximum vertical component (except SUB:  $\phi = 0^{\circ}$ ). Using the radial component, the omnidirectional transmission can be improved by 14 to 21 dB (MET) and 9 to 11 dB (SUB) in realistic conditions.

# 12.2. DISCUSSION OF THE LIMITATIONS OF EXPERIMENT AND COMPUTATION

Generally, the experimental data agree with the computational data within  $\pm$  3 dB at all frequencies and antenna-body-distances as small as 0.1 m.

Some experimental data in the resonance region differ more than 3 dB from the computational data, especially  $E_V$  at  $\phi = 0^{\circ}$  in proximity to the ground is larger than computed. However, the important data from the shadow zone and the big difference between  $E_V$  and  $E_r$  are of satisfactory agreement. Taking into account the large signal range from -24 to + 6 dB the agreement between experiment and theory is satisfactory. A difference of  $\pm$  3 dB corresponds to a power variation of only 1% F.S., related to 0 dB = FSL = 100 %.

The experimental errors are caused mainly by five reasons:

- Capacitive coupling of the body with the ground. The resonant frequency depends on the proximity to the ground, as demonstrated by GANDHI et al. [24] in FIGURES 4 and 6. In our experiments with s = 0.2 and 0.7 m the difference amounts to maximum 3 dB at  $\phi = 180^{\circ}$ .
- Transverse polarization of the test antenna  $A_1$ . The applied helical monopole and dipole antennas are elliptically polarized, so that theoretical signals below -10 dB can be superimposed by stronger transversally polarized field components which determine the recorded data.
- Transverse polarization of the remote antenna  $A_2$ . The LPD antenna is only specified for the 100-1000 MHz range. A cross-talk of transversally polarized field components (see TABLE 91) is very probable.
- Symmetry of fields in the proximity of the test body. The antenna-manipulator contains no metallic parts, but the dielectric material may cause field disturbances. The test transmitters are not infinitesimally small and the orientation of the antenna A<sub>1</sub> may vary from the ideal value by about 5°, causing phase errors and thus amplitude errors.
- Position and shape of the test bodies. The metallic cylinder with its sharp ends (vessel without top and bottom plates) does not correspond completely to the computational cylinder FZYL with its round end caps. The human test subject is not rotationally symmetric and during the measurements a change of the position (vertical axis inclined by a few degrees) and a change of the shape (breathing, etc.) cannot be excluded. The shape of the human body is a very important factor in the resonance region, and the computational differences between MANMOD1 and MANMOD 2 are in the same order of magnitude as the difference between the experimental SUB and computational MANMOD data.

The limitations and the accuracy of the computational model have been discussed in section 10.3.4.. Principally, the accurate computation is limited to frequencies below 500 MHz with the standard parameter set and each computational result needs to be carefully checked. The computation of a test point is not accurate a priori: only if the frequency, the position of the test point and the test segment length have been varied, without large changes of the result, are the computational data reliable.

The computational errors are caused mainly by three reasons:

- At small dat one can only compute the averaged field components in the environment of the selected test point.
- The computational field data depend very much on the shape of the body. If the body model is of complicated shape (MANMOD1&2) the standard number of contour points is at the lower limit and the field data vary greatly at small variations of  $d_{at}$ ,  $h_B$  and f. The experiments in 9.1.1. have lead to the conclusion that the body material is of little significance at  $d_{at}$  above 0.05 m and f above 200 MHz. Thus, the computational model for a human body should be first adapted to the asymmetric body shape and for frequencies below 200 MHz later on to the body material. However, an improvement of the body model is of secondary significance with respect to the practical applications of the obtained data, because the agreement between experimental and computational data is already satisfactory for the fields <u>outside</u> the human body.
- Computational effort. The present state of art allows the computation of 4 test points with a computational time of about 700 seconds on a CDC 6500 computer. More accurate computations are only possible with considerably improved computers with higher speed and more storage capacity.

# 13. CONCLUSIONS AND PERSPECTIVES

## 13.1. IMPORTANT INVESTIGATED PARAMETERS OF THE ANTENNA-BODY SYSTEM

## 13.1.1. OVERVIEW OF THE INVESTIGATED ANTENNA-BODY SYSTEM

The purpose of the antenna-body study has been defined in section 2.. A standard test situation according to FIGURE 11 has been selected which represents the actual operational conditions on one hand and which could be computed on the other hand. Different antenna-body models have been computed and the results have been compared with corresponding experimental data obtained with representative body models. A relatively simple, analytically treatable, computational model was found which explains the effects of the human body on the EM field at frequencies above 200 MHz. A more complicated, numerically treatable, computational model was found which explains the effects in the entire investigated frequency range from 10 to 1000 MHz. Reliable experiments could be performed in the frequency range from 65 to 900 MHz. The agreement between theory and experiment was generally better than 3 dB (corresponding to a 1 % F.S. accuracy at power levels ranging from -25 to + 6 dB) with some few exceptions at extremely small antenna-body distances (see section 9. and 12.).

Interesting systematic computable correlations were found among frequency, body geometry, relative position of the antenna and transmission loss. The main question concerned the worst-case transmission loss in the azimuthal radiation pattern  $0 < \phi < 180^{\circ}$ . The parameters determining the performance may be listed in the order of their importance as follows:

- frequency (f) and size (length) of the test subject  $(L_B)$
- antenna-body distance (dat)
- polarization of the body-mounted antenna (p1)
- relative antenna height of the body-mounted antenna (h<sub>R</sub>)
- lateral and sagittal diameter of the test subject (D<sub>B</sub>)
- body material

The most interesting feature of the human body is the field polarization transformation effect which will lead to a new class of electrically small antennas for extremely close mounting on the human body. As found by computations and experiments, omnidirectional transmissions with less than 6 dB transmission loss are practically possible with antennas of less

 $6 \times 6 \times 6$  cm dimensions mounted directly on the surface of the body. These antennas can be used for transmitters and receivers in the frequency range from about 60 to 160 MHz if the remote antenna A<sub>2</sub> is vertical polarized, without biological problems at power levels up to about 2 W.

# 13.1.2. THE EFFECT OF THE FREQUENCY ON THE FIELD DISTRIBUTION

In the following discussions we regard only the  $p_2$  = vertical polarization at  $p_1$  = vertical, radial and horizontal. The  $p_2$  = horizontal polarization is of little practical interest with respect on omnidirectional transmission, as can be seen from the data in section 10.3.5..

At  $p_2$  = vertical the wavelength of frequencies between 10 and 1000 MHz are of comparable magnitude like the circumference (from head to feet) of the human body. Thus one distinguishes three principal frequency regions:

- below 50 MHz : Rayleigh region (below first resonance)
- 50 to 200 MHz : Mie- or resonance region (including first resonance at about 65 MHz, second resonance at about 140 MHz, and first anti-resonance at about 100 MHz)
- above 200 MHz : Optical region (above second resonance)

Vertical polarized E-field components ( $p_1 = vertical$ ): Considering FIGURE 77 one notices the largest transmission Loss<sub>B</sub> above 85 MHz at  $\phi \sim 180^{\circ}$ . Comparing the d<sub>at</sub> dependences in FIGURE 78 one notices the largest Loss<sub>B</sub> at minimum d<sub>at</sub>. Thus, we have to look at only the Gain<sub>B</sub> versus f diagrams at  $\phi = 180^{\circ}$  at d<sub>at</sub> = 0.1 m for maximum Loss<sub>B</sub> considerations, e.g., FIGURES 92, 93 and 96. At a given (small) d<sub>at</sub> the Loss<sub>B</sub> is relatively small (10 dB) and almost constant from 10 to 50 MHz. Below the first resonance a body is <u>not transparent</u> for EM-waves as could be assumed from the well-known radar cross section (RCS) pattern in FIGURE 17, if we regard small d<sub>at</sub>'s. From 50 to 65 MHz Loss<sub>B</sub> decreases a few, but less than could be assumed by the RCS pattern. At 65 to 120 MHz Loss<sub>B</sub> increases drastically (especially at  $\phi \sim 90^{\circ}$ , see FIGURE 77c), amounting to 15-20 dB at d<sub>at</sub> = 0.1 m. From about 150 to 1000 MHz Loss<sub>B</sub> remains large as demonstrated best by FIGURE 72. The best agreement between theory and experiment is obtained at frequencies above 200 MHz, due to the small radial Er's above 200 MHz.

Radial polarized E-field components  $(p_1 = radial)$ : The radial LossB versus f pattern is completely different from the vertical LossB pattern.

The radial field is maximum at minimum dat and is almost independent f  $\phi$  in the frequency range from 60 to 250 MHz. At frequencies below 30 MI the radial component is small and the minimum is located at  $\phi = 0^{\circ}$  (FIGI 77 ) The radial component becomes interesting above 50 MHz, and its properties are best demonstrated in the Gain<sub>B</sub> versus f diagrams at  $\phi = 18$ (FIGURES 92,93 and 96). Loss<sub>B</sub> decreases considerably from 50 to 65 MHz and at dat=0.1m an actual gain of some dB can be observed. A first Los maximum of about 5 dB occurs at about 100 MHz, but not so accentuated a in the vertical polarization. A second Loss<sub>B</sub> minimum is at about 150 MF with about 1 to 3 dB. With increasing frequency Loss<sub>B</sub> increases, too. *F* frequencies above 250 MHz Loss<sub>B</sub> is higher than 6 dB ( $\phi = 0^{\circ}$ , FIGURE 77 b and above 350 MHz the maximum radial Loss<sub>B</sub> is larger than the maximum vertical Loss<sub>B</sub>. However, at  $\phi = 180^{\circ}$  the radial Loss<sub>B</sub> remains small up t about 400 MHz (FIGURES 93 and 96).

The experimental and computational azimuthal radiation patterns in FIG-URES 97 and 98 reveal excellent omnidirectional characteristics with up to 11 dB (human test subject) and 21 dB (metallic cylinder) better performance than compared with the vertical component. For practical appli cations the radial component is limited to frequencies between first and second resonance, that is about 50 to 200 MHz.

Horizontal polarized E-field components ( $p_1 = horizontal$ ): The transmission Loss<sub>B</sub> at horizontal polarization is of little practical significanc as demonstrated with FIGURE 77. At  $\phi = 0$  and  $180^{\circ}$  Loss<sub>B</sub> is very high, onl at  $\phi = 90^{\circ}$  a Loss<sub>B</sub> of less than 20 dB is noticed. Small losses, but only for  $\phi$  around 0 and  $180^{\circ}$ , are only obtained at  $p_2 = horizontal$  which we do not discuss here. Some data for that case are to be found in TABLES 69 and 70 and in the Appendix 16.2.4. for the frequencies 65 to 425 MHz.

#### 13.1.3. THE EFFECT OF THE ANTENNA-BODY DISTANCE

In the following discussion we regard again only the  $p_2$  = vertical polarization. From the practical point of view the small  $d_{at}$ 's are of main interest, but only the knowledge of the Gaing- $d_{at}$  dependence lead to an understanding of the shape of the azimuthal radiation patters.

Vertical polarized E-field components ( $p_1 = vertical$ ): As long as  $d_{at}$  is smaller than  $\lambda/4$  the transmission LossB increases with decreasing  $d_{at}$ . At larger  $d_{at}$ 's standing waves with highly varying LossB are observed.

The human body consists of a material which reflects an EM wave similar to a perfect conductor (see details in section 5.2.4.). The reflection factor for the E-component is thus -1, so that the superimposed signal from scattered and incident field is maximum at  $d_{at} \sim \lambda/4$  at  $\phi = 0$  (see FIGURES 56,57 and 68). A similar situation occurs at  $\phi = 90^{\circ}$ , where the maximum is at  $d_{at} \sim \lambda/2$ . At such extraordinary  $d_{at}/\phi$  conditions the transmission  $Loss_B$  is very low or even a gain of some dB can be measured, so that the human body can be regarded as a reflector similar to the reflector of a Yagi antenna. A guasi-parabolic antenna can be arranged with 3 persons (FIGURE 54), effecting a directive gain of more than 4 dB. Considering the fact that + 3 dB is the double power, this result is quite interesting. This  $\lambda/4$  to  $\lambda/2$ -effect (depending on  $\phi$ ) is also observed in daily life, if one approaches a mobile receiver (e.g. FM, 80 to 120 MHz) which is tuned to a weak radio station : at some distances the signal is very clear and varying some centimeters distortions can be heard. Thus, for frequencies above 300 MHz practical antenna-body distances may be larger than  $\lambda/4$ , effecting large Loss<sub>B</sub> variations in the azimuthal radiation patterns (see e.g. FIGURE 59), which can now be understood and well predicted.

<u>Radial polarized E-field components ( $p_1 = radial$ )</u>: The radial field effect occurs only in the proximity to the body, as can be seen in FIGURE 78. At very small dat and frequencies between 50 to 250 MHz the Loss<sub>B</sub> is very low or even a gain can be observed, independent on  $\phi$ . In this frequency range Loss<sub>B</sub> decreases considerably with decreasing dat, e.g. from 11 dB/dat = 1.0 m to -2 dB/dat = 0.05 m at f = 150 MHz and  $\phi = 180^{\circ}$ . At about dat = 0.3mthe amplitude of the radial components are as large as the amplitude of the vertical components, and above 250 MHz a similar, but inverted pattern like the dat/Gain<sub>B</sub> versus  $\lambda$  in the vertical component can be noticed. Horizontal polarized E-field components ( $p_1$  = horizontal): The horizontal components depend little on dat, because they are essentially only the residual, little disturbed horizontal components of the incident field (see effect of the irradiation angle in FIGURE 76).

#### 13.1.4. THE DOMINANT RULE OF THE RADIAL E-FIELD COMPONENT

At frequencies from about 50 to 200 MHz (first to second resonance of the human body) the human body acts like a very efficient polarization con-

verter. The only conditions are :  $d_{at}$  below 0.3m and  $p_2$  = vertical. At  $d_{at}$  = 0.1m the transmission loss amounts to  $\pm$  5 dB which is much better than the widely varying transmission Loss<sub>B</sub> of -4 dB to -20 dB obtained with a vertical polarized antenna. The mechanism of the polarization transformation effect can be briefly summarized as follows:

- Receiving case: We assume a standing human test subject, irradiated by a plane, mostly vertical polarized wave by a remote antenna with an incident angle of 70 to 90°. At frequencies from about 50 to 200 MHz the relative and the specific absorption cross section (FIGURES 3 and 4) is high. Thus, the body collects not only RF-energy corresponding to its shadow area, but also from outer regions similar to a good receiving antenna with a large effective area. The RF- currents are flowing mainly in the outer layers of the body and surface charges are generated. Both currents and charges produce a scattered field around the body. Because the human body is now the new RF-source, it is not astonishing that a certain field component is largest very close to the surface, and because the vertical and the horizontal components are weak, the radial component must be large due to the total collected RF-energy (this explanation is logical for a metallic body, but only partially acceptable for biological bodies). The source of the tremendous radial field component must be the surface charges. The integral effect is completely described by the Maxwell equations and is proven by experiments. It explains the practically important transmission loss or gain, but a further study on the detailed mechanism may be interesting from the scientific point of view.
- <u>Transmitting case</u>: The reciprocity theorem (section 5.2.1.) says that the transmission loss from  $A_2$  to  $A_1$  is the same as from  $A_1$  to  $A_2$ . With this answer the explanation is given why a radial transmitting antenna performs best, because the receiving case is mathematically clear. However, the detailed mechanism is not clear at all, only in the integral sense is this explanation satisfactory if we speak only of the transmission loss, which was the purpose of this study. It might be that a radial antenna produces above all large surface charges which evoke axial currents. These currents are comparable to those in a thick mono- or dipole, producing a ver-

tical polarized far field. As we shall see in section 13.3.2., an applicable radial antenna will be equipped with a metallic plate on the surface of subject, so that biological problems can be excluded at reasonable RF-power (body shielded from the large reactive nearfield of the antenna). However, a future study on the detailed coupling mechanism might be of scientific interest. In this context it should be mentioned that some unexplainable (parapsychological ?) effects occured with the test transmitters described in section 11. 3. : Some test subjects complained to feel an irritating sensation when the antenna was held radial to the head. Further, an irregular signal could be heard on the monitor receiver if the transmitter was positioned at two distinct points of the upper forehead and to the wrist of the hands. It should be mentioned that the transmitters were quartz stabilized (modulation only possible by selective RF-absorption) and that the CW power was much less than 10 mW.

- Practical significance : The polarization transformation effect is not only very important for specially designed radial antennas, but also for most of the generally applied electrically small antennas, such as the helical monopole antennas of walkie talkies : If selfresonant, electrically small antennas are developed, it is very difficult to obtain a strictly linear polarization. There is always a transverse polarization, e.g., in the case of helical antennas with its elliptical polarization (see Appendix 16.1.) the secondary polarization axis increases with  $\P^2 D_h^2/p \lambda$ , where p is the pitch and  $D_h$  is the helical diameter. Greatly simplified, the transverse polarization is relatively large compared with the desired axial polarization, if the helical antenna length h is small compared with  $\lambda/4$ . In addition, a helical monopole antenna would require a counterpoise of about  $\lambda/2$ diameter in order to operated properly. Transmitting devices are, however, generally much smaller than  $\lambda/4$  (maximum housing dimension), so that the electromagnetic counterpoise is not sufficient. As a consequence, the antenna does not radiate the computed elliptical polarized field, but also additional field components of arbitrary polarizations. For constant antenna and transmitter length the ratio of radial to vertical polarization increases with decreasing frequency. At frequencies of 100 MHz this ratio may amount to about -5 dB, at 200 MHz to about -10 dB and at 400 MHz to about -20 dB. Due to this
inevitable transverse sensitivity (receiving case) and transverse radiation (transmitting case) the transmission loss does not sharply increase above 75 MHz (FIGURE 96) but increases slowly up to about 300 MHz (FIGURE 92). Above 250 MHz not only the investigated radial effect is of smaller influence, but also the antenna and its counterpoise approaches an ideal, strictly linear polarized radiation system. In this case, the computed Gain<sub>B</sub>/f dependence for vertical polarization becomes accurate also for practically realizable mobile radio sets. Thus, the unexpected good performance of practical body-mounted antennas in the proximity of a human test subject at frequencies between 50 and 200 MHz can be now explained by the polarization transformation effect and the technically inevitable transverse polarization of electrically small antennas.

- <u>Perspectives</u> : The polarization transformation effect is of great practical significance and could be proven theoretically and experimentally. Only roughly has the dependence of the relative antenna height been studied (FIGURES 94 a,b,c,d) and little is known about this effect in extreme proximity to the human body. A future study of the parameters  $d_{at}$  (0 to 0.1 m),  $h_B$  (0.8 to 1.8 m),  $\theta_i$  (70 to 90°), bodygeometry and body material would be of great interest. Because the computational cost would be very high (improved model of man with more than 200 contour points required, investigation of the significance of the body material, many dat and  $h_B$  steps necessary), such a future study should start with experiments. Test transmitters similar to those described in section 11., but of reduced size, and an improved antenna manipulator would be required.

# 13.2. INTERESTING ADDITIONAL FEATURES OF THE ANTENNA-BODY SYSTEM

# 13.2.1. THE BROAD-BAND CHARACTERISTICS OF THE HUMAN BODY

The presented study dealt with an antenna-body system where the antenna A<sub>1</sub> was separated from the body. In an other study the active radiation of the human body was investigated (NEUKOMMM [63]). Briefly summarized, the main results of that study are:

 A thick monopole antenna was mounted on a large counterpoise and fed properly by a continuously enlarge coaxial line according to HEILMANN [42], page 110. The radiation pattern of metallic rotational cylinder of a diameter  $D_B = 0.25 \text{ m}$  (diameter of a human body) and a monopole length 1/2  $L_B = 0.9 \text{ m}$  (1/2 human size) was an omnidirectional pattern in the azimuthal plane, with a gain of more than -3 dB at frequencies between 60 to 200 MHz at an elevation angle of about 0 to 10°. At frequencies above 200 MHz a side lobe at an elevation angle of about 45° is developed, causing high losses at 0 to 10°. The results agreed partially with whose of KRAUS [51].

- The same experiment was repeated with a phantom (comparable with PHA in section 5.4.1., but equipped with a similar feeding as mentioned above). The astonishing result was, that the phantom showed almost the same broad-band characteristics, with only little additional losses: from 60 to 180 MHz an omdirectional radiation pattern was obtained with a gain of more than -5 dB (reference: isotropic antenna).
- Actual experiments with human test subject could not be performed, but there is no doubt that the human body would show a similar performance. If an RF-current is flowing along the vertical axis of the human body, it flows not only in the outer layers, but also in depths of some centimeters (TABLE 1). Thus, the current carrying area is relatively large, resulting in a small resistance  $R_{loss}$ . Due to the shape of the human body ('thick cylinder') the radiation resistance is relatively large compared.with the reactive impedance (general theory of thick monopoles and dipoles). This means that the human body is an efficient broad-band antenna for the frequency range 60 to 180 MHz, as proven with the phantom experiments.

Unfortunately, the practical feeding of the human body causes severe problems. Theoretical and experimental investigations by FISCHER and CASTEL-LI [23] revealed the impossibility of the RF-coupling by means of a toroid. This idea looks very promising at first sight: A toroid coil generates a circular H-field in the horizontal plane and is theoretically appropriated to induce an axial current in a body situated in the center of the toroid. However, in the regarded frequency range 60 to 200 MHz the circumference around the hip is larger than 0.12  $\lambda$ . Because of the helical construction of the toroid a phase difference of mimimum 90° occur because the signal velocity along the toroid is reduced by a factor of 2 to 5 (see Appendix 16.1.), so that the induced axial currents are not in phase. Several improved feeding methods were suggested , including segmental toroids and capacitive coupling, but up to now a practically applicable solution is missing (FISCHER and CASTELLI [23]).

The broad-band characteristics of the human body has its main significance for personal radio sets with inefficient, poorly matched antennas, as for instance wireless microphones in the frequency range 30 to 150 MHz. Below 100 MHz the design of tuned, efficient and body-mounted antennas for vertical polarization is very difficult. The main problems are the detuning effect of the body proximity and the efficiency at small  $h/\lambda$ ratio. If a short monopole or dipole cable antenna is in the proximity of the human body, the body acts like a lengthening of the antenna, as can be noticed with any short-wave or ultra-short-wave receiver. If the original antenna is relatively bad (general situation), the human body is a considerable improvement. Since many transmitters operated with such cable antennas (e.g. biotelemetry in the 37 MHz band), an investigation of the active radiation pattern of the human body and a detailed study on the coupling mechanism would be sensible. Principally, there are two ways for a theoretical analysis:

- Integration of the near-field data of section 10.4. along a multidimensional antenna. By designing a special antenna with matched amplitude- and phase- correlations along the selected polarization axes the desired coupling might be obtained.
- Computing of aperture radiation according to HARRINGTON and MAUTZ [40]. In this case the RF-energy has to be coupled into the body by means of surface electrodes or capacitive plates. This method seems to be risky with respect to biological effects and encumbrance, but perhaps an acceptable solution is possible.

#### 13.2.2. BODY-MOUNTED ANTENNA ARRAYS

The presented study dealt essentially with body-mounted antennas of very small dimensions, after having proved the field homogeneity around the test points. The data computed in section 10.3.5. may be also used for larger antennas, antenna arrays and antennas with more than one polarization axes. Such antennas would be of interest, if quasi-isotropic transmission is required. For such applications the computer programs are already prepared to compute similar data for changing incident angles  $\theta_i$ .

## 13.3. PROPOSALS FOR EFFICIENT, BODY-MOUNTED ANTENNAS

## 13.3.1. VERTICAL POLARIZED ANTENNAS

Most of the currently used antennas are vertical polarized. Constructional details can be found by GOUBAU and SCHWERING [32], KANDOIAN and SICHAK [47], LI and BEAM [54], NEUKOMM [62,63], OEHEN and BALZARINI [67], TONG [80] and WHEELER [85]. It is not possible to discuss these antennas here, but the main limitations should be mentioned:

- Transmission loss of the vertical polarized E-field component. As studied in this report, the transmission loss increases with increasing frequency and with decreasing antenna-body distance. Above 100 MHz and  $d_{at}$  below 0.1 m it is not possible to obtain a worst-case transmission loss of less than 10 dB. Usually the transmission loss is physically limited on 10 to 30 dB at higher frequencies and smaller  $d_{at}$ 's, and the largest loss occurs usually in the shadow zone.
- Bandwidth. If an electrically small antenna occupies only a small fraction of the radiansphere (WHEELER [85]), the bandwidth decreases with decreasing volume (at constant frequency), if no additional resistive elements are involved. If a body is near to the near-field zone of the antenna, severe detuning effects are most probable.
- Efficiency. Usually, the radiation resistance is small, decreasing with decreasing volume of the antenna (at constant frequency).Each additional loss (ohmic losses in the conductor, earth losses,matching losses, etc.) decreases the efficiency. At frequencies below 200 MHz an important loss is caused by the insufficient counterpoise of monopole antennas.

A comparison of some helical antennas is given in Appendix 16.1.2. A special group of non-resonant antennas are the cable antennas for frequencies below 100 MHz: in contrast to the resonant antenna they may perform better in proximity to the body as in free-space, but also here the overall loss, compared with an ideal dipole in free-space, is in the region of 5 to 20 dB.

#### 13.3.2. RADIAL POLARIZED ANTENNAS

These antennas are up to now little explored. The test transmitters of section 11.3. are not developed for optimal efficiency and are not intend-

ed for practical applications. FISCHER and CASTELLI [23] tried to design some radial polarized antennas for practical applications and performed some experiments with metallic and lossy body models (1/3 scale of MET and PHA, see also section 5.4.1.). In principle, these antennas are short helical monopoles with a top capacitor, mounted on a limited counterpoise and feed from a  $50 \Omega$  coaxial cable (FIGURE 99).

Top C-Helix mounted on Phantom



#### FIGURE 99

Radial polarized antenna (Top C-Helix) mounted on a Phantom.

Antenna dimensions:

- h : 34 mm (monopole length)
- D<sub>h</sub> : 12 mm (helical diameter)
- n : 1/2 to feeding point
  5 V2 to top C
  (number of turns)
- $D_c$ : 47 mm (top C diameter)
- L<sub>C</sub> : 100 x 100 mm (dimensions of the counterpoise)

Body dimensions: (1/3 scale PHA)

 $L_B$  : 0.6 m (length)

D<sub>B</sub>: 0.084 m (diameter)

(Source: FISCHER and CASTELLI [23])

The experimental data are of limited accuracy, because the measurements have been performed in a very small anechoic chamber:

Bandwidth on metallic cylinder (1/3 MET) : 311 to 319 MHz Bandwidth on model phantom (1/3) PHA) : 313 to 321 MHz Minimum/maximum azimuthal gain on 1/3 MET :  $-12 \, dB / -5 \, dB$ Minimum/maximum azimuthal gain on 1/3 PHA :  $-16 \, dB / -9 \, dB$ Bandwidth when mounted on a large c-poise : 309 to 317 MHz Gain at vertical polarization " " :  $-0.5 \, dB$ 

All gain data are related to the ideal, isotropic radiator (0 dB), and not to the FSL as usually used in this study. The comparison of the data

with those studied in section 12. lead to the following conclusions:

- The experiments of FISCHER and CASTELLI [23] are comparable to the experiments in section 12. performed at 101 MHz (1/3 of the model size = 1/3 of the nominal frequency). At this frequency the radial Gain<sub>B</sub> is minimum and amounts to -5 dB. The new measured data are about 7 to 11 dB lower in the worst case.
- The efficiency of the presented radial antenna could be improved by a larger counterpoise. The increase of the center frequency from 313 to 315 to 317 MHz from mounting on a large counterpoise to 1/3 MET to 1/3 PHA points to a too small counterpoise and efficiency loss.

The presented radial antenna is a promising solution, albeit the predicted high gain could not be achieved completely. However, comparing the radial antenna with a standard vertical polarized antenna of similar volume and mounting, the radial antenna prototype is already better in many respects:

- High absolute gain when mounted very close to the body, little variation of the field amplitude from 0 <  $\phi$  < 360<sup>0</sup>
- Human body well shielded from the near-fields of the antenna (Safety)
- Simple mounting, small volume and little encumbrance
- little or no detuning of the resonant frequency when counterpoise is sufficiently large
- Constant VSWR, direct matching to a 50  $\Omega$  coaxial line
- Little difference in gain when separated from the body (however, axis of the antenna must be vertical when separated from the body).

The physically given disadvantage of all electrically small antennas, the limited bandwidth, is of minor importance , because there are only very small detuning effects from the body. The reduced efficient (in context with the extremely small length/ $\lambda$  ratio) is balanced by a better matching but is still a subject for improvement.

The improvement of the radial antenna and the optimization of hB and  $d_{at}$  for frequencies between 50 and 200 MHz would be an interesting field of research. The literature on electrically small antennas and the experimental and theoretical data in this study may lead to practical body-mounted antenna with better performance and less biological risks.

# 13.4. THE OPTIMAL FREQUENCY RANGE FOR BODY-MOUNTED ANTENNAS

#### 13.4.1. CONCLUSIONS FROM THE OBTAINED DATA

As a result of this study the optimal frequency range for omni-directional transmission with body-mounted antennas can be defined as follows:

# OPTIMAL FREQUENCY RANGE : 35 TO 180 MHz

Some remarks are needed with respect to the practical use of personal radio sets. The antenna type, the antenna efficiency, the dimension of the radio set, the matching of the antenna to the RF-terminal and some other factors may lead to a slightly different choice. Thus, we may define four categories of optimal frequencies:

- 35 to 65 MHz : Strictly vertical polarized, well-matched, efficient, electrically small antennas (very theoretical, exist very seldom in practice).The transmission loss is minimum at 65 MHz and increases slightly with decreasing frequency. The lower the frequency, the more difficulties occur with bandwidth, detuning and efficiency.
- 50 to 200 MHz : Strictly radial polarized antennas. Good (experimentally proved) and excellent (predicted theoretically) performance can be obtained at small antenna-body distances in this range.
- 60 to 150 MHz : Technically realizable antennas with dominant vertical polarization but additional transverse (radial) polarization, such as helical monopoles on small transmitting and receiving devices, especially small walkie-talkies. The unexpectedly good performance in moderate proximity to the human body is mainly caused by a cross-talk of the radial component. Usually the transverse polarization was considered as an unavoidable lack of electrically small antennas. Above 150 MHz the design of more vertically polarized antennas becomes possible. As a consequence, the performance becomes better in free space, but worse in the proximity to the body, because both transverse polarization and radial component decreases with increasing frequency.
- 20 to 100 MHz : Cable antennas, inefficient, usually not matched antennas of less than  $\lambda/4$  length (wireless microphones, biotelemetry transmitters in the 37 MHz band). The human becomes a part of the antenna due to different effects with resulting better performance.

#### 13.4.2. FUTURE FREQUENCIES FOR BIOTELEMETRY

į.

As a member of the working group TC 62 of the IEC the author took part in the preparation of the IEC document 62(Secretariat) 38. This document recommends new biotelemetry frequencies to be proposed at the ITU world conference of frequency allocation. The official proposal concludes with:

- 8.1. A frequency band, ranging from 36.7 to 37.9 MHz, power 50 mW ERP, should be reserved and allocated for biotelemetry.
- 8.2. Two frequency bands between 70 to 200 MHz, each of 1 MHz width and with a power of 50 mW ERP, should be reserved and allocated for biotelemetry.

The IEC document explains the special needs of biotelemetry, especially the urgently requested international standardization of frequencies which allow omnidirectional transmission with body-mounted antennas. It points to the potential risks of frequencies above 300 MHz at power densities above 10  $\mu$ W/cm<sup>2</sup> and recommends therefore lower frequencies with only 50 mW effective radiated power (ERP). If the IEC recommendations became accepted at the ITU conference, at least 110 small band channels (37 MHz) 40 medium band channels (70 - 200 MHz) and 9 broad band channels (37, 70-200 MHz) would be internationally usable during the next 20 years.

#### 14. SUMMARY

The influence of the human body on the radiation pattern of body-mounted antennas has been investigated in the frequency range 10 to 1000 MHz (below, up to above main resonances of the human body). An analytically formulated, computable antenna-body-model has been developed which explains the correlations between the electrical field (amplitude and phase) and antenna location (antenna-body distance dat and azimuthal rotation angle  $\phi$ ) at frequencies above 200 MHz (above resonance region) at vertical polarization (E-field parallel to the largest axis of the body). With the method of moments a computational antenna-body model has been investigated which explains the correlations among electrical field, antenna location  $(d_{at}, \phi, and relative antenna height h_B)$  and irradiation angle  $\theta_i$ , at all polarization axes and frequencies, especially within the resonance region. Experimental data with human test subjects and body models have been collected with special measuring antennas and field generators at frequencies between 25 to 900 MHz, whereby  $d_{at}$ ,  $\phi$  and  $h_B$  have been varied continuously (or in small steps) within large limits. The agreement between theoretical and experimental data amounted to 3 dB, except at extreme conditions (measuring range: -20 to +5 dB). The main conclusions from the complete study are: 1. There is a mathematical correlation between transmission loss (from a body-mounted to a remote antenna) and frequency, antenna location, body geometry, and polarization. 2. Within 50 to 200 MHz (just below first, up to just above second resonance) the human body (and other, in material and gecemetry comparable bodies) acts like an efficient polarization transformer. At dat below 0.3m a radial polarized , small antenna allows omnidirectional (i.e. little depending on  $\phi$ ) transmission, where the transmission loss decreases with decreasing dat. Even a gain compared with an ideal isotropic radiator in free space can be achieved. 3. The antenna-body distance dat is the parameter of greatest influence in antennas close to a body. Especially at frequencies above 200 MHz and vertical polarization the dat determines the different reflections of the E-field at the body and determines the azimuthal radiation pattern of the antenna-body system. 4. The shape of a (lossy) body is much more important than the (biological) material. 5. The reciprocity theorem is applicable on body-mounted antennas, as experimentally proven at low power,

at frequencies between 65 and 900 MHz, and at  $d_{at}$  ranging from 0.05 - 4m. 6. The optimal frequency range for omnidirectional transmission with body-mounted, electrically small antennas is between 35 to 180 MHz, as resulting from both theoretical and experimental data.

The order of magnitude of electric and magnetic field strengths of the near-fields of electrically small antennas have been approximatively determined and compared with the maximum permissible limits of international safety standards. At higher transmitter power (above approximately 100 mW, very much depending on antenna type and location) these limits may be exceeded, especially by walkie-talkies with vertical polarized helical antennas. Based on the recent results on the biological significance of non-ionizing radiation, RF-induced biological effects are possible, above all at frequencies above the first resonance. These effects may lead to artifacts (Biotelemetry transmitter) and perhaps to health hazards (transmitters of walkie-talkies). Comparing a standard antenna (e.g. vertical polarized helical antenna) with a radial antenna at the same extreme antenna-body distance (below 0.1 m) and at the same input power, the radial antenna does not only decrease the risk, but also offers a smaller transmission loss.

#### 14.2. ZUSAMMENFASSUNG

Der Einfluss des menschlichen Körpers auf das Strahlungsmuster von körpernahen Antennen wurde im Frequenzbereich 10 bis 1000 MHz (unterhalb bis oberhalb der Hauptresonanzen des menschlichen Körpers) untersucht. Es wurde ein analytisch beschreibbares, berechenbares Antennen-Körper-Modell entwickelt, das die Zusammenhänge zwischen elektrischem Feld (Amplitude und Phase) und Antennenposition (Antennen-Körperabstand dat und azimutalem Drehwinkel  $\phi$ ) bei Frequenzen oberhalb der Hauptresonanz ab 200 MHz bei vertikaler Polarisation (E-Feld parallel zu Körperlängsachse) erklärt. Mit der 'Method of Moments' wurde ein numerischberechenbares Antennen-Körper-Modell untersucht, das die wichtigen Zusammenhänge zwischen elektrischem Feld, Antennenposition  $(d_{at}, \phi$  und relativer Antennenhöhe  $h_B$ ) und Einstrahlungswinkel  $\theta_i$  bei allen Polarisationsrichtungen und Frequenzen,insbesondere im Resonanzbereich, erklärt. Experimentelle Daten mit Versuchspersonen und Körpermodellen wurden im Frequenzbereich 25 bis 900 MHz mit Hilfe von speziellentwickelten Messantennen und Feldgeneratoren gesammelt, wobei  $d_{at}$ ,  $\phi$  und  $h_{B}$  kontinuierlich (oder in feinen Abstufungen) in weiten Grenzen variiert wurden. Mit Ausnahme bei extremen Bedingungen wurde eine Uebereinstimmung von 3 dB (Messbereich -20 bis +5 dB) zwischen den experimentellen und berechneten Daten erzielt. Die wichtigsten Folgerungen aus der gesamten Untersuchung lauten: 1. Es besteht ein mathematischer Zusammenhang zwischen Uebertragungsverlust (von körpernaher Antenne zu entfernter Antenne), der Frequenz, der Antennenposition, der Körpergeometrie und den Polarisationsrichtungen. 2. Innerhalb 50 bis 200 MHz (knapp unterhalb erster bis knapp oberhalb zweiter Resonanz) wirkt der menschliche Körper (und andere, geometrisch und materiell vergleichbare Körper) als effizienter Polarisations-Transformator: Bei dat unterhalb 0.3 m erlaubt eine radial polarisierte (d.h. E-Feld radial zur Körper-Längsachse) kleine Antenne eine omnidirektionale (d.h. wenig von  $\phi$  abhängige) Uebertragung, wobei der Uebertragungsverlust mit abnehmendem dat abnimmt und sogar ein Gewinn gegenüber einer idealen, isotropen Antenne im freien Raum möglich ist. 3. Der Antennen-Körperabstand dat ist der wichtigste Parameter bei körpernahen Antennen. Speziell bei Frequenzen oberhalb 200 MHz und vertikaler Polarisation bestimmt dat die unterschiedlichen Reflexionen des E-Feldes am Körper und damit die Azimutal-Strahlungsdiagramme des Antennen-Körper-Systems. 4. Die Form des (verlustbehafteten) Körpers ist von weitaus grösserer Bedeutung als das (biologische) Körpermaterial.

5. Experimente im Frequenzbereich 65 bis 900 MHz bei kleinen Leistungen (1 - 10 mW) und d<sub>at</sub> von 0.05 bis 4m haben gezeigt, dass das Reziprozitätsgesetz auch für elektrisch kleine, körpernahe Antennen gilt. 6. Aus Theorie und Experiment geht hervor, dass der optimale Frequenzbereich für omnidirektionale Uebertragung mit kleinen, körpernahen Antennen zwischen 35 und 180 MHz liegt.

Die Grössenordnung der elektrischen und magnetischen Feldstärken im Nahfeld elektrisch kleiner Antennen wurde abgeschätzt und mit den entsprechenden zulässigen Grenzwerten internationaler Sicherheitsvorschriften verglichen. Bei höherer Sendeleistung (ab ca. 100 mW, Grenze sehr stark von Antennentyp und Position zu Körper abhängig) werden diese Grenzwerte überschritten, besonders von Funksprechgeräten mit vertikal polarisierten Helix-Antennen. Auf Grund der neuesten Erkenntnisse sind, vor allem bei Frequenzen oberhalb der ersten Resonanz, HF-induzierte biologische Effekte möglich, die zu Artefakten (Biotelemetrie-Sender) und Gesundheitsschäden (Funksprech-Sender) führen können. Vergleicht man eine Standardantenne (z.B. vertikal polarisierte Helix-Antenne) mit einer Radial-Antenne bei extrem kleinen Antennen-Körperabständen (dat kleiner als 0.1 m) so kann mit der Radial-Antenne bei gleicher Eingangsleistung nicht nur das Sicherheitsrisiko, sondern auch der Uebertragungsverlust verringert werden.

- [1] ANDREASEN M.G. Scattering from Bodies of Revolution. IEEE Trans., Vol. AP-15, No.2, March 1965, pp. 303-310.
- [2] ANSI American National Standards Institute: Safety Level of Electromagnetic Radiation with Respect to Personnel. ANSI C95.1-1974, IEEE, New York, 1974.
- [3] ARRL The ARRL Antenna Book. American Radio Relay League, Newington, Conn., 1974.
- [4] BACH S.A. Effects of Radiofrequency Energy on Human Gamma
   LUZZIO A.J. Globulin. In: Biological Effects of Microwave Ra-BROWNELL A.S. diation, (Ed. M.F. Peyton), Plenum, New York, 1961.
- [5] BAGGENSTOS H. Vorlesung Elektromagnetische Felder, ETH 1977, Kapitel 4: Rasch veränderliche Felder. Electr. Dept., Swiss Federal Institute of Technology, Zurich, 1977.
- [6] BARANSKI S. Biological Effects of Microwaves. Dowden, Hutchinson CZERSKI P. & Ross, Inc., Stroudsburg, Pa., 1976.
- [7] BECKER K.D. Ausbreitung Elektromagnetischer Wellen. Springer Verlag, Berlin, 1974.
- [8] BELDING H.A. Index for Evaluating Heat Stress in Terms of Resulting Physiological Strains. Heating, Piping, Air Conditioning, Aug. 1955, pp.129-136.
- [9] BELSHER D.R. Development of Near-Field Electric Energy Density Meter Model EDM-2. HEW Publ. No.(NIOSH) 75-140, 1975.
- [10] BEVENSEE R.M. The Syracuse Computer Code for Radiation and Scattering from Bodies of Revolution, Extended for Near-Field Computations. Report UCRL-51622, TID-4500, UC-32, Lawrence Livermore Laboratory, 1974.
- [11] BRONSTEIN I. Taschenbuch der Mathematik, Verlag Harri Deutsch, SEMENDJAJEW K. Zürich, 1975.
- [12] BUCHANAN H. MOORE W.F. RICHTER C.R. Human Body Effect on Signal Patterns of Personal Telemetry Transmitters. Report SAM-TR-70-4, USAF School of Aerospace Medicine, Aerospace Medical Division (AFSC), Brooks Air Force Base, Texas, Jan. 1970.
- [13] CHANG D.C. The Electromagnetic Field Very Near to a Monopole. HALBGEWACHS R.D. IEEE Trans., Vol. EMC-17, No.2, May 1975,pp. 97-105. HARRISON C.W.
- [14] CHEN K.M. Internal EM Field and Absorbed Power Density in GURU B.S. Human Torsos Induced by 1-500-MHz EM Waves. IEEE Trans., Vol. MTT-25,No. 9,Sept. 1977, pp. 746-756.
- [15] CHEN K.M. Induced EM Fields Inside Human Bodies Irradiated GURU B.S. by EM waves of up to 500 MHz. Internal Report by the Department of El.Eng. and Syst. Sci., Michigan State Univ., 1977.

[16] CHOU C.K. The Effect of the Electromagnetic Fields on the Nervous System. Scientific Report No.6, Aug. 1975, GUY A.W. Bioelectromagnetic Res. Lab., Univ. of Washington School of Medicine, Seattle, Wash.98195. Microwave Ovens: We can tell you how well they per-[171 CONSUMER formed - but no one really knows how safe they are. REPORTS Consumer Reports, (USA), June 1976, pp. 314-321. Arbeitshygiene: Elektrische, Magnetische und Elek-[18] DDR-STANDARD tromagnetische Felder und Wellen. TGL 32602/01. Staatsverlag DDR, Berlin, 1975. Zusätze: ASAO 5: MfGe Nr.1/1977, Gesetzesblatt 109, 17. Marz 1978, Teil 1. Nr.8. Trends in Nonionizing Electromagnetic Radiation [19] DODGE C.H. Bioeffects. Research and Related Occupational GLASER Z.R. Health Aspects. J. of Microwave Power, Vol. 12(4), Dec. 1977, pp. 319-334. The Biologic Action and Hygienic Significance of [20] DUMANSKIJ J.D. Electromagnetic Fields of Superhigh and Ultrahigh SANDALA M.G. Frequencies in Densely Populated Areas. In: ref. [69], pp. 289-293. [21] EGGERT S. Results at the Development of One Near-Field Strength Meter at the Measurement of the Electrical GOLTZ S. Component of the Electromagnetic RF Field in the KUPFER J. GDR. Proc. of the 1977 USNC/URSI Meeting, Arlie, See also: Nahfeldstärke-Messgerät Va. (in press). NFM-1, Zentralinst.f.Arbeitsmed.der DDR,Berlin,1976. The Numerical Thermal Simulation of the Human Body [22] EMERY A.F. SHORT R.E. When Absorbing Non-ionizing Microwave Irradiation -With Emphasis on the Effect of Different Sweat Mo-GUY A.W. dels. In: Biological Effects of Electromagnetic KRANING K.K. Waves, Vol.II, Selected Papers of the 1975-USNC/ LIN J.C. URSI Annual Meeting, Boulder, Colo.1976, pp. 96-118. [23] FISCHER T. Aktivierung des menschlichen Körpers als Sendeantenne. Study work at the Electronics Inst. and Bio-CASTELLI J. mechanics Lab., ETH Zurich, 1979. [24] GANDHI O.P. Some Recent Results on Deposition of Electromagnetic Energy in Animals and in Models of Man. HAGMANN M.J. In: Abstracts of Scientific Papers, International D'ANDREA J.A. USNC/URSI Symposium on the Biological Effects of Electromagnetic Waves, Arlie, Va., 1977. Distribution of Electromagnetic Energy Deposition [25] GANDHI O.P. in Models of Man with Frequencies Near Resonance. SEDIGH K. In: Biological Effects of Electromagnetic Waves, BECK G.S. selected papers of the 1975-USNC/URSI Annual Meet-HUNT E.L. ing, Boulder, Colo., (HEW publ. (FDA) 77-8011), pp.44-67. Deposition of Electromagnetic Energy in Animals and [26] GANDHI O.P. in Models of Man With and Without Grounding and Re-HUNT E.L. flector Effects. Radio Science Vol. 12, No. 6 (S), D'ANDREA J.A. 1977, pp. 39-47.

- [27] GANDHI O.P. Recent Results on Deposition of Electromagnetic Energy in Animals and in Models of Man. Lecture presented at the ETH, Aug.17, 1978.
- [28] GLASER Z.R. Biomedical Aspects of Radio Frequency and Microwave Radiation: A Review of Selected Soviet, East European, and Western References. In: Biological Effects of Electromagnetic Waves. Selected papers of the 1975-USNC/URSI meeting, Boulder,Colo.Vol.1,Dec.1976 (HEW publ. (FDA) 77-8011), pp. 2-34.
- [29] GLASER Z.R. BROWN P.F. ALLAMONG J.M. NEWTON R.C. BROWN R.C. Bibliography of Reported Biological Phenomena ('Effects') and Clinical Manifestations Attributed to Microwave and Radio-Frequency Radiation. Ninth Supplement, Nov.1977. DHEW (NIOSH) Publ. No. 78-126. First bibliog.: Res.Rep.No.2, NMRI,Bethesda,Md.1971. Available from NTIS,Springfield,Va. 22151.
- [30] GORDON Z.V. Main Directions and Results of Research Conducted ROSCIN A.V. in the USSR on the Biologic Effects of Microwaves. BYCKOV M.S. In: ref. [69], pp. 22-35.
- [31] GORDON Z.V. Biological Effect of Microwaves in Occupational Hygiene. Izdatel'stvo Meditsina, Leningrad, 1966. US-Translation TT 70-50087,NASA TT F-633, 1970.
- [32] GOUBAU G. SCHWERING F. Proceedings of the ECOM-ARO Workshop on Electrically Small Antennas. U.S. Army Electronics Command, Fort Monmouth, May 1976.
- [33] GREEN F.M. Development of Magnetic Near-Field Probes. HEW Publ. No. (NIOSH) 75-127, 1975.
- [34] GREEN F.M. Development of an RF Near-Field Exposure Synthesizer (10 to 40 MHz). Interagency Agreement NIOSH-IA-75-16. Available from NIOSH, Cincinnati,Ohio 45226.
- [35] GUY A.W. Quantitation of Induced Electromagnetic Field Patterns in Tissue and Associated Biological Effects. In: ref.[69], pp. 203-217.
- [36] GUY A.W. Determination of Power Absorption in Man Exposed to
   WEBB M.D. High Frequency Electromagnetic Fields by Thermo SORENSEN C.C. graphic Measurements on Scale Models. IEEE Trans.,
   Vol. BME-23, No.5, Sept. 1976, pp.361-371.
- [37] GUY A.W. Microwave-Induced Acoustic Effects in Mammalian CHOU C.K. Auditory Systems and Physical Materials. Ann.of the LIN J.C. N.Y. Acad. of Sciences, Vol. 242, Febr. 1975, CHRISTENSEN D. pp. 194-218.
- [38] GUY A.W. Phantom Models for Muscle, Brain, Fat, and Bone at MW Frequencies, Private communication, 1976.

[39] HANKIN N.N. TELL R.A. ATHEY T.W. JANES D.E.
High Power Radiofrequency and Microwave Radiation Sources: A Study of Relative Environmental Significance. Operational Health Physics, Proc. Ninth Midyear Topical Symp. of the Health Physics Society. (Compilated by P.L. Carson et al.), Febr. 1976.

[40] HARRINGTON R.F. Radiation and Scattering from Bodies of Revolution. MAUTZ J.R. Rep. AFCRL-69-0305, 1969, and : Generalized Network Parameters for Bodies of Revolution. Rep. TR-68-7. 1968, Syracuse Univ., El. Eng.Dept. Syracuse, N.Y. [41] HARRINGTON R.F. Field Computation by Moment Methods, Macmillian Co., N.Y.. 1968 (Second printing: write to the author). 1421 HEILMANN A. Antennen I. BI Hochschultaschenbücher 140/140a, Mannheim, 1970. [43] HELLER J.H. The Effect of Electromagnetic Fields on Unicellar Organisms. In: IRE, AIEE, ISA Conf.Electrical Technology in Medicine and Biology, Vol.7, 1959. [44] HIRSCH F.G. Bilateral Lenticular Opacities Occurringin a Technician Operating a Microwave Generator. AMA Arch. PARKER J.T. Ind.Hyg.Occup. Med.6:1952, pp. 512-517. [45] JOHNSON C.C. Nonionizing Electromagnetic Wave Effects in Bio-GUY A.W. logical Materials and Systems. Proc. IEEE, Vol.60, No.6, 1972, pp. 692-718. [46] KALADA T.V. Biologic Effects of Radiation in the 30-300 MHz FUKOLOVA P.P. Range. In: ref.[69], pp. 52-57. GONCAROVA N.N. [47] KANDOIAN A.G. Wide-Frequency-Range Tuned Helical Antennas and SICHAK W. Circuits. IRE Nat. Conv. Rec. Part 2, Antennas and Components, 1953, pp. 42-47. [48] KING H.E. Characteristics of Body-Mounted Antennas for Personal Radio Sets. IEEE Trans., Vol. AP-23, 1975, pp. 242-244. [49] KING H.E. Effects of a Human Body on a Dipole Antenna at 450 WONG J.L. and 900 MHz. IEEE Trans., Vol.AP-25, 1977, pp. 376-379. The Scattering and Diffraction of Waves. Harvard [50] KING R.W.P. WU T.T. Univ. Press, Cambridge, Mass., 1959, Chapter 2. [51] KRAUS J.D. Antennas. McGraw Hill, 1950, Chapter 9. [52] KRITIKOS H.N. The Distribution of Heating Potential Inside Lossy SCHWAN H.P. Spheres., IEEE Trans., Vol.BME-22, No.6, Nov. 1975, pp. 457-463. The Effect of the Human Body on the Radiation Pro-[53] KRUPKA Z. perties of Small-Sized Communication Systems. IEEE Trans., Vol. AP-16, 1968, pp. 154-163. [54] LI T. Helical Folded Dipoles and Unipoles. Proc. Nat. BEAM R.E. Electr. 13: 1957, pp. 89-105. Conditioned Reflex Activity in Dogs under Local In-[55] LIVESHITS N.N. fluence of a VHF Field upon Certain Zones of the Cerebral Cortex. Biophys. J., Vol.2,1957, p. 198. [56] LIU L.M. The Relation of Teratogenesis in Tenebrio Molitor to ROSENBAUM H.J. the Incidence of Low-Level Microwaves. IEEE Trans., Vol. MTT-23, Nov. 1975, pp. 929-931. PICKARD W.F.

- [57] MICHAELSON S.M. Effect of Exposure to Microwaves: Problems and Perspectives. Environmental Health Persp. Vol.8, 1974, pp. 133-156.
- [58] MICHAELSON S.M. The Tri-Service Program A Tribute to George M. Knauf, USAF (MC). IEEE Trans., Vol. MTT-19, No. 2, Febr. 1971, pp. 131-146.
- [59] MILROY W.C. Microwave Cataractogenesis: A Critical Review of MICHAELSON S.M. the Literature. Aerospace Medicine, Vol. 43, No.1, 1972, pp. 67-75.
- [60] MOOR F.B. Microwave Diathermy. In: Therapeutic Heat and Cold, (Ed.:Licht S.), New Haven, Conn, 1965, Sec. 12, pp. 310-320.
- [61] MUTH E. Appearance of Pearl-Chain Formation of Particles in Emulsions Caused by Alternating Fields. Kolloid-Z., Vol. 41, 1927, p. 97.
- [62] NEUKOMM P.A. Body-Mounted Transmitting Antennas: Radiation Patterns and Design of Helical Dipole Antennas. In : Biotelemetry III (Eds.:Fryer T.B., H.A. Miller and H.Sandler), Acad, Press, Inc,N.Y., 1976,pp. 345-348.
- [63] NEUKOMM P.A. Biotelemetry Antennas: The Problem of Small Body-Mounted Antennas. Proc.int.conf. BIOSIGMA 78, Vol. 2, Paris 1978, pp. 99-106.
- [64] NEUKOMM P.A. Artifacts from RF and MW Telemetry: Estimation of Safety Aspects and Review on Biological Effects.In: Biotelemetry IV (Eds.: Klewe H.J. and H.P.Kimmich), Döring-Druck,Druckerei und Verlag,Braunschweig,1979.
- [65] NEUKOMM P.A. The Rubber Band Goniometry. J.Biotelemetry 1, 1974, pp. 12-20.
- [66] NYQUIST D.P. Coupling Between Small Thin-Wire Antennas and a CHEN K.M. Biological Body. IEEE Trans., Vol.AP-25, No.6, Nov. GURU B.S. 1977, pp. 863-866.
- [67] OEHEN W. BALZARINI N. Helix Biotelemetrie Antennen. Study work at the Lab. Microwave Tech. and Biomechanics Lab., ETH Zurich, 1975.
- [68] OLSEN R.G. Microwave-Induced Chronotropic Effects in the Iso-LORDS J.L. lated Rat Heart. Ann. of Biomed. Engineering 5, DURNEY C.H. 1977, pp. 396-409.
- [69] PROCEEDINGS WARSAW
   Biologic Effects and Health Hazards of Microwave Radiation. Proc. Int.Symposium Warsaw, 15-18 Oct., 1973, Polish Medical Publishers, Warsaw, 1974.
- [70] ROMERO-SIERRA C. Effect of an Electromagnetic Field on the Sciatic HALTER S. TANNER J.A. Nerve of the Rat. In: The Nervous System and Electric Currents,(Ed.:Wulfsohn N.L.),Vol.2, Plenum, N.Y., 1971, pp. 81-85.

### [71] ROTHAMMEL K. Antennenbuch. Telekosmos-Verlag, Franckh'sche Verlagshandlung, Stuttgart, 1976.

1

[72]	SCHWAN H.P.	Microwave Radiation: Biophysical Considerations and Standards Criteria. IEEE Trans.,Vol.BME-19, No.4, July 1972, pp. 304-312.
[73]	SCHWAN H.P.	Biophysics of Diathermy. In: Therapeutic Heat and Cold, (Ed.: Licht S.),New Haven, Conn.,1965, Sec.3, pp. 63-125.
[74]	SCHWAN H.P. PIERSOL G.M.	The Absorption of Electromagnetic Energy in Body Tissues. Int. Review of Physical Medicine and Re- habilitation, Vol. 33, 1954, pp. 371-404.
[75]	SCHWAN H.P. LI.K.	Capacity and Conductivity of Body Tissues at Ultra- high Frequencies. Proc. IRE, Vol. 41, Dec. 1953, pp. 1735-1740.
[76]	TELL R.A.	Microwave Energy Absorption in Tissue. Reportby the Environ. Protection Agency, Twinbrook Res. Lab., 12709 Twinbrook Parkway,Rockville,Md.20852.Feb.1972.
[77]	TELL R.A.	An Analysis of Radiofrequency and Microwave Absorp- tion Data with Consideration of Thermal Safety Stan- dards. (Draft). U.S. Environ. Protection Agency,Las Vegas, Nev. 89114, P.O. Box 15027, Dec. 1977.
[78]	TELL R.A. O'BRIEN P.J.	Radiation Intensities Due to Mobile Communication Systems. U.S. Environ. Protection Agency, Office of Radiation Program, Silver Spring, Md 20910, 1976.
[79]	TOLER J. SEALS J.	RF Dielectric Properties Measurement System: Human and Animal Data. DHEW (NIOSH) Publ. No. 77-176,1977.
[80]	TONG D.A.	The Normal Mode Helical Aerial. Radio Communication, July 1974, pp. 432-437.
[81]	VAN BLADEL J.	Electromagnetic Fields. McGraw-Hill, 1964, Chapt.12.
[82]	VREELAND R.W. SHEPHERD M.D. HUTCHINSON J.C.	The Effects of FM and TV Broadcast Stations upon Cardiac Pacemakers. IEEE Symp. Record (Publ. No. IEEE 74CH0803-7 EMC), 1974.
[83]	WHEELER H.A.	Fundamental Limitations of Small Antennas. Proc. IRE, Dec. 1947, pp. 1479-1484.
[84]	WHEELER H.A.	The Radiansphere Around a Small Antenna. Proc. IRE, Aug. 1959, pp. 1325-1331.
[85]	WHEELER H.A.	Small Antennas. IEEE Trans., Vol.AP-23, No.4, July 1975, pp. 462-469.

# 16. APPENDIX

<u>NTENTS</u>	page
.1. Helical Antennas	237
16.1.1. Properties, Design, Efficiency Measurement and Matching	237
16.1.2. Comparison of some Antenna Types	243
.2. Computer Programs and Additional Results	245
16.2.1. Program PANA and IZYL Results	245
16.2.2. Program HARRA and Output Sample	253
16.2.3. Program PANB and Output Sample	262
16.2.4. Program PANC, Data Cards for Test Bodies FZYL, MANMO	1 DC
and MANMOD 2, Field Homogeneity Results with FZYL	281
16.2.5. Additional Results from Field Computations with FZYL	- 3
MANMOD 1 and MANMOD 2	280

# Leer - Vide - Empty

# 16.1.1. PROPERTIES, DESIGN, EFFICIENCY MEASUREMENT AND MATCHING

e,1

A helical antenna is a typical electrically small antenna. Its maximum dimension is a small fraction of the wavelength. The "normal mode" helical antenna (FIGURE 106) consists of a helical conductor in the shape of a long cylinder with the diameter  $D_h$  ( $D_h << \lambda$ ) and with the axial (monopole) length h ( $h < \lambda/4$ ). The polarization of the radiated E-field is elliptical, with a dominant axis parallel to the helical cylinder axis (see FIGURE 106, bottom). The radiation pattern is very similar to that of an ordinary whip antenna. i.e. maximum radiation radial to the helical axis.

#### Helical Normal Mode Antenna







Helical Normal Mode Antenna Vertical E<sub>V</sub> far-field generated from the

FIGURE 106

vertical flowing antenna currents, Horizontal E<sub>h</sub> far-field generated from the horizontal flowing antenna currents.

p = pitch

The helix is a slow-wave structure. When used as a waveguide, the axial phase velocity of the wave guided by the helix is less than the velocity of light in free space. Accordingly, the resonant length of a helix is

shorter than the corresponding resonant length of a linear wire antenna. Thus, one may reduce the axial length h by a factor of 3 to 8 without adding external tuning elements. Compared with the short (non-resonant) whip antenna one obtains a better current distribution (see FIGURE 107), resulting in a higher radiation resistance:



FIGURE 106 Current distribution on the helical antenna z: vertical antenna axis I: current in the conductor

The radiation resistance of a half-wave helical dipole can be evaluated by integrating the far-field Poynting vector over a large spherical surface. For a thin half-wave helical dipole of length 2 h, an approximate expression for the radiation resistance is : (LI and BEAM [54])

$$R_{rad}$$
 Helix = 1280 (h/ $\lambda$ )<sup>2</sup> [Ohm] (246)

As a comparison, the radiation resistance of a small linear dipole of the length 2h and with linear (triangular) current distribution is

$$R_{rad}$$
 Whip = 790 (h/ $\lambda$ )<sup>2</sup> [Ohm] (247)

Hence the radiation resistance of a small-diameter half-wave helical dipole is approximately 62 percent greater than that of a small linear dipole of the same length. If the antennas are operated above a perfect ground, the radiation resistances are reduced by a factor of 2 and one obtains the values of KANDOIAN and SICHAK [47]

 $R_{rad}$  Helix above ground =  $(25.3 h/\lambda)^2$  [Ohm] (248)

 $R_{rad}$  Whip above ground =  $(20 h/\lambda)^2$  [Ohm] (249)

Principally, there are four methods to design a helical antenna:

 Computation with the method of moments: OEHEN and BALZARINI [67] adapted an existing antenna modelling program (BURKE and SELDEN, Microfiches AD - 767 420, 1973) to the helical dipole problem. This method is very accurate, if the number of subsegments is large enough ( $\sim$  200) and offers many results : impedance, gain, bandwidth, effect of near-by conducting surfaces, etc. The ROUND HELICAL DIPOLE (RHD) in FIGURE 107 has been computed with this method, and the experimental data (16.1.2.) agree quite well with the computed data. However, this method is very expensive and should only be applied for antenna optimization.

- Analytical approach. LI and BEAM [54] investigated the characteristic equation for helical waveguide and presented the results in nomograms. This method offers an insight in the complicated correlations between antenna geometry, bandwidth and general performance.
- 3. Approximative computation. KANDOIAN and SICHAK [47] evaluated approximative computational methods, which were adapted by TONG [80] for computation on pocket calculators. For a given antenna length (h), diameter ( $D_h$ ) and wavelength ( $\lambda$ ) the number of turns ( $N_h$ ) respectively the number  $n_h=N_h/h$  can be approximatively computed for long helices:

$$\log n_h = 0.4 \left( \log(\frac{\lambda}{h} - 4) + \log(\frac{\lambda}{h} + 4) + 0.5 \log \lambda - 3 \log D_h \right) - 1$$
 (250)

This method offers data for design with an accuracy of about 20 %, if  $h/\lambda$  is not smaller than 1/10 and if  $D_h$  is smaller than 0.3 h.

4. Experimental approach. If a network analyser and a small anechoic chamber is available, a well performing helical antenna can be designed as follows: For a monopole helical antenna a wire of a length of  $\lambda/2$  is wrapped in a shape of a helix with the desired  $D_h$  and h. The 'hot end' is contacted on a large counterpoise, and the feeding coaxial cable (inner conductor) is contacted at the m<sup>th</sup> turn (m $\cong$ N/10) from the now grounded 'hot end'. The Smith Chart (see FIGURE 109) shows the resonance frequency, the bandwidth and the input impedance (transformed  $R_{rad} + R_{loss}$  at  $f_{res}$ ). Varying the feeding point (changing m) one obtains a match to 50 Ohm (with a resulting relative bandwidth) and by cutting the upper antenna ends one obtains the wanted resonance frequency, because the initial  $f_{res}$  is usually ~ 30% too small.

The problems of all helical antennas are : 1.) small bandwidth , 2.) reduced efficiency, 3.) low radiation resistance,4.) sensitivity to detuning effects from proximity to obstacles, 5.) transverse polarization. Because the helical is an electrically small antenna the fundamental laws



2һ

**NNNN** 

# HELICAL DIPOLE ANTENNA RHD

Specifications:

GEOMETRY :			
LENGTH	2h	:	22 см
DIAMETER	D	:	2 см
WIRE DIAMETER	D,	;	2 MM
NUMBER OF TURNS	2N <sub>h</sub>	:	18
COMPUTER DATA :			
EFFICIENCY	Eff	:	94 %
GAIN VERT, POLARIZATION	GVER	T١	+1.4 pB
GAIN HOR. POLARIZATION	GHOR	1	– 17 pB
RADIATION RESISTANCE	R <sub>R</sub>	:	9.2 Онм
NETWORK ANALYSER DATA :			
RESONANT FREQUENCY	FRES	;	236 MHz
BANDWIDTH (-3 dB)	Bw	:	17 MHz
INPUT RESISTANCE	RI	:	24 Онм



LAMBDA / 4

50 OHM COAXIAL



FIGURE 107 (above) FIGURE 108 (left)

Round helical dipole (RHD)

mechanical details, computer predictions (antenna modelling program,see [67]) network analyser data and (left) performance of the antenna mounted on the human test subject (0 dB : isotropic radiator)



FIGURE 109 Smith Chart of the round helical dipole antenna RHD. Solid line : antenna in free space (anechoic chamber), dashed line : antenna mounted on human test subject at  $d_{at} = 6$  cm . Due to the losses effected by the body the resistance increases with decreasing  $d_{at}$  and the resonant frequency lowers with decreasing  $d_{at}$ .

found by WHEELER [83,85] must be considered. The bandwidth is determined by the  $h/\lambda$  ratio (see equations 9 to 13 in section 4.5) but can be controlled within small limits with matching (good match = smaller bandwidth and perhaps a better efficiency) and increasing the radiation resistance (higher radiation resistance = smaller bandwidth). The efficiency is mainly determined by ground losses, by losses in the matching network and, in complicated helical structures (double helix, etc.), by resistive losses in the antenna conductor. An example may illustrate the importance of the ground losses:

The standard helical monopole antenna of 173 MHz walkie-talkie (MOTOROLA HT 220) is specified as:

ht	=	114 mm	(total length)
h	=	106 mm	(length of the helix)
Nh	=	42	(total number of turns)
Dw	=	1.2 mm	(wire diameter)
Dh	=	9.6-6.7 mm	(tapered helical diameter)

The computation according to equation 250 and 246 resultin an N<sub>h</sub> of  $\sim$ 40 and a radiation resistance of 5.5 Ohm. The housing of the walkie talkie is maximum 180 mm  $\doteq \lambda/10$ . The VSWR of the complete antennatransmitter system is close to 1:1, so that the ground losses amount to about 44.5 $\Omega$ . The efficiency in radiation is thus 5.5 / 44.5+5.5 = 11 %, so that the complete system radiates about -9.5 dB less than an ideal dipole. A helical dipole according to FIGURE 107 has an efficiency of about 94 (theoretical) and 89 % (measured) and a gain of - 0.75 dB (theoretical ) and - 2 dB (measured) compared with an ideal full-length dipole.

The low radiation resistance is a potential source of bad efficiency, if the match to the feeding line is poor , if there are losses in the antenna conductor or in the matching network and if (in the case of monopole antennas) the counterpoise is not large enough. According to LI and BEAM [54] special helical antennas with multi-conductors were designed. As can be seen in the comparison FIGURE 110, 16.1.2., a higher gain can be obtained, but paying the price of a very small bandwidth.Thus, a maximum radiation resistance or a perfect matching is not very sensible with respect to detuning sensitivity: the more the antenna is "improved" for free-space operation, the more delicate it responds to external influencies. One further problem could be the transverse polarization. If the height is about 0.9 times the diameter, the antenna becomes circular polarized. However, as discussed in section 13.1.4., a transverse polarization at frequencies between 50 to 200 MHz may be even an advantage at small antenna-body distances, also when the radiation is reduced at

- 242 -

axial polarization. The transverse polarization could be computed by the formulas indicated in FIGURE 106,[54] and [74], but the agreement with the experimental data is so poor, that one should trust only the actually measured data. The same situation happens with the actual bandwidth: it is better not to cite the formulas here. If the bandwidth becomes really very important, one should use the accurate method of moments or one should perform representative experiments.

An important point is the matching of a dipole antenna on a 50  $\Omega$  coaxial cable and the determination of the efficiency. Below 1 Watt power and below 300 MHz a ferrit 1:1 balun (manufactured from a 0.5 cm<sup>3</sup> standard 1:4 balun for TV-application) leads to very good results. The additional loss is below 10 % , the volume of the network is very small and is not critical with respect to bandwidth. The parallel  $\lambda/4$  bazooka in FIGURE 106 is only a few percent better, but cumbersome and of limited bandwidth. However, an investigation by the author has shown, that the total bandwidth of dipole antenna plus bazooka is slightly larger than that of the antenna alone, because the reactance of the antenna is partially compensated by the reactance of the bazooka at changing input frequencies. The best method to determine the efficiency follows from the application of the equations by WHEELER [85]: The Smith Chart of the antenna in free-space is recorded, and one reads the real part of the impedance R at resonant frequency fres. The antenna is then located in a conducting vessel with the dimension of the radiansphere (see section 4.5.). The Smith Chart is again recorded, and the highest ohmic resistance near f<sub>res</sub> represents the total loss resistance R<sub>loss</sub>. The efficiency Eff can be calculated with

$$Efficiency = \frac{R - R_{1oss}}{R}$$
(251)

The absolute accuracy is in the region of 25 %, but the relative accuracy is much better than 5 %, e.g., if only the feeding point of an antenna is varied. If one combines the efficiency measurement with transmission tests (network analyser, second input channel), the actual performance of a helical antenna can be reliably quantified.

#### 16.1.2. COMPARISON OF SOME ANTENNA TYPES

Five antennas have been selected for a discussion of the performance: The GROUNDPLANE ANTENNA GA is a vertical  $\lambda/4$  whip on 4 ground rods, each  $\lambda/4$ 

of length and at an angle of 135° to the radiating whip. The HELMET ANTEN-NA HGA is a vertical  $\lambda/4$  whip (32.6 cm) on a plastic helmet, coated with a copper mesh. The ROUND HELICAL DIPOLE RHD is the antenna shown in FIG-URE 107. The FLAT HELICAL DIPOLE FHD is a flat helix with  $2N_h = 10.5$  turns, 2h = 20 cm,  $D_{h1} = 0.5$  cm,  $D_{h2} = 5.3$  cm. In the center section the antenna conductor is parallel to the antenna axis at a total length of 6 cm , representing the feeded antenna segment ( $\Delta$ -match with bazooka, similar to the feeding in FIGURE 107). The FLAT FOLDED HELICAL DIPOLE FFHD consists of a 240  $\Omega$  parallel line with the shape of a helix, with single conductors at the antenna ends (see LI and BEAM [54]).The size of the FFHD is the same as that of the FHD, but  $2N_h$  is 13.5.

COMP	ARISON OF SOME BODY-MOUNTED	ANTENNNAS USED IN BIOTELEMETRY
ANT. TYPE	ANTENNA DATA IN FREE SPACE WITHOUT TEST SUBJECT	VERTICAL ANTENNA MOUNTED DORSALLY ON THE TEST SUBJECT WITH $d_{at} = 57 \text{ mm}, h_B = 1.4 \text{ m}, \phi = 180^{\circ}/0^{\circ}$
	RES. BAND- GAIN EFFI- VSWR FREQ. WIDTH CIEN- CY [MHz] [MHz] [dB] [%] [1]	RES. BAND- GAIN GAIN FREQ. FREQ. WIDTH at $\phi$ at $\phi$ SHIFT 180° 0° [MHz] [MHz] [dB] [dB] [MHz]
GA HGA RHD FHD FFHD	220         65         2.15         -         1:1.4           237         47         +0.5         -         1:1.3           236.3         18         +0.2         89         1:2.5           237.5         25         +0.8         79         1:4.0           242.1         8.8         +1.0         78         1:1.3	229.2       23       -20       -7.7       7.1         231.8       24       -21       -6.1       5.7         236.8       9.6       -20       -4.0       5.3

TABLE 110 Comparison of some body-mounted antennas.

TABLE 110 shows the performance of these antennas. The helical antennas were mounted on the phantom PHA, the antenna center was spaced 57 mm from the surface of the phantom. These results hold true within 2 dB when mounted dorsally on a human test subject SUB. The performance of the GA was measured at an absolute antenna height of 1.4 m, and the helmet antenna HGA was mounted at the head of the SUB, with the head at the same absolute antenna height. This comparison shows clearly that a good antenna in free space may perform poorly in extreme proximity to a body. The interesting FFHD with its high gain and its excellent VSWR cannot be applied in practice, because varying antenna-body distances may detune the antenna in excess of its bandwidth. A good compromise seems to be the flat helical dipole FHD.

#### 16.2. COMPUTER PROGRAMS AND ADDITIONAL RESULTS

#### General remarks

In the following sections the listings of the used computer programs are presented with all necessary comments. The source programs are those of HARRINGTON and MAUTZ [40] (program A is essentially the here presented program HARRA) and of BEVENSEE [10] (program HARRDF is a part of the here presented program PANB). Program PANA and PANC are new programs.

The programs are written in FORTRAN IV for a CDC computer. Card decks are available from HARRINGTON, BEVENSEE or from the author. Depending on your computer system, some of the characters need to be changed or the punched characters do not agree with the listing obtained from the punched cards. Please check above all the following characters:

- C : for comment
- = : might be printed (and read) as a >
- \* : might be printed (and raad) as a +
- + : might be printed (and read) as a {

If you notice some differences between your listing (from the card deck) and the presented listing, use a subroutine DECODE for character replacement. Such subroutines should be available at your computer center.

Depending on your computer system, the organisation of the main programs and the subroutines may be different. Problems may occur with the COMMON statements. Check the listings and ask the specialists of the computer center. Before actual computing the punched cards beginning with a C,R or E must be replaced by the corresponding control cards :

- C : This card is only a comment card and has no influence on the computation
- R : Replace that card by an appropriate control card
- E : Take this punched card out of the program

Except program PANA, the execution of the programs is quite expensive. The minimum computational time on a CDC 6500 for a series of 4 test points at one single frequency is in the order of 1000 seconds.

#### 16.2.1. PROGRAM PANA AND IZYL RESULTS

E PROCRAM #PANA#, COMPUTATION OF THE TWO-DIMENSIONAL MODEL PROCRAM #PANA##################################	 \$:	49 FORMAT(F7, 3) PANA 50 READ(1,51) F
PAN, 3571, 0460000, 0715.		21 FURTRATICA, 137 FIE(GPF (1)) 52,53 FAUA
PUBLIC, BRUSLIB.	4 <b>4</b> 7 10	
PUBLIC, FTNLIB.	91 81	LAM=300./F
LLISET, LLIB=BKUSLLIB/FINLIB. LGO.	~ ®	K≡ba.28315/LAM AK≝K#A
R HERE END OF RECORD CARD	6 (	520 FORMAT(1H1, 1X, *A Z I M UT H A L R A D I A T I O N P A T T E R NPANA
PROGRAM PANA (INPUT, OUTPUT, TAPE1=INPUT, TAPE3=OUTPUT) PAN	NA 11	IFRENUE ANTENNA-TANA 2BODY SYSTEM",/1H0,1X,"TESTBODY: INFINITE ROT.SYN.CTLINDER", 7X, PANA
C HERBERSER	NA 12	3*NUMBER OF MODI:",I3,/1H ,11X, *DIAMETER:",F4.2,* M*,17X, PANA
C THIS PROGRAM COMPUTES THE SCATTERING OF A PLANE WAVE FROM A FEREFILY PAN C CONDUCTING, CIRCULAR CYLINDER OF INFINITE LENGTH. COMPUTED ARE THE PAN	n ∎≇	4*POLARISATION: VERTICAL/VERTICAL*) PANA 521 FORMAT(140,1X,*BAT(I) : ANTENNA-BODY DISTANCE IN METERS*,/1H ,1X,PANA
C E-FIELD COMPONENTS PARALLEL TO THE CYLINDER AXIS (VERT.POLARISATION) PAN C THE TUPHT PATA SET CONSTSTS OF THE FOLLOWING PARAMETERS.	X N ک	1*GAI(I) : GAIN, FIELD STRENGTH AT THE ANTENNA A1 IN DB, O DB = FRPANA DEF SEMPTERAN 1V #DUT : UNDITANTAN AND 5 TH DEFERSENDAND
CF = FREQUENCY IM MHZ	NA 17	522 FORMAT (1H0, 1X, * DAT (1) DAT (2) DAT (2) DAT (4) DAT (5) * / , 1H , 4X, PANA
C A = RADIUS OF THE CYLINDER IN METERS	NA 18	15F7.3,/,1H0,1X, PHI GAI(1) GAI(2) GAI(3) GAI(4) GAI(5) <sup>4</sup> ,5X, <sup>4</sup> -20 DBPANA
C UMINT = MINIMUM ANTENNA-BOUT SURFACE DISTANCE FOR PART 1 C DMAXT = MAXTMUM ANTENNA-BODY SURFACE DISTANCE FOR PART 1	N N	Z",4X,*=10 UB",4X,*=4 UB",4X,*=10 UB",7,1 523 FORMAT(1H .1X.I3.5F7.1.08X,*.*.9X.*.*.9X.*.*.9X.*.*) PANA
C MR1 = NUMBER OF COMPUTED ANTENNA-BODY DISTANCES FOR PART 1 PAN	NA 21	524 FORMAT(1H+, 38X, 41(R1)) PANA
C M1 = NUMBER OF REGARDED MODI FOR PART 1 PAN	88 8	WRITE(3,520) F,M1,DU PANA
C UMING = MINIMUM ANIENNA-BOUT SURFACE DISTANCE FUR PART 2 C DMAX2 = MAXTMIM ANTENNA-PODY SIRFACE DISTANCE FOR PART 2	र त. इ. इ.	MALTE(5,521) FANA Martine(5,521) FANA
C MR2 = NUMBER OF COMPUTED ANTENNA-BODY DISTANCES FOR PART 2 PAN	12 12	DM=(DMAX1-DMIN1)/(HR1-1) PANA
C M2 = NUMBER OF RECARDED MODI FOR PART 2 PAIL	92 K N 150	DO 7 JJ=1,MR1 PAN4 7 PAN4 7 PAN4 7 PAN4 PAN4 PAN4 PAN4 PAN4 PAN4 PAN4 PAN4
C THE OUTPOIT OF THIS PROCHAM PANA IS THE FLELD STRENGTH AT THE BUDY- PAN C MAINTED BECETVING ANTENNA WITH THE ECVITATIO PARAMETERS.	57 60 19 10	「 DAII(いい)= DMINI+(UM*(いい・)) WRTTE(3.522)(DAI1(1、1=3.MR1) PANA
C PHI = HORIZONTAL ROTATION ANGLE IN DEGREES	8 8 8 8	DO 2 1=1,37 PANA
C DAT = DISTANCE OF THE ANTENNA FROM THE CYLINDER SURFACE PAN	8 8	PAILE(I_1)#5 PANA
$\mathbf{U}$ LAI = FIELU SIKENGIH IN US AI INE ANIENNA, (U UB = FREE SFACE VALUE FRI PAI	5 22 5 22	C DURING THE COMPUTATION THE ARGUMENT (PHI+PI) IS REPLACED BY PHH PANA
C THE FIRST PART CONSISTS OF THE COMPUTATION OF THE AZIMUTHAL RADIATION PAI	NA 33	DO 3 L=1,41 PANA 2 D1/( 2-10
C PAITERN FOR FIVE (MR1) ANTENNA-BODY DISTANCES PRESENTED IN A TABLE PAI C AND TN A CRAPH.	s, ¥ N N	DO 14 J=1. PANA
C THE SECOND PART CONSISTS OF THE COMPUTATION OF REFLECTING PROPERTIES PAI	19 19 19	RK=(DAT1(J)+A)#K
C (PHI = 0 DEG) AND DIRECTING PROPERTIES (PHI = 180 DEG) OF THE ANTENNA-PAI C CVI INDER SYSTEM AT THENTY (MR2) ANTENNA-PODY DISTANCES	NN 258	CALL BESS(M, AK, RK, PHH, EZI, EZSC, Y, G) FF(G, GT, 10, CPB, G, 17,00-) (2010, 2
	968 11 M	GAI1(J)=G
C THIS PROCRAM TAKES USE OF THE LIBRARY BRUSLIB VIMCODE C306 FOR THE PAI	NA 40	IF(G.GT.10OR.G.LT20.) COTO 4 PANA
C COMPUTATION OF THE BESSEL AND HANKEL FUNCTIONS. C ################	N 4	KK=LFLX(G+30,5) D1(KK)=1R0+J PANA
PAN	NA 43	4 CONTINUE PANA
COMPLEX EZI, EZSC PATAGON CATAGEN PAI	NA 44	WALTE(3,523) PHL,(GAIT(11),11=1,MKT) URITE(3,5201) (Dif(1),1-1 bt)
REAL RK.LAM.AK.A.K.G	29 8 9	2 CONTINUE
INTEGER F, D1(41), D2(51), D3(51), PHI	AL 17	PANA
READ(1,448) DWIN1,DMAX1,MR1,M1 READ(1,448) DMIN2,DMAX2,MR2,M2	81 AN	620 FORMAT(1H1, 1X, *DIRECTIVE RADIATION PATTER NPANA 1 FREOUENCY*.I4. #MHZ*./140.1X. *T40-DIMENSIONAL ANTENNA-PANA
48 FORMAT(2F6.2, 214)	NN 20	ZBODY SYSTEM*, 10X, *ROT. ANGLE PHI : 0 ABOVE, 180 BELOM*, /1H0, 1X, PANA
READ(1,49) A	NA 51	3*TESTBODY: INFINITE ROT.SYM.CYLINDER#,10X, #NUMBER OF MODI: #, I3, /, PANA

2	<b>3 じ 6 日</b>	×
6686511511	222222222222222222222222222222222222222	.808999898999955555555555555555555555555
н, 3X, ****, 6X, * 108 - 8, ММ Рим 2 X, *, *, 9X, * *, 9X, * * , 9X, * Рим Рами Рами	7488 7488 7488 7488 7488 7488 7488 7488	PANN PANN PANN PANN PANN PANN PANN PANN
и (110, 13, 1941 (1.) (м.1.(.) тит., эм си , б., - 5 DB*, W, *, 0 DB*, W, *, 5 DB*, /1 T(11 , 1X, F4. 2, F9. 2, 3X, 13, 08X, *, *, 9X, *, * T(11 , 1X, F4. 2, F9. 2, 3X, 51 (A1)) T(11 , 1X, *0.00*, 23X, 51 (A1)) T(11 , 1X, *0.00*, 23X, 51 (A1))	(13, pcr) MAXZ-JMINE)/(MR2-1) 10 1.14, 10 1.14, 10 1.14, 10 1.14, 10 1.14, 10 1.14, 10 2.15, 10 2	1 Jai, MEZ 1 Jai, MEZ MT2(J) MT2(J) MT2(J) MT2(J) MT2(J) MT2(J) MT2(J) MT2(J) MT2(J) HIH CT5CH.C.LI20) COTO 92 LEI HIL DEIH CT5CH.C.LI20) COTO 92 CT5CH.C.LI20) COTO 92 TOUC CT5CH.C.LI20) COTO 92 TOUC CT5CH.C.LI
21 FORM 25X, #1 23 FORM 23 FORM 24 FORM 25 FORM 25 FORM	MART MART MART MART MART MART MART MART	DO 12 CONTRACTOR CONTR

<pre>N(I)=-JN(I)/CMFLX(JN(I),-YN(I)) CALL BESYN(-RK,MP1,JN,YN) D0 2 I=1,MP1</pre>	<pre>PHICE(I)=CHELX(JN(I),-YN(I)) EZSC=AN(1)#HNCR(1) EZ1=IN(1)</pre>	M.1-N.1 NU=N+1	BN=-CMPLX(0., N*3. 1415926/2.) BN=2.*COS(N*P)*CEXP(BN) 577-577.004810/011)	EZSC=EZSC+BN#HNKR(NU)#AN(NU) V=CARS(1_4F7SC/F71)	G=(ALOG10(V)*20.) BETTIRN & FND	(11) ・ 11・)に、	R HERE END OF RECORD CARD	C DATA CARDS	0.05 0.25 05 12	0.05 1.00 20 25	0.125	150 MHZ	205 MHZ	250 MHZ	R HERE END OF INFORMATION CARD	
<del>.</del>	2			~		0	<u>~</u>	ы							<b>F</b>	

N			8	1			_					_						-	-				T	771	1	25	MHZ
Ŧ	₹		22	·	••	• •	•••	•	•••	••	•••	•••	•	•••	1	•••	•••	٠	•••	·	•••	• •	Ľ				1012
μ	BEL	CAL	+					. *	* *																		
12	80	RTI	~			_*		-			<b>.</b> •	*															
5	щ. Т	E.	DE	.	·_•*			•		••	• •	• •	*				• •	•	•••	٠	•••	• •	٠	••	•	•••	••
z	BOV	CAL	+		•								-														
5	A O	82 11 12 12 12 12 12 12 12 12 12 12 12 12											-														
ø	••	ц <sup>я</sup> ,	80	1										. *.													*.*.
8	E	MOL	5	1.	•••	• •		•		•••			•									. *	* *			-	
<u> </u>	ΞE	OF ŠATI																		* *		-					
N	ANC	BER	æ															. •	•								
ω	Ĕ	UMB CIC	1 01	•	•••	•••	•••	•	••	•••	•••	•••	•	••••		• •	•		•••	•	•••	• •	•	• •	•	•••	••
E L	-		Ϋ́.	1													*										
4			m																								
_ ₽			D 2	•	••	• •	•••	•	••	•••	•••	••	٠			• *••	• •	•	• •	٠	••	• •	٠	••	•	•••	••
Z	δ	e:	T																								
H H	ЦSТ	INDI																									
L.	S N	E.	B	.											.   •	· .											
H	BOD	ΜW	5	1																							
AD	-W	25.S													-												
8	TEN	8.	но	0	000			00	00				00		ç	20	<u>.</u>	00	29	99	20	00	89	20	00	20	<u>.</u>
ш	AN	TER	EB												ä	ē⇔	<u>۳</u>	80	200	₽;	<u>≃</u> ₽	<u>۳</u>	Ψ.		<u>۳</u>	<u>=</u>	≌₽
L A	NAL	FIN	î		ar. 90 m	0 vo g	۰ <u>م</u>	ω.c	2		tω	e z	Ng	~ <u>@</u> @	5	≥≂	ສຍ	210	- 2	<u></u>	2.00	22	8	- 5	<u>∞</u> =	- 200	<b>5</b> 8
H	SIC	NIG	DB (AI		- I		- ~	~i~	i mir	-00 nmr	101		1-	= 9		n at	NO	9	စ်ထုံ		÷φ	φφ	ភូរ	с in	ហេដ	i f	न न
μ	MEN	:YQ	0	1.	•										•	īΤ	77	•	•••	•	•••	• •	Ċ	• •	•		•••
<b>~</b>	Ę	Ĩ.	9	8	£84	284	ng	លទ	មេន	វភគ្	ŝ	ສກ	۶¥	າວຮ	88	82	БS	សខ	ያኤ	<u>9</u>	ក្ខន	មូន	នទ	212	84	38	88
	M	E	LAC	· ] - '			•••				••••			••••	6	•			••••	•	•			•••	•	•••	
2			- 1																								
Z H			-		DB			•			•••			 			<u> </u>	•					•				 • •
5 M H Z					+10 DB	•	•••	•			•••		•		•	•••		•		•	•••				•		•••
125 M H Z		CAL	ACE		+10 DB		•••	•			•••		•		•	•••		•			•••	• •			•		•••
C Y 125 M H Z		RTICAL	: SPACE		DB +10 DB	ري د	оъ • •	رت بر • • •			•••		•		•	•••	 	•	 	•	•••	•••	•		•	 	•••
N C Y 125 M H Z		VERTICAL	REE SPACE		+ 0 DB +10 DB	3 45 3 45		3 45		345	· · ·	34 5	45 ME	15. 5.		•••	•••	•	 • •	•	•••	•••	•	•••	•	 	•••
UENCY 125 MHZ		CAL/VERTICAL	= FREE SPACE		+ 0 DB +10 DB	2 3 45 2 3 45	2345	2 3 45	5	2 345		2 34 5	2 3 45	34 5	34 5	34.5	45	45		•	•••	•••	•	•••	•		•••
EQUENCY 125 MHZ		12 CRTICAL/VERTICAL	DB = FREE SPACE		DB + 0 DB +10 DB	1 2 3 45		2 2 2 15		2 345		2 3 4 5 · · ·	2 3 45	2 34 5.	2 345	2 34 5	345	3_45	3 45	45 · · ·	15 · · ·		•	•••	•	•••	•••
REQUENCY 125 MHZ		DI: 12 : VERTICAL/VERTICAL	, 0 DB = FREE SPACE		10 DB + 0 DB +10 DB	.1 23 45 	.1 23 45	1 2 3 45 1 2 3 45		2 345		1. 2 3 45	1. 2345	1.2345. .2345.	.2345 2345.	.2 34 5	2 3 45 · · · ·	2,3,45	2.3 45	2345	3.45	3455		34.5	345	345	345
FREQUENCY 125 MHZ		MODI: 12 CON: VERTICAL/VERTICAL	DB, 0 DB = FREE SPACE		-10 DB + 0 DB +10 DB	.1 2345 1 2345	.1 23 45	1 2 3 45 1 2 3 45	1010 200 200 200 200 200 200 200 200 200	1 2 345 2 15 2 15		1. 2 3 4 5	1. 23 45	1. 2 3 45	1.2 345	1.2 345	1 2 3 45	2.3 45	2.3 45	2345 · · ·	2 3.45	2 34 5	2 3 45	2 34.5	2 345 · · ·	2 345	2 345
IN FREQUENCY 125 MHZ		OF MODI: 12 SATION: VERTICAL/VERTICAL	IN DB, 0 DB = FREE SPACE		3 -10 DB + 0 DB +10 DB	-1 2 3 45 	1 23 45	1 2 3 45	100 200 200			1. 2 3 4 5	1. 2345.	1.2345. 1.2345.	1.2345	1 .2 34 5	1 2 3 45 · · · · · · · ·	1 2, 3 45	1 2.3 45	1 2345	1 2 3.45	2 34 5 · · · · · · · · · · · · · · · · · ·	2 3 45 · · ·	2 34.5	2 345	2 345	2 345
ERN FREQUENCY 125 MHZ		BER OF MODI: 12 ARISATION: VERTICAL/VERTICAL	A1 IN DB, 0 DB = FREE SPACE		0 DB -10 DB + 0 DB +10 DB			1 2 3 US		2450		. 1. 2 3 45	. 1. 2345		• 1.2 345. •	. 1.2 345	· 1 2 3 45 · · · ·	• 1 2.3 45 • •	. 1 2 .3 45	. 1 2345	.1 23.45	.1 2 34 5	·1 2 3 45 · · ·	.1 2 34.5	1 2 345	1 2 345	1 2 345
LTERN FREQUENCY 125 MHZ		NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	NNA A1 IN DB, 0 DB = FREE SPACE REES		-20 DB -10 DB + 0 DB +10 DB		.1 23 45	1 2 3 45				. 1. 2 3 45	. 1. 2345		. 1.2 345	. 1 .2 34 5	· 1 2 345 · · ·	. 1 2.3 <sup>45</sup>	. 1 2.3 45	. 1 2 3 45	.1 23.45	.1 2 34 5 2 3 45	-1 23 45	.1 2 34.5	1 2 345 · · ·	1 2 345	1 2 345
ATTERN FREQUENCY 125 MHZ		NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	NTENNA A1 IN DB, 0 DB = FREE SPACE DEGREES		-20 DB -10 DB + 0 DB +10 DB			1 2 3 45				. 1. 2 3 45 . 1. 2 3 4 5	. 1. 2345		. 1.2345 1.3 45	. 1 .2 345	· 1 2 3 45 · · ·	. 1 2.3 <sup>45</sup>	. 1 2 3 45	. 1 2 3 45	. 1 2 3.45	-1 2 34 5 · · · · · · · · · · · · · · · · · ·	·1 2 3 45	.1 2 34.5	1 2 345	1 2 345	1 2 345
PATTERN FREQUENCY 125 MHZ		NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	E ANTENNA A1 IN DB, 0 DB = FREE SPACE IN DECREES	50	5) -20 DB -10 DB + 0 DB +10 DB	.01 2 3 45		9 1 2 3 45		55		.2 . 1. 2 3 45	.7 . 1. 23 45		.3 . 1.2 345		.8 . 1 2 3 45	.0 . 1 2.3 45 .	.2 . 1 2 .3 45	. 1 2 3 45	.9 .1 2 3.45	.41 2 34 5			1.5 1 2 345 · · ·	1 2 345	
ON PATTERN FREQUENCY 125 MHZ		ER NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	N TALLENS THE ANTENNA A1 IN DB, 0 DB = FREE SPACE LE IN DEGREES	AT(5) .250	AI(5) -20 DB -10 DB + 0 DB +10 DB	1.01 2.3 45	1.01 23.45	.1 2 3 45 1 2 3 45				21 · 1. 2 3 4 5 · 1	7 . 1. 2 3 45.		-2.3 . 1 . 2 34 5	-3.2 . 1 .2 34 5	-3.8 · 1 2 3 45 · · · · · · · · · · · · · · · · · ·	-5.0 . 1 2.3 45 .	-5.0 . 1 2.3 45	-6.8 . 1 2.3 45	-7.9 . 1 2 3.45	-8.4 .1 2 34 5	-9.1 .1 23 45 .	-9.5 1 2 34.5	-9.5 1 2 345		-9.6 1 2 345
ION PATTERN FREQUENCY 125 MHZ	YSTEM	INDER NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	L IN THLEMA A1 IN DB, 0 DB = FREE SPACE ANGLE IN DEGREES	) DAT(5) 0250	) GAI(5) -20 DB -10 DB + 0 DB +10 DB	1 1.01 2.3 45 .	2 1.0 1	2 .9 .1 2 3 45 3 .0 1 2 3 45				74 . 1. 2 345	07 . 1. 2 3 45		6 -2.3 . 1 . 2 34 5	6 -3.2 · 1 .2 34 5 · · ·	1 -3.8 · 1 2 3 45 · · · · · · · · · · · · · · · · · ·	3 -5.0 · 1 2.3 45 · ·	5 -6.2 . 1 2 .3 45	1 -6.8 . 1 2 3 45	1 -7.9 . 1 2 3.45	0 -8.4 . 1 2 34 5	3 -9.1 -1 2 3 45	6 -9.5 1 .1 2 34.5	7 -9.5 1 2 345	7 -9.6 1 2 345	.7 -9.6 1 2 345
ATION PATTERN FREQUENCY 125 MHZ	Y SYSTEM	CYLINDER NUMBER OF MODI: 12 POLARISATION: VERTICAL VERTICAL	ANGL IN FILE ANTENNA AI IN DB, O DB = FREE SPACE OM ANGLE IN DECREES	T(4) DAT(5) .200 .250	I(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	1 1.0 1 2 3 45	-2 1.01 23 45 .	2 .91 2 3 45				-1.42 · · · 1. 2 3 45 · · ·			-3.6 -2.3 . 1 . 2 34.5		-5.1 -3.8 · 1 2 345 · · · ·	-6.3 -5.0 . 1 2.3 <sup>45</sup>	-0.9 -5.0 · · · 2.3 45 · · ·	-8.1 -6.8 . 1 2.3 45	-9.1 -7.9 . 1 2 3.45	-9.6 -8.4 ] .1 2 34 5	-10.3 -9.1 .1 2 3 45 .	-10.6 -9.51 2 34.5	-10.7 -9.5 1 2 345	-10.79.6 1 2 345	-10.7 -9.6 1 2 345
IATION PATTERN FREQUENCY 125 MHZ	BODY SYSTEM	W.CYLINDER NUMBER OF MODI: 12 POLARISATION: VERTICAL	LEARCE AT THE ANTENNA A1 IN DB, O DB = FREE SPACE ATTOM ANGLE IN DEGREES ATTOM ANGLE IN DEGREES	DAT(4) DAT(5) .200 .250	GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	1 1.01 2.3 45 .							-2.07 . 1. 23 45				-5.1         -3.8         .         1         2         345         .         1           5.7         -4.4         .         1         2         345         .         .         1	5.5.0 . 1 2.3 <sup>45</sup>		3 –8.1 –6.8 . 1 2.3 45		39.68.4 ] .1 2 34 5		2 -10.6 -9.5 1 .1 2 34.5	3 -10.7 -9.5 1 2 345	3 -10.7 -9.6 1 2 345	3 -10.7 -9.6 1 2 345
ADIATION PATTERN FREQUENCY 125 MHZ	NA-BODY SYSTEM	T.SYM.CTLINDER NUMBER OF MODI: 12 25 M POLARISATION: VERTICAL/VERTICAL	A DESAMPLE A THEATENNA AI IN DB, O DB = FREE SPACE NOTATION ANGLE IN DEGREES	(3) DAT(4) DAT(5) 150 .200 .250	(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	1.8 -1 1.0 1 2 3 45	1.82 1.0	1.92 .91 2 3 45 .		201		3.1 -1.42			<b>5.4</b> -3.6 -2.3 . 1 . 2 34 5 2 34 5	6.4	-6.9 -5.1 -3.8 · 1 2 3 45 · · · · · · · · · · · · · · · · · ·	-8.0 -6.3 -5.0 . 1 2.3 45	-6.0 -0.3 -5.0 . $-0.2$ . $1 2.3 +5$	-9.8 -8.1 -6.8 . 1 2 3 45	10.8 -9.1 -7.9 . 1 2 3.45	11.3 -9.6 -8.4 .1 2 345 11.6 -10.0 -8.8 .: 2 3 45		12.2 -10.6 -9.5 1 .1 2 34.5	12.3 -10.7 -9.5 1 2 345	12.3 -10.7 -9.6 1 2 345	12.3 -10.7 -9.6 1 2 345
RADIATION PATTERN FREQUENCY 125 MHZ	TENNA-BODY SYSTEM	ROT.SYM.CYLINDER NUMBER OF MODI: 12 	educi distante an Hicker al IN DB, O DB = FREE SPACE ELD STRUCTH AT THE ANTENNA A1 IN DB, O DB = FREE SPACE AL ROTATION ANGLE IN DECREES	DAT(3) DAT(4) DAT(5) .150 .200 .250	GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	-1.8 -1 1.01 2.3 45 .	-1.82 1.01 23 45 .					-3.1 -1.42 1. 2 3 45	-3.8 $-2.0$ $7$ . 1. 2 3 $45.$ . $-1.2$ .		-5.4 -3.6 -2.3 · 1 · 2 34 5 · ·	-5.4 -4.6 -3.2 · 1 .2 34 5 · ·	-6.9 -5.1 -3.8 · 1 2 3 45 · · · · · · · · · · · · · · · · · ·	-8.0 -6.3 -5.0 . 1 2.3 45 .	-0.2 -7.5 -6.2 · 1 2 .3 45 · · ·	-9.8 -8.1 -6.8 . 1 2.3 45		-11.3 -9.6 -8.4 1 .1 2 34 5 11.6 -10.0 -8.8 1 .1 2 3 45	-11.9 -10.3 -9.1 .1 2 3 45	-12.2 -10.6 -9.5 1 .1 2 34.5	-12.3 -10.7 -9.5 1 2 345	-12.3 -10.7 -9.6 1 2 345	-12.3 -10.7 -9.6 1 2 345
L RADIATION PATTERN FREQUENCY 125 MHZ	ANTENNA-BODY SYSTEM	ITE ROT.SYM.CXLINDER NUMBER OF MODI: 12 ITER: 25 M	MARADUI DISIMULAI THE ANTENNA AI IN DB, O DB = FREE SPACE ONTAL ROTATION ANGLE IN DECREES	22) DAT(3) DAT(4) DAT(5) 000 .150 .200 .250	(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB		1,4 -1,82 1,0	1,5 -1,92 .91 2 3 45				5.1 -3.4 -1.74	5.4 - 3.8 - 2.07 . 1. 2 3 45		B.0         -5.4         -3.6         -2.3         -1         -2         3.45         -	3.0 -6.4 -4.6 -3.2 · 1 .2 34 5 · · ·	9.6 -6.9 -5.1 -3.8 · 1 2 3 45 · · ] 0.1 -7.5 -5.7 -4.4 · 1 2 34 5 · · ·		1.3 -6.0 -0.3 -5.0 · · · · 2.3 45 · · · ·	2.4 -9.8 -8.1 -6.8 . 1 2.3 45	3.4 -10.8 -9.1 -7.9 . 1 2 3.45	3.8 -11.3 -9.6 -8.4 .1 2 34 5		4.7 -12.2 -10.6 -9.51 2 34.5	4.8 -12.3 -10.7 -9.5 1 2 345	1.0         1.0 <th1.0< th=""> <th1.0< th=""> <th1.0< th=""></th1.0<></th1.0<></th1.0<>	4.8 -12.3 -10.7 -9.6 1 2 345
HAL RADIATION PATTERN FREQUENCY 125 MHZ	NAL ANTENNA-BODY SYSTEM	FINITE ROT. SW. CYLINDER NUMBER OF MODI: 12 AMERIE: 25 M COLLARISATION: VERTICAL/VERTICAL	THENR-POLY DISTANCE A MILLION AI IN DB, O DB = FREE SPACE PRIZONTAL ROTATION ANGLE IN DEGREES	AT(2) DAT(3) DAT(4) DAT(5) .100 .150 .200 .250	AI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB							-5.8 $-3.1$ $-1.4$ $2$ $. 1. 2 3.45 . -6.1 -3.4 -1.7 4 . 1. 2 3.45 .$	-6.4 -3.8 -2.07 . 1. 2.3 45			-0.0 -6.4 -4.6 -3.2 · 1 · 2 34 5 · ·	-9.6 -6.9 -5.1 -3.8 · 1 2 345 · · 1 -10.1 -7.5 -5.7 -4.4 · 1 2 345 · · ·		-11.9 -9.2 -7.5 -6.2 . 1 2 .3 45	-12.4 -9.8 -8.1 -6.8 . 1 2.3 45	-13.4 -10.8 -9.1 -7.9 . 1 2 3.45	-13.8 -11.3 -9.6 -8.4 1 .1 2 34 5		-14.0 -12.1 -10.0 -9.5 $.1 2 34.5$ $.1 2 -10.6 -9.5$	-14.8 -12.3 -10.7 -9.5 1 2 345	-14.8 -12.3 -10.7 -9.6 1 2 345 · · ·	-14.8 -12.3 -10.7 -9.6 1 2 345
ITHAL RADIATION PATTERN FREQUENCY 125 MHZ	SIONAL ANTENNA-BODY SYSTEM	INFINITE ROT.STM.CXLINDER NUMBER OF MODI: 12 DIAMETER 25 M DOLARISATION: VERTICAL VERTICAL	ANTENNAM-COLUMDATION ALLA NELLORA AL IN DB, O DB = FREE SPACE (ANT) FIELD STRUCH AT THE ANTENNA AI IN DB, O DB = FREE SPACE HORIZONTAL ROTATION ANGLE IN DEGREES	) DAT(2) DAT(3) DAT(4) DAT(5) 0. 100 . 150 . 200 . 250	) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3		H         H				.7 -5.8 -3.1 -1.42 ] . 1. 2 3 45 . ]	H         -6.4         -3.8         -2.0        7         .         1.         2.3         45.         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.         2.3         1.5         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         1.5         2.45         .         .         1.5         2.45         .         .         .         1.5         2.45         .		0 - 8:0 - 5:4 - 3:6 - 2:3 · 1 - 2 34 5 · ·	0 -9.0 -6.4 -4.6 -3.2 · 1 .2 34 5 · ·	.5 -9.6 -6.9 -5.1 -3.8 . 1 2 3.45	.6 -10.7 -8.0 -6.3 -5.0 . 1 2.3 45 .		.3 -12.4 -9.8 -8.1 -6.8 . 1 2.3 45	2 -13.4 -10.8 -9.1 -7.9 .1 2 3.45	.6 -13.8 -11.3 -9.6 -8.4 ] .1 2 34 5	2 -14.4 -11.9 -10.3 -9.1 .1 2 3 45 .	.5 -14.7 -12.2 -10.6 -9.51 2 34.5	.5 -14.8 -12.3 -10.7 -9.5 1 2 345	6 -14.8 -12.3 -10.7 -9.6 1 2 345	.6 -14.8 -12.3 -10.7 -9.6 1 2 345
MUTHAL RADIATION PATTERN FREQUENCY 125 MHZ	MENSIONAL ANTENNA-BODY SYSTEM	DDY: INFINITE ROT.SYM.CYLINDER NUMBER OF MODI: 12 DIAMETER: 25 M DATANALSYN, 2000 NOT NUMBER OF MODI: 12	: ANTIANTEOLO JUSTIANCH AT THE ANTENNA AT IN DB, O DB = FREE SFACE : HORIZONTAL ROTATION ANGLE IN DEGREES	TT(1) DAT(2) DAT(3) DAT(4) DAT(5) .050 .100 .150 .200 .250	AI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	-9.3 -4.4 -1.81 1.01 2 3 45 .	-9.4 -4.4 -1.82 1.0 · · · · · 2 3.45 ·	-9.4 ±4.5 -1.92 .91 2 3 45				-10.7 -5.8 -3.1 -1.42 ] . 1. 2 3 45 . 1. -11.0 -6.1 -3.4 -1.74 ] . 1. 2 34 5 . ]		12.1 -7.2 -4.5 -2.7 -1.4 - 1 - 2 -4.5	-13.0 -8.0 -5.4 -3.6 -2.3 · 1 . 2 34.5 · · ·		-14.5 -9.6 -6.9 -5.1 -3.8 · 1 2 3 45 · · 1 -15.1 -10.1 -7.5 -5.7 -4.4 · 1 2 345 · · ·	-15.6 -10.7 -8.0 -6.3 -5.0 . 1 2.3 <sup>45</sup>	$-10.2 -11.3 -0.0 -0.3 -5.0$ $\cdot 1 2.3 +5 \cdot \cdot -16.8 -11.9 -9.2 -7.5 -6.2$ $\cdot 1 2.3 +5 \cdot \cdot -16.8 -11.9 -9.2 -7.5 -6.2$	-17.3 -12.4 -9.8 -8.1 -6.8 . 1 2 3 45	-11.0 -12.9 -10.3 -0.0 -1.4 .1 -7.9 .1 -2.345	-18.6 -13.8 -11.3 -9.6 -8.4 ] .1 2 34 5	-19.2 -14.4 -11.9 -10.3 -9.1 1 2 3 45	-19.5 -14.7 -12.2 -10.6 -9.5 1 .1 2 34.5	-19.5 -14.8 -12.3 -10.7 -9.5 1 2 345		-19.6 -14.8 -12.3 -10.7 -9.6 1 2 345
IMUTHAL RADIATION PATTERN FREQUENCY 125 MHZ	D-DIMENSIONAL ANTENNA-BODY SYSTEM	STBODY: INFINITE ROT.SW.CYLINDER NUMBER OF MODI: 12 DIAMPHAR: 25 M PARAMON NUMBER POLARISATION: VERTICAL/VERTICAL	(11) ANTENNE-BOLT VESTIME LAT THE ANTENNA AI IN DB, O DB = FREE SPACE I : HORIZONTAL ROTATION ANGLE IN DEGREES	DAT(1) DAT(2) DAT(3) DAT(4) DAT(5) .050 .100 .150 .200 .250	[GAI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	20 -9-3 -1.4 -1.81 1.01 2 3 45 .	0 -9.4 -4.4 -1.82 1.0 · · · · · 2 3 45 ·	5 -9.4 -4.5 -1.92 .91 2 3 45 .		5 -10.0 -5.1 -2.4 -7 -5 - 1 2 345 - 1 345 -		0 -10.7 -5.8 -3.1 -1.42 0 -10.7 -5.8 45 · 1 5 -11.0 -6.1 -3.4 -1.74 · 1 · 2 34 5 · 1	0 -11.4 -6.4 -3.8 -2.07 . 1. 2.3 45 5 -11.7 - 6 - 1.1 - 2.2 -1.1	0 -12.1 -7.2 -1.5 -2.7 -1.4 1 - 1 - 2 -3 -5. 5 -12.6 -7.6 -4.9 -3.1 -1.8 1 - 2 -34.5.	0 -13.0 -8.0 -5.4 -3.6 -2.3 . 1 .2 34.5		5 - 14.5 - 9.6 - 6.9 - 5.1 - 3.8 · 1 2 3.45 · · · ] 0 - 15.1 - 10.1 - 7.5 - 5.7 - 4.4 · · 1 2 3.45 · · · ]	5 -15.6 -10.7 -8.0 -6.3 -5.0 . 1 2.3 45 .	51 - 16.8 - 11.9 - 9.2 - 7.5 - 6.2 . 1 2 .3 $45$	0 -17.3 -12.4 -9.8 -8.1 -6.8 · 1 2.3 45 · · ·	0 -18.2 -13.4 -10.8 -9.1 -7.9 .1 2 3.45	5 -18.6 -13.8 -11.3 -9.6 -8.4 1 1 2 34 5	5 -19.2 -14.4 -11.9 -10.3 -9.1 .1 2 3 45 .	5 -19.5 -14.7 -12.2 -10.6 -9.51 2 34.5	0 -19.5 -14.8 -12.3 -10.7 -9.5 1 2 345	0 -19.6 -14.8 -12.3 -10.7 -9.6 1 2 345	5 -19.6 -14.8 -12.3 -10.7 -9.6 1 2 345
AZIMUTHAL RADIATION PATTERN FREQUENCY 125 MHZ	TWO-DIMENSIONAL ANTENNA-BODY SYSTEM	TESTBODY: INFINITE ROT.SW.CYLINDER NUMBER OF MODI: 12 DIAMPERI: 25 M MARTIN: DIAMPERI: 25 M	DATIL) : ANTENNE-BOLT ULBLING TO THELLING (ANTL) : CANN, FELLO STREPT AT THE ANTENNA AI IN DB, 0 DB = FREE SFACE PHI : HONIZONTAL ROTATION ANGLE IN DEGREES	DAT(1) DAT(2) DAT(4) DAT(5) .050 .100 .150 .200 .250	PHI GAI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	0 -9.3 -1.4 -1.8 -1 1.01 2.3 45 .		15 -9.4 -4.5 -1.92 -91 2 3 45 20 -0.5 -4.6 -2.0 -3 -91 2 3 45		30 -9.0 -1.9 -2.30 .0 . 1 2 34.5 . 35 -10.0 -5.1 -2.4 -7 .5 . 1 2 345 .		50         -10.7         -5.8         -3.1         -1.4        2         .         1.         2         3.45         .         1           55         -11.0         -6.1         -3.4         -1.7        4         .         1.         2         34         5         .         1	60         -11.4         -6.4         -3.8         -2.0        7         .         1         2         345.         .           66         -11.7         -6.8         -11         -2         -1         .         1         2         345.         .	700 -12:1 -7:2 -4:5 -2:7 -1:4 1 2 34 5: 751 -12:6 -7:6 -4:9 -3:1 -1:8 1 2 34 5:	80 -13.0 -8.0 -5.4 -3.6 -2.3 . 1 .2 34 5	90 -14.0 -9.0 -6.4 -4.6 -3.2 . 1 .2 34 5	95         -14.5         -9.6         -5.1         -3.8         .         1         2         3.45         .         .         1           100         -15.1         -10.1         -7.5         -5.7         -4.4         .         1         2         3.45         .         .         1		$\begin{bmatrix} 110 \\ -10.2 \\ -11.9 \\ -9.2 \\ -7.5 \\ -6.2 \\ -7.5 \\ -6.2 \\ -1 \\ -5.2 \\$	120 -17.3 -12.4 -9.8 -8.1 -6.8 . 1 2 3 45		135     -18.6     -13.8     -11.3     -9.6     -8.4     .1     2     24.5     .       1400     -18.9     -14.1     -10.0     -8.8     .     2     34.5     .	145 -19.2 -14.4 -11.9 -10.3 -9.1 .1 2 3 45	130         -19.4         -14.7         -12.2         -10.6         -9.5         .1         2         34.5         .         .	160 -19.5 -14.8 -12.3 -10.7 -9.5 1 2 345		1175 -19.6 -14.8 -12.3 -10.7 -9.6 1 2 345

,

<b>x</b> 3	<u>م</u> ا		• •	• •	••	••	•••	•••	•••		1		•••	••	•••	••	IZ	<u>/L 3</u>	500	MH
QUENCY 300 ( 0 ABOVE, 180 BE 25 ERTICAL AVERTICAL	+ 0 DB +	••	.*. *	* * ' 	* **	* •	•••	* *	•••	. <b>*</b>				••	••		••	••		•
IFREQ IGLEPHI: COFMODI: SATTON: V	- 5 DB	• •	•••	• •	••	•••	• •	•••	••	• • •			•••	•••			•	• • '	*• *•	<b>*</b> . *
T T E R N ROT.AN NUMBER	-10 DB		•••	• •	• •	•••	••	• • •	••	•••		• • •	* *	•••	•••	•••	•••	• •	•••	•
ONPA Tem Der	-15 DB		• •	•••	•••	••	••	• • •	• •	•••	•			•••		•••	• •	• •	•••	•
D I A T I -BODY SYS SYM.CYLIN	-20 DB		•••	•••	••	••	•••	•••	••	•••	.		•••	•••	••	••	•••	• •	•••	•
V E R A L ANTENNA NITE ROT.	DBG	00	00	00	00	00	000	000	000	000	180	8 <u>8</u> 8	88	222 222	<u>8</u> 88	2 <u>8</u> 8	282	88	88	85
E C T I IMENSIONA ODY: INFI	CAI(I)	12 S	88	1.78	2•31 1.60	8° 8	ខនះ តំហិត	2.75	~~~ 838	-3.81	-19.97 -15.13	-12.61	ምም	882. 977	ດ ສະ 	ក្តតុ ម ពុំពុំ	ዓድ የጥ	ያም ዋዋ	₹8. Ƴ†	88 1 1
D I R TWO-D	DAT(I M	88	88	8.6		<u>ទ</u> មុះ	ខ្ទះខ្ទ		សុនុម	55.8	8.8.5	58	សុខ្ល	ອີອັງ	 ÷ខ់ព	 ບໍ່ອີ ເ	8 <u>8</u>	68	ສີຮູ	ຮູ
2 H W			BC 01	•••			•••	•••	• • •					•••	•••	• •	•••			•
N C Y 300	REE SPACE		0 DB +	44 88	<i>տ</i> , տ՝ សូស្ល	ំ។ សូតំ	2 0 0 0 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	ភ្លៃ ភ្លេស្ត្រ ភ្លេស្ត្រ	2,315 2,315 2,15	2 3 4 2 3 4 2 3 4 2		34.5	ស្នំសំ	•••				•••		•
REQUE DI: 12 VERTICAL	0 DB = F		+ 80			·				·~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	<sup>م</sup> <sup>م</sup>	~			n				
			2			• •	•••	•••	•••	•••		·~	-	?∾;	ູີ	ູ່ສີ່	ີ. ເ	ងដ	តិភិ	랷
E R N F BER OF MC	A1 IN DB,		0 DB -10		•••	•••	•••	•••	•••	•••		•		?~?; ~~,	· 1 1 2.3		1 2 34.5			1 2345
PATTERNF NUMBERGEMO	ETERS E ANTENNA A1 IN DB, IN DECREES	68	5) -20 DB -10	<u>ر</u> وبور 		 	وىرىم • • •		<u></u>										9 . 2345	
ATIONPATTERNF SYSTEM Y SYSTEM CTLINDER NUMBER OF MC	ANCE IN METERS GTH AT THE ANTENNA A1 IN DB, ON ANGLE IN DECREES	T(4) DAT(5) .200 .250	I(4) GAI(5) -20 DB -10	3.6 3.5	3.50	3.5 3.6 3.6	••••••••••••••••••••••••••••••••••••••		2.6 3.2 2.4 3.1 2.1 2.0	1.4 2.4		6	-2.07			-6.9 -7.6 · · 1 2 3.45			11.2 -10.0 · 23 45 11.1 -9.9 · 23 45	11.0 -9.9 1 2 3 45 11.0 -0.8 1 2 3 45
RADIATION PATTERNF TENNA-BODY SYSTEM ROT.STMCTLINDER NUMBER OF MO	BODY DISTANCE IN METERS ELD STRENGTH AT THE ANTENNA AT IN DB, AL ROTATION ANGLE IN DEGREES	DAT(3) DAT(4) DAT(5) .150 .200 .250	GAI(3) GAI(4) GAI(5) -20 DB -10	2.8 3.6 3.5 2.8 3.6 3.5	2.7 3.6 3.6		2.2 3.3 3.5 2.2 3.3 3.5 2.5 3.3 3.5		1.3 2.6 3.2 1.0 2.4 3.1 6 2.1 2.0		-1.2 .5 1.6	-2.46 .6 1	-3.9 -2.07 . 1. 2 -4.7 -2.8 -1.5 . 1 . 2	-5.6 -5.7 - 16.2 -6.6 -15.7 - 13.2 -13.2 - 13.2			-12.3 -10.5 -9.3 1 2 34.5	-12.6 -11.0 -9.8 · 23 45	-12.8 -11.2 -10.0 . 23 45 -12.7 -11.1 -9.9 . 23 45	-12.6 -11.0 -9.9 1 2 3 45 -12.6 -11.0 -0.8 1 2 3 45
THAL RADIATION PATTERNF SLOWL ANTENNA-BODY SYSTEM INFINITE ROTSTM. CNLINDER NUMBER OF NO DIAMETRE SOFT	ANTENNA-BOY DISTANCE IN METERS GAIN, FIELD STRENGTH AT THE ANTENNA A1 IN D9, HORIZONTAL ROTATION ANGLE IN DEGREES	) DAT(2) DAT(3) DAT(4) DAT(5) ) .100 .150 .200 .250	) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10		· · 7 2·8 3.6 3.6			1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	9 -1.1 1.3 2.6 3.2 7 -1.4 1.0 2.4 3.1	2 - 2.3 .2 1.7 2.6 - 2.8 - 2 1.4 2.4		2 -5.2 -2.46 .6				5 -13.5 -10.7 -8.9 -7.6 1 1 2 3.45 5 -13.5 -10.7 -8.9 -7.6 1 1 2 3.45	3 -14.9 -12.3 -10.5 -9.3 1 2 34.5	2 -15.3 -12.8 -11.2 -10.0   2 3 45	1 -15.2 -12.7 -11.1 -9.9 . 23 45	0 -15.2 -12.6 -11.0 -9.9 1 2 3 45 0 -15.1 -12.6 -11.0 -0.8 1 2 3 45

1 1					_		_																ZYL	-40	υM	HZ I
Η	ð	ч	- 5 D		•••	••	•	•••	• •	••	••	•••		•••	.	• •	•	•••	• •	•	•••					
ş	SO BE	RTIC/	Ŧ	•	*			*	* *			*		•												
L L	, щ.	- VEI	昭〇〇	*	*.	•••	*	• •	• •	- 181-	••		•••	• •	.	•••	•	••	• •	•	•••	••	••	•••	• •	•
2 12	ABON	LICAL	+			*	*							*												
n o	,	LER.	B			*									.										** **	
ы С	IHA	NOL	2	• •				•••	•	•••	••	•••	•••							*	* *	* *	***	* *		
й.   2	NGLE	T OF																*	• *	*						
е ш	ŭ.A	OLAR	ad of	•	•••	•••	••	••	•	•••	•••	•••	••	••	·	•••	• *	• •	• •	•	•••	••	•••	••	•••	•
н 1	•	-	ï													*	*									
A 4	-	l	的												.					•	•••				• •	
z	្ត	ER	-15													•										
П	SYSTI	CIND	m																							
N I	, MO	M. CY	20 D	• •	••	••	••	•••	•	•••	•••	•••	• •	•••		•••	••	•••	•	••	•••	•••	•••	• •	•••	•••
D	NA-B	П. SY 25 м		-		_																		•	·'	
~	NTEN	E RO	ΗB	00	000	000	00	00	00	00	00	00	00	00	180	888	389	<u>8</u> 8	88	86	385	889	38	88	88	8
2	ALA.	TINIT				·	- 0/1	~ ~	<b>~</b>	2 10	ഗന	~ ~	mo	00	ر.		- 40 0	20	90	1 - 1 - 5	5.00	2 = 1	≍ 190'	مو چ		
E E	ISION	NIC N		1 2.2	500		- N 8 - N	2.2	50	5 <del>-</del> -	~ ~	1.4	0 	*8. ~-	20.3	15.15		2 2 7 7 7	φ' Γ			ŶŶ	ና የ	- - - - -	หาส เ	Ŧ
5	IME	SODY:											_											~		
		TEST	MTC	88	688	98K	22	ຮູຮູ	សំព	ý₽.	3.H	ຮຸ່	ຮູ	⊇.ຄຸ	88	;e#	181	ά Η	សំគ	÷.	, right	5.99	2.6	8	8.9	
F				<b></b>	<u> </u>			_			-															
											_															
2 н	7 1				B			•							•					•••						
1 1 2 2	7 H M M	-1	ω		+10 DB	•	•••	• •				•••	••		•				•	•••	••	•••		•	•••	
сн и UVII v	7 H M 00+ I	TICAL	SPACE		B +10 DB	5 4			254	 S2n		 	<u>ب</u>	ۍ . تو	55		•••	•••	•	•••		•••		•	••	
N C V BOO M H	N C I 400 M H 2	VERTICAL	REE SPACE		0 DB +10 DB	1.54		1.00 	1.254				هي کي کي	2.35 2.35	2.345 2.345	2 3 45 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2	3.45	34.5 . 345 .	45. IE	· · ·	•••	•••		•	•••	 
	U E N C I 400 M N 2	ICAL/VERTICAL	= FREE SPACE		+ 0 DB +10 DB				1. 254		 	1 25		1 2.35 1 2.35	1 2 345	2 345	2345 	2 34.5 2 3 45	2 3 45.	3 45	34 5 • • • • •	•••		•	•••	 
тория 1900 и и 2 2 стария – С. А. ВОО и и 2	Е Ц О Е И С I 400 М И 2	: 12 VERTICAL/VERTICAL	0 db = Free Space		) DB + 0 DB +10 DB	. 1.5 H			1.254		 5.25 			· 1 2.35 ·	• 1 2 345 1 2 345		1 2 3 45	1 2 34.5 1. 2 3 45	. 2 3 45.	.2 3 45	2345 · · · · · · · · · · · · · · · · · · ·	3.45 · · · ·	4.5 45			5 
5 9 5 7 1 6 M C X 800 M 6 7	F К Е 4 О Е И С I 400 М И 2	MODI: 12 ON: VERTICAL/VERTICAL	DB, O DB = FREE SPACE		-10 DB + 0 DB +10 DB				1.254	. 1.254 .	 222	. 1 .254		· 1 2.35 ·	· 1 2 .345		2345	1 2 34.5 1. 2 3 45	1.23 <sup>45.</sup>		2.345	2 3 45	2 34.5			2 3 45
	K N F K E Q U E N C I 400 M N Z	R OF MODI: 12 CSATION: VERTICAL/VERTICAL	I IN DB, O DB = FREE SPACE		DB				1.254	. 1.254 .	 22	. 1 .254 	. 1 235	· 1 2.35 ·				1 2 34.5 . 1. 2 3 45 .	1. 2 3 45.		1 2 3 4 5	1 2 3 45	2 34.5 2 3 45			2 3 45
	ТЕКИ F КЕQUENCI 400 МИ.	MBER OF MODI: 12 DIARISATION: VERTICAL/VERTICAL	A A1 IN DB, 0 DB = FREE SPACE Ees		-20 DB -10 DB + 0 DB +10 DB				1.254	· · · · · · · · · · · · · · · · · · ·								. 1 2 34.5 . . 1. 2 3 45 .	1.2345.			.1 2 3 45	· 2 34.5 · · · ·			. 2345
, , , , , , , , , , , , , , , , , , ,	АТТЕКИ ҒКЕФОЕИСІ 400 МИ 2	NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	SS VTENNA A1 IN DB, 0 DB = FREE SPACE DEGREES		-20 DB -10 DB + 0 DB +10 DB	. 1.54			1.254					. 1 2.35				1 2 34.5 1. 2 3 45	1.2345.			.1 2 3 45 .	2 34.5 · · · · · · · · · · · · · · · · · · ·		ດ ເຊັ່ນ ສາກາດ ເຊັ່ນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັ	. 2345
ים איזינים ער מערכם שרע לאומא אני איני א	м РАТТЕКИ ККЕЧОЕИСИ 400 М И 2	NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	METERS HE ANTENNA A1 IN DB, 0 DB = FREE SPACE IN DEGREES	250	(5) ~20 DB ~10 DB + 0 DB +10 DB	2.1 1.54 .		2.3 2 H	2.6	2.7	3.1	3.3	3.4	3.2	2.9 1 2 345		1.8	-7 . 1 2 34.5 .	9 1.2345.			-0.7 1 2 3.45	9.2 . 2 34.5			0.1 2345
ייה היוורים א ניסייבן ארי אומי איז	LON PATTERN FREQUENCI400MHZ STEM	NDER NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	IN METERS AT THE AMTENNA A1 IN DB, O DB = FREE SPACE NGLE IN DEGREES	DAT(5) .250	GAI(5) -20 DB -10 DB + 0 DB +10 DB	2.1 1.54 .	2.2	2.3 2.4	2.6	2.9	3.1	3.3 3.4	3.4 3.4 	3.3 3.2 	2.9				<b>1.</b> 2 3 45.			+	5 -9.2 . 2 34.5			3 -10.1 . 2 3 45
יפירים יידירים אין הייריא וואי אין א	ATTON PATTERN FREQUENCI 400 M H.C. (SYSTEM	CYLINDER NUMBER OF MODI: 12 POLARISATION: VENTICAL/VERTICAL	ance in meters cth at the antrinna a1 in db, o db = free space on angle in degrees	T(4) DAT(5) .200 .250	I(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3.6 2.1 1.54 .	3.6 2.2	3.7 2.3 3.7 2.4 1.54	3.7 2.6	3.7 2.9 1 . 254	3.7 3.1 1 .2 54	3.6 3.3 3.5 3.4	3.3 3.4 3.1 3.4 	2.8 3.3 1 2.35 . 2.5 3.2 1 2.35 .	2.2 2.9 1 2 345		1 1.3 . 1 2 3 45 .	6 .7 . 1 2 34.5 . -1.40 1. 2 345 .	-2.39 1. 2 3 45.			-8.1 -6.7 1 2 3 45	-10.5 -9.2 · 2 34.5 · · · 2 34.5 · · · ·			-11.3 -10.1 . 2 3 45
יייפייטה מיניפינטה בסבטובתיע 1000 את י	DIATION PATTERN FREQUENCI400 M 1.2 -BODY SYSTEM	SM.CTLINDER NUMBER OF MODI: 12 M	DISTANCE IN METERS TRENCTH AT THE ANTENNA A1 IN DB, O DB = FREE SPACE TATION ANGLE IN DEGREES	) DAT(4) DAT(5) 0 .200 .250	) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	8 3.6 2.1 1.54 .	8 3.6 2.1 8 3.6 2.2	7 3.7 2.3 . 1.54 7 3.7 2.3	6 3.7 2.6	5 3.7 2.7	3 3.7 3.1 1 .254	9 3.6 3.3	3 3.3 3.4 0 3.1 3.4 1 2 35	6 2.8 3.3 · · 1 2.35 · · 2.5 3.2 · · 1 2.35 · ·				46 .7 . 1 2 34.5 . 3 -1.40 . 1. 2 345 .	2 -2.3 -9 1. 2 3 45.		5 -5.5 -4.0 - 1 2 34 5	.1 -8.1 -6.71 2 3 45 . .3 -9.4 -8.0 -1 2 3.45 .	3 -10.5 -9.2   · 2 34.5 · · ·	3 -11.6 -10.4 2 34 5 · · · · · · · · · · · · · · · · · ·		.0 -11.3 -10.1 . 2 3 45
ע הייני יוען הייני וויין איז	RADIATION PATTEKN FREQUENCI400 M H 4 ENNA-BODY SYSTEM	ROT.SYM.CYLINDER NUMBER OF MODI: 12 .25 M .CYLINDER POLARISATION: VERTICAL/VERTICAL	ODY DISTANCE IN METERS LD STRENGTH AT THE ANTRINAA A1 IN DB, O DB = FREE SPACE L ROTATION ANGLE IN DEGREES	AT(3) DAT(4) DAT(5) .150 .200 .250	AI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3.8 3.6 2.1 1.54 .	3.8 3.6 2.1	3.7 3.7 2.3 3.7 3.7 2.3	3.6 3.7 2.6	3.5 3.7 2.7 [ 1 . 254	3.3 3.7 3.1 1 .2 54	2.9 3.6 3.3 1 .254	2.3 3.3 3.4 2.0 3.1 3.4 2.0 3.1 3.4	1.6 2.8 3.3 · · · 1 2.35 · · · 1 2.35 · ·				-2.46 .7 . 1 2 34.5 . -3.3 -1.40 . 1 2 3 45 .	4.2 -2.39 1. 2 3 45.			-10.1 -8.1 -6.7   .1 2 3 45	-12.3 -10.5 -9.2 · 2 34.5 · · · · · · · · · · · · · · · · · · ·	-13.3 -11.6 -10.4 2 34 5	-13.3 -11.6 -10.4 · · · · · · · · · · · · · · · · · · ·	-13.0 -11.3 -10.1 . 2 3 45
י הייייייה יכוביים אימייים אימייים אימיייייים אימיייייייייי	L RADIATION PATTERN FREQUENCI 400 MH 2 ANTENNA-BODY SYSTEM	ITE ROT.SYM.CYLINDER NUMBER OF MODI: 12 TER: .25 M POLARISATION: VERTICAL/VERTICAL	NA-BODY DISTANCE IN METERS FIELD STRENGTH AT THE ANTENNA A1 IN DB, 0 DB = FREE SPACE ONTAL ROTATION ANGLE IN DEGREES	2) DAT(3) DAT(4) DAT(5) 00 .150 .200 .250	2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	.4 3.8 3.6 2.1 · · 1.5 4 ·	2.4 3.8 3.6 2.1 · · · · · · · · · · · · · · · · · · ·	23 3.7 3.7 2.3 2.1 54 . 1.54 . 1.54	1.0 3.6 3.7 2.6 1	1.9 3.5 3.7 2.7 5.9 1 . 254 .	2.1 3.3 3.7 3.1 1 .2 54	5 2.6 3.6 3.3 . 1 2.5 <sup>4</sup>	.2 2.3 3.3 3.4 1 2 35	.7 1.6 2.8 3.3 1 2.35 . 			3.7 -1.0 .7 1.8 23.45	5.2 -2.46 .7 . 1 2 34.5 . 5.1 -3.3 -1.40 1. 2 3 45 .	7.0 - 4.2 - 2.3 - 9 . 1. 2 3 45.	3.2 -6.3 -4.3 -2.9 · 1 .2 3 45 · ·	0.4 -7.5 -5.5 -4.0 1 2 34 5	2.9 -10.1 -8.1 -6.7   .1 2 3.45	5.1 -12.3 -10.5 -9.2   . 2 34.5		5.7 -13.1 -11.5 -10.4 · · · · · · · · · · · · · · · · · · ·	5.5 -13.0 -11.3 -10.1 · 2 3 45 · · ·
ייי הייריפירטת הייניים בייני ווא אניין איין איין איין איין איין איין אי	'HAL RADIATION PATTERN FREQUENCI 400 M 4 4 OMALANTENNA-BODY SYSTEM	NFINITE ROT.SYM.CYLINDER NYMBER OF MODI: 12 TAMETER: .25 M FOLANISATION: VERTICAL/VERTICAL	NTENNA-BOOY DISTANCE IN METERS WIN, FIELD STRENGTH AT THE ANTENNA A1 IN DB, O DB = FREE SPACE IORIZONTAL ROTATION ANGLE IN DEGREES	DAT(2) DAT(3) DAT(4) DAT(5) .100 .150 .200 .250	GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	2.4 3.8 3.6 2.1 1.54 .	2.4 3.8 3.6 2.1 · · · 1.54 · · 2.3 3.8 3.6 2.2 · · · · · 1.54 · · ·	2.3 3.7 3.7 2.3 1.54 .	2.0 3.6 3.7 2.6	1.9 3.5 3.7 2.7   1 . 254 . 1.7 3.4 3.7 2.9 1 . 254 .	1.4 3.3 3.7 3.1 1 .2 54 1.2 3.1 3.7 3.2	5 2.6 3.6 3.3 . 1 2 5 <sup>4</sup> . 1 25 5 <sup>4</sup> . 1 25 5 <sup>4</sup> .	-2 2.3 3.3 3.4 1 2 35 -3 2.0 3.1 3.4 1 2 35	7 1.6 2.8 3.3 1 2.35 . -1.2 1.2 2.5 3.2 1 2.35 .	-1.8 .7 2.2 2.9 1 2 .345 .			-5.2 $-2.4$ $6$ $.7$ . 1 2 $34.5$ . $-6.1$ $-3.3$ $-1.4$ $0$ . 1. 2 $345$ .			$-10.4$ $-7.5$ $-5.5$ $-4.0$ $\cdot$ $1$ $\epsilon$ $34$ $5$ $\cdot$ $-11.6$ $-8.8$ $-6.8$ $-5.3$ $\cdot$ $1$ $2$ $\cdot$ $34$ $5$ $\cdot$ $\cdot$ $-11.6$ $-8.45$ $-6.8$ $-5.3$ $\cdot$ $1$ $-7$ $-7$	-12.9 -10.1 -8.1 -6.7   .1 2 3 45	-15.1 -12.3 -10.5 -9.2   . 2 34.5		-15.8 -13.5 -11.6 -10.4 · · · · · · · · · · · · · · · · · · ·	-15-5 -13.0 -11.3 -10.1 · 2 3 45 · · ·
יייייי יייייייייייייייייייייייייייייי	IUTHAL RADIATION PATTERN FREGUENCY 400 M 42 ENSIONAL ANTENNA-BODY SYSTEM	Y: INFINITE ROT.SYM.CYLINDER NUMBER OF MODI: 12 DIAMETER: 25 M FOLARISATION: VERTICAL/VERTICAL	: ANTENNA-BODY DISTANCE IN METERS : GAIN, FIELD STRENGTH AT THE ANTENNA A1 IN DB, O DB = FREE SPACE : HORIZONTAL ROTATION ANGLE IN DECREES	(1) DAT(2) DAT(3) DAT(4) DAT(5) (050 .100 .150 .200 .250	T(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	-1.9 2.4 3.8 3.6 2.1 1.54 .	-1.9 2.4 3.8 3.6 2.1 1.54	20 2.3 3.7 3.7 2.3 . 1.54	2.3 2.0 3.6 3.7 2.6 1.254 .	2.5 1.9 3.5 3.7 2.7 1 2.5 <sup>4</sup>		Lit 5 2.6 3.6 3.3 . 1 2.54	1.6     .2     2.3     3.4     .     1     2.35       5.1    3     2.0     3.1     3.4     .     1     2.35	-5.67 1.6 2.8 3.3 1 2.35 . -6.1 -1.2 1.2 2.5 3.2 1 2.35 .				$10.3 - 5.2 - 2.4 - 6 \cdot 7   \cdot 1 2 34.5 \cdot 11.2 - 6.1 - 3.3 - 1.4 - 0   \cdot 1.2 3 45 \cdot .$	12.2 - 7.0 - 4.2 - 2.3 - 9 1. 2 $3 - 45$		15.5 -10.4 -7.5 -5.5 -4.0   . 1 2 34 5	18.0 -12.9 -10.1 -8.1 -6.7   .1 2 3.45	20.1 -15.1 -12.3 -10.5 -9.2   . 2 34.5	20.8 -15.9 -13.3 -11.6 -10.4 2 34 5	20.7 -15.8 -13.3 -11.6 -10.4 · · · · · · · · · · · · · · · · · · ·	20.4 -15.5 -12.9 -11.3 -10.1 . 2 3 45
ע הייטין איזייקר איזירט איז	ZIMUTHAL RADIATION PATTEKN FREGUENCY 400 M L 4	STBODY: INFINITE ROT.SYM.CYLINDER NUMBER OF MODI: 12 DIAMETER: -25 M	r(I) : ANTENNA-BODY DISTANCE IN METERS ((I) : GAIN, FIELD STRENGH AT THE ANTENNA A1 IN DB, 0 DB = FREE SPACE (: : HORIZONTAL ROTATION ANGLE IN DEGREES	DAT(1) DAT(2) DAT(4) DAT(4) DAT(5) .050 .100 .150 .200 .250	r Gai(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	0 -1.9 2.4 3.8 3.6 2.1 1.54 .	5 -1.9 2.4 3.8 3.6 2.1 · · · 1.54 · · ] 9 -1.9 2.3 3.8 3.6 2.2 · · · 1.54 · · ]		5 -2.3 2.0 3.6 3.7 2.6 1.254 .	0 -2.5 1.9 3.5 3.7 2.7 · · · 1 · 254 · · · · · 1 · 254 · · · · · · · · · · · · · · · · · · ·	0 -3.1 1.4 3.3 3.7 3.1 1 .2 54	0 -3.7 .9 2.9 3.6 3.3 . 1 .254 .	0         4.6         2         2.3         3.3         3.4         .         1         2.35         .         .         1         2.35         .         .         1         2.35         .         .         .         .         .         .         .         .         .         .         .         .         .         .         .         . <th< td=""><td>0 -5.67 1.6 2.8 3.3 · · 1 2.35 ·</td><td></td><td></td><td>5     -3.7     -1.0     .7     1.8     .     .1     2     3.45     .       0     -9.5     -4.4     -1.7     .1     1.3     .     1     2     345     .</td><td>5 -10.3 -5.2 -2.46 .7 . 1 2 34.5 . 0 -11.2 -6.1 -3.3 -1.40 . 1 2 3 45 .</td><td><math display="block">5 \begin{bmatrix} -12.2 \\ -12.2 \\ -7.0 \\ -1.2 \\ -7.0 \\ -1.2 \\ -2.3 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -</math></td><td>5 -14.4 -9.2 -6.3 -1.3 -2.9 · 1 .2 3 45 · ·</td><td>0 -15.5 -10.4 -7.5 -5.5 -4.0 · 1 2 34.5 · · · 55.5 -16.8 -11.6 -8.8 -6.8 -5.3 · 1 2 34.5 · · ·</td><td>0 -18.0 -12.9 -10.1 -8.1 -6.7   . 1 2 3 45</td><td>201 –20.1 –15.1 –12.3 –10.5 –9.2   ・ 2 34.5 ・ 5 – 20.6 –15.6 –13.0 –11.3 –10.0   ・ 2 3.45 ・</td><td></td><td><math display="block">\begin{array}{c ccccccccccccccccccccccccccccccccccc</math></td><td>0 -20.4 -15.5 -13.0 -11.3 -10.1 . 2 3 45</td></th<>	0 -5.67 1.6 2.8 3.3 · · 1 2.35 ·			5     -3.7     -1.0     .7     1.8     .     .1     2     3.45     .       0     -9.5     -4.4     -1.7     .1     1.3     .     1     2     345     .	5 -10.3 -5.2 -2.46 .7 . 1 2 34.5 . 0 -11.2 -6.1 -3.3 -1.40 . 1 2 3 45 .	$5 \begin{bmatrix} -12.2 \\ -12.2 \\ -7.0 \\ -1.2 \\ -7.0 \\ -1.2 \\ -2.3 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -9 \\ -$	5 -14.4 -9.2 -6.3 -1.3 -2.9 · 1 .2 3 45 · ·	0 -15.5 -10.4 -7.5 -5.5 -4.0 · 1 2 34.5 · · · 55.5 -16.8 -11.6 -8.8 -6.8 -5.3 · 1 2 34.5 · · ·	0 -18.0 -12.9 -10.1 -8.1 -6.7   . 1 2 3 45	201 –20.1 –15.1 –12.3 –10.5 –9.2   ・ 2 34.5 ・ 5 – 20.6 –15.6 –13.0 –11.3 –10.0   ・ 2 3.45 ・		$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	0 -20.4 -15.5 -13.0 -11.3 -10.1 . 2 3 45

Z			四							. ]	ZYL 700 MHz	
HW	ELOW	AL	<u>د</u>	1.	••		• • •	••••	• • • • •			
20	808	RTIC			*	*	•	• .	 			
C X	Е,1	LVE	80	*.	• *•	.**.						
ыN	ABO	5 FTCAI	Ĭ				*	* _	-			
0	•	EE S	6			*	*	-				
н В Ш	IH	ION:	- 21	1.	•••	• • • •	•••	••••	*			
<u>н</u>	J. E	R OF	'								•	
8	т. А	MBEI	B	.						***		
E	Å	28	۴							*		
-			e									
Z			15 D	•	• •	• • • •	•••	•••	• • • • • •			
01	STEM	NDER	'							-		
F.	Y SY	CHLI	8	.								
ЦЦ	1000	M.	ې م									
RA	ENNA.	Ę,										
ш	ANT	ER:	H		00	0000	000	000	0000000	868888888888888888888888888888888888888	<u>.</u> 	
2	ONAL	(FIN)	E	5	60	E858	855	23.29	じんちびにおお	8000-8590380851	0-0-300	
н 0	NSIC	A A A	GAI	3 -	- 1	N N			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	266666666666666666	ዸ፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟፟	
ы ж	DIME	BODY	<b>1</b>			_				T I I I I		
H	ę.	TEST	DAT(	E   8 	សំខំ	ខ្លួនមុខ	ខំនំអំដ	ਤੇਲੇ ਤੇ।	<i>ស់មក់</i> ទំន	<u>ទីស់តក់ខ្លស់ម៉ស់</u> ទំតំថ្នស់ខ្លស់	ទកនេះខ្លួន	
							•					
ΖH					DB							
Σ					₽ ₽	• • •		•••				
7 7		ICAL	PACE		_	~~~	N N N	หมห	៳៳៷៷៳៹	ດດ ດ ດ ດ ດ		
U X		VERT	E		8	ທີ່ທີ່ທີ່			* <u>*</u> ****	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		
ы Э		CAL/	#		+	====	*£	កំណ័្ណ	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	~~_ <sup>~</sup> ~~ <sup>~</sup> ~~ <sup>~</sup> ~~ <sup>~</sup> ~~		
о ш		12 ERTI	8		B					  	n	
84 14		N: V			우	• • •	•••	•••	• • • • • • •	······································		
z		NTIO NTIO	N							· ' '	Nan man	
БR		ARIS	۲		BB							
H H		E S S S S S S S S S S S S S S S S S S S	REES		Ň							
P A		e la construction de la construc	SEA .		-	<b>N N N</b>	+ 0					
×	F	8	벌	Π(5) .250	1(5)	117	- 4 4		የተጥተ ነ -		0111111	
H	<b>TSTE</b>	INDE	ANG	50 20	C C	===	10100	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	- 01- 08 01 -	∩≠ N∞m≻ON≻∞oN∞ou	11111	
15		E E	H SU S	AT (4 . 20	AI (4	ร่งร่ง	ייי ד ד	i - i	• • ๙๙ ๓๓	mmmvv 1		
1 -	DY S	· · 2	222	а 28	3 0	00-0	vi= voic	ب ت ش م	• ⊢∞∞∞•• <del>-</del>			
D I A	A-BODY S	5 M.C	-2221			ເດັດເດັ	งเจ้าเจ้า	u m m n	๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛๛	ኯፙፙቝ <u>፞</u> ፞ዀ፟ኯ፟ኯ፟ኯ፟ኯ፟ኯ፟፟፟ኯ		
RADIA	TENNA-BODY S	ROT.SYM.(	L Russ	5AT (3	Ĭ							
IL RADIA	ANTENNA-BODY S	VITE ROT.SYM.( TER: .25 M	FIELD ST	(2) DAT (3	(2) CAI(		ာက္ရင္	:::::	8970508		-#10.001-10	
HAL RADIA	CONAL ANTENNA-BODY S	INFINITE ROT. SYM. ( IAMETER: .25 M MTENNA BODY DIST	AIN, FIELD ST	DAT(2) DAT(3 .100 .15	GAI(2) GAI(	ពុកុកុ គាត់តំ	າຕູດ * * * =	, <del>,</del> , , , , , , , , , , , , , , , , ,	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶ ੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶ ੶੶੶੶	-17.5	
UTHAL RADIA	ENSIONAL ANTENNA-BODY S	Y: INFINITE ROT.SYM.( DIAMETER: .25 M	CAIN, FIELD ST HORIZONTAL ROT	(1) DAT (2) DAT (3 050 .100 .15	(1) GAI(2) GAI(	8.8.8. 8.8.8. 8.3.4.4. 8.5.7.	- 6	10,00 1 - 0,0 1 - 0,0	25565939 25565939 25579939 25579939	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2.7 -16.4	
IMUTHAL RADIA	-DIMENSIONAL ANTENNA-BODY S	TBODY: INFINITE ROT.SYM.( DIAMETER: 25 M	(I) : CAIN, FIELD SI : HORIZONTAL ROT	DAT(1) DAT(2) DAT(3 .050 .100 .15	GAI(1) GAI(2) GAI(	888 888 1111		10,00 1 - 0,00	, , , , , , , , , , , , , , , , , , ,	2 2 2 2 2 2 2 2 2 2 2 2 2 2	22.7 -17.5 - 22.7 -17.5 - 22.2 -17.5 - 21.8 -16.7 - 21.6 -16.5	
AZIMUTHAL RADIA	TWO-DIMENSIONAL ANTENNA-BODY S	TESTBODY: INFINITE ROT.SYM.( DIAMETER: .25 H	GAI(1) : GAIN, FIELD ST PHI : HORIZONTAL ROT	DAT(1) DAT(2) DAT(3 .050 .100 .15	PHI GAI(1) GAI(2) GAI(	0 5 0 0 0 5 0 0 0 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0	22 22 22 22 24 24 25 24 25 25 25 25 25 25 25 25 25 25 25 25 25		42688888 4444 4444 4444 4444 4444 4444 4	889 899 900 110 110 110 110 110 110 110 110 1	155 - 21.7 - 16.4 166 - 22.7 - 17.5 176 - 22.2 - 17.5 177 - 21.8 - 16.7 178 - 21.6 - 16.7	

<b>L</b> .			m	1												-									Т	ΖŶ	L (	897	7. M	Hzl
H	ð		10 2	•		• •	•	• •	••	••	•	•••	•	• •	* _	•	٠	•••	•	•••	•	•	••	• •						
66	0 BE	LICA	+			*					*		•		-															
7	.18	VER	8	.*						.*.			• •	<b>.</b>		.				• •	•				•	•				•
×	BOVE	CAL/	0 +			*	*					*																		
ы	A O	ERTI		*	•		•			*	1																			
о ш	::	: 100 1: VI	5 DB			•		•		•••	•		•	. *.	•	•	•	• •	•	• •	•	•	•••	•	•	•	• • •	· ·	•	
8	ಕೆ: ಬ್ರ	TIO)	ï																					. *	*	* *	Ī	•	-	
z	ANG	RISA	æ																			* *	*							
Ш	RQT.	POLA	6	·	•••	••	••	•	•••	•••	•	• •	•	• •	•	•	•	• •	•			•	•••	•	•	•	•••	•	• •	••
1			1																*	-										
P A			留								•					.		.*.		•						•		•		••
z	x	<b>6</b>	÷																											
L L	YSTE	INDE																												
AT	X S	Ę.		1.	•••	• •	•••	•	•••	• •	•	• •	• •	•	• •	.	•	•		٠	•••	·	•••	•	•••	•	• •	•	•	•••
I	DQ-1	MIS	Ŷ.	-			_		-		-				_	_				_										
RA	ENNA	Б <sub>У</sub>					. ~		~~	~~		~ ~		~ ~		-	0			00		00		00		00	2	0	oc	0
ш	ANT	ËË	PHI		500	,00	50	00	.0	00	. 0.	00					18	<u>8</u> 8	ŝ	₫	¢₽	₫¢	è₿ P	₿¢	2₩2	₿ţ	<u>°</u> <u>∞</u>	2 ₩	≊≅	2₩2
ΛI	NAL	UT IN	8	8	5.2.5	ះម្ន	ເສ	ສະ	3%	88	18	88	នគ	8,	- 5 5 6 7 6 7 6 7 7 7 7 7 7 7 7 7 7 7 7 7	R	ጽ	€ ä	3	₹K	ក	នុទ	₽ <i>\</i> S	នុត	5	đ≦	≩⇔	8	52	8
L C	NSIC	HA .:	GAI	η N			1	η <b>ι</b> Γ	v-	ŵ-	~	កុំ	'n	·Ŧ	'n	m.	ц.	51	Ĥ	÷f	יי	٣́°	٩ŗ	57	ï٩	φ́ч	ŶŶ	ŝ	ዓ	፞ኯ፟
ы В В	DIME	BODY	<b>1</b>							~ ~				~ "		~ ~		<b>.</b>		ŝ	<u>س</u> د	0.	0	ы	<u>م</u> ر	0		6	0 4	.0
H		TEST	DAT		ŕ×'s	, w i	P	5,3	δ là	N, P	ä	۳ <b>,</b> ۳	S.N.	<b>⊼</b> i =	=	00	; •		- ~	Q, u	<u>;</u>		ι.	ŵĸ	9.0	÷.		, w	ο, o	2
$\vdash$							_						_	_																
Z H					鋄																									
ZHWZ					+10 DB	•		•	•••			•		•	••	•		•		•		•	••	•		•	•		•	•••
Z H W 168 J		ICAL	PACE		+10 DB				•••		•••		 		 		 			<u>،</u>	•••	•	• •	•		•	• `		•	•••
ICY 897 MHZ		ERTICAL	E SPACE		) DB +10 DB		• • •	. 22		3.52 .	352	.512	.132		1.1 23	1. 423 .	.524	.254	 ເຄີ	2,35	3.45	ŝ		•			•	 • •	•	•••
E N C Y 897 M H Z		AL/VERTICAL	FREE SPACE		+ 0 DB +10 DB	3 4 52	• • •	3 4 52 ·	3 4. 52 . 4 4 . 52 . 4	43.52 ·	+ • • • • • • • • • • • • • • • • • • •	512	45.132	54132	5 1.4 23	51. 423 .	1.524	1.254 .	- · ·	2.35	2 3.45	2 3 45	3 4 5 · · ·	3 45 .	 	•	•	•••	•	•••
Q U E N C Y 897 M H Z		12 RTICAL/VERTICAL	DB = FREE SPACE		B + 0 DB +10 DB	3 4 52		3 4, 52	3 4. 52 . 4 . 52 .	43.52 ·	1 T T T	# 512	45.132	5 4 1 32 ·	5 1.4 23	51. 423 .	1.524	1 .254 .	1 2.35	1 2,35	1 2 3.45	2 3 45	2 3 4 5	2.3.55.	345	<u>.</u>	•	•••	•	•••
REQUENCY 897 MHZ		DI: 12 : VERTICAL/VERTICAL	, 0 DB = FREE SPACE		10 DB + 0 DB +10 DB			3 4 52	. 3.4. 52 . . 4. 52 .	. 43.52 .		. 4 .512 . 	. 45.132 .	. 541 32 .	5 1.1 23	. 51. 423 .	. 1.524 .	. 1.254 .	. 1 2.35	. 1 2.35 .	. 1 2 3 45	1 2 3 45	1.2345 1.2345	2 3 45	2 3 45	3.45	3 4 5 · · · · · · ·			
I FREQUENCY 897 MHZ		MODI: 12 TION: VERTICAL/VERTICAL	I DB, O DB = FREE SPACE		-10 DB + 0 DB +10 DB	. 3 # 52		3 4 52	. 34. 52 .	. 43.52 . 135.5	. + * * * * · · ·	. 4 .512 . , E12.2	. 45.132	. 5 4 1 32 .	5 1.4 23	. 51. 423 .	. 1.524 .	. 1.254 .	. 1	. 1 2.35	.1 2 3.45	1 2 3 45	1. 2345 1.2345	1 2 3 45		2 3.4 5	2 3 4 5	3 45	345	2 3 t 5
RN FREQUENCY 897 MHZ		R OF MODI: 12 LISATION: VERTICAL/VERTICAL	(1 IN D8, 0 DB = FREE SPACE		DB -10 DB + 0 DB +10 DB	3 # 52		3 4 52		· 43.52 ·		. 4 .512 . * 512 .	. 45.132 .	. 5 4 1 32 .	5 1.4 23	51. 423 .	. 1 .524 .	· · · · · · · ·		1 2.35		1 2 3 45	. 1. 2.345 . 1.2345	1 2 3 45		1 2 3.4 5	· 2 3 4.5 · · ·	.2 3 45	. 2 34 5	. 2 34 5
TERN FREQUENCY 897 MHZ		UMBER OF MODI: 12 DLARISATION: VERTICAL/VERTICAL	NA A1 IN DB, 0 DB = FREE SPACE Ees		-20 DB -10 DB + 0 DB +10 DB			3 4 52	. 34. 52 .	43.52 . 	. t 2004 .	. 4 .512 . * 512		. 54132 . 5133 .	5 1.4 23	· · · 51. 423 ·	. 1.524	. 1.254 .	. 1 2 35	1 2.35		1 2 3 45	. 1.23455 .	. 1 .2 3 45 .		1 2 3 4 5	2 3 4 5 · · 2 3 1 5	.2 3 45	. 2 34 5	· 2 34 5. · · · ·
ATTERN FREQUENCY 897 MHZ		NUMBER OF MODI: 12 FOLARISATION: VERTICAL/VERTICAL	TERNIA A1 IN DB, 0 DB = FREE SPACE		-20 DB -10 DB + 0 DB +10 DB	3 # 52		34.52	· · · · · · · · · · · · · · · · · · ·		••••••••••••••••••••••••••••••••••••••	. 4 .512 . * 512 .		5 4 1 32 .	5 1.4 23		. 1 .524 .	. 1 .254 .	. 1 2.35 .	1 2.35		1 2 3 45	. 1.2345 . 1.2345	1 2 3 5		1 2 3.45	2 3 4 5 · · · 2 3 15	.2 3 45	. 2 34 5	. 2 34 5
I PATTERN FREQUENCY 897 MHZ		NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	TE ANTENNA A1 IN DB, 0 DB = FREE SPACE IN DEGREES	55)	(5) -20 DB -10 DB + 0 DB +10 DB	3.3 4 52	3.3 · · · · · · · · · · · · · · · · · ·	3.3	3.2	2.8 · · · · · · · · · · · · · · · · · · ·	1.8 · · · · · · · · · · · · · · · · · · ·	.9512		3.8 . 5 4 1 32 . E 4 1 32 .	4.3 · · · 5 1.4 23 ·	2.4		2.1	3.1	3.1 1 2.35 .		1.5 1 2 3 45	-6 · 1 · 2 3 4 5 · ·	1.7 . 1 .2 3 45		6.8 1 2 3.4 5	9.3 · 2 3 4.5 ·	2.6 .2 3 45	2.3 .2.34.5.	1.4 . 2 345
ON PATTERN FREQUENCY 897 MHZ	Here and the second sec	DER NUMBER OF MODI: 12 POLARISATION: VERTICAL/VERTICAL	T THE ANTENNA A1 IN D8, 0 DB = FREE SPACE GLE IN DEGREES	DAT (5) 250	GAI(5) -20 DB -10 DB + 0 DB +10 DB	3.3	3.3 3.3 	3.3	3.2 · · · · 3.4. 52 · · 3.1 · · 52 · ·	2.8 · · · · · · · · · · · · · · · · · · ·	1.8 · · · · · · · · · · · · · · · · · · ·	.9]512		-3.8 · · · 5 4 1 32 ·	14.3 · · · 5 1.4 23 ·	-2.4 51. 423 .	1.1	2.1 · · · · · 254 · ·	3.1	3.1		1.5 1 2 3 45	6 - 1 - 2 3 4 5	-1.7 . 1 .2 3 45 .		-6.8 1 2 3.4 5	-9.3 · 2 3 4.5 · · ·	-12.6 .2 3 45	-12.3 .2 34 5	
TION PATTERN FREQUENCY 897 MHZ	SYSTEM	YLINDER NUMBER OF MODI: 12 POLARISATION: VENTICAL/VENTICAL	TH AT THE ANDRINA A1 IN DB, O DB = FREE SPACE In Angle IN Decrees	(4) DAT(5) 200 - 250	(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	.0 3.3 · · · 3 # 52 ·		-7 3.3	1.3 3.2 · · · · · · · · · · · · · · · · · · ·	3.1 2.8 · · · · · · · · · · · · · · · · · · ·		5.9 .9	3.7 -2.0	1.9 -3.8 · · · 5 4 1 32 ·			3.1 1.1 1 .524	3.4 2.1 · · · 1 .254 · ·	3.1 3.1	2.6 3.1 1 2.35	1.9 2.8	2 1.5 1 2 3 45	9 .6 . 1 . 2 3 4.5 .	-3.2 -1.7 . 1 .2 3 45 .		-8.6 -6.8 1 2 3.4 5 · ·	10.9 -9.3 · 2 3 4.5 · · ·	14.0 -12.6 .2 3 45	13.6 -12.3 . 2 34 5	12.6 -11.4 . 2 34.5
IATION PATTERN FREQUENCY 897 MHZ	ADY SYSTEM	M. C'LINDER NUMBER OF MODI: 12 A THOUSE TU METER POLARISATION: VERTICAL VERTICAL	ENGINE AT THE ANTENNA A1 IN DB, O DB = FREE SPACE ANTION ANGLE IN DEGREES ATTON ANGLE IN DEGREES	DAT (4) DAT (5) 200 .250	GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	.0 3.3 · · · 3 # 52		7 3.3	-1.3 3.2 · · · · · · · · · · · · · · · · · · ·	-3.1 2.8 · · · · · · · · · · · · · · · · · · ·		-5.9 .9				2.0 -2.4 51. 423 .				2.6 3.1	3 1.9 2.8 · · · · · 2 3.45 · ·	3 .2 1.5 . 1 2 3 45	0 - 0 . 0 . 0 . 1 . 2 3 4 5	1 -3.2 -1.7 . 1 .2 3 45 .		3 -8.6 -6.8 1 2 3.45			t -13.6 -12.3 . 2 34 5	t -12:6 -11.4 . 2 34 5
ADIATION PATTERN FREQUENCY 897 MHZ	NA-BODY SYSTEM	TT.S.M.CTLINDER NUMBER OF MODI: 12 2 M 2 M DETAILSATION: VERTICAL/VERTICAL	STRUCTURE ATTENDED AT IN DB, O DB = FREE SPACE STRUCTURE AT THE ANTENNA AT IN DB, O DB = FREE SPACE ROTATION ANGLE IN DEGREES	[3] DAT(4) DAT(5) 150 .200 .250	I(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB			-3.57 3.3		-1.3 -3.1 2.8 4 3. 52 .		1.2 -5.9 -9 · · · · · · 512 · ·		<u>3.0 -1.9 -3.8</u> 5 4 1 32 .	3.6 .9 <u>14.3</u> 5 1.4 23 .	3.7 2.0 -2.4 · · · 51. 423 ·	3.4 3.1 1.1 1 .524	3.1 3.4 2.1 · · · · · · · · · ·		1.2 2.6 3.1 1 2.35 .	-7 1.1 2.2   · · · 1 2 3.45 ·	-1.8 .2 1.5 . 1 2 3 45		-5.4 -3.2 -1.7 . 1 .2 3 45 .		-10.8 -8.6 -6.8 1 2 3.4 5 ·	-13.1 -10.9 -9.3 · 2 3 4.5 ·	-15.9 -14.0 -12.6 2 3 45	-15.4 -13.6 -12.3 . 2 34 5	14.4 -12.6 -11.4 2 3 34 5
RADIATION PATTERN FREQUENCY 897 MHZ	NTENNA-BODY SYSTEM	E ROT.SM.CHLINDER NUMBER OF MODI: 12 Bar. 25 M BORN FILVERER IN REFERSATION: VERTICAL/VERTICAL	LEAD STREAMS AT THE ANTENNA A1 IN DB, 0 DB = FREE SPACE TIAL REPORTION ANGLE IN DECREES	DAT(3) DAT(4) DAT(5) 1 .150 .200 .250	GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3 4.4 .0 3.3				1 -1.3 -3.1 2.8 · · · · · · · · · · · · · · · · · · ·	2 - 4 - 1-3 - 2.4 · · · · · · · · · · · · · · · · · · ·				3 3.6 .9 ±.3 5 1.4 23 .	9 3.7 2.0 -2.4 · · · 51.423 ·				5 1.2 2.6 3.1 1 2.3.5		8 -1.8 .2 1.5 . 1 2 3 45 .	0 -2.99 .6 1. 2 3 4.5	6 -5.4 -3.2 -1.7 · 1 .2 3 45 · ·		0 -10.8 -8.6 -6.8 1 2 3.4 5 · ·	3 = 13.1 = 10.9 = 9.3 · · 2 3 4.5 · · · ·	8 -15.9 -14.0 -12.6 2.3 45	2 -15.4 -13.6 -12.3 . 2 34 5	2 -14.4 -12.6 -11.4 2 345
AL RADIATION PATTERN FREQUENCY 897 MHZ	AL ANTENNA-BODY SYSTEM	INTE ROT.SW.CTLINER NUMBER OF MODI: 12 METRI: .25 M METRI: .26 M	ENTERPONT INTERCAT MATERIA AI IN DB, O DB = FREE SPACE IN FILLD STREAMENT AT THE ANTENNA AI IN DB, O DB = FREE SPACE IZONTAL ROTATION ANGLE IN DECREES	T(2) DAT(3) DAT(4) DAT(5) 100 .150 .200 .250	I(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3.8 -4.4 .0 3.3 3 4 52	3.8 -4.30 3.3 · · · · · · · · · · · · · · · · · ·	3.9 -3.57 3.3	4.02.91.3 3.2 · · · · · · · 52 · · · · · · 52 · · · ·	4.1 -1.3 -3.1 2.8 · · · 4 3. 52 ·	4.2 - 4 - 4.3 2.4 4 9.2 C 4.2 4. 3.5 2	4.2 1.2 -5.9 .9 4 .512 .		3.9 3.0 -1.9 -3.8 · · 5 4 1 32 ·	3.3 3.6 .9 4.3 5 1.4 23 .	2.9 3.7 2.0 -2.4 · · · 51.423 ·				-1.5 1.2 2.6 3.1 1 2.3.5	-2.5 .3 1.9 2.8 2 3.45	4.8 -1.8 .2 1.5 . 1 2 3 45	-6.0 -2.99 .6 . 1. 2 3 4.5	-8.6 -5.4 -3.2 -1.7 . 1 .2 3 <sup>45</sup> .		-14.0 -10.8 -8.6 -6.8 1 2 3.4 5	-16.3 -13.1 -10.9 -9.3 · 2 3 4.5 · · ·	-16.6 -15.6 -12.6 .2 3 45	-18.2 -15.4 -13.6 -12.3 . 2 34 5	-17.2 -14.4 -12.6 -11.4 . 2 34 5
THAL RADIATION PATTERN FREQUENCY 897 MHZ	SIONAL ANTENNA-BODY SYSTEM	INFINITE ROT.SW.CTLINDER NUMBER OF NODI: 12 DIAMETER:	ANTENNATIONAL PLATING AT THE ANTENNA AI IN DB, O DB = FREE SPACE ANN: FIELD STRUCH AT THE ANTENNA AI IN DB, O DB = FREE SPACE HORIZOWTAL ROTATION ANGLE IN DECREES	) DAT(2) DAT(4) DAT(4) DAT(5) 0 .100 .150 .200 .250	) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3 3.8 -4.4 .0 3.3 3 4 52	3 3.8 ±4.3 ±0 3.3 · · · · · · · · · · · · · · · · · ·		0 4.0 -2.9 -1.3 3.2 · · · · 3.4. 52 · · · · · · 52 · · · · · · · · 52 · · · ·	7 4.1 -1.3 -3.1 2.8 · · · 4 3. 52 ·	4     4     5     5     6       1     4     2     4     5     1	8 4.2 1.2 -5.9 -9 4 .512		5 3.9 3.0 -1.9 -3.8 · · 5 4 1 32 ·		3 2.9 3.7 2.0 -2.4 · · · 51. 423 ·		8 1.1 3.1 3.4 2.1 · · 1 .254 · ·		8 -1.5 1.2 2.6 3.1 1 2.35		3 4.8 -1.8 .2 1.5 . 1 2 3 45	5 -6.0 -2.99 .6 . 1. 2 345 . 8 -7.2 -4.1 -2.05 . 1 . 2 345 .	1 -8.6 -5.4 -3.2 -1.7 . 1 .2 3 45 .		5 -14.0 -10.8 -8.6 -6.8 1 2 3.45	.8 -16.3 -13.1 -10.9 -9.3 . 2 3 4.5		4 -18.2 -15.4 -13.6 -12.3 . 2 34 5	4 -17.2 -14.4 -12.6 -11.4 2 2 34 5.
MUTHAL RADIATION PATTERN FREQUENCY 897 MHZ	MENSIONAL ANTENNA-BODY SYSTEM	DOY: INFINITE ROT.SYM.CALINDER NUMBER OF MODI: 12 DIAMETRI:	: ANLENDATIONAL PLANENCH A. THELENDA AI IN DB, O DB = FREE SPACE : ANN' FIELD STRUCTH AT THE ANTENNA AI IN DB, O DB = FREE SPACE : HORIZONTAL ROTATION ANGLE IN DEGREES	rr(1) DAT(2) DAT(3) DAT(4) DAT(5) .050 .100 .150 .200 .250	AI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	3.3 3.8 -4.4 .0 3.3 · · · 3 4 52	3.3 3.8 ± 3 ± 0 3.3 · 0 3.3 · · · · 3 ± 52 · ·	3.1 3.9 -3.57 3.3	3.0 4.0 -2.9 -1.3 3.2 · · · · · · · · 52 · · · · · · 52 · · · ·	2.7 4.1 -1.3 -3.1 2.8 · · · 4.3.52 ·	2.1 4.2 -4 -4.3 2.4 · · · · · 4 3.5 · · · 2.1 4.2 · 4 -5.4 1.8 · · · · 4 3.52 · ·	1.8         4.2         1.2         -5.9         .9         . <th< td=""><td></td><td>.5 3.9 3.0 -1.9 -3.8 5 4 1 32 ·</td><td>-1 3.0 3.4490 7 3.3 3.69 -4.3 5 1.4 23 .</td><td>-1.3 2.9 3.7 2.0 -2.4 51.423 ·</td><td></td><td>-3.8 1.1 3.1 3.4 2.1 · · 1 .254 · ·</td><td></td><td>-6.8 -1.5 1.2 2.6 3.1 1 2.35</td><td></td><td>-10.3 -4.8 -1.8 .2 1.5 . 1 2 3 45 .</td><td>-11.5 -6.0 -2.99 .6 · 1 . 2 3 4.5 · 1 . 2 3 4 5 · 1</td><td>-14.1 -8.6 -5.4 -3.2 -1.7 . 1 .2 3 45 .</td><td></td><td>-19.5 -14.0 -10.8 -8.6 -6.8 1 2 3.4 5</td><td>-21.8 -16.3 -13.1 -10.9 -9.3 · 2 3 4.5 · ·</td><td>-24.0 <math>-16.6</math> <math>-15.6</math> <math>-116.6</math> <math>-12.6</math> <math>-2.345</math> <math>-2.6</math> <math>-2.61</math> <math>-2.</math></td><td>-23.4 -18.2 -15.4 -13.6 -12.3 . 2 34 5</td><td>-22.4 -17.2 -14.4 -12.6 -11.4 . 2 34 5</td></th<>		.5 3.9 3.0 -1.9 -3.8 5 4 1 32 ·	-1 3.0 3.4490 7 3.3 3.69 -4.3 5 1.4 23 .	-1.3 2.9 3.7 2.0 -2.4 51.423 ·		-3.8 1.1 3.1 3.4 2.1 · · 1 .254 · ·		-6.8 -1.5 1.2 2.6 3.1 1 2.35		-10.3 -4.8 -1.8 .2 1.5 . 1 2 3 45 .	-11.5 -6.0 -2.99 .6 · 1 . 2 3 4.5 · 1 . 2 3 4 5 · 1	-14.1 -8.6 -5.4 -3.2 -1.7 . 1 .2 3 45 .		-19.5 -14.0 -10.8 -8.6 -6.8 1 2 3.4 5	-21.8 -16.3 -13.1 -10.9 -9.3 · 2 3 4.5 · ·	-24.0 $-16.6$ $-15.6$ $-116.6$ $-12.6$ $-2.345$ $-2.6$ $-2.61$ $-2.$	-23.4 -18.2 -15.4 -13.6 -12.3 . 2 34 5	-22.4 -17.2 -14.4 -12.6 -11.4 . 2 34 5
ZIMUTHAL RADIATION PATTERN FREQUENCY 897 MHZ	O-DIMENSIONAL ANTENNA-BODY SYSTEM	STBODY: INFINITE ROT.SW.CTLINDER NUMBER OF MODI: 12 DIAMEDER: 25 M KORV.CTLINDER POLARISATION: VERTICAL/VERTICAL	11(1) : ANIEMANG-DOLUZISTANCH AL MULLENNA AI IN DB, O DB = FREE SPACE III : HORIZONTAL ROTATION ANGLE IN DEGREES II : HORIZONTAL ROTATION ANGLE IN DEGREES	DAT(1) DAT(2) DAT(4) DAT(4) DAT(5) .050 .100 .150 .200 .250	II GAI(1) GAI(2) GAI(3) GAI(4) GAI(5) -20 DB -10 DB + 0 DB +10 DB	0 3.3 3.8 -4.4 .0 3.3 3 4 52	5 3.3 3.8 H.3 +.0 3.3 · · · · · · · · · · · · · · · · · ·	15 3.1 3.9 -3.5 -7 3.3 · · · · · · · · · · · 52 · ·	20 3.0 4.0 -2.9 -1.3 3.2 · · · · · · · · · · · · 22 · · · · ·	30 2.7 4.1 -1.3 -3.1 2.8 4.3 52 ·	35 2.4 4.24 -4.3 2.4 · · · · 4 3.2 C · · · · 4 3.5 C · · · · · · · · · · · · · · · · · ·	45 1.8 4.2 1.2 -5.9 .9 4 .512 .	56 1.0 4.0 2.5 -3.7 -2.0 45 .1 32 .	50 5 3.9 3.0 -1.9 -3.8 · · 5 4 1 32 ·	707 3.3 3.6 .9 -1.3 5 1.4 23 .	75 -1.3 2.9 3.7 2.0 -2.4 · · · 51. 423 ·	86 -2.9 1.8 3.4 3.7 1.1 · · · · 1.524 · ·	<u>50</u> -3.8 1.1 3.1 3.4 2.1 · · · 1 .254 · ·		05 -6.8 -1.5 1.2 2.6 3.1	10 -8.0 -2.5 .3 1.9 2.8 2 2.45 2 3.45	20 -10.3 -4.8 -1.8 -2 1.5 . 1 2 3 45 .	25 -11.5 -6.0 -2.99 .6 . 1.2 3 4.5 . 30 -12 8 -7.2 -4.1 -2.05 . 1 .2 3 4 5 .	35 -14.1 -8.6 -5.4 -3.2 -1.7 · 1 .2 3 45 · ·	40 - 15.7 - 10.1 - 5.9 - 14.7 - 5.6 - 14.8 - 1 2 3 45	50 -19.5 -14.0 -10.8 -8.6 -6.8 1 2 3.4 5	25 -21.8 -16.3 -13.1 -10.9 -9.3 . 2 3 4.5	661 - 24.0 - 16.8 - 15.9 - 14.0 - 12.6 2.3 45	70 -23.4 -18.2 -15.4 -13.6 -12.3 2 34 5 · · ·	10         -22.4         -17.2         -14.4         -12.6         -11.4         23.4         5.
## 16.2.2. PROGRAM HARRA AND OUTPUT SAMPLE

E PROGRAM "HARRA", COMPUTATION OF THE Y-MATRIX		READ(1,53)(ZH(I),I=1,NP)	HARRA 52
1. 李雯宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇宇	HARRA 1	53 FORMAT(10F8.4)	HARRA 53
E CONTROL CARDS (START RUN)	HARRA 2	TO WRITE(3,54) NN,NP,NPHI,BK,F	HARRA 54
TAN, 3D/11, CM/COUCH, CI 240.	HAKKA 3	74 FUNDAL(IA//* NN=*,12,* NT=*,12,* NTILE*',12,* DAF*',5,14./, 1 * FREGENCY - * FG 2 * M47*)	HARRA 23
	HARRA 5	55 FORMAT(1X/* RH*)	HARRA 57
	HARRA 6	WRITE(3, 55)	HARRA 58
PUBLIC, COLLECT.	HARRA 7	WRITE(3, 46)(RH(I), I=1, NP)	HARRA 59
COLLECT, N=AMAT2.	HARRA 8	46 FORMAT(1X, 10F8.4)	HARRA 60
CALALUL, AMAIZ. P HEDE END OF DEMODI MADD	HAKKA 9	WALLE(3,50) 56 FORMAT(1Y/# 7H#)	HAKKA 01 HARRA 63
E CONTROL CARDS (FOT LOUTING RINS)	HARRA 11 HARRA 11	WRITE(3, 46)(ZH(I), I=1, NP)	HARKA 63
PAN, 3571, CM70000, CT240.	HARRA 12	IF((RH(1)-RH(NP)).NE.00R.(ZH(1)-ZH(NP)).NE.0.) GO TO 58	HARRA 64
ATTÁCH, AMATZ, PW=3571	HARRA 13	RH (NP+1)=RH(2)	HARRA 65
"NL	HARRA 14	ZH(NP+1)=ZH(2)	HARRA 66
	HARRA 15	NS(NF+Z)=KH(3) 21(AP 0) 21(AP 0)	HAKKA O
TUBLIC, CULLECT.	HARRA 16 UAPPA 17	CTU(NY+Z)=CH(S) CP=UP_C	HARRA OC
R HERE END OF RECORD CARD	HARRA 18	58 D0 57 I=2.NP	HARRA 70
	HARRA 19	I2=I-1	HARRA 71
PROGRAM HARRA (INPUT, OUTPUT, DISK, TAPE1=INPUT, TAPE3=OUTPUT,	HARRA 20	RR 1=RH(I)-RH(I2)	HARRA 72
1TAPE6=DISK)	HARRA 21	RR2=ZH(I)-ZH(I2)	HARRA 73
	HARRA 22	DH(IZ)=SQRT(RR1#RR1#RR2#RR2)	HARRA 72
C SOURCE PROGRAM IS DESCRIBED IN REPORT "RADIATION AND SCATTERING	HARRA 23	(21)HZ+(1)HZ+(1)HZ+(1)Z	HARRA 7
C FROM BODIES OF REVOLUTION* BY R.F. HARRINGTON AND J.R. MAUTZ, JULY	HARRA 24	N(12)="2"(NH(12)) N(12)-1004/PUT(12))	HAKKA 70
U 1909, ELECTRICAL ENG. LEFT., STRACUSE UNIV., STRACUSE, N.Y.	HAKRA 25	(CT/DA)-DB2/AU/1721	HANNA ()
C POULY LCATIONS BY P.A. NEUKOMM AT BIOMECHANICS LABORATORY, SAISS FEDE-	HARRA 26	57 CONTINIE	HARRA /C
C PITTER OF THE RTH_7. THE MONTERCETTONS IN THIS DECOLLED ON INE VUS UN-	12 AAAAA	KG=NP-1	HARRA 80
C MAINLY THE CONVERSION FROM IBM TO CDC STANDARD AND FILE GENERATION.	HARRA 29	N=KG/2	HARRA 81
C THE RESULTS OF THIS PROCRAM ARE THE Y-ADMITTANCE-MATRICES FOR THE	HARRA 30	XX=X-1	HARRA 82
C MODI O UP TO KK FOR A GIVEN BODY AND A GIVEN FREQUENCY. IN THIS CASE	HARRA 31		HARRA 8
C THE BODY IS MANMOD2, THE FREQUENCY 164 MHZ, THE MAXIMUM MODE KX IS 7	7 HARRA 32		HAKKA 54
V ANU THE RESULTS ARE STORED IN THE FILE STORMOT IN THE CULLECT FILE	HARKA 33 UADDA 20		UARDA OC
C ATTENTION: SCALING IN THE PROCRAM. PHYSICAL DIMENSIONS IN TIMES	HARRA 34	M5=NN+2	HARRA
C ACTUAL SIZE, FREQUENCY 10 TIMES LOWER THAN ACTUAL.	HARRA 36	M6=NN+4	HARRA 80
	HARRA 37	Pre-NN	HARRA 89
	HARRA 38	FM2=NN#NN	HARRA 90
COMPLEX A3, A4, Z, (CS (40), G, U	HARRA 39	FL=3.141593	HAKKA 91
CUMPLEX 61,62,63,17(50) COMMON//6(500) 7/1600)	HAKKA 40	LIA≚5(0./U/ DP≞PT /NPHT	HARRA 9
DIMENSION RH(43).ZH(43).ZH(43).TJ(20)	HARRA 42	CA=BK * BK * ETA	HARRA 94
DIMENSION SV(42), CV(42), 2S(42), R(42), ANG(40), AC(40), CSM(120)	HARRA 43	CQ=ETA	HARRA 95
DIMENSION TP(80), T(80), TR(80), JK(4)	HARRA 44	SS=0.	HARRA 96
REWIND 6	HARRA 45	100 117 I=1,NM	HARRA 97
U=(U+1+) En dean (1 E1) with unburger dy F	HARRA 40	1   = Z * ( 1 = 1 ) + 1 T 2= T 1 ± 1	HAKKA 90
51 FORMAT(313,E14,7,F8,2)	HARRA 41	S=SS+DH(I1)+DH(I2)	HARRA 100
IF(EOF(1)) 52,49	HARRA 49	TJ(I)=SS=(1)/T	HARRA 10
49 CONTINUE	HARRA 50	117 CONTINUE	HARRA 102
READ(1,53)(RH(I),I=1,NP)	HARRA 51	WKIIE(3,118)	HAKKA TU

118 FORMAT(1X/" TJ") WRITE(3_46) (TJ(I).1=1.NN)	HARRA 104 HARRA 105	DO 68 MM=1,3 M1=MM-1	HARRA 15 HARRA 15
DO 2 J=1, NPHI ANC(1)= (1_ 5)#nP	HARRA 106 HABPA 107	CH-JN48 N=tH IHdN81 N=tH	HARRA 15
AC(J)=COS(ANC(J))	HARRA 108	G(M2)=0.	HARRA 160
2 CONTINUE M3-D	HARRA 109 UA DDA 110	DO 13 K=1, NPHI V3_V.Hh	HARRA 16
DO 10 MM #15. H6	HARRA111	N.C=N.H14 G (M2)=G (M2)+CS (K)=CSM(K2)	HARRA16
M1=M-3	HARRA 112	13 CONTINUE	HARRA 16
M2H43#NPHI Do 11 K -1 NBH7	HARRA113	68 CONTINUE	HARRA 165
	HARRA115 HARRA115	16 CONTINUE	HARRA 160
CSH(K1)=DP#COS(M1#ANG(K))	HARRA 116	W. 1 - 1 the Decision of the D	HARRA 16
11 CONTINUE	HARRA 117	J2=2#(J=1)+1	HARRA 16
ID CONTINUE	HARRA 118 UADDA 110	J3=J2+1 .11-12-1	HARRA 170
D0 16 J=1,KG	HARRA 120	J5⊒J4+1	HARRA 172
DEL=.5+DH(J)	HARRA 121	J6=4 *(.J-1)+1	HARRA17
UELT=UE(J)*OK	HARRA 122	J7±J5+1	HARRA 17
D0 17 I=1.KG	HARRA 123 HARRA 124	10=18+1 19=18+1	HARKAT7
z3=zs(j)-zs(i)	HARRA 125	DEL1=DH(J2)+DH(J3)	HARRA 17
RR1±SV(J)#R(J)+CV(J)#Z3	HARRA 126	DEL2=DH(J4)+DH(J5)	HARRA 178
RR2=-SV(J)#R(I) DD2 D1124140411 D124141 1241	HARRA 127	TP(J6)=DH(J2)/DEL1	HARRA 179
RRJSER(U J=R(U J=R(L J=R(L J=L)"C) RRB2 #R(1)#R(T )	HAKKA 128 HABBA 120	IP(J7)=DH(J3)/DEL1 TP(J8)DH/J4)/DEL1	HARRA 180
X1=ABS(R(J)-R(I))+ABS(Z3)	HARRA 130	TP(J9)==DH(J5)/DEL2	HARRA 182
DO 5 K=1, NPHI	HARRA 131	T(J6)=DH(J2)*DH(J2)/2,/DEL1	HARRA 18
IF(K. NE. 1. CR. X1. NE. 0.) CO TO 7	HARRA 132	T(J7)=DH(J3)*(DH(J2)+DH(J3)/2.)/DEL1	HARRA 18 <sup>1</sup>
X=R(J)=DP VV-SCAPT (NET #NET .V#V)	HARRA 133	T(J8)=DH(J4)#(DH(J5)+DH(J4)/2,)/DEL2	HARRA 185
MI=(X#ALOG((DEL+XX)/X)+DFL#ALOG((X+XX)/DFL))/AA	HARRA 134 HARRA 135	211 CONTTANTS THE CONTTANTS THE CONTTANTS	HARRA186
W2=-1.	HARRA136	DO 75 J=1.MM	HARRA 189
GS(1)=W1+U#W2	HARRA 137	TR(J)=T(J)	HARRA 18
	HARRA 138	75 CONTINUE	HARRA 190
/ I=AUS(KM1+AUC(K)=KKZ) DA_DD3.DD#846/VV1	HARRA 139 HARRA 139	15 IF((ZH(1)-ZH(NP-2)).EQ.0AND.(RH(1)-RH(NP-2)).EQ.0.) GO TO 78	HARRA 19
RK=BK#SORT(RD)	HARRA 140 HARRA 141	TT DEL 1=DH(1)+DH(2)	HARRA 19
D2=RD-Y #Y	HARRA 142	TR(1)=DH(1)#(1.+(DH(2)+DH(1)/2,)/DEL1)	HARRA 194
Y1=Y-DEL	HARRA 143	TR(2)=DH(2)#(1.+DH(2)/2./DEL1)	HARRA 195
Y2=Y+DEL B1-Sourtense	HARRA 144	23 IF(RH(NP)) 79,78,79	HARRA 196
N 1224RT (Y28Y24D)	HAKKA 145 HARPA 116	(y u = 1 ≤ 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1	HARRA 197
IF(Y1) 72, 73, 73	HARRA 147	DEL2=DH(NP-2)+DH(KG)	HARRA 190
72 TIN-ALOG((-Y1+R1)*(Y2+R2)/D2)	HARRA 148	TR(J1)=DH(NP-2)*(1.+DH(NP-2)/2./DEL2)	HARRA200
10 20 20 20 20 20 20 20 20 20 20 20 20 20	HARRA 149 HARPA 150	16 TK(JZ)=DH(KG)=(1.+(DH(NP-2)+DH(KG)/2.)/DEL2) 78 no 30 1-1 w	HARRAZO
25 SH-SIN(RK)	HARRA 151	JL=(J=1)*M42	HARRA20
CS=COS(RK)	HARRA 152	J3=(J-f)#4	HARRAZO
GS(K)=(CS-U=SN)=(TIN-U=(EK=DH(J)-EKT=TIN))/DELT 5 CONTINUE	HARRA 153 Harra 154	J1=Z*(J-1) D0 31 T=1 NM	HARRA205
M3=(J-1)"KG+I	HARRA 155		HARRAZOT

HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO HARRAZO	HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZYO HARKAZO	HARKA 304 HARKA 304 HARKA 304 HARKA 305 HARKA 305 HARKA 307 HARKA 311 HARKA 311 HARKA 311
C SUBROUTINE LIMEQ(LL, C) E SUBROUTINE LIMEQ(LL, C) SUBROUTINE LIMEQ(LL, C) CONTINE X (C), STOR, STO, ST, S DD 20 20 1=1, LL DD 20 CONTINUE 20 CONTINUE 20 CONTINUE	D 18 M=1,LL K=4 D 2 I=4,LL K1=41.4 K2=41.4 K2=41.4 K2=41.4 K2=41.4 K2=41.4 K2=41.4 K1=1.4 K	C(K1)=C(K2)*ST 1 = 1 = 1. 1 CONTINUE 11 CONTINUE 11 CONTINUE 11 CONTINUE 12 CONTINUE 12 CONTINUE 13 CONTINUE 14, 8, 14 15 (J-LR(J)) 14, 8, 14
HARRA208 HARRA209 HARRA210 HARRA211 HARRA213 HARRA213 HARRA213 HARRA215 HARRA215	HARRA218 HARRA220 HARRA220 HARRA221 HARRA221 HARRA223 HARRA223 HARRA223 HARRA223 HARRA223 HARRA223 HARRA230 HARRA231 HARRA231 HARRA233 HARRA231 HARRA331 HARA331 HARA	HARRASSU HARRASSI HARRASSI HARRASSI HARRASSI HARRASSS HARRASSS HARRASSS HARRASSS
L2=L1.4MM L2=LM1WM2-L1 L4=L3+MM Z(L1)=0. Z(L2)=0. Z(L2)=0. Z(L2)=0. Z(L2)=0. Z(L2)=0. Z(L2)=1. Z(L2)=1.24 Z(L2)=1.24	22=11+4J 17=23+11 D0 71 III=1,4 IT=211+II IT=21+11 IT=21+11 IT=21+11 IT=21+11 J==(22-1)*KG+12 J==(22-1)*KG+12 J==2F(G(J6)+G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==5F(G(J6)-G(J4)) A==2F(J6)-G(J7)-G(J6)-G(J7)-G(J7)-G(J7)-G(J7)-F(J7)-	R COMMAT(1X* T=,11) DD 92 T=1,M K2=K1+M-1 K2=K1+M-1 K1=K1+MC F COMTINUE 3 CONTINUE 3 CONTINUE 3 CONTINUE 3 CONTINUE 3 CONTINUE 3 CONTINUE 3 CONTINUE

-							-	-							-	-			0																			
15.000	뵤	0.500	100	0.760	1.770	6.510	11.820	15.000	Ħ	0.50	0.710	0.760		1.270	11.820	15.000			T UP T		2H	0.500	0.710	992.0	200	1.270	6.510	11.80										
14.700	16.4 M	0.515	1.085	0.620	0.760	5.960	11.340	14.700	16.4 M	0.515	0.660	0.620		0,760	11.380	14.700			DATA SE		16.4 M	0.515	0.660		0.3U.S	0.760	5.960	11.340		0			N CARD					
14.550	008E+00	0.515	200.1	0.490	0.350	5.50	10.770	14.550	308E+00	0.515	0.625	0.1100		0.350	10.770	14.550			PLETE 1		SOR -DO	0.515	0.625		064.0	0.350	2.50	10.770	DCC **!	CORD CAR	R0.		FORMATIO					
14.310	0.34348	0.50	08.0	0.480	0.00	1.840	10.270	14.310	0.34348	0.50	800	200 200 200 200 200		0000	10.270	14.310			DE COM		10 221205	005.0	0,60		00 <b>-</b> • 0	0.00	4 840	10.270	14.510	AD OF RE	NOT, DISK		ND OF IN					
13.920 16.800	003041 20	0.000	0.980	0.630	000.0	4.260	9.740	13.920	004041 20	000.0	0.575	0.630	0000	0.000	9.740	13.920	10*01		EACH MC		007001 OC	00.00	0.575	005.0	000.0	0.000	4.260	9.740	15.800	R HERE EN	ADD, S164h	END	R HERE EN					
ARRA312 Arra313	ARRA314	ARRA315 Appa316	ARRA317	ARRA318 ARRA319	ARRA 320	ARRA 321	IARRA 322	IARRA 323 IAPPA 324	ARRA 325	IARRA 326	IARRA 327	IARRA 328 IARRA 329	IARRA330	LARRA331 LARRA332		0.525	1.000	0.500	3 KRU	9.200	13.510	16.680		0.525	1.000	0.850	0,05.0	3.680	9.200	13.510	16.680		0022 0007	0.850	0.500	2 680	0.200	210-01
ΞΞ	Ŧ	<b>x</b> 3	- 20					<b>-</b>		-	-		-		-	0.525	06.0	0.700	120	8.700	13.190	16.420		0.525	1.050	0.930	00/-0	3.120	8.700	13.190	16.420		0.525	0.930	0.700	120	200	2
																0.515	88	80 80 80	2 580	8.240	12.820	16.050		0.515	1.050	88	0.820	2.580	8.240	12.820	16.050		0.515	89	0.820 0.820	000 0	8.200	12,000
																0.515	52.0		084 C	7.630	12.550	15.700		0.515	0.925	8	0.870	2.180	7.630	12.550	15.700		0.515		0.870		2.630 2.630	14.200
																0.500	0.780	0.850	. 700	090	12.270	15.400		0.500	0.780	66.1	0.850	1.740	7.060	12.270	15.400		0.50		0.850			12.610
															2	0.500	0.710	0.760		6.510	11.820	15.000	6	0.500	0.710	-1 81	0.760	1.270	6.510	11.820	15.000	5	0.500		0.760	1	6.510	11.65
															16.4 MH	0.515	0.660	- 650 - 650	076 0	200	11.340	14.700	16.4 MH	0.515	0,660	- 085	0.620	0.760	2.960	11.340	14.700	16.4 MF	0.515		0.620		200	<u> </u>
Н						-		5	14.8.14				*	ORD CARD	085-400	0.515	0.85 83	021.0	2		10.770	14.550	085400	0.515	0.625	1.020	0.490	0.750	5.500	10.770	14.550	308E+00	0.515		0.190		- - - - - - - - - - - - - - - - - - -	017.01
LR (J)	1 I-1 E	19 19	Ŧŝ	()=C(K1)		D-1 B C B.	RJ)=LRJ	1)=LR(LR		1+1	LINUE	JRN	*******	OC REC	0. 34348	0.500	0.00	286 863 863	5		10.270	14.310	0. 241348	0.50	0.00	0.980	0.480	000	4, BHO	10.270	14.310	0.34348	0	3	8 8 8 8 9 9		0.80	10.2/0
14 LRJ= J2=(	21 D0 1	K2=1						LRU	IF(J	8 J1=1	9 CON1	EN ON	·******	R HERE ENI	0000M1 20	0.000	0.575	0.630	0.00	1.260	0#7.6	13.920	001041 20	0.000	0.575	0.980	0.630		4.260	9.740	13.920	002041 20	0.00		0.630	0.000	0.000 14,260	0#/.*6

.

								m≠10v0r
16.680	0.525 0.850 0.850	3.680 9.200 13.510 16.680	0.525 0.850 0.500	3.680 9.200 13.510 16.680		0.525 0.850 0.850	3.680 9.200 13.510 16.680	HARRA 33 HARRA 33 HARRA 33 HARRA 33 HARRA 33 HARRA 33
16.420	0.525 0.930 0.930 0.700	3.120 8.700 13.190 16.420	0.525 0.525 0.930 0.700	3.120 8.700 13.190 16.420		0.525 0.930 0.930 0.700	3.120 8.700 13.190 16.420	
16.050	0.515 1.050 1.820 0.820	2.580 8.240 12.820 16.050	0.515 1.050 0.820 0.820	2.580 8.240 12.820 16.050	- 7	0.515 1.050 1.000 0.820	2.580 8.240 12.820 16.050	
15.700	0.515 0.925 1.040 0.870	2.180 7.630 12.550 15.700	0.515 0.925 1.040 0.870	2.180 7.630 12.550 15.700	ode nn=	0.515 0.925 1.040 0.870	2.180 7.630 12.550 15.700	
15.400	0.500 0.780 0.850	1.740 7.060 12.270 15.400	0.500 0.780 1.070 0.850	1.740 7.060 12.270 15.400	LAST M	0.500 0.780 0.850	1.740 7.060 12.270 15.400	
15.000	z 0.500 1.100 0.760	1.270 6.510 11.820 15.000	z 0.500 0.710 0.710 0.760	1.270 6.510 11.820 15.000	01 dn .	z 0.500 1.100 0.760	1.270 6.510 11.820 15.000	
14.700	16.4 MH 0.515 0.660 1.085 0.620	0.760 5.960 11.340 14.700	16.4 MH 0.515 0.660 1.085 0.620	0.760 5.960 11.340 14.700	ata set	16.4 MH 0.515 0.660 1.085 0.620	0.760 5.960 11.340 14.700	CARD
14.550	08E+00 0.515 0.625 1.020 0.490	0.350 5.500 10.770 14.550	08E+00 0.515 0.625 0.490 0.490	0.350 5.500 10.770 14.550	PLETE D	08E+00 0.515 0.625 1.020 0.490	0.350 5.500 10.770 14.550	CRD CAR
14.310	0.34348 0.500 0.980 0.480	0.000 4.840 10.270 14.310	0.34348 0.500 0.600 0.480 0.480	0.000 4.840 10.270 14.310	DE COM	0.34348 0.500 0.980 0.480	0.000 4.840 10.270	D OF REC O7, DISK,
ស្តី	82282 82282 82282 82282 8282 8282 8282	8882288	8822 8822 8822 8822 8822 8822 8822 882	888288	CH MQ	11 20 20 20 20 20 20 20 20 20 20 20 20 20	000 00 00 000	.800 ST64M I ST64M

.3434809E+Jn FREGENCY = 16.40 MHZ NN= 0 NP= 41 NPHI= 20 BK=

RH 0.

results fro	only mode
.5250 1.0000	.8500 .5000
.5250 1.0509	• 7000 • 7000
.5150 1.1500	1.0000
.5150 .9250	1.0400
.7890	1.0700
.7110	.7623
.5150	1.0359 .620f
.5150	0020°T
. 5000 . 6000	0087*
0.0000 .5750	0.0000

NN = 0 SHOWN HERE

**M HARRA** 

ž

0.0666 6.0000 .3549 .7599 1.2779 1.7409 2.1800 2.5599 3.1200 3.6800 4.2600 4.8400 5.5009 5.5600 6.5197 7.0600 7.6500 8.7400 8.7600 9.2000 9.7400 10.47709 11.7700 11.8500 11.8200 12.5590 12.8200 13.1900 13.5100 13.9200 14.3100 14.5500 14.7500 14.7000 15.4000 15.7706 16.0500 16.4200 16.6500

2

.8523 1.7705 2.6808 3.6209 4.7570 6.0041 7.0177 8.1502 9.2429 10.2058 11.3174 12.3713 13.1639 13.7534 14.5485 15.2066 15.7362 16.4463 17.1894

2810E-04	2418E-04	.2660E-04		5013E-04	5463E-04	4518E-04		5105E-04	7508E-04	. 5325E-04		4571E-04	-+9065E-04	.4838E-04		7752E-04	9853E-04	3077E-04		7809E-04		• 5896E-06		1031E-03	7528E-04	2823E-04		1232E-03	3938E-04	6030E-04
.7697E-04	- 3335E-04	7610E-04		.1673E-C3	6739E-04	1520E-03		.2233E-03	8369E-04	- •2119E-03		*2625E-03	896804	2426E-03		-2778E-03	8114E-C4	24572-03		.2561E-03	5371E-C4	2088č-03		-2105E-03	1989E-04	1504E-03		.1329E-03	.2758E-04	58936-04
9309E-05	3697E-04	• 2 2 9 5 E - 0 4	.8630E-05	7142E-05	7882E-04	•4015E-04	.1647E-04	.3560E-06	1031E-03	•4241E-04	<ul> <li>1936E-04</li> </ul>	.1096E-03	1180E-03	.3552E-04	.1936E-04	4609E-04	1189E-03	.1706E-04	.15585-04	9433E-04	3858E-04	1207E-04	.7634E-05	1221E-03	6394E-04	3850E-04	7550E-06	1319E-03	1790E-04	6629E-04
.7421E-04	5359E-05	7967E-04	1628E-04	.1605E-03	7186E-05	1692E-03	34935-04	.2134E-03	4160E-05	2207E-03	4598E-04	.2496E-03	.2557E-05	2520E-03	5303E-04	.26235-03	•1+33E-04	2542E-03	5431E-04	.2332E-03	• 30 895-0 t	2143E-C3	47015-04	.1937E-U3	.4551E-04	15246-03	34875-04	•1178E-C3	.6191E-04	5614E-04
.1155E-04	-+46825-04	.1634E-04	.2154E-04	.3U21E-04	9597E-04	•2632E-04	.4045E-04	1445E-03	1209E-03	•2462E-04	4665E-04	.1438E-06	1319E-03	.15516-04	.4537E-04	5114E-04	12296-03	2747E-05	•3436E-94	77066-94		2828E-04	.1 302E-04	1165E-03	4212E-34	49466-94	8723E-05	i210E-03	.1961E-03	
.64135-04	•3018E-04	78548-04	47555-04	.13825-03	•6887€-04	16622+03	1017E-03	1832E-03	.959UE-04	21622-03	13355-03	.21355-03	<ul> <li>11315-13</li> </ul>	24586-03	15358-93	.22335-03	.13315-33	24655-03	15655-03	.2021d-03	.13515-93	20562-03	1345E-03	.16205-03	.12505-33	14355-93	98622-04	.95735-14	<ul> <li>10185-03</li> </ul>	46-32224
.2518E-04	4761E-04	•5896E=05	•2621E-04	<ul> <li>1614E-03</li> </ul>	9502E-04	.4934E-05	.482AE-04	.3032E-04	1153E-03	2263E-05	<ul> <li>5446E-04</li> </ul>	7333E-05	1219E-93	13996-04	•5116E-04	5031E-04	19256-03	3079E-04	.35642-04	8221E-04	587PE-04	4951E-04	.7637E-05	9535E-04	.1291E-03	61745-54	1333E-04	961_E-14	4353E-04	6851E-04
.4878E-04	.5337E-n4	7196E-04	67105-04	.104BE-03	.1135E-0X	15126-03	1431E-03	.1335E-03	.1621E-03	1956E-03	18755-03	<ul> <li>1609E-03</li> </ul>	.1946E-73	2210E-03	2150E-03	.16765-03	.21066-03	2133E-03	2133E-63	.1507E-03	.2021E-93	1735E-03	1854E-JJ	.1136E-J7	.17495-03	536(2T	1353E-03	.68335-04	<ul> <li>1243E-03</li> </ul>	31796-04
<ul> <li>1244E-03</li> </ul>	42446-04	1061E-04	.2686E-04	.2554F-04	8221E-04	28n9E-04	.490JE-04	.1124E-04		4311E-04	■ 5464E=04	9384E-05	94655-04	5764E-04	•5C38E-94	27855-04	-•78395-94	70456-04	.37455-04	4186E-04	.8612E-04	7660E-04	.3678E-05	+0-26024	5835E-04	72405-94	25046-04	46225-04	82565-04	5986E-04
 •2244E-04	•6853E-04	5427E-04	7341E-04	•4800E-04	.15015-03	1127E-03	1564E-03	• 6322E-04	•2018E-03	1439E-03	2647E-03	.7320E-04	• 2391E-03	1600E-03	23455-03	• 7583E-04	2557E-03	1546E-03	23785-03	•6764E-64	-2490E-03	1198E-53	20245-03	• 5311E-04	.2018E-03	72140-04	1463E-03	• 2364E-04	•1349C-03	1754E-D5

-.3098E-04 -.3793E-04 -.5870E-04 .1572E-04 -.2907E-04 .2628E-05 .1534E-04 -.9171E-04 -.9865E-04 .2707E-03 -.1023E-03 -.7032E-04 .1719E-04 -.9428E-04 -.2750E-05 -.7530E-04 -.1069E-04 .1713E-04 -.9427E-04 .3377E-04 .3083E-04 -.1024E-03 .2291E-03 .3356E-04 -.1018E-03 .9673E-04 .3581E-04 -.9693E-04 .7446E-04 .3436E-04 -.7561E-04 .2658E-04 -.1189E-03 .... -.2542E-03 .1985E-03 .3539E-03 .2461E-03 .1845E-03 .3370E-03 .2381E-03 .1594E-03 •3741E-04 •7306E-04 .93556-04 .7663E-04 .8112E-04 .1434E-03 .1842E-03 .1550E+03 .1996E-03 -2465E-03 -.1564E-03 .1104E-03 35226-03 -.2181E-03 2097E-03 2641E-03 •2191E-63 .32466-03 -5504E-04 .1437E-04 ... . ī ī . í. -.1099E-04 -.1178E-03 .2613E-03 -.9052E-04 -.2261E-04 .3160E-04 -.2291E+04 -.8485E+04 .1438E+03 • 33445-04 - 81595-04 • 37195-03 • 22795-04 • 74622-04 - 17585-04 • 74622-04 - 17585-04 .4529E-04 -.9068E-04 .1552E-04 .8610E-04 -.9058E-04 .3729E-03 .2542E-03 .7546E-04 -.8503E-05 .5122E-04 -.8379E-04 .1952E-04 -.2455E-63 .1545E-04 .4837E-04 -.9189E-04 .3520E-04 .2336E-05 .5047E-04 .1362E-04 -.7122E-05 .1984E-03 -.9438E-04 --8992E-04 .1501E-04 -.6241E-04 .2641E-04 ..... -.2427E-03 .7683E-04 .3545E-03 +0-3 .3108E-03 .6360E-04 .8602E-04 .1358E-04 -.8953E-04 -.2205E-04 .7271E-04 .3413E-63 .4289E-04 .2200E-03 .76485-04 -.16025-03 .7219E-04 -.2345E-03 -.2145E-03 .6453E-04 --1533E-03 .4535E-C4 .15915-04 -.1550E-04 .26285-05 .8120E-04 • 9363E-04 •6965E-04 --5370E-04 .13601 .....  $\begin{array}{c} 0.4 & 0.51835 - 0.4 & -0.8145 F = 0.4 \\ 0.4 & -0.2955 - 0.4 & -0.8145 F = 0.4 \\ 0.4 & -0.2955 - 0.4 & -0.8145 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5945 F = 0.4 \\ 0.4 & -0.2037 F = 0.3 & -0.5935 F = 0.4 \\ 0.4 & -0.2145 F = 0.3 & -0.5935 F = 0.4 \\ 0.4 & -0.2145 F = 0.3 & -0.5935 F = 0.4 \\ 0.4 & -0.2105 F = 0.3 & -0.5339 F = 0.4 \\ 0.4 & -0.2105 F = 0.4 \\$ • 4657E-04 -.41995-04 -.34355-04 .21825-03 -.34895-04 .13155-03 .26006-03 -.1113E-04 .2716E-04 .19536-04 -.1766E-04 -.3446E-04 -.1031E-03 ..... .75656-04 -.13342-03 -.15735-04 • #2 385 - 3 # --40745-05 --46592-04 ..... 04 -1422E-03 9344E-04 05 -1422E-03 4039E-04 -. 95 -1353E-03 -.6539E-04 -. 14 -2605-03 -.6539E-04 -. 4 - 2605-03 2549F-14 + 20145-05 + -10185-03 + -10185-03 + -28555-03 + -59016-04 + -20266-03 + -55016-04 + -5575-04 + -5575-04 + -5575-04 + -5555-04 .2582E-03 .4046E-04 -.7839E-06 -.7834E-04 -.6394E-14 --2547E-04 .1383E-04 ..... -.5439E-04 .6938E-04 .45485-04 --35+6E-04 --3535E-04 đ •6434E-..... .9264E-04 .2129E-04 .1277E-04 -.5739E-04 -.3644E-04 --98226-04 .25945-04 .7486E-05 -.9310E-04 -.7587--04 8615E-05 .7577E-05 .6544E-05 -10-28-01-..... 256405-03 . .32485-03 . .32485-04 - 72435-04 - 253455-03 . .253455-03 . .253455-03 . .253455-03 . .253455-03 . .253455-03 . -.18576-03 .22976-03 .28526-03 -.4683E-04 -.1340E-03 .1607E-03 .2023E-D3 -.1626E-04 --4750E-04 .5510E-04 .7051E-04 .311CE-04 -.5814E-04 -.5304E-05 ¥20.00.00.

	··· · · ·	1680E-03 -11992E-05 5899E+06 5899E+06 5401E-03 36853E+05	137E-02 .1967E-04	.3365E-02 4719E-04 .1253E-06	6642E-02 .1149E-03 2493E-05 .3457E-02	2009E-03 .4023E-05 1376E-02 .0499E-03	1356E-04 .6325E-03 .6325E-02 .3571E-04 .3571E-04 .260276E-02
		-5030E-05 -2053E-05 -3191E-05 -3638E-05 -2630E-05	.4426E-05 .3030E-05 .7692E-07	.4642E+05 .3412E-05 .2990E+06	.4846E-05 .4672E-05 .6598E-05 .5714E-05	.61076-05 .12546-05 .48296-03 .67206-05	.1619E-05 -7121E-05 -1090E-04 -3211E-05 -3211E-05 -160vE-04 -5425E-05
••••		- + 650E-03 7221E+05 - 1508E-05 - 1406E-05 - 1379E-06 - 1379E-06			-3377E-02 -2607E-03 -3114E-05 -1031E+05	-12016-04 -12016-04 -11826-05 -61476-03	. 3384E-05 . 1526E-05 . 1528E-02 . 5423E-02 . 8481E-02 . 1049E-02 . 11049E-05 . 11049E-05 . 2490E-01 . 2490E-01
		.4970E-05 .4970E-05 .3567E-05 .3264E-05 .133E-06 .1338E-05 .4067E-05		4/805-U0 .37885-U5 .61446-05 .12726-05 28865-06	.7953E-05 .7953E-05 .2259E-05 .4316E-07	.1012E-04 .3810E-05 .4616E-06 .4351E-05	66726-05 89206-05 17856-05 88175-05 88175-05 63406-05 63406-05 61400-05 610
****		1293E-02 .1461E-02 .2151E-05 7057E-05 .4048E-05 3895E-06		1636-05 .37396-02 25126-03 .39426-03 14536-05		1585E+02 .3147E-04 2281E+05 2458E-03 .4584E-02	
	***** ***	•55585+05 •55685+05 •76812-05 •11085-05 •41315-05 •41315-05		-++597E-46 ++526E-85 +5689E-35 -1869E-35	4301-05 4301-05 3019-05 52761-05 65981-05	.483310-05 .487210-05 .46760-05 .40940+05	55871-05 55871-05 55871-05 122721-05 102721-04 55951-09 15733-04 15733-04
••••		.35126-02 .186566-05 .186566-05 .86556-05 .85556-05 .85556-05 .87376-05 .727876-05		1155E-05 1333E-02 6053E-02 1156E-04 454805+06			-1972E-03 -4603EE-03 -4603EE-03 -4603EE-03 -44703EE-03 -44708E-03 -1952E-04 -1952E-04 -1357E-04
		• 5123E-05 • 4037E-05 • 7497E-05 • 5358E-05 • 3155E-05 • 3155E-05	.10196-00 .10196-00 .41726-05 .41.96-05 .22416-05	. 56644 . 33436 . 33436 . 29396 . 29396 . 29396 . 105	.371510 .371510 .48586105 .445366105 .945666105	64)76+05 65418-05 .19528+65 .33178-05	755322 426952 426952 425552 425555 4132955 4132955 413295 410345 403 410345 403 403 403 403 403 403 403 403 403 403
		6989E-02 .8202E-04 2978E-05 1679E-05 .3563E-05 .3563E-05		7514E-07 .4973E-03 1445E-02 .2067E-04		72296-02 72296-03 23246-05 30736-04	• 34535 - 03 • 51435 - 03 • 15225 - 03 • 96741 - 03 • 15225 - 04 • 15225 - 05 • 15225 - 05 • 15225 - 05 • 15245 - 04 • 71945 - 05 • 24196 - 05 • 45115 - 06
• • • •		(4 .7626E-05 .4773E-05 .1712E-05 .1039E-06 .5186E-05 .3495E-05	-19046-07 -15106-07 -55816-05 -47196-05 -29106-05	. 3094E-07 .4901E+05 .4248E-05 .3429E-05	.1002E-05 .5842E-05 .4803E-65 .2277E-06	.5707E-05 .6580E-05 .4347E-05	71876 05 56022-06 49256 05 49256 05 12366 05 11096 05 41096 05 110366 04 110366 04 110366 04 110366 05

200

te en lo

.5047E-06 -.1027E-03 .1025E-01 -.1029£-05 .5011£-05 -.3813£-03 .11622-03 -.11186-01 .23916-03 -.4646E-04 .6380£-02 -.5816E-03 .2433E-04 -.3121E-02 .1667E-02 -.1282E-04 .1583E-02 -.4659E-02 .3061E-05 -.5550E-03 .1098E-01 -.2451E-05 .2354E-03 -.1727E-01 -.1572E-05 .4979E-04 -.1698E-05 -.1064E-04 .1538E-02 -.3718E-02 .4604E-05 .9559E-03 .4452E-03 .4803E-05 .1357E-04 .5788E-05 .4376E-05 .1433E-04 .8051E-05 .3127E-05 .1264E-04 .22196-05 .11196-04 .7954E-03 .6307E-05 .4377E-05 .2772E-06 .7307E-05 .5386E-06 .9226E-05 .8303E-05 .4694*E*-05 .4961£-05 7257E-05 .2584E-05 .1755E-05 -.79246-07 54 90 E-05 112926-04 -5981E-03 -122926-05 -7506E-03 -122926-05 -7506E-03 -122926-05 -7506E-03 -122926-05 -7506E-03 -122926-05 - 7506E-03 -1241E-03 -1241E-04 -2441E-05 -2441E-05 -2441E-05 -2441E-05 -2441E-05 -2441E-05 -2441E-02 -.65066-07 -.86956-05 -29256-05 -.11686-04 -19566-04 -.13566-04 -19466-04 -.138266-02 -99166-05 -.38676-04 -9956-05 -.38676-04 -.48076-04 .60416-02 -.56716-03 .1785-04 -.66285-03 .13795-04 -.66585-03 -.29835-06 -.68765-05 .48785-05 -.28085-05 .2104E-04 -.2427E-02 .15226-02 .2258E-03 -.9653E-02 .5176E-05 .34026-05. .16056-04. .11136-04. .41136-04. .46526-05. .346326-05. .34826-05. .314426-04. .14426-04. .1050E-04 .8414E-05 .2028E-04 -.2173E-32 .1617E-32 -.1113E-04 -.9286E+05 -.9240E-03 -.3916E-02 -.4931E-04 .11026-01 -.79406-04 .2400E-03 -.2289E-01 .3609E-03 -.7701E-06 -.8265E-04 .1232E-01 -.6703E-03 -.3323E-06 .1538E-02 -.1283E-02 -.1554E-04 -21945-J2 -30936-D2 -11026-U5 • •1093E-01 • •1214E-U5 • •3960E-35 .3506E-04 -.4691E-J2 -.5.361E-06 .2771E-05 -.4055E-03 .66294-05 -.8136E-03 .3u14E-03 -.1)95E-05 -.7+40E-04 -.1725E-01 .66395-02 .13196-04 .20455-04 .1745-04 .13045-04 17210 657356 657356 657356 657356 657356 176576 176576 176576 175576 .10795-04 .93622-05 .28552-35 .12465-34 .15146-94 .12275-04 .21576-05 .10645-04 .16735-04 .16995-04 .20625-04 .56511-05 31-=4162 -.83)56-U5 .84326-03 -.31446-02 .5119E-04 .2439E-04 ..3457E-03 ..8195e-03 -.2104E-01 -.8846E-06 -.8563E+06 -.8572E-04 .3347E-03 --97546-04 -.3237E-U5 .1858E-03 -.1608E-01 -.60035-06 -.8147*ē-*03 -.1612£-05 .3292E-03 -.76395-04 .1092E-01 -.3257E-06 --2894E-04 114,32-04 175,162-05 175,162-05 191,162 .5013E+05 .5059E-05 -.2455E-07 - 18116-06 - 24566-05 - 13396-04 - 92756-04 .2014E-05 .3356E-05 .3051E-05 .13236-04 -.49566-07 .1531E-U5 • 2 4 4 6 E - 9 5 .1818i-04 .152GE-04 --+6356-06 .%601E-06 .10115-04 • 8941E-05 18966-04 81522-02 18062-04 2915-03 8772215-05 - 12915-03 8772215-05 - 12915-03 15025-05 - 12915-02 15025-05 - 114702-04 14915-05 - 21645-05 - 114915-02 - 21645-05 - 114915-02 - 21722-05 - 50175-02 - 21726-05 - 114915-02 - 21726-05 - 114915-02 - 21726-05 - 10164-04 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-01 - 113605-05 - 101542-05 - 113605-05 - 101540 - 6116E-06 - 6233E-06 -1912E-05 - 5247E-06 9425E-05 - 59547E-04 - 2207E-05 - 1065E-01 - 1104E-05 - 22011E-05 -- 1104E-05 - 22011E-05 .12136-05 -.2819E-03 .6380E-U2 -.1077E-03 -.3009c-05 - 1284c-03 - 12026-01 - 23046-03 - 17846-05 .24502-05 .26965-03 .16516-07 -.19726-01 -.60546-04 .1226E-04 .4846E-04 .2682E-05 -.3778E-02 -.7457E-06 -.1379E-06 .4649E-06 -.1192E-05 -.8423E-05 .1055E-02 .137uc-04 -.20216-05 .71816-05 .6483E-05 1527E-05 1693E-05 .4649E-06 19786-05 .6056E-05 .1348E-u4 177 UE-05

NN= 1 NP= 41 NPHI= 20 0<= .3434903E+0.0 FREQENCY = 16.40 MH2

6 \* (Y2,Y3 ) ~ = Z 2 ₽ Idom ALL FOR GENERALIZED ADMITTANCE MATRICES PROGRAM DELIVERS THE 6.2.3. PROGRAM PANB AND OUTPUT SAMPLE

E PROCRAM *PAND* , COMPUTATION OF THE NEAR FIELD COMPONENTS		•	CCMMON/B/RH(041), ZH(041), DH(041), TJ(19), R(041), ZS(041),	PANB 52
	UNA U		COMMON/TEET/PT(U+1), MNU(2U), WOT(U) CUT(U) TT(U) TFT(U)	PANR 54
E UUNITAUE VARIAS	PANG	<b>.</b>	DIMENSION REST(20). ZTEST(20). DTEST(20). SINT(20). COST(20)	PANB 55
ATTACH. MAT2. PU=3571.	PAND	ז=ר	U=CMPLX(0,,1.)	PANB 56
CALL, SI64MOT, P=AMAT2, B=DISK.	PANB	ŝ	ETA=376.707	PANB 57
FTN.	PANB	9	PI=3. 141593	PANB 58
LGO.	PANB	~ `	PR=180./PI	PANE CONTRACT
PUBLIC, CONTECT.	PANB	<b>σ</b> α		PANB 61
B HERF FUD OF RECORD CARD	PANB	• e	50 READ(1.51)KK, NP, NPHI, NT, NTEST, NNPHI, BK, RUN	PANB 62
	PANB	=	51 FORMAT(613,E14.7,2X,12)	PANB 63
PROCRAM PANE (INPUT, OUTPUT, DISK, RESULT, TAPE 1=INPUT, TAPE 3=OUTPUT,	PANB	12	IF(EOF (1)) 52,490	PAND 64
ITAPE6=DISK, TAPE7=RESULT)	FANE	<b>ت</b>	490 CONTINUE	PANB 65
C FREFFERERFERE	PANB	Z :	KST=NIEST*NNPHL BEAD(1 E2)/DH(T) I_1 ND)	PANU 00 DANID 67
C SUURCE PRUJARY IS DESCRIBED IN NEIURI "THE STRAUGSE CUMPULEN COUR. C FOR RADIATION AND SCATTERING FROM PODIES OF REUMINITION FYTENDED FOR	PAND	υ¥	READ(1.53)(ZH(I).I=1.NP)	PAND 68
C NEAR FIFLD COMMITATIONS <sup>®</sup> BY R.M. REVENSEE, MAY 1074 LAWRENCE	PANB	2	53 FORMAT (10F8.4)	PANB 69
C LIVERMORE LADORATORY.	PANB	18	TEXT(1)=10HVERTIKAL	DV ID 10
C MODIFICATIONS BY P.A. NEUKOMM AT BIOMECHANICS LABORATORY, SWISS FEDE-	PAND	19	TEXT(2)=10HHORIZONTAL	PANB 71
C RAL INSTITUTE OF TECHNOLOCY, PROVIDE THE E-FIELD VECTOR AT 9 TEST	ENVE	ଛ	TEXT(3)=100RADIAL	PAND 72
C POINTS HEAR THE BODY OF REVOLUTION BY SUMMING UP ALL NEEDED MODI	PANG	2	TEAL (4 )= 10HDINENTE EL	CI GNAT
C FOR ALL AZIMUTHAL ANGLES. THE STORAGE REQUIREMENT IS LOWER THAN IN	PANS PANS	28		
C THE UKTOTINAL FRUUKKIN, THE CUMPRIDULIUM OF THE J-TH MODE IN FRINKLU	DAM	01	TEXT(7)=104RFFLEKTTER	PANB 76
C IN UNDER TO UTILITE THE CONTUNITOR, A FROCEOURE FOR UNATION RE-	DANE	5 K	TEXT (8) = 100TE EINSTRA	PANB 77
C TAL RADIAL ARE STORED FOR FACH TEST DOINT AND ATTWITIAL AND E TH	PANA	3%	TEXT (9)=10HILUNG	PANB 78
C THE FILE STORDS IN THE COLLECT FILE AMATZ.	PANE	32	JA=RU1*3+1	PANB 79
C THIS PROGRAM HAS BEEN DIMENSIONED TO COMPUTE:	PAND	82	JB=RUN*3+2	PANB 80
C KK \$13, NPS41 , NPHI \$20, NT\$2, NTEST\$9, NNPHI \$37.	PANB	53	JC=RUN*3+3	PANB 81
C IN THIS PROGRAM IS NT=1, AND BY SELECTING ON THE DATA CARD *RUN=1,	PANB	ខ្ល	76 WRITE (3, 54)KK, NP, NPHI, NT, NTEST, NNPHI, BK, TEXT (JA), TEXT (JB), TEXT (	PANB 82
C RESP, RUN=2# THE DIRECT WAVE, RESP, THE GROUND REFLECTED WAVE IS RE-	PANB	Ē	54 FORMAT(1X//* KK=#,I3,* NP=#,I3,* NPHI=*,I3,* NT=*,I3,* NTESI=* • T 2 * MURKIT_* T * BV_* E1M 7 EV D10 D10 D10	PANIS 83 PANIS 81
C GARDED FOR FIELD COMPUTATIONS.		ж,	1,410, MNT11=",410," EX=",9514.1,900,010,010,010) WRTTF(2,55)	PANB 85
C ALTERITOR: SCALLING IN THE FRUCKART: FRISLOAL DIFFENSIONS TO LIFES		รล	55 FORMAT(1X/# RH#)	PANB 86
C ACTINE DATA IN ORIGINAL SCALE.	PANA	۲¥	WRITE(3.46)(RH(I),I=1.NP)	PANB 87
	PAND	38	46 FORMAT(1X, 10F8.4)	PANB 88
	PAND	31	WRITE(3,56)	PANB 89
COMPLEX SYV(333), SYI(333), SYS(333)	PANB	ŝ	56 FORMAT(1X/* ZH*)	PANU 90
DIMENSION PUTOT (09, 2, 37), PITOT (09, 2, 37), PSTOT (09, 2, 37)	PANR	۶.	WK1E(3,40)(20(1),1=1,NY) 21 - 1	PAND 91
DIMENSION GVIOT (09, 2, 37), GRITOT (09, 2, 37), GSTOT (09, 2, 37)	LANU	₽1	ALEI 126 TE/(RH/1)_RH/3P)) NE D. OR (7H(1)_7H(NP)) NE D.) CO TO 58	PANB 93
COMPLEX ENTUL, ELTUL, EVIUL, ENTUL, ENTUL, ESTUL	PAND	- - -		PANB 94
	DA NT	5	RII(NP+1)=RH(2)	PANB 95
COMPLEX FTR(13.9.2).FTT(13.9.2).FTP(13.9.2)	PANB	िच	ZH(NP+1)=ZH(Z)	PANB 96
DIMENSION PHI(37), SITU(20), COTN(20), DI (20), PHJ(37), THR(3)	PANB	₽ ₽	RH(NP+2)=RH(3)	PANB 97
INTECER D(61), E(37), TEXT(9), POL, RUN, JA, JB, JC	PANB	¥!	ZH(UP+Z)=ZH(3)	
CUMPLEX U, E3(U2,UM1) ,E4(U2,UM1) ,E1(U2),E2(U2),E3U(2,4),PP		÷		PANE 100
CURPON/A/EX,NN, IK(UTO), I(UTO), UK(Z), UZ(Z), AC(40), CA, CQ, IXC BENT TP(CAS) H	PANR	şg	JU DO JI 1-5,00 I2=I-1	PANB 101
COMPLEX VVR(2.760).Y(1444).ZM(2.076).G(4764)	PAND	្ល	RR 1=RH(I)-RH(I2)	PANB 102
COMMON WR, Y, ZM, G	PAND	21	RR2=ZH(I)-ZH(I2)	PANE 103

PAND 156 PANB 157 PANB 158 PANB 159 PANB 160 PANB 160	PANB 161 PANB 162 PANB 163 PANB 163	PANB 165 PANB 166 PANB 167	PANB 168 PANB 169 PANB 170 PANB 170	PANB 172 PANB 173 PANB 174	PANB 175 PANB 176 PANB 177	PANB 179 PANB 180 PANB 180	PANB 182 PANB 183 PANB 184 PANB 184	PANB 186 PANB 186 PANB 187 PANB 187	PANIB 189 PANIB 190 PANIB 191 PANIB 192	PANE 194 PANE 194 PANE 195 PANE 195 PANE 195	PANB 198 PANB 199 PANB 200 PANB 201 PANB 201 PANB 201	PANB 203 PANB 204 PANB 205 PANB 205 PANB 205
DO 7 I=1,NM 11=2*(I-1)+1 I2=1+1 SS=SS+N(I1)+DN(I2) T_1(I)=SS 7 community	r Continue DEL=1J(NN) TF(KL.NE.O) DEL=DEL+DH(HP-2)+DH(RP-1) DEL=DEL/10.	D0 8 J=1, M TJUJ=TJUJ/DEL 8 CONTURE	D0 42 J=1,NT E1(J)=0. E2(J)=0.	IF(NTEST.EQ.O)GO TO 707 HEAD(), 49)(RTEST(J), 2TEST(J), DTEST(J), J=1, NTEST) 49 FORMAT(F88,4)	D0 708 J=1, NTEST RRT=SGRT(RTEST(J)**2-275ST(J)**2) - SIMT(J)=RTEST(J)/RRT \$COST(J)=2TEST(J)/RRT 708 CONTINUE	707 DP-EI/IPHI CA-BM*BX*ETA \$CQ=ETA DO 709 J=1,NIPHI	ANG(J)=(J5)*DP 709 AC(J)=COS(ANG(J)) DF=PL/(MNH1-1)	R4T(J) H4=(L) L41 \$ PHJ(J) + PHI(J) +	COPHI(M, J)= CGS (NN*PHI(J)) SIPHI(M, J)= U*SIN (NN*PHI(J)) 501 CONTINUE 500 CONTINUE	DO 502 I=1,MTEST ABSTA-SORT(STEST(I)+ZTEST(I)+RTEST(I)*RTEST(I)) SITN(I) = RTEST(I)/ABSTA \$ COTN(I) = ZTEST(I)/ABSTA 502 CONTINE DO MO H-1 KC	NN-AM-1 NN-AM-1 CALL PLAME(VVR, THR, NP, MT, 1, R, 2S, SV, CV, T, TR) CALL PROGA 127 READ(6)(Y(I), I=1, NZ)	WRITE(3,112)NN 112 FORMAT(*1*,* 0*,6X,4SIG 00*,6X,*MAG SOO*,6X,*SIG 00*,6X, 1*MG SOO*,10X,*MODE NN=*,12) WRITE(3,113)
PANB 104 PANB 105 PANB 105 PANB 107 PANB 108	PAND 109 PANB 110 PANB 111 PANB 112	PANB 113 Panb 114 Panb 115	PANB 116 PANB 117 PANB 118 PANB 119	PANB 120 PANB 121 PANB 122	PANB 123 PANB 124 PANB 125	PAND 127 PAND 127 PAND 128 PAND 128	PANB 130 PANB 131 PANB 132 PANB 132	PANB 134 PANB 135 PANB 135	PANB 137 PANB 138 PANB 138 PANB 140	PARE PARE	PAND 147 PAND 147 PAND 147 PAND 148 PAND 149 PAND 150	PANB 151 PANB 152 PANB 153 PANB 154 PANB 154
DH([[2)=SQRT(RR1#RR]+RR2#RR2) SS([22)=5 <sup>6</sup> (2H(1)+RH([2)) R([22)=5 <sup>6</sup> (RH(1)+RH([2)) SV(122)=RR1/DH([2) SV(122)=RR2/DH([12) C7 COVUM	C BRETESTELLING DES ETHFALLS- UND REFLEKTIONSMINKELS DO 1 J=1,NT DO 1 J=1,NT	THR(J)=DT*(J-1) + 1.410 IF(RUM.EQ.2) THR(J)=DT*J + 1.410 1 CONTINUE	BK2=BK*EK P1=SCRT(BK2*ETA*ETA/4,//P1**3) RM=L(P=3//2	NN2=NNEZ NZ=NNEZ NZ=NP-1 KG=NP-1	DO 74 J=1,RM JZ=2*(J=1)+1 J3=J2+1 M=12-3	J5=14+1 J5=14+1 J7=16+1	J8=J7+1 J9=J8+1 DELL=H/L(Z2)+DH(J3) DELL=H/L(Z2)+DH(J5)	TP(de)= M(d2)/DEL1 %TP(J7)= M(J3)/DEL1 TP(d8)==M(d4)/DEL2 %TP(d9)==-M(J3)/DEL2 T(d6)==M(J2)^EM(J2)/Z2./DEL1	T(JT)=bH(J3)*(DH(J2)+DH(J3)/2.)/DEL1 T(J8)=DH(J4)*(DH(J5)+DH(J4)/2.)/DEL2 T(J9)=DH(J5)*DH(J5)/2./DEL2 74 COPTIVUE	DO 75 JEL, IM4 TR(J)=T(J) 75 GRUTURE 115 IF(KL E0, 0) GO TO 78	T7 DELI-DH(1)-MH(2) TR(1)-DH(1)-MH(2)-(1)-C(1)-C(1)-C(1)-C(1)-C(1)-C(1)-C(1)	J2=J1+1 DEL2=PH(NP=2)+DH(NP=1) TR(J1)=DH(NP=2)*(1,+DH(NP=2)/2,/DEL2) TR(J2)=DH(NP=1)*(1,+(DH(NP=2)+DH(NP=1)/2,)/DEL2) 78 SS=0.

2       0.00000000000000000000000000000000000	,J2+MM)#E3(L,J2) PAND 265 )#E4(L,J2) PAND 266 ,J2+MM)*E4(L,J2) PANB 267 ,J2+MM)*E4(L,J2) PANB 268 PANB 268	EARLY CONTRACT TARK 200 PARK 200 PARK 200 PARK 200 PARK 201 PAKK 2	EMAZ 278 EMAZ 708 210 III=*,2212,4,1X, PANB 280 III=*,2212,4,1X, PANB 280 III)=FIEST PANB 281 III)=FIEST PANB 285 20)=FIEST PANB 285 20)=FIEST PANB 285 20)=FIEST PANB 285 EMAZ PANB 285 201=FIEST PANB 285 FIELT PANB 285 FIELT PANB 285 FIELT PANB 285	LTTTLL TANK 200 LTTTL TANK 201 LTTTL TANK 20	CUVRICI, 131) PAMB 301 CUCVRICI, 131) PAMB 301 CUCVRICI, 141) PAMB 301 CUVRICI, 141) PAMB 302 CUVRICI, 141) PAMB 302 ETTPIREETHIR PAMB 305 PAMB 305
2     MMB 213     ESC(IT, J)= FMB 214     ESC(IT, J)= FMB 217     ESC(IT, J)= FMB 219     ESC(IT, J)= FMB 219     ESC(IT, J)= FMB 219     ESC(IT, J)= FSC(IT, J)= FSC(	SS(IT, 1)-ZM(IT, J1)*E3(L, I1)-ZM(IT, J2 SS(IT, 2)-ZM(IT, J1)*M915(L, I1)-ZM(IT, J2 SS(IT, 2)-ZM(IT, J1)*M9154(L, I1)-ZM(IT, J2 SS(IT, 4)-ZM(IT, J1*M9)*E4(L, I1)-ZM(IT, J2)	Z=11+1 \$13=L2+1 \$14=L3+1 "REAL(UVR(1,11)) \$ERI =FTEST*AIMAG "REAL(UVR(1,11)) \$ERI =FTEST*AIMAG "REAL(UVR(1,12)) \$ETH1 =FTEST*AIMAG "REAL(UVR(1,12)) \$ETH1 =FTEST*AIMAG "REAL(UVR(2,12)) \$ETH21=FTEST*AIMAG ETH1=EN1 \$ETHR=ETHR \$ETHR=ETH2R &STTM=ETH1 \$ETTR=ETH2R \$ETTP21 "NOT \$ETTR=ETH1 \$ETTR=ETH2R \$ETTP21	<pre>"EINC.* (NF5.1, UX, "ETHETA INC"/) "PRR. EAL, FH1R, EM1, FH1A, EM14, EM14, EM24, EM24</pre>	RR SETTIALETULAETURAETURAETURAETURAETURAETURAETURAETUR	TTRAL(WW(1,13)) ERI ==TTESTAIN TTRAL(WW(2,13)) ERI ==TTESTAIN TRAL(WW(2,11)) ERII ==TTESTAIN TRAL(WW(2,11)) ERIII==TTESTAIN TEAL(WW(2,11)) ERIII==TTESTAINAG TETTEAR STTIRE=TH =TTIL STTPAR=EMPR &TTIL=ENIZ TOTICALINE STTPAR TETTEAR STTIRE=THA TTICALINE STTIRE=THA TTICALINE STTIRE=THA TTICALINE STTIRE=THA TTICALINE STTIRE TETTEAR STTIRE=THA TTICALINE STTIRE TETTIAR TTICALINE STTIRE TTILE STTIRE
2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 21         1.243,17411           3 21         ESC(TT, 1)=E           3 21         ESC(TT, 2)=E           3 214         ESC(TT, 2)=E           3 214         ESC(TT, 2)=E           3 214         ESC(TT, 1)=E           3 214         ESC(TT, 1)=E           3 216         703 CONTINUE           2 216         703 CONTINUE	218 Jack Charles (C1) at 218 Jack Charles (C1) at 218 228 CBR =7TES 128 228 CBR =7TES 128 228 CBR =7TES 228 CB	277         704         FOMMAT(), 100           228         NUTE(2, 705           229         705         FOMMAT(), 100           229         1*EFH=*, 2612           230         1*EFH=*, 2612           231         EFH<*, 2612	238         ETHLETHING           229         ETHLETHING           229         ETHLETHING           229         ETHLETHING           220         ETHLETHING           229         ETHLETHING           229         ETHLETHING           229         ETHLETHING           229         ETHLETHING           220         ETHLETHING           221         ETHLETHING           222         ETHLETHING           223         ETHLETHING           224         Z11           224         Z11           224         Z11           224         Z11           224         Z11	248         248         571         571         571           239         571         571         571         571           250         571         1714         571         571           551         571         1714         571         571           553         714         7000         703         571           555         714         7000         703         571           555         714         5000         703         571           555         714         5000         703         571           555         714         5000         703         571           555         714         5000         703         571           555         571         571         571         571           555         571         571         571         571           555         571         571         571         571           573         573         571         571         571           573         573         571         571         571           574         573         571         571         571           574         574
2 J)-Y(J1)*VVR(1,11)-Y(J2)*VVR(1,12) J)-Y(J1)*VVR(1,11)-Y(J2)*VVR(1,14) 3(L,J)*VVR(1,J1) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J)*VVR(1,J2) 4(L,J2)*VVR(1,J2)*VVR(1,J2) 4(L,J2)*VVR(1,J2)*VVR(1,J2) 4(L,J2)*VVR(1,J2)*VVR(1,J2)*VVR(J1)*VVR(J1,J2) 4(L,J2)*VVR(1,J2)*VVR(J1,J2)*V	PANB PANB PANB PANB PANB	PANB PANB PANB PANB PANB PANB PANB PANB	PANG PANG PANG PANG PANG PANG PANG PANG	PANB PANB PANB PANB PANB PANB PANB PANB	PANG PANG PANG PANG PANG PANG PANG PANG
	ŭ	. J)+Y (J1)*VVR(1, I 1)-Y (J2)*VVR(1, I2) (J)-Y (J1)*VVR(1, I3)+Y (J2)*VVR(1, I4)	(1,1,1,1) (1,1,1,1,2) (1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,	X1,X2,X3,X4,X5 2,ET2,4) 000 TO 40 1.MTEST 1.MTEST 1.Strong 4,4X,*IEST="F8.4,4X,*DTEST=" 1.Strong 4,4X,*IEST="F8.4,4X,*DTEST=" 1.Strong 4,4X,*IEST="F8.4,4X,*DTEST="	ab2(2)=-DR(1) abf=21(1E) EST(1E),2FELST EST(1E),2FELST R,1H,5,M,JT,RT,2ST,SVT,CVT,TT,TRT) R,1H,5,M,JT,RT,2ST,SVT,CVT,TT,TRT) R,1H,5,M,JT,RT,2S=CHPLX(0.,0.) LX(0.,0.) \$ESC(1T,4)=CHPLX(0.,0.)

TE,L) * COPHI(M,J) PANB 364 TE,L) * SIPHI(M,J) PANB 365 TE,L) * SIPHICH,J) PANB 365	LILC(3,545) NN, ERIOT, ELIUL FAILS 300 TITE(3,545) NN, ERIOT, EFTOT, ETTOT PAUB 367 PAUB 367	PANB 369 PANB 370	TE,L) * SIPHI(M,J) PAND 372	TELU * COLTIL(N, 0) PAND 3/3 TTELU * COLTIL(N, 1) PAND 3/4 TTELU * COLTIL(N, 1) PAND 3/4 PAND 3/4	ITE(3,545) NN, ERTOT, ETTOT PANE 375 ITE(3,545) NN, ERTOT, ETTOT PANE 375 PANE 375	PARTS 278 278 278	c) - ETTOT * SITN(ITE) PANE 380	PANB 381 PAND 382	PANB 383 • + FTTYCT * COTIN(TTF) PANB 384	PANE 385	$\mathbf{TOT} = \mathbf{PP}$ $\mathbf{TOT} = \mathbf{PP}$ $\mathbf{PANB}  380$	510T = PP PANB 388 X10/24DS/EUTOP1)	COO(CABS (EHTOT)) PAND 300	JGTO(CABS(ESTOT)) PANE 391	REAL(ENTUL), ATTACUCUTUL) JTTR	REAL(ESTOT), AIMAG(ESTOT))*PR PANB 394 PANB 394	PANE 396	PANB 397 PANB 397 PANB 398	PANB 399	PAKB 400 PARB 401	TEXT(POL),DA,TEXT(1),TEXT(JA),TEXT(JB), PAND 402 PANR 403	PANE 404	AHLUNGS DIAGRAMM*, PANB 405 Z = *,F6.1,* MHZ*,/110,*KOERPER ; VERTPANB 406	DD. MESSANTENNE : PUNKTANTENNE AUF EFAMB 407 27a = * F5. 1 * CRAD* /14 -102 *MINCHMESSFR PANB 408	24.2,* METER HOEHE *, 20X, *POLARISATION = * PANB 409	ENGE = *, F5.2, * METER*, 21X, *PHI UND DIST VAPANB 410 : POLARISATION = *, R 10,/1H0, 3R10,/) PANB 411	PARIS 412 skel phi = *.f6.2.* grad*./; Pavis 413	PAND DER MESSANTENNE VON DER KOERPEROBEREPAND 415	
PARE 312 EFTOT = ETTOT = ETTOT + ETT(M, I PARE 313 EFTOT = EFTOT + ETP(M, I PARE 313 EFTOT = EFTOT + ETP(M, I	PANB 314 LFUJECTION.J.EQ.377 WK PANB 315 IF(J.EQ.19) WR PANB 316 506 CONTINUE	PANB 317 GO TO 913 PANB 318 910 DO 911 M=1,KK	PARE 319 ERIOT = ERIOT + FTR(M, I PANE 320 EFITOT - ERIOT + FTR(M, I EFITOT - EFITOT - EFITOT - EFITOT	PANB 321 E-1101 = E-1101 + F-11( $r_{11}$ , PANB 322 E-FT07 + E-FT07 + FTP( $M_{11}$ )	PANIS 323 IF (J. CO. 19) WR PANIS 324 IF (J. CO. 19) WR PANIS 324 01 CONTRINE	PAND 326 913 CONTINUE	PANB 328 EVIDT = ERTOT * COTN (ITE	PANB 329 SYV(JDI) = EVTOT PANB 330 EHTOT = EPTOT	PAND 331 SYH(JDI) = EHTOT BAND 332 ESTIVIT = FRIDT # SITN(TTF	PANB 333 SYS(JDI) = ESTOT	PANB 334 IF(CABS(EVTOT).LT.QQ) EV PANR 335 IF(CABS(EHTOT).LT.QQ) EH	PANB 336 IF(CABS(ESTOT).LT.QQ) ES	PANB 337 GHTOT (ITE,L,J) = $20. \text{ ALC}$ PANB 338 GHTOT (ITE,L,J) = $20. \text{ ALC}$	PANB 339 $GSIUT(ITE,L,J) = 20.*ALC$	PANB 340 FYLOI (ITE,L,J) = AIANCIA CPANB 341 PHTOT (ITE,L,J) = ATAN2(R	PANB 342 PSTOT(ITE,L,J) = ATAN2(F PANB 343 505 CONTINUE	PAND 344 504 CONTINUE	PANB 345 WRITE(7)(SYV(1),1=1,NST) PANB 245 WRITE(7)(SYH(1),1=1,NST)	PANB 347 WRITE(7)(SYS(1), I=1, NST)	PAND 348 DO 918 1=1,3 PAND 349 $J = (I^*18) - 17$	PANB 350 WRITE(3,509) F,X1,DU,HO,	PAUB 352 509 FORMAT(1H1, 11X,	PANB 353 1*A Z I M U T A L S T F PANB 354 1 11X, FF R E Q U E N	PANB 355 ZIKALER ROT.SYM. MENSCHMC	PAND 350 $4 = *, 155.2, *$ METER*, 21X, 1	PANB 358 5, R10,/1H , 10X,*ACHSENLAF PANB 359 6R.*, 5X,*MESSANTENNE	PAND 360 WRITE(3,553) PHJ(J) PAND 361 553 FORMAT(1H0.*ROTATIONSWIP	PANB 362 552 FORMAT(11 .*DIST : AF	
EPH2R=REAL(ESC(2,4))*FTEST \$EHH2!=AIMAC(ESC(2,4))*FTEST WRITE(3,720)X1	720 FGMAT(/9X, FESAT, *, 1XF5, 1, 1X, FEYI IN*1) WITE(3, 705)EM: FAIL FENH 1, FEHIL, FEHIL, FEHIL, FEHZR, FEHZI ETRR-FERK-AFER SETRI-FETU-FENT SETTHR-FETHR-FETHR	ETDIL=ETDIL	FIR(N, ITE,L) = CMFLX(EINN, EINL) FIT(N, ITE,L) = CMFLX(EITNN, EITHL)	FPR = (ETP/R+ETP2R)/2 \$ FPI = (ETP11+ETP21)/2 FTP(M,TTE_L) = CMPLX(FPR,FPI)	721 FORMAT(/JX,*ETOT,*JXF5,1,4X,*EFHI INC*/) DISTRET/JX,*ETOT,*JXF5,1,4X,*EFHI INC*/)	WHILE(3, 102) CHAN, EIKL, EIK IN, EIK IL, EILINN, EILAL, EIKEN, SIFEL	722 FORMAT(V///) 7101 CONTINUE	7100 CONTINUE 40 CONTINUE	F=3000, #BK/(2#PI)	PP = CMPLX(1.0E-32, 1.0E-32), $PD = CMPLX(1.0E-32, 1.0E-32)$	00 = 1.0E-32 D0 508 10-1 MIFET	508 DI(ID) = (RTEST(ID)-RH(22))*0.1	TUC 503 L=1,NT X1=THR(L)*PR	D0 912 P0L=1,2	WRITE(3,543)F,X1,TEXT(POL),HO,TEXT(JA),TEXT(JB),TEXT(JC) 543 FORMAT(1H1,8X,*EINFLUSSDERANZSDERANZAHLBERUE	IKSICHTIGTEN MODI AUF ETOT*//HO,13X, 2*EPECNEWZ -* EG 1 * MHZ EVERTRANTING - UTWEET THETA - * EG 1	3" GRAD POLARISATION : #, R10,/1H , 13X, #ANTENNEGHE = #, F42.2,	4* METER*,/1H0,13X,3R10,/) CM1 ECOMAT/1U0 #TESTECTAENT NDF T2 * ADSTAND DIST _* EE 2 * METED	THE CONNECTION TELEDISCONDER WATLES, HASTAND VISIT = , F2.5, THEICH ROT.WINKEL PHI = $\frac{1}{2}$ , F4.0, $\frac{1}{2}$ GRAD*,/)	545 FORMAT(1H, *MODE 0 BIS*, I2, * : ERTOT = *, 2E12.4, 3X, *EPTOT = *, 12F12_4.3X, *ETTOT = * 2F12.4)		DO 504 JIE=1,NIESI WRITE(3,546)	546 FORMAT(/) DO 505 J=1.NUPHT	IF(J.E0.1.08.J.E0.37) WRITE(3,544) ITE, DI (ITE), PHJ(J)	IF(J.EQ.19) WKILE(3,544) ILE, DI(IL), FRJ(J) ERTOT = CMPLX(00.)	ETTOT = CMPLX(0, 0.) $EPTOT = CMPLX(0, 0.)$	IF(POL.EQ.1) 909,910	A MATHAON * (1 att Ware ) A Market D - Merk	EXIOL = EXIUL + ELK( $M$ , LIE, L) " WEAT ( $N$ , V)

PAND 468 PANB 469	PAND 470	PAND 471	PANB 473	PANB 474	C/ P ENAT	PAND 477	PANB 478	PANE 479	PANE 480	PANB 482	PANB 483	PANE 484	PANE 485	PANB 487	PANB 488	PANE 489	PANB 4190	PANE 491	PANB 493	PANB 494	PANB 495 PANR 106	PANB 497	PANB 498	PANB 499	PAKE 501	PANB 502	PANB 503	PANE 504	PANB 506	PANB 507	PANB 508	PARE 500	PANB 511	PANB 512	PANB 513	PANE 515	PANB 516	PANB 517	PANB 519
C ####################################	E SUBROUTINE	SUBROUTINE FLANE (VVK, INK, NP, NI, IT, K, ZS, SV, CV, T, TR) COMPLEX VVR(2.1). A5. A6.U	COMMON / A/ BK, NN	DIMENSION BJ(120), THR(1), FK(20), R(041), 2S(041), SV(041), CV(041),		KG=NP-1	NM=KG/2=1	M2=NN+2 AE_0 #3 10160380580444		FK(1)=1.	DO 153 J=1,M2	01=0+1 CV/14/_FV/1/#1	153 CONTINE	D0 156 L=1,MT	L1=(L-1)*I/V	CS=COS(THR(L))		DO 302 J=1,KG	X=R(J)*BK*SN	J1=J + 1 - M	I1≅NR IF(I1) 303.304.303	304 I1=I1+1	J1=J1+KG 202 TN 205 11 14 H2	5 TE/Y -1 E-K/ 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 IF(JJ-1) 3,3,4	3 BJ(J1)=1.		4 BJ(J1/50. GO TO 206	2 RH=X/2.	RH2=RIF*RH		DV(01)=RRJ(A) SS=BJ(J1)	8 SST=SS*1.E-7	D0 155 K=1,20	SS==SS*RH2/K/(K+JJ-1) BILII)-BILII).60	IF(ADS(SS)-SST) 306. 306. 155	155 CONTINUE	MKIIE(3,1000) 1000 FORMAT(1145TOD AT 155)	RETURN
(NPARB 416 JEPARB 417	S PAND 418	*PANB 419	PAND 421	PAND 422	PANE 423	PANB 425	PANB 426	PAND 427	PANE 428	PARR 430	PANB 431	PAND 432	PANB 433 PAND 431	PANE 135	PAND 436	PANB 437	PANE 438	PANB 439	PANE 441	PANB 442	PANB 443	PAND 445	PAND 446	Land and	PANE 448	PAND 450	PANB 451	PANB 452	PANB 454	PAND 455	PANB 456	PANB 457	PANA 450	PANB 460	PANB 461	PAND 462	PANB 464	PANB 465	PAND 400
1LAECHE IN METER *,/1H , *PHI : HORIZONTALER ROTATIONSWINKEL 3 2 GRAD; 180 GRAD = SCHATTEN*,/1H ,*GAIN : FELDSTAERKE BEI DER )	3SSANTENNE IN DB : 0 DB = MAX, FREESPACE*, / 1H , *PHASE : FHASE DE	HIESSPUNKIES IN UKAD",/140, 22X,FAMPLITU 5DE DER FELDSTAFRKE IM MESSPINKT* 35Y PPHASENIACE FER FELDSTAEDVE	64 MESSPONKT #,/)	WRITE(3,562)	ייד דישראל וווט, "טבא דישראל איז	2 + 90 +180 <sup>4</sup> ,/) 20 -180 <sup>4</sup> ,/)	563 FORMAT(1H/,F4.0,2F8.2)	267 FURMAT(1H+,23X,7(1H.,9X),2X,4(1H.,6X),1H.) E48 ECMMAT(1U. 222 64784)	569 FORMAT(18+, 97X, 37(R1))	D0 512 ITE=1, NTEST	WRITE(3,513) DI (ITE), GVTOT(ITE,L,J), PVTOT(ITE,L,J)	513 FORMAT(TH/,F4.2,ZF8.2)	DU 214 NA ≡ 1,01 514 D(NA)=1H	Q=GVTOT (ITE.L.J)	IF(Q.GT.5OR.Q.LT25.) GO TO 515 \$ Q = Q*2.	KV=IFIX(Q+50.5)	D(KV)=1RA	515 JU 320 NEN=1,37 526 E(NPH)=1H	B=PVTOT (ITE,L,J)/10.	KVP=IFIX(B+19.0)	L(NVY)= INY WRITE(2.516)	516 FORMAT(1H+, 24X,7(1H+, 9X), 2X, 4(1H., 8X), 1H.)	WRITE(3,517)(D(NL),ML=1,61)	URTERS CANNERS (KI))	531 FORMAT(18+,97X,37(R1))	528 FORMAT(1H , F4.2, 20X, 7(1H., 9X), 2X, 4(1H., 8X), 1H.)	IF(ITE.EQ.3)WRITE(3,528) 0.20	IF (ITE, EQ. 4)WHIJE (3,528) 0,50 IF (TTE, FD, 5)WRITE (3,528) 0,40	IF(ITE.EQ.5)WRITE(3.528) 0.45	IF(ITE.EQ.6)WRITE(3,528) 0.55	IF(ITE.EQ.6)WRITE(3,528) 0.60	IF(ITE EV. 7)WRITE(3,528) 0.70 IF(ITE EV. 7)WRITE(3,528) 0.75	IF(ITE.EQ.8)MRITE(3.528) 0.85	IF(ITE.EQ.8)WRITE(3,528) 0.90	IF(ITE.EO.8)WRITE(3,528) 0.95	918 CONTINUE	912 CONTINUE	503 CONTINUE 52 STOP	EID

PAUB 572 PAUB 573 PAUB 574 PAUB 574 PAUB 576 PAUB 576 PAUB 576 PAUB 577 PAUB 577 PAUB 587 PAUB 581 PAUB 581 PAUB 581 PAUB 581 PAUB 585 PAUB 595 PAUB 591 PAUB 577 PAUB 581 PAUB 581 PAU	PANB 594 PANB 595 PANB 597 PANB 597 PANB 597 PANB 599 PANB 509 PANB 600 PANB 601	PANB 603 PANB 603 PANB 604 PANB 604 PANB 605 PANB 605 PANB 605 PANB 611 PANB 613 PANB 613
M5=NH+2 H6=NH+4 FN=AIN FN=AIN FN=2NN*NN FN=2NN*NN FN=2NN*NN FN=2N1 PN=2 D0 10 M=M5,M6 N1=M=3 N2=M3=NHI D0 11 K=1,HPHI N2=M3=NHI D0 11 K=1,HPHI N2=M3=NHI D0 11 K=1,HPHI N2=M3=NHI D0 11 K=1,HPHI N2=M3=NHI D0 11 K=1,HPHI N2=M3=NHI D1 CONTINUE RETURN RETURN RETURN RETURN RETURN RETURN RETURN RETURN RETURN	E SUBROUTINE RECRUCK(1,K3,L) DIMENSION K1(1),K3(1) DIMENSION K1(1),K3(1) DO 81 J=1,L K8=K3(1) K6-J DO 82 I=J,L IF(K3(1-K8) 82,82,84 B4 K8=3(11) B4 K8=3(11)	<pre>K6=T K6=T K6=T(L) K3(K0)=H2(L) K3(K0)=H2(L) K3(K0)=H2(L) K1(L)=H3 B1 CONTINUE K3(L+1)==1 B1 CONTINUE K3(L+1)==1 B1 CONTINUE K3(L+1)==1 EID B1 CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)==1 EID CONTINUE K3(L+1)=1 EID CONTINUE K3(L+1)=1 EID CONTINUE K3(L+1)=1 CONTINUE CONTINU</pre>
PANB 520 PANB 521 PANB 521 PANB 521 PANB 522 PANB 523 PANB 533 PANB 522 PANB 523 PANB 533 PANB 533 PAN	PANB 542 PANB 543 PANB 544 PANB 544 PANB 546 PANB 548 PANB 548 PANB 548 PANB 550	PANB 551 PANB 551 PANB 552 PANB 554 PANB 554 PANB 554 PANB 555 PANB 556 PANB 566 PANB 567 PANB 567 PANB 567 PANB 567 PANB 571
306 Ji=Ji4C 302 CONTINE 302 CONTINE 303 CONTINE 303 CONTINE 301 0 309 Ja1,6C 301 0 309 Ja1,6C 301 0 300 Ja1,4C 3312-4M Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+L Ji=J+C VR(TT,J2)=0.VR(TT,J2)=0. VR(TT,J2)=0.VR(TT,J2)=0. VR(TT,J2)=0.VR(TT,J2)=0.VR(TT,J2)=0.VR(TT,J2)=0.VR(TT,J2)=0.VR(TT,J	A6=(COS(2S(I)*DCS)+U*SIN(2S(I)*BCS))*A5 bJ1=(BJ(I3)+BJ(I1))*.5 BJ2=(BJ(I3)+BJ(I1))*.5 VR(IT,21)=VR(IT,21)-A6 <sup>6</sup> C(SS <sup>1</sup> SV(I)*BJ2-SN <sup>4</sup> CV(I1)*BJ(I2)*U)*T(I <sup>4</sup> ) VR(IT,21)=VR(IT,21)-A6 <sup>6</sup> C(SS <sup>1</sup> S1 <sup>4</sup> U*T(I4)) VR(IT,21)=VR(IT,21)-A6 <sup>6</sup> SC <sup>2</sup> TR(I1 <sup>4</sup> ) VR(IT,41)=VR(IT,31)-A6 <sup>6</sup> SL <sup>2</sup> TR(I4)	301 CONTINUE 156 CONTINUE REJTRN REJTRN REND C INTERT SUBJOUTINE PROCA COMPLEX U COMPLEX U TST(041), AIG(20), T(1941), TJ(19), R(041), ZS(041), TST(041), CV(041), AIG(20), CSM(50) TST(041), CV(041), CV(041), AIG(20), CSM(50) TST(041), CV(041), CV(041), AIG(20), CSM(50) TST(041), CV(041), CV(041), CV(041), CV(041), CV(041), CV(041), CV(041), CV(041), CV(041), AIG(20), CSM(50) TST(041), CV(041), CV(041), AIG(20), CSM(50) TST(041), CV(041),

PANB 676	PANB 677 PANR 678	PANB 679	PANB 680 DAND 601	PANE 682	PANB 683	PANB 684	PANB 685	PANB 687	PAND 698	PANB 689	PANB 690	PANB 691	PANB 692	PAND 50%	PAND 695	PARE 696	PANB 697	PANB 698	PANB 699	PANE 701	PANB 702	PANB 703	PANB 704	207 BING	PAND 700	PANB 708	PANE 709	PAND 710	PAND 712	PAND 713	PANB 714	PANB 715	PAND 717	PANB 718	PAND 719	PANB 720	PANB 722	PAND 723	PANB 724 PANB 725	1) PANB 726 PANB 727	121 101
GST(1)=#1+U*#2 CO TO E	7 Y0=ABS(RR1+AC(K)*RR2)	RD=RR3+RR4*AC(K) pv_pvescorver)	D2±RD-Y0#Y0	Y1=Y0-DEL \$Y2=Y0+DEL	R1=SQRT(Y1#Y1+D2) \$R2=SQRT(Y2*Y2+D2)	If (11)/2, /3, /3 72 TTN=A1/CC//_V1_P1/#/V2_P2/ AN2/	G0 T0 25	73 TIN=ALOG((Y2+R2)/(Y1+R1))	25 SN=SIN(RK) \$CS=COS(RK)	GST(K)=(CS-U*SN)*(TIN-U*(BK*DH(J)-RK*TIN))/DEL1	5 CONTINUE	1) A ANL-1 -2	MI=MM=1 SM4=M1FNIFIT SM2=M1+KCF4KCF4KCF4KCF4KCF4KCF4KCF4KCF4KCF4KCF4	GT (M2)=CMPLX(00.)	DO 13 K=1, NPHI		1.5 UL (MZ.)=UL (MZ.)+UST (K.)=USY (K2.)	17 CONTINUE	16 CONTINUE	DEL 1=DHT(1)+DHT(2)	DEL2=DHT(3)+DHT(4)	TPT(1)=DHT(1)/DEL1 \$TPT(2)=DHT(2)/DEL1	111(3)=-DH1(3)/DEL2 \$1PT(4)=-DHT(4)/DEL2 TT(1)=DHT(1)*DHT(1)/2 ^DEL4	TT(2)=DHT(2)*(DHT(1)+DHT(2)/2.)/DF1.1	TT(3)=DHT(3)*(DHT(4)+DHT(3)/2.)/DEL2	74 TT(4)=DHT(4)*DHT(4)/2./DEL2	INT(1)=TI(1) \$INT(2)=TT(2) \$TNT(3)=TT(3) \$TNT(4)=TT(4) 78 DO 30 J=1.NM		J3=(J-J)#ti	J1=2*(J=1)	LIEU \$LZ2LIANM \$L3=J+NH2 \$L4EJ3+NH2 \$L4=L3+NH 74/1T 13/-CNN2 Y/A 3/ 43/14 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/ 2000 10/	ZM(IT.I.3)=CMPLX(0.,0.) \$ZM(II,L2)=CMPLX(0.,0.) ZM(IT.I.3)=CMPLX(00.) \$ZM(IT ! U)-CMPLY(0.0.)	Do 70 JJ=1, h	J2=J1+JJ \$J7=J3+JJ	DO 71 LI=1,4 T2-TT &TT-TT	J4=(J2-1)*KG+I2	NG=KG ₹KG	J5=J4+NG \$J6=J5+NG SS=SVT172)#SVZ123	A3=.5*(GT(J6)+GT(J4)) \$A4=.5*(GT(J6)-GT(J4))	ZM(IT,L1)=ZM(IT,L1)+(CA*TT(I7)*T(J7)*(SS*A3+CC*GT(J5))-CQ*TPT(I7) 1*TP(J7)*GT(J5))*U	
PANB 624 PANB 624	PANB 626	PAND 627 PANR 628	PAND 629	PANB 630	PAND 631	PAND 032	PANB 634	PANB 635	PAND 636	PAND 637	PAND 030	DANA 640	PANB 641	PAND 642	PAND 643	PANB 044	CHO GUAN	PAND 647	PAND 648	PANB 649	PAND 650	PANB 651	PANE 653	PAND 654	PANB 655	PAND 656	PANB 658	PANB 659	PANB 660	LON UNAY	PART 662	PANB 664	PAND 665	PANB 666	PARE 668	PAND 669	PAND 670	PANG 571 PANG 572	PAND 673	PAND 674 PAND 675	
1SV(041), CV(041), ANG(20), CSM(60) COMPLEX VVR(2, 760), Y(1444), ZM(2, 076), CT (4764)	CORPON VVR, Y, ZM, GT	NM=KG(Z=1 \$1NM2=NM#2	D0 2 1=1,5	X1=I=3	2 ZHT(I)=X1*0Z(TT) +7TEST	D0 57 I=2,5	I2=1-1	RK1=Rkif(I)-Rkif(I2) \$RR2=Zlif(I)=ZHT(I2) Nir(I3)-\$C\$T{\$B\$4#B\$4, \$B\$3#B\$23	UNI (12)#3441 (KK   7KK   4KK2*KK2) 7557(12)= 5,54(2)[7(1)_2](7(13))	SVT(T2)=881/04/17(T2) SVT(T2)=881/04/17(T2)	57 CVT(12)=RR2/DHT(12)	FM=HN \$FM2=NN#NN	D0 16 J=1,KG	DEL=.5*DH(J) \$DEL1=DH(J)*DK	AAEUF*K(J)*DEL*LK DO 17 T-1 4	Z 3=ZS(J) -ZST(T)	RR1= SV(J)*R(J)+CV(J)*Z3	RR2=-SV(J)*RT(I)	RR3=R(J)*R(J)+RT(I)**2+Z3*Z3	KK44=2.**R(J)*RT(I) V4 /2/17 SEV-1/2010	X1=(K(^)-HI(T)**2+Z3**2 YA-(RT(T)-P/1)/#6V/1)./757/T> 75/1//#2//1	xx=xxx x + x + x + x + y + + + + + + + + +	IF(DELA2.NE.0.)GO TO 707	DELA=0. \$60 TO 708	/U/ UELA=SQRT(DELA2) 708 PM E V-1 NRVI	IF (K. NE. 1. OR. DELA. OF. DTEST. OR. ARS (VA). CT. DE1 YOU TO 70	X=R(J)*DP	YAA=DEL+YA \$YBB=DEL-YA	IRACEINATIAN \$150221(1)TUG YYA-CAPT/YEV, YAAD.NEY AD.	XXB=SQRT (X *X +YBD2+DELA2)	XX1=SQRT (YAA2+DELA2)	XX2=SQRT (YBD2+DELA2)	W1=X *ALDG((YB0+XXB)/(-YAA+XXA))	TL VARE EXTURNING (X + XXR) / XX2)	801 IF(XX1.EQ.0.)GO TO 802	W1=W1+YAA <sup>R</sup> ALOG((X+XXA)/XX1)	804 IF (DELA, EQ. 0, )GO TO 803 Y1-Y ART A	ATN=ATAN2((YBB*X1),XXB) + ATAH2((YAA*X1),XXA)	W1±W1→DELA#ATN 803 W1→D EFU1XA		

CONTINUE CONTINUE ERETURN ERETURN ERETURN CONTINUE ERETURN CONTINUE CONTINU	R CARD RT CARD 1.0.3434 0.490 0.400000000	+(CA*A3-4) 808E +00 0.515 0.660 0.660 0.660 11.085 5.960 5.960 11.340	60111111111111111111111111111111111111	TIT)*RIT RT(12)*RI RT(12)*RI 16.4 MIZ 0.500 0.780 0.700 1.070 1.700 1.710 1.710 1.710 1.5400 15.400	7.44 (177)78( (127)78( (122)767( (122)767( 0.515 0.515 0.670 0.670 0.670 0.670 15.700 15.700	25)*TT(1 1.050 1.050 0.822 15.050 15.050 16.050	(7)* 0.525 0.7050 0.7050 0.7050 0.7050 15.420 16.420	PARAT PA
	~~~~~							
10.0 0. 10.0 0. END OF RECC 64D9, RESULT, END OF INFC	2 RD CARD R0.	CARD						PANB PANB PANB PANB PANB PANB

- 270 -

RESULTS FROM PROGRAM \* P A N B \* (ONLY MODE NN = 1 SHOWN HERE)

DIREKTE EINSTRAHLUNG .3434808E+00 KK= 8 NP= 41 NPHI= 20 NT= 1 NTEST\* 9 NNPHI= 37 BK\*

£

.5250 1.0000 .5000 .5250 9300 .5150 1.0503 1.0000 .8200 .5150 .9250 1.0400 .8700 .5000 .7800 1.0700 .8500 .5000 .7122 1.1000 .7699 . 5150 . 6550 1. 0850 . 6240 .5150 .6250 1.0200 0067. .5000 .6000 .9800 .9900 .630C 0.0630 0.0006

ZH

--40DE NN= PAG SOB SIG AA •23395-01 11AG S 83 • 2021E+03 SIG 88 00.79 •

•4495E+39

.1703E+00

.2000 07ES1= ZTEST= 10.0000 1.7500 RTEST=

ETHETA INC EINC, AD.8

- 1305E+00 -.8302E-01 .1540E+00 EPH= .6653E-01 .1306E+00 ETH= --30045-01 -\*#5792+00 EPH= ETHETA INC ESCAT, BC.8 •3090E+00 ERAD=
- -.4374E-01 .7253E-02 .5339E-01 EPH= .6585E-01 E TH= -.4437E-01 • 3599E+02 •21035+00 €PH= ETHETA INC ETOT, 80.8 -.8361E-01 ER AU=
- .8679E-01 -.7276E-01 .2074E+00 EPH= .1324E+00 E1H= .8521E-01 -.70445-01 •22546+00 -.24766+00 EPH= EPHI INC EINC, 80.8 ERAD=
  - 4548E +00 7418E+00 -.8031E+00 EPH= •4923E+0C ETH= .4532E+00 •7427E+00 •8618E-01 -.1406E+00 EPH= EPHI INC ESCAT, 80.8 ERAD=
- -.21495+00 -.2356E+00 -.42966+00 EPH= •1088E+00 ETH= -.2110E+00 -.2277E+00 -\*5590E+J1 EPH= DNI IHGE ETOT, 80.5 .14306-01 ERAD=
- •2398£+00 •5360E+00 -.1233E+01 EPH= .6011E+10 ETH= .24425+00 •5150E+00 -.1965E+00 EPH= .1005E+00 ERAU=

.3619E+03 -.1301E+UO .2318E+00 -.2565E-01 .96492-01 .1221E+00 -,1588E+00 .2539E+00 .94296-01 .4127E+00 -.3246E-U1 .1268E+00 •5488E+96 .1039E-01 •5903E+00 -.8149E-01 -.1282E+00 -.64502-01 •5449E+0G .2419E+00 EPH= -.74865-01 .1J21E-01 .6731E+00 -.7771E-01 -.67506-01 -.2254E+00 EPH= .2271E+00 EPH= -.7356E+00 EPH= -,961UE+06 EPH= .4680E-01 -.3026E+00 EPH= -.1487E-01 EPH= -.1075E+01 EPH= .1952E+06 EPH= .2036E+00 EPH= -.7724E+00 EPH= .8401E-02 EPH= .46235+30 .5203E+16 .2847E-01 .5570E-02 .451UE+JC .1183E-01 .4735E+90 -.2290E-01 .6858E-01 .2499E-01 .4359E-01 ETH= .23276+00 ETH= €TH= £1H= ETH≃ = H1 3 E T H = .4130E+00 ETH= ETH= .2557E+00 ETH= .1268E+00 ETH= ETH= .9657E-01 -.2561E-01 •3622E+00 -.1235E+00 -.1573E+00 .1222E+00 .9434E-01 -.3246E-01 .2000 .2690 -.8025E-01 .1754E-01 -.5436E-01 .5107E+00 .591CE+00 -.1253E+00 -.7490E-01 +6739E+00 .5486E+00 .1077E-01 -.6696E-01 -.7773E-01 DTEST= oTES1≞ -,59852-01 EPH= ZTEST= 10.0000 .4011E+C0 -.5308E+J0 EPH= .12406+30 -.20245+30 EPH= .4403E+00 -.6618E+00 EPH= .1310E+00 EPH= .1066E+00 -.1739E+9C EPH= .77226+02 -.5367E-J1 EPH= .1143E+00 -.2275±+00 EPH= ETHETA INC -**+**5672±+)0 EPH= .3193E+00 -.4126E+00 EPH= ETHETA INC -.6107E-01 .1547E+00 EPH= ETHEIA INC ETHETA INC ETHETA INC ETHEIA INC EPHI INC EPHI INC EPHI INC EPHI INC ETOT, 89.8 SPHI INC EPHI INC ESCAT, 8J.8 ESCAT, '80.8 ETOT, 80.5 80.3 ETOI, 80.5 8.0.8 ETOT, 89.8 8.08 8.0.8 EINC, 80.8 ESCAT, 80.5 .2830E-02 -.39165-01 .3804E+00 2.7500 RTEST= 2.2500 EINC, ESCAT, EINC. EINC, RTE ST= ERAD= ERAD= ERAD= ERADI ER A0= ER AD= ERAD= ERAD= ERAD= ERAD= ERAD=

- 272 -

ZTEST= 10.0090

.12698+00 -.25326+39 dPH= ERAD= KTEST= 3.7500 ZTEST= 10.0000 TTEST= .2030 CINC, 80.A ETHETAINC D= .52016+99 -.79995+00 EPH= -.67866-01 .1107E+00 ETH= -.12806+00 .

- .1107E+J0 -.6785E-01 .3403E+01 EPH= ETHE TA INC SCAT, 80.8 ER A C=
  - -.1849E-01 • 2 3 0 E - 0 2 -.3195E-01 EPH= .1096E-01 -.1848E-01 ETH= .8189£-02 -H42 C(+36001. ETHELA INC ETOT, 89.8 -.7614E-C2 ERAD=
- .92195-01 -.5962E-01 .3084E+60 EPH= -.1171E+00 ETH= .92236-01 --5967E-01 .51255+00 -.6983E+00 EPH= EPHI INC EINC, A0.8 ERAU≃
- .2396E+00 •3906E+00 -.6473E+00 EPH= .3968E+0C £ TH= .2396E+00 •3111E+00 ,14886+00 -,24286+30 EPH= EPHI INC 60.0 SCAT, ERAD=
- -.9966E-01 -.5255E-01 -.1340E+00 EPH= -.2376E-01 -.9351E-01 ETH= -.5228E-01 -.6382E-J2 -.4475E-J1 EPH= EPHI INC ETOT, 81.8 ERAD=
- .1400E+00 .3381E+06 -.7813E+0C EPH= .3731E+90 .1401E+00 ETH= .3388E+00 .1424E+00 -.2876E+00 EPH= ERAU=
- <u>rtest= 4.7550</u> ztest= 10.0009 ntest= .2000 Einc, 80.8 etheta inc
- .96752-01 -.5932E-01 .4262E+00 EPH= ETH= -.2291E+00 .9677E-01 -.5331E-01 .5431E+30 -.8603E+00 EPH= ETHETA INC SCAT, 80.8 ERAD=
  - -.15202-01 .5552E-02 -.3572E-61 EPH= -.1875E-02 ±H1 = -.1519E-01 •5486E-02 •78366-01 EPH= ETHEIA INC ETOT, 89.8 13562-01 ERAU=
- .6155E-01 -.5377E-01 .3905E+0C EPH= -.2310E+0C .8153E-01 ETH= -.5382E-01 •5566E+00 -•7826E+JC EMI= SPHI INC CINC, 81.9 ER AC=
- .99315-01 .1517E+00 -.5458£+00 £PH= .3347E+00 E T H = .93056-01 .1519E+00 .1589E+00 -.2593E+0C EP4= EPHI INC 80.b ESCAT, ERAD=
- -.7916E-01 -.5179E-01 -.8122E-01 EPH= -.376 % -01 ETH= -.7912E-01 -.5175E-01 -.1447E-01 -.37376-11 EPH= EPHI INC ETOT, 59.8 ERAC=
- .2015É-01 •10 99E +00 -.6271E+00 EPH= .2970E+0C •1393č-01 ETH≈ .1102E+06 .14456+00 -.29676+30 EPH= ERAD=

.2000 DTEST= ZTEST= 10.0019 ETHETA INC EINC, 80.8 RTEST= 6.2500

- .72946-61 -.4473E-01 •4889E+00 EPH= -.3352E+06 £TH= .7295E-01 .4791E+00 -.8178E+00 EPH= ETHEIA INC CSCAT, 80.8 ERAU=
- -.1269E-01 .1709E-02 -.3144E-01 EPH= -.1819E-01 -.1268E-01 ETH= 1649E-02 .3226E-61 .4930E-01 EPH= ETHETA INC ETOT, 80.8 ERAU=
- •6025**Ξ-01** -.4302E-01 .4575E+00 EPH= --3534E+30 .6027E-01 ETH= --43056-01 .5114E+00 -.7685E+00 EPH= EPHI INC EINC, 80.8 ERAU=
- -.1161E+00 -.1898E+00 -.3863E+00 EPH= .2369E+30 ETH= -.1166E+00 -.1398E+00 .1480E+00 -.2415E+J0 EPH= EPHI INC ESCAT, 80.8 ERAU=
- -.5012E-01 -.6320E-01 -.3419E-01 EPH= --4030E-01 -.5011E-01 ETH= -.6323E-01 -.2263E-01 -.2444E-J1 EPH= EPHI INC ETOT, 80.8 ERAD=
- -.1663E+00 -.2500E+00 -.4205E+00 EPH= 1966E+00 -.1667E+00 ETH= -.2500E+00 -.2659E+30 EPH= .1253C+60 ERAD=

.2040 **JIEST=** ZTEST= 10.0030 ETHETA INC FTEST= 7.7500

EINC, 81.8

- .4810E-01 -.2951E-01 .4393E+00 EPH= --3542E+00 .4312E-01 ETH= --2948E-01 .3296E+00 -.6451E+00 EPH= ETHETA INC SCAT, 80.8 ERAU=
- -.1064E-01 -.1309E-02 -.2957E-01 -.1973E-01 EPH= -.1362E-01 ETH= -.1955E-02 -24005-01 SPH= ETHELA INC ETOT, 80.8 .38735-01 EKAD=
- .37465-01 --31 32E-01 .4195E+00 EPH= -.3837E+90 ETH= .37+9E-01 -.3133E-01 .3683E+60 -.5211E+]0 EPH= SNI IHGE EINC, 83.8 ERAD≡
- -.2984E+00 -.4371E+00 -.23756+00 EPH= .1457E+JC E T H = -.2388E5+00 -.4372E+00 .1128E+30 -.1841E+00 2PH= EPHI INC ESCAT, 80.8 ERAD=
  - -.2056E-01 -.6398E-01 -.3289E-01 -.7642E-02 EPH= -.6401E+01 -.2353E-01 ETH= -.2541E-01 -.11225-31 EPH= EPHI INC ET01, 80.8 ERAU=
- -.31905+00 -.5511E+00 -.2451E+00 EPH= .1128E+0C -.5512E+80 -.319+E+80 ETH= .87382-01 -.19535+00 EPH= ER AU=

			~ •				_				_													
	.2497E-01		81205-02		.1685E-01		41165+00		.6702£-ú		40495+00			.5146E-03		3937E-02		3422 <u>E</u> -U2		4226£+DÚ		.3308£-01		38955+00
	-,1532E-01		4706E-02		2303E-01		67156+00		-,5845E-01		-*7300E+00			3266E-03		6963E-02		7289E-02		6893E+00		3900£-01		7273E+00
	.2791E+00 EPH=		4286E-02 EPH=		.2748E+00 EPH=		1145E+00 EPH=		.6349E-02 EPH=		1081E+0G EPH=			3611E-01 EPH=		•1530E-01 EPH=		2081E-01 EPH=		2135E-02 EPH=		.1231E-01 EPH=		.1018E-01 EPH=
	•.27996+00		3385E-01		3137E+30		.7025E-01		••2193E-01		.4832E-01			-,7894E-01		2787E-01		1068E+00		.1355E-02		7785E-02		6430E-02
	ETH=		ETH=		ETH≤		€TH=		E 1H=		ETH≖			E 1H =		ETH=		€1H=		ETH=		ETH=		E 1H =
	.24985-01		8093E-02		.1689E-01		4117E+00		.6708E-02		4050E+03	.2000		•5290E-03		3933E-02		3380E-02		42256400		.3337E-01		3394E+00
	1530E-01		4735E-02		2003E-01		6715E+00		5847E-01		7300E+00	91ES1=		3132E-03		-•637UE+02		7283E-02		-+6893E+00		3801E-01		72736+00
EINC, 80.8 EFHETA INC	.1440E+003399E+30 EPH=	ISCAT, 80.8 ETHETA INC	.3604E-U1 .3607E-32 EPH=	ETOT, 80.8 ETHEIA INC	.1800E+C03954E+30 EPH=	EINC, 80.8 EPHI INC	.6487E-011059E+00 ZPH=	ESCAT, 80.8 EPHI INC	2319E-01 .2557E+03 EPH=	ETOT, 80.8 EPHI INC	.41685-0119575+)0 @PH=	ST= 11.2500 ZTEST= 10.0000	EINC, 80.5 ETHETA INC	-,7773E+01 -,5859E-11 EP4=	ESCAT, 80.8 ETHETA INC	•2379E-011381C-31 EPH=	ETOT, 89.8 ETHETA INC	5394E-017240E-01 EPH=	EINC, 80.8 EPHI INC	.1462E-f22473é+32 5PH=	ESCAT, 80.8 EPHI INC	1515E-01 .1044E-11 EPH=	ETOT, 81.8 EPHI 140	1368E-01 .7972E-02 EPH=
	ER AD=		ER AC=		ER AO=		ER AD=		ER AN=		ER A0=	RTE		ER A 0=		ER AD =		ER AD=		ER A D=		ERAD≖		ER AU=

RTEST= 9.2500 ZTEST= 13.0009 OTEST= .2010

- 275 -

ż

E T O T AUF I 0 0 H BERUECKSICHTIGTEN ANZAHL 0 E R EINFLUSS

POLARISATION : VERTIKAL FREQUENZ = 164.0 MHZ EINSTRAHLUNG # WINKEL THETA = 80.8 GRAD Antennenhoehe = 1.00 'teter

DIREKTE EINSTRAHLUNG

	.4954E+00 .764E+00 .7345E+00 .7385E+00 .7380E+00 .7380E+00 .7380E+00 .7380E+00		.4990E+00 .4990E+00 .4620E+00 .4620E+00 .4650E+00 .46555E+00 .46555E+00 .46555E+00 .46555E+00 .45555E+00 .3330E+00 .3330E+00 .3330E+00 .3330E+00 .3330E+00 .3330E+00
	.13215+01 .1454E+01 .144126+01 .144126+01 .14136+01 .14136+01 .14136+01 .14136+01 .14136+01		1321E+01 1357E+01 1357E+01 1357E+01 1357E+01 1357E+01 1357E+01 1357E+01 1357E+01 1351E+01 1351E+01 1456E+01 1156E+01 1156E+01
	ETT0T = ETT0T = ETT0T = ETT0T = ETT0T = ETT0T = ETT0T = ETT0T =		
GRAD		GRAD	0. 7160E-01 7160E-01 75166E-01 7516E-01 7516E-01 7516E-01 7516E-01 7516E-01 0. 2.7516E-03 3754E-00 0. 2710E-00 2710E-00 2710E-00 2710E-00 2710E-00
•0 = IH4		PHI = 90.	0. 
ROT.HINKEL	EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01	ROT.WINKEL	EP101 = EP101 =
09 NETER	1610E+00 4095E+00 3605E+00 3543E+00 3548E+00 3548E+00 3548E+00 3548E+00 3548E+00 3548E+00	.08 METER	16106+00 16106+00 16106+00 209006+00 209066+00 209466+00 209466+00 209466+00 209466+00 209466+00 209466+00 209466+00 23666+00 -12866+00 -128466+00 -128466+00 -128466+00 -128466+00 -12846+00 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000 -12846+000000000000000000000000000000000000
ND 01ST = .	.1792E+00 .4045E+00 .4045E+00 .4643E+30 .4591E+30 .4587E+0 .4587E+0 .4587E+0 .4587E+0	ND DIST =	$\begin{array}{c} 1.7926\pm0.0\\ 1.7926\pm0.0\\ 1.19946\pm0.0\\ 1.19946\pm0.0\\ 1.18946\pm0.0\\ 1.18946\pm0.0\\ 1.1895\pm0.0\\ 1.1896\pm0.0\\ 1.18966\pm0.0\\ 1.18966\pm0.0\\ 1.1876\pm0.0\\ 1.187$
1 ABSTA	ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT =	1 ABST	$ \begin{array}{c} \mbox{ERTOT} = \\ \mbox{ERTOT} = $
TESTSEGMENT NR	MODE       0       81S       0         MODE       0       81S       1       1         MODE       0       81S       1       1         MODE       0       81S       2       1         MODE       0       81S       3       1         MODE       0       81S       3       1         MODE       0       81S       3       1         MODE       0       81S       4       1         MODE       0       81S       4       1         MODE       0       81S       4       1         MODE       0       81S       5       1         MODE       0       81S       5       1	TESTSEGMENT NR	MODE         BIS         1

FOR THE TESTSEGMENTS NR. 2 TO 9 ONLY THE VALUES FOR PHI = 180° ARE SHOWN HERE

180.
H
H
ROT.WINKEL
METER
5
•
n
0151
ABSTAND
~
Ħ
TESTSEGHENT

GRAD

256+0 256+0 256+0 256+0 256+0 216+0 216+0 216+0 216+0 0 216+0 0	3655 3655 3655 3655 3655 3655 400 3655 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 4000 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 400 3755 200 3755 200 3755 200 200 200 200 200 200 200 200 200 2	337 337 467 547 547 547 101 101 101 101 101 101 101 101 101 10
	,	04033333 040000000000000000000000000000
.1027E+01 .9584E+00 .9164-E+00 .9116E+00 .9125E+00 .9125E+00 .9125E+00	.8271E+00 .8215E+00 .7441E+00 .7551E+00 .7551E+00 .7548E+00 .7548E+00 .7548E+00	<pre>5679E+00 6849E+00 5437E+00 5514E+00 5514E+00 5514E+00 5575E+00 5575E+00 575E+00</pre>
	8 8 8 8 8 8 8 8 8 9 8 9 .	
ETT0T ETT0T ETT0T ETT0T ETT0T ETT0T ETT0T	ETT01 ETT01 ETT01 ETT01 ETT01 ETT01 ETT01	ETT01 ETT01 ETT01 ETT01 ETT01 ETT01 ETT01
0. 23296-07 98066-08 14189-07 14186-07 14186-07 14076-07 14076-07	6RAD 1. 22322-07 .22322-07 .224925-07 .225925-07 .233785-07 .23595-07 .23595-07 .23555-07	6RAD 0.20665-07 -21425-07 -472926-07 -33426-07 -33876-07 -33876-07 -33416-07 -33416-07 -33416-07
0. 3267E-07 5274E-07 4288E-07 4188E-07 4188E-07 4204E-07 4205E-07 4205E-07	- PHI = 180. 0. 33446-07 58076-07 58076-07 43516-07 442076-07 42146-07 42146-07 42146-07 42146-07 42146-07	PHI = 180. 0. 3194E-07 36333E-07 3602E-07 33497E-07 35497E-07 35497E-07 35497E-07 35138E-07
EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01	R01. H EPT01 EPT01 EPT01 EPT01 EPT01 EPT01 EPT01	ROI. 4 EPTO EPTO EPTO EPTO EPTO EPTO EPTO EPTO
25556+00 .15706+00 .23496+00 .22176+00 .22176+00 .22076+00 .22076+00	<ul> <li>.13 HETER</li> <li>.3 224640</li> <li>.3 2126400</li> <li>.2 37126400</li> <li>.2 9516400</li> <li>.2 9516400</li> <li>.2 9516400</li> <li>.2 9546400</li> <li>.2 9546600</li> </ul>	<ul> <li>23 HETER</li> <li>3972E+00</li> <li>3098E+00</li> <li>4932E+00</li> <li>4232E+00</li> <li>4232E+00</li> <li>4233E+00</li> <li>42304E+00</li> <li>4304E+00</li> <li>4304E+00</li> </ul>
*1704540 *1465540 *14655410 *46185411 *35205411 *3645611 *3645611 *366515411 *366515411	AND DIST = 10755 = 0 10755 = 0 107555 = 0 10755 = 0 107555 = 0 107555 = 0 107555 = 0 107555 = 0 10755	AND DIST = .1683E+00 .1683E+00 .3441E+00 .3636-01 1716-01 1706-01 1706-01 706-01 
E2101 # ERT01 # ERT01 # ERT01 # ERT01 # E2101 # E2101 # E2101 #	3 Ansi E8101 = E8101 = E8101 = E8101 = E8101 = E8101 = E8101 =	4 4 ABS1 ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT = ERTOT =
HODE 0 015 0 HODE 0 015 0 HODE 0 015 1 HODE 0 015 2 HODE 0 015 5 HODE	TESTSEGMENT NR MODE 0 UIS 0 1 MODE 0 UIS 1 1 MODE 0 UIS 2 1 MODE 0 UIS 2 1 MODE 0 UIS 5 1 MODE 0 UIS 5 1 MODE 0 UIS 5 1 MODE 0 UIS 7 1	TESTSLGHENT NP TESTSLGHENT NP MODE C UIS 0 1 MODE 0 015 1 1 MODE 0 015 2 1 MODE 0 015 4 1 MODE 0 015 6 1 MODE 0 015 6 1

GRAD
180.
ΡΗΙ
ROT.WINKEL
S METER
•
0
DIST
AJSTAND
5
NP.
TL STSEGNENT

.2847E+3C 1059E+0C 1589E+0C 8702E-01 8899E-01 9182E-01 9187E-01
.3994E+00 .6303E+00 .4058E+00 .4058E+00 .4159E+00 .4193E+00 .4186E+00 .4186E+00
H TOTTA H TOTA
0. 1864E-07 -4215E-07 -6506E-07 -5160E-07 -5160E-07 -5102E-07 -5102E-07 -5102E-07 -5112E-07
0. 2826E-07 6557E-07 2657E-07 2257E-07 2144E-07 .2144E-07 .2144E-07 .2144E-07 .2144E-07 .2144E-07
страна с 10143 с 10143 н н 10143 н н 10143 с 10143 с 10143 н н н 10143 с 10145 с 10
**************************************
<pre>.174444.10 36234.10 71364.12 .75354.14 .55354.14 .55354.14 .51354.14 .51354.14</pre>
E&101 = CRT01 = CRT01 = CRT01 = ERT01 = ERT01 = CRT01 = ERT01 =
MODE 0 DIS 0 1 MODE 0 DIS 0 1 MODE 0 DIS 2 1 MODE 0 DIS 3 1 MODE 0 DIS 4 1

GRAD	•
. 180.	
÷	
ā	•
KEI	п
Ĩ.	10
KO1	EPJ
.53 METER	3547E+00
"	00
ST	15+
10	181
AND	•
ISE	Ð
6 A	ERTOT
€,	••
÷	0
GMEN	015
SE (	0
TEST	MODE

•2142E+0C	2433E+00	4016E+00	2452E+00	2384E+00	24956+00	24945+00	2496E+00
.22036+00	.5737E+00	.2070E+00	.1643E+U0	• 2123E+00	.2124E+00	.2102E+00	.2103E+00
ETT01 =	ETTOT =	ETT0T =	ETTOT =	ETTOT =	ETT 0T =	ETT0T =	ETTOT =
<b>.</b> .	.1491E-07	5259E-07	89145-07	5930E-07	5299E-07	5569E-07	5603E-07
<b>.</b> .	.2087E-U7	.62286-07	.2654E-08	1564E-07	5351E-08	3700E-08	4267E-08
EPT01 =	£PT01 =	= 10143	EPT0T =	E 10143	EP107 =	EPT07 =	EPT0T =
3547E+00	.4138E+00	*7752E+00	. 5782E+00	.53156+00	•5429E+00	• 5450E+00	•5447E+00
.18116+30	330 ZE+30	.19045+0C	.3447E+00	.29135+30	•2803E+00	.2823E+30	.28265+00
ERTOT =	ERT07 =	ERTOT =	ERTUT =	EPT01 =	EXTOT =	ERTOT =	ERI0T =
••	••	••	••	••	••	••	
6	-	ده د	m 	\$	5	و م	~
015	516	310	516	10	310	515	BIS
0	0	•	-	e	•	c	•
MODE	MODE	MODE	MODE	MOCE	MODE	MODE	MODE

	FITO
180. GRAD	.0
RU1.WINKEL PHI =	EPTOT = 4.
.69 METER	25256+00
AND DIST =	.1741E+30
7 AHST	ERTOT =
SMENT NR	BIS 0 :
TESTSE	MODE 0

1 .1066E+L	03129E+	15808E+	03060E+L	12763E+L	13069E+L	13079E+L	130646+1	
• 8880 E-U	.4725E+0	1785E-0	1293E+0	<523E-0	1863E-U	2598E-0	2608E-0	
"	-	*	*	1		-	1	
ETTOI	ETTON	ETTO1	ETTOI	E1101	ETT 01	ETTON	ETT 01	
.0	.1085E-07	5562E-07	1034E-06	53266-07	3984E-07	47065-07	4823E-07	
.,	.1298E-07	.5374E-07	2413E-07	5483E-07	3294E-07	2852E-07	3042E-07	
n	н	11	11	n	N	H	н	
LPT01	EPTOT	EPTOT	EPT0T	EPTOT	EPTOT	EPTOT	ÉPT0T	
2525E+00	.3687E+00	.7642E+00	• 4636E+00	• 3745E+00	.4027E+00	.4092E+00	• 4080E+00	
.1741E+30	1942E+00	•;4032E+80	.63575+90	•5311E+30	• 5 ü 44E+00	•5106E+30	•5119E+00	
11	0	н	ß	Ħ	n	0	11	
ERTOT	ERTOT	ERTOT	ERTOT	ER101	EPIOT	ER 101	ERTOT	
••		••	••	••	••	••	••	
0		~1 ~	m 	÷.	ۍ ۱۵	و د	~	
BIS	516	318	315	319	18	818	318	
		<b>c</b> •	0	<b>.</b>	<u>.</u>	0	•	
<sup>1</sup>	100	100	100£	ĝ	10 D	100 100	100	

	ETTOT =
18 <b>9.</b> GRAD	
ROT.WINKEL PHI =	EPTOT = 0.
.83 METER	14296+00
A3STAND DIST =	0 <u>F</u> = .1434E+ <u>3</u> 3
£	ERIC
SEGMENT NR	0 BIS 0
TEST:	HODE

-.4312E-02 -.2379E-01

• ت ۍ 000

2335E+0	2497E+00	0	ETT01	2900E-07	4942E-07	n	EPTOT	1525E+00	00+36649.	H	ERTOT	••	<u> </u>	015	ш ш	100
- 2363E+0	2487E+00	Ħ	ETTOT	2593E-07	4441E-07	H	<b>EPTOT</b>	.15596+00	•5463E+3N	11	ERIOT	•	0	81S	ы ш	100
2330E+0	2292E+00	16	ET 101	1027E-07	5401E-07	It	EPT01	•14056+00	• 5 3 0 9E + 00	н	ERTOT	••	5	9 BIS	ت س	100
1654E+0	2518E+00	91	ETTOT	3420E-07	9305E-07	n	EPTOT	.3421E-01	.54246+99	11	ERTOT	**	*	I BIS	<i>ш</i>	1001
2428E+0	4413E+00	H	ETTOT	1069E-06	4850E-07	11	EPT01	.2244E+00	.9531L+00	н	ERTOT	••	m 	BIS 0	ې س	100
6434E+0	2423E+00	#	ETTOT	51586-07	.41726-07	Ħ	EP101	.6162E+00	•5702E+00	н	ERTOT	••	~	O DIS	ш ш	1001
2986E+0	• 3094E+00	11	ETTOT	.69385-08	.5842E-08	a	EPT01	.2525E+00	3160E-01	11	ERTOT	•	-1	312	وت النا	1001

ROT.WINKEL PHI = 180. GRAD A3STAND DIST = 1.03 METER 5 TESTSEGMENT NP

1824E+J0	1616E+00	5126E+00	.2299E-01	.1901E+00	.3803E-01	.16996-01	.33396-01
6683E-01	.3998E-01	4478E+00	7570E+00	4217E+00	3552E+00	4093E+00	4154E+00
H	H	0	n	H	n	II	н
ETIOT	ETTOT	ETT0T	ETT 0T	ETT0T	ETTOT	ETIOT	ETT 0T
3.	.2524E-08	3774E-07	95056-07	• 5262E-08	.47585-07	.1273E-07	.4217E-08
•0	117dE-08	.2352E-67	6997E-07	1315E-06	624+E-07	41376-07	5496E-07
H	0	ų	ų	M	Ħ	ŋ	n
EPT 0 T	EPTOT	EPTOT	EPT0T	EPT0T	EPT01	EP101	LPT0T
27836-01	.460E-01	.28135+00	1776E+00	78246+00	2712E+00	2342E+00	24476+00
.99686-31	14366+30	€234c+30	•3262E+30	.5627E+J0	.5644E+ 3	.50195+03	•5134E+30
ERTOT =	ERTUT =	= 101d3	EKT0T =	ER10T =	EPTOT =	ERTOT =	ERTUT =
	••	••	••	••		••	••
-	-1	~	m		5		~
ers	ers	BIS	OIS	BIS	BIS	518	EI S
e	•	0	0	0	0	0	0
MODE	MODE	HODE	MODE	HODE	MODE	MODE	MODE

........

ع

U 11 11 11 11 11 11 11 11 11 11 11 11 11	A Z I M U T A 3 * VERTIXALEN POT 0 URCHMESSER = ACHSENLACHGE = ACHSENLACHGE = ACHSENLACHGE = A HOSTAINL UER MCS # HOSTAINLER HO # HOSTAINTER HO # HOSTAI	L S T 7 A H L U N G S J I A G R A M M F R E Q U E N Z = 164.0 M H Z 514. Metricimod. Ressantenne i puktantenne auf Einstrahlung i minkel theta = 80.0 Grad 20 metri 1.69 metri 1.69 metri 1.69 metri 1.69 metri 1.69 metri 1.69 metri 1.60 grad	SANTENAE VAN TER KOERPEROBERLAECHE IN METER Taitonskingel in Grau = Schatten Obe Aessaytevpe in OB = L 0J = Max,Freespace Unites II 7an Lifuue der Felnstaerke im Messpunkt Lifuue der Felnstaerke im Messpunkt		
	A Z I M U T A L A VERTIXALEK POT.3Y1. DUBCHTES: CR A CHSCRLAERGE = 1.6 C EINSTFAILUNU E EINSTFAILUNU A STAIU UER MESSANT A ABSTAIU UER MESSANT A ABSTAILUNU A APPLITU G A IN FHASE DE. 05 PISS POINKT A PLITU G A IN FHASE - 25 U3 -5.6 137.11 1.88 144.32 -5.13 156.39 -5.29 123.91 -5.29 123.91 -1.95 70.02 -1.95 70.02 -1.95 16.20	STRAHLUNGS J Hethochhod, Messantenn Meter 9 Meter 9 Meter 64AD	ZNNE VNH TCK KOERPEROBER DISHTITCL IN GRADI 180 G AGSSATTETVE IN DB 2 U 0 55 I' 73AT UE DER FELNSTAERKE IM ME	1- 00 51- 	

- 279 -

0151	GAIN	FHA SE	-25 08	-23 33	-15 00	-10 08	-5 08	<b>60 0</b>	+5 08	-180	96 <b>-</b>	c	16 +	+180
.08	+10.21	-157.77	•	•	•	]	•	•	•	•	•	•	•	•
•13	-7.03	-157.28	•	•	•		./	•	•	•	•	•	•	•
.18	-5.01	-156.11	•	•	•	•	4	•	•	•	•	•	•	•
- 20			•	•	•	•		•	•	•	•	•	•	•
• 28	-2.43	-152.63	•	•	•	•	•	•	•	•	•	•	•	•
.30			•	•	•	•	•		•	•	•	•	•	•
.38	82	-148.52	•	•	•	•	•	۸.	•	•	•	•	•	•
3			•	•	•	•	•	·	•	•	•	•	•	•
-			•	•	•	•	•	~	•	•	•	•	•	•
. 53	170	-141.65	•	•	•	•	•	. <b>A</b> .	•	•	•	•	•	•
. 55			•	•	•	•	•		•	•	•	•	•	•
.60			•	•	•	•	•		•	•	•	•	•	•
. 68	1.53	-134.59	•	•	•	•	•	۰.	•	•	•	•	•	•
. 70			•	•	•	•	•		•	•	•	•	•	•
. 75			•	•	•	•	•	-	•	•	•	•	•	•
• 83	1.88	-127.74		•	•	•	•	•	•	•	•	•	•	•
.85			•	•	•	•	•	•	•	-		•	•	•
<b>96</b> .			•	•	•	•	•	•	•	•	•	•	•	•
• 95			•	•	•	•	•	-	•	•		•	•	•
1.03	1.76	-119.51	•	•	•	•	•	۰.	•	•	•	•	• •	• •
	,													
E	= 180 <sup>0</sup>		AMPLITU	0E DER FEL	DSTAERKE I	H HESSPUNKI				PHASEI	NLAGE DER I	FELOSTAE	RKE IN MES	SPUNKT
DIST	GAIN	PHASE	-25 08	6U iZ-	-15 08	-10 08	-5 08	0 08	4C 24	-186	J6 -	J	n6 +	+180
. 0.8	-14.23	-69.27	•	•	/ 4 •		•	•	•	•	•	•	•	•
	-10.92	-56.02	•	•	•	., 4	•	•	•	•	<u>م</u>	•	•	•
•18	-9.02	-48.19	•	•	•	Å.	•	•	•	•		•	•	•
• 20			•	•	•		•	•	•	•	•	•	•	•
8 Z P	-5.82	-31.71	•	•	•	•	•	•	•	•	•		•	•
	5 ° ° -	110-06	•	•	•	•		•	•	•	•		•	•
			•	•	•	•		•	•	•	•		•	•
5.7					•	• •	~	•	•	•	•	~	•	•
53	-4.33	12.18	• •	• •			۰ ۹		• •	• •	•	: .	•	•
			•	•		• •		•	• •	•	•	•	•	•
99.			•	•	• •	• •		• •		• •	•		•	•
.58	-3.59	39.50	•	•	• •	•			• •		• •	•	•	•
.70			•	•	•	•		• •		• •	• •			•
.75			•	•	•	•		• •		• •		• •		• •
. 83	-3.09	67.36	•	•	•	•	۰.	•	•	•	• •		• •	• •
• 85			•	•	•	•		•	•	•	•	•	•	•
.90			•	•	•	•	•	•	•	•	•	•	~	•
- 95			•	•	•	•	-	•	•	•	•	•	1.	•
1.03	-2.59	104.64	•	•	•	•	•	•	•	•	•	•	٩.	•

16.2.4. PROGRAM PANC, DATA CARDS FOR TEST BODIES FZYL, MANMOD 1 AND MANMOD 2, FIELD HOMOGENEITY RESULTS WITH FZYL

·. .

E DEPOSITION REALINE CONCERTATION OF THE HOMOGENEITY OF THE FIELDS	PANC		PP= CMPLX(1.0E-32.1.0E-32)	ANC 52
1、「FROGRAM」「FAMO」。 しんいFOLMALAON OF 1115 124100011111 とうしょう いっとうかから またまたまたまたまたまたまたまたまたまたまたまたまたまたまたまたまたまたまた	PANC	~~~	QQ=1.0E-32 \$ Q=10000.	22
E CONTRACE CARDS	PANC	m	PR=180./PI	
PAN, 3571, CM70000, CT5.	PANC	<b>a</b> 1	TEXT(1) =10HVERTICAL TEXT(2)_10HUDDI70HTAL	28
FTN.	PANC	nv		ANC 57
ATTACH, HOMOG.	PANC	01	KEMINU 0 Po 701 Pol -1 2	ANC 58
CALL, F10005, P=HOMUG, B=DISK.	DAMC	- a	READ/61/CYV/T1 T-1 NST)	ANC 59
LGO. R HERF FND OF RECORD CARD	PANC	°0	READ(6)(STH(L), L=1, NST)	ANC 60
	PANC	₽	READ(6)(SYS(I), I=1, NST)	ANC 61
PROGRAM PANC (INPUT, OUTPUT, DISK, TAPE1=INPUT, TAPE3=OUTPUT,	PANC	= 9	520 FORMAT(1H1,1X, 	
XTAPE6=DISK)	PANC	24	T*A 2 IM UINAL AND	ANC 64
C THIS PROCRAM TO AN EVTENSION TO DECTEAN DAME AND CONDINES THE FIELD	PANC	2	2 0.1 METER DIFIDIE ANTENNA#//140.1X. #TESTBODY: ROT.SYM.CYLINDER*, 8X	ANC 65
C HONOGENETTY ALONG A DIPOLE ANTENNA (2H=0. IM) ORIENTED (P1) VERTICALL)	C, PANC	Έ	4*FIELD FOINT*, 7X, *INCIDENT WAVE*/1H , 11X,	ANC 66
C HORIZONTALLY OR RADIALLY, SPACED (DAT= 0. 1M) FROM THE BODY SURFACE	PANC	91	5*AXIAL LENGTH =*,F5.2,* M DAT =*,F4.2,* M*,7X,*POLAR. = *,H1U/II	ANC OF
C AND FOR BOTH INCIDENT FIELD FOLARIZATIONS (P2).	PANC	<b>2</b> 9	6 , IIX,*DIAMELER =",F5.2," M ND =",F4.2," M',IA,"INCLA =",	ANC 69
C THE OUTPUT IS A TABLE WITH FOLLOWING INFOMMATIONS:	PANC	2₽	(1), 1), 050, 1) 501 FORMBIT(140 1Y 5Y)	ANC 70
C PHL: HUKILONIAL KULAIIUN ANGGE VALINUTARI ANGGE) C CAIN CENTER: ETELD COMPONENT AT THE CENTER POINT OF THE ANTENNA	PANC	2	1*VERTICAL POLARIZED ANTENNA*, 13X, *RADIAL POLARIZED ANTENNA*/1H0, 1	ANC 71
C PHASE CENTER: PHASE OF THE FIELD COMPONENT AT THE CENTER POINT	PANC	2	2, *PHI GAIN PHASE MEAN MAXIMUM MAXIMUM <sup>*</sup> , 4X,	ANC 72
C MEAN ERROR: LOCARITHMIC DIFFERENCE (DELTA U) BETWEEN THE INDUCED	PANC	ខ	3 #CAIN PHASE MEAN MAXIMUM MAXIMUM*/1H, 1X, 5X,	
C VOLTAGE (CENTER E * 2H) AND THE ACTUAL INDUCED VOLTAGE ( INTEGRAL E	PANC	ແ	4 *CENTER CENTER ERROR GAINVAK PHASEVAR <sup>*</sup> ; 3A, * *CENTER PERCOR CAINVAR PHASEVAR*;	ANC 75
C FROM -H TO +H)	PANC	1	5 *CENTER CENTER EXHOR GALINARIA FILAGEVAN" /	ANC 76
C MAXIMUM CAINVAR: LOCARITHMIC DIFFERENCE (SMALL UELIA E/ BELWEEN INE	DANC	3 %	1440RTZCNTAL, POLARIZED ANTENNA <sup>#</sup> , 11X, *RADIAL POLARIZED ANTENNA <sup>#</sup> /1HO	ANC 77
C FIELUS AT THE ENUS OF THE DIFOLE ANIENNA.	PANC	22	21X #PHIT GAIN PHASE MEAN MAXIMUM MAXIMUM*, 4X,	PANC 78
C OF THE FIELDS AT THE ENDS OF THE DIPOLE ANTENNA.	PANC	8	3 *GAIN PAINSE MEAN MAXIMUM MAXIMUM*/7H, 1X,5X,	ANC 79
C PRIOR TO THE RUN OF THIS PROGRAM PANC THE FOLLOWING COMPUTATIONS	PANC	ଝ	A *GENTER CENTER ERROR CAINVAR PHASEVAR*, 3X,	ANC 80
C ARE NEEDED:	PANC	2	5 *CENTER CENTER ERROR GALNVAR PHASEVAN")	ANC 82
C Y-MATRICES WITH PROGRAM HARRA (FZYL DATA SET, MODEL FREQUENCY TO MHZ	, PANC	58	Dec FURMALLIN, IA, 14DFG DB DE DB DB DEG#, 8K,	PANC 83
C FULL U 10 () C FTFID PATURS WITH PROCRAM PANE (FZYL DATA SET. MODEL FREQUENCY 10 MH	Z, PANC	ŝ	2 *DB DEG DB DB DEC*,/)	ANC 84
C KK-8. RUN=1. NTEST=5, TESTSECMENTS 2.25 10.0 0.2 / 1.75 10.0 0.2 /	PANC	ħ	523 FORMAT(1H , 1X, I3, 2X, F5. 1, 1X, F6. 1, 2X, F5. 2, 1X, F5. 1, 1X, F6. 1, TX,	PANC 85
c 2.75 10.0 0.2 / 2.25 9.5 0.2 / 2.25 10.5 0.2)	PANC	ይ	1  F5.1, 1X, F0.1, ZX, F5.2, 1X, F0.1)	DANC 00
C THE RESULTS OF PANE SHOULD BE STORED IN THE FILE F10005 IN THE COLLE	CIPANC	R.F	WRITE( $3,520$ ) IFIX(F), DA, DL( $1$ ), IEAL (PUL), DV, FU, A	PANC 88
C FILE HOMOG.	CHANC	58	IF(FUL:E4.2) GUIU 1 UDTTE/2 E21) & COTO 2	PANC 89
2 法非法法法法法法法法法法法法法法	PANC	88	1 WRITE(3,524) \$ 60TO 2	PANC 90
CCMPFEX_SYV(185)_SYH(185)_SYS(185)	PANC	9	2 CONTINUE	PANC 91
REAL F. DA. DU.HO.X1	PANC	1	WRITE(3, 522)	
INTEGER TEXT (2), POL, NNPHI	PANC	2		
COMPLEX PP	PANC	<u></u>	IH1=J-2 \$ [H2=J+2 \$ NC=NNH1~1 Tri FO \$ 1 TH3 1.0 \$ TE(1 TO 2) TH1-1	PANC 95
DIMENSION DI (5)	DANC	\$ ¥	TF(J, FO, NC) TH2=J & JF(J, FO, NPHI) TH2=J-2	PANC 96
READ(1, 10) F	PANC	淂哥	ITE1 = J	PANC 97
$DA=1.8 \pm DI(1)=0.1 \pm DU=0.25 \pm HO=1.0 \pm X1=80.8$	PANC	11	ITE2 = JANNAHI	PANC 98
NNPH1=37 \$ NST=185	PANC	° ₽	ITE3 = J4NNPHI#2 TTEA - 1.NNPHI#2	PANC 100
PI=3, 141593	DANG	6.5	THE PART AND A	PANC 101
$C_{S} = (DU/2.401(1))^{-1} (INRTA1-1)$ FAC=DI(1)/(4.*CS)	PANC	5	-111 = (1 + 2) - 2	PANC 102

.

	1.2500 1.2500 1.2500 0.4735	3.6577 8.5144 13.3712 17.9068		1.2800 1.6600 1.7000 0.5000	3.6700 8.2700 12.9400 16.6500		0.525 0.850 0.500	3.680 9.200 13.510
1 0 N	1.2500 1.2500 1.2500 0.8738	3. 1720 8. 0288 12. 8855 17. 64 14		1.2300 1.6600 1.6000 0.7600	3.2000 7.8000 12.5000 16.4100		0.525 1.050 0.930 0.700	3. 120 8. 700 13. 190 16. 420
	1.2500 1.2500 1.2500 1.1488	2.6863 7.5431 12.3998 17.2426	TRIC	1.1500 1.6500 1.5500 0.8500	2.7800 7.3300 12.0500 16.0000	TRIC	0.515 1.050 1.000 0.820	2.580 8.240 12.820 16.050
SCR	1.2500 1.2500 1.2500 1.2499	2.2007 7.0574 11.9141 16.7707	AL SYMME	1.0700 1.5800 1.5000 0.8200	2.3000 6.8400 11.6000 15.5800	AL SYMME	0.515 0.925 1.040 0.870	2.180 7.630 12.550 15.700
Y D E	1.2500	1.7150 6.5717 11.4285 16.2852	ROTATION	0.9000 1.5100 1.4800 0.6500	1.8500 6.4200 11.1100 15.1500	ROTATION	0.500 0.780 1.070 0.850	1.740 7.060 12.270 15.400
T B O D D CAPS	1.2499	1.2293 6.0861 10.9428 15.7995	T VIEW,	0.7500 1.4400 1.4500 0.6500	1.4200 6.0000 10.6300 14.6500	E VIEW,	0.500 0.710 1.100 0.760	1.270 6.510 11.820 15.000
T E S	1.1488 1.2500 1.2500	0.7574 5.6004 10.4571 15.3139	ODY, FRON	0.6500 1.3800 1.4500 0.8500	0.9500 5.4900 10.1700	ODY, SID	0.515 0.660 1.085 0.620	0.760 5.960 11.340
LINDER W	ADIUS) 0.8738 1.2500 1.2500 1.2500	IGHT) 0. 3586 5. 1147 9. 9714 14. 8282	HUMAN B	ADIUS) 0.6700 1.3300 1.5300 1.2500	IGHT) 0.4700 5.0300 9.7000	HUMAN B	ADIUS) 0.515 0.625 1.020 0.490	IGHT) 0.350 5.500 10.770 14.550
C A R FZYL: CY	FONENT R 0.4735 1.2500 1.2500 1.2500	PONENT H 0.0932 4.6290 9.4858 14.3425	HANMOD 1:	PONENT R 0.7000 1.3000 1.6100 1.7000	PONENT H 0.0000 1.6000 9.2000 13.8000	MANMOD2:	FONENT R 0.500 0.980 0.480	PONENT H 0.000 1.840 10.270 14.310
D A T A	RH (X-COM 0.0000 1.2500 1.2500 0.0000	ZH (Z-CCM 0.0 4.1434 9.0001 13.8568 18.0000	TESTBODY	RH (X-COM 0.0000 1.2600 1.6800 1.6800 1.7000 0.0000	ZH (Z-CCM 0.0000 1.1000 13.3600 16.8000	TESTBODY	RH (X-COM 0.000 0.575 0.580 0.630 0.630	ZH (Z-COM 0.000 1, 260 9.740 13.920 15.800
PANC 103 PANC 104 PANC 105 PANC 105 PANC 105	PANC 108 PANC 108 PANC 109 PANC 111 PANC 112 PANC 113 PANC 113	PANC 115 PANC 115 PANC 116 PANC 117 PANC 119 PANC 120 PANC 120 PANC 121	PANC 123	PANC 125 PANC 125 PANC 125 PANC 127 PANC 127 PANC 129 PANC 130	PANC 132 PANC 132 PANC 133 PANC 134 PANC 135 PANC 135 PANC 136 PANC 136	PANC 140	PANC 142 PANC 143 PANC 144 PANC 145 PANC 145 PANC 147 PANC 147 PANC 148	PANC 149 PANC 150 PANC 151 PANC 152
TO 4 TE1) H1) H2)	T. QO) SYV(TTE1)=PP T. QO) SYV(TTE1)=PP T. QO) SYV(TTE1)=PP T. QO) SYS(TTE1)=PP T. QO) SYS(TTE3)=PP T. QO) SYS(TTE3)=PP T. QO) SYS(TTE1)	\$ AMR=GABS(SYS(TTE2)) \$ AMR=CABS(SYS(TTE2)) \$ AMR=CABS(SYS(TTE2)) FE5), ATMAG(SYV(TTE3)) PR FE3), ATMAG(SYV(TTE3)) PR FE3), ATMAG(SYS(TTE3)) PR FE3), ATMAG(SYS(TTE3)) PR	DPV2=360DPV1 DPR2=360DPV1	S(DPV1)) DPV1=DPV2 & DPV=ABS(DPV1) S(DPR1)) DPP1=DPR2 & DPR=ABS(DPR1) BVV1-ANV4 ANV4=ANV1-D2*FAC 5500 \$ IF(ANV4.LT.00) AMV4=00	1) \$ AR1=20."ALLOC10(ANR1) 5) \$ AR3=20."ALLOC10(ANR3) 1) \$ AR3=20."ALLOC10(ANR2) 1) \$ AR3=20."ALLOC10(ANR2) 2011/(ANV5-ANVV4.2."ANV1)) 2011/1/(ANV5-ANV2) 2011/2012/2012/2."ANV1)) 2011/2012/2012/2012/2012/2012/2012/2012	PEVEQ \$ DAVEQ \$ DAVEQ	100 (k=0 \$ DAR=0 \$ DPR=0 (V1, PV1, EAV, DAV, DPV, AR1, PR1, EAR, DAR, DPR	D K CARD

		_																					1 E	7	1	61	5 N	1117	
	7 11 12 60				MAXIMUM PHASEVAR DEG	6.7 6.6	6°2	6.1	ۍ د 1000		- 1 1 1 1 1 1 1 1	- ~ • ~	2.1	1.5	ŗ.	ņ		5.1 5.1	- 0.	80 m m =	, e.	2°5	<b>ل</b> و، ہ	6.3	9.6	0.0	1.7	7.2	
	د د ع		ZONTAL	NNA	AXTMUM A A I NVAR	2.1 2.1	~ ~		2.0		1000	20.0	00	2010	20	5.0	0.0 V V	2.0	20	0.0 2.0	0.0 V	2.0	2.0 2.0	5.0	5.0	2.0	5.0	2°0	
1	- - - -	TENNA	NT WAVE = HORIV = 80.8	ed ante	MEAN MEAN MEAN DB DB DB	* * * *	Ę.	== / /	 	= = \$ / /	:=: //	= = ; ;	; ; ; ;	: ;;;	14 1	4	≓₽ / /	- 11	- 1	1	::	- 11	2 1 1	:	11	1	11	- 17	
4	2) X.	POLE AN	INCIDE POLAR. THETA	POLARIZ	PHASE CENTER DEG	64.7 64.7	555	66.0 66.0	66.6 57.5	89.0	69.7 69.7	2.17	72.8	22	79.2	78.7	79.9 81.1	82.3	88 19	82.6	80.0 87.6	88.5	8.9	8.7	91.3	2.10	8	92.5	
	ж. Э. Г.	IETER DI	FIX	RADIAL	GA IN CENTER DB	-18.6 -18.6		 9 <del>1</del>		999 77	* m (	. <del>.</del>	1.8	1	5°0	5.1	2.4	5.4	2.2	5	2.4	?		i "	1		-12.5	-18.5	
	ΡΑΤ	A 0.1 V	T = .10		IMUM SEVAR	000	n 01 0	5,60	~	0,00,0		≠.~	ŵ.		- 0		ಸಂ	5	ون	1	mo					@ C		0.0	
	NOI	ALONG	E M H	ANA	JM MAX AR PHA DEG	<i></i>	- 11	ಗೆಸ್	<b>.</b>		≈°£°;	= = =	228	v R	115	3 <u>6</u>	8 8 1 8 1 8 1 8 1		25		۳ ۳	~~ > 0		0 ~	י <i>י</i> ד הסה	ш		= 0	
	IAT	FIELD	1.801	ANTEI	MAXIM GAINVI DB	0,0,0			~	N N M	ມ ສີສີ 1	5°-	5	ž≅	ల్లాం ల్లాం		σ̈́ς	,≊ ,≊	20		ശ്⊐ നെ	; <del></del>		• •	:-:		- · ·		;
	RAD	f THE	KTH =	ARIZEI	MEAN Error DB	ຮູຮູຍ	58	88	88	88	88	88	5	<u>8</u> 5	59	55	កុ	10	8,5	3	ਲ ਟ	58	ġ	50	••	••		•	?
	ΊΛĽ	CHECK 0	r. Sym. C Ial Len Ameter	tal. Poi	PHASE CENTER DEG	59.1	. 0.92 . 0.92	60.6 61.5	68.5	63.7 65.2	66.8 68.6	7.07	5.8	83.5 1	9.0	2.6 <u>8</u>	-128.2	-106.1	-101.7	-95-1	6 6 7 7 7 7		-87.9	8	) a	89. 19.			
	MUTI	CNEITY (	DY: RO AX DI	HORIZON	GA IN CENTER DB	ά. 1. ν. ι	ኯ <del>፟</del>	ې ۳۹	-	e.v. عرب	-1.0	0 °	-10.6	-12.4	-18.1	γ'n	-53.1	-10.0	-12.4		6.7-	è è	5	* † =	- - - -	ς. Υ.Υ.	η η η	- - - - -	?
	A 2 I	HOMOCE	TESTBO	<u> </u>	DEG	0.01	55	2%	្ល្រ្ត	<del>ا</del>	훈망	53	5	86	85	88	8	Ξĉ	21	<u> 8</u>	5	зř.	01	5 5 5 5	5£	35	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	12	2
ľ										_																			
- 1													_				_					_							
ł	2				M AR																								
	65 M H Z			-	MAXIMUM Phasevar Deg		0.4 0.4	6. 	- 00 mm		6.0 5'60	۳. ۲	1.5	2. 1. 5	, <b></b>	- ^	ių.	- •		22	2		0.1	ę,	<u>,</u>		<u>.</u> ,,		۰.
	NCY 65 MHZ		ICAL DEG	NNA	IAXIMUM MAXIMUM Alinvar Phasevar Beg	4.3 4.1 4.3 4.1	ц.3 ц.0 ц.3 д.0		н.н 3.6 1.1 3.6	स २ २ स म म म	ц. ц. 2.9 ц.5 2.6	4.5 2.3	4.4 1.5	ц. ц. 1. 2 и и В	-	- ^ -	5	1.1 1.1	4.0 1.0	3.9 7.1	3.8	3.7	3.6	3.6	5. 5. 5.	3.5	4.0°	 	9. 
	QUENCY 65MHZ	TENNA	VT WAVE = VERTICAL = 80.8 DEG	ED ANTENNA	MEAN MAXIMUM MAXIMUM Error gainvar phasevar db db db	- 32 4.3 4.1 - 32 4.3 4.1	31 4.3 4.0	-31 -31 -31 -31 -31 -31 -31 -31 -31 -31	-31 4.4 3.6	31 4.4 3.4	30 u.u 2.9	29 4.5 2.3	28 4.4 1.5	28 4.4 1.2	-26 4.4	- S		23 4.1	22 4.0 1.0	21 3.9 1.1	15 3.8 1.1	19 3.7 1.1	-18 3.6 1.0	17 3.6 .9	- 17 3.5 -9	- 16 3.5 .7	16 3.4 .7	- 16 3.4 .6	16 3.4 .6
	FREQUENCY 65MHZ	OLE ANTENNA	INCIDENT WAVE POLAR. = VERTICAL THETA = 80.8 DEG	OLARIZED ANTENNA	PASE MEAN MAXIMUM MAXIMUM Center Error Gainvar Phasevar Deg db db db	-27.332 4.3 4.1 -27.332 4.3 4.1	-27.531 4.3 4.0	-28.131 4.4 3.9	-28.631 4.4 3.6	-29.9 - 31 4.4 3.4 -30.6 - 30 4.4 3.2	-31.530 4.4 2.9	-33.529 4.5 2.3	-35.728 4.4 1.5	-36.828 4.4 1.2	-39.126 4.4 .4		12.424 4.2 .5		15.122 4.0 1.0		1.1 3.8 1.1		18.6 -18 3.6 1.0	-48.917 3.6 .9		19.6 16 3.5 .7	19.8 - 16 3.4 · 7	19.9 - 16 3.4 .6	-49.916 3.4 .6
	ERN FREQUENCY 65MHZ	TER DIPOLE ANTENNA	INCIDENT WAVE POLAR. = VERTICAL THETA = 80.8 DEG	ADIAL POLARIZED ANTENNA	AIN PHASE MEAN MAXIMUM MAXIMUM Senter Center Error Calinvar Phasevar De deg db db de	-1.7 -27.332 4.3 4.1 -1.7 -27.332 4.3 4.1	-1.7 -27.531 4.3 4.0	-1.7 -28.131 4.4 3.9	-1.7 -28.631 4.4 3.6 -1.7 -29.231 4.4 3.6	-1.7 -29.931 4.4 3.4 -1.7 -30.630 4.4 3.2	-1.7 -31.530 4.4 2.9		-1.7 -35.728 4.4 1.5	-1.7 -36.828 4.4 1.2	-1.6 -39.126 4.4 .4		-1.3 -12.424 4.2 .5		-1.0 -15.122 4.0 1.0	9	7 17.215 3.8 1.1	6 -47.719 3.7 1.1		3 -48.917 3.6 .9	2	1	0 -19.816 3.4 .7	.0 19.9 - 16 3.4 .6	.0 -49.916 3.4 .6
	ATTERN FREQUENCY 65MHZ	0.1 METER DIPOLE ANTENNA	POINT INCIDENT WAVE - 10 M FOLAR. = VERTICAL 1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	M GAIN PHASE MEAN MAXIMUM MAXIMUM AR CENTER CERRER CAINVAR PHASEVAR DB DEC DB DB DEC	-1.7 -27.332 4.3 4.1 -1.7 -27.332 4.3 4.1	-1.7 -27.531 4.3 4.0	-1.7 -28.131 4.4 3.9	-1.7 -28.631 4.4 3.6	-1.7 -29.931 4.4 3.4	-1.7 -31.5 30 4.4 2.9	-1.7 -33.529 4.5 2.3	-1.7 -35.728 4.4 1.5	-1.7 -36.828 4.4 1.2	-1.6 -39.126 4.4 .4	-1.5 -10.225 4.3 .1	-1.3 -12.424 4.25		-1.0 -15.122 4.0 1.0	9 45.921 3.9 1.1	7 -47.215 3.8 1.1			3 -48.917 3.6 .9		-1 -19.616 3.5 .7	0 49.816 3.4 .7	0 149.910 3.4 .6	
	ON PATTERN FREQUENCY 65MHZ	ONG A 0.1 METER DIPOLE ANTENNA	FIELD POINT INCIDENT WAVE DAT = .10 H POLAR. = VERTICAL HB =1.00 H THETA = 80.8 DEG	RADIAL FOLARIZED ANTENNA	MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM MAXIMUM PHASEVAR PHASEVAR CENTER CENTER ERROR GAINVAR PHASEVAR Deg db deg db deg deg db deg	6.5 -1.7 -27.332 4.3 4.1 6.6 -1.7 -27.332 4.3 4.1	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	7.5 -1.7 -28.1 -31 4.4 3.9	8.0 $-1.7$ $-28.6$ $31$ $4.4$ $3.68.7$ $-1.7$ $-29.2$ $31$ $4.4$ $3.6$	9.4 -1.7 -29.931 4.4 3.4 10.0 -1.7 -30.630 4.4 3.2	10.3 -1.7 -31.5 - 30 4.4 2.9 10.0 -1.7 -32.530 4.5 2.6	8.8 -1.7 -33.5 -29 4.5 2.3	u.5   -1.7 -35.728 4.4 1.5	2.4 -1.7 -36.828 4.4 1.2	.6 -1.6 -39.126 4.4 .4		2.3 -1.3 -12.424 4.2 .5	2.5 -1.2 -13.323 1.1 .7	2.6 -1.0 -15.1 -22 4.0 1.0	2.69 15.921 3.9 1.1	2.5 -7 -7 -7.2 -15 3.8 1.1	2.46 -47.719 3.7 1.1		2.33 -48.917 3.6 .9		2.1 -1 -19.6 -16 3.5 .7	2.10 -19.816 3.4 .7		2.0 1 .0 -49.9 - 16 3.4 .6
	ATION PATTERN FREQUENCY 65 MHZ	IELD ALONG A 0.1 METER DIPOLE ANTENNA	FIELD POINT INCIDENT WAVE 80 M DAT = .10 M POLAR. = VENTICAL .25 M HB =1.00 M THETA = 80.8 DEG	TENNA RADIAL POLARIZED ANTENNA	AXTANIM MAXTMUM GAIN PIASE MEAN MAXTMUM MAXIMUM MAXIMUM MAXIMUM Ajiwar Piasevar center center error gaiwar Piasevar B. Deg de deg de deg de deg	1.0 6.5 -1.7 -27.3 -32 4.3 4.1 1.0 6.6 -1.7 -27.3 -32 4.3 4.1	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	.9 7.5 -1.7 -28.131 4.4 3.9	.8 8.0 -1.7 -28.631 4.4 3.6 .7 8.7 -1.7 -29.231 4.4 3.6	5 9.4 -1.7 -29.931 4.4 3.4	0 10.3 -1.7 -31.5 - 30 4.4 2.9	.6 8.8 -1.7 -33.5 -29 4.5 2.3	1.0 4.5   -1.7 -35.728 4.4 1.5	1.0 2.4   -1.7 -36.828 4.4 1.2	.8 .6 -1.6 -39.126 u. h .4	. 6 1.5 - 1.5 - 10.2 - 25 4.3 .1	u 2.3 -1.3 -12.424 4.2 .5	· 3 2.5   -1.2 -13.323 1.1 ·7	. 1 2.6 -1.0 -15.1 -22 4.0 1.0	.1 2.69 45.921 3.9 1.1	.0 2.57 -17.215 3.8 1.1	1 2.4 - 6 -47.7 - 19 3.7 1.1		.1 2.3 -3 -18.9 -17 3.6 .9		2 2.1 -1 -1 -1 -16 3.5 .7	-2 2.10 49.816 3.4 .7		.2 2.0   .0 -49.916 3.4 .6
	LADIATION PATTERN FREQUENCY 65MHZ	THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA	ALINDER FIELD POINT INCIDENT WAVE TH = 1.80 H DAT = .10 H POLAR. = VENTICAL = .25 H HB =1.00 H THETA = 80.8 DEG	IZED ANTENNA RADIAL FOLARIZED ANTENNA	VEAN MAXIMUM MAXIMUM GALIN PIASE MEAN MAXIMUM MAXIMUM Error Galinvar Piasevar Center Center Error Galinvar Piasevar Nac dr deg db db deg db db deg	01 1.0 6.5 -1.7 -7.3 -32 4.3 4.1	.01 .9 6.7 -1.7 -27.5 -31 4.3 4.0	00 .9 7.5 -1.7 -28.131 4.4 3.9	01 .8 8.0 $-1.7$ $-28.6$ $31$ $4.4$ 3.6 $07$ 8.7 $-1.7$ $-29.2$ $31$ $4.4$ 3.6	-03 5 9.4 -1.7 -29.9 -31 4.4 3.4	-07 0 10.3 -1.7 -31.530 4.4 2.9	-10 .6 8.8 -1.7 -33.5 - 29 4.5 2.3	-11 .9 0.9 1 -1.7 -34.528 4.4 1.5	-11 1.0 2.4 $-1.7$ $-36.8$ $28$ $4.4$ 1.2	10 .9 .0 -1.6 -39.126 4.4 .4	08 .6 1.5 -1.5 -10.225 4.3 .1				- 04 .1 2.6 9 45.9 - 21 3.9 1.1	0 .0 2.57 17.215 3.8 1.1	03 .1 2.46 -47.719 3.7 1.1		03 .1 2.3318.917 3.6 .9		03 .2 2.11 49.616 3.5 .7	03 .2 2.10 -19.816 3.4 .7		- 02 .2 2.0   .0 -49.916 3.4 .6
	AL RADIATION PATTERN FREQUENCY 65MHZ	HECK OF THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA	T.SMCALINDER FIELD POINT INCIDENT WAVE AL LENGTH = 1.80 M DAT = .10 M POLAR. = VENTICAL Meter = .25 M HB =1.00 M THETA = 80.8 DEG	. POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	HASE MEAN MAXIMM MAXIMM GALN PLASE MEAN MAXIMM MAXIMM MAXIMM Edner Error Galnvar Phasevar Center Center Error Galnvar Phasevar Foc dr dr dr dr dc dr dr de db db db db db	167.5 .01 1.0 6.5 -1.7 -27.3 -32 4.3 4.1 167.5 .01 1.0 6.5 -1.7 -27.3 -32 4.3 4.1	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	162.600 .9 7.5 -1.7 -28.131 4.4 3.9	159.4 - 01 .8 8.0 $-1.7 -28.6 -31 -31 -3.4155.2 - 07 -7 -8.7 -1.7 -29.2 -31 -4 -3.6$	149.7 - 03 5 9.4 - 1.7 - 29.9 - 31 4.4 3.4 10.7 - 03 5 9.4 - 1.7 - 20.6 - 30 4.4 3.2	1344 - 07 - 0 10.3 -1.7 -31.5 - 30 4.4 2.9	112.8 - 10 .6 8.8 -1.7 -33.5 - 29 4.5 2.3	$100.9 - 11 \cdot 9  0.9  1 - 1.1 - 54.529  7.2  7.2  7.2  7.5  -1.7  -35.7 28  4.4  1.5$	-78.8 -11 1.0 2.4   -1.7 -36.828 4.4 1.2	-62.209 .8 .6 -1.6 -39.126 4.4 .4	-55.808 .6 1.5 -1.5 -10.225 1.3 .1		-12.206 .3 2.5 -1.2 -13.323 1.1 .7	-38.9 -05 .2 2.0 -1.0 -15.1 -22 4.0 1.0	-33.404 .1 2.69 -45.921 3.9 1.1	-29.2040 2.57 17.215 3.8 1.1	-27.503 .1 2.4  6 -47.719 3.7 1.1		-23.403 .1 2.33 -48.917 3.6 .9			-20.303 .2 2.10 49.816 3.4 .7	-20.002 .2 2.10 -49.910 3.4 .6 -10.702 .2 2.0 1 .0 -49.916 3.4 .6	<u>-10 7 - 02 2 2.0   0 -19.9 - 16 3.4 6</u>
	MUTHAL RADIATION PATTERN FREQUENCY 65MHZ	VELTY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	DY: ROT.SYM.CKLINDER FIELD POINT INCIDENT WAVE AXIAL LENCTH = 1,80 M DAT = .10 M POLAR. = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	ERTICAL POLARZZED ANTENNA RADIAL POLARIZED ANTENNA	AIN PHASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM Enter center error gainvar phasevar Exter center error gainvar phasevar de feg de	194 -167.8 . 01 1.0 6.5 -1.7 -27.3 -32 4.3 4.1	19.6 - 166.6 .01 .9 6.7 $-1.7 - 27.5 - 31 4.3 4.0$	20.1 -162.600 .9 7.5 -1.7 -28.131 4.4 3.9	20.5 -159.401 .8 8.0 -1.7 -28.631 4.4 3.1	21:5-149.7 -03 5 9.4 -1.7 -29.9 -31 4.4 3.4 20.5 -149.7 -03 5 9.4 -1.7 -30.6 -30 4.4 3.2	255 - 1341 - 07 - 0 10.3 - 1.7 - 31.5 - 30 4.4 - 2.9 225 - 1341 - 07 - 0 10.3 - 1.7 - 32.5 - 30 4.5 - 2.6	-22.9 -112.8 -10 .6 8.8 -1.7 -33.5 -29 4.5 2.3	$-22.7 - 100.9 - 11 \cdot 9 \cdot 0.9 \cdot 1 - 1.7 - 34.528 4.4 \cdot 1.5$	-21.4 -78.811 1.0 2.4 -1.7 -36.828 4.4 1.2	-20.5 - 69.6 - 10 - 3 - 10 - 3 - 10 - 3 - 10 - 30.1 - 26 - 11 - 3 - 10.5 - 62.2 - 09 - 8 - 6 - 10.6 - 39.1 - 26 - 11 - 3		-1(.5 -50.5015 2.9 -1.3 -1.224 4.2 -5		-15.1 - 38.9 - 05 - 2 - 2.0 - 1.1 - 44.5 - 22 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0 - 1.0	-13.8 -33.404 .1 2.69 45.921 3.9 1.1	-12.7 -29.204 .0 2.57 -17.215 3.8 1.1	-12.3 -27.503 .1 2.4  6 -47.719 3.7 1.1		-11.2 -23.403 .1 2.3 -13 -48.917 3.6 .9	-10.9 - 22.403 - 2 - 2.2 2 - 19.2 - 17 - 3.5 - 9 10.1 - 17 - 3.5 - 8 - 8 10 - 17 - 3.5 - 8 - 8 - 8 - 10 - 17 - 3.5 - 8 - 8 - 8 - 8 - 8 - 8 - 8 - 8 - 8 -			-10.3 -20.002 .2 2.1  0 -49.910 3.4 .6	<u>-10.5 -19.7 - 02 .2 2.0   .0 -19.9 - 16 3.4 .6</u>
	IZIMUTHAL RADIATION PATTERN FREQUENCY 65MHZ	CONCEENELTY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	IESTBODY: ROT.SYM.CALINDER FIELD POINT INCIDENT WAVE AXIAL LENGTH = 1.80 M DAT = .10 M POLAR. = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	VERTICAL POLARZED ANTENNA RADIAL POLARIZED ANTENNA	THI CAIN PHASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM Center Canter Cainvar Phasevar Center Center Error Cainvar Phasevar Dec Dr Dr Dr Dr Dr Dr Dr Dec Dr	6 -194 -167.8 -01 1.0 6.5 -1.7 -27.3 -32 4.3 4.1	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	20 -20.1 -162.600 .9 7.5 -1.7 -28.131 4.4 3.9	25 - 20.5 - 159.401 .8 8.0 -1.7 -28.631 4.4 3.7 an 21.0 - 155.202 .7 8.7 -1.7 -29.231 4.4 3.6	35 21.5 -149.7 -03 5 9.4 -1.7 -29.931 4.4 3.4	Terror         Control         Control <th< td=""><td></td><td>60[-22.7-100.911 .9 <math>6.9</math> <math>[-1.1 -34.529 4.4</math> <math>1.5</math> <math>60[-22.728 4.4 1.5</math></td><td>70 -21.4 -78.8 -11 1.0 2.4 -1.7 -36.828 4.4 1.2</td><td>75 -20.5 -09.8 -110 .9 .0 -1.6 -30.1 -26 4.4 .4 .4</td><td><math>\begin{array}{c ccccccccccccccccccccccccccccccccccc</math></td><td>90 -1(1,5 -50.50(1 -2 -2.0 -1.3 -1.2 -1.2 -1.2 -2.9 -1.2 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5</td><td>100 -15.8 -12.206 .3 2.5 -1.2 -13.323 1.1 .7</td><td>105 -15.1 -38.905 .2 2.0   -1.1 -4.222 4.0 1.0</td><td><math>115 - 13.8 - 33.4 - 04 \cdot 1 \cdot 2.6 - 9 - 15.9 - 21 \cdot 3.9 \cdot 1 \cdot 1 \cdot 2.6 - 13.8 - 31.4 \cdot 1 \cdot 2.6 - 10 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot </math></td><td>125 -12.7 -29.204 .0 2.57 -17.215 3.8 1.1</td><td>130 -12.3 -27.503 .1 2.46 -47.719 3.7 1.1</td><td>135 -11.9 -25.903 .1 2.4 3.1 1.1 1.1</td><td>145 -11.2 -23.403 .1 2.33 -48.917 3.6 .9</td><td></td><td></td><td>165 -10.4 -20.303 .2 2.10 -49.816 3.4 .7</td><td>170 -10.3 -20.002 .2 2.1  0 -49.910 3.4 .6</td><td>1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2</td></th<>		60[-22.7-100.911 .9 $6.9$ $[-1.1 -34.529 4.4$ $1.5$ $60[-22.728 4.4 1.5$	70 -21.4 -78.8 -11 1.0 2.4 -1.7 -36.828 4.4 1.2	75 -20.5 -09.8 -110 .9 .0 -1.6 -30.1 -26 4.4 .4 .4	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	90 -1(1,5 -50.50(1 -2 -2.0 -1.3 -1.2 -1.2 -1.2 -2.9 -1.2 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5	100 -15.8 -12.206 .3 2.5 -1.2 -13.323 1.1 .7	105 -15.1 -38.905 .2 2.0   -1.1 -4.222 4.0 1.0	$115 - 13.8 - 33.4 - 04 \cdot 1 \cdot 2.6 - 9 - 15.9 - 21 \cdot 3.9 \cdot 1 \cdot 1 \cdot 2.6 - 13.8 - 31.4 \cdot 1 \cdot 2.6 - 10 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.6 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot 2.0 \cdot 2.0 \cdot 1.1 \cdot 2.0 \cdot $	125 -12.7 -29.204 .0 2.57 -17.215 3.8 1.1	130 -12.3 -27.503 .1 2.46 -47.719 3.7 1.1	135 -11.9 -25.903 .1 2.4 3.1 1.1 1.1	145 -11.2 -23.403 .1 2.33 -48.917 3.6 .9			165 -10.4 -20.303 .2 2.10 -49.816 3.4 .7	170 -10.3 -20.002 .2 2.1  0 -49.910 3.4 .6	1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2

N						_	_	_	_	_	_					_			_	_			_						_	
Ξ				FUM										_				•				~	~		F	ZY	L	75	M	Hz
75 M				MAXII	*****	7.6	 	9.9	9.9	0 0	5	8 ° 7 4	5	- 4 - 1	- 8	27	: ••		- ~		ມ ແ ຕໍ່ສ	=			6.1	-		~	2° 0	
CΥ		DEG	¥,	NUAR	***	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~											:	0 0 7 0	200	5.0	0.0	5.0	5.0 5.0	2	2.1			1.2		
) E N	VA 1 AVE	JORIZ(	ANTEND	N MA DB DB	** **	ŝ	<u>8</u> 8	<u>o</u> <u>e</u>	1		:=	22	₽		:=	55	:=	51	==	:=	22	=	5		1	<b>F</b> !	55	1	2 €	
EQ	ANTEN <sup>®</sup>	ни ни	IZED /	MEA ERR DB	**	1	i	i i	1	i i	1	: ·	1	i	ii	i i	; ;	i	i i 	' i	••	' i	i .a.	i i 	```	' 	; i ~~	· ·	i n r	
8	POLE /		FOLAR	PHASE CENTE DEG	***** 61-3	6.5	61.8	32	63.4	5.0	6.9	67.0		5.0 2	13.2	1.5		7.8.7	200	8	5 %	8	6.6	82	8	5	59	ŝ	s a	
R N	ER DII		DIAL 1	NTER	* * *	5.0		- 6	- ਜ ਨ	~~~	17	a	25	<b>6</b> , - , 0	14	1 0 1 0	- 8.	2.1	0 U 1 0	5.5	6. 6. 4	2	ΰ	2	0.0	∾, m	ר- ער קיק	0		
I I	1 MET	¥¥ 28	RA	558	17	ī T	•																			·	· ·		<u> </u>	1 1
P A	ELD P	н. Н.		SEVAR	0-	- m	<b>.</b>	<u>_</u>		مو	, m	r- 0	9.00	0,0	0	N 0	o m	0,1	<u>.</u>	-	œ. =	. –	ŝ	- «	-	و	<b>∿</b> =	ŝ	N. •	- 9
I O N	ALONG	₫£ 	N	M MAX R PHA DEC	d -	- ~;	<b>ന്</b> =	รับร	6	~ 0	ē	=:	çΨ	ב <u>י</u> א	12	Ë₿	s E	ŝ,		ŝ	φĽ	20	Ξ	2œ	-	9	ທີ່ສ	m	~ ~	-0
AT	IELD	8.2	ANTEN	AXIMI AINVA B	0.0	?:-	22		:	80 m N N	10	8°9		00	17.9	24.5	,	2.6	2. 2 2	12.0	6 F	5.8	8.4	0.4	3	2	 	23		
ADI	THE F	- 	RIZED	ROR P	શ્વ	98	ຣູຊ	88	38	88	38	8,8	38	5,8	35	នុរ	ñ	۳. ۳	នុខ	5	ŝ	8	8	88	8	8	s's	8	ຮຸຍ	ŝŝ.
L R	жоғ м.сп	LENG	POLAF		L. 0	<u>.</u>	~ "	Ů.R.		«		ŵ,	أم		2-		, n	י יי	ņ.	<u>.</u>	ŝ	<u>,</u>	ņ	~~~~		ņ	50	-	<u>م</u> ،	ບສ.
H A	r CHEC	DIAME	DNTAL	R CENT	15	18	53	Ś.C	ŝ	33	50	æ.	82	₽°2	28	26	<u>8</u> 8	÷.	9 	38	6¥	የቸ	5	φ¥	5 KG	ដី	ም ዓ		සි දි 	78 78
L N M	CNEIT)		10fi IZ	CENTEI DB	0.0 10	i n n	ιψ. Wi	71	1	5.0 7 4	ŶŶ	0.1-	200	-10.7		-18.0	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	-22.8	6.2 - 1		-10.5	1	Ý	φų	F	ร	ή'n	i T	ή	79 79
IZV	HOMOCH TESTB(			DEC	0 4	2	ΰ	₹ K	28	ЮЗ	<u>ئ</u>	2	3	£	212	8	88	5	8 4	22	115	3 22	130	ы К	5	5	εç ş	5	21	5 <u>8</u>
F		1										Ó		_					_								_			_
N		1		- 4							-																			
ZHWS				AXIMUM ASEVAR 5G	0.0							0.0		1.6	1.2	ون	•••	0	ņ٠		<u>.</u>		1			1.4	= (	<u>.</u>	<u>.</u>	<u></u>
Y 75 MHZ				HUM MAXIMUM VAR PHASEVAR DEG	2.0	1 2.1	1 2.1	1.2	2.1	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		2.0		3.1.6	3.1.2	6. 6.		0	~~ ~		0.1.0					6 1.4	رن 1		4 1.3	
ENCY 75MHZ	VE	RTICAL .8 DEG	TENNA	MAXIMUM MAXIMUM Gainvar Phasevar DB DEG	4.1 2.0	4.1 2.1	4.1 2.1	4.1 2.1 # 1 2.1	4.2 2.1	4.2 2.1	4.2 2.1	1.2 2.0	1.8	1.6	4.3 1.2	4.3 .9		1,2			1.0	2.4	3.8			7 3.6 1.4	7 3.5 1.4	0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	5 3.4 1.3	5 3.4 1.3 5 3.4 1.3
QUENCY 75MHZ	ttenna Nt Wave	= VERTICAL = 80.8 DEG	ED ANTENNA	MEAN MAXTMUM MAXIMUM Error Cainvar Phasevar DB DB DB	31 4.1 2.0	31 4.1 2.1	-31 4.1 2.1	-30 4.1 2.1	-30 4.2 2.1	30 4.2	20 4.2 2.1	29 4.2 2.0	-28 4.3 1.8	-28 4.3 1.6	26 4.3 1.2	26 4.3 .9		24 4.2 0	-23 4.2	22 4.1 .8	21 4.0 1.0	- 20 3.9 1.3	1.1 8.6	- 18 3.7 1.4	- 17 3.6 1.4	17 3.6 1.4	17 3.5 1.4	16 3.4 16 3.4 1.1	16 3.4 1.3	16 3.4 1.3 16 3.4 1.3
FREQUENCY 75MHZ	LE ANTENNA NCIDENT WAVE	OLAR. = VERTICAL HETA = 80.8 DEG	DLARIZED ANTENNA	ASE MEAN MAXIMUM MAXIMUM ENTER ERROR GAINVAR PHASEVAR 30 DB DB DEG	2.531 4.1 2.0	2.431 4.1 2.1	2.2 - 31 4.1 2.1	2.030 4.1 2.1	1.330 4.2 2.1			-1.129 4.2 2.0	-2.828 4.3 1.8	-3.828 4.3 1.6	-6.026 4.3 1.2	-7.226 4.3 .9		11.024 4.2 .0	12.323 4.2 .3	14.722 4.1 .8	15.921 4.0 1.0		18.9 19 3.8 1.4	19.7 18 3.7 1.4	21.1 -17 3.6 1.4	21.617 3.6 1.4	22.017 3.5 1.4	22.716 3.4 1.3	22.916 3.4 1.3	23.0 - 16 3.4 1.3 23.0 - 16 3.4 1.3
RN FREQUENCY 75 MHZ	R DIPOLE ANTENNA Incident Mave	FOLAR. = VERTICAL THETA = 80.8 DEG	GAL POLARIZED ANTENNA	V PHASE MEAN MAXTMUM MAXIMUM TER CENTER ERROR CAINVAR PHASEVAR DEG DB DB DG	9 2.531 4.1 2.0	9 2.4 -31 4.1 2.1	9 2.2 -31 4.1 2.1		1 1.330 4.2 2.1	1 .830 4.2 2.1		<u>1</u> -1.129 4.2 2.0	5 -2.828 4.3 1.8		7	. <u>8</u> -7.226 4.3 .9		.9 -11.024 4.2 .0		.8 -14.722 4.1 .8	.8 -15.921 4.0 1.0	7 -18.020 3.9 1.3	.6 -18.919 3.8 1.4			.4 -21.617 3.6 1.4	.3 -22.0 -17 3.5 1.4	2 -22.7 -16 3.4 1.3	.2 -22.916 3.4 1.3	.2 -23.016 3.4 1.3 .2 -23.016 3.4 1.3
LTERN FREQUENCY 75MHZ	METER DIPOLE ANTENNA INT INCIDENT MAVE	H FOLAR. = VERTICAL H THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	CAIN PHASE MEAN MAXIMUM MAXIMUM Center Center Error Cainvar Phasevar DB DEG DB DB DEG	9 2.531 4.1 2.0	9 2.431 4.1 2.1	9 2.231 4.1 2.1		-1.1 1.330 4.2 2.1	-1.1 .830 4.2 2.1	-1.3 -1 -20 4.2 2.1	-1.4 -1.129 4.2 2.0	-1.5 -2.828 4.3 1.8	-1.6 -3.828 4.3 1.6	-1.7 -6.026 4.3 1.2	-1.8 -7.226 4.3 .9		-1.9 -11.024 4.2 .0	-1.9 -12.323 4.2 .3	-1.8 -14.722 4.1 .8	-1.8 -15.921 4.0 1.0		-1.6 -18.9 19 3.8 1.4	-1.5 -19.718 3.7 1.4		-1.4 -21.617 3.6 1.4	-1.3 -22.017 3.5 1.4	-1.2 -22.7 -16 3.4 1.3	-1.2 -22.916 3.4 1.3	-1.2 -23.0 - 16 3.4 1.3 -1.2 -23.0 - 16 3.4 1.3
PATTERN FREQUENCY 75MHZ	A 0,1 METER DIPOLE ANTENNA .D POINT INCIDENT WAVE	= 10 M FOLAR. = VERTICAL =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	ALM CAIN PHASE MEAN MAXIMUM MAXIMUM EVAR CENTER CENTER CENTER CENTER CALINVAR PHASEVAR DB DE DEC DB DB DEC	9 2.531 4.1 2.0	9 2.431 4.1 2.1	9 2.231 4.1 2.1		-1.1 1.330 4.2 2.1	-1.1 .830 4.2 2.1	-1.3 -4 - 29 4.2 2.1	-1.1 -1.129 4.2 2.0	-1.5 -2.828 4.3 1.8	-1.6 -3.828 4.3 1.6	-1.7 -6.026 4.3 1.2	-1.8 -7.226 4.3 .9		-1.9 -11.024 4.2 .0	-1.9 -12.323 4.2 .3	-1.8 -14.722 4.1 .8	-1.8 -15.921 4.0 1.0		-1.6 -18.9 19 3.8 1.4			-1.4 -21.617 3.6 1.4	-1.3 -22.0 -17 3.5 1.4	-1.2 -22.7 -16 3.4 1.3	-1.2 -22.9 - 16 3.4 1.3	
ON PATTERN FREQUENCY 75MHZ	LONG A 0.1 METER DIPOLE ANTENNA Field foint incident wave	HE = 10 M FOLAR. = VERTICAL HB =1.00 M THETA = 80.8 DEG	RADIAL FOLARIZED ANTENNA	MAXTNUM CAIN PHASE MEAN MAXIMUM MAXIMUM PHASEVAR CENTER CENTER EARCH CAINVAR PHASEVAR DEG DB DB DEG DB DB DEG	1.79 2.531 4.1 2.0		1.6 -9 2.2 -31 4.1 2.1			1.3 -1.1 .830 4.2 2.1		1.0 -1.4 -1.129 4.2 2.0	.7 [ -1.5 -2.828 4.3 1.8	.5 -1.6 -3.828 4.3 1.6	.1 -1.7 -6.026 4.3 1.2	. 4 -1.8 -7.226 4.3 .9		2.3 -1.9 -11.024 4.2 .0	<b>3.6</b> -1.9 -12.323 4.2 .3	12.0   -1.8 -14.722 4.1 .8	18.0 -1.8 -15.921 4.0 1.0	A 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	4.9 -1.6 -18.9 -19 3.8 1.4	4.0 -1.5 -19.718 3.7 1.4	3.0 -1.4 -21.1 -17 3.6 1.4	2.7 -1.4 -21.6 -17 3.6 1.4	2.5 -1.3 -22.0 -17 3.5 1.4	2.2	2.1 -1.2 -22.9 -16 3.4 1.3	2.1 -1.2 -23.0 -16 3.4 1.3 2.0 -1.2 -23.0 -16 3.4 1.3
ITION PATTERN FREQUENCY 75MHZ	ELD ALONG A 0,1 METER DIPOLE ANTENNA FIELD POINT INCIDENT WAVE	80 H MT = .10 H FOLAR. = VERTICAL 25 H HB =1.00 H THETA = 80.8 DEG	ENNA RADIAL POLARIZED ANTENNA	XININ MAXINIM GATN FHASE MEAN MAXINIM MAXINIM Inwar Fhasevar Center Center Brog Gainvar Phasevar deg db deg db deg	-5 1.79 2.531 4.1 2.0	.5 1.69 2.431 4.1 2.1	-9 2.2 -31 4.1 2.1		.5 1.4 -1.1 1.330 4.2 2.1	-5 1.3 -1.1 .830 4.2 2.1		. 1.0 -1.1 -1.1 -29 4.2 2.0	.4 .8 -1.5 -2.828 4.3 1.8	· · · · 5   -1.6 -3.828 4.3 1.6		.3 .4 -1.8 -7.226 4.3 .9		.0 2.3 -1.9 -11.024 4.2 .0	.3 3.6 -1.9 -12.3 -23 4.2 .3	3.8 12.0   -1.8 -14.722 4.1 .8	7 18.0 -1.8 -15.921 4.0 1.0		.9 4.9 1 -1.6 -18.9 19 3.8 1.4	.9 4.0 -1.5 -19.718 3.7 1.4		.8 2.7 -1.4 -21.6 -17 3.6 1.4	.8 2.5 -1.3 -22.0 -17 3.5 1.4		.7 2.1 -1.2 -22.9 -16 3.4 1.3	.7 2.1 -1.2 -23.016 3.4 1.3 .7 2.0 -1.2 -23.016 3.4 1.3
DIATION PATTERN FREQUENCY 75 HHZ	HE FIELD ALONG A 0.1 HETER DIFOLE ANTENNA VIDER FIELD POINT INCIDENT MANE	= 1.80 M DT = .10 M FOLAR. = VERTICAL = .25 M HB =1.00 M THETA = 80.8 DEG	D ANTENNA RADIAL POLARIZED ANTENNA	N MAXTNUM MAXTNUM GAIN PHASE MEAN MAXIMUM MAXTNUM MAXTNUM Gr Gainwar Phasewar center center error gainwar Phasewar db deg db deg db deg db db db	88 -5 1.79 2.531 4.1 2.0	02 .5 1.69 2.431 4.1 2.1	02 .5 1.69 2.231 4.1 2.1		02 .5 1.4 -1.1 1.3 -30 4.2 2.1	03 .5 1.3 -1.1 .830 4.2 2.1	03 . 2 1. 2 - 1. 2 - 3 - 30 4. 2 2. 1 - 03 . 4 . 2 2. 1 - 1. 3 - 1. 4 - 29 4. 2 2. 1	03 .4 1.0 -1.4 -1.129 4.2 2.0	04 .4 .8 -1.5 -2.8 -2.8 4.3 1.8	CH	05 .4 .1 -1.7 -5.026 4.3 1.2	06 3 .4 -1.8 -7.226 4.3 .9		11 .0 2.3 -1.9 -11.024 4.2 .0	15 .3 3.6 -1.9 -12.323 4.2 .3	06 3.8 12.0   -1.8 -14.722 4.1 .8	46 .7 18.0 -1.8 -15.921 4.0 1.0	12 1.0 9.5	11 .9 4.9 -1.6 -18.919 3.8 1.4	00 .9 4.0 -1.5 -19.718 3.7 1.4		06 .8 2.7 -1.4 -21.6 -17 3.6 1.4	05 .8 2.5 -1.3 -22.0 -17 3.5 1.4	05 .8 2.3   -1.3 -22.4 10 3.5 1.3 05 .8 2.2   -1.2 -22.7 16 3.4 1.3	04 .7 2.1 -1.2 -22.9 -16 3.4 1.3	04 7 2.1 -1.2 -23.0 -16 3.4 1.3 04 7 2.0 -1.2 -23.0 -16 3.4 1.3
RADIATION PATTERN FREQUENCY 75MHZ	OF THE FIELD ALONG A 0,1 METER DIPOLE ANTENNA .cv1.inder field point incident vave	RMTH = 1,80 M INT = 1,0 M FOLAR. = VERTICAL R = .25 M HB =1.00 M THETA = 80.8 DEG	ARIZED ANTENNA RADIAL POLARIZED ANTENNA	HEAN MAXTMUM MAXTMUM GATN PHASE MEAN MAXTMUM MAXTMUM R ERROR GATWAR PHASEVAR CENTER EAROR GATWAR PHASEVAR DB DB DB DE DEG DB DEG DB DB DEG			.02 .5 1.69 2.231 4.1 2.1	.02 .5 1.69 2.030 4.1 2.1		.03 .5 1.3 -1.1 .830 4.2 2.1	.03 .5 1.2 -1.2 .5 -5.0 4.2 2.1 . .03 .4 1.1 -1.3 -429 4.2 2.1	. 03 . 4 1.0 -1.4 -1.1 29 4.2 2.0		.04 .4 .5 -1.6 -3.828 4.3 1.6	05 .4 .1 -1.7 -5.026 4.3 1.2	06 .3 .4 -1.8 -7.226 4.3 .9		.11 .0 2.3 -1.9 -11.024 4.2 .0		06 3.8 12.0 1 -1.8 -14.722 4.1 .8	46 .7 18.0 -1.8 -15.921 4.0 1.0					06 .8 2.7 -1.4 -21.617 3.6 1.4	305 .8 2.5 -1.3 -22.017 3.5 1.4	105 .8 2.3   -1.3 -22.410 3.5 1.3 505 .8 2.2   -1.2 -22.716 3.4 1.3	204 .7 2.1 -1.2 -22.916 3.4 1.3	004 .7 2.1   -1.2 -23.016 3.4 1.3 904 .7 2.0   -1.2 -23.016 3.4 1.3
4AL RADIATION PATTERN FREQUENCY 75MHZ	CHECK OF THE FIELD ALONG A 0.1 HETER DIPOLE ANTENNA 1.SYM.CYLINDER FIELD POINT INCIDENT WAVE	TAL LENGTH = 1.00 M MT = .10 M FOLAR. = VENTICAL IAL LENGTH = .25 M HB =1.00 M THETA = 80.8 DEG	L POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	HASE HEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM MAXIMUM Center Error Gainwar Phasevar Center Center Error Gainwar Phasevar dec db db deg deg db deg db	157.3 .02 .5 1.79 2.531 4.1 2.0	15/.5 .02 .5 1.69 2.431 4.1 2.1	157.7 .02 .5 1.69 2.231 4.1 2.1	158.1 .02 .5 1.6 -9 2.0 -30 4.1 2.1	159.2 .02 .5 1.4 -1.1 1.330 4.2 2.1	159.9 .03 .5 1.3 -1.1 .830 4.2 2.1	161.6 .03 .4 1.1 -1.3 -1.4 -2.9 4.2 2.1	162.6 .03 .4 1.0 -1.4 -1.129 4.2 2.0	163.7 .03 .4 .8 -1.4 -1.929 4.3 1.9 164.8 .04 .4 .7 -1.5 -2.828 4.3 1.8	166.1 .04 .4 .5 -1.6 -3.828 4.3 1.6		170.8 .06 .3 .4 -1.8 -7.226 4.3 .9	172.8 .07 .3 .9 -1.8 -0.525 4.5 .0 175.3 no .2 1.5 1 -1.8 -0.724 4.3 .3	178.5 .11 .0 2.3 -1.9 -11.024 4.2 .0	-176.5 .15 .3 3.6 -1.9 -12.323 4.2 .3	122.3 .06 3.8 12.0 1 -1.8 -14.7 -22 4.1 .8	-40.6 - 46 .7 18.0 -1.8 -15.921 4.0 1.0		-11.4 -11 .9 4.9 -1.6 -18.9 -19 3.8 1.4	-9.009 .9 4.0 -1.5 -19.718 3.7 1.4		-4.706 .8 2.7 -1.4 -21.617 3.6 1.4	-3.805 .8 2.5 -1.3 -22.017 3.5 1.4	-3.105 .8 2.3   -1.2 -22.410 3.5 1.3	-2.204 .7 2.1 -1.2 -22.916 3.4 1.3	-2.004 .7 2.1 -1.2 -23.016 3.4 1.3 -1.904 .7 2.0 -1.2 -23.016 3.4 1.3
UT HAL RADIATION PATTERN FREQUENCY 75 MHZ	EITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA 1. ROT.SM.CALINDER FIELD POINT INCIDENT WAVE	XXAL LENGTH = 1,80 M INT = 1,0 M FOLAR. = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	TICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	TIN FHASE MEAN MAXTMUM MAXTMUM GATN FHASE MEAN MAXTMUM MAXTMUM MAXTMUM TER CENTER ERROR GATWAR FHASEVAR CENTER CENTER ERROR GATWAR FHASEVAR CENTER CENTER ERROR GATWAR FHASEVAR CENTER CENTER OF DB DB DEC DB DB DB DEC DB DB DB DEC DB DB DEC DB DB DB DEC DB DB DB DEC DB DB DEC DB DB DB DB DB DEC DB DB DB DB DB DB DB DB DEC DB	3.0 157.3 .02 .5 1.79 2.531 4.1 2.0	0.1 157.5 .02 .5 1.69 2.431 4.1 2.1	0.2 157.7 .02 .5 1.69 2.231 4.1 2.1	0.4 158.1 .02 .5 1.69 2.030 4.1 2.1	0.9 159.2 .02 .5 1.4 -1.1 1.3 -30 4.2 2.1	1.2 159.9 .03 .5 1.3 -1.1 .830 4.2 2.1	2,1 161,6 .03 .3 1.2 -1.3 -1.429 4.2 2.1	2.6 162.6 .03 .4 1.0 -1.4 -1.129 4.2 2.0	3.2 103.7 .03 .4 .6 -1.4 -1.929 4.3 1.9 4.0 164.8 .04 .4 .7 -1.5 -2.828 4.3 1.8	4.8 166.1 .04 .4 .5 -1.6 -3.828 4.3 1.6	5.8 169.1 .05 .4 .4 .1 -1.7 -6.026 4.3 1.2	8.1 170.8 .06 .3 .4 -1.8 -7.226 4.3 .9	9.6 172.8 .07 .3 .9 -1.8 -6.7 -2.5 4.3 .0 1 1 175 2 10 2 1.5 -1.8 -0.7 -24 4.3 .3	3.8 178.5 .11 .0 2.3 -1.9 -11.024 4.2 .0	7.0 -176.5 .15 .3 3.6   -1.9 -12.323 4.2 .3	0.3 -122.3 .06 3.8 12.0 1 -1.8 -14.7 -22 4.1 .8	5.3 -40.6 - 46 .7 18.0 -1.8 -15.921 4.0 1.0	9.5	3.6 -11.411 .9 4.9 -1.6 -18.919 3.8 1.4	1.8 -9.009 .9 4.0 -1.5 -19.718 3.7 1.4		8.6 -4.706 .8 2.7 -1.4 -21.617 3.6 1.4	7.9 -3.805 .8 2.5 -1.3 -22.017 3.5 1.4	7.0 -2.605 .8 2.2   -1.2 -22.710 3.9 1.3	6.8 -2.204 .7 2.1 -1.2 -22.916 3.4 1.3	6.6 -2.004 .7 2.1   -1.2 -23.016 3.4 1.3 6.6 -1.904 .7 2.0   -1.2 -23.016 3.4 1.3
LIMUTHAL RADIATION PATTERN FREQUENCY 75MHZ	AGGENEITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA ;1900Y: ROT.SYM.CYLINDER FIELD POINT INCIDENT WAVE	DIAMETER = 1.25 H HB =1.00 H FULAR. = VERTICAL DIAMETER = .25 H HB =1.00 H THETA = 80.8 DEG	VERTICAL FOLARIZED ANTENNA RADIAL FOLARIZED ANTENNA	I GAIN PHASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM CEMTER CENTER ERROR GAINWAR PHASEVAR 3 DB DEC DB DB DEC DB DEC DB DB DEC	0-10.0 157.3 .02 .5 1.79 2.531 4.1 2.0	0 -10.1 157.5 .02 .5 1.69 2.431 4.1 2.1	5 -10.2 157.7 .02 .5 1.69 2.231 4.1 2.1	0 -10.4 158.1 .02 .5 1.69 2.030 4.1 2.1	0 -10.0 159.2 .02 .5 1.4 -1.1 1.3 -30 4.2 2.1	5 -11.2 159.9 .03 .5 1.3 -1.1 .830 4.2 2.1	51_12, 161.6 .03 .4 1.1 1.3 -1.429 4.2 2.1	0 -12.6 162.6 .03 .4 1.0 -1.4 -1.129 4.2 2.0	01 -14.0 164.8 .04 .4 .7 1 -1.5 -2.828 4.3 1.8	5 -14.8 166.1 . OH .4 .5 -1.6 -3.828 4.3 1.6	5 -16.8 169.1 .05 .4 .1 -1.7 -6.026 4.3 1.2	0 -18.1 170.8 .06 .3 .4 -1.8 -7.226 4.3 .9	5 -19.6 172.8 .07 .3 .9 -1.8 -8.529 4.3 .0	5 -23.8 178.5 .11 .0 2.3 -1.9 -11.024 4.2 .0	0 -27.0 -176.5 .15 .3 3.6 -1.9 -12.3 -23 4.2 .3	01 -30.7 -102.0 .23 1.2 2.3 1 -1.0 -1.5 -1.5 7.5 4.1 .8	5 -35.3 -40.6 - 46 .7 18.0 -1.8 -15.921 4.0 1.0	01 – 29. 2 – 21. 1 – 1 – 1 – 1 – 1 – 1 – 1 – 1 – 1 – 1	0 -23.6 -11.411 .9 4.9 -1.6 -18.919 3.8 1.4	5 -21.8 -9.009 .9 4.0 -1.5 -19.718 3.7 1.4		0 -18.6 -4.706 .8 2.7 -1.4 -21.617 3.6 1.4	5 -17.9 -3.805 .8 2.5 -1.3 -22.017 3.5 1.4	00 -17.4 -3.105 .8 2.3   -1.5 -22.410 3.5 1.3 5.5 1.5 5.5 1.5 5.5 1.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 1.5 5.5 5	0 -16.8 -2.204 .7 2.1 -1.2 -22.916 3.4 1.3	5 -16.6 -2.004 .7 2.1 -1.2 -23.016 3.4 1.3 00 -16.6 -1.904 .7 2.0 -1.2 -23.016 3.4 1.3

N												-		_							1_				
ΗΨ			SEVAR	*	n (~ =	, <del>.</del> .	<u>-</u> و	٥ð	<b>N N</b>	1		- 0	- -	τın	⇒. u	ΩŅ	0,0	٥. V	m,	<u>.</u>	F		. 1 	<u>5.</u> 20	MHz
100	-1		M MAX IR PHA DEC		iġ;	20	റ്റ്	φ.r.		ιń Η	म (	า๋ณ์	÷	• •	·	i m	=i =	ţ	ý	0	ω	χoσ	6	o 0	5
N C X	E LZONTA B DEC	NNA	AXIMU	10°3	100	~~~ ~~~	2.2 2.2	200	200	~	~	N N	~ ~		~ ~	ŝ	~		~	2	N.	NN	2	~ ~	
9 U E	ENNA T WAVE = HOR) = 80.8	D ANTE	EAN P RROR 0	# [] # []	:::: / /	==	== 	22 7 7	14	:=:	1	2 2 7 7	11	== /	: ;	:₽ //	₽: 1	:⊧ ¦'i	₽! `	≥⊈ ii	° 	28	° 	₽ ₽ • •	292
ы ж	E ANT ICIDEN ILAR. HETA	ARIZE	LSE M TER E	* * ~ u	າ ດ ເ ດໍ່ ຕໍ່-	0 00	1.1	25	6.7	ດແ		- 6.0			~ ~ ~			0 0 0	8.0	4.6 9.4	1.1	2.2	2	∞ ≓'u	
NN		AL POI	ER CEI DEC	* 101	- 0.0	ი იი	 ചെ വ	ល័ល លុង				-10 -10	⊷r ∞c	~~ ~~	80 v	0 80 81	~ ·	~~~	9	~ 0	5	~ U	-	œ، م،	101
ш.	METER NT M	RADI	CA IN CENT DB	∎ 18: 19:	į r	i₫	ኯ፟ኯ፞	71	· · -		~	ง่ณ่	ດໍ ເ	ini	N, C	ŝ	N,		•	11	7	ท่า	Ý	φţ	
P A T	A 0.1 LD POI =1.00		MUM EVAR										_		~	•		~ ~		• •		~ ^	~	~ ~	0
NO	FIEI FIEI HB	A	HAXI PHAS DEC	0.00	- 00 ( n # 0	2.7	6.9 8.0	12.5	1.61	21.6	3.5	115.4	168.4	166.2	601		5.5	17.7	15.4	2 2 2	10	80 1	5	= 0	1-0
ATI	ЕLD А 25 М	ANTENN	AXIMUM A INVAR 3	o,#,!	:::	<u>.</u>	2.2	m0 m4		0.7	1.4	17.3	6.7	• • •	18.1	-1 -1	8.9		6.4	u 		a -	5.	- '	. 7. 0
IQV	THE F	IZED	CAN M. ROR G	ຮອ	ទំនុ	88	88	88	888	58	3	52	88	56	89	25. 25.	ទុ	<u>5</u> 5	5	88	38	88	8	ຮຸເ	ទំនំន
۲ د	TENGT	POLAR	SE ME TER ER DB	<b>~</b> @,	<u></u> .		~	<i>=</i> , c		) <b>-</b> 4		٥٩	ιĝ.	977 977	= 0	2	ō,	• •	80.	o, v		ño	5	~ •	ງ ທີ ⇒
T H A	Y CHEC ROT.SI AXIAL DIAMET	ONTAL	R CENT	59	121	85	\$G		388.2 	88	212	88	115		121		101	77	. ନ ମ	894 94	<u>ନ୍</u>	ត្តក្ត	-1-		
D H	ENEIT SODY:	HORIZ	CA IN CENTE DB	44.	ពំកំ	र्ग न	74	Ϋ́Υ		5	-15		ដុ	977 977			9	Ϋ́Υ	Ŷ	φ'n	f	ร่ ๆ	÷۳	Ϋ́	ሳሳሳ
IZV	TESTE		PHI	000	558	៩ ស	R	킁뷰	ទន	84	22	68	88	ያጽ	85	<u>8</u> 2	115	<u>8</u> <u>8</u>	9	មិគឺ	ŝ.	ខ្មរ	3	165	568
1																									
2 H			VAR								<u> </u>														
Z H W 001			MAXTMUM PHASEVAR Deg	8.1.	- 0-		مىن	~	<u>.</u>	0.1			1.2		ŵ	- v,		1.1	9.1	1.8	1.9		1.9	1.8	
C Y 100 M H Z	CAL	NA	XIMUM MAXIMUM INVAR PHASEVAR DEG	3.2 1.8	3.5	3.3	0.5 0.5			3.7 1.0	4.0 1.4	4.1 1.4 4.2 1.4	1.2	4.5 1.0 4.4 .7			4.5	4.1 1.2 1.1	4.4 1.6	4.3 1.8	1.9	4.2 1.9	1.1	4.1 1.8 1.1 1.8	
U E N C Y 100 M H Z	INA WAVE VERTICAL 80.8 DEG	ANTENNA	AN MAXIMUM MAXIMUM Rok gainvar phasevar db deg	15 3.2 1.8 15 3.2 1.8	15 3.2	16 3.3 1.1 -	16 3.3 16 3.4 2 4 5 2 4	17 3.4 .2	18 3.6	19 3.7 1.0	21 4.0 1.4	.22 4.1 1.4 .23 4.2 1.4	24 4.3 1.2	2	.27 4.4 .3	28 4.5 .5	-29 4.5 .8		.31 4.4 1.6	-32 4.3 1.8 32 4.3 1.8	.33 4.2 1.9	.33 4.2 1.9	33 4.1 1.9	.33 4.1 1.8	-33 4.1 1.8 -33 4.0 1.8
REQUENCY 100 MHZ	ANTENNA Ident Wave Ar. = Vertical Ta = 80.8 dec	RIZED ANTENNA	E MEAN MAXIMUM MAXIMUM Er Error Gainvar Phasevar DB DB DB DEG	1 - 15 3.2 1.8 - 15 3.2 1.8 - 15 3.2 1.8		716 3.3 1.1	5 - 16 3.3 - 16 3.4 - 5 3.4 - 5	9 17 3.4 .2 - 17 3.5 1		5 - 19 3.7 1.0	5 21 4.0 1.4	322 4.1 1.4 923 4.2 1.4	424 4.3 1.2		327 4.4 .3	2 28 4.5 .5	9 29 4.5 .8	1	531 4.4 1.6	8 - 32 4.3 1.8 0 - 33 4.3 1.8	8 - 33 4.2 1.9	733 4.2 1.9	9 33 4.1 1.9		8 - 33 4 1 - 50 - 33 4 1 - 1 - 8 - 33 4 - 1 - 1 - 8
N F R E Q U E N C Y 100 M H Z	DIFOLE ANTENNA INCIDENT MAVE POLAR. = VENTICAL THETA = 80.8 DEG	POLARIZED ANTENNA	HASE MEAN MAXIMUM MAXIMUM R CENTER ERROR GAINVAR PHASEVAR DEG DB DB DB	18.1 - 15 3.2 1.8 18.1 - 15 3.2 1.8	18.0 - 15 3.2 1.6	17.716 3.3 1.1	17.5 - 16 3.3 .9 17.3 - 16 3.4 .5	16.9 - 17 3.4 .2 16.5 - 17 3.4 .2		14.5 19 3.7 1.0	12.521 4.0 1.4	11.322 4.1 1.4 9.923 4.2 1.4	8.424 4.3 1.2	5.126 4.4 .7	3.327 4.4 .3	228 4.5 .5	-1.929 4.5 .8	-5.030 4.5 1.2	-6.531 4.4 1.6	-7.832 4.3 1.8	-9.8 -33 4.2 1.9		-11.933 4.1 1.9	-12.333 4.1 1.8	-12.8 - 33 4.1 1.8 -12.9 - 33 4.1 1.8
TERN FREQUENCY 100 MHZ	ETER DIPOLE ANTENNA T INCIDENT WAVE M POLAR. = VERTICAL M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	GATN FHASE MEAN MAXIMUM MAXIMUM CENTER CENTER ERROR GAINVAR FHASEVAR DB DEG DB DB DEG	-3.1 18.115 3.2 1.8 -3.1 18.115 3.2 1.8 -3.1 18.115 3.2 1.8		-3.3 17.9 -15 3.2 1.4 1	-3.4 17.5 - 16 3.3 .9 -3.5 17.3 - 16 3.4 .5	-3.7 16.917 3.4 .2		L.2 14.5 - 19 3.7 1.0	4.5 12.521 4.0 1.4	-1.7 11.322 1.1 1.1 -1.8 9.923 1.2 1.1	-4.9 8.424 4.3 1.2	-5.1 5.126 4.4 .7	-5.1 3.327 4.4 .3	-5.1228 4.5 .5	-5.1 -1.929 4.5 .8	-5.0 -5.130 4.5 1.2	-n.9 -6.531 u.u 1.6		Late -9.8 33 4.2 1.9				
ATTERN FREQUENCY 100 MHZ	0.1 METER DIFOLE ANTENNA D POINT INCIDENT WAVE 10 M FOLAR. = VENTICAL - 1.00 M THETA = 80.8 DEC	RADIAL POLARIZED ANTENNA	M GAIN HASE HEAN MAXIMUM MAXIMUM AR CENTER ERROR GAINVAR PHASEVAR DB DB DEG DB DB DEG	-3.1 18.115 3.2 1.8 -3.1 18.115 3.2 1.8 -3.1 18.115 3.2 1.8		-3.4 17.716 3.3 1.1	-3.4 17.5 - 16 3.3 .9 -3.5 17.3 - 16 3.4 .5	-3.7 16.917 3.4 .2		1.2 14.5 - 19 3.7 1.0		L. 7 11.3 22 4.1 1.4 L. 8 9.9 23 4.2 1.4		-5.1 5.126 4.4 .7	-5.1 3.327 4.4 .3	-5.1 -2 -28 4.5 .5	-5.1 -1.929 4.5 .8		-4.9 -6.531 4.4 1.6						
ON PATTERN FREQUENCY 100 MHZ	ONG A 0.1 METER DIFOLE ANTENNA FIELD POINT INCIDENT WAVE DAT = .10 M POLAR. = VENTICAL HB =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	MAXTMUM GAIN HASE MEAN MAXTMUM MAXIMUM Phasevar center error gainvar phasevar deg db deg db db	1.1 -3.1 18.115 3.2 1.8 1.1 -3.1 18.115 3.2 1.8		1.2	1.2 -3.4 17.516 3.3 .9 1.2 -3.5 17.316 3.4 .5	1.2 -3.7 16.917 3.4 .2 1.2 -3.8 16.517 3.4 .2		1.2 4.2 14.5 - 19 3.7 1.0	1.1 4.5 12.521 4.0 1.4	1.1			-1 -5.1 3.327 4.4 .3	.6 -5.1 -2 -28 4.5 .5	1.1 -5.1 -1.929 4.5 .8	2.2 -5.0 -5.1 -3.0 4.5 1.2	2.7 -4.9 -6.531 4.4 1.6	3.0 H.B7.8 32 4.3 1.8	3.2 4.6 -9.833 4.2 1.9	<b>3.0 1 -1.5</b> -10.733 4.2 1.9	2.1 1.4 -11.9 -33 4.1 1.9	1.7 <u>–</u> 1.3 –12.3 –.33 4.1 1.8	
ITION PATTERN FREQUENCY 100 HHZ	ELD ALONG A 0.1 METER DIPOLE ANTENNA FIELD POINT INCIDENT WAVE 30 M DAT = . 10 M FOLAR. = VERTICAL 55 M HB =1.00 M THETA = 80.8 DEG	SNNA RADIAL POLARIZED ANTENNA	CIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM MAXIMUM INVAR PHASEVAR CENTER CENTER ERROR GAINVAR PHASEVAR DE	.3 1.1 -3.1 18.115 3.2 1.8 .3 1.1 -3.1 18.115 3.2 1.8 .18.115 3.2 1.8		3 1.2 -3.4 17.716 3.3 1.1	.3 1.2 -3.4 17.5 - 16 3.3 .9 .3 1.2 -3.5 17.3 - 16 3.4 .5	.3 1.2 -3.7 16.9 -17 3.4 .2 -3 1.2 -3.8 16.5 -17 3.4 .2		2 1.2 4.2 14.5 - 19 3.7 1.0	2 1.1 4.5 12.521 4.0 1.4	.2 1.1 -4.7 11.322 4.1 1.4 .2 1.0 -4.8 9.923 4.2 1.4	2 .8 .4.9 8.4	· · · · · · · · · · · · · · · · · · ·	·1 ·1 -5.1 3.327 4.4 ·3	.1 .6 -5.1 -2 -28 4.5 .5	2 1.1 -5.1 -1.9 -29 4.5 .8		.4 2.7 -4.9 -6.531 4.4 1.6	.5 3.0 -4.8 -7.832 4.3 1.8 7 3.2 -4.7 -8.0 - 32 4.3 1.8	.8 3.2 Jul 6 -9.833 4.2 1.9	.9 3.0   -1.5 -10.733 4.2 1.9	1,1 2,1 1,4,4,1,9 -33 4,1 1,9	1.2 1.7 - H.3 -12.333 4.1 1.8	1.2 1.2 1.2 1.2 1.2 1.2 1.2 1.2 1.2 1.2
DIATION PATTERN FREQUENCY 100 MHZ	HE FIELD ALONG A 0.1 METER DIFOLE ANTENNA NDER FIELD POINT INCIDENT WAVE = 1.80 M DAT = .10 M POLAR. = VENTICAL = .25 M HB =1.00 M THETA = 80.8 DEG	D ANTENNA RADIAL FOLARIZED ANTENNA	N MAXIMUM MAXIMUM GATN FHASE MEAN MAXIMUM MAXIMUM AXIMUM CR GAINVAR FHASEVAR CENTER CENTER EAROR GAINVAR FHASEVAR DB DB DEC DB DB DEC	00 .3 1.1 -3.1 18.1 -15 3.2 1.8 00 .3 1.1 -3.1 18.1 -15 3.2 1.8 1.1 -3.1 18.1 -15 3.2 1.8		00 .3 1.2   -3.4 17.716 3.3 1.1	00 .3 1.2 -3.4 17.5 - 16 3.3 .9 00 .3 1.2 -3.5 17.3 - 16 3.4 .5	00 .3 1.2 -3.7 16.9 -17 3.4 .2 00 .3 1.2 -3.8 16.5 -17 3.4 .2			01 .2 1.1 .4.5 12.521 4.0 1.4	01 .2 1.1 .4.7 11.322 4.1 1.4 . 01 .2 1.0 .4.8 9.923 4.2 1.4 .	01 .2 .8 .4.9 8.424 4.3 1.2	01 .1 .4 -5.1 5.1 -26 4.4 .7	01 .1 .1 -5.1 3.327 4.4 .3	01 .1 .6 -5.1228 4.5 .5	01 .2 1.1 -5.1 -1.929 4.5 .8	01 .2 1.0 1 -5.0 -5.1 -5.0 1.2 1	00 .4 2.7 -4.9 -6.531 4.4 1.6	00 .5 3.0 -4.8 -7.832 4.3 1.8 -	01 .8 3.2 4.6 -9.8 -33 4.2 1.9	01 .9 3.0 J -J.5 -10.733 4.2 1.9 J	02 1.1 2.1 J -1.4 -11.9 -33 4.1 1.9	02 1.2 1.7   -H.3 -12.3 -133 4.1 1.8	02 1.2 1.4 1.5 1.5 1.5 1.5 1.5 1.5 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6
. RADIATION PATTERN FREQUENCY 100 MHZ	( OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA 1. CYLINDER FIELD POINT INCIDENT WAVE ENCTH = 1.80 M DAT = .10 M POLAR. = VENTICAL 3. = .25 M HB =1.00 M THETA = 80.8 DEG	ARIZED ANTENNA RADIAL POLARIZED ANTENNA	E MEAN MAXTMUM MAXTMUM GATN HASE MEAN MAXTMUM MAXTMUM MAXTMUM CA CATTER CATINVAR FHASEVAR Errer Gatinvar Fhasevar DB DB DE DEG DB DB DEG	3 .00 .3 1.1 -3.1 18.115 3.2 1.8 2 .00 .3 1.1 -3.1 18.115 3.2 1.8		7 .00 .3 1.2 1 -3.4 17.7 -16 3.3 1.1 1	8 .00 .3 1.2 -3.4 17.5 - 16 3.3 .9 0 .00 .3 1.2 -3.5 17.3 - 16 3.4 .5	4 .00 .3 1.2 -3.7 16.9 -17 3.4 .2			2 .01 .2 1.1 4.5 12.521 4.0 1.4	5 .01 .2 1.1 -4.7 11.322 4.1 1.4 5 5 .01 .2 1.0 -4.8 9.923 4.2 1.4	4 .01 .2 .8 -4.9 8.424 4.3 1.2	2 .01 .1 .4 -5.1 5.126 4.4 .7	2 .01 .1 .1 -5.1 3.327 4.4 .3	9 .01 .1 .6   -5.1228 4.5 .5	6 .01 .2 1.1 -5.1 -1.929 4.5 .8	7 .01 .2 1.0 1 -5.0 -5.1 -5.0 -5.1 2.1 1.2 1 7 .01 .3 2.2 1 -5.0 -5.131 4.4 1.4 1	3 .00 .4 2.7 .4.9 -6.531 4.4 1.6			$\begin{array}{cccccccccccccccccccccccccccccccccccc$	502 1.1 2.1 ] -4.4 -11.933 4.1 1.9	302 1.2 1.7 ] -4.3 -12.333 4.1 1.8 ]	3         -100         1.2         -1.3         -1.0         1.0           5         -100         1.3         1.2         -1.2         -1.2         -1.0           5         -100         1.3         1.1         -1.0         -1.0         -1.0
HAL RADIATION PATTERN FREQUENCY 100 MHZ	CHECK OF THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA OT.SYM.CYLLINER FIELD FOINT INCIDENT WAVE XIAL LENGTH = 1.80 M INT = .10 M FULAR. = VENTICAL IAMETER = .25 M HB =1.00 M THETA = 80.8 DEC	AL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	PHASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM CENTER ERROR GAINVAR PHASEVAR CENTER CENTER ERROR GAINVAR PHASEVAR DEG DB DB DEG DB DB DEG	167.3 .00 .3 1.1 -3.1 18.115 3.2 1.8 167.4 .00 .3 1.1 -3.1 18.115 3.2 1.8		108.9 .00 .3 1.2 1 -3.3 17.9 15 3.2 1.4 1 169.7 .00 .3 1.2 1 -3.4 17.7 16 3.3 1.1 1	170.8 .00 .3 1.2 -3.4 17.5 -16 3.3 .9 172.0 .00 .3 1.2 -3.5 17.3 -16 3.4 .5	173.4 .00 .3 1.2 -3.7 16.917 3.4 .2	176.8	-179.1 .01 .2 1.2 -1.2 14.519 3.7 1.0	-174.2 .01 .2 1.1 .4.5 12.521 4.0 1.4	-171.5 .01 .2 1.1 -4.7 11.3 -22 4.1 1.4 - -168.6 .01 .2 1.0 -4.8 9.9 -23 4.2 1.4	-165.4 .01 .2 .8 -4.9 8.424 4.3 1.2	-102.0 .01 .1 .1 -1 -5.1 5.1 -26 4.4 .7 -	-154.2 .01 .1 .1 -1 -5.1 3.327 4.4 .3	-149.6 .01 .1 .2 $-5.1$ -5.2 $-28$ $4.5$ .1 $-144.9$ .01 .1 .6 $-5.1$ 2 $-28$ $4.5$ .5	-139.6 .01 .2 1.1 -5.1 -1.929 4.5 .8	-133.7 .01 .2 1.0 1 -5.1 -5.030 4.5 1.2 1	-120.3 .00 .4 2.7 -4.9 -6.531 4.4 1.6	-113.000 .5 3.0 -4.8 -7.832 4.3 1.8 -	-98.001 .8 3.2   -11.6 -9.833 4.2 1.9	-91.001 .9 3.0 $-1.5$ -10.733 4.2 1.9 $-91.6$	-79.502 1.1 2.1 ] -1.4 -11.933 4.1 1.9	-75.302 1.2 1.7 ] -1.3 -12.333 1.1 1.8 ]	-12:302 1.2 1.4
MUTHAL RADIATION PATTERN FREQUENCY 100MHZ	INEITY CHECK OF THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA DXY: ROT, SWL.CYLINDER FIELD POINT INCIDENT WAVE AXTAL LENGTH = 1.80 M DAT = .10 M POLAR. = VENTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	GERTICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	AIN PHASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM SUTTER CENTER ERROR GAINVAR PHASEVAR Setter center error gainvar phasevar Setter de de de de deg deg deg de deg	-5-8 167.3 .00 .3 1.1 -3.1 18.115 3.2 1.8 -5-8 167.4 .00 .3 1.1 -3.1 18.115 3.2 1.8	-5.9 168.2 .00 .3 1.2 -3.2 18.0 -15 3.2 1.6	-6.1 169.7 .00 .3 1.2 1 -3.4 17.716 3.3 1.1 1	-6.3 170.8 .00 .3 1.2 -3.4 17.5 - 16 3.3 .9 -6.5 172.0 .00 .3 1.2 -3.5 17.3 - 16 3.4 .5	-6.8 173.4 .00 .3 1.2 $-3.7$ 16.9 $-17$ 3.4 .2 $-7.0$ 175.0 $-10$ 2.8 16.5 $-17$ 3.4 .2		-8.1 -179.1 .01 .2 1.2 4.2 14.5 -19 3.7 1.0	-8.9 -174.2 .01 .2 1.1 4.5 12.521 4.0 1.4	-9.4 -171.5 .01 .2 1.1 -4.7 11.322 4.1 1.4 - -10.0 -168.6 .01 .2 1.0 -4.8 9.923 4.2 1.4	-10.5 -165.4 .01 .2 .8 -4.9 8.424 4.3 1.2	-11.6 -162.0 .01 .1 .1 .1 -5.0 0.025 4.3 1.0 - -11.8 -158.2 .01 .1 .4 -5.1 5.126 4.4 .7	-12.6 -154.2 .01 .1 .1 -1 -5.1 3.327 4.4 .3	-15.3 -149.6 .01 .1 .2 $-5.1$ $-5.1$ $-2.8$ $4.5$ .5 $-14.1 -144.9$ .01 .1 .6 $1 -5.1$ $-2$ $-2.8$ $4.5$ .5	-14.9 -139.6 .01 .2 1.1 -5.1 -1.929 4.5 .8	-15.0 -133.7 .01 .2 1.0 1 -5.0 -5.1 -5.030 4.2 1.2 1 -16.4 -127.3 .01 .3 2.2 1 -5.0 -5.131 4.4 1.4 1.4	-17.1 -120.3 .00 .4 2.7 -4.9 -6.531 4.4 1.6	-17.7 -113.000 .5 3.0 -14.8 -7.832 4.3 1.8	-18.6 -98.001 .8 3.2 4.6 -9.833 4.2 1.9	-18.9 -91.001 .9 3.0 $-1.5 -10.733 4.2 1.9$		-19.1 -75.302 1.2 1.7 -4.3 -12.333 4.1 1.8	-19.0 -76.5 - 02 1.2 1.4

				_	_	_						_					_	_	_	_					_		
5 H .			MUM EVAR			-								~ ~										ZY	LJ	125	MHz
125 4			A MAXI R PHAS	****	1 4	5	m e	<u>i ci</u>	= 0	õ	o «				; <del>-</del>		-	N	ກໍສໍ	<u>ب</u> ري	~	ື່		2	2=	=	
NCY	ZONTAI DEG	NNA	IAXTMU AINVAI 8	***	, . , . , .	5.2			ਸ ਸ ਨਾ ਨ	5. N	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	1	v ~ v ~	2.0	, - , -, -,	۲. ۲.					5	~ ~		2	 	2.1	101 101 101
Q U E Enna	T WAVE = HORJ = 80.8	D ANTE	EAN M RROR O	* ***	2 <u>99</u>	<u>∞</u>	° °	2 <u>8</u> 2	≊≊ ii	° 	₽₽ 7 7	: <b>-</b> :	= <b>:</b> / /	₽₽ 1	= = / /	₽! -	==	=	2 2 7 7	24 1	2 ≌	°₽ ?	2 ∰ / /	≌: '	°≌ ''	œ: i	₽₽ <b>₽</b> ₽₽ <b>₽</b>
E ANTI	ICIDEN ILAR.	ARIZE	TER M	* *	v r	0		~~	0.0	1	۵ ۲-	~		N=		0	n e		25			# 0	<u> </u>	5	n.≂.	5	* **
R N F	484 -	AL POL	E E H	***		जन	5 - 0 -	- 9:			5 7 7 7 7		5 3 0 N		- @	6	5 8 5 8	~~	v v 2 20	80 K	 	۵. ۳.		6. 6.	+ ∾	9.9	
T E   METER	THE M	RADI	CAIN CENT	***		Ŷ	- - -	i m	ที่ร่	i	• •	· ·	- ~	~ ~	i ni	~	N N	~	Ň		: •	•	i f	ญ่.	İΫ	တံ့	v 80 1 11
P A 7 A 0.1	LD P0] = 1( = 1.00		MUM EVAR												_		•••	_									
I O N	FIE	¥,	A MAXI R PHAS DEG	0.0	,	6	8 0	2	14.4	18.9	21.5	82	19 19	115.02	155.1	165	2021		Å Å	80	19.6	1.1	<u>1</u>	5°.	~~	<u>م</u> ،	n- c
A T I	8 K	ANTEN	AXIMU A INVAL B	0.0		-		- ~ ~	20 ar N m	0.4	- 9. 2. <del>1</del> .	8.9 9	10.6	13.3	- 5-9 9	•	0.0	5.5	8.0 8.0		50	2 4 7 7	0 0 0 m	2°2	9.1 1	~;-	0 - F C
A D I THE F	LINDER TH = 1	RIZED	EAN M B ROR O	શ્ક	88	8	ŝŝ	88	88	5	55	88	9 8	99 19	2.22	3.72		₽ •	55	38	8	<u>5</u> 5	58	ຮຸ	ŝŝ	ຮຸຍ	5° = =
L R CK OF	LENG LENG	, POLA	TER E	5.1		2	<u>г</u> и		a	6.2	8. C.	<u> </u>	v =	8.2	r Em		। च न च न		2	æ-	: 5	<u>ر</u>				9	- ~ c
T H A	ROT.S AXIAI DIAME	ZONTAL	ER CEN		-, ~ -, ~	ŝ	ññ o o	ήm. hmi	53 x7	1 1 1	00 مع	0		- 666 mc	22	9 9 1 9	ጓሯ ጉጉ ጉጉ		- - - - - - - -	ې م - 0	ት ሻ ካ ማ	õ o	ρά γφγ	ې ۲	- m		747 500 r
I M U	BODY:	HORI	CA IN CENTI DB	<b>1</b> 1	i f	Ŧ	7 7	i vî i	ก๋ จ๋	4	ή. 		÷Ŧ	ιų.	ក៍	22	77	7	ÌĤ	Ϋ́	φ	¢ 4	न	٩, i	ŕŕ	Ϋ́	¥ Y Y
A Z HOMO	TESI		PHI	04	<u>∩</u> 2	5	ດ ເ	38	ન્ <i>≌</i>	÷	88	33	58	5.8	88	81	មខ្ម	5	25	₿Ķ	2 2 2 2	5	<u>ا</u>	ខ្ល	63	ភិះ	2228
																										-	
2 H F			SEVAR																	<u>~</u>			<u> </u>		2 01	~	
125 M H Z			M MAXIMUM R MASEVAR DEG	2.6	2.5	2.3	2.1	1.5		=.	o.=			1.5	9.1	1.6		0.1	۰. •	<b>ب</b> -			; m	, ,	Ĵ.	~ •	<u>, v v</u>
N C Y 125 M H Z	E TICAL B DEG	ENNA	HAXIMUM MAXIMUM Gainvar filasevar DB	3.2 2.6	3.2 2.5	3.2	3.2 2.1	3.3		3.5 .4	3.5	3.7 .7	3.9 1.3	4.0 1.5 1.5	H.2 1.6	4.3 1.6 	مر مر مر مر	1.0	4.5 -	د. د.					,	#.3 .2	
QUENCY 125 HHZ	rr wave = vertical = 80.8 deg	ED ANTENNA	FEAN MAXIMUM MAXIMUM ERFOR CAINVAR FHASEVAR DB DB DEG	12 3.2 2.6	-12 3.2 2.5	- 12 3.2 2.3	- 13 3.2 2.1	-13 3.3	15 3.4 1.2	- 15 3.5 .4	16 3.5 .0	17 3.77	19 3.9 1.3	20 4.0 1.5	-22 4.2 1.6	24 4.3 1.6		0.1 4.4 72	29 4.5 .5	30 4.5 .3		33 4.4 .2	34 4.4 .3	-35 4.3		36 4.3 .2	200
FREQUENCY 125 MHZ. LE ANTENNA	NCIDENT WAVE Diar. = Vertical Heta = 80.8 deg	LARIZED ANTENNA	ASE MEAN MAXIMUM MAXIMUM NTER ERROR CAINVAR FHASEVAR G DB DB DB	5.8 12 3.2 2.6	5.8 12 3.2 2.5	5.8 12 3.2 2.3	5.813 3.2 2.1	5.9 - 13 3.3 1.5	5.815 3.4 1.2	5.7 -15 3.5 .4	5.516 3.5 .0	4.9 17 3.7 .7	3.8 19 3.9 1.3	3.120 4.0 1.5	1.322 4.2 1.6	0.324 4.3 1.6	7.9 - 26 4.4 1.2	6.727 4.4 1.0	4.229 4.5 .5	3.030 4.5 .3		-, 1 -, 33 4, 4 .2	1.734 H.4 .3	2.335 4.3 .3	3.335 4.3 .2	3.636 4.3 .2	
RNFREQUENCY 125 HHZ AD FOLE ANTENNA	INCIDENT MAVE FOLAR. = VERTICAL THETA = 80.8 DEG	IAL POLARIZED ANTENNA	H PHASE MEAN MAXIMUM MAXIMUM Ler center Error Cainvar Phasevar Deg DB DB	1 15.812 3.2 2.6	1 15.8 12 3.2 2.5	1 15.8 12 3.2 2.3	2 15.813 3.2 2.1	3 15.9 -13 3.3 1.5	5 15.815 3.4 1.2 5 15.815 3.4 .8	6 15.715 3.5 .4	.7 15.516 3.5 .0 .8 15.217 3.6 .4	9 14.9 -17 3.7 .7	2 13.819 3.9 1.3	3 13.120 4.0 1.5	5 11.322 4.2 1.6	6 10.324 4.3 1.6		7 6.727 4.4 1.0	7 4.229 4.5 .5	7 3.030 4.5 .3	6 .8 -32 4.5 .1	5133 4.4 .2	. H -1.734 H.H .3		2 -3.335 4.32	1 -3.636 4.3 .2	
TTERN FREQUENCY 125 MHZ Meter Dipole Antenna	INT INCIDENT WAVE 0 M FOLAR. = VERTICAL 0 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	CAIN FHASE MEAN MAXIMUM MAXIMUM Center Error Cainvar Fhasevar DB Deg DB DB	-1.1 15.812 3.2 2.6	-1.1 15.8 12 3.2 2.5	-1.1 15.8 12 3.2 2.3		-1.3 15.9 -13 3.3 1.5		-1.6 15.715 3.5 .4	-1.7 15.516 3.5 .0 -1.8 15.217 3.6 .4	-1.9 14.917 3.7 .7	-2.2 13.8 -19 3.9 1.3		-2.5 11.322 4.2 1.6	-2.6 10.324 4.3 1.6		-2.7 6.727 4.4 1.0		2.6 1.030 4.5 .3	-2.6 .832 4.5 .1	-2.5133 4.4 .2 -2.4 -1.0 - 24 4.4 .2	-2.4 -1.734 4.4 .3	-2.3 -2.335 4.3 .3	-2.2 -3.335 4.3 .2	-2.1 -3.636 4.3 .2	
PATTERNFREQUENCY 125 HHZ A 0.1 HETER DIPOLE ANTENNA	ELD POINT INCIDENT WAVE T = . 10 M FOLAR. = VERTICAL =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	HUM CAIN HASE MEAN MAXIMUM MAXIMUM Sevar center Error Cainvar Phasevar DB deg db db deg	-1.1 15.812 3.2 2.6		-1.1 15.8 12 3.2 2.3		-1.3 5.9 -13 3.3 1.5	2 -1.4 15.914 3.4 1.2 -	2 -1.6 15.715 3.5 .4	2 -1.7 15.516 3.5 .0	1 -1.9 14.9 -17 3.7 .7	9 -2.2 13.8 -19 3.9 1.3		3 -2.5 11.322 4.2 1.6	0 -2.6 10.324 4.3 1.6		3 -2.7 6.727 4.4 1.0	5 2.7 4.2 -20 4.5 .5	2 -2.7 3.0 -30 4.5 .3	2.6 .8 .32 4.5			7 -2.3 -2.335 4.3 .3	3 -2.2 -3.335 4.3 .2	6 -2.1 -3.636 4.3 .2	-         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -
ION PATTERN FREQUENCY 125 HHZ ALONG A 0.1 METER DIPOLE ANTENNA	FIELD POINT INCIDENT MAVE DAT = . 10 M FOLAR. = VERTICAL HB =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	M MAXTMUM GAIN HASE MEAN MAXTMUM MAXIMUM R PHASEVAR CENTER ERROR GAINVAR FHASEVAR R DB DB DEG DB DB DEG	1.2 -1.1 15.812 3.2 2.6		1.2 -1.1 15.812 3.2 2.3			1.2 -1.4 15.914 3.4 1.2 1.2 -1.5 15.815 3.4 .8	1.2 -1.6 15.7 -15 3.5 .4	1.2 -1.7 15.516 3.5 .0 1.1 -1.8 15.217 3.6 .4	1.1 -1.9 14.9 -17 3.7 .7	.9 -2.2 13.8 -19 3.9 1.3		.3 2.5 11.322 4.2 1.6	-0 -2.6 10.324 4.3 1.6		1.3 -2.7 6.7 -27 4.4 1.0	2.5 2.7 4.2 -20 4.5 .5	3.2 -2.7 3.0 -30 4.5 3	4.2 -2.6 .8 -32 4.5 .1	4.52.5133 4.4 .2 4.52.4 -1034 4.8 2		3.7 -2.3 -2.335 4.3 .3	2.3 -2.2 -3.335 4.3 .2	1.6 2.1 -3.636 4.3 .2	
IATION PATTERN FREQUENCY 125 MHZ. FIELD ALONGA O.1 METER DIPOLE ANTENNA	R FIELD FOLMT INCLENT WAVE 1.80 M DAT = .10 M FOLAR. = VERTICAL .25 M HB =1.00 M THETA = 80.8 DEG	NTENNA RADIAL POLARIZED ANTENNA	MAZTMUM MAXIMUM GAIN HASE MEAN MAXIMUM MAXIMUM MAXIMUM Gainvar Phasevar center center error gainvar fhasevar DB deg db deg db db teg	.3 1.2 -1.1 15.812 3.2 2.6	.3 1.2 -1.1 15.812 3.2 2.5	.3 1.2 -1.1 15.812 3.2 2.3	.3 1.2 -1.2 15.8 -13 3.2 2.1		.2 1.2 -1.5 15.815 3.4 1.2	.2 1.2 -1.6 15.715 3.5 .4	2 1.2 -1.7 15.516 3.5 .0 2 1.1 -1.8 15.217 3.6 .4	2 1.1 -1.9 14.9 -17 3.7 .7	2 .9 -2.2 13.8 -19 3.9 1.3	2 .7 -2.3 13.120 4.0 1.5	2 .3 2.5 11.322 4.2 1.6	.2 .0 -2.6 10.324 4.3 1.6		·2 1.3 -2.7 6.727 4.4 1.0	.3 2.5 2.7 4.2 -29 4.5 .5	4 3.2 -2.7 3.0 -30 4.5 3	7 4.2 -2.6 .832 4.5 .1		1.3 4.2 -2.4 -1.734 4.4 .3	1.5 3.7 -2.3 -2.3 -35 4.3 .3	1.7 2.3 2.2 -3.335 4.3 .2	1.8 1.6 -2.1 -3.636 4.3 .2	
RADIATION PATTERN FREQUENCY 125 HHZ THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	NLINGER FIELD POINT INCIDENT WAVE 3TH = 1.80 M DAT = .10 M FOLAR. = VERTICAL = .25 M HB = 1.00 M THETA = 80.8 DEG	IZED ANTENNA RADIAL POLARIZED ANTENNA	TEAN MAXTMUM MAXTMUM GATN THASE MEAN MAXTMUM MAXTMUM MAXTMUM CATNVAR PHASEVAR CENTER ERROR GATIVVAR FHASEVAR 39 DB DB DEG DB DB DB DEG DB DB DB DEG DB DB DEG DB DB DB DG DB DB DB DG DB DB DB DG DB DB DB DG DB DB DB DB DB DB DG DB DB DB DG DB DB DB DG DB DB DB DG DB DB	.00 .3 1.2 -1.1 15.812 3.2 2.6	.00 .3 1.2 -1.1 15.8 -12 3.2 2.5	.00 .3 1.2 -1.1 15.8 -12 3.2 2.3	.00 .3 1.2 -1.2 15.8 -13 3.2 2.1		.00 .3 1.2 -1.4 15.914 3.4 1.2 .	.00 .2 1.2 -1.6 15.7 -15 3.5 .4	.00 .2 1.2 -1.7 15.516 3.5 .0 - .00 .2 1.1 -1.8 15.217 3.6 .4			$\begin{array}{cccccccccccccccccccccccccccccccccccc$	.00 .2 .3 .2.5 11.322 4.2 1.6	.00 .2 .0 -2.6 10.324 4.3 1.6	.00 .2 .8 -2.7 7.925 4.3 1.4	.00 .2 1.3 -2.7 6.727 4.4 1.0	.00 .3 2.5   -2.7 4.2 -2.9 4.5 .5	.00 4 3.2 -2.7 3.0 -30 4.5 .3			-01 1.3 4.2 -2.4 -1.734 4.4 .3	01 1.5 3.7 -2.3 -2.335 4.3 .3	01 1.7 2.3 -2.2 -3.335 4.3 .2	01 1.8 1.6 -2.1 -3.636 4.3 .2	
AL RADIATION PATTERN FREQUENCT 125 H H Z ECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	SYM.CVLINER FIELD FOINT INCIDENT MAVE L LENGTH = 1.80 M DAT = .10 M FOLAR. = VERTICAL LETER = .25 M HB =1.00 M THETA = 80.8 DEG	POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	ASE MEAN MAXIMUM MAXIMUM GAIN PASE MEAN MAXIMUM MAXIMUM MAXIMUM Witer Error Calinvar Phasevar C DB DB DB DEC DB DEC DB DEC DB DEC	1.8 .00 .3 1.2 -1.1 15.8 -12 3.2 2.6	2.3 .00 .3 1.2 -1.1 15.8 -1.2 3.2 2.5	2.9 .00 .3 1.2 -1.1 15.8 -12 3.2 2.3	13.7 .00 .3 1.2 -1.2 15.8 -13 3.2 2.1		9.4 .00 .2 1.2 -1.4 15.914 3.4 1.2 9.4 .00 .2 1.2 -1.5 15.815 3.4 .8 1	1.3 .00 .2 1.2 -1.6 15.7 -15 3.5 .4	3.5 .00 .2 1.2   -1.7 15.5 - 16 3.5 .0   5.9 .00 .2 1.1   -1.8 15.2 - 17 3.6 .4	8.5 .00 .2 1.1 -1.9 14.9 -17 3.7 .7	5.7 .00 .2 .9 2.2 13.8 -19 3.9 1.3	2:5 .00 .2 .7 -2.3 13.120 4.0 1.5	5.3 .00 .2 .3 2.5 11.322 4.2 1.6	il.4 .00 .2 .0 -2.6 10.324 4.3 1.6	2.6 .00 .2 .8 -2.7 7.9 -2.5 4.4 1.2 -	7.7 .00 .2 1.3 -2.7 6.7 -27 4.4 1.0	16.7 .00 .3 2.5   -2.7 4.229 4.5 .5	30.5 .00 .4 3.2 -2.7 3.0 -30 4.5 .3		39.400 .9 4.5 1 -2.5133 4.4 .2 1	10.1     1.3     1.2     -2.4     -1.7    34     4.4     .3	36.001 1.5 3.7 2.3 -2.335 4.3 .3	0.8 -01 1.7 2.3 -2.2 -3.335 4.3 .2	2.801 1.8 1.6 -2.1 -3.636 4.3 .2	7:0 - 1:0 1:0 1:1 - 2:1 - 5:9 - 50 4:3 :2 16:1 - 02 1:9 :8 - 2:1 - 4:0 - 36 4:2 :2 16:0 1:9 :7 - 5:1 - 4:0 - 36 4:2 :2
JTHAL RADIATION PATTERN FREQUENCY 125 HHZ TY CHECK OF THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA	ROT.SYM.COLINER FIELD POINT INCIDENT WAVE AXIAL LENGTH = 1.80 M DAT = .10 M POLAR. = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	IICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	I FHASE MEAN MAXTMUM MAXTMUM CAIN FHASE MEAN MAXTMUM MAXTMUM MAXTMUM Ter center error cainvar phasevar center center error cainvar fhasevar dec db db de dec db db dec db db	5 161.8 .00 .3 1.2 -1.1 15.812 3.2 2.6	5 162.3 .00 .3 1.2 -1.1 15.812 3.2 2.5	6 162.9 .00 .3 1.2 -1.1 15.8 -12 3.2 2.3	7 163.7 .00 .3 1.2 -1.2 15.8 -13 3.2 2.1 8 164.8 00 3 1.2 1.2 15.0 13 3.2 2.1		2 10/10 .00 .3 1.2   -1.4 15.9 -14 3.4 1.2   4 169.4 .00 .2 1.2   -1.5 15.8 -15 3.4 .8	7 171.3 .00 .2 1.2 -1.6 15.7 -15 3.5 .4	9 173-5 .00 .2 1.2   -1.7 15.516 3.5 .0   3 175-9 .00 .2 1.1   -1.8 15.217 3.6 .4	6 178.5 .00 .2 1.1 -1.9 14.9 -17 3.7 .7		13.1 - 12.5 = 0.0 = 2 = 7 = -2.3 = 13.1 = -2.0 = 1.0 = 1.5	0 -165.3 .00 .2 .3 -2.5 11.322 4.2 1.6	.5 -161.4 .00 .2 .0 -2.6 10.324 4.3 1.6	8 -152.6 .00 .2 .8 -2.7 7.9 -26 4.4 1.2 1	.5 -147.7 .00 .2 1.3 1 -2.7 6.727 4.4 1.0	0 -136.7 .00 .3 2.5   -2.7 4.2 -2.9 4.5 .5	17 -130.5 .00 .4 3.2 -2.7 3.030 4.5 .3 4 -123.8 00 5 3.8 -2.6 1.0 -31 4.5 .1	0-116.800 .7 4.2 -2.6 .832 4.5 .1	0 -109.400 .9 4.5   -2.5133 4.4 .2	4 -94.801 1.3 4.2 -2.4 -1.734 4.4 .3	.6 -88.001 1.5 3.7 -2.3 -2.335 4.3 .3	8 -76.8 -01 1.7 2.3 -2.2 -3.335 4.3 .2	.8 -72.801 1.8 1.6 -2.1 -3.636 4.3 .2	
IMUTHAL RADIATION PATTERN FREQUENCY 125 M H Z. DOGNETTY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTANNA	TBODY: ROT.SM.CYLINDER FIELD POINT INCIDENT MAVE AXIAL LENGTH = 1.80 M DAT = .10 M POLAR. = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	VERTICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	GAIN FHASE MEAN MAXIMUN br>Genter Error Ramita Firida Fir Deficia Firida FIR		1 -1.5 162.3 .00 .3 1.2 -1.1 15.812 3.2 2.5	-4.6 162.9 .00 .3 1.2 -1.1 15.812 3.2 2.3			-5.2 10(.0 .00 .3 1.2 -1.4 15.914 3.4 1.2 - -5.4 169.4 .00 .2 1.2 -1.5 15.815 3.4 .8	-5.7 171.3 .00 .2 1.2 -1.6 15.7 -15 3.5 .4	-5.9 173.5 .00 .2 1.2   -1.7 15.516 3.5 .0     -6.3 175.9 .00 .2 1.1   -1.8 15.217 3.6 .4	-6.6 178.5 .00 .2 1.1 -1.9 14.9 -17 3.7 .7		-7.9 - 172.5 .00 .2 .7 $-2.3$ 13.1 $-2.0$ 4.0 1.5 $-8$ 12.1 $-160$ 0 0 2 5 $-2$ 1 12.2 $-10$ 1 5 $-10$ 12.2 $-10$ 1 5 $-10$ 12.2 $-10$ 1 5 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2 $-10$ 12.2	-9.0 -165.3 .00 .2 .3 -2.5 11.322 4.2 1.6	-9.5 -161.4 .00 .2 .0 -2.6 10.324 4.3 1.6	1 = 10.6 = 152.6 . 00 . 2 . 8 $1 = 2.7$ 7.9 . 26 $1.4$ 1.2 $1 = 10.8 = 152.6$ . 00 . 2 . 8 $1 = 2.7$ 7.9 . 26 $1.4$ 1.2		-13.0 -136.7 .00 .3 2.5 -2.7 4.229 4.5 .5		-15.0 -116.800 .7 4.2 -2.6 .832 4.5 .1	1 - 15.6 - 109.400 .9 4.5 1 - 2.5 - 1 - 33 4.4 .2	1 -16.4 -94.801 1.3 4.2 -2.4 -1.734 4.4 .3	1 -16.6 -88.001 1.5 3.7 -2.3 -2.335 4.3 .3	-10.6 -76.801 1.7 2.3 -2.2 -3.335 4.3 .2	-16.8 -72.801 1.8 1.6 -2.1 -3.636 4.3 .2	-10.0 -09.0 -102 1.0 1.1 -2.1 -3.9 -3.0 -4.3 .2 -16.8 -6.1 -02 1.9 .8 -2.1 -4.0 -3.6 4.2 .2 -16.8 -6.7 -07 1.9 .7 -2.1 -4.0 .36 4.2 1

_																						1	C7\	1	200	N N	1471			
ΖH				EVAB		~ ~	0.0		~ ~			0.01	- 00	<b>љ</b> и	o			<del>م</del> ع	. 6.	N 6	<u>~~</u>	<mark>ا</mark> ل	52	- (	0.00	<u>.</u>	-#			
м 8		ľ		PHAS	44.5	1. CH	្អ	ŝ	31	พ่อ	28.0	ູ່	ĕ₽		° 'n	••		æ =	m	<u>é</u> e	20	ສ່ະ	ೇಸೆ	ສູ່	22	58	8			
₩ ₩		TAL D		WWW	‡ ≞ ∾	~~~	N, C	y m	n n			1 <b>-</b> 1	ņņ	m, c	'n			oʻc	. e.	ņ		9	25		55	~	N.			
U ×		B DE	ENNA	MAX MAX B G IN	•	<b>ณ</b> ์ ณี	i ni r	้เก่	N N	i ni d	ú ní d	ŇŇ	v v	N C	ŇŇ	~ ~	101	~ ~	1-	- (	10	~								
Э	NN	NAN BOR	ANT	ROR	∎ R	ភុគ្គ	គ្គទ	កស្	22	សុខ	ទុសុខ	ຊ ຊະ 	22	ក្	0≇ 11	₫: ;	200	÷	22	₽≑ ſ	1 I 7 7	₹\ 7	1.1 515	ម្ព	្រុះ	-				
ы Ш	ANTE	DENT	IZED	988 988					 			 	 o=	m.	- ∞		~ ∞	-0	J =7		~ @	ŝ	ρvα	-		. 00	<u>ب</u>			
5	OLE	1921	OLAB	HASE	-37.6	9°.	N.	λų γ	2	Ę	, i	° Ž	5.8	ŝ	şğ	ŝ	<u>5</u> .6	17	8	5	ະ ເຕ	ŝ	Ξĝ	<b>1</b> 08.	ËĒ	112	133			
ž	DIP 1		AL F	LER L	* 8		مونو	ې د	~ ~	i	ပဲထဲ	- 9	<b>,</b> .	æ,	₹.	ŝ		~ •	; ņ	<u>.</u>	• •	<u>م</u>	~ ~	~.	0,=	8.	8.2			
ы н	ETER	HEE	RADI	CEN B	÷.	55	9	የዋ	ዮ ተ	ጥ	77	<b>7 '</b>	1	•		- 1				-	'	•	7 Y	٦	77	7	71			
A T	1.	<u>2</u> 28		-5			_																							
2	A C			ASEVI	ەب	<u>م</u> ۽		<b>~</b> @	<u>m</u> 2	35	N	0 0	<u></u>	~	₹ <u>0</u>	9	2°	~		<del>م</del> ،	- ~ - ~	5	m 0.	6.4	~ 0 6 #	6	۰.0 0,5			
ē	NLONC	C 7 R	M	WE BO	00	41	- 6 (	22	52	ន្ល	2.6	552	22	Ĕ:	56	ыX і	22	20	6.9	2	ōin		m w	Ñ		•				
E	013	¥Σ 8%	NTENI	INVAL	0.0	1.0	- 61	~ æ.	6.0 7.6			ຊ.ຍ ເອີຍ ເອີຍ	2.7 6.9	(6·9)	9.1	5	0.0 m N	2.2	v m	9. 		7	 	5	2.5	<u> </u>	•••			
Ē	E FII	ЕН. 	ED A	¥38	90	29	2 E E	υΞ	ΩΞ	:5	22	9 8 2	: 2	رچ 1	នភ	8	ю=	5	≘ನ	សា	ຂຕ	28	58	8	₽₹	ξ	ъъ			
A I	H	YL IN GTH	ARIZ	MEAN Erro DB					<b>-</b> .			Ϋ́	1	•	•••			7	11	i	11		•	•	• •	•••	••			
5	CK OI	TER TER	DC.	LEB .	<b></b> = 0		, <u>.</u> ,		90	0.00	0.0	9°2	* 0 * -	6.1	9 O.H	5	<del>ہ</del> ۔ ۔		?.~	6.6	2.0	8 8 8 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	9.0	5		5.9	1.5			
Η	CHB	OT.S XIAL	NTAL	DEC	<b>គឺ</b> គឺ	ဆိုမ်	Ř	ř.	44	ŶΫ	፝፝ጘ	ή'n	äφ	1	ųφ	2	29	÷	? <b>?</b>	ę	ግግ 	7	ዋ ኘ ····	ነጥ · · · ·	"ግ	17	77 			
5	EITY	¥: ₽	RIZO	LIN ENTER	2.0	2	ကိုလို	ਕ ∾ ਯੋਜ	2° 4	- 0- 9 @	10.0	16.1 21.9	24.5	18.8	15.8	12.9	12.1	2.5	0.4	2-6-	۵.C	5	φ'n	Ĩ	ร้า	ĥ	ጥጥ			
Ξ	IOCEN	TBOD	Ŧ	358	0.0			50	in c	2.0	<u></u>	1 T 9 0	11	1	<u></u>	1 10	<u>8 </u>		<u>. 0</u>	ۍ ۲	2.5	20	<u> </u>	10	S.f	22	128			
Z V	ě	122	L_	Eğ			- 2	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	- m =	1 #1 		<u>و ہ</u>	~~	- @	~~~		20	=:	= =	2			~ =		<u>~</u>		22			
																•														
	<u> </u>				1												_													
2 H F				SEVAR	0.0	=	 > ~	च न	50	- ~	- =	- ~	r-==	=	<u>.</u>		<u>~-</u>	5		6	<u>ب</u> ہ			i o			<u>N</u> -			
300 M H Z				MAXIMUM PHASEVAR DEG	31.0 30.6		2.2	22.4	16.5	11.2	9.1 7.4	6.1 5.2	н. 1 1	п п	9°4	ŝ	2.1	6.5	6.9	6.9	6.7		5.7	1°0	4°	- = - -	0- 0			
C Y 300 M H Z		AL EG	A	JHUM MAXIMUM INVAR MASEVAR DEG	5.1 31.0 5.1 30.6	1.53.1	5.1 25.2	5.0 22.4 1.9 19.4	16.5	11.2	u.2 9.1 u.0 7.4	3.8 6.1 3.7 5.2	3.5 H.7 3.4 H.4	3.3 4.4	3.2	3.2	3.2 5.7	3.3 6.5	3.3 6.7	3.5 6.9	3.6 6.7	3.8	3.9 5.7	4.0 4.6	4.1 4.1	4.1	4.1 3.2			
ENCY 300 MHZ		VE RTICAL • 8 DEG	ITENNA	MAXIMUM MAXIMUM I GAINVAR PHASEVAR DB DEG	1 5.1 31.0 5.1 30.6	5.1 29.4	3 5.1 25.2	9 5.0 22.4	0 4.8 16.5	1 4.0 13.1	2 4.2 9.1 0 4.0 7.4	9 3.8 6.1 9 3.7 5.2	9 3.5 H.7	9 3.3 4.4	9 3.2 4.6	0 3.2	0 3.2 5.7	1 3.3 6.5	23.3 6.7	2 3.5 6.9	2 3.6 6.7	3.8.6.1	3 3.9 5.7	23 <b>1.0</b>	24 P. 1 24 P. 1 25 P. 1 27		24 4.1 3.2			
Q U E N C Y 300 M H Z	ENNA	T WAVE = VENTICAL = 80.8 DEG	ED ANTENNA	IEAN MAXIMUM MAXIMUM Error gainvar phasevar 38 db db df.g	54 51 53 1 31.0	-51 5.1 29.4	-4 2.1 25.2	39 5.0 22.4	-30 4.8 16.5	24 4.4 11.2	22 4.2 9.1 20 4.0 7.4	19 3.8 6.1 19 3.7 5.2	19 3.5 4.7 19 3.4 4.4	- 19 3.3 4.4	19 3.2 4.6	- 20 3.2	20 3.2 5.7	21 3.3 6.5	21 3.3 6.7	-22 3.5 6.9	22 3.6 6.7	23 3.8 6.1	23 3.9 5.7	-23 4.0 4.6	24 4.1 4.1		24 4.1 3.2			
REQUENCY 300 MHZ	ANTENNA	IDENT WAVE AR. = VENTICAL TA = 80.8 DEG	RRIZED ANTENNA	SE MEAN MAXIMUM MAXIMUM Fer error cainvar phasevar dê db de	2 - 54 5.1 31.0 - 53 5.1 30.6		.3 -43 5.1 25.2	.539 5.0 22.4	5 - 30 4.8 16.5	.624 4.4 11.2	.922 4.2 9.1 .620 4.0 7.4	.419 3.8 6.1 .519 3.7 5.2	-7 19 3.5 4.7 19 3.4 4.4	.319 3.3 4.4	.719 3.2 4.6	.320 3.2 5.3		3.621 3.3 6.5	0.321 3.3 6.7	3.322 3.5 6.9	1.422 3.6 6.7	5.123 3.8 6.1	5.723 3.9 5.7	7.523 4.0 4.6	7.724 4.1 4.1	7.024 4.1 3.4	7.924 4.1 3.2 7.924 4.1 3.1			
FREQUENCY 300 MHZ	POLE ANTENNA	INCIDENT WAVE POLAR. = VERTICAL THETA = 80.8 DEG	POLARIZED ANTENNA	FHASE MEAN MAXIMUM MAXIMUM Center Error Gainvar Phasevar Dec db db db	118.254 5.1 31.0 118.253 5.1 30.6	118.251 5.1 29.4	118.3 -41 5.1 25.2	118.539 5.0 22.4	119.5 - 30 4.8 16.5	121.624 4.4 11.2	122.922 4.2 9.1 124.620 4.0 7.4	126.419 3.8 6.1 128.519 3.7 5.2	130.7 19 3.5 4.7 133.0 19 3.4 4.4	135.3 19 3.3 4.4	137.719 3.2 4.6	142.3 20 3.2 5.3	144.620 3.2 5.7 146.621 3.2 6.1	148.621 3.3 6.5	150.3 21 3.3 6.7	153.322 3.5 6.9	154.422 3.6 6.7	156.123 3.8 6.1	156.723 3.9 5.7	157.5 - 23 4.0 4.6	157.724 4.1 4.1	157.024 4.1 3.4	157.924 4.1 3.2 157.924 4.1 3.1			
RNFREQUENCY 300 MHZ	CR DIPOLE ANTENNA	INCIDENT WAVE Polar. = VERTICAL Theta = 80.8 deg	DIAL POLARIZED ANTENNA	IN FHASE MEAN MAXIMUM MAXIMUM NTER CENTER ERROR GAINVAR FHASEVAR Dec db db db dfg	2.7 118.254 5.1 31.0 2.6 118.254 5.1 30.6	2.5 118.2 -51 5.1 29.4	1.8 118.3 -43 5.1 25.2	1.4 118.539 5.0 22.4 0.0 0 118.934 1.0	0.4 119.5 - 30 4.8 16.5	9.2 121.624 4.4 11.2	8.6 122.922 4.2 9.1 8.0 124.620 4.0 7.4	7.4 126.4 19 3.8 6.1 6.8 128.5 19 3.7 5.2	6.3 130.7 19 3.5 4.7 5.8 133.0 19 3.4 4.4	5.3 135.3 19 3.3 4.4	4.9 137.719 3.2 4.6	4.3 142.320 3.2 5.3	4.0 144.620 3.2 5.7 3 8 146.621 3.2 6.1	3.6 148.621 3.3 6.5	-3.5 150.321 3.3 6.7	3.4 153.322 3.5 6.9	-3.3 154.422 3.6 6.7	-3.3 156.123 3.8 6.1	-3.4 156.723 3.9 5.7	-3.4 157.523 4.0 4.6	-3.4 157.724 4.1 4.1	-3.4 15/10 -1.24 4.1 3.4	-3.5 157.924 4.1 3.2			
TERN FREQUENCY 300 MHZ	METER DIPOLE ANTENNA	NT INCIDENT WAVE M POLAR. = VERTICAL M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	GAIN FHASE MEAN MAXIMUM MAXIMUM Center Center Error Gainvar Fhasevar DB deg db db db	-12.7 118.254 5.1 31.0 -12.6 118.253 5.1 30.6	-12.5 118.251 5.1 29.4	-11.8 118.3 -43 5.1 25.2	-11.4 118.5 39 5.0 22.4	-10.4 119.5 - 30 4.8 16.5	-9.6 121.624 4.4 11.2		-7.4 126.419 3.8 6.1 -6.8 128.519 3.7 5.2	-6.3 130.7 19 3.5 4.7	-5.3 135.3 - 19 3.3 4.4	-4.9 137.719 3.2 4.6		-4.0 144.620 3.2 5.7 -3.8 146.621 3.2 5.1	-3.6 148.621 3.3 6.5	-3.5 150.321 3.3 6.7	-3.4 153.322 3.5 6.9	-3.3 154.422 3.6 6.7	-3.3 156.123 3.8 6.1	-3.4 156.723 3.9 5.7	-3.4 157.523 4.0 4.6	-3.4 157.724 4.1 4.1	-3.4 15/10 -1.24 4.1 3.4	-3.5 157.9 - 24 4.1 3.2			
PATTERN FREQUENCY 300 MHZ	0.1 METER DIPOLE ANTENNA	D POINT INCIDENT WAVE = 10 M POLAR. = VENTICAL =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	UN CAIN FHASE MEAN MAXIMUM MAXIMUM Var center center error cainvar fhasevar db dec db db dec	-12.7 118.254 5.1 31.0 -12.6 118.253 5.1 30.6	-12.5 118.2 - 51 5.1 29.4	-11.8 118.343 5.1 25.2	-11.4 118.539 5.0 22.4	-10.4 119.5 - 30 4.8 16.5	-9.6 120.421 4.0 13.1	-8.6 122.922 4.2 9.1 -8.0 124.620 4.0 7.4	-7.4 126.419 3.8 6.1 -6.8 128.519 3.7 5.2	-6.3 130.7 - 19 3.5 4.7	-5.3 135.3 19 3.3 4.4	-4.9 137.7 19 3.2 4.6	1 11.3 112.320 3.2 5.3	-4.0 144.620 3.2 5.7	-3.6 148.621 3.3 6.5	-3.5 150.321 3.3 6.7	-3.4 153.322 3.5 6.9	-3.3 154.422 3.6 6.7	-3.3 156.123 3.8 6.1	-3.4 156.723 3.9 5.7	-3.4 157.5 -23 4.0 4.6	-3.4 157.724 4.1 4.1	-3.5 157.024 4.1 3.4	-3.5 157.9 - 24 4.1 3.2 -3.5 157.9 - 24 4.1 3.2			
N PATTERN FREQUENCY 300 MHZ	WG A 0.1 METER DIPOLE ANTENNA	FIELD FOINT INCIDENT WAVE MAT = .10 M FOLAR. = VENTICAL HB =1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	AZTHUM GAIN FLASE MEAN MAXIMUM MAXIMUM Hasevar center center erron gainvar hasevar eg db db deg db db deg	7.3 -12.7 118.254 5.1 31.0 -12.6 118.253 5.1 30.6	7.3 -12.5 118.2 -51 5.1 29.4	7.3 -11.8 118.3 -4.4 5.1 25.2	7.4   -11.4 118.539 5.0 22.4   7.4   -10.0 118.934 4.9 19.4	7.5 -10.4 119.5 - 30 4.8 16.5	7.6 -9.8 120.421 4.9 13.1	7.6 -8.6 122.922 4.2 9.1 7.7 -8.0 124.620 4.0 7.4	7.7 -7.4 126.419 3.8 6.1 -7.8 -19 3.8 5.1 -7.8 -5.8 128.5 -19 3.7 5.2	7.9 -6.3 130.7 -19 3.5 4.7 7.6 -5.8 133.0 -19 3.4 4.4	8.0 -5.3 135.3 -19 3.3 4.4	8.1 - 4.9 137.7 - 19 3.2 4.6	8.2 1 4.3 142.320 3.2 5.3	8.3 4.0 144.620 3.2 5.7 8.1 -3.8 146.621 3.2 6.1	8.4 -3.6 148.621 3.3 6.5	8.3 -3.5 150.3 -21 3.3 6.7 1 2.1 151 0 -22 3.4 6.9	7.8 -3.4 153.322 3.5 6.9	7.2 -3.3 154.422 3.6 6.7	4.7 -3.3 156.1 -23 3.8 6.1	2.8 -3.4 156.723 3.9 5.7		2.3 -3.4 157.724 4.1 4.1	3.0 -5.4 15/.024 4.1 3.4	3.4 -3.5 157.9 -24 4.1 3.2 3.4 -3.5 157.9 -24 4.1 3.2			
ION PATTERN FREQUENCY 300 MHZ	ALONG A 0.1 METER DIPOLE ANTENNA	FIELD FOINT INCIDENT WAVE MIT = .10 M FOLAR. = VENTICAL HB =1.00 M THETA = 80.8 DEG	A RADIAL FOLARIZED ANTENNA	UN MAZTMUM GATN FHASE MEAN MAZTMUM MAZTMUM MAZTMUM AR PHASEVAR CENTER CENTER ERROR GATWAR PHASEVAR DE	7.3 -12.7 118.254 5.1 31.0 7.3 -12.6 118.253 5.1 30.6	7.3 -12.5 118.2 -51 5.1 29.4	7.3 -11.8 118.3 -43 5.1 25.2	7.4 - 11.4 118.5 - 39 5.0 22.4	7.5 -10.4 119.5 - 30 4.8 16.5	7.6 -9.2 121.624 4.4 11.2	7.6 -8.6 122.922 4.2 9.1 7.7 -8.0 124.620 4.0 7.4	7.7 -7.4 126.419 3.8 6.1	7.9 -6.3 130.7 - 19 3.5 4.7		8.1 4.9 137.7 - 19 3.2 4.6	2 8.2 14.3 142.3 -20 3.2 5.3	8.3 Ju.0 144.620 3.2 5.7	th 8.4   -3.6 148.621 3.3 6.5	5 8.3 -3.5 150.321 3.3 6.7	8 7.8 2.4 153.322 3.5 6.9	0 7.2 -3.3 154.422 3.6 6.7	2 4.7	2 2.8 -3.4 156.723 3.9 5.7	1 1 1	9 2.3 -3.4 157.7 -24 4.1 4.1	7 3.0 -5.4 10/0 -5.4 10/0 - 5.4 10 5.1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4 1 3.4	L 3.4 -3.5 157.9 -24 4.1 3.2			
ATION PATTERN FREQUENCY 300 MHZ	IELD ALONG A 0.1 METER DIPOLE ANTENNA	FIELD POINT INCIDENT WAVE 80 M DAT = . 10 M POLAR. = VERTICAL .25 M HB =1.00 M THETA = 80.8 DEG	TENNA RADIAL POLARIZED ANTENNA	AXINUM MAXIMUM CATIN FHASE MEAN MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM FHASEVAR Ativara Phasevar Conter Center Error Catination Dec B dec db db dec db db dec	·1 7.3 -12.7 118.254 5.1 31.0 -12.6 118.253 5.1 30.6	1 7.3 -12.5 118.2 -51 5.1 29.4	.1 7.3 -12.6 118.2 -41 3.1 61.0	1 7.4 -11.4 118.539 5.0 22.4		1 7.6 -9.2 121.6 -24 4.4 11.2	.0 7.6 -8.6 122.922 4.2 9.1 0 7.7 -8.0 124.620 4.0 7.4	0 7.7 -7.4 126.419 3.8 6.1 0 7.8 -6.8 128.519 3.7 5.2	.0 7.9 -6.3 130.7 19 3.5 4.7 .1 7.6 -5.8 133.0 19 3.4 4.4	.1 8.0 -5.3 135.3 -19 3.3 4.4	1 8.1 4.9 137.7 -19 3.2 4.6	2 8.2 4.3 112.3 -20 3.2 5.3	.3 8.3 _4.0 144.620 3.2 5.7	. 4 8.4   -3.6 148.621 3.3 6.5	5 8.3 -3.5 150.321 3.3 6.7	8 7.8 -3.4 153.3 -22 3.5 6.9	. 9 7.2 -3.3 154.422 3.6 6.7	1.1 0.2	1.2 2.8 -3.4 156.723 3.9 5.7		.9 2.3 -3.4 157.724 4.1 4.1	-7 3.0 -5.4 15/.024 4.1 3.4				
DIATION PATTERN FREQUENCY 300MHZ	HE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	NDER FIELD FOINT INCIDENT WAVE = 1.80 M DNT = .10 M FOLAR. = VERTICAL = .25 M HB =1.00 M THETA = 80.8 DEG	D ANTENNA RADIAL POLARIZED ANTENNA	M HAZIMUM MAZIMUM GAIN FHASE MEAN MAZIMUM MAZIMU Ga cativer fister fi	00 .1 7.3 -12.7 118.254 5.1 31.0	00 .1 7.3 -12.5 118.2 -51 5.1 29.4	00 .1 7.3 -16.6 118.3 -4.4 5.1 25.2 1 00 .1 7.3 -11.8 118.3 -4.43 5.1 25.2	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		00 .1 7.5 -9.6 120.421 4.0 15.1 0 00 .1 7.6 -9.2 121.624 4.4 11.2	00 .0 7.6 -8.6 122.922 4.2 9.1 00 .0 7.7 -8.0 124.620 4.0 7.4		00 0 7.9 -6.3 130.7 - 19 3.5 4.7 7.8 132.0 - 19 3.4 4.4	00 .1 8.0 -5.3 135.3 -19 3.3 <sup>1.1</sup>	.00 .1 8.1 _4.9 137.719 3.2 4.6	00 .2 8.2 4.3 112.320 3.2 5.3	00 3 8.3 4.0 144.6 - 20 3.2 5.7	.00 .4 8.4 -3.6 148.621 3.3 6.5	01 5 8.3 -3.5 150.3 -21 3.3 6.7	01 8 7.8 3.4 153.3 -22 3.5 6.9	.01 .9 7.2 -3.3 154.422 3.6 6.7	.01 1.1 0.2 -5.3 125.422 3.8 5.1	-01 1.2 2.8 -3.4 156.723 3.9 5.7				01 4 3.4 -3.5 157.9 -24 4.1 3.2			
RADIATION PATTERN FREQUENCY 300 MHZ	OF THE FIELD ALONG A 0.1 HETER DIPOLE ANTENNA	CVLINDER FIELD FOLMT INCIDENT WAVE WITH = 1,80 M INT = , 10 M FOLAR. = VENTICAL = ,25 M HB = 1.00 M THETA = 80.8 DEG	RIZED ANTENNA RADIAL POLARIZED ANTENNA	MEAN MAXIMUM MAXIMUM GAIN FHASE MEAN MAXIMUM MAXIMUM MAXIMUM FERROR GAINVAR PHASEVAR CENTER CENTER ERROR GAINVAR PHASEVAR DB DB D	00 .1 7.3 -12.7 118.254 5.1 31.0	00 .1 7.3 -12.5 118.251 5.1 29.4	00 .1 7.3   -12.6 110.641 2.1 21.0   00 .1 7.3   -11.8 118.343 5.1 25.2	00 1 7.4 -11.4 118.539 5.0 22.4		00 .1 7.5 -9.6 120.421 4.0 13.1	00 .0 7.6 -8.6 122.922 4.2 9.1 00 .0 7.7 -8.0 124.620 4.0 7.4	00 .0 7.7 -7.4 126.419 3.8 6.1	00 .0 7.9 -6.3 130.719 3.5 4.7 		-500 $1$ $8.1$ $-4.9$ $137.7$ $-19$ $3.2$ $4.6$ $-10$ $3.2$ $1.6$	00 .2 8.2 4.3 142.320 3.2 5.3	00 .3 8.3 _4.0 144.620 3.2 5.7	00 .4 8.4 -3.6 148.621 3.3 6.5	-01 5 8.3 -3.5 150.3 -21 3.3 6.7	-01 8 7.8 -3.4 153.3 -22 3.5 6.9	01 .9 7.2 -3.3 154.422 3.6 6.7		01 1.2 2.8 -3.4 156.723 3.9 5.7							
AL RADIATION PATTERN FREQUENCY 300 MHZ	ECK OF THE FIELD ALONG A 0.1 HETER DIPOLE ANTENNA	SW-CALINER FIELD POINT INCIDENT WAVE L LEWGTH = 1.80 M DAT = . 10 M POLAR. = VENTICAL ETER =	POLARIZED ANTENNA RADIAL FOLARIZED ANTENNA	ASE MEAN MAXIMUM MAXIMUM CAIN FRASE MEAN MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM RIASEVAR Niter Error Cainvar Phasevar Center Center Brror Cainwar Hasevar C DB DB DB DE DEG DB DB DE DEG DB DB DB	3.600 .1 7.3 -12.7 118.254 5.1 31.0 -12.6 118.253 5.1 30.6	Hic00 .1 7.3 -12.5 118.251 5.1 29.4	<b>55.900 .1 7.3   -12.6 118.24 5.1 25.2  </b> <b>77.700 .1 7.3   -11.8 118.34 5.1 25.2  </b>	No.0        00         1         7.4         -11.4         118.5        39         5.0         22.4           No.7         7         7         1         -10.0         118.9        34         4.9         19.4	200 - 00 1 7.5 - 10.4 119.5 - 30 4.8 16.5	73.7 - 00 $1$ $7.5$ $-9.8$ $120.4$ $-21$ $4.9$ $13.7$ $11.2$ $13.9$ $-00$ $1$ $7.6$ $-9.2$ $121.6$ $-24$ $4.4$ $11.2$	18.400 .0 7.6 -8.6 122.922 4.2 9.1 33 400 .0 7.7 -8.0 124.620 4.0 7.4	28.8 - 00 0 7.7 - 7.4 126.4 - 19 3.8 6.1	10.5 - 00 0 7.9 -6.3 130.7 - 19 3.5 <sup>1</sup> .7	52.400 -1 8.0 -5.3 135.319 3.3 4.4	50.200 .1 8.1 -4.9 137.719 3.2 4.6	07.200 .2 8.2 4.3 142.320 3.2 5.3	78.2 - 00 3 8.3 -4.0 144.6 - 20 3.2 5.7	62.800 .4 8.4   -3.6 148.621 3.3 6.5	54.701 .5 8.3 -3.5 150.321 3.3 6.7	70 - 01 - 8 7.8 1 - 3.4 153.3 - 22 3.5 6.9	27.2 -:01 :9 7.2 -3.3 154.4 -:22 3.6 6.7	16.501 1.1 0.25.3 125.422 3.4 6.1	92.501 1.2 2.8 -3.4 156.723 3.9 5.7	79.8 -01 1.2 (1 -3.4 157.5 -23 4.0 4.6		<b>A8.9</b> 01 .7 3.0 -5.4 15/0024 4.1 3.4	330,4 -01 -1 3.4 -3.5 157.9 -24 4.1 3.2 330,4 -01 -4 3.4 -3.5 157.9 -24 4.1 3.2			
THAL RADIATION PATTERN FREQUENCY 300 MHZ	Y CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	ROT.SDM.CRLINDER FIELD POINT INCIDENT WAVE AXIAL LENCTH = 1,80 M DAT = . 10 M POLAR. = VENTICAL DIAMETER =	CAL FOLARIZED ANTENNA RADIAL POLARIZED ANTENNA	PHASE MEAN MAXIMUM MAXIMUM CAIN FHASE MEAN MAXIMUM MAXIMUM MAXIMUM MAXIMUM RASEVAR R center error cainvar Phasevar R center error cainvar Phasevar R center dror cainvar R center dror cainvar R center dror cainvar Phasevar R center dror cainvar R center d	93.600 .1 7.3 -12.7 118.254 5.1 31.0	94.600 .1 7.3 -12.5 118.251 5.1 29.4	5 95.900 .1 7.3   -12.6 110.64 2.1 21.6   1.0   1 97.700 .1 7.3   -11.8 118.343 5.1 25.2	3 100.000 .1 7.4 -11.4 118.539 5.0 22.4	106.000 .1 7.5 -10.4 119.530 4.8 16.5	5 113.900 .1 7.5 -9.8 120.421 4.9 13.7 5 113.900 .1 7.692 121.624 4.4 11.2	9 118.400 .0 7.6 -8.6 122.922 4.2 9.1	5 128.800 .0 7.7 -7.4 126.419 3.8 6.1 - 5 128.500 0 7.8 -6.8 128.519 3.7 5.2	5 140.5 - 00 : 0 7.9 - 6.3 130.7 - 19 3.5 4.7 5 146 - 00 : 0 7.9 - 5.8 133.0 - 19 3.4 4.4	5 153.400 .1 8.0 1 -5.3 135.319 3.3 4.4	1 160.2 - 00 $1 8.1 - 4.9 137.7 - 19 3.2 4.6 + 6 100 - 10 3.2 4.6$	1 167.200 .2 8.2 1 140.0 3.2 5.3	1-178.2 - 00 .3 8.3 -4.0 144.6 - 20 3.2 5.7	9 - 1 (0.000 .3 0.4 1 - 3.6 148.6 - 21 3.3 6.5	7 -154.701 .5 8.3 -3.5 150.321 3.3 6.7	8 -170 - 01 - 8 7.8 1 - 3.4 153.3 - 22 3.5 6.9	9 -127.201 .9 7.2 -3.3 154.422 3.6 6.7	0 - 116.5 - 01 1.1 0.2 - 3.3 156.1 - 23 3.8 6.1 1 1 - 01 1.2 4.7 - 3.3 156.1 - 23 3.8 6.1	9 -92.501 1.2 2.8 -3.4 156.723 3.9 5.7			9	7 -38.4 -01 -0 -3.4 -5.5 157.9 -24 4.1 3.2 7 -38.4 -01 -4 3.4 -5.5 157.9 -24 4.1 3.1			
MUTHAL RADIATION PATTERN FREQUENCY 300 MHZ	NEITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	DY: ROT.SWCALINER FIELD FOINT INCIDENT WAVE AXIAL LEWGTH = 1.80 M DMT = .10 M FOLAR = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	ERTICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	AIM FHASE MEAN MAXIMUM MAXIMUM CAIM FHASE MEAN MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM FAREVAR CENTER ERROR CAINVAR FHASEVAR Estiter center error cainvar fhasevar center center error cainvar fhasevar B dec db db de teg teg teg teg teg teg teg teg teg te	.6 33.600 .1 7.3 -12.7 118.254 5.1 31.0	.6 94.600 .1 7.3 -12.5 118.251 5.1 29.4	1 07.700 .1 7.3 -11.6 118.347 5.1 25.2	3 100.0 - 00 $1 7.4 - 11.4 118.5 - 39 5.0 22.4 3 100.7 - 00 1 7.4 - 10.0 118.9 - 34 4.9 19.4 - 10.0 118.9 - 34 4.9 19.4$	-1 106.000 -1 7.5 -10.4 119.530 4.8 16.5	3 109.700 .1 7.5 -9.6 120.424 4.9 11.2	9 118.400 .0 7.6 -8.6 122.922 4.2 9.1 -1 2 123 400 .0 7.7 -8.0 124.620 4.0 7.4	-1.6 128.800 .0 7.7 -7.4 126.419 3.8 6.1 -2.0 128.500 0 7.8 -6.8 128.519 3.7 5.2	-2.5 140.500 .0 7.9 -6.3 130.7 -19 3.5 4.7	-3.0 140.000 .1 8.0 1 -5.3 135.319 3.3 4.4	1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1 <th1< th=""> <th1< th=""> <th1< th=""> <th1< th=""></th1<></th1<></th1<></th1<>		-6.1 -178.200 .3 8.3 -4.0 144.620 3.2 5.7	-0.9 -1/0.000 .3 0.4 -3.6 148.621 3.3 6.5	$-9.7 - 154.7 - 01$ $\cdot 5$ $8.3$ $-3.5$ $150.3$ $-21$ $3.3$ $6.7$		-11.9 -127.201	-13.0 - 116.5 - 01 1.1 0.2 - 5.3 125.4 - 22 5.1 0.2 - 11.1 - 104.9 - 00 1.2 4.7 - 3.3 156.1 - 23 3.8 6.1	-14.9 -92.501 1.2 2.8 -3.4 156.723 3.9 5.7		$-16.0 -57.3 -01 \cdot 9 \cdot 2.3 -3.4 \cdot 157.7 -2.4 \cdot 1 \cdot $		-15.6 -15.0 -101 -1 5.4 -5.5 157.9 -24 4.1 5.2 -15.7 -39.4 -01 -4 3.4 -5.5 157.9 -24 4.1 3.1 -1.5 157.9 -24 4.1 3.1			
ZIMUTHAL RADIATION PATTERN FREQUENCY 300 MHZ	MOCENEITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	STBODY: ROT.SNL.CR.INDER FIELD FOINT INCIDENT WAVE AXIAL LENCTH = 1.80 M DMT = .10 M POLAR. = VERTICAL DIANETER = .25 M HB =1.00 M THETA = 80.8 DEG	VERTICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	II GAIN PHASE MEAN MAXIMUM MAXIMUM GAIN FHASE MEAN MAXIMUM MAXIMUM AXIMUM CENTER CENTER ERROR CAINVAR PHASEVAR CENTER CENTER ERROR CAINVAR PHASEVAR C DB DEC DB DB DB DE DEC DB DB DEC DB DB DFC	0 .6 33.600 .1 7.3 -12.7 118.254 5.1 31.0	10 .6 94.600 .1 7.3 -12.5 118.251 5.1 29.4	15 .5 95.900 .1 7.3   -12.6 110.641 2.1 21.0   10 01 .14 07.700 .1 7.3   -11.8 118.343 5.1 25.2	55 .3 100.000 .1 7.4 -11.4 118.539 5.0 22.4	30         1         106.0        00         1         7.5         -10.4         119.5        30         4.8         16.5           55        1         106.0        00         .1         7.5         -10.4         119.5        30         4.8         16.5	103 109.700 .1 7.5 -9.8 120.421 4.9 13.7 1 156 113.900 .1 7.6 -9.2 121.624 4.4 11.2	50 - 9 118.400 .0 7.6 -8.6 122.922 4.2 9.1 56 -1 2 123 800 .0 7.7 -8.0 124.620 4.0 7.4	50 -1.6 128.800 0 7.7 -7.4 126.419 3.8 6.1 51 -1.6 128.800 0 7.8 -6.8 128.519 3.7 5.2	70 -2.5 140.500 .0 7.9 -6.3 130.719 3.5 4.7	20	85 4.1 160.200 .1 8.1 4.9 137.719 3.2 4.6	90 4.7 167.200 .2 8.2 4.4.903 .2 5.3 1	20 -6.1 -178.200 .3 8.3 -4.0 144.620 3.2 5.7 -	10 -7.8 -162.800 .4 8.4 -3.6 148.621 3.3 6.5	15 -8.7 -154.701 .5 8.3 -3.5 150.321 3.3 6.7	>>         -1         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0         -0<	<u>30 -11.9 -127.201 .9 7.2 -3.3 154.422 3.6 6.7</u>	35 -13.0 -116.501 1.1 0.2 -3.3 120.422 3.1 0.2	45 -14.9 -92.501 1.2 2.8 -3.4 156.723 3.9 5.7	50 - 15.5 - 79.8 - 01 1.2 .4 - 5.4 .01.6 - 5.3 4.0 4.6		[65] = 15.9 = 48.9 = .01  .7  3.0  -5.4  17.0  -2.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4  3.4	175 -15.0 -45.0010000000000			
-	-					-	_	_		_	_	-	_	_	_		_									_				
---------------------------------------------	--------------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------	--------------------------------------------	---------------------------------------------------------------------------------------------	-----------------------------------------------	--------------------------------------------	---------------------------------------------------------------------------------------	-------------------------------------------------	-----------------------------------------------	---------------------------------------------	-------------------------------------------------------------------------------	--------------------------------------------	------------------------------------------------	--------------------------------------------------------------------------------------------------------	-------------------------------------------------	-----------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------	-----------------------------------------------------	-------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------	-----------------------------------------------	----------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------
ZH				HUM													<b>~</b> .							~		LF	ΖY	L 4	25	MHz
425 M		. 1		A MAXI R PHAS DEG	****	50.4	5 5 2 2 2		16.0	# 9	₹.8. 8.65	20.00	R	5.5		5	55	-	2	~ ~	; <u>-</u>	5.5	2 2	2	5	10	ŝ	200	ŝ	
NCY		ZONTAI 3 DEG	ENNA	AXIMU AINVA	****	2.2	2.5 2.5		2.0	6. -	-9-1	ς Π	<u> </u>		22	-	5°-		:	6.0	5 0. 7	2.0		0.0			~.	- 2	2,0	
QUE	ENNA	T WAVE = HOR] = 80.8	D ANTE	EAN P	****	26	88	ŝĸ	ស	1.2			នុះ	នុខ		2		:œ	11-	<u>۽</u> ۽	: : :	1 57	2 ¥2	₽: ``	≥ °	2	- 10 - 10	2 2 2 2 2 2	€ 8	₹ ₹ ₹
8	LE ANT	NCIDEN DLAR. HETA	LARIZE	ASE NTER E	* **	8.2			н. 6	m. 	5 CL	0-	- 27 - - 27 -		· · ·	9.6	00	10	1.6	- <del>-</del>	17	8.0 8.0				~~ 	بې م	7.8 1.8	0 v	• • سم • • • •
R N F	ION IG 1	426	AL POI	£93	***	ې و	η η	ሻ	φ 	Ξρά = (	ρ <sub>Γ</sub> . νm	γ γ γ	ំភ្នំ៖ ស	٩.٩ - ٩	ስሳ ኦኮ	: آ <sup>ت</sup> روب	יי היח		~ ~	a v o o	i no i no		- ec	- i	⊇⊊ ຄ	2 5	5	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 3 5 2 2 3 5 2 3 5 5 2 5 5 5 5	eis mi	-9
T T E	METER	T M M	RADI	BENI	Ĩ	ຈຸ່	Ŧ	ŕ	ý	∽ 	ťΎ	<b>ทำ</b>	17	<del>.</del>	/ /	<u> </u>	: i								۰. -	1	റ്- 	ተጥ	ኆ :	
ΡA	A 0.1	ELD PO 1 = .1		SEVAR			<u>ر</u>	<b>.</b>	~	÷.	<b>n</b> con	- ^	-	~ 0	ייח ה		- م	- @	ŝ	- 0	2	m=	, 0	~ 0	יי ט	- L	ۍ د د	n		0 <b>~</b> 0
ION	ALONG		ANA	AR PHA	0	n,	ഗ്ര	°õ	2	ź;	<u>ំង</u>	z z	N,	é :	గజ్	8	5.5	ຸ່ື	υ. Έ	ຮູ່ຂ	ŝ	ŝ	68	6	28	66	ις Έ	Ϋ́Ε	ដះ	<u>n</u> oc
IAT	FIELD	3.8	ANTEI	MAXIM Gainvi DB	0.0	\$		2	2.7	n.	າ <del>ສ</del> ເທ		10.7	<u> </u>	<u>.</u>	-	- 0	6.00	9.0	- m	<u>.</u>			N.		4		1 00 1 00 1 00	5.0	, 
RAD	F THE	STLINDE	ARIZEI	MEAN Error DB	. 12	₽.	<u>5</u> 5	: ₽	₽:	÷	2 2 2 2	28	8	28	88 7	5.0	% 	Ē	3.	- 3	. %	ຕຸເ	9 K	°≌:	R ⊆ 1 1	ନ୍ଦୁ 1	88	5≓	£.8	រុស្ភ
A L	HECK O	. SYM. C AL LEN METER	AL POL	HASE Enter Eg	32.6	, R	5.5	, e. 8	26.0		11.0	200	08.6	2. S	3.9	5.0	29.9 69.6	67.3	69.7	74.7	6.69	59.7	32.6	15.4	20.02	63.7	51.5	35.4	2.0	26.1
υTΗ	UTY C	1: ROT AXI DIM	LINOZIS	VIER C	1- 9-1	5.1			٦ ج		5			۳ ۳	11	5.2 -1				  	17			5.6 -1			0.1	- ~	<u>ب</u>	
H I Z	MOGENE	100ELS	1011	898	0	5	0.0	10	<u>ارد</u>	2 4			9 10	7 ÷	17	5	77	- 	5	77 2 <u>9</u>		<u>5</u>	ະ ກ ເ	8.4		<u>ک</u>		0.91	ະ ກ	1 1 1 1 2 E S
<	Ξ	¥		E S																= =	:=	= :				~	<u> </u>		2:	
1																														
N			<u> </u>	_ @	Τ							_							_	_			-							
2 H H S		·		AXTHUM TIASEVAR EG	5.4	5.0	5.0	5.6					8.0		1.1	ين ه و	5.5 5.5	6.7	15.1	2	6.1		1.7			0.1	0.0	5.0	с. 2	5.5 2.5
C Y 425 M H Z		EG.	A	IMUM MAXTMUM NVAR MIASEVAR DEG	.9 65.4	.9 65.0	.9 63.9 62.1	9 59.5	.8 56.3	C.7C 81	5 43.8	25.00	8.08	-7 -27.4 6 -25 -3	.3 24.7	.8 25.4	24.9	4 16.7	.1 12.1	2 2 2	.1 1.9		.5 1.7	- 8.4		.5 10.1	.7 10.9	.1 12.0	.3 12.3	5 12.6 12.6
UENCY 425 MHZ	NA	AAVE VERTICAL 80.8 DEG	ANTENNA	N MAXIMUM MAXIMUM Or cainvar miasevar DB dG	97 5.9 65.4	96 5.9 65.0	93 5.9 63.9 88 5.9 63.1	82 5.9 59.5	75 5.8 56.3	C.7C 8.C 85	19 5.5 43.8		26 4.9 30.8	21 4.7 27.4	16 4.3 21.7	16 3.8 25.4	17 2.9 24.9	15 1.4 16.7	14 1.1 12.1	13 1.0 8.2 13 1.0 1.8.2	1.1 1.9	14 1.2 .6	15 1.5 1.7	16 1.8 6.4	18 2.2 9.1	19 2.5 10.1	20 2.7 10.9	22 3.1 12.0	22 3.3 12.3	23 3.5 12.6 3.5 12.6 23 3.5 12.7
R EQUENCY 425 M H Z	ANTENNA	IDENT WAVE AR. = VERTICAL TA = 80.8 DEG	RIZED ANTENNA	E MEAN MAXIMUM MAXIMUM Er Error Cainvar Masevar DB DB DB reg	497 5.9 65.4	096 5.9 65.0	0 93 5.9 63.9	B - 82 5.9 59.5	8 75 5.8 56.3			7	7 - 26 4.9 30.8	321 4.7 27.4	8 16 4.3 24.7	916 3.8 25.4	1 - 17 2.9 24.9	3 15 1.4 16.7	0 14 1.1 12.1		1.1 1.9	- 14 1.2 .6	9 15 1.5 1.7	2 - 16 1.8 5.4	3 18 2.2 9.1	2.5 10.1	320 2.7 10.9	122 3.1 12.0	422 3.3 12.3	
N FREQUENCY 425 MHZ	DIPOLE ANTENNA	INCIDENT WAVE POLAR. = VENTICAL THETA = 80.8 DEG	, POLARIZED ANTENNA	PHASE MEAN MAXIMUM MAXIMUM CENTER ERROR GAINVAR MIASEVAR DEG DB DB CB	-97.497 5.9 65.4	-97.096 5.9 65.0	-95.993 5.9 63.9 -94.288 5.9 62.1	-91.882 5.9 59.5	-88.875 5.8 56.3		-78.2 49 5.5 43.8	-71.7 33 5.1 34.8	-69.7 26 4.9 30.8	-09.321 4.7 27.4	-77.8 16 4.3 24.7	-88.916 3.8 25.4	-103.017 2.9 24.9	-123.3 15 1.4 16.7	-127.014 1.1 12.1	-127.3 -13 1.0 8.2	-125.513 1.1 1.9	-123.214 1.2 .6	-117.9 15 1.5 4.7	-115.216 1.8 6.4	-110.3 18 2.2 9.1	-108.2 19 2.5 10.1	-106.320 2.7 10.9	-103.422 3.1 12.0	-102.422 3.3 12.3	-101.223 3.4 12.5 -101.223 3.5 12.6 -101.123 3.5 12.7
TERN FREQUENCY 425 MHZ	IETER DIPOLE ANTENNA	T INCIDENT WAVE M POLAR. = VENTICAL M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	GATN PHASE MEAN MAXIMUM MAXIMUM CENTER CENTER ERROR CAINVAR MASEVAR DB DEG DB DB DG	-15.6 -97.497 5.9 65.4	-15.6 -97.096 5.9 65.0	-15.7 -95.993 5.9 63.9 -15.8 -04.288 5.0 62.1	-15.9 -91.882 5.9 59.5	-16.0 -88.875 5.8 56.3		-16.9 -78.2 - 49 5.5 43.8	-17.9 -71.733 5.1 34.8	<b>-18.6</b> -69.726 4.9 30.8	-19.5 -09.3 - 21 4.7 27.4 -20 5 -71 5 - 17 1 6 25 3	-21.5 -77.8 16 4.3 24.7	-22.2 -88.916 3.8 25.4	-20.6 -115.4 - 17 2.9 24.9	-18.8 -123.3 - 15 1.4 16.7	-17.0 -127.014 1.1 12.1	-13.8 -120.0 - 13 1.0 8.2 -13.8 -127.3 - 13 1.0 8.8	-12.6 -125.5 -, 13 1.1 1.9	-11.5 -123.214 1.2 .6	-9.8 -117.9 15 1.5 1.7	-9.2 -115.216 1.8 6.4 87 -115 7 -15 1.6 7.0	-8.3 -110.3 18 2.2 9.1	-8.0 -108.2 19 2.5 10.1	-7.7 -106.320 2.7 10.9	-7.4 -103.422 3.1 12.0	-7.3 -102.422 3.3 12.3	-7.2 -101.023 3.5 12.6 -7.2 -101.123 3.5 12.6 -7.1 -101.123 3.5 12.7
PATTERN FREQUENCY 425 MHZ	0.1 METER DIPOLE ANTENNA	D POINT INCIDENT WAVE = .10 M POLAR. = VENTICAL = 1.00 M THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	UM GATIN PHASE MEAN MAXIMUM MAXIMUM VAR CENTER ERROR GATINVAR FINSFVAR DB DEC DB DB CEC	-15.6 -97.497 5.9 65.4	-15.6 -97.096 5.9 65.0	-15.7 -95.993 5.9 63.9 -15.8 -94.288 5.0 62.1	-15.9 -91.882 5.9 59.5	-16.0 -88.875 5.8 56.3		-16.9 -78.2 - 49 5.5 43.8	-17.9 -71.7 33 5.1 34.8	-18.6 -69.7 26 4.9 30.8		21.5 -77.816 4.3 24.7	-22.2 -88.916 3.8 25.4	20.6 -115.4 - 17 2.9 24.9	-18.8 -123.3 15 1.4 16.7	-17.0 -127.014 1.1 12.1		-12.6 -125.5 13 1.1 1.9	-11.5 -123.214 1.2 .6	-9.8 -117.9 - 15 1.5 <sup>1</sup> .7	-9.2 -115.216 1.8 6.4 8.7 -115.716 1.8 6.4		-8.0 -108.2 19 2.5 10.1	-7.7 -106.320 2.7 10.9	-7.4 -103.422 3.1 12.0	7.3 -102.422 3.3 12.3	-7.2 -101.023 3.4 12.6 -7.2 -101.223 3.5 12.6 -7.1 -101.123 3.5 12.7
ON PATTERN FREQUENCY 425 MHZ	LONG A 0.1 METER DIFOLE ANTENNA	FIELD FOINT INCIDENT WAVE DAT = . 10 H FOLLR. = VERTICAL HB =1.00 H THETA = 80.8 DEG	RADIAL POLARIZED ANTENNA	HAZEMUM GATN PHASE MEAN MAXIMUM MAXIMUM Phasevar center error Galivvar Masevar deg db deg db db reg	8.0 -15.6 -97.497 5.9 65.4	8.0 -15.6 -97.096 5.9 65.0	8.0   -15.7 -95.9 - 93 5.9 63.9   8.0   -15.8 -94.2 - 88 5.9 62.1	7.9 -15.9 -91.882 5.9 59.5	7.9 -16.0 -88.875 5.8 56.3		7.7 -16.9 -78.2 - 49 5.5 43.8	7.7	7.7 -18.6 -69.726 4.9 30.8		8.0 -21.5 -77.8 -16 4.3 24.7	8.2 -22.2 -88.916 3.8 25.4	8.7   -20.6 -115.4 - 17 2.9 24.9	9.0 -18.8 -123.3 -15 1.4 16.7	9.4 -17.0 -127.014 1.1 12.1	10.1 1.2.8 1.2.1.2.1.2.1.2.1.2.1.2.1.2.1.2.1.2.1.2	10.6 -12.6 -125.5 -, 13 1.1 1.9	11.011.5 -123.2 14 1.2 .6	12.2 1 -9.8 -117.915 1.5 1.7	13.0 -9.2 -115.216 1.8 6.4		16.8 -8.0 -108.2 19 2.5 10.1	18.27.7 -106.320 2.7 10.9	18.6 -7.4 -103.422 3.1 12.0	17.4 -7.3 -102.422 3.3 12.3	15.0 -7.1 -101.0 -7.3 3.4 12.5 15.0 -7.2 -101.2 -23 3.5 12.6 14.6 -7.1 -101.1 -23 3.5 12.7
ATION PATTERN FREQUENCY 425 MHZ	IELD ALONG A 0.1 METER DIPOLE ANTENNA	FIELD POINT INCIDENT WAVE BO M DAT = .10 M POLAR. = VERTICAL .25 M HD =1.00 M THETA = 80.8 DEG	TENNA RADIAL POLARIZED ANTENNA	AZTIKUM MAZTHUM GATN FHASE HEAN MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM Alivvar Phasevar Center Center Error Galivar Thasevar B deg db db db db db teg	.0 8.0 -15.6 -97.497 5.9 65.4	-0 8.0 -15.6 -97.096 5.9 65.0	.0 8.0  -15.7 -95.993 5.9 63.9	.0 7.9 -15.9 -91.882 5.9 59.5	.0 7.9 -16.0 -88.875 5.8 56.3		· 1 7.7 -16.9 -78.249 5.5 43.8	1 7.7 - 17.9 - 71.7 - 33 5.1 34.8	.1 7.7 -18.6 -69.7 26 4.9 30.8	2 7 0	.2 8.0 21.5 -77.8 -16 4.3 24.7	-3 8.2 -22.2 -88.9 -16 3.8 25.4	.3 8.7 20.6 -115.4 - 17 2.9 24.9	.3 9.0 -18.8 -123.3 - 15 1.4 16.7	.3 9.4 -17.0 -127.014 1.1 12.1	3 10.1 -13.8 -125.0 -13 1.0 8.2 -	.3 10.6 -12.6 -125.5 -13 1.1 1.9	.3 11.0   -11.5 -123.214 1.2 .6	.3 12.2 1 -9.8 -117.9 15 1.5 1.7	· 3 13.0   -9.2 -115.216 1.8 6.4	.2 15.3 -8.3 -110.3 -18 2.2 9.1	.0 16.8 -8.0 -108.2 19 2.5 10.1	.3 18.2 -7.7 -106.320 2.7 10.9	1.1 18.6 -7.4 -103.422 3.1 12.0	1.4 17.4 -7.3 -102.422 3.3 12.3	1.7 15.0 -7.2 -101.0 -2.3 5.4 16.5 1.7 15.0 -7.1 -101.1 -23 3.5 12.6 1.7 14.6 1 -7.1 -101.1 -23 3.5 12.7
ADIATION PATTERN FREQUENCY 425 HHZ	THE FIELD ALONG A 0.1 METER DIFOLE ANTENNA	LINDER FIELD POINT INCIDENT WAVE TH = 1.00 M DAT = . 10 M POLAR. = VENTICAL = .25 M HB =1.00 M THETA = 80.8 DEG	ZED ANTENNA RADIAL POLARIZED ANTENNA	EAN MAXTMUM MAXIMUM MAXI 3 DB CALIMA E CENTER ERROR CALIWAR MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXI 3 DB CALIMA E CENTER E CENTER ERROR CALIWAR MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXI	.00 .0 8.0 -15.6 -97.497 5.9 65. <sup>4</sup>	.00 .0 8.0 -15.6 -97.096 5.9 65.0	.00 .0 8.0   -15.8 -95.9 - 93 5.9 63.9	.00 .0 7.9 -15.9 -91.882 5.9 59.5	.00 .0 7.9 -16.0 -88.875 5.8 56.3			.00 .1 7.7   -17.9 -71.7 33 5.1 34.8	.00 .1 7.7 -18.6 -69.726 4.9 30.8		.00 .2 8.0 21.5 -77.8 -16 4.3 24.7	.00 .3 8.2 -22.2 -88.9 -16 3.8 25.4	.00 .3 8.7   -20.6 -115.4 - 17 2.9 24.9	.00 .3 9.0 -18.8 -123.3 -15 1.4 16.7	.00 .3 9.4 -17.0 -127.014 1.1 12.1		.01 .3 10.6 -12.6 -125.5 -13 1.1 1.9	.01 .3 11.0 -11.5 -123.214 1.2 .6	.01 .3 12.2   -9.8 -117.9 - 15 1.5 1.7	.01 .3 13.0 -9.2 -115.2 -16 1.8 6.4		01 .0 16.8 -8.0 -108.219 2.5 10.1	03 .3 18.2 -7.7 -106.320 2.7 10.9	-06 1.1 18.6 -7.4 -103.422 3.1 12.0	07 1.4 17.4 [ -7.3 -102.422 3.3 12.3 ]	- 08 1.7 15.0 - 7.2 - 101.0 - 2.3 3.4 12.5 - 08 1.7 14.6 - 7.1 - 101.1 - 2.3 3.5 12.6 - 08 1.7 14.6 - 7.1 - 101.1 - 2.3 3.5 12.7
AL RADIATION PATTERN FREQUENCY 425 MHZ	ECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	SIN-CILINDER FIELD FOINT INCIDENT WAVE L. LENCTH = 1.80 M DML, = VERTICAL STER = .25 M HB =1.00 M THETA = 80.8 DEG	OLARIZED ANTENNA RADEAL FOLARIZED ANTENNA	ASE MEAN MAXIMUM MAXIMUM GAIN PHASE MEAN MAXIMUM MAXIMUM MAXIMUM VIER ERROR GAINVAR PHASEVAR CENTER ERROR GAINVAR FIASEVAR 26 08 08 16G	7.1 .00 .0 8.0 -15.6 -97.497 5.9 65.4	7.5 .00 .0 8.0   -15.6 -97.096 5.9 65.0	5.5 .00 .0 5.0 -15.7 -95.9 -, 93 5.9 63.9 1 3.2 .00 .0 8.0 -15.8 -94.2 - 88 5.9 62.1	2.6 .00 .0 7.9 -15.9 -91.882 5.9 59.5	5.7 .00 .0 7.9 -16.0 -88.875 5.8 56.3	C.22 8.2 00 - C.20 2.01 0.1 0. 0. 8.6 C.01 2.2 2.2 0.10 3.5 0.1 0.1 0. 0. 0. 8.6	8.8 .00 .1 <u>7.7</u> -16.9 -78.2 -149 5.5 43.8	1.5 .00 .1 7.7 -17.9 -71.733 5.1 34.8	7.3 .00 .1 7.7 -18.6 -69.726 4.9 30.8	1,2 ,W ,Z /,0  = 9,509,3,21 4,7 2/,4   2,3 M 2 7 0  _20 E_71 E_17 1 K 25 2	24 .00 .2 8.0 21.5 -77.8 -16 4.3 24.7	9.0 .00 .3 8.2 -22.2 -88.9 - 16 3.8 25.4	7.2 .00 .3 8.7   -21.9 -103.0 -17 2.9 24.9	5.7 .00 .3 9.0 -18.8 -123.3 -15 1.4 16.7	5.3 .00 .3 9.4 -17.0 -127.014 1.1 12.1	5.8 .00 .3 10.1 -13.8 -127.3 -13 1.0 8.2 1 5.8 .00 .3 10.1 -13.8 -127.3 -13 1.0 1.8	1.4 .01 .3 10.6 -12.6 -125.5 -13 1.1 1.9	4.7 .01 .3 11.0 -11.5 -123.2 -14 1.2 .6	5.0 .01 .3 12.2   -9.8 -117.9 -15 1.5 <sup>1</sup> .7	1,9 $0,01$ $3$ $13,0$ $-9.2$ $-115.2$ $-16$ $1.8$ $6.4$ $1.2$ $0.3$ $11.1$ $1.2$ $0.3$ $11.1$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$ $1.3$	2.800 .2 15.3 -8.3 -110.318 2.2 9.1	3.001 .0 16.8 -8.0 -108.219 2.5 10.1	5.703 .3 18.2 -7.7 -106.320 2.7 10.9	1.806 1.1 18.6 -7.4 -103.422 3.1 12.0	1.507 1.4 17.4 1-7.3 -102.422 3.3 12.3	5.708 1.7 15.0 -7.2 -101.223 3.5 12.6 5.708 1.7 15.0 -7.2 -101.223 3.5 12.6 1.708 1.7 14.6 -7.1 -101.123 3.5 12.7
THAL RADIATION PATTERN FREQUENCY 425 MHZ	TY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	ROT.SIM.CILINDER FIELD FOINT INCIDENT WAVE AXIAL LENCTH = 1.80 M DAT = .10 M FOLAR = VERTICAL DIAMETER = .25 M HB =1.00 M THETA = 80.8 DEG	ICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	PHASE MEAN MAXIMUM MAXIMUM CAIN PHASE MEAN MAXIMUM MAXIM ER CENTER ERROR CAINVAR PHASEVAR CENTER ERROR CAINVAR MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MA DEC DB DB DB DEC DB DEC DB DB DEC DB DB DC DC DB DB DAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIMUM MAXIM	6 47.1 .00 .0 8.0 -15.6 -97.497 5.9 65.4	5 47.5 .00 .0 8.0 -15.6 -97.096 5.9 65.0	2 48.2 .00 .0 8.0 -15.7 -95.9 -93 5.9 63.9 1 4 50.2 .00 .0 8.0 -15.8 -94.2 - 88 5.9 62.1	3 52.6 .00 .0 7.9 -15.9 -91.882 5.9 59.5	2 55.7 .00 .0 7.9 -16.0 -88.875 5.8 56.3	C.25 8.5 80 C.55- 2.01- 0.1 0. W. P.65 1		1 80.5 .00 .1 7.7 -17.9 -71.733 5.1 34.8	7 87.3 .00 .1 7.7 -18.6 -69.7 - 26 4.9 30.8	1 1/0.3 M 2 7 0 1.9/1.7 1.4/1.7 2/1.4	5 110.4 .00 .2 8.0 21.5 -77.8 -16 4.3 24.7	0 119.0 .00 .3 8.2 -22.2 -88.9 -16 3.8 25.4	7 126.0 . W . 3 8.7   -21.9 -103.0 17 2.9 24.9 ]	7 146.7 .00 .3 9.0 -18.8 -123.3 -15 1.4 16.7	4 156.3 .00 .3 9.4 -17.0 -127.014 1.1 12.1	9 175.8 0.0 5 10.1 1.25.4 1.25.0 1.0 5.2 1 1 10 5.2 1 10 5.2 1 10 10 5.2 1 10 10 5.2 1 10 10 5.2 10 10 5.2 10 10 10 5.2 10 10 10 10 10 10 10 10 10 10 10 10 10	7 -174.4 .01 .3 10.6 -12.6 -125.5 -13 1.1 1.9	6 - 164.7 .01 .3 11.0   -11.5 - 123.214 1.2 .6	7 -145.0 .01 .3 12.2 1 -9.8 -117.9 - 15 1.5 1.7	9 -134.9 .01 .3 13.0 -9.2 -115.216 1.8 6.4	7 -112.800 .2 15.3 -8.3 -110.318 2.2 9.1	1 -100.001 .0 16.8 -8.0 -108.219 2.5 10.1	5 -85.703 .3 18.2 -7.7 -106.320 2.7 10.9	2 -54.800 1.1 18.6 -7.4 -103.422 3.1 12.0	4 -41.507 1.4 17.4 -7.3 -102.422 3.3 12.3	2 -21.2 -1.0 1.7 15.0 -7.2 -101.2 -23 3.5 12.6 2 -23.7 -00 1.7 14.6 -7.1 -101.1 -23 3.5 12.6
IHUTHAL RADIATION PATTERN FREQUENCY 425 MHZ	DERIGITY CHECK OF THE FIELD ALONG A 0.1 METER DIPOLE ANTENNA	IBODY: ROT.SYM.CH.LINDER FIELD POINT INCIDENT WAVE AXALL LENGTH = 1.80 M DAT = . 10 M POLAR. = VERTICAL DIAMETER = . 25 M HB = 1.00 M THETA = 80.8 DEG	VERTICAL POLARIZED ANTENNA RADIAL POLARIZED ANTENNA	GAIN PHASE MEAN MAXIMUM MAXIMIM MAXIMUM MAXIMIX M CENTER CENTER ERROR GALIVAR MAXIMUM MAXI DB DEG DB DB DB DEG DB DB DEG DB DB DEG DB DB MAXIMUM M DAMITAN MAXIMUM MA	2.6 47.1 .00 .0 8.0 -15.6 -97.497 5.9 65.4	2.5 47.5 .00 .0 8.0 -15.6 -97.096 5.9 65.0	2.4 50.2 .00 .0 8.0 -15.7 -95.993 5.9 63.9	2.3 52.6 .00 .0 7.9 -15.9 -91.8 - 82 5.9 59.5	2.2 55.7 .00 .0 7.9 -16.0 -88.875 5.8 56.3	C 22 C 20 - C 20 - 2 10 - 0 1 0 1 0 1 0 1 12 - 12 12 12 12 12 12 12 12 12 12 12 12 12		1.1 80.5 .00 .1 7.7 -17.9 -17.7 - 33 5.1 34.8	.7 87.3 .00 .1 7.7 -18.6 -69.726 4.9 30.8		5 110.4 .00 .2 8.0 21.5 -77.8 -16 4.3 24.7	-1.0 119.0 .00 .3 8.2 -22.2 -88.9 -16 3.8 25.4	-1.2 126.0 . W .3 8.7 -20.4 -103.0 -17 2.9 24.9 1 -2.1 137.2 . 00 .3 8.7 -20.6 -115.4 - 17 2.0 21.5	-2.7 146.7 .00 .3 9.0 -18.8 -123.3 -15 1.4 16.7	-3.4 156.3 .00 .3 9.4 -17.0 -127.014 1.1 12.1		-5.7 -174.4 .01 .3 10.6 -12.6 -125.5 -13 1.1 1.9	-5.6 - 164.7 .01 .3 11.0 $- 11.5 - 123.2$ 14 1.2 .6	-8.7 -145.0 .01 .3 12.2   -9.8 -117.9 -15 1.5 1.7	$\begin{vmatrix} -9.9 & -134.9 & 01 & .3 & 13.0 \\ -112 & -124.2 & 00 & .3 & 11.1 \\ -112 & -124.2 & 00 & .3 & 11.1 \\ \hline -112 & -124.2 & 00 & .3 & 11.1 \\ \hline -112 & -124.2 & 00 & .3 & 11.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 00 & .3 & 0.1 \\ \hline -112 & -124.2 & 0.1 & 0.1 \\ \hline -112 & -124.2 & 0.1 \\$	-112.7 -112.800 .2 15.3 -8.3 -110.318 2.2 9.1	-14.1 -100.001 .0 16.8 -8.0 -108.219 2.5 10.1	-15.5 -85.703 .3 18.2 -7.7 -106.320 2.7 10.9		-17.4 $-41.5$ $-07$ $1.4$ $17.4$ $-7.3$ $-102.4$ $22$ $3.3$ $12.3$	-17.2 -25.7 -30 1.7 15.0 -7.2 -101.2 -23 3.5 12.6 -17.2 -23.7 -30 1.7 15.0 -7.1 -101.1 -23 3.5 12.6 -17.2 -23.7 -30 1.7 14.6 -7.1 -101.1 -23 3.5 12.7

16.2.5. ADDITIONAL RESULTS FROM FIELD COMPUTATIONS WITH FZYL, MANMOD 1 AND MANMOD 2





FIGURE 78d Field components  $E_v$  and  $E_r$  versus  $d_{at}$  at  $\phi = 0$  and  $180^{\circ}$ , with the parameter f 20 to 700 MHz. Constant:  $p_2 = vertical$ ,  $\theta_i = 80.8^{\circ}$ ,  $h_B = 1.0$  m.



FIGURE 77d Field components  $E_v$ ,  $E_r$  and  $E_h$  versus  $\phi$  with the parameter f 20 to 700 MHz. Constant:  $d_{at}$  = 0.1m, p<sub>2</sub> = vertical,  $\theta_i$  = 80.8°,  $h_B$  = 1.0 m.

# Leer - Vide - Empty

## AZIMUTHAL RADIATION PATTERNS IN THE 11 MHz RANGE

- FZYL : 11 MHz
- MANMOD1 : 15 MHz
- MANMOD 2 : 15 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_{V}$ :  $E_r$  and  $E_h$  at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4 m.

- $E_V$  varies only within 2 dB
- Er varies extremely, especially at small dat
- E<sub>h</sub> varies only within 2 dB



FIGURE 100a Azimuthal radiation pattern FZYL at 11 MHz.



FIGURE 100b Azimuthal radiation pattern MANMOD1 at 15 MHz.



FIGURE 100c Azimuthal radiation pattern MANMOD 2 at 15 MHz.

## AZIMUTHAL RADIATION PATTERNS IN THE 50 MHz RANGE

- FZYL : 50 MHz
- MANMOD 1 : 50 MHz
- MANMOD 2 : 50 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_v$ ,  $E_r$  and  $E_h$  at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4m.

- E<sub>V</sub> varies only within 2.5 dB
- Er varies only within 2.5 dB
- E<sub>h</sub> varies only within 2.0 dB



FIGURE 101a Azimuthal radiation patterns FZYL at 50 MHz.



FIGURE 101b Azimuthal radiation patterns MANMOD1 at 50 MHz.



FIGURE 101c Azimuthal radiation patterns MANMOD 2 at 50 MHz.

## AZIMUTHAL RADIATION PATTERNS IN THE 75 MHz RANGE

- FZYL : 75 MHz
- MANMOD 1 : 75 MHz
- MANMOD 2 : 80 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_v$ ,  $E_r$  and  $E_h$  at dat = 0.1, 0.2, 0.3 and 0.4 m.

- $E_{\rm V}$  varies extremely between FZYL 75 MHz and MANMOD1 75 MHz, but  $E_{\rm V}$  varies less (about 10 dB) between FZYL 75 MHz and MANMOD2 80 MHz.
- $E_{r}$  is almost constant versus  $\varphi$  and the amplitude varies within 5 dB between the three bodies at constant  $d_{at}.$
- Eh varies only within 1 dB



FIGURE 102a Azimuthal radiation patterns FZYL at 75 MHz.



FIGURE 102b Azimuthal radiation patterns MANMOD1 at 75 MHz.



FIGURE 102c Azimuthal radiation patterns MANMOD 2 at 80 MHz.

## AZIMUTHAL RADIATION PATTERNS IN THE 85 MHz RANGE

- FZYL : 85 MHz
- MANMOD 1 : 90 MHz
- MANMOD 2 : 90 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_v$ ,  $E_r$  and  $E_h$  at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4 m.

- $E_V$  varies very much between FZYL 85 MHz and MANMOD1 90 MHz (12 dB), but  $E_V$  varies less between FZYL 85 MHz and MANMOD2 90 MHz (3 dB)
- $E_{\textbf{r}}$  is almost independent on  $\varphi$  and differs in amplitude within 5 dB between the three bodies
- Eh varies only within 2 dB



FIGURE 103a Azimuthal radiation patterns FZYL at 85 MHz.



FIGURE 103b Azimuthal radiation patterns MANMOD1 at 90 MHz.



FIGURE 103c Azimuthal radiation patterns MANMOD 2 at 90 MHz.

- 301 -

## AZIMUTHAL RADIATION PATTERNS IN THE 200 MHz RANGE

- FZYL : 200 MHz
- MANMOD1 : 200 MHz
- MANMOD 2 : 200 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_v$ ,  $E_r$  and  $E_h$  at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4 m.

- Ev varies within 5 dB
- $E_r$  is almost independent on  $\phi$  up to  $d_{at} = 0.3m$  and varies only 3 dB between the three bodies. At  $d_{at} = 0.4$   $E_r$  becomes dependent on  $\phi$  and varies within 5 dB between the three bodies.
- E<sub>h</sub> varies within 10 dB and develops two peaks, especially with MANMOD1



FIGURE 104a Azimuthal radiation patterns FZYL at 200 MHz.



FIGURE 104b Azimuthal radiation patterns MANMOD1 at 200 MHz.



FIGURE 104c Azimuthal radiation patterns MANMOD 2 at 200 MHz.

## AZIMUTHAL RADIATION PATTERNS IN THE 800 MHz RANGE

- FZYL : 800 MHz
- MANMOD 1 : 800 MHz
- MANMOD 2 : 800 MHz

The three figures show the azimuthal radiation patterns of the field components  $E_v$ ,  $E_r$  and  $E_h$  at  $d_{at}$  = 0.1, 0.2, 0.3 and 0.4 m.

The accuracy of the results are very doubtful, because the computer program limitations have been exceeded at this high frequency (see section 10.3.4.).

- $E_v$  varies within 6 dB
- Er varies within 16 dB
- E<sub>h</sub> changes very much in wave form and amplitude



FIGURE 105a Azimuthal radiation patterns F7YL at 800 MHz.



FIGURE 105b Azimuthal radiation patterns MANMOD1 at 800 MHz.



FIGURE 105c Azimuthal radiation patterns MANMOD 2 at 800 MHz.

## LEBENSLAUF

- 28. 7. 1943 Geboren in Baden (AG)
- 1950 1955 Primarschule in Wettingen (AG)
- 1955 1958 Bezirksschule in Baden und Wettingen
- 1958 1959 Oberrealsschule der Kantonsschule Zürich Stadt
- 1959 1963 Lehre als Maschinen-Schlosser bei Brown, Boverie & Cie
- 1963 1966 Teilzeit-Arbeit als FEAM bei Brown, Boverie & Cie und zweiter Bildungsweg bei der Akademikergemeinschaft in Zürich,

Eidgenössische Matura, Typus C

- 1966 1970 Studium der Elektrotechnik an der Abt. III B der ETH Zürich. Hilfs-Assistent am Lehrstuhl für Apparatebau der Elektrotechnik der ETH Zürich, Industrietätigkeit. Abschluss als Dipl. El. Ing. ETH
- seit 1971 Leiter der Gruppe Messtechnik am Laboratorium für Biomechanik der ETH Zürich
- 1974 Organisator und Chairman des Second International Symposium on Biotelemetry, Davos, May 20-24, 1974
- 1974 1976 Vice-president der International Society on Biotelemetry und seither member of the scientific council
- seit 1978 Beirat der Schweizerischen Gesellschaft für Biomedizinische Technik, Mitglied des Schweizerischen Fachkommission FK-62 der IEC-SEV, Experte des Internationalen TC-62 der IEC (Probleme der Biotelemetrie)